DRAFT SUPPLEMENTAL ENVIRONMENTAL IMPACT REPORT 597 SECOND AMENDMENT TO THE 2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN

VOLUME 1

Prepared for:

Orange County Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, CA 92703

> Contact: Rochelle Carpenter (714) 834-4060

> > Prepared by:

BonTerra Consulting 151 Kalmus Drive, Suite E-200 Costa Mesa, CA 92626

Contact: Joan Patronite Kelly, AICP (714) 444-9199





TABLE OF CONTENTS

Section

Page

Acronym	าร			vii
Glossary	/			XII
Executiv	e Summ	ary	t	:S-1
1.0	Introduc	tion		.1-1
	1.1	Purpose of	the Draft Supplemental EIR	.1-1
		1.1.1 A	uthority	
		1.1.2 P	Preparation of a Supplemental EIR	.1-1
		1.1.3 Ir	ncorporation by Reference	.1-2
		1.1.4 Ir	ntended Uses of the Draft Supplemental EIR	.1-2
		1.1.5 A	Agencies Having Jurisdiction	.1-3
		1.1.6 A	vailability of the Supplemental EIR	.1-6
	1.2	Methodolog	gy and Approach	1-8
		1.2.1 E	xisting Environmental Setting	.1-8
		1.2.2 S	Significance Criteria	1-9
		1.2.3 E	nvironmental Impact Analysis	1-9
		1.2.4 C	Cumulative Impacts	1-9
		1.2.5 N	litigation Measures	
		1.2.6 L	evel of Significance After Mitigation	1-10
2.0	Existing	Conditions	History and Evolution of the 2001 General Development Plan.	.2-1
	21	Background	d and Evolution of the GDP	2-1
	22	Overview o	of the General Development Plan (Existing Conditions)	2-1
		221 C	biectives of the 2001 General Development Plan	2-2
		2.2.1 0	001 General Development Plan Elements: Project Site	
		2.2.2 Z	Description Existing Conditions	2-3
		223 A	Iternatives Analyzed Within FIRs 548 and 575	2-0
		2.2.3 A	Amoranda of Understanding (MOU) Between the County of	.2-5
		2.2.4	Viende and the Cities of San Clemente and San Juan	
			Sonistrano	2-11
		225 1	mandmant No. 1 to the 2001 Coneral Development Plan	2-11 0.44
		2.2.J A	Summary of Canaral Development Plan Environmental	2-11
		2.2.0 3	mooto	0 1 1
	• •	II Eviating On	npacis	2-14
	2.3		be 2001 CDD. Eviating DDL Dermite and Subacquart	2-30
		2.3.1	ne 2001 GDP, Existing PDL Permits, and Subsequent	0.05
		A	ACTIONS	2-35
		2.3.2 Ir	mplementation and Status of 2001 GDP Biological Mitigation	2-38
3.0	Purnose	and Need	for Amendment No. 2 to the 2001 General Development Plan	3-1
0.0	3 1	Physical Si	te Stahility	3-1
	3.2	Accommod	late Future Landfill-Related Features	3-2
	3.2 3.3	Re-Design	Desilting System to Reduce Biological Resources Impacts	3-2
	0.0	331 Δ	woid Least Bell's Vireo Territories	3-2
		333	Sustain the Downstroam Piparian Corridor	3-2
	2 /	5.5.2 5	and term Stewardship of On Site Biological Persources	22
	3.4		Olig-term Stewaluship of Oli-Sile Diological Resources	
		3.4.1 R	Relationship to Other Regional Flamming Enorts (NCCF, SAMF).	. 3-3
		3.4.2 D	vevelopment of a Pre-iviligation Program	.3-3
		3.4.3 D		<u> </u>
4.0			pportunities Plan	.3-3
4.0	The Pro	posed Proje	ЭСТ	
	4.1	Project Loc	cation	4-1
	4.2	Environmei	ntal Setting	

TABLE OF CONTENTS (Continued)

Section

<u>Page</u>

	4.3	Project Description		
		4.3.1	Proposed Project Elements	4-1
		4.3.2	Timing of the Proposed Project	4-9
		4.3.3	Regulatory Requirements for Implementation	4-10
		4.3.4	Relationship to Other Regional Planning Efforts	4-16
5.0	Enviror	nmental A	Analysis	5-1
	5.1	Introduc	stion	5-1
	5.2	Geophy	sical	5-1
	-	5.2.1	Existing Conditions	
		5.2.2	Significance Criteria	
		5.2.3	Potential Impacts	5-5
		5.2.4	Mitigation Measures	
		5.2.5	Level of Significance After Mitigation	5-8
	5.3	Hvdrolo	av and Water Quality.	
	0.0	5.3.1	Existing Conditions	5-8
		5.3.2	Significance Criteria	
		5.3.3	Potential Impacts	5-14
		534	Mitigation Measures	5-17
		535	Level of Significance After Mitigation	5-18
	54	Air Qua	lity	5-18
	0.1	541	Existing Conditions	5-19
		542	Significance Criteria	5-27
		543	Potential Impacts	5-27
		544	Mitigation Measures	5-28
		545	Level of Significance After Mitigation	5-38
	55	Biologic	al Resources	5-38
	0.0	5.5.1	Existing Conditions	5-38
		552	Significance Criteria	5-76
		553	Potential Impacts	5-77
		554	Mitigation Measures	5-87
		555	Level of Significance After mitigation	5-95
	56	L Itilities	and Service Systems	5-95
	0.0	561	Existing Conditions	5-95
		562	Significance Criteria	5-97
		563	Potential Impacts	5-97
		564	Mitigation Measures	5-98
		565	Level of Significance After Mitigation	5-100
6.0	Enviror	mental le	ssues Not Requiring Substantial Additional Analysis	6-1
0.0	61	Introduc	tion	
	6.2	Summa	ry of Issues Not Requiring Additional Analysis	0-1 6-1
	0.2	621	L and Lise and Planning	6-1
		622	Agriculture	0-1 6-1
		623	Population and Housing	0-1 6-2
		624	Transportation and Circulation	0-2 6-2
		625		2-2 ຄ_ງ
		626	Aesthetics	0-2 6.2
		627	Cultural and Scientific Resources	2 ⁻ 0 6_2
		628	Recreation	2-0 6-3
		620	Mineral Resources	د-ی ۲-۲
		0.2.3		0-3

TABLE OF CONTENTS (Continued)

Section

<u>Page</u>

		6.2.10	Hazards	6-3	
		6.2.11	Public Services	6-3	
7.0	Alternat	tives to the	e Proposed Project	7-1	
	7.1	Overview	1	7-1	
		7.1.1	Alternatives Considered but Not Evaluated in SEIR 597	7-1	
	7.2	Alternativ	ves to the Proposed Project for Amendment No. 2 to the GDP		
		Evaluate	d in SEIR 597	7-2	
		7.2.1	No Action Alternative	7-2	
		7.2.2	Alternative 1: Maintain 2001 GDP Zone 4 Footprint and		
			Detention/Desilting Basin Between Zones 1 and 4	7-4	
		7.2.3	Alternative 2: Maintain 2001 GDP Zone 4 Footprint and		
			Detention/Desilting Basin Between Zones 1 and 4 with Surface		
			Water Augmentation	7-5	
		7.2.4	Alternative 3: Modify Zone 4 Footprint to Avoid Permanent		
			Impact to Three Least Bell's Vireo Territories	7-8	
		7.2.5	Alternative 4: Shift Zone 4 Footprint Southwest for Recharge		
			Purposes	.7-10	
	7.3	Environm	nentally Superior Alternative/Least Environmentally Damaging		
		Practicab	ble Alternative	.7-12	
8.0	Cumula	tive Impac	cts	8-1	
9.0	Growth-Inducing Impacts, Unavoidable Adverse Impacts, and Significant				
	Irreversible Environmental Changes9-1				
	9.1	Growth-Inducing Impacts9-1			
	9.2	Unavoida	able Adverse Impacts	9-1	
	9.3	Significar	nt Irreversible Environmental Changes	9-2	
10.0	Persons	s and Orga	anization Consulted/List of Preparers	.10-1	
	10.1	Persons	and Agencies Consulted	.10-1	
	10.2	Preparer	s and Contributors	.10-2	
11.0	List of References11-1			.11-1	

LIST OF TABLES

_			
μ	а	a	е

<u>Table</u>		<u>Page</u>
1.1-1	List of Lead Agency and Local and State Responsible Agencies for	
	Amendment No. 2 to the 2001 Prima Deshecha General Development Plan	1-5
1.1-2	Federal Permitting Agencies for Amendment No. 2 to the 2001 Prima Deshecha	
	General Development Plan	1-5
1.1-3	Letters Received on the NOP	1-7
2.2-1	Summary of Alternatives Analyzed in EIRs 548 and 575	2-10
2.2-2	Permitted Land Uses and Viewshed Protection Requirements, County of	
	Orange/San Clemente MOU, County of Orange/San Juan Capistrano MOU	2-12
2.2-3	Permitted Land Uses and Viewshed Protection Requirements Contained	
	Within the County of Orange/RMV Settlement Agreement	2-13
2.2-4	Previously Adopted Mitigation 2001 GDP and EIR 575	2-16
4.3-1	Proposed Zone 4 Desilting Basins	4-4
4.3-2	Landfill Activities Matrix	4-20
5.3-1	Historical Summary Data – Groundwater Spring Geotechnical Investigation	5-13
5.4-1	State and Federal Ambient Air Quality Standards	5-20
5.4-2	Air Quality Levels Measured at the Saddleback Valley Monitoring Station	5-24
5.4-3	Existing (2005) Daily Pollutant Emissions Inventory at Prima Deshecha	
	Landfill (4,000 tpd)	5-25
5.4-4	Forecast (2020) Daily Pollutant Emissions Inventory at Prima Deshecha	
	Landfill (4,000 tpd)	5-26
5.4-5	SCAQMD Regional Pollutant Emission Thresholds of Significance	5-27
5.4-6	Required Best Available Control Measures (SCAQMD Rule 403, 1)	5-31
5.4-7	Fugitive Dust Control Actions (SCAQMD Rule 403, 1)	5-34
5.4-8	Contingency Control Measures for Large Operations (SCAQMD Rule 403, 3)	5-36
5.4-9	Track Out Control Options	5-37
5.5-1	Vegetation Types and Other Areas Acreage in 2001, 2003, and 2005	5-39
5.5-2	Special Status Plant Species Potentially Occurring on the Project Site	5-51
5.5-3	Special Status Wildlife Species Potentially Occurring on the Project Site	5-62
5.5-4	Vegetation Impacts by Phase for the Proposed Project (As of 2004)1	5-80
5.5-5	Vegetation Impact Differences	5-81
7.2-1	Alternatives Matrix	7-2

LIST OF EXHIBITS

Follows Page

FS-1	Proposed Project ES-12
2.1-1	General Plan Zones as Contained Within the GDP2-2
2.1-2	2001 GDP Landslide Remediation Area Project Elements
2.2-1	Existing Facilities
2.2-2	Zone 1 Fill Phasing Limits as Approved2-8
2.2-3	Excavation Phasing Limits in Zone 1 as Approved
2.2-4	Zone 4 Fill Phasing Limits as Approved
2.2-5	2001 General Development Plan Final Landfill Grades
2.2-6	PDL Burdened Property RMV Benefited Property per Amendment #1
2.2-7	Mitigation Site Location
2.2-8	Site A Mitigation Site
2.2-9	Site B Mitigation Site
2.2-10	Site C Mitigation Site
2.2-11	Site D Mitigation Site
4.1-1	Location Map4-2
4.1-2	Local Vicinity and 2001 GDP Zones
4.3-1	Proposed Project
4.3-2	Proposed Buttress Schematic
4.3-3	Shear Key Schematic
4.3-4	Revised Zone 4 Desilting System
4.3-5	Subsurface Storage Reservoir
4.3-6	Proposed Zone 4 Excavation Phasing Limits in Zone 4
4.3-7	Pre-Mitigation Plan On-Site
4.3-8	Off-site Pre-Mitigation Location (San Juan Creek Giant Reed Removal
4.3-9	Environmental Opportunities
4.3-10	Study Boundaries NCCP and SAMP
5.2-1	Geologic Features
5.3-1	Pre-Development Site Hydrology
5.3-2	Water Source Evaluation for Zone 4 Spring
5.3-3	Groundwater Monitoring Wells and Gradients
5.3-4	Post-Project Hydrologic Conditions at Project Site
5.5-1	Biological Resources
5.5-2	NCCP/HCP Designations
5.6-1	Easements and Leases
6.2-1	County Connector Trail
7.2-1	Alternative 1 – Maintain Zone 4 with Basin North of Prima Deshecha Creek
7.2-2	Alternative 2 – Maintain Zone 4 with Surface Water Augmentation7-6
7.2-3	Alternative 3 – Modify Zone 4 Footprint to Avoid Three LBV Territories
7.2-4	Alternative 4 – Zone 4 Modified for Recharge Purposes
8.1-1	La Pata Avenue Potential Roadway Alignments8-2

<u>Exhibit</u>

LIST OF APPENDICES

- A Notice of Preparation, February 9, 2004 and Notice of Preparation Comment Letters
- B 2001 General Development Plan Amendment 2
- C Memorandum of Understanding Between the City Of San Clemente, the Orange County Flood Control District, and the County of Orange Regarding the Prima Deshecha Landfill and the Memorandum of Understanding Regarding the Prima Deshecha Landfill Between the City of San Juan Capistrano and the County of Orange
- D Amendment No. 1 to the 2001 General Development Plan
- E An Analysis of Groundwater Resources in Zone 4, Prima Deshecha Landfill (GLA 2004)
- F Delineation of Jurisdictional Waters, Prima Deshecha Landfill, June 7, 2004
- G Pre-Mitigation Plan
- H Regional Environmental Enhancement Opportunities Plan (REEOP)

ACRONYMS

Acronym	Acronym and Abbreviation Description
A AAM AB ADT af AGM AQMD AQMD AQMP ARAR ASTM ASW	Annual arithmetic mean Assembly Bill Average Daily Traffic Man-made fill Annual geometric mean Air Quality Management District Air Quality Management Plan Applicable or Relevant and Appropriate Requirement American Society for Testing and Materials Automobile Shredder Waste
B BAS BMP Board BO	Bryan A. Stirrat & Associates Best Management Practice Orange County Board of Supervisors Biological Opinion
C CAA Caltrans CARB CCAA CCR CDFG CDMG CEQA CESA cfs CFR CIWMB CIWMP CNDDB CNF CNDDB CNF CNPS CO COVC CSS CSSMP CUP CVC CVWD CWA CWA	Clean Air Act California Department of Transportation California Air Resources Board California Clean Air Act California Code of Regulations California Department of Fish and Game California Department of Fish and Geology California Environmental Quality Act California Environmental Quality Act California Endangered Species Act Cubic feet per second Code of Federal Regulations County of Orange Integrated Waste Management Board County of Orange Integrated Waste Management Plan California Natural Diversity Database Cleveland National Forest California Native Plant Society Carbon Monoxide Orange County Vector Control Coastal sage scrub Coastal Sage Scrub Mitigation Program Conditional Use Permit California Vehicle Code Capistrano Valley Water District Clean Water Act Cubic yards

Acronym	Acronym and Abbreviation Description
D DAMP DBH DSEIR	Drainage Area Management Plan Diameter at breast height Draft Supplemental Environmental Impact Report
E EG EGR EIR EIS ERF ESA	Emission Guidelines Exhaust gas recirculation Environmental Impact Report Environmental Impact Statement Energy Recovery Facility Federal Endangered Species Act
F °F FEIR FESA FHWA FIRM ft FTC fy	Degrees Fahrenheit Final Environmental Impact Report Federal Endangered Species Act Federal Highway Administration Flood Insurance Rate Map Feet Foothill Transportation Corridor Fiscal year
G GDP gpm	General Development Plan Gallons per minute
H H ₂ S HBP HCA HCA/LEA HCP HCS HDPE HHW HHWCC hr HRMP	Hydrogen sulfide Harbors Beaches and Parks Orange County Health Care Agency County of Orange Health Care Agency/Solid Waste Local Enforcement Agency Habitat Conservation Plan Habitat Classification System High density polyethelene Household Hazardous Waste Household Hazardous Waste Collection Center hour Habitat Reserve Management Plan
I IA IHLMP IS IWMB IWMD	Implementation Agreement Interim Habitat Loss Mitigation Plan Initial Study Integrated Waste Management Board Integrated Waste Management Department

Acronym	Acronym and Abbreviation Description
J TD	Joint technical document
K km kV	Kilowolt
L Ibs/day LBV LCRS LCSA LEA LFG	Pounds per day least Bell's vireo Leachate Collection and Recovery System Load Check Storage Area Local Enforcement Agency Landfill Gas Generation
M mcy MDL MEP mg/L MM MMRP MPAH MOU MPRR MRF MSAA msl MW	Million cubic yards Method detection limit Materials Exchange Program Milligrams per liter Mitigation Measure Mitigation Monitoring and Reporting Program Master Plan of Arterial Highways Memorandum of Understanding Monitoring Program and Reporting Requirements Materials Recovery facility Master Streambed Alteration Agreement Mean Sea Level Monitoring Well
N NCCP NCCP/HCP NEPA NG NMFS NO ₂ NOCLATS NOI NOP NOX NPDES NRCS NSPS	Natural Communities Conservation Plan Natural Communities Conservation Plan/Habitat Conservation Program National Environmental Policy Act Native grassland National Marine Fisheries Service Nitrogen Dioxide North Orange County Landfill and Alternative Technologies Study Notice of Intent Notice of Preparation Nitric Acid National Pollution Discharge Elimination System Natural Resources Conservation Service New Source Performance Standards

Acronym	Acronym and Abbreviation Description
O O ₃ OCCP OCFA	Ozone Orange County Circulation Plan Orange County Fire Authority
P Pb PDF PDL PDSD PM2.5 PM10 ppm PQL	lead Project Design Feature Prima Deshecha Landfill Planning and Development Services Department Particulate Matter – Fine Particulate Matter Parts per million Practical quantitiation limit
Q Q ₁₀₀	100-year flood
R RCB RCP RDMD RDMD/HBP REEOP RHMP RMV ROG RWQCB	Reinforced concrete box Reinforced concrete pipe Orange County Resources and Development Management Department Orange County Resources Development and Management Department, Department of Harbors, Beaches, and Parks Regional Environmental Enhancement Opportunities Plan Riparian Habitat Mitigation Program Rancho Mission Viejo Company, LLC Reactive Organic Gas Regional Water Quality Control Board
S SAA SAMP SAMP/MSAA SB SCA SCAB SCAQMD SCE SCGC SCPWD SCTM SCWD SDG&E SDRWQCB SEIR SJCTM SMWD SO ₂	Streambed Alteration Agreement Special Area Management Plan Special Area Management Plan/Master Streambed Alteration Agreement Senate Bill Standard Condition of Approval South Coast Air Basin South Coast Air Quality Management District Southern California Edison Southern California Edison Southern California Gas Company San Clemente Public Works Department San Clemente Traffic Model San Clemente Water Department San Diego Gas and Electric San Diego Regional Water Quality Control Board Supplemental Environmental Impact Report San Juan Capistrano Traffic Model Santa Margarita Water District Sulfur Dioxide

Acronym SO4 SOS SOX SWF SWFP SWPPP SWRCB	Acronym and Abbreviation Description Sulfate Supplemental Open Space sulfur oxides Solid Waste Facility Solid Waste Facilities Permit Storm Water Pollution Prevention Program State Water Resources Control Board
T T-BACT TCA TCO tpd	Toxics–Best Available Control Technology Transportation Corridor Agency Technical Change Order tons per day
U UCI USACE USEPA USFWS USGS	University of California, Irvine United States Army Corps of Engineers US Environmental Protection Agency United States Fish and Wildlife Service United States Geological Survey
V VCD VMT VOC VPL VPP	Vector Control District Vehicle miles traveled Volatile Organic Compounds Viewshed Protection and Landscaping Plan Viewshed Protection Program
W WDR WMU WMU1 WMU2	Waste Discharge Requirements Waste Management Unit Waste Management Unit 1 Waste Management Unit 2
Symbols µg/m³	micrograms per cubic meter

GLOSSARY

Adverse Impact: A term used to describe unfavorable, harmful, or detrimental environmental changes. Adverse impacts may be significant or not significant (see Significant Impact).

Agricultural Land Use: The use of land primarily for farming, ranching, horse breeding, dairy farming and other forms of food and crop production. From a planning perspective, agricultural land use connotes primary economic use of the property.

Agriculture: Farming, including animal husbandry and the production and management of crops (including aquatic crops) for food, fiber, fuel, and ornament (source: Moorpark Municipal Code Section 17.08.010).

Air Basin: An area of the state designated by the Air Resources Board pursuant to Subdivision (a) of §39606 of the California Health and Safety Code for air quality planning purposes.

Air Monitoring: The periodic or continuous sampling and analysis of air pollutants in ambient air or from individual pollutant sources.

Air Pollutants: Substances that are foreign to the atmosphere or are present in the natural atmosphere to the extent that they may result in adverse effects on humans, animals, vegetation, and materials. Common air pollutants are ozone, nitrogen dioxide, particular matter, and carbon monoxide. Air pollution is defined in the *California Heath and Safety Code* as any discharge, release, or other propagation into the atmosphere and includes, but is not limited to, smoke, charred paper, dust, soot, grime, carbon, fumes, gases, odors, particulate matter, acids, or any combination thereof.

Air Pollution Control District (APCD): A local agency with authority to regulate stationary sources of air pollution (such as refineries, manufacturing facilities, and power plants) within a given county and that is governed by a District Air Pollution Control Board composed of elected county supervisors and city representatives.

Air Quality Management Plan (AQMP): A plan prepared by an air pollution control district or agency to comply with either the federal Clean Air Act or the California Clean Air Act. An AQMP contains measures that will be taken to attain and maintain federal and state ambient air quality standards. In California, air districts prepare air quality management plans that are included in the state's State Implementation Plan (SIP) that is required by the federal Clean Air Act. Such plans are also referred to as Clean Air Plans or Clean Air Attainment Plans.

Air Quality Model: An algorithmic relationship between pollutant emissions and pollutant concentrations used in the prediction of a project's pollutant impact.

Air Quality Standards: Standards promulgated by state or federal pollution control districts. It is the specified average concentration of an air pollutant in ambient air during a specified time period at or above which undesirable effects may be produced.

Air Toxics: Any air pollutant for which a national ambient air quality standard (NAAQS) does not exist (i.e., excluding ozone, carbon monoxide, PM10, sulfur dioxide, nitrogen dioxide) that may reasonably be anticipated to cause cancer, developmental effects, reproductive dysfunctions, neurological disorders, heritable gene mutations, or other serious or irreversible chronic or acute health effects in humans.

Ambient Noise: The background noise associated with a given environment, usually a composite of sounds from many sources near and far. The ambient noise level constitutes the normal or existing level of environmental noise at a given location.

Applicant: An applicant is a person who proposes to carry out a project which needs a lease, permit, license, certificate, or other entitlement for use or financial assistance from one or more public agencies when that person applies for the governmental approval or assistance (source: CEQA Guidelines §15351).

Approval: An approval is the decision by a public agency which commits the agency to a definite course of action in regard to a project intended to be carried out by any person. The exact date of approval of any project is a matter determined by each public agency according to its rules, regulations, and ordinances. Legislative action in regard to a project often constitutes approval. With private projects, approval occurs upon the earliest commitment to issue or the issuance by the public agency of a discretionary contract, grant, subsidy, loan, or other form of financial assistance, lease, permit, license, certificate, or other entitlement for use of the project (source: CEQA Guidelines §15352).

Aquifer: A natural underground formation that is saturated with water, and from which water can be withdrawn. A geologic formation of sand, rock, and gravel through which water can pass and which can store, transmit, and yield significant quantities of water to wells and springs.

Artifact: A single, portable man-made or man-altered object; usually culturally diagnostic.

Attainment: Achieving and maintaining the air quality standards (both state and federal) for a given air pollutant.

Attainment Area: An area considered to have air quality as good as or better than the National Ambient Air Quality Standard as defined in the Clean Air Act. An area may be an attainment area for one pollutant and a non-attainment area for others.

Automobile Shredder Waste: California's Automobile Shredder Waste Initiative (Initiative) was financed with grant funds provided by the United States Environmental Protection Agency through the Resource Conservation and Recovery Act. The goals of the Initiative were three fold: evaluate the adequacy of the Department of Toxic Substances Control's (DTSC) automobile shredder waste policy; affirm the regulatory status of the automobile shredders operating in California; and ensure compliance by the automobile shredders with the existing statutes and regulations.

Average Daily Traffic (ADT): The number of vehicles (trips) passing a given point on a road going in a direction during a 24-hour period.

Background Concentration: Air pollutant concentration due to natural sources and distant unidentified man-made sources.

Background View: View beginning at a distance from the observer and extending as far toward the horizon as the eye can detect the presence of objects. Skylines or ridge lines against other land surfaces are the strongest visual elements of the "background."

Base Flood: In any given year, a 100-year flood that has a one percent likelihood of occurring and is recognized as a standard for acceptable risk.

Base Flow: River surface flow, not counting storm flow and/or purchased imported water.

Baseline conditions: see existing conditions.

Basin Plan: A water quality control plan developed by a Regional Water Quality Control Board (RWQCB) for a specific geographic area. The Basin Plan identifies beneficial uses of waters, the water quality objectives needed to maintain these beneficial uses, and an implementation plan.

Beneficial Uses: The resources, services, and qualities of state waters that may be protected against quality degradation. The uses include, but are not limited to, domestic, municipal, agricultural and industrial supply, power generation, recreation, aesthetic enjoyment, navigation, and preservation and enhancement of fish, wildlife, and other aquatic resources or preserves. The specific uses such as "cold freshwater habitat" and "water contact recreation" are defined in Section 2 of the Regional Water Quality Control Boards' Basin Plans. Beneficial Uses are defined in *California Water Code*, Section 13050.

Best Management Practice (BMP): A BMP is any program, technology, process, siting criteria, operating method, measure, or device which controls, prevents, removes, or reduces pollution.

Biotic Community: A group of living organisms characterized by a distinctive combination of both animal and plant species in a particular habitat.

Buffer Zone: An area of land separating two distinct land uses that acts to soften or mitigate the effects of one land use on the other.

Buildout: Development of land to its full potential or theoretical capacity as permitted under current or proposed planning or zoning designations. Buildout also refers to the year in which project construction has been completed.

California Air Resources Board (CARB): California's lead air quality agency, consisting of a nine-member Governor-appointed board that is responsible for motor vehicle air pollution control and that has oversight over California's air pollution management program.

California Ambient Air Quality Standards (CAAQS): These are specified concentrations of air pollutants, recommended by the California Department of Health Services and adopted into regulation by the Air Resources Board, which relate the intensity and composition of air pollution to undesirable effects. CAAQS are the standards that must be met per the requirements of the California Clean Air Act.

California Clean Air Act (CCAA): A California law passed in 1998 that provides the basis for air quality planning and regulation independent of federal regulations and which establishes new authority for attaining and maintaining California's air quality standards by the earliest practicable date. A major element of the CCAA is the requirement that local Air Pollution Control Districts in violation of the California Ambient Air Quality Standards must prepare attainment plans that identify air quality problems, causes, trends, and actions that must be taken for attainment.

California Code of Regulations (CCR): The regulations that implement California laws.

California Department of Transportation (Caltrans): The state government agency responsible for the construction, maintenance, and operation of state and federal highways in California.

California Endangered Species Act: The California Endangered Species Act (CESA) (Fish & Game Code §§ 2050, *et seq.*) generally parallels the main provisions of the Federal Endangered Species Act and is administered by the California Department of Fish and Game (CDFG).

California Environmental Protection Agency (CalEPA): The state agency that incorporates the State Water Resources Control Board, the Integrated Waste Management Board, and other agencies with environmental responsibilities.

California Environmental Quality Act (CEQA): The California Environmental Quality Act, *California Public Resources Code,* Sections 21000 et seq (Source: CEQA Guidelines §15353).

Carbon Dioxide (CO₂): A colorless gas that enters the atmosphere as the result of natural and artificial combustion processes. It is also a normal part of the ambient air.

Carbon Monoxide (CO): A colorless, odorless gas resulting from the incomplete combustion of fossil fuels. CO interferes with the blood's ability to carry oxygen to the body's tissues and can result in adverse health effects. CO is a criteria air pollutant.

Carbon Monoxide Hot Spot: An area, usually an intersection or congested segment of a highway, that exceeds the federal or state carbon monoxide standard.

Catch Basin: A storm drain inlet having a sump below the outlet to capture settled solids.

Clean Air Act (CAA): A federal law passed in 1970 (and amended in 1977 and 1990) that sets primary and secondary National Ambient Air Quality Standards for major air pollutants and forms the basis for the national air pollution control effort.

Clean Water Act (CWA): Growing public awareness and concern for controlling water pollution led to enactment of the Federal Water Pollution Control Act Amendments of 1972. As amended in 1977, this law became commonly known as the Clean Water Act. The Act established the basic structure for regulating discharges of pollutants into the waters of the United States. It gave the EPA the authority to implement pollution-control programs such as setting wastewater standards for industry. The Clean Water Act also set water quality standards for all contaminants in surface waters. The Act made it unlawful for any person to discharge any pollutant from a point source into navigable waters, unless a permit was obtained under its provisions. It also funded the construction of sewage treatment plants under the Construction Grants Program and recognized the need for planning to address the critical problems posed by non-point source pollution. Subsequent enactments modified some of the earlier Clean Water Act provisions. Revisions in 1981 streamlined the municipal construction grants process, improving the capabilities of treatment plants built under the program. Changes in 1987 phased out the construction grants program and replaced it with the State Water Pollution Control Revolving Fund, more commonly known as the Clean Water State Revolving Fund. This new funding strategy addressed water guality needs by building on EPA-State partnerships.

Code of Federal Regulations (CFR): The document that codifies all rules of the executive departments and agencies of the federal government. It is divided into 50 volumes, known as titles. Title 40 of the CFR (referenced as 40 CFR) lists all the environmental regulations.

Community Noise Equivalent Level (CNEL): A noise compatibility level established by *California Administrative Code*, Title 21, Section 5000. It represents a time-weighted 24-hour average noise level based on the A-weighted decibel. The CNEL scale includes an additional 5 dB adjustment to sounds occurring in the evening (7 PM to 10 PM) and a 10 dB adjustment to sound occurring in the late evening and early morning between (10 PM and 7 AM).

Compost: The product resulting from the controlled biological decomposition of organic wastes that are source-separated from the municipal solid waste stream or which are separated at a centralized facility. Compost includes vegetable, yard, and wood wastes which are not hazardous wastes (*Public Resources Code*, Section 40116).

Conditional Use: A land use which is not permitted by right, but which may be appropriate in a given zoning district under certain circumstances. The use may occur only upon approval of a conditional use permit.

Conditional Use Permit (CUP): A permit based on a discretionary decision that is required prior to the initiation of particular uses that are not allowed as a matter of right. Such permits are subject to site plan review and may be conditioned at the time of approval (source: Moorpark Municipal Code Section 17.44.030).

Conformity: A requirement of the federal Clean Air Act that no department, agency, or instrumentality of the federal government shall engage in, support in any way, or provide financial assistance for license, permit, or approve any activity that does not conform with the State Implementation Plan (SIP) by causing or contributing to an increase in air pollution emissions, violation of an air pollution standard, or frequency of violating that standard.

Construction: Any site preparation, assembly, erection, substantial repair, alteration, or similar action for or of public or private rights-of-way, structures, utilities, or similar property.

Contiguous: Lands or legal subdivisions having a common boundary; lands having only a common corner are generally not contiguous.

Contour Grading: A grading technique which uses curvilinear, horizontal, and vertical undulations in order to simulate the characteristics of natural topography.

Co-Permittee: A permittee to a National Pollutant Discharge Elimination System (NPDES) permit that is only responsible for permit conditions relating to the discharges from its area of jurisdiction.

Cumulative Impact: A cumulative impact refers to two or more individual effects which, when considered together, are considerable or which compound or increase other environmental impacts. The individual effects may be changes resulting from a single project or a number of separate projects. The cumulative impact from several projects is the change in the environment which results from the incremental impact of the project when added to other closely related past, present, and reasonable foreseeable probable future projects. Cumulative impacts can result from individually minor but collectively significant projects taking place over a period of time (source: CEQA Guidelines §15355).

Cumulatively considerable: This term can be used to describe the proposed project if, when compared to other projects occurring or that will occur in the area, will cause a significant impact.

Debitage: Debris; waste products or by-products of the manufacturing process. Lithic debitage would include unused flakes, exhausted cores, and broken artifacts.

Decibel (dB): A unit for expressing the relative intensity (loudness) of sounds. The decibel is the logarithm of the ratio of the intensity of a given sound to the faintest sound discernible by the human ear.

Decibel, A-Weighted (dBA): A-weighting is a frequency correction that correlates overall sound pressure levels with the frequency response of the human ear.

Decision-making Authority: Decision-making authority means any person or body vested with the authority to make recommendations or act on application requests. The final decision-making authority is the one which has the authority to act on a request by approving or denying the request. This may include the Community Development Director or his/her designee, the Planning Commission, or the City Council.

Decision-making Body: Any person or group of people within a public agency that is permitted by law to approve or deny the project at issue (source: CEQA Guidelines §15356).

Deep Percolation: The percolation of surface water through the ground beyond the lower limit of the root zone of plants into a groundwater aquifer.

Demolition: Any dismantling, intentional destruction, or removal of structures, utilities, public or private rights-of-way surfaces, or similar property.

Desilting: The physical process of removing suspended particles from water.

Density: The gross site area which shall include local roadways, slopes, and open space areas, unless otherwise specified. Density is usually expressed "per acre." For example, a development with 100 dwelling units located on 20 acres has a density of 5 units per acre.

Detention Device: Facilities designed to collect and temporarily detain the initial volume of storm water runoff for a specified period of time to permit settlement of particulate pollutions.

Discretionary approval: An action taken by a government agency that calls for the exercise of judgment in deciding whether to approve or how to carry out a project.

Direct Effects: Effects which are caused by an action and occur at the same time and place.

Dispersion: The process by which atmospheric pollutants disseminate due to wind and vertical stability.

Drainage: An area that collects and diverts rain water and urban runoff down slope.

Drainage Area: The portion of the earth's surface from which precipitation or other runoff flows to a given location. With respect to a highway, this location may be a culvert, the farthest point of a channel, or an inlet to a roadway drainage system.

Effects: "Effects" and "impacts," as used in the CEQA Guidelines, are synonymous. Effects include: (a) Direct or primary effects which are caused by the project and occur at the same time and place; (b) Indirect or secondary effects which are caused by the project and are later in time or farther removed in distance, but are still reasonably foreseeable. Indirect or secondary effects may include growth-inducing effects and other effects related to induced changes in the pattern of land use, population density, or growth rate, and related effects on air and water and other natural systems, including ecosystems. Effects analyzed under CEQA must be related to a physical change (source: CEQA Guidelines §15358).

Effluent: Wastewater or other liquid, partially or completely treated or in its natural state, flowing from a treatment plant.

Emission: An air contaminant released to the atmosphere. The act of passing into the atmosphere of air contaminant or a gas stream that may or may not contain an air contaminant or the material so passed into the atmosphere (source: Ventura Air Pollution Control District).

Emission Standards: The federal Environmental Protection Agency (EPA), California Air Resources Board (ARB), or South Coast Air Quality Management District (SCAQMD) standards or limits for air contaminant emissions.

Encroachment: The occupancy of project right-of-way by non-project structures or objects of any kind or character; also, activities of other parties within the operating right-of-way.

Endangered Species Act: The Federal Endangered Species Act of 1973 provids a means whereby the ecosystems upon which endangered species and threatened species depend may be conserved, to provide a program for the conservation of such endangered species and threatened species, and to take such steps as may be appropriate to achieve the purposes of the treaties and conventions set forth in subsection (a) of the Endangered Species Act.

Energy Recovery Facility: The EPA expects about 30 percent of waste to be recycled or composted, leaving approximately 150 million tons of waste to be managed. This remaining waste will be handled using other elements of EPA's integrated waste management approach, primarily landfilling and energy recovery, also referred to as waste-to-energy (WTE). Energy recovery plants burn waste to heat water to produce steam in much the same way power plants burn coal, oil, natural gas, or wood. The steam can drive a turbine to produce electricity or be used directly for industrial processes or community heating. Today, there are 114 energy recovery plants that operate in 32 states throughout the United States. The EPA estimates that energy recovery plants will dispose of more than 15 percent of the nation's waste by the year 2000. These plants perform the following services:

- Burn about 15 percent of the waste generated nationwide, or about 101,000 tons each day.
- Generate enough electricity to meet the power needs of 1.2 million homes and businesses.
- Generate enough energy to replace nearly 30 million barrels of oil annually.
- Serve the disposal needs of more than 40 million people.

Environment: The physical conditions which exist within the area which will be affected by a proposed project including land, air, water, minerals, flora, fauna, ambient noise, and objects of historical or aesthetic significance. The area involved shall be the area in which significant effects would occur either directly or indirectly as a result of the project. The "environment" includes both natural and man-made conditions (source: CEQA Guidelines §15360).

Environmental Documents: Environmental documents means Initial Studies, Negative Declarations, draft and final EIRs, documents prepared as substitutes for EIRs, and Negative Declarations under a program certified pursuant to *Public Resources Code*, Section 21080.5 and documents prepared under NEPA and used by a state or local agency in the place of an Initial Study, Negative Declaration, or an EIR (source: CEQA Guidelines §15361).

Environmental Impact Report: A detailed statement prepared under the California Environmental Quality Act (CEQA) that describes and analyzes the significant environmental effects of a project and discusses ways to mitigate or avoid the effects. The term "EIR" may mean either a draft or a final EIR depending on the context. A Draft EIR means an EIR containing the information specified in CEQA Guidelines §§15122–15131. A Final EIR means an EIR containing the information contained in the draft EIR, comments (either verbatim or in

summary) received in the review process, a list of persons commenting, and the response of the Lead Agency to the comments received (source: CEQA Guidelines §15362).

Environmental Impact Statement: An environmental impact document prepared pursuant to the National Environmental Policy Act (NEPA). NEPA uses the term "EIS" instead of the term "EIR," which is used in CEQA (source: CEQA Guidelines §15363).

Environmental Protection Agency (EPA): The EPA is the federal agency with primary responsibility for the implementation of federal environmental statutes, including the Clean Water Act, Clean Air Act, Safe Drinking Water Act, and the Resource Conservation and Recovery Act. California is included within EPA Region IX, which is headquartered in San Francisco.

Equivalent Noise Level (Leq): A single-number representation of the fluctuating sound level in decibels over a specified period of time. It is a sound-energy average of the fluctuating level.

Erosion: The process by which material is removed from the earth's surface (including weathering, dissolution, abrasion, and transportation), most commonly by wind or water.

Erosion Control: The stabilization of cut-and-fill slopes and other areas.

Existing conditions: These are the regional and local settings that occur on the site at the time the Notice of Preparation/Initial Study was issued.

Expansive Soils: Soils that swell when they absorb water and shrink as they dry.

Farmland of Local Importance: These are lands of importance to the local agricultural economy and are determined by each county's board of supervisors and local advisory committee. Each county has developed its own definition of Farmland of Local Importance. (Source: Natural Resource Conservation Service)

Farmland of Statewide Importance: Lands similar to Prime Farmland but with minor shortcomings, such as greater slopes or with less ability to hold and store moisture. These lands have the same reliable source of adequate quality irrigation water available during the growing season as required for Prime Farmland. The land must have been used for the production of irrigated crops at some time during the two update cycles prior to the mapping date. (Source: U.S. Department of Agriculture Soil Conservation Service)

Fault: A fracture in the earth's crust forming a boundary between rock masses that have shifted. An active fault is a fault that has moved recently and which is likely to again. An inactive fault is a fault which shows no evidence of movement in recent geologic time and no potential for movement in the relatively near future.

Feasible: If something is feasible it is capable of being accomplished in a successful manner within a reasonable period of time when taking into account economic, environmental, legal, social, and technological factors (source: CEQA Guidelines §15364).

Filtration: The mechanical process that removes particulate matter from water by passing through sand or other media.

Flood: A general and temporary condition of partial or complete inundation of normally dry land areas from: (1) overflow of inland or tidal waters; (2) the unusual and rapid accumulation or runoff of surface waters from any source; (3) mudslides (i.e. mudflows) which are proximately caused by flood, and are akin to a river of liquid and flowing mud on the surface of normally dry

land areas, as when earth is carried by a current of water and deposited along the path of the current; and (4) the collapse or subsidence of land along the shore of a lake or other body of water as a result of erosion or undermining caused by waves or currents of water exceeding the cyclical levels which result in flood.

Flood, 100-Year (Q₁₀₀): The magnitude of a flood expected to occur on the average every 100 years, based on historical data. The 100-year flood has a 1/100, or one percent, chance of occurring in any given year.

Floodplain: Any land area susceptible to being inundated by flood waters from any source. It is the relatively level land area on either side of the banks of a stream regularly subject to flooding. That part of the floodplain subject to a one percent chance of flooding in any given year is designated as an "area of special flood hazard" by the Federal Insurance Administration.

Flow: A flow is a sudden movement of a soil mass in which individual particles travel separately in a fluid motion. Debris and mudflows are rapid and can be related to excess rainfall on slopes, often where vegetation has been removed. Debris flows often have the consistency of cement and can result in catastrophic effects to structures.

General Development Plan: see General Plan

General Plan: A compendium of city or county policies regarding long-term development, in the form of maps and accompanying text. A General Plan is a legal document required of each local agency by *State of California Government Code*, Section 65301 and adopted by a city council or board of supervisors. California law requires the preparation of seven elements or chapters in a General Plan: Land Use, Housing, Circulation, Conservation, Open Space, Noise, and Safety. Additional elements are permitted.

Geomorphic: Relating to the form or surface features of the earth.

Glare: A light source, either reflected or direct, that is annoying or distracting. The effect produced by lighting sufficient to cause annoyance, discomfort, or loss of visual performance and visibility. Glare can occur when the luminaire or associated lens of a light fixture is directly viewable from a location off the property that it serves (Source: Moorpark Municipal Code Section 17.30.020)

Grade: Adjacent ground level. For purposes of building height measurement, grade is the average of the finished ground level at the center of all walls of a building or other datum point established by the division of building and safety (source: Moorpark Municipal Code, Section 17.08.010).

Grading: Grading is the alteration of existing slope and shape of the ground surface.

Ground Failure: Ground Failure is ground movement or rupture caused by strong shaking during an earthquake. It includes landslide, lateral spreading, liquefaction, and subsidence.

Ground Shaking: This is ground movement that results from the transmission of seismic waves during an earthquake.

Groundwater: The term usually refers to the "saturated" zone in the ground where all the pore space between the soil particles is occupied by water. Water under the earth's surface, often confined to aquifers capable of supplying wells and springs, does not include water which is being produced with oil in the production of oil and gas or in a genuine mining operation.

Groundwater Basin: A groundwater reservoir defined by the entire overlying land surface and the underlying aquifers that contain water stored in the reservoir. Boundaries of successively deeper aquifers may differ and make it difficult to define the limits of the basin.

Grub: This is the process of removing vegetation by mechanical or manual methods.

Habitat: A place where a plant or animal naturally or normally lives or grows.

Habitat Conservation Plan (HCP): In 1983 Congress adopted Section 10 of the Endangered Species Act as a way to promote "creative partnerships between the public and private sectors and among governmental agencies in the interest of species and habitat conservation." Section 10 authorizes states, local governments, and private landowners to apply for an Incidental Take Permit for otherwise lawful activities that may harm listed species or their habitats. To obtain a permit, an applicant must submit an HCP outlining what he or she will do to "minimize and mitigate" the impact of the permitted take on the listed species.

Habitat linkage: Areas of natural habitat that function to join two larger blocks of habitat.

Hazardous Material: Dangerous, poisonous, corrosive, oxidizing, volatile, flammable, explosive, or toxic materials for which federal, state, or local industrial safety or other limits have been established. It is a material or form of energy that could cause injury or illness to persons, livestock, or the natural environment.

Hazardous Waste: A waste or combination of wastes that, because of its quantity, concentration, or physical, chemical, or infectious characteristics may either cause or significantly contribute to an increase in mortality or an increase in serious irreversible illness or pose a substantial present or potential hazard to human health or the environment when improperly treated, stored, transported, disposed of, or otherwise managed. A hazardous waste possesses at least one of four characteristics—ignitability, corrosivity, reactivity, or toxicity—or appears on special EPA or state lists. Hazardous waste is regulated under the federal Resource Conservation and Recovery Act and the *California Health and Safety Code*.

Herbicides: Chemical compounds that are used to destroy or control the growth of weeds and other undesirable plants.

High Fire Hazard Areas: Certain areas in the unincorporated territory of the county classified by the county fire protection district and defined as any areas within 500 feet of uncultivated brush, grass, or forest-covered land wherein authorized representatives of said district deem a potential fire hazard to exist due to the presence of such flammable material (source: Moorpark Municipal Code Section 17.08.010).

Hillside Area: Any property containing slope areas of 20 percent or greater. The steepness of a slope is defined as the relationship (the ratio) between the changes in elevation (rise) and the horizontal distance (run) over which that change in elevation occurs. The percent of steepness of any given slope is determined by dividing the rise by the run on the natural slope of land, multiplied by 100. (Source: City of Moorpark)

Household Hazardous Waste Collection Center (HHWCC): A collection center that gives residents a legal and cost-free way to dispose of unwanted household chemicals that cannot be disposed of in the regular trash.

Hydrology: The study of the water cycle.

Impact: The effect, influence, or imprint of an activity on the environment. Impacts include: direct or primary effects which are caused by the project and occur at the same time and place; indirect or secondary effects which are caused by the project and are later in time or farther removed in distance, but are still reasonably foreseeable. Indirect or secondary effects may include growth-inducing effects and other effects related to induced changes in the pattern of land use, population density, or growth rate and related effects on air and water and other natural systems, including ecosystems.

Impervious Surface: Ground surface that cannot be penetrated by water. It includes paved and compacted surfaces, as well as those covered by buildings.

Indirect Impact: Effects caused by an action that are later in time or farther removed in distance, but are still reasonably foreseeable. Indirect effects may include growth-inducing effects and other effects related to induced changes in the pattern of land use, population density or growth rate, and related effects on air and water and other natural systems, including ecosystems.

Indirect Source: Any structure or installation which attracts an activity that creates emissions of pollutants. For example, a major employment center, a shopping center, an airport, or a stadium can all be considered to be indirect sources. For purposes of air quality, facilities, buildings, structures, properties, and/or roads which, through their construction, indirectly contribute to air pollution. This can also include projects and facilities that attract or generate mobile sources activity (autos and trucks), such as shopping centers, employment sites, schools, and housing developments that result in emissions of any regulated air pollutant.

Infiltration: The introduction of underground water, such as groundwater, into wastewater collection systems. Infiltration results in increased wastewater flow levels.

Infiltration System: An infiltration system is a basin that is designed to capture runoff volume from the water quality design storm and infiltrate it to the soil.

Inflow: Surface water, such as rainfall runoff, that enters a wastewater collection system through manhole covers and joints or cracks in pipes. Inflow results in increased wastewater flow levels.

Infrastructure: Permanent utility installations, including roads, water supply lines, sewage collection pipes, and power and communications lines.

Initial Study: Under CEQA, a preliminary analysis that the lead agency prepares in order to determine whether an EIR, a Negative Declaration, or Mitigated Negative Declaration must be prepared, or to identify the significant environmental effects to be analyzed in an EIR (source: CEQA Guidelines §15365).

Integrated Waste Management Board (IWMD): The state agency within CalEPA responsible for solid waste management (non-hazardous).

Isolated Landform: An isolated landform is a parcel which has an isolated topographical feature that has a slope greater than 20 percent, a horizontal run of less than 200 feet, and a vertical rise of less than 50 feet.

Kilowatt (kW): One kilowatt is equal to 1,000 watts. Refers to the instantaneous amount of electricity used or generated.

Kilovolt (kV): A kilovolt is a unit of potential equal to a thousand volts.

Landfill: An area of land or an excavation in which wastes are placed for permanent disposal, and that is not a land application unit, surface impoundment, injection well, or waste pile.

Landmark: (1) A building, site, object, structure, or significant tree that has historical, architectural, social, or cultural significance and marked for preservation by the local, state, or federal government; (2) A visually prominent or outstanding structure or natural feature that functions as a point of orientation or identification.

Landscape linkage: see habitat linkage.

Landslide: A landslide is downslope movement of soil and/or rock, which typically occurs during an earthquake or following heavy rainfall.

Land Use: The purpose or activity for which a piece of land or its buildings is designed, arranged, or intended, or for which it is occupied or maintained.

Land Use Classification: A system for classifying and designating the appropriate use of properties.

Land Use Plan: An adopted map depicting the approximate location of residential, commercial, public, semi-public and private uses, open space, and road systems with a statistical summary of areas and densities for these land uses.

Lateral Spreading: Lateral spreading is the lateral movement of soil, often as a result of liquefaction, during an earthquake.

Leachate Collection and Recovery System (LCRS): Landfills generate leachate, a liquid that is a complex mixture of organic and inorganic pollutants. Leachate is produced as precipitation, groundwater, or other forms of moisture reacts with the waste material within the landfill. The exact composition of the leachate is dependent on site-specific conditions such as waste composition, temperature, pH, and site nutrients. In order to protect the environmental quality of nearby ground and surface waters, leachate collection, control, and treatment is a required component for all types of waste-disposal operations. A leachate collection system must be in place and operating prior to the placement of any waste material. In addition, the collection system must continue to operate effectively over the entire lifespan of the landfill.

Lead Agency: The public agency which has the principal responsibility for carrying out or approving a project. The Lead Agency will decide whether an EIR, Mitigated Negative Declaration, or Negative Declaration will be required for the project and will cause the document to be prepared (source: CEQA Guidelines §15367).

Liquefaction: A geologic phenomenon in which surface and near-surface materials (soils, alluvium, etc.) behave like a liquid during seismic shaking, often causing failure of soils to support structures.

Local Agency: Local agency means any public agency other than a state agency, board, or commission. Local agency includes but is not limited to cities, counties, charter cities and counties, districts, school districts, special districts, redevelopment agencies, local agency formation commissions and any board, commission, or organizational subdivision of a local agency when so designated by order or resolution of the governing legislative body of the local agency (source: CEQA Guidelines §15368).

Local Enforcement Agency (LEA): The local government entity that acts in the capacity as solid waste enforcement agency (14 CCR 18011[a][14]).

Mass Grading: A grading technique in which all lots, building pads, and streets are generally graded over the entire area resulting in the disruption of the majority of the on-site natural grade and vegetation and/often resulting in, but not required to result in, a successive pad/terrace configuration.

Master Plan of Arterial Highways (MPAH): A diagram in the Circulation Element which illustrates the arterial designation of roadways. Each arterial designation defines the number of ultimate lanes planned for a given roadway. Arterial designations include: Freeway, Transportation Corridor, Expressway, Major Highway, Primary Highway, Secondary Highway, and Commuter Highway.

Materials Exchange Program: A program designed to help businesses find markets for nonhazardous materials which they have traditionally discarded.

Materials Recovery Facility (MRF): A solid waste facility where solid wastes or recyclable materials are sorted or separated (by hand or by use of machinery) for the purposes of recycling or composting. (14 CCR 18720[a][36]).

Maximum Credible Earthquake: The largest Richter magnitude (M) seismic event that appears to be reasonably capable of occurring under the conditions of the presently known geological framework.

Memorandum of Understanding (MOU): An informal written record containing the terms of a transaction or agreement between two or more parties. An MOU is generally not legally binding.

Mineral Resource: Land on which known deposits of commercially viable mineral or aggregate deposits exist. This designation is applied to sites determined by the State Division of Mines and Geology as being a resource of regional significance, and is intended to help maintain the quarrying operations and protect them from encroachment of incompatible land uses.

Mitigation: Mitigation refers to: (1) avoiding the impact altogether by not taking a certain action or parts of an action; (2) minimizing impacts by limiting the degree or magnitude of the action and its implementation; (3) rectifying the impact by repairing, rehabilitating, or restoring the impacted environment; (4) reducing or eliminating the impact over time by preservation and maintenance operations during the life of the action; or (5) compensating for the impact by replacing or providing substitute resources or environments (source: CEQA Guidelines §15370).

Mitigation Measure: Action taken to reduce or eliminate environmental impacts. Mitigation includes: avoiding the impact altogether by not taking a certain action or parts of an action; minimizing impacts by limiting the degree or magnitude of the action and its implementation; rectifying the impact by repairing, rehabilitating, or restoring the affected environment; reducing or eliminating the impact over time by preservation and maintenance during the life of the action; and compensating for the impact by replacing or providing substitute resources or environments.

Mudflow (Mudslide): A river flow or inundation of liquid mud down a hillside, usually as a result of a dual condition of loss of brush cover and the subsequent accumulation of water on or under the ground, preceded by a period of unusually heavy or sustained rain.

National Pollutant Discharge Elimination System (NPDES): NPDES is the national program for administering and regulating discharges to waterways according to the Clean Water Act (CWA), Sections 401 and 402. In California, the State Water Resources Control Board and the nine Regional Water Quality Control Boards are responsible for administering the NPDES storm water program.

Natural Communities Conservation Plan (NCCP): The NCCP is an unprecedented effort by the State of California and numerous private and public partners that takes a broad-based ecosystem approach to planning for the protection and perpetuation of biological diversity. An NCCP identifies and provides for the regional or area-wide protection of plants, animals, and their habitats, while allowing compatible and appropriate economic activity.

Natural Grade: The grade unaffected by construction techniques such as fill, landscaping, or berming.

National Environmental Policy Act (NEPA): Enacted in 1969, the National Environmental Policy Act established a national policy for the environment to provide for the establishment of a Council on Environmental Quality, and for other purposes.

New Source Performance Standards: Section 111 of the Clean Air Act, "Standards of Performance of New Stationary Sources," requires the EPA to establish federal emission standards for source categories which cause or contribute significantly to air pollution. These standards are intended to promote use of the best air pollution control technologies, taking into account the cost of such technology and any other non-air quality, health, and environmental impact and energy requirements.

Nitrogen Dioxide (NO₂): A secondary contaminant formed through a reaction between nitric oxide (NO) and atmospheric oxygen, irritates the lungs at high concentrations and contributes to ozone formation.

Nitrogen Oxides (NOx): Chemical compounds containing nitrogen and oxygen; NOx reacts with volatile organic compounds in the presence of heat and sunlight to form ozone. It is also a major precursor to acid rain.

Noise: Annoying, harmful, or unwanted sound.

Noise Attenuation: Reduction of the level of a noise source using a substance, material, or surface, such as earth berms and/or solid concrete walls.

Noise Barrier: A wall or other solid structure constructed with the objective of attenuating (i.e., reducing) noise behind the barrier; commonly, a noise wall along a roadway.

Noise Contour: A line connecting points of equal noise level as measured on the same scale. Noise levels greater than the 60 Ldn (Day-night average sound level) contour (measured in dBA) require noise attenuation in residential development.

Noise Disturbance: Any sound which exceeds the noise standards by the County of Orange, the City of San Juan Capistrano, and the City of San Clemente.

Noise-sensitive Land Use: Any land use (i.e., residential development) or designated geographic area (i.e., hospital complex) where "intrusive noise" is incompatible with the conduct of the noise-sensitive uses or constitutes a "noise disturbance" for residents or works.

Non-attainment: The condition of not achieving a desired or required level of performance. This term is frequently used in reference to air quality.

Notice of Intent (NOI): Notice of intent to prepare an environmental impact statement.

Notice of Preparation (NOP): A brief notice sent by a Lead Agency to notify responsible agencies, trustee agencies, and involved federal agencies that the Lead Agency plans to

prepare an EIR for the project. The purpose of the notice is to solicit guidance from those agencies as to the scope and content of the environmental information to be included in the EIR. Public agencies are free to develop their own formats for this notice (source: CEQA Guidelines §15375).

Objective: A description of a desired condition for a resource. Objectives can be quantified and measured and, where possible, have established time frames for achievement.

Open Space: Land that has been left in its natural state and has not been developed with primary or accessory structures.

Ordinance: A law or regulation set forth and adopted by a governmental authority, usually a city or county.

Oxides of Nitrogen: A reddish-brown gas with an odor similar to bleach. The major source of this pollutant is the high temperature combustion of fossil fuels. Health effects include irritation and damage to lungs and lower resistance to respiratory infections.

Ozone (O₃): A compound consisting of three oxygen atoms that is the primary constituent of smog. It is formed through chemical reactions in the atmosphere involving volatile organic compounds, nitrogen oxides, and sunlight. Ozone can irritate the lungs as well as damage to trees, crops, and materials. There is a natural layer of ozone in the upper atmosphere which shields the earth from harmful ultraviolet radiation. Ozone is a criteria pollutant.

Paleontological Site: Any area or location containing a trace, an impression, or the remains of plants or animals from past ages.

Parcel: The basic unit of land entitlement. A designated area of land established by plat, subdivision, or otherwise legally defined and permitted to be used or built upon.

Particulate Matter–Fine (PM2.5): PM2.5 is a mixture of very small particulates with an aerodynamic diameter equal to or less than 2.5 microns. PM2.5 consists of particles directly emitted into the air and particulates formed in the air from the chemical transformation of gaseous pollutants. PM2.5 particulates are emitted from activities such as industrial and residential combustion, and from vehicle exhaust. Particles 2.5 microns or smaller infiltrate the deepest portions of the lungs and increase the risks of long-term disease, including chronic respiratory disease, cancer, and increased and premature death.

Particulate Matter (PM10): PM10 is any particulate matter with an aerodynamic diameter equal to or less than 10 microns. PM10 consists of particles directly emitted into the air and particulates formed in the air from the chemical transformation of gaseous pollutants. PM10 particulates are emitted from activities such as industrial and residential combustion, and from vehicle exhaust. PM10 causes adverse health effects and atmospheric visibility reduction. It is a criteria pollutant.

Permeability (soil): That quality of the soil or other geologic formations that enables it to transmit water or air.

Permit: The possession of a permit issued by the city, or where no permits are issued, the sanctioning of the activity by the jurisdiction as noted in a public record.

Pesticide: Any material used to control pests. This encompasses insecticides, herbicides, and rodenticides.

R:\Projects\OCIWMD-S\J004\Final Draft SEIR 597\TOC-083006.doc

pH: A measure of acidity or alkalinity of a material, liquid, or solid. It is measured on a scale of 0 to 14 with 7 representing a neutral state, 0 representing the most acid, and 14 the alkaline.

Prime Farmland: Lands with the best combination of physical and chemical features able to sustain long-term production of agricultural crops. The land must be supported by a developed irrigation water supply that is dependable and of adequate quality during growing season. The land must have been used for the production of irrigated crops at some time during the two update cycles prior to the mapping date (source: Natural Resource Conservation Service).

Project: Project means the whole of an action which has a potential for resulting in either a direct physical change in the environment, has a reasonably foreseeable indirect physical change in the environment, and that is any of the following: (a) an activity directly undertaken by any public agency including but not limited to public works construction and related activities, clearing or grading of land, improvements to existing public structures, enactment and amendment of zoning ordinances, and the adoption and amendment of local General Plans or their elements, pursuant to Government Code. Sections 65100-65700; (b) an activity undertaken by a person which is supported in whole or in part through public agency contacts, grants, subsidies, loans, or other forms of assistance from one or more public agencies; or (c) an activity involving the issuance to a person of a lease, permit, license, certificate, or other entitlement for use by one or more public agencies. The term "project" does not include: (a) Proposals for legislation to be enacted by the State Legislature; (b) Continuing administrative or maintenance activities, such as purchases for supplies, personnel-related actions, general policy and procedure-making (except as they are applied to specific instances covered above); (c) The submittal of proposals to a vote of the people of the state or of a particular community; (d) The creation of government funding mechanisms or other government fiscal activities, which do not involve any commitment to any specific project which may result in a potentially significant physical impact on the environment. The term "project" refers to the activity which is being approved and which may be subject to several discretionary approvals by governmental agencies. The term "project" does not mean each separate governmental approval. Where the Lead Agency could describe the project as either the adoption of a particular regulation under subsection (a)(1) or as a development proposal which will be subject to several governmental approvals under subsections (a)(2) or (a)(3), the Lead Agency shall describe the project as the development proposal for the purpose of environmental analysis. This approach will implement the Lead Agency principle as described in Article 4 (source: CEQA Guidelines §15378).

Prominent Landform or Ridgeline: A visually prominent landform or ridgeline means any landform visible from the valley floor that forms a part of the skyline or is seen as a distinct edge against a backdrop of land at least 500 feet horizontally behind it.

Public Agency: Public agency includes any state agency, board, or commission and any local or regional agency, as defined in the CEQA Guidelines. It does not include the courts of the state. This term does not include agencies of the federal government (source: CEQA Guidelines §15379).

Public Facilities: Institutional response to basic human needs, such as health, education, safety, recreation, and inspiration. Also, includes facilities and services such as, but not limited to, police, fire, libraries, parks, and flood control.

Reactive Organic Compound (ROC)/Reactive Organic Gases (ROG): Any organic compound containing at least one carbon atom, except for specific exempt compounds found to be non-photochemically reactive and thus not participating in smog formation. Classes of hydrocarbons (olefins, substituted aromatics, and aldehydes) that are likely to react with ozone and nitrogen dioxide in the atmosphere to form photochemical smog. These are also referred to as non-methane organic compounds or volatile organic compounds.

Recharge: The physical process where water naturally percolates or sinks into a groundwater basin.

Recharge Basin: A surface facility, often a large pond, used to increase the infiltration of surface water into a groundwater basin.

Regional: Pertaining to activities or economies at a scale greater than that of a single jurisdiction, and affecting a broad geographic area.

Responsible Agency: A public agency which proposes to carry out or approve a project, for which a Lead Agency is preparing or has prepared an EIR or Negative Declaration. For the purposes of CEQA, the term "Responsible Agency" includes all public agencies other than the Lead Agency which have discretionary approval power over the project (source: CEQA Guidelines §15381).

Risk Assessment: The qualitative and quantitative evaluation of the risk posed to human health and/or the environment by the actual or potential presence and/or use of specific pollutants.

Risk of Upset: The risk associated with potential explosions, fires, or release of hazardous substances in the event of an accident or natural disaster.

Special Area Management Plan (SAMP): The goal of Special Area Management Plans (SAMPs) is to achieve a balance between aquatic resource protection and reasonable economic development. SAMPs are designed to be conducted in geographic areas of special sensitivity under intense development pressure. These comprehensive and complex efforts require the participation of multiple local, state, and federal agencies. In addition, the U.S. Army Corps of Engineers considers public and stakeholder involvement an essential part of a successful SAMP.

Sanitary Landfill: The controlled placement of refuse within a limited area, followed by compaction and covering with a suitable thickness of earth and other containment material.

Sanitary Sewer: A system of subterranean conduits that carries refuse liquids or waste matter to a plant where the sewage is treated, as contrasted with storm drainage systems (that carry surface water) and septic tanks or leech fields (that hold refuse liquids and waste matter on the site). This can also refer to underground pipes that carry off only domestic or industrial waste, not storm water.

Sediment: Organic or inorganic material that is carried by or is suspended in water and that settles out to form deposits in the storm drain system or receiving waters.

Sedimentation: Process by which material suspended in water is deposited in a body of water.

Seismic: Caused by or subject to earthquakes or earth vibrations.

Sensitive Receptors: Locations where individuals especially sensitive to chemical exposure (such as children, the infirm, and the elderly) or are expected to be located on a regular basis. These sites include hospitals, daycare centers, and schools. Sensitive receptors were evaluated with residential exposure duration assumptions.

Significance threshold: A quantitative, qualitative, or performance level of a particular environmental effect that would normally be determined to be significant by the (lead) agency if the threshold is exceeded (CEQA).

Significant Impact or Significant Effect on the Environment: As defined by the CEQA Guidelines, a substantial, or potentially substantial, adverse change in any of the physical conditions within the area affected by the project including land, air, water, minerals, flora, fauna, ambient noise, and objects of historic or aesthetic significance. An economic or social change by itself shall not be considered a significant effect on the environment. It also refers to a social or economic change related to a physical change that may be considered in determining whether the physical change is significant. The lead agency will determine whether a project may have a significant effect on the environment based on substantial evidence in light of the whole record (source: CEQA Guidelines §15382).

Slide: A slide is a downslope movement of a soil or rock mass occurring dominantly on shallower slopes at surfaces of rupture or on relatively thin zones of intense shear strain. The displaced mass often slides beyond the toe of the surface rupture covering the original ground surface of the slope. Slides consist of two main types: rotational and translational. Rotational slides move along a surface of rupture that is curved and concave. Translational slides move along a planar or undulating surface of rupture

Slope Face: The slopes located directly below, or leading up to, the crest of a significant ridgeline or prominent landform.

Slope Steepness: The relationship (the ratio) between the change in elevation (rise) and the horizontal distance (run) over which that change in elevation occurs. The percent of steepness of any given slope is determined by dividing the rise by the run on the natural slope of land, multiplied by 100.

Solid Waste: Any non-hazardous garbage, refuse or sludge, which is primarily solid but may also include portions of liquid, semi-solid, or contained gaseous material that results from residential, industrial, commercial, agricultural, mining operations, and community activities.

South Coast Air Basin (SCAB): A geographic area defined by the San Jacinto Mountains to the east, the San Bernardino Mountains to the north, and the Pacific Ocean to the west and south. The SCAB is under the jurisdiction of the South Coast Air Quality Management District (SCAQMD).

South Coast Air Quality Management District (SCAQMD): The agency responsible for protecting public health and welfare through the administration of federal and state air quality laws, regulations, and policies in the South Coast Air Basin.

Southern California Association of Governments (SCAG): The organization, known in federal law as a Council of Governments or Metropolitan Planning Organization. As the designated Metropolitan Planning Organization, SCAG represents the counties of Imperial, Riverside, San Bernardino, Orange, Los Angeles, and Ventura, and the cities within these six counties. SCAG is mandated by the federal government to research and prepare plans for transportation, growth management, hazardous waste management, and air quality. Additional mandates exist at the state level.

Storm Water: Storm water means storm water runoff, snow melt runoff, and surface runoff and drainage.

Structure: Anything, including a building, located on the ground in a permanent location or attached to something having a permanent location on the ground.

Subsidence: Sinking of the land surface due to a number of factors, of which groundwater extraction is one.

Sulfur Dioxide (SO₂): A colorless, extremely irritating gas or liquid. Sulfur dioxide enters the atmosphere as a pollutant mainly as a result of burning high sulfur-content fuel oils and coal and from chemical processes occurring at chemical plants and refineries. There are National Ambient Air Quality Standards and California Air Quality Standards for sulfur dioxide.

Sump: In drainage, any low area that does not permit the escape of water by gravity flow.

Swale: An elongated or depressed landform within a landscaped area, which is designed to carry storm or other runoff.

Take: Significantly modifying the habitat of a listed endangered animal or habitat (FESA). The "hunt, pursue, catch, capture, or kill, or attempt to hunt, pursue, catch, capture, or kill" (CESA, *California Fish and Game C ode,* Section 86).

Threshold of Significance: Impact criteria which determines whether a project causes a significant impact.

Travel route: A landscape feature (such as a ridgeline, drainage, canyon, or riparian strip) within a larger natural habitat area that is used frequently by animals to facilitate movement and to provide access to necessary resources (e.g., water, food, cover, den sites).

Trustee Agency: A state agency having jurisdiction by law over natural resources affected by a project which are held in trust for the people of the state of California. Trustee agencies include the California Department of Fish and Game, State Lands Commission, the State Department of Parks and Recreation, and the University of California (with regard to sites within the Natural Land and Water Reserves System) (source: CEQA Guidelines §15386).

Unavoidable significant adverse impact: An effect that either cannot be mitigated or that remains significant even after mitigation is incorporated into the Proposed Project.

Unique Farmland: Lands of lesser quality soils used for the production of the state's leading agricultural cash crops. These lands are usually irrigated but may include non-irrigated orchards or vineyards as found in some climate zones in California (Source: Natural Resource Conservation Service).

Urban: Of, relating to, characteristic of, or constituting a city. Urban areas are generally characterized by moderate and higher density residential development (i.e., three or more dwelling units per acre), commercial development, and industrial development, and the availability of public services required for that development, specifically central water and sewer, an extensive road network, public transit, and other such services (e.g., safety and emergency response). Development not providing such services may be "non-urban" or "rural." CEQA defines "urbanized area" as an area that has a population density of at least 1,000 persons per square mile (Public Resources Code 21080.14[b]).

Urbanized Area: Urbanized area means a central city or a group of contiguous cities with a population of 50,000 or more, together with adjacent densely populated areas having a population density of at least 1,000 persons per square mile. A Lead Agency shall determine whether a particular area meets the criteria in this section either by examining the area or by referring to a map prepared by the U.S. Bureau of the Census which designates the area as urbanized. Use of the term "urbanized area" in Section 15182 is limited to areas mapped and designated as urbanized by the U.S. Bureau of the Census. (Source: CEQA Guidelines §15387).

Viewshed: The surface area that is visible from a given viewpoint or series of viewpoints. It is also the area from which that viewpoint or series of viewpoints may be seen (a collection of viewpoints). The viewshed aids in identifying the views that could be affected by the proposed action.

Volatile Organic Compound (VOC): Any organic compound containing at least one carbon atom, except for specific exempt compounds found to be non-photochemically reactive and thus not participating in smog formation. VOC is synonymous with reactive organic gases and reactive organic compounds.

Waste Stream: Any and all waste that has been generated and is being processed toward permanent disposition.

Wastewater: Water that has been previously used by a municipality, industry, or agriculture and has suffered a loss of quality as a result of use.

Wastewater Reclamation: Treatment and management of municipal, industrial, or agricultural wastewater to produce water of suitable quality for additional beneficial uses.

Watershed: The drainage basin contributing water, organic matter, dissolved nutrients, and sediments to a stream, estuary, or lake.

Wildlife corridor: A piece of habitat, usually linear in nature, that connects two or more large blocks of habitat that would otherwise be fragmented or isolated from one another.

Wildlife crossing: A small, narrow area, relatively short in length and generally constricted in nature, that allows wildlife to pass under or through an obstacle or barrier that otherwise hinders or prevents movement.

Zone: A specifically delineated area or district in a municipality within which regulations and requirements uniformly govern the use, placement, spacing and size of land and buildings.

EXECUTIVE SUMMARY

SECOND AMENDMENT TO THE 2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN AND SUPPLEMENTAL ENVIRONMENTAL IMPACT REPORT 597

Introduction. The 2001 General Development Plan (GDP), as currently amended, represents the latest planning document that guides actions and activities for the Prima Deshecha Landfill and does so through the landfill's projected closure in 2067. Environmental Impact Report (EIR) 575 contains a detailed environmental analysis of the 2001 GDP, and its conclusions and commitments remain applicable to the project. Amendment No. 2 updates the GDP with proposed new or revised project elements, and Supplemental Environmental Impact Report (SEIR) 597 analyzes the impacts of these new features of the Proposed Project. The Proposed Project elements in Amendment No. 2 are: (1) project features for which detailed information is now available (as they have moved into preliminary or final design); (2) project feature revisions needed for landslide stabilization of the site; and/or (3) project features required for maintenance of environmental mitigation and restoration areas. SEIR 597 will also provide an analysis of project elements that may require State and Federal agency permits.

Accordingly, SEIR 597 analyzes the incremental effects of the Proposed Project that are contained within Amendment No. 2 to the 2001 GDP, and it incorporates EIR 575 by reference.

Proposed Project Overview. In Amendment No. 2, the Proposed Project, discussed in greater detail below, addresses updates to the Landfill and Recreation elements of the GDP and consists of the following changes and/or additions to the approved project:

- An increase in the temporary limits of disturbance around the perimeter of the 2 landfill zones by approximately 278 acres to accommodate features for site-stabilization purposes and landfill-support activities (Zones 1 and 4)
- Re-design of the de-silting system for Zone 4
- Implementation of features to supplement water supply in Prima Deshecha Cañada stream channel
- Presentation of potential excavation phasing limits for Zone 4 and an update of Zone 1's fill phasing limits
- Development of a Pre-mitigation Program to offset project-related biological impacts
- Development of a Regional Environmental Enhancement Program that identifies environmental enhancement opportunities on site

Location. The 1,530-acre Prima Deshecha Landfill (PDL) site is located in the hills of southeastern Orange County, and includes acreage within the jurisdictions of the cities of San Juan Capistrano (570 acres) and San Clemente (133 acres). The remaining 827 acres are located within unincorporated Orange County. The Prima Deshecha Cañada watercourse traverses the site from the northeast to the southwest.

The 2001 GDP. The 2001 GDP and its first amendment were accompanied by EIR 575 (Notice of Determination was issued on November 6, 2001). This document serves as the planning guide for the PDL site and represents the product of updates to previous GDPs (1979 GDP, 1995 GDP); it was drafted to incorporate additional landfill design considerations and actions

required for remediation of an on-site landslide in 1998. Amendment No. 1 to the 2001 GDP incorporated additional design amendments and boundary constraints that were negotiated with the Rancho Mission Viejo Company, LLC (adjacent landowner).

The GDP utilizes a five-zone concept to guide planning decisions at the PDL site and to manage landfill operations. Zone 1 is the area of current landfill operation; Zone 2 represents the recreational trail area around the perimeter of the site; Zone 3 is an area of designated open space; Zone 4 represents the area where future landfilling operations will occur once Zone 1 has been closed (estimated around the year 2019); and Zone 5 is the Master Plan of Arterial Highways (MPAH) alignment of the La Pata Avenue extension through the center of the site. The 2001 GDP as amended, provides the baseline condition for the assessment of the impacts that are associated with Amendment No. 2 which are contained within SEIR 597.

The GDP identifies multiple uses for the Prima Deshecha site including solid waste management; regional park and recreational development; and circulation and transportation linkages. Accordingly, the GDP considers three primary focal points: a landfill element, a recreation element, and a circulation element. The Proposed Project addresses updates to the Landfill and Recreation elements of the GDP within Amendment No 2.

EIR 575 concluded that implementation of the 2001 GDP would result in impacts that remain significant after mitigation to in the following areas: topography, aesthetics, and biological resources; all other project impacts were found to be less than significant.

Purpose and Need for Amendment No. 2 to the 2001 GDP. Amendment No. 2 contains project modifications necessary to accomplish the goals discussed below.

Provide for physical site stability. The PDL site is comprised of bedrock which has had extensive landslides and which is prone to instability and slope failure. Slope-stabilization regulatory requirements must be met for future development of the landfill and for updates to the solid waste disposal permits that are needed to continue landfill operations. Without updated solid waste disposal permits, future landfill operations would prematurely cease and would trigger the transport of a significant amount of waste that was originally designated for disposal at the PDL to other, more distant, facilities. Amendment No. 2 and SEIR 597 considers the increase in the temporary limits of ground disturbance within and around Zones 1 and 4 that are likely required to construct the landslide remediation features that are needed for site stabilization.

Accommodate future, landfill-related support features. These features could include environmental-control facilities such as landfill gas perimeter probes, extraction wells and/or header lines, stockpile areas, temporary construction staging areas, or other support features. Although these features were anticipated in the 2001 GDP, Amendment No. 2 increases the temporary limits of ground disturbance that are associated with construction and operation of these facilities in order to assess biological impacts and obtain long-term resource agency permits. As these temporary limits of disturbance are conservative, they represent an area larger than may actually be necessary for these project features; however, the resource agency permits must be acquired before facility installation is required to ensure a seamless continuation of landfill operations throughout the life of the landfill.

Make adjustments in Zone 4 to improve drainage and to avoid sensitive biological resources. The Prima Deshecha Cañada stream originates from an existing spring in Zone 4 and is an important source of water that sustains biological resources downstream (including existing mitigation areas) at the project site. Impact to the spring and/or its recharge area could potentially result in adverse environmental impacts to

these biological resources. In addition, the original GDP-designated location for a desilting basin between Zones 1 and 4 directly impacts the least Bell's vireo, a federally listed endangered bird species. Amendment No. 2 re-configures the Zone 4 desilting system in an effort to avoid or reduce impacts to this species and sustain surface water flows to riparian resources downstream.

Acquire long-term resource agency permits to ensure seamless continuation of landfill operations and to integrate landfill activities with regional planning efforts. Landfill closure is not expected to occur until approximately the year 2067. Operations must proceed seamlessly in order to ensure the landfill functions as needed. Implementation of landslide remediation measures and landfill support features; impacts associated with an increase in the temporary limits of disturbance for these features; and the progression of landfill operations into Zone 4 will all require new resource agency and updated landfill operating permits. While updated landfill operating permits are not anticipated to be required until right before Zone 4 development is initiated. The IWMD intends to pursue biological resource agency permits at this time in order to implement a Pre-mitigation Program years ahead of the occurrence of any impact. Development (and coordination) of a comprehensive Pre-mitigation Program to offset these future impacts is necessary to ensure the PDL has the necessary resource agency approvals to proceed with Zone 4 development where required. In addition, the South Orange County Natural Communities Conservation Plan/Habitat Conservation Plan program (NCCP/HCP), a subregional conservation planning program, is being coordinated with all major stakeholders in the region in order to provide long-term protection of natural vegetation and wildlife diversity while allowing compatible land uses and appropriate development and growth. Amendment No. 2 incorporates the NCCP/HCP planning goals in the GDP through the proposed Pre-mitigation Program, and through a proposed regional environmental enhancement program that identifies enhancement opportunities on site that are consistent with the NCCP/HCP.

The Proposed Project. The Proposed Project for Amendment No. 2, as analyzed within SEIR 597 (Exhibit ES-1), will include the following elements:

- 1. An increase in the temporary limits of disturbance for the 2 landfill zones within the PDL site from 800 acres (2001 GDP) to 1,078 acres. This represents an increase of 278 acres (or 35 percent). Of this 278 acres, approximately 110 acres are around the perimeter of Zone 1 and 168 acres are around the perimeter of Zone 4. It should be noted that it may not be necessary to impact all 278 acres; this is intended to be a conservative estimate to provide for the development of a comprehensive Pre-mitigation Program for the maximum anticipated impacts to biological resources based on available documentation.
- 2. The re-design of the PDL desilting basin for Zone 4 that was originally proposed downstream of Zone 4. It includes construction of 4 desilting basins around the perimeter of Zone 4 for rainfall collection, runoff, and sediment management in lieu of the 2001 GDP desilting basin location that was originally proposed to be located between Zones 1 and 4.
- 3. Implementation of features to supplement water supply in the Prima Deshecha Cañada stream channel that includes the of construction of one or more sub-surface reservoirs beneath the relocated desilting basin(s) for flow storage and release over time into the downstream channel.

- 4. Presentation of the potential excavation phasing limits for Zone 4 to accommodate the construction of landslide remediation features. Fill phasing limits in Zone 1 have also been updated based on anticipated landslide areas.
- 5. Development, coordination, and implementation of a comprehensive Pre-mitigation Plan to compensate for biological impacts.
- 6. Development of a comprehensive conceptual plan that identifies regional environmental enhancement opportunities on site.

These elements represent the features of the Proposed Project which are contained within Amendment No. 2. However, **the following should be noted**:

- The Proposed Project for Amendment No. 2 does not change the landfill refuse prism or the amount of tonnage that is brought into the PDL.
- The increase in the temporary disturbance limits around the site (278 acres) is a conservative estimate for environmental pre-mitigation and agency coordination purposes; it is possible that landfill activities will not impact this entire acreage.
- The subsurface reservoir is a potentially promising concept for supplementing water supply in the Prima Deshecha Cañada stream channel. However, implementation of this feature is dependent on additional investigations to confirm engineering and operations maintenance feasibility, and permitting requirements; SEIR 597 contains other water supply alternatives that may be implemented if it is determined that the sub-surface reservoir concept is not feasible.

Environmental Analysis. SEIR 597 is a Supplemental EIR authorized by CEQA to address circumstances where changes have been made in the previously analyzed project, but only minor additions or changes are necessary to make the existing EIR (EIR 575) adequate. SEIR 597 analyzes the incremental effects of the Proposed Project for Amendment No. 2 over the effects that were already identified in the GDP. Based on the nature of the Proposed Project elements, a detailed analysis is presented for geophysical resources, hydrology and water quality, air quality, biological resources, and utilities and service systems. Environmental resource categories that do not require substantial additional analysis include land use and planning, agriculture, population and housing, transportation and circulation, noise, aesthetics, cultural resources, recreation, mineral resources, hazards, and public services.

SEIR 597 concludes that all impacts from the Proposed Project can be mitigated to a less-thansignificant level with the exception of air quality. However, with the implementation of updated mitigation measures and best available control measures, the PDL will have taken every reasonable effort to reduce emissions from the project site to the maximum extent practicable.

Project Alternatives. Five final alternatives, including the no project alternative, were analyzed in detail in SEIR 597 in addition to the Proposed Project. Several other alternatives were considered for detailed analysis but eliminated due to infeasibility.

Final Alternatives: The five final alternatives to the Proposed Project (as analyzed within SEIR 597) are as follows:

The No Project Alternative. The No Project Alternative consists of the approved project in the 2001 GDP, as revised by Amendment No. 1 and the Memoranda of Understanding between the County of Orange and the cities of San Juan
Capistrano and San Clemente. This alternative includes a desilting basin downstream of Zone 4 within the Prima Deshecha Cañada stream channel.

Alternative 1: This alternative would maintain the 2001 GDP Zone 4 footprint and shift the detention/desilting basin between Zones 1 and 4 to the north and out of the stream channel in order to reduce impact(s) to sensitive species.

Alternative 2: Alternative 2 would maintain the 2001 GDP Zone 4 footprint and relocate the detention/desilting basin between Zones 1 and 4 with surface water augmentation.

Alternative 3: This alternative would modify the Zone 4 footprint in order to avoid permanent impacts to 3 least Bell's vireo territories.

Alternative 4: Alternative 4 would shift the Zone 4 footprint southwest for recharge purposes.

Alternatives Eliminated from Further Consideration:

Reduce the Zone 4 Footprint and Deepen to Maintain Capacity. Significant uncertainties relative to site stability and potential impacts on local and regional groundwater resources eliminated this alternative from further analysis.

Shift the Zone 4 footprint over the ridge into Segunda Deshecha. The value of this area as an open space preserve and existing viewshed commitments resulted in the elimination of this alternative from further consideration.

Shift the Zone 4 footprint to the northeast. Significant impacts accrue from this alternative to utility line right-of-ways through the project site which resulted in the elimination of this alternative from further consideration.

Environmental Analysis of the Proposed Project

Environmental Impacts. Table ES-1 summarizes the environmental impacts of the Proposed Project. All impacts can be mitigated to a level that is less than significant except those for air quality. Implementation of updated mitigation measures and best available control technology will minimize air quality impacts to the maximum extent practicable, but not below significance thresholds.

Growth-Inducing Impacts. The Proposed Project does not include elements that alter refuse capacity at the site, create permanent employment opportunities, or affect regional housing trends. Accordingly, there will be no growth-inducing impacts.

Cumulative Impact Analysis. A thorough cumulative impact analysis was completed for the 2001 GDP within EIR 575. As a supplement to EIR 575, SEIR 597 has refined the analyses and updated mitigation measures as appropriate. It, therefore, does not induce cumulative effects above those identified within EIR 575.

The La Pata Avenue Gap Closure Study and the Proposed Project, as defined in Amendment No. 2/SEIR 597, have been closely coordinated to ensure that alternatives under consideration for the La Pata extension (and any mitigation requirements that are likely to arise) are compatible with one another, and that they do not result in unacceptable cumulative impacts to the area. If an alternative alignment is eventually chosen for the La Pata Avenue extension that

is different from the alignment shown in the MPAH, supplemental documentation will be completed at that time, if necessary.

Recommended Mitigation Measures and Updates for the Proposed Project

The following Executive Summary table (Table ES-1) contains a comprehensive list of all identified potential impacts, their associated mitigation measures, and the level of significance following mitigation for the Second Amendment to the 2001 Prima Deshecha General Development Plan and Supplemental Environmental Impact Report 597. All previously adopted mitigation measures and the level of significance following mitigation for the 2001 GDP and EIR 575 can be found in Section 2.2, Overview of the General Development Plan (Existing Conditions); Table 2.2-4, Previously Adopted Mitigation, the 2001 GDP; and in EIR 575.

AESTHETICS	Level of Significance after Mitigation		
No additional mitiga			
AIR QUALITY	Level of Significance after Mitigation		
Recommended M	itigation Measure Updates for the Proposed Project	Significant	
Particulate Emiss	ion (PM ₁₀) Control		
MM 5.4-1:	IWMD and its contractors shall be required to comply with regional rules to reduce air pollutant emissions. SCAQMD Rule 401 sets limits on the opacity of visible plumes of dust resulting from activities at the landfill. SCAQMD Rule 402 requires that air pollutant emissions generated at the landfill not be a nuisance off-site. SCAQMD Rule 403 requires that fugitive dust be controlled with the best available control measures so that the presence of such dust does not remain visible in the atmosphere beyond the property line of the emission source. Two options are presented in Rule 403: monitoring of particulate concentrations, or active control. Monitoring involves a sampling network around the project with no additional control measures unless specified concentrations are exceeded. The active control option does not require any monitoring, but requires that a list of measures be implemented on a daily basis.	The Prima Deshecha Landfill is currently implementing several mitigation measures to reduce potential air quality impacts. Implementation of the mitigation measures described in Section 5.4.4 would help to further reduce air quality impacts that result from operations at the	
	SCAQMD Rule 403 requires that "best available control measures" be utilized whenever a dust-generating activity occurs in the Air Basin. These measures are listed in Table 1 of Rule 403 and are presented in Section 5.4, Air Quality. It is important to note that all applicable measures from Table 5.4-6 should be implemented to achieve the required PM_{10} emissions reductions.	Prima Deshecha Landfill. However, even with implementation of all existing and recommended mitigation measures, operations at the Prima Deshecha Landfill would continue to result in significant and unavoidable air quality impacts.	
	Rule 403 requires that "Large Projects" implement additional measures. A Large Project is defined as "any active operations on property which contains 50 or more acres of disturbed surface area; or any earth-moving operation with a daily earth-moving or throughput volume of 3,850 cubic meters (5,000 cubic yards) or more than three times during the most recent 365 day period. The Prima Deshecha Landfill would be considered a Large Project under Rule 403. Therefore, the landfill is required to implement the applicable actions specified in Table 2 of the Rule. Table 2 from Rule 403 is presented in Section 5.4, Air Quality.		
	As a Large Operation, the landfill will also be required to:		
	 Submit a fully executed Large Operation Notification (SCAQMD Form 403N) to the SCAQMD Executive Officer within 7 days of qualifying as a large operation; Include, as part of the notification, the name(s), address(es), and phone number(s) of the person(s) responsible for the submittal, and a description of the operation(s), including a map depicting the location of the site; Maintain daily records to document the specific dust-control actions taken, maintain such records for a period of not less than three years; and make such records available to the Executive Officer upon request; Install and maintain project signage with project contact signage that meets the minimum standards of the <i>Rule 403 Implementation Handbook</i>, prior to initiating any earthmoving activities; 		

AIR QUALITY		Level of Significance after Mitigation
	 Identify a dust-control supervisor that is employed by or contracted with the property owner or developer, is on the site or available on-site within 30 minutes during working hours, has the authority to expeditiously employ sufficient dust mitigation measures to ensure compliance with all Rule requirements, and has completed the AQMD Fugitive Dust Control Class and has been issued a valid Certificate of Completion for the class; and Notify the SCAQMD Executive Officer in writing within 30 days after the site no longer qualifies as a large operation. 	
	Rule 403 also requires that the construction activities "shall not cause or allow PM_{10} levels to exceed 50 micrograms per cubic meter when determined by simultaneous sampling, as the difference between upwind and down wind sample." Large Projects that cannot meet this performance standard are required to implement the applicable actions specified in Table 3 of Rule 403. Table 3 from Rule 403 is presented in Section 5.4, Air Quality.	
	Further, Rule 403 requires that that the project shall not "allow track-out to extend 25 feet or more in cumulative length from the point of origin from an active operation." All track-out from an active operation is required to be removed at the conclusion of each workday or evening shift. Any active operation with a disturbed surface area of five or more acres, or with a daily import or export of 100 cubic yards or more of bulk materials must utilize at least one of the measures listed in Table 5.4-9 (see Section 5.4, Air Quality) at each vehicle egress from the site to a paved public road.	
Mobile Equipmen	at Emission Control	
MM 5.4-2:	To reduce equipment emissions, the following measures shall be implemented when feasible.	See above.
	 Use low emission mobile construction equipment. "CARB Certified" heavy construction equipment conforms to the latest off-road CARB emission standards and is the lowest polluting equipment available. The use of this equipment PM₁₀ emissions by approximately 50 percent from the emissions levels shown in Tables 5.4-3 through 5.4-5. This is a substantial reduction but will not reduce emissions to less than the significance thresholds. Maintain construction equipment engines by keeping them tuned. Use low sulfur fuel for stationary construction equipment. This is required by SCAQMD Rules 431.1 and 431.2. Utilize existing power sources (i.e., power poles) when feasible. This measure would minimize the use of higher polluting gas or diesel generators. Use aqueous diesel fuel where feasible and reasonably commercially available. 	

AIR QUALITY		Level of Significance after Mitigation
	Several of the mitigation measures listed above are advanced emission control technologies that are currently not commercially available. For example, aqueous diesel fuel reduces NO _X formation by reducing combustion temperatures, resulting in lower NO _X emissions. According to the SCAQMD, the current availability of this fuel technology is limited, and it may not be available for use at the landfill. In addition, with EGR diesel engines, a small amount of hot exhaust gas is routed through a cooler and mixed with fresh air entering the engine. The exhaust gas helps reduce the temperature during combustion, which lowers the formation of thermal NO _X . EGR technology is in the development phase, and has not been fully commercialized. To the extent that the advanced emissions control technologies become reasonably commercially available, or are required by the CARB from grading contractors, then such advanced emissions control technologies will be used.	
	Furthermore, a requirement to install diesel particulate filters on construction equipment used at the landfill was considered to further reduce emissions. However, the availability of construction equipment retrofitted with diesel particulate filters is limited. This is a result of operational problems in diesel engines equipped with these filters. Therefore, this potential mitigation measure for construction is considered infeasible.	
BIOLOGICAL RES	SOURCES	Level of Significance after Mitigation
Recommended M	itigation Measures/Updates for the Proposed Project	Less Than Significant
Additional Provisi	ions for Thread-Leaved Brodiaea	
MM 5.5-1:	Prior to the Initiation of construction within Phase C3, OCIWMD will obtain authorization to take the thread- leaved brodiaea may be obtained from CDFG through the provisions of Section 2081(b) of the California Fish and Game Code if no federal nexus is present such as a USACE Section 404.	The Proposed Project will affect approximately 255.02 additional acres that are needed to address
	If a USACE Section 404 Permit is being pursued, IWMD would request consultation with the USFWS under Section 7 of the FESA. Consultation is required between the USFWS and a federal agency (such as the USACE) whenever a federal action is likely to adversely affect species listed as Threatened or Endangered, such as thread-leaved brodiaea. The anticipated federal action is the issuance/amendment of a 404 permit that will affect the thread-leaved brodiaea.	the landslide remediation requirements necessary to stabilize the existing landslide complexes within Zones 1 and 4. Of this amount, approximately 55.24 acres contains sensitive
	At the conclusion of the consultation, the USFWS will prepare a Biological Opinion based upon its review of the information provided herein. The final Biological Opinion may include an incidental take statement.	habitats. With the implementation of the Pre-mitigation Plan that requires the installation and
	As part of the consultation process under Section 7 of the FESA, the California Department of Fish and Game (CDFG) will be consulted pursuant to Section 2080.1 of the California Fish and Game Code. Because the Project will affect a state-listed species, the thread-leaved brodiaea, CDFG concurrence with the Project conservation measures is required. The mitigation for the thread-leaved brodiaea will include the following requirements:	establishment of coastal sage scrub and southern needlegrass grassland prior to the occurrence of any impacts to these habitat types, the long-term net habitat
	• A pre-construction survey during the peak flowering period, approximately March through June, will be conducted by a qualified biologist. The limits of each brodiaea location within the impact area will be clearly delineated with lath and brightly colored flagging.	habitat types within the region will be sustained through project buildout with no cumulative loss.

BIOLOGICAL RESOURCES	Level of Significance after Mitigation
 The loss of thread-leaved brodiaea will be mitigated by seed and bulb collection, and revegetation into suitable mitigation site(s). A qualified biologist shall prepare a mitigation plan for review/approval by the United States Fish and Wildlife Service and oversee its implementation. The detailed mitigation plan shall include the following requirements: 	Therefore, impacts associated with the removal of this additional habitat will be reduced to a less- than-significant level through the implementation of this plan.
 The known populations of thread-leaved brodiaea on the project site shall be determined and mapped as the "collection area". The collection area shall include only areas within the impact footprint. 	In addition, SEIR 597 identifies potential impacts to the
 The existing locations of thread-leaved brodiaea shall be monitored every two weeks by a qualified biologist to determine when the seeds are ready for collection. A qualified seed collector shall collect all of the seeds from the plants within the collection area when the seeds are ripe. The seeds will be cleaned and stored by a qualified nursery or institution with appropriate storage facilities. 	increased least Bell's vireo numbers as well as the identification of potential impacts to San Diego fairy shrimp,
 Following the seed collection, the bulbs should be removed by an approved method (e.g., bulb collection or block transplantation). The bulbs shall either be transplanted directly or stored by a qualified nursery or institution with appropriate storage facilities. If the bulbs are collected and the block transplantation method is not used, then the top 12 inches of topsoil from the thread-leaved brodiaea locations shall be scraped, stockpiled, and used at the selected mitigation site. 	Riverside fairy shrimp, and western spadefoot toad that were not addressed in EIR 575. Implementation of the mitigation measures included in the SEIR
 The mitigation site(s) shall be located in open space. The site(s) shall not attempt to enhance existing populations and shall be located so as not to be impacted by any pesticides or herbicides used on adjacent properties. 	will reduce the effects of this project to less-than-significant levels.
 The thread-leaved brodiaea mitigation site(s) will be prepared for seeding as described in a conceptual restoration plan. 	
 The topsoil shall be re-spread in the selected location as approved by the project biologist. Approximately 60 percent of the seeds and bulbs collected shall be spread/placed in the fall following soil preparation. Forty percent of the seed and bulbs shall be kept in storage for subsequent seeding, if necessary. 	
 A detailed maintenance and monitoring plan shall be developed by a qualified biologist. The plan shall include detailed descriptions of maintenance appropriate for the site, monitoring requirements, and annual report requirements, and shall have the full authority to suspend any operation in the study area which is, in the qualified biologist's opinion, not consistent with the restoration plan. Any disputes regarding the consistency of an action with the restoration plan will be resolved by the 	
 appropriate Project Applicant and the biologist. The performance criteria shall be developed in the maintenance and monitoring plan and approved by a qualified biologist. The performance criteria shall also include percent cover, density, and seed production requirements. These criteria shall be developed by a qualified biologist following habitat analysis of an existing high-quality thread-leaved brodiaea population. This information will be recorded by a qualified biologist. 	

BIOLOGICAL RES	SOURCES	Level of Significance after Mitigation
	 If the germination goal is not achieved following the first season, remediation measures shall be implemented prior to seeding with the remaining 40 percent of seed and bulbs. Remedial measures shall include at a minimum: soils testing, control of invasive species, soil amendments, and physical disturbance (to provide scarification of the seed) of the planted areas by raking or similar actions. Additional mitigation measures may be suggested as determined appropriate by the project biologist. Potential seed sources from additional donor sites shall also be identified in case it becomes necessary to collect additional seed for use on the site following performance of remedial measures. IWMD is currently pursuing authorization to collect seed and propagate the brodiaea as well as transplantation of the plants and soils containing plants from CDFG under Section 2081(b). 	
Fairy Shrimp Surv	/eys	
MM 5.5-2:	Prior to the initiation of construction activities that involve the removal of any pond within Zone 4, the IWMD shall have focused surveys conducted for the San Diego fairy shrimp and Riverside fairy shrimp by a biologist possessing the necessary resource agency permits. The surveys will be performed during the winter season prior to any construction activities on the site that may impact appropriate habitat for the fairy shrimp (i.e., ponds). The surveys will follow the protocol developed by the USFWS for these species. If it is determined that either or both fairy shrimp species are not present, then no further mitigation is necessary. However, if one or both fairy shrimp species are present, then consultation with the USFWS will be necessary in order to obtain a take authorization prior to any construction activities that may impact the species. The permitting process would require the preparation of a Biological Assessment which would include a mitigation plan to avoid or minimize impacts on this species.	Less than significant
Western Spadefoo	ot Toad Surveys	
MM 5.5-3:	Prior to the initiation of construction activities that involve the removal of habitat that is known and/or has the potential to support the western spadefoot toad, the IWMD shall have a focused survey conducted, where appropriate, on the project site prior to any potential impacts and during the breeding season for this species (February through May). The survey results will be submitted within 30 days after completion of the last survey to the CDFG for concurrence. Based on the May 3, 2005 survey results, a relocation program will be developed for western spadefoot on the project site. The relocation program will include a detailed methodology for locating, capturing, and relocating individuals prior to capture. The relocation program will require a biologist with the necessary permits for handling the western spadefoot. Prior to implementation of the relocation program, the program and the biologist(s) implementing the program will be subject to approval of the CDFG.	Less than significant

BIOLOGICAL RES	Level of Significance after Mitigation	
Existing Mitigation		
MM 5.5-4:	Less than significant	
CULTURAL/SCIEN	NTIFIC RESOURCES	Level of Significance after Mitigation
No additional mitiga	ation measures are required for the Proposed Project.	
GEOPHYSICAL		Level of Significance after Mitigation
No additional mitiga	ation measures are required for the Proposed Project.	
HAZARDS/HAZAR	RDOUS MATERIALS	Level of Significance after Mitigation
No additional mitiga	ation measures are required for the Proposed Project.	
HYDROLOGY AND	Level of Significance after Mitigation	
Recommended Mi		
Water Quality		
MM 5.3-1:	The Proposed project will comply with Section 7 of the Drainage Area Management Plan (DAMP) for Orange County through the development of a Water Quality Management Plan.	Less Than Significant
LAND USE AND P	LANNING	Level of Significance after Mitigation
No additional mitiga	ation measures are required for the Proposed Project.	
NOISE		Level of Significance after Mitigation
No additional mitiga	ation measures are required for the Proposed Project.	
UTILITIES AND SE	Level of Significance after Mitigation	
Recommended Mi		
PDF 5.6-1:	SCE and SDG&E electrical transmission facilities will be relocated or re-routed, if necessary, in order to avoid service interruptions during construction of landslide remediation measures through the center of the site. IWMD will coordinate closely with SCE and SDG&E in the development of a plan to ensure cost-effective and efficient temporary facility relocation and post-construction re-establishment of transmission lines through the site.	Less than Significant



R:/Projects/OCIWMD/J004/SEIR 597PublicReview/ExES-1_project_020606.pdf

SECTION 1.0 INTRODUCTION

1.1 PURPOSE OF THE DRAFT SUPPLEMENTAL EIR

1.1.1 AUTHORITY

This Supplemental Environmental Impact Report (SEIR) has been prepared in accordance with the California Environmental Quality Act (CEQA), *Public Resources Code*, Section 21000 *et seq.*; the State CEQA Guidelines (Section 15000 *et seq.* of Title 14, *California Code of Regulations*); and the guidelines adopted by the County of Orange. Specifically, this Draft SEIR was prepared in accordance with the most recently adopted State CEQA Guidelines which were last amended on July 11, 2006.

As with project or program Environmental Impact Reports (EIRs), an SEIR is an informational document prepared pursuant to CEQA to provide for informed decision-making. That is, it provides decision-makers, public agencies, and the public in general with detailed information about the potential significant environmental effects of a proposed project. It also identifies the ways in which the significant effects of a project might be avoided, minimized or mitigated, and addresses alternatives to the project. In situations where an EIR has been finalized for a proposed project, Section 15163 of the State CEQA Guidelines states that where only minor changes or additions would be necessary to make the previous EIR adequate for the project as revised, the lead agency may prepare a Supplemental EIR, rather than a subsequent EIR. Since this is the case with this project, SEIR 597 need only contain the information necessary to make the previous EIR adequate.

In cases where a program EIR has been prepared, SEIRs will usually contain the same major topical elements of the program EIR; however, pursuant to State CEQA Guidelines, SEIRs may focus discussion solely on new effects, which had not been considered in the previous document. Within the context of a focused analysis of incremental project changes and their effects, the general elements of an SEIR usually contain the following:

- Executive Summary
- Project Description
- Environmental Setting, Impacts, and Mitigation Measures
- Alternatives to the Proposed Project
- Cumulative Impacts
- Growth-Inducing Impacts
- Effects Not Found to be Significant
- Organizations and Persons Consulted

1.1.2 PREPARATION OF A SUPPLEMENTAL EIR

In accordance with State CEQA Guidelines, the County of Orange Integrated Waste Management Department (IWMD) will act as the lead agency in the preparation of a supplemental EIR for this Prima Deshecha Landfill (PDL) project. The PDL 2001 General Development Plan (GDP) and its First Amendment were accompanied by EIR 575 for which a Notice of Determination was issued on November 6, 2001. EIR 575 served as a Project EIR for near-term projects and as a Programmatic EIR for long-term projects which covered known actions at the project site related to: current and future project operations; foreseeable and known site improvements related to regional transportation requirements; and subsequent land uses through the post-buildout period of project completion. At the time EIR 575 was issued, engineering designs had still not been developed for future (above and beyond the design in EIR 575) operations, and engineering and geotechnical considerations for future phases of

project operation had not been fully addressed with respect to impacts on biological resources. Accordingly, EIR 575 stated its intent to function as a Program EIR for these portions of the project site. Specifically mentioned within this context were landfill operations in the eastern portion of the site and unanticipated projects elsewhere on the property. Accordingly, EIR 575 was intended to simplify and narrow the scope of the necessary documentation that is required by CEQA for these portions of the site once additional project information became available.

Amendment No. 2 to the 2001 GDP further defines the extent of ground disturbance at the PDL site over the boundaries identified within EIR 575 and the 2001 GDP. Although there is no change to the refuse prism or refuse volumes being accepted at the PDL, additional information has become available regarding probable future operational activities that will result in a larger area being affected by temporary ground disturbance. These activities include landslide remediation requirements and implementation of landfill-related support features; adjustments to the Zone 4 desilting system; and measures needed to ensure the long-term success of the environmental mitigation and restoration components of the overall GDP (required for long-term State and Federal agency permits). Although consistent in concept with the 2001 GDP, these elements constitute incremental additions, refinements, or minor changes to the approved 2001 GDP, the effects of which are analyzed within SEIR 597. Prior to approval of the Proposed Project (Section 4.0, Proposed Project Description), the Board of Supervisors will consider Final EIR 575 as updated and revised by Supplemental EIR 597.

1.1.3 INCORPORATION BY REFERENCE

Pursuant to Section 15150 of the State CEQA Guidelines, SEIR 597 has incorporated by reference the entirety of the 2001 GDP (as previously amended), Final EIR 575, and relevant technical studies, analyses, and reports. Information from these documents (Appendices B and D) have been briefly summarized in the appropriate section(s) that follow; the relationship between the incorporated part of the referenced document and the Draft SEIR has also been described. The documents and other sources that have been used in the preparation of Draft SEIR 597 are identified in Section 11.0 (References). In accordance with Section 15150(b) of the State CEQA Guidelines, the location where the public may obtain and review these referenced documents and other sources used in the preparation of Draft SEIR 597 is listed below in Section 1.1.6, Availability of the Supplemental EIR.

SEIR 597 incorporates by reference all stated project objectives, overall project information, environmental analyses, mitigation measures, and construction elements contained within the Final EIR 575, and all supporting documentation. Analyses contained within Final EIR 575 for the 2001 GDP will be summarized, but will not be reiterated in detail unless there is a change in the analysis that has been necessitated by the Proposed Project.

1.1.4 INTENDED USES OF THE DRAFT SUPPLEMENTAL EIR

This Draft SEIR is necessarily broad in analytical approach in an effort to address certain elements of the Proposed Project that are part of the long-range program for full buildout of the Prima Deshecha Landfill. These elements are typically scheduled for implementation well into the future (e.g., 15 years or beyond from the date of issuance), and assumptions made regarding the specifics of project execution will be confirmed through final design efforts as implementation nears. This document also provides more in-depth analysis and environmental documentation on one or more specific projects contained within the 2001 GDP, for which additional design information is now available (refer to Section 4.0, Project Description).

Discretionary Approvals

A "discretionary approval" is an action taken by a government agency that calls for the exercise of judgment in deciding whether to approve or how to carry out a project. For this project, the lead agency responsible for SEIR certification is the Orange County Board of Supervisors.

County of Orange

To approve and implement the Proposed Project as outlined in Amendment No. 2 to the 2001 *Prima Deshecha General Development Plan*, the following specific discretionary actions by the Orange County Board of Supervisors are required:

- Certification of Final EIR 575 as modified by Supplemental EIR 597
- Approval of Amendment No. 2 to the 2001 Prima Deshecha General Development Plan

Full implementation of the proposed project would also require the following subsequent discretionary actions by the County of Orange Health Care Agency/Solid Waste Local Enforcement Agency (HCA/LEA) and the Regional Water Quality Control Board (RWQCB):

- Approval of Amended Waste Discharge Requirements (WDRs)
- Approval of Revised Solid Waste Facility Permit (SWFP)

It is likely that these permits will be issued as operations in Zone 4 near initiation. In addition to the four approvals identified above, County agencies may also be required to approve new and updated utility permits, including but not limited to: building, grading, encroachment, electrical, gas, fire, water, and other various permits. This document will provide information (as available) to support these permit application processes.

Integration of Site Planning into Larger-Scale Regional Planning Efforts

On-site environmental resources and restoration and enhancement opportunities at the PDL make the site an important partner in a number of significant and ongoing regional planning efforts. These planning efforts include, but are not limited to, the development of a San Juan Creek and Western San Mateo Creek Watershed Special Area Management Plan (SAMP); the Proposed Southern Subregion Natural Communities Conservation Plan (NCCP) for the Orange County Southern Subregion; and the La Pata Avenue Gap Closure Study (all of which are discussed in greater detail in Section 4.3.4 and Section 8.0). In accordance with the County of Orange's commitment to actively participate and further the goals of natural resources conservation efforts within its jurisdictional areas, SEIR 597 will be used as a framework within which to integrate proposed project actions at the PDL with these larger-scale regional planning efforts.

1.1.5 AGENCIES HAVING JURISDICTION

The lead agency for this SEIR is the:

County of Orange Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, CA 92703 Advisory bodies to the Board of Supervisors include the Orange County Planning Commission; Orange County Waste Management Commission; and the Orange County Harbors, Beaches and Parks Commission.

Responsible agencies expected to utilize Amendment No. 2 to the 2001 GDP and SEIR 597 in their decision-making and permitting processes (see **Error! Reference source not found.** below) include, but are not limited to, the following:

- Orange County Fire Authority (OCFA)
- Orange County Health Care Agency, Solid Waste Local Enforcement Agency (HCA/LEA)
- Orange County Resources and Development Management Department (RDMD)
- City of San Juan Capistrano
- City of San Clemente
- South Coast Air Quality Management District (SCAQMD)
- California Regional Water Quality Control Board, San Diego (SDWQCB)
- California Integrated Waste Management Board (CIWMB)
- United States Army Corps of Engineers (USACE)
- United States Fish and Wildlife Service (USFWS)
- California Department of Fish and Game (CFDG)

The approval process for Amendment No. 2 to the 2001 GDP and SEIR 597 will include a hearing at the County of Orange Planning Commission with a recommendation; the project will then proceed to the Board of Supervisors. Once the SEIR for the project is certified by the Board, local and state responsible agencies can rely on the environmental clearance for issuing or updating permits and for implementing projects, as appropriate. As discussed in Section 1.1.4, issuance of Amended WDRs for Zone 4 landfill operations is not anticipated until Zone 4 landfill operations near.

This document was specifically prepared to pursue resource agency permits and approvals that are necessary for the Proposed Project elements discussed below. The agencies expected to use SEIR 597 in their decision-making include the OCFA, the HCA/LEA, the RDMD, the City of San Juan Capistrano, the City of San Clemente, the SCAQMD, the SDRWQCB, the CIWMB, the USACE, the USFWS, and the CDFG. Additional detail on these agency actions may be found in Tables 1.1-1 and 1.1-2.

TABLE 1.1-1 LIST OF LEAD AGENCY AND LOCAL AND STATE RESPONSIBLE AGENCIES FOR AMENDMENT NO. 2 TO THE 2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN

Agency	Approval/Permit		
County of Orange (Lead Agency)			
Board of Supervisors	Second Amendment to the 2001 General Development Plan		
	Certification of Final EIR 575 as amended by SEIR 597		
Responsible State Agencies			
Regional Water Quality Control Board	Section 401 Certification Waste Discharge Requirements (WDRs) National Pollution Discharge Elimination System (NPDES) Permit		
Department of Fish and Game	Section 1602 Streambed Alteration Agreement NCCP Authorization (Note: MSAA may be approved for this)		
South Coast Air Quality Management District	 New Source Performance Standards (NSPS)/Emission Guidelines (EG) Title V (1990 Clean Air Act) – Regulation XXX Rule 1150 (Excavation of Landfill Sites) Rule 1150.1 (Landfill Gas Emissions) Rule 431.1 (Sulfur Content of Gaseous Fuels) Rule 431.2 (Sulfur Content of Liquid Fuels) 		
California Integrated Waste Management Board	Solid Waste Facilities Permit Revision (Concurrence)		
Responsible County of Orange Agencies			
Orange County Fire Authority	Fuel Modification Plan and Program Fire Break Roads		
Orange County HCA/Local Enforcement Agency	Solid Waste Facilities Permit Revision		
Planning & Development Services Department	Grading/Miscellaneous Permits		
Source: EIR 575, as modified			

Federal agencies and their respective permits and/or approvals are identified in Table 1.1-2. Section 4.3.3 provides an expanded discussion of these approvals and permits as they apply to the PDL site.

TABLE 1.1-2FEDERAL PERMITTING AGENCIES FORAMENDMENT NO. 2 TO THE2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN

Agency	Approval/Permit	
Federal Agencies		
U.S. Fish and Wildlife Service	Section 7 or 10(a) Permit or NCCP Authorization	
U.S. Army Corps of Engineers	Section 404 Permit	

1.1.6 AVAILABILITY OF THE SUPPLEMENTAL EIR

The County of Orange has complied with Sections 15063 and 15082 of the State CEQA Guidelines by preparing and issuing a Notice of Preparation (NOP) of a Draft SEIR. The NOP, which was distributed on February 9, 2004, included a brief description of Proposed Project elements covered by this Draft SEIR. The NOP was circulated to responsible agencies and potential interested groups and parties, as required by CEQA; the review and comment period for the NOP officially ended on March 3, 2004. The Initial Study responses and NOP comments were used to establish the scope of the issues addressed in this Draft SEIR. Appendix A contains a copy of the NOP, its distribution list, and comment letters received. A total of 8 letters were received during the NOP comment period from the public agencies, and are listed in Table 1.1-3 below. Copies of these letters can be found in Appendix A.

The Draft SEIR and all related technical studies are available for review at the County of Orange, Integrated Waste Management Department. All agencies, organizations, and individuals are invited to comment on the information presented in the Draft SEIR during the public review period. Specifically, comments are requested on the scope and adequacy of the environmental analysis. Respondents are also asked to provide or identify additional environmental information that is germane but that may not have been used in the analysis. Following the public review period, a response to all substantive public comments will be prepared and compiled into a Final SEIR, which will be considered by the Orange County Board of Supervisors (Board) for certification. Any parties interested in reviewing the SEIR and/or the documents incorporated by reference may do so at the following location:

County of Orange Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, CA 92703

Contact Persons for the Proposed Project are:

For telephone and email comments: Linda Hagthrop Public Information Officer (714) 834-4176 (phone) linda.hagthrop@iwmd.ocgov.com For written comments via mail or fax: Rochelle Carpenter Project Manager Orange County Integrated Waste Management Department 320 N. Flower St., Suite 400 Santa Ana, CA 92703 (714) 834-4001 (fax)

TABLE 1.1-3LETTERS RECEIVED ON THE NOP

Agency	General	Traffic	Air Quality	Biological Resources
Talega Associates, LLC	Reserves right to future			
	informed			
Transportation Corridor Agency (TCA)	Foothill Transportation Corridor alternatives may impact landfill operations; wants to review future documents			
OC Fire Authority (OCFA)	No comment; wants to be provided with updated information for Emergency Business Plan			
California Department of Transportation (Caltrans)	wants to be kept informed	 Would like traffic study for future La Pata work for regional traffic impacts, any changes in Level of Service (LOS), and types of traffic on the road. "Lead Agency should include the DOT Department of Transportation in its close coordination on regional transportation needs with regards to La Pata Avenue." "Care should be taken to contain loads resulting from construction and operation of the landfill, from blowing over or onto State Right of Way or facilities" 		
OC Health Care Agency (Local Enforcement Agency [LEA])	Clarify "landfill footprint" vs. "waste footprint"; Clarify which zones will be affected.			
California Integrated Waste Management Board (CIWMB)	No comments; If proposed project changes landfill operations, the Board will have comments on Draft SEIR			
SCAQMD	Use 1993 CEQA Air Quality Handbook		Calculate impacts from construction operations, indirect sources, and all toxic air contaminants; Utilize all feasible mitigation measures	
CDFG	CDFG appreciates and encourages pre-mitigation measures.	Include Analysis of Alternative La Pata Alignments, and potential effects of wildlife movement/design of preserve. Project should ensure that construction and operation of road does not affect NCCP.		Minimize impacts to existing resources; include measures to ensure wildlife connectivity not precluded

1.2 METHODOLOGY AND APPROACH

SEIR 597 analyzes the impacts of the Proposed Project for Amendment No. 2 to the 2001 GDP (herein referred to as the 'Proposed Project'). Accordingly, the approved project in the 2001 GDP provides the existing condition, or baseline condition, from which the incremental impacts of the Proposed Project are assessed. Section 2.0 of SEIR 597 presents an overview and description of the 2001 GDP and contains exhibits that illustrate the existing project condition at the site. The history of project actions on the site is summarized, as are mitigation measures in EIR 575.

Section 3.0 presents the purpose and need for the Proposed Project and Amendment No. 2 to the GDP. Section 4.0 then describes the Proposed Project elements which constitute new information, changed conditions, or increased detail over the 2001 GDP and Amendment No. 1. All exhibits within Section 4.0 illustrate features or elements of the Proposed Project.

Section 5.0 contains an analysis of those resource categories for which the Proposed Project results in either a new significant impact or an impact that is more severe than that analyzed in previous project environmental documentation. For these resources, organization of each topical section will vary slightly depending upon the environmental impact generated by the Proposed Project. Due to the nature of the Proposed Project, certain resource categories require in-depth analysis in order to conclude an impact designation of "no substantial change." These resources and detailed analyses are also presented in Section 5.0.

Section 6.0 presents those resources that were identified in the NOP/Initial Study (IS) and the accompanying environmental checklist provided by the County as sustaining "no substantial change" from implementation of the Proposed Project. For these resources, detailed analyses were not required; the results of prior analyses are briefly discussed in Section 6.0, along with an explanation of why Amendment No. 2 generates no substantial incremental environmental effect. For more information or detail on prior impact analyses conducted for the 2001 GDP, refer to FEIR 575 (2001).

Each resource category in Sections 5.0 and 6.0 is presented according to the outline below. As indicated above, the analyses in Section 5.0 are discussed in greater detail than those in Section 6.0:

- Existing Environmental Setting
- Significance Criteria
- Environmental Impact Analysis
- Mitigation Measures
- Level of Significance After Mitigation

1.2.1 EXISTING ENVIRONMENTAL SETTING

This section introduces the resource under consideration by describing the existing conditions related to each resource in the Draft SEIR. In accordance with Section 15125 of the State CEQA Guidelines, both the local and regional settings are discussed as they exist at the time of NOP issuance which is prior to implementation of the Proposed Project. In this case, the environmental setting will include a description of the resource under existing conditions including, if appropriate, conditions as affected by approved actions at the project site. This documentation will serve as the baseline upon which the Proposed Project-related impacts will be evaluated. Section 15125 also stipulates that an EIR shall discuss the inconsistencies between the Proposed Project and any applicable General and Regional plans including (but not limited to) habitat conservation plans (HCPs), NCCPs, and regional land use plans.

1.2.2 SIGNIFICANCE CRITERIA

The specific criteria upon which the significance of the project-related potential impacts are determined are considered the same as those utilized within FEIR 575. These have been derived from the significant effects identified in local (i.e., County of Orange), state, and/or federal policies and programs that may apply to the project as well as other commonly accepted technical and non-technical standards determined to be appropriate by the lead agency (pursuant to Section 15064.7 and Appendix G [Environmental Checklist] of the State CEQA Guidelines).

1.2.3 ENVIRONMENTAL IMPACT ANALYSIS

The environmental analysis for each issue is contained in Section 5.0 of this document. The extent of the analysis and documentation for each issue analyzed in the Draft SEIR were identified in the Initial Study (Appendix A). In some cases, the discussion will be a summary of the analysis conducted for FEIR 575; in other instances a technical study may have been prepared to specifically analyze the project-related impacts of the Proposed Project. In such cases, the technical report will be summarized to present the existing environmental conditions; to provide an assessment of the potential project-related impacts; and to identify and/or recommend appropriate mitigation measures in order to ensure that the environmental consequences are eliminated or reduced to a less-than-significant level, if feasible. All project-related impacts, including those associated with all phases of the Proposed Project, will be clearly and adequately analyzed in accordance with the State CEQA Guidelines. With respect to each potential effect, an analysis will be conducted to determine if, in fact:

- The project causes the identified "effect";
- The effect produces a substantial or potentially substantial change in the physical conditions within the area affected by the project as measured against the significance criteria and the previously identified effects as analyzed within FEIR 575; and
- The changed conditions are "adverse."

Where the investigation of a potential effect concludes that the effect is too speculative for evaluation, that conclusion is noted and the discussion of that effect is ended. In cases where the investigation demonstrates that a potential effect does or may (without undue speculation) occur but is beneficial, that conclusion is noted. Similarly, if the investigation demonstrates that a potential effect is not significant or not adverse, that conclusion is noted.

1.2.4 CUMULATIVE IMPACTS

Section 15130 of the CEQA Guidelines requires that an EIR discuss the cumulative impacts of a project when the project's incremental effect is "cumulatively considerable." A cumulative impact consists of an impact which is created as a result of the combination of the project evaluated in the EIR together with other projects causing related impacts. Section 15130(d) indicates that previously approved land use documents such as general plans may be used in cumulative impact analyses for consistency and demonstration of cumulative effect. Section 8.0 of the document considers the incremental effects of the Proposed Project together with the effect of other projects in the region in accordance with these Guidelines.

1.2.5 MITIGATION MEASURES

Where the analysis described above demonstrates that a potential effect does or may occur and is found to have a substantial or potentially substantial and adverse impact above those identified in FEIR 575, that conclusion is noted and:

- Mitigation measures are provided which will minimize these effects and, in most cases, reduce them to less-than-significant levels; and/or
- If feasible mitigation measures are not identified which can reduce the significant effects to less-than-significant levels, the effect will be identified as one which will result in "significant unavoidable adverse impacts."

1.2.6 LEVEL OF SIGNIFICANCE AFTER MITIGATION

Unavoidable significant adverse impacts are those effects that either cannot be mitigated or that remain significant even after mitigation is incorporated into the Proposed Project. These significant effects, if any, will be identified in the Draft SEIR. If significant unavoidable impacts are identified with the Proposed Project and are found as such by the decision-makers, it will be necessary for the Orange County Board of Supervisors to adopt a Statement of Overriding Considerations that identifies and describes public benefit(s) associated with project implementation which the Board finds sufficient to offset or override the significant impacts. This Statement of Overriding Considerations must be adopted in order to approve the Proposed Project if unavoidable significant adverse impacts occur.

SECTION 2.0 EXISTING CONDITIONS: HISTORY AND EVOLUTION OF THE 2001 GENERAL DEVELOPMENT PLAN

2.1 BACKGROUND AND EVOLUTION OF THE GDP

In February 1973, the Board of Supervisors established the Prima Desecha Landfill (PDL) as a multi-use concept for refuse disposal and recreation. The landfill began accepting municipal waste in 1976 in an area now known as Waste Management Unit 2 (WMU2). In December 1976, a General Development Plan (GDP) was initiated to combine both refuse disposal and, ultimately, recreational plans for the site upon closure.

An Interim Project Report/Environmental Impact Analysis for the PDL site was submitted in August 1978 to the County of Orange (County) Harbors, Beaches, and Parks (HBP) Commission. The report contained an Interim Plan and two ultimate Alternative Schematic Plans. Alternative 2 (an 81 million cubic yard refuse plan covering 800 acres of landfill area and 200 acres of borrow area) was recommended by the Commission and subsequently adopted by the Board in December 1978. That Alternative Schematic Plan was further refined and provided the basis for the 1979 Prima Deshecha GDP as well as the initial and current Solid Waste Facilities Permit (SWFP) No. 30-AB-0019 for the site. In 1980, the disposal operations were moved to a second active area known as Waste Management Unit 1 (WMU1) (Exhibit 2.1-1).

In 1994, an updated draft GDP was prepared and analyzed in a Program Environmental Impact Report (EIR 548), which was circulated for public review and comment in September and October of 1995. On November 21, 1995, the Board certified Final EIR 548 as adequately assessing potential environmental impacts associated with the 1994 GDP, but decided not to approve the 1994 GDP project in an effort to address viewshed concerns of the City of San Clemente. At Board direction, a revised plan was developed, and a Memorandum of Understanding (MOU) was executed between the City of San Clemente and the County of Orange for the Prima Deshecha property on July 1, 1997 (Appendix C).

The negotiated design amendments and boundary constraints were incorporated into the 2001 GDP (EIR 575), which now serves as the future planning guide for the Prima Deshecha site. This latest GDP utilizes a five-zone concept to guide planning decisions at the PDL site (see Section 2.2.2 and Exhibit 2.1-1) and to manage landfill operations. The 2001 GDP resembles the 1994 GDP but reflects the plan agreed to by the City of San Clemente and also incorporates actions required for remediation of a landslide which occurred in May 1998 in a stockpile area south of the Prima Deshecha Cañada stream (Exhibit 2.1-2). This landslide was first observed on May 21, 1998, following severe El Niño rains.

The landslide remediation measures incorporated into the 2001 GDP involved removing the landslide material from the soil stockpile, recompacting the materials within the Prima Deshecha Cañada stream, and rerouting the stream as a natural channel south of the compacted material. The recompacted material was designed to serve as a buttress to stabilize the area and to provide a stable subgrade for Zone 1 refuse disposal. Mitigation requirements for these landslide remediation activities have also been incorporated into the 2001 GDP, and are being implemented within the Zone 1 boundaries as coordinated with the Resource Agencies (see Section 2.3.2 for additional information on project biological mitigation activities to date).

2.2 OVERVIEW OF THE GENERAL DEVELOPMENT PLAN (EXISTING CONDITIONS)

As summarized above, the 2001 GDP is the product of updates to previous GDPs (1979 GDP, 1994 GDP) and was drafted to reflect a landfill plan agreed to by the cities of San Clemente and

San Juan Capistrano. The 2001 GDP also addresses additional landfill design considerations and actions required for remediation of the 1998 on-site landslide (Exhibit 2.1-2). Mitigation measures adopted in EIR 575 for the 2001 GDP are currently being implemented within Zone 1, as coordinated with the State and Federal Resource Agencies.

Since the site began operations in 1976, there has been substantial residential development in the region, particularly to the south within the cities of San Clemente and San Juan Capistrano. Accordingly, the 2001 GDP took the current and projected proximity of urban development into consideration.

The 2001 GDP, updated as appropriate with recent on-site additional information, provides the basis for the existing description of the site and the baseline for analyses contained within this SEIR. The following sections provide an overview of the objectives of the 2001 GDP and a corresponding description of existing conditions at the project site.

2.2.1 OBJECTIVES OF THE 2001 GENERAL DEVELOPMENT PLAN

The established objectives of the 2001 GDP remain applicable for the Proposed Project in Amendment No. 2, which is analyzed within this SEIR. The GDP identifies multiple uses for the Prima Deshecha site which encompass solid waste management, regional park and recreational development, and circulation and transportation linkages. It is important to note that the primary objective for the site is its function as a solid waste disposal facility; this objective has and will continue to take precedence over other identified future uses until such time as the landfill achieves full build out and can transition to recreational and open space uses once closure is complete. Secondary objectives include the provision of interim and ultimate recreational opportunities for the general public and the accommodation of regional highway circulation alignments. Listed below are the specific project objectives that are contained within the 2001 GDP and which were used to guide the development of the Proposed Project for Amendment No. 2. The purpose and need for the Proposed Project for Amendment No. 2 directly supports these GDP objectives.

Solid Waste Management Objectives

- Optimize the use of the site as a long-term waste disposal facility which operates in compliance with local, state, and federal regulations governing landfill operations, and in compliance with regulations protecting the environment.
- Provide for consistency with the County of Orange Integrated Waste Management Plan (CIWMP), adopted County and applicable City General Plans and zoning regulations, and compliance with City MOU design and operational conditions.
- Provide a long-term, regional solid waste management facility with appropriate safeguards to protect public health and safety as well as water, air, soil and other important resources which exist on-site and on surrounding property.

Circulation Objectives

- Provide for regional as well as local access to landfill operations and recreational activities on the site.
- Accommodate adopted Master Plan of Arterial Highways (MPAH) arterial highway alignments through the site.



Topography date: 9/23/04





R:/Projects/DCIW/MD/J004/SEIR 597/graphics/Ex2.1-1_GPZones_120605.pdf



2001 GDP Landslide Remediation Area and Project Elements

Prima Deshecha Landfill Supplemental Environmental Impact Report 597

*----

805 õ

D: P

Source: EIR 575 Scale: 1" = Approximately 500'







Recreation and Open Space Objectives

- Identify preferred activities that include a variety of passive and limited active recreational uses which respond to the changing recreational needs in the region.
- Provide a phased recreation concept for implementation of both interim and ultimate recreational uses as solid waste management activities allow.
- Consider recreation goals and objectives of the Orange County Master Plan of Regional Parks as well as with those identified in the Orange County Master Plan of Regional Riding and Hiking Trails and the San Juan Capistrano and San Clemente General Plans.
- Provide opportunities for the benefit of the public to develop and operate recreation facilities within the regional park.
- Provide essential linkages to the existing multiple use trails in the area which will also serve the recreation elements of the GDP.
- Preserve regionally significant habitat on the site which will be set aside as natural reserves and which can be used throughout the region for educational purposes.
- Provide linkage and open space opportunities for wildlife corridors.

2.2.2 2001 GENERAL DEVELOPMENT PLAN ELEMENTS: PROJECT SITE DESCRIPTION, EXISTING CONDITIONS

The 2001 GDP serves as the currently approved planning document that guides actions and activities at the PDL. The primary elements of the GDP, as currently approved, are summarized here and represent a current site description, as well as the baseline condition for impact analysis of the Proposed Project for Amendment No. 2 (as described in Section 4.0).

The Prima Deshecha 2001 GDP provides for the effective management of multiple uses on the site including solid waste disposal; various regional park and recreational uses included in the Orange County Master Plan of Regional Recreational Facilities; and implementation of a key arterial highway and road extension included in the Master Plan of Arterial Highways (MPAH), Orange County Circulation Plan (OCCP), and Circulation Elements of the Cities of San Juan Capistrano and San Clemente. It incorporates three primary elements including a Landfill Plan, a Circulation Plan, and a Recreation Plan. The 2001 GDP updated future project implementation plans with modifications to five planning zones. These modifications consist of zone boundary adjustments, landfill grading, and height limits which were developed through discussions with regional partners about updated design information.

The following sections describe existing conditions at the site (as covered by the approved 2001 GDP), subsequent modifications adopted with Amendment No. 1, and permitted project features and actions implemented to date. The focus of this section is the Landfill Plan Element of the 2001 GDP, as Amendment No. 2 primarily focuses on modifications to future landfill planning. A summary of the environmental impacts and mitigation measures incorporated into project commitments as a part of the 2001 GDP is then presented. (Note: *All exhibits and references in this section refer to the approved existing condition at the site and do not reflect Proposed Project modifications*.)

Description of Existing GDP Project Features

Zone 1: Zone 1 is currently designated for landfilling operations and comprises a total area of 327 acres, which includes non-refuse cut areas and the currently active refuse disposal area. WMU1 is located within Zone 1 (Exhibit 2.1-1). This zone provides for refuse disposal within 271 acres and will be filled over the next 13-year period. At the time of closure (anticipated in the year 2019), Zone 1 is expected to be completely filled; this is based on current assumptions for refuse inflow rates, daily cover use, and diversion. After closure activities have been completed, satisfactory access has been established and sufficient settlement has occurred, a needs analysis will be conducted to identify the recreational uses that are considered to be both compatible with the site and desirable for the general public.

The major phases of development for Zone 1 (as originally contained in the 2001 GDP) include Phases A, B, C, and D. Several subphases of fill and excavation within major Phases A through D are being implemented for the full development of Zone 1; they are Phases A, B, and C3. Filling has occurred in landfill areas designated as WMU1 (Summer 1980), Phases A (February 1999), A1 (November 2000), C1 (July 2002), B (July 2004), A2 (September 2005), and B1 (December 2005). Future development will continue to the east of these active landfill areas. Phase C2 is currently under construction (completion expected in October 2006). The ultimate development of Zone 1 will provide a total of 271 acres for refuse fill.

On-Site Facilities

The San Juan Capistrano Regional Household Hazardous Waste Collection Center (HHWCC) was constructed adjacent to Zone 1 as a temporary facility and opened in August 1991 (Exhibit 2.2-1). A new facility has recently been constructed near the landfill entrance and will function as a permanent facility (replacing the temporary facility) to meet current regulatory standards. The new facility consists of: (1) a secured area with covered storage to collect and store various household hazardous wastes in designated areas and (2) a Materials Exchange Program (MEP) area which allows materials received through the HHWCC to be set aside and reused by the public (at no cost) if regulatory requirements are met. The location of the new HHWCC is near the landfill entrance to provide for quick public access. In addition, the PDL's new Load Check Storage Area (LCSA) will be relocated adjacent to the new, permanent HHWCC. Both the LCSA and the HHWCC was demolished as part of the construction for Phase B1. Collected household hazardous waste that is temporarily stored in the HHWCC is disposed off site or recycled appropriately; no hazardous waste is permanently disposed at the PDL.

Landfill Operations

Landfill operations are proceeding in phased increments with soil excavation and installation of a liner system prior to refuse disposal and cover. Recently completed actions within the Zone 1 area include the installation of a liner system between 2 areas of the Zone 1 landfill in order to facilitate existing landfill operations. The presently completed Phase A2 liner system, located in the mid-northern portion of the site, was constructed as a small easterly extension of the Phase A1 liner to eliminate a gap between the Phase A1/C1 liner areas (see Section 2.2, Project Phasing). The recently completed Phase B1 area connects the existing liner system within Phase B (to the south) and Phase C1 (to the north). The Phase A2/B1 liner construction was completed in December 2005.



Source: BAS, 2005 Topography date: 9/23/04





R /Projects/OCIW/MD/J004/SEIR 597/Ex2.2-1_exist_fac_061506.pdf

Landslide Remediation Features

Landslide remediation activities implemented to date within Zone 1 have included: (1) the removal and re-compaction of stockpile and landslide materials as a buttress fill within an approximately 88-acre area, including a portion of the Prima Deshecha Cañada stream; (2) the relocation of the stream to the south into a new 85-to-101-foot-wide channel on coarse-grained alluvial-type soil material underlain with a low-permeability clay layer; (3) the planting of the stream channel with 7.1 acres of riparian and wetland habitat; and (4) the installation of gabion drop structures for channel erosion control within mitigation Site A of the realigned channel (see Section 2.3 and Exhibit 2.1-2).

Mitigation for landslide remediation activities, as coordinated and permitted by the resource agencies, also occured at a 7-acre site known as mitigation Site B (see Section 2.3.2) within Zone 1 and consisted of site excavation, planting of 4.5 acres of riparian and wetland vegetation, and construction of a stilling basin and a low-flow sediment release pipe. This mitigation program was initiated in Sites A and B in the spring of 2003 with the planting of Site B, followed by the planting of Site A in spring 2004. Section 2.3.2 provides additional detail and exhibits on the implementation and status of existing mitigation areas.

Aesthetic Requirements

Final landfill grades for Zone 1 are to be kept below the major ridgelines which form the northern and western edges of the landfill site boundary. This will ensure that landfill grades will not be visible from Ortega Highway, the valleys of San Juan Capistrano, and the Truman Benedict Elementary School in San Clemente's Forster Ranch community which will be consistent with the MOUs on record with the cities of San Juan Capistrano and San Clemente.

A Viewshed Protection Plan (VPP) is also required to protect views from the south in the City of San Clemente, and will be developed cooperatively between the County and the City as part of a condition for certification of EIR 575. The VPP needs to be implemented prior to Zone 4 operations.

Additional aesthetic mitigation requirements are discussed in Sections 2.2.4, 2.2.5, and 2.2.6.

Desilting Features

The 2001 GDP proposed enlargement of the existing desilting basin at the south end of Zone 1. This enlargement was completed in 2005, with a basin capacity increase from 9 acre-feet (af) to approximately 18 af (to the spillway bottom). Other basin improvements included lining the earthen facility with concrete to facilitate access (so earthmoving equipment could remove collected silt) and the construction of a new basin riser and emergency spillway. An existing dirt service road was also paved with an all-weather surface, and a bridge was installed for vehicles using the service road to safely travel over the new spillway. A pedestrian trail that crosses the southern and western limits of the basin was also incorporated into the basin improvement project. The Zone 1 basin improvements required resource agency permit modifications due to the removal of 0.09 acre of low quality mule fat scrub.

Lastly, a small portion of the stream bank (approximately 0.008 acre) at the interface between the newly realigned Prima Deshecha Cañada channel bank (Site A) and the existing riparian zone had been subject to potential erosion. A rock gabion structure approximately 60 feet in length and 4-to-6 feet in height was recently installed to provide a more permanent resource management and protection solution. The structure was installed within the realigned channel area and no vegetation was disturbed.

Zone 2: Zone 2 is designated for Multiple Use Trails and identifies the recreational trails that will eventually circumnavigate the landfill property (Exhibit 2.1-1). On-site public trails around Zone 1 can be used throughout the Zones 1 and 4 development as long as protection of public health and safety can be provided. Existing trails on the west and north sides of Zone 1 are currently in use. Trails depicted along the perimeter of Zone 4 will be available for interim recreational use during filling operations in Zone 1. When landfill operations move to Zone 4, this trail will be closed or relocated to protect public safety. When landfilling operations in Zone 4 are completed, the perimeter trails will be reconstructed and made available for use. The trails along the perimeter of Zone 4 will be restricted per the agreements between the County of Orange and Rancho Mission Viejo Company LLC (RMV) to a location 10 feet below and southwest of the existing ridgeline between the PDL property and adjacent RMV property to the north and east (this agreement is the basis for Amendment No. 1 to the 2001 GDP; a copy can be found in Appendix D). The GDP proposes to eventually connect the County trail along the edge of Zone 4 with on-site City trails proposed along Zone 1 to provide a complete loop for trail users. Discussions are ongoing with the Cities of San Clemente and San Juan Capistrano to identify specific trail alignments around Zone 1. The on-site trail connections would not be completed until Zone 4 is closed. A trail crossing under the future La Pata Avenue would have to be provided as part of an internal loop trail.

Zone 3: Zone 3 occurs within the Segunda Deshecha Cañada and the southern portions of the landfill. This zone contains native vegetation, including coastal sage scrub habitat used by the California gnatcatcher and mixed chaparral. Two open space areas on the site comprise Zone 3 and are proposed to be protected or restored to their natural state in concert with the Southern Subregion NCCP program currently under development by the County of Orange, landowners, environmental resource groups, and the Resource Agencies.¹ The intent of the GDP is to retain the majority of Zone 3 in a native state. Habitat restoration or enhancement could be implemented in Zone 3 where portions of these areas have been disturbed. Other opportunities exist in Zone 3 to provide mitigation for impacts associated with the development of the GDP in other zones, or as mitigation for impacts associated with other off-site development efforts in Orange County. These opportunities have been incorporated into the Proposed Project and are described in Section 4.0.

Zone 4: Zone 4 of the 2001 GDP is a 473-acre open space area (including non-refuse cut acreage) of which 409 acres is to be used for future refuse fill operations (Exhibit 2.1-1). Development of Zone 4 has a maximum elevation of 1,010 feet in accordance with MOU requirements between IWMD and the City of San Clemente. The final slopes and deck area have also been modified per the MOU to provide a more natural, undulating appearance. Additional aesthetic considerations were addressed in an agreement between the County of Orange and the RMV in 2002; this agreement constituted Amendment No. 1 to the 2001 GDP (Appendix C).

The scalehouse entrance facilities, personnel offices, equipment maintenance facilities, Energy Recovery Facility, and landfill gas flare station are located in the north-central portion of the site just west of Zone 4 (Exhibit 2.2-1). Fuel, water, and gas-condensate tanks are also located in this area. Additional facilities or modifications to existing facilities at the entrance to the landfill will be occurring prior to the La Pata Avenue extension project. However, there are no plans currently available so they are not considered in the Proposed Project (and are therefore not included in this SEIR).

¹ Resource Agencies include the US Fish and Wildlife Service, the California Department of Fish and Game, and the U.S. Army Corps of Engineers.

Landfill Operations

Zone 4 is planned for future refuse disposal activities after Zone 1 is filled to capacity; this is currently projected to occur in 2019. Zone 4 would be in active operation for approximately 48 years based on current projections for refuse inflow rate, daily cover use, and diversion. Landfill operations within this zone will fill a substantial portion of the Prima Deshecha Cañada channel. The 2001 GDP proposes 9 major refuse fill phases of development (Phases A through I) for Zone 4. Prior to the final closure of Zone 1, the IWMD will prepare Zone 4 to receive refuse. Zone 4 will then serve as the final refuse disposal area on the property.

Landslide Remediation Features

The 2001 GDP discusses a major landslide feature located within the Zone 4 refuse footprint and proposes an extensive shear key for stabilization of the refuse prism. However, the potential limits of excavation for construction of a shear key in Zone 4 were not identified in the 2001 GDP.

Desilting System

The 2001 GDP originally envisioned a detention and desilting basin to be located between Zones 1 and 4 to meet stormwater requirements for ultimate development of the landfill in Zone 4. This system would be implemented with interim desilting basins as part of on going Zone 4 landfill operations.

Zone 5: The future alignment and construction of the La Pata Avenue extension was anticipated within Zone 5 and is scheduled to pass through the center of the PDL site between Zones 1 and 4 (Exhibit 2.1-1). The currently approved MPAH provided the basis for the ultimate capacity and conceptual alignment for this roadway as it is presented in the 2001 GDP.

Project Phasing in the 2001 GDP

Fill phasing limits represent the areal extent of liner placement in support of refuse filling operations for each phase of landfill development. Excavation phasing limits represent the area needed for grading in order to provide for liner placement. These limits also differ in that the area that will be excavated prior to refuse disposal will reflect the total area needed for support activities such as site grading and subdrain installation, liner installation, material stockpiling, landslide remediation, and other design elements that are required to support future refuse disposal. Generally, excavation phasing limits will extend beyond fill phasing limits and will represent the greatest limit of ground-disturbing activity in the area being prepared for active refuse disposal.

Phasing limits of excavation and fill operations for Zone 1 at the PDL have been updated in the landfill operating permit documents since the approval of the 2001 GDP. Zone 4 refuse fill phasing limits have not changed from those proposed the 2001 GDP, and Zone 4 excavation phasing limits were not defined within the 2001 GDP. Exhibits 2.2-2 through 2.2-4 illustrate these excavation and fill phasing limits based on current site knowledge. Phasing of the 2001 GDP's Circulation and Recreation elements are not proposed for modification and are not considered in subsequent analyses; brief summaries of these elements are included below for information only.

Landfill Phasing

The 2001 GDP provides for the lateral and vertical development of the first refuse disposal area (WMU1) within Zone 1 from 125 acres to 271 acres over approximately the next 13 years, from 2006 through 2019. Zone 4 would then be utilized within its 409-acre refuse footprint for about 48 years, based on a disposal rate of 4,000 tons per day (tpd). Neither the refuse footprint on or the capacity are proposed for modification within SEIR 597. The total life of the site for landfilling purposes (as of 2006) is 61 years and is estimated to close in the year 2067. The life of the site could change if assumptions for the daily refuse inflow rate change or if new technologies are developed which enhance landfill capacity.

Phasing scenarios that were developed in the 2001 GDP (which include 4 major fill and excavation development phases—A, B, C, and D— within Zone 1) are located to the east of WMU1 (Exhibits 2.2-2 and 2.2-3). Excavation and lining of Zone 1 are currently proceeding in subphases and will move progressively east until the development boundary is reached. The completed development subphases for Zone 1 include Phases A, A1, and A2; B and B1; C1, and C2 (to be completed in October 2006). Additional subphases are proposed in the future for major Phases C and D.

Exhibit 2.2-4 illustrates current fill phasing limits for Zone 4. As indicated in FEIR 575, excavated material in initial phases will be used for daily cover and compacted fills that are proposed for future phases in Zone 4. Excess excavation material from earlier phases can be stockpiled in future phase areas. Once fill operations reach the final phases, soil material excavated to develop these phases will be stockpiled on previously filled phases (above the interim fill and below the final fill grades proposed). There is no anticipated need or plan to excavate trash currently buried in that portion of WMU2 located within Zone 4. A liner system would likely be proposed over previously filled areas. However, should refuse excavation be required by the regulatory agencies, then IWMD would secure necessary permits during final design for removing refuse.

Exhibit 2.2-5 presents the final grades of the completed landfill as stated within the 2001 GDP.

Recreation Plan Phasing

The phasing and implementation of recreational uses at the site are constrained in that landfilling activities and uses will take precedence over recreation improvements. In addition, the phasing of landfilling activities, post-closure maintenance of the landfill itself, and physical conditions at the site (e.g., settlement) will significantly affect the timing and development of future recreational site features and facilities. Once Zone 1 has been closed, satisfactory access has been established, and sufficient time has passed to allow for settling of the fill material, the area (Zone 1) would be available for interim recreational use during the period of active landfill operations within Zone 4. A needs analysis will be conducted to identify the recreational uses that are compatible with the ongoing conditions and the desires of the general public.

The Orange County Resources Development and Management Department, Department of Harbors, Beaches, and Parks (RDMD/HBP) will coordinate with IWMD on the development of trails on the Prima site with the development of trails outside the site proposed by the cities of San Clemente and San Juan Capistrano. Trails depicted around Zone 4 will be available on an interim basis only during Zone 1 filling operations, and would be closed to the public upon commencement of landfill activities within Zone 4 for public safety reasons. Trail heads adjacent to Zones 1 and 4 will be developed after each zone is closed to active landfill operations. Onsite City trails around Zone 1 can be used throughout the development of both Zones 1 and 4.



Zone 1 Fill Phasing Limits (EIR 575)

Exhibit 2.2-2

Prima Deshecha Landfill Supplemental Environmental Impact Report 597



R:\Projects\OCIWMD/J004/SERI597PublicDraftReview/Graphics/Ex2.2-2_z1_phasing_575_072505.pdf



Source: BAS, 2005

Excavation Phasing Limits in Zone 1 (Current)

Prima Deshecha Landfill Supplemental Environmental Impact Report 597

Exhibit 2.2-3



Scale: 1" = 1000'

 $R: \label{eq:resonance} R: \label{eq:resonance} R: \label{eq:resonance} Projects \label{eq:resonance} OCIWMD \label{eq:resonance} J004 \label{eq:resonance} SIR597 \mbox{PublicReviewDraft} \mbox{Graphics} \mbox{Ex2.2-3_z1} \mbox{excavation} \mbox{OCIWMD} \label{eq:resonance} J004 \label{eq:resonance} SIR597 \mbox{PublicReviewDraft} \mbox{Graphics} \mbox{Ex2.2-3_z1} \mbox{excavation} \mbox{OCIWMD} \mbox{J004} \mbox{SEIR597 \mbox{PublicReviewDraft} \mbox{Graphics} \mbox{Ex2.2-3_z1} \mbox{excavation} \mbox{OCIWMD} \mbox{J004} \mbox{SEIR597 \mbox{PublicReviewDraft} \mbox{Figure} \mbox{F$





The GDP proposes to eventually connect the County trail along Zone 4 with on-site City trails proposed along Zone 1 to provide a complete loop for trail users.

Subsequent to the closure of Zone 4 (projected for the year 2067), the 2001 GDP envisioned the completion of a recreational needs analysis that is closely coordinated with the public in order to prepare for final designation of post-closure site uses.

Circulation Element Phasing

The County RDMD, in coordination with the cities of San Clemente and San Juan Capistrano and the IWMD, has initiated a Feasibility Study Project Design Report, and an Environmental Impact Analysis for the extension of La Pata Avenue from Ortega Highway (through the Prima Deshecha property to Calle Saluda in San Clemente. This work will generate a proposed final alignment for the La Pata Avenue extension through the project site. Pending further studies after the completion of the La Pata Avenue extension, the extension of Camino de los Mares through the southwestern corner of the site may be initiated by the cities of San Clemente and/or San Juan Capistrano. Construction of any roadway through the site would be closely coordinated with local cities, the IWMD, and the RDMD. The MPAH also indicates that Camino Las Ramblas will continue in a northeasterly direction from its intersection with the proposed Camino de los Mares extension to eventually connect with the La Pata Avenue extension, adjacent to the westerly and northerly landfill boundaries. Final implementation is subject to approval by all responsible agencies and certification of requisite CEQA documentation. The currently proposed schedule for the extension of La Pata Avenue through the Prima Deshecha property is 2012. The extension of Camino de los Mares and Camino Las Ramblas will be further studied upon completion of the La Pata Avenue extension.

2.2.3 ALTERNATIVES ANALYZED WITHIN EIRs 548 AND 575

Alternatives Analyzed in Detail

Table 2.2-1 summarizes the alternatives analyzed within EIRs 548 and 575. EIR 548 analyzed impacts due to the 1994 GDP, which was an update to the 1979 GDP plan and addressed the latest regulatory requirements for solid waste management, accommodated elements of the County MPAH, and added flexibility for future planning of recreational uses on the site. The No Action Alternative for EIR 548 was the implementation of the 1979 GDP. Alternatives considered within EIR 548 included a Maximum Capacity Alternative, a Reduced Height Alternative, and an Interior Ridge Protection Alternative. These alternatives were eventually not selected for either negatively impacting landfill capacity, failing to accommodate the MPAH, and/or failing to meet the goal of reduced solid waste disposal.

The 2001 GDP/EIR 575 was issued as an update to EIR 548, and was also based upon implementation of the 1979 GDP as the No Project Alternative. The establishment of height restrictions to accommodate the MOU with the City of San Clemente and to remediate the 1998 landslide within Zone 1 was one of the primary project purposes for the 2001 GDP. The alternatives within the document reflect this purpose. Table 2.2-1 provides additional detail on these alternatives which include the following: no change to the 1979 GDP; no action to remediate the 1998 landslide; landslide stabilization with the re-routing of the Prima Deshecha Cañada channel south of its proposed location via a reinforced concrete pipe (RCP) culvert; landslide stabilization with the re-routing of the stream channel north of its proposed location via an RCP culvert; and stockpile on top of existing fill. These alternatives were not selected for a variety of reasons including reasons such as: a negative impact in landfill capacity; the failure to mitigate an existing flood hazard resulting in significant environmental impacts; and/or the

violation of the Conditional Use Permit (CUP) issued by the City of San Juan Capistrano and/or the MOU with the City of San Clemente.

Alternative	Source Document	Description	Analysis
No Project Landfilling Alternative	EIR 548 EIR 575	1979 Development Plan	Does not meet MPAH goals based on La Pata alignment, no Materials Recovery Facility (MRF) facility, greater impacts and less capacity than Proposed Projects.
Maximum Capacity Alternative	EIR 548	Includes re-alignment for La Pata; protects perimeter ridges; maximizes solid waste disposal	Does not provide site for MRF. Does not meet goal to reduce total solid waste for disposal.
Reduced Height Alternative	EIR 548	Lowers height of Zone 1 landfill from the 1979 GDP. Landfill footprint same. Top deck contours imitate natural contours.	Reduces landfill capacity by 44%.
Interior Ridge Protection Alternative	EIR 548	Interior ridgeline within Zone 1 is preserved resulting in 2 interior operating areas within Zone 1	Provides less solid waste disposal capacity than 1979 GDP.
No Action Landslide Remediation Alternative	EIR 575	No action to remediate slide; landfill operations restricted to the north of stream	Is infeasible due to: impacts to stream channel, creation of potential flooding hazard, material to overflow to stockpile areas off site, stability issues in Zone 1 remain, reduced landfill capacity.
Stabilize stockpile landslide, re-route natural stream south of proposed alignment, and install 96-inch RCP	EIR 575	Buttresses landslide by placing 30– 40 feet of fill over the stream and diverting water farther south downstream; requires 96-inch pipe to convey flows from the east; segments riparian corridor	Eliminated due to significant impacts to riparian habitat that was associated with design modifications for a larger landslide complex than what was anticipated.
Stabilize stockpile landslide and install a 96-inch pipe north of the stream	EIR 575	Places 1.1 million cubic yards (mcy) of fill over the stream as a buttress fill; uses 96-inch underground pipe to convey flows north of stream channel	Eliminated once data showed landslide was larger than anticipated for same reasons as indicated above for previous alternative.
Stockpile on top of Existing Fill	EIR 575	Places stockpile material on top of Zone 1	Eliminated due to violation of CUP for the site, San Juan Capistrano General Plan height restrictions, and MOU with San Clemente.

TABLE 2.2-1SUMMARY OF ALTERNATIVES ANALYZED IN EIRs 548 AND 575

Alternatives Eliminated from Detailed Consideration in EIRs 548 and 575

Several alternatives were considered but eliminated from detailed analysis for infeasibility reasons. These alternatives included unconventional approaches to waste management, including rail haul and waste-to-energy facilities (i.e., Long Haul By Rail Alternative, Transfer/Loading Station Alternative, and Unloading Station Alternative). These alternatives were not considered feasible for a variety of reasons, including: (1) inconsistency with the project objective to optimize use of the site as a long-term integrated waste management facility; (2) significant environmental effects including land use compatibility and siting; (3) air quality concerns; (4) public safety concerns; and (5) site-specific issues such as biological resources, circulation, and public services.
2.2.4 MEMORANDA OF UNDERSTANDING (MOU) BETWEEN THE COUNTY OF ORANGE AND THE CITIES OF SAN CLEMENTE AND SAN JUAN CAPISTRANO

The cities of San Juan Capistrano and San Clemente executed MOUs (Appendix C) with the County of Orange to establish formal agreements on the methods for mitigating impacts of the landfill on the respective cities. Elements of the MOU between the County of Orange and the City of San Juan Capistrano that are applicable to concerns over potential impacts of the 2001 GDP include road improvements, ridgeline preservation requirements, sound and visual berms along La Pata Avenue, a park use financing plan, and traffic circulation stipulations.

The MOU between the County of Orange and the City of San Clemente focuses primarily on the City's concern over ridgeline preservation and aesthetic impacts from viewpoints located in various residential and public viewpoints in the City. Accordingly, this MOU established project commitments regarding grading and height limits in both Zone 1 and Zone 4; zone boundary adjustment constraints relative to height limits; and requirements for interim and final landscape treatments within Zone 1. In addition to the requirements established to minimize aesthetic impacts, this agreement also established 100-year flood control design criteria for on-site drainage facilities and partnership arrangements for a watershed drainage study; water quality commitments relative to potential future on-site water needs; and circulation and landfill access commitments. Specific requirements set forth within these MOUs (that will be maintained by the Proposed Project as described in Section 4.0) are presented in Table 2.2-2.

2.2.5 AMENDMENT NO. 1 TO THE 2001 GENERAL DEVELOPMENT PLAN

Amendment No. 1 to the GDP was issued in October 2002 (Appendix D) as an outcome of negotiations between the County and the Rancho Mission Viejo Company, LLC (RMV) who owns the adjacent land to the north and east of the PDL. These negotiations resulted in an amended Settlement Agreement and a Covenant and Declaration of Restrictions ("RMV agreements") which affect 945 acres of the eastern portion of the PDL (referred to in the RMV agreements as the "Burdened Property") and specify conditions and restrictions for the site. The Landfill Operations Area (needed for future Zone 4 landfill activities) and the Restricted Area (outside the boundaries of active landfill operations) both constitute the Burdened Property (Exhibit 2.2-6). One design change to the 2001 GDP due to the RMV agreements was a reduction in the Zone 4 refuse footprint from 412 acres to 409 acres. The total Zone 4 disturbed surface area (including exterior cut slopes) remained the same at 473 acres. The agreements also identify RMV's Benefited Property and place conditions on a No-Build Area within the Benefited Property adjacent to and east of the PDL property (Exhibit 2.2-6). They also contain requirements for La Pata Avenue funding, maintenance, and litter control. Amendment No.1 to the GDP did not result in any additional environmental impacts, but resulted in additional project commitments as contained within Table 2.2-3. Requirements set forth in the County of Orange/City of Rancho Mission Viejo MOU (Appendix D), contained below in Table 2.2-3, will be maintained by the Proposed Project (Section 4.0).

TABLE 2.2-2 PERMITTED LAND USES AND VIEWSHED PROTECTION REQUIREMENTS, COUNTY OF ORANGE/SAN CLEMENTE MOU, COUNTY OF ORANGE/SAN JUAN CAPISTRANO MOU

Document	Permitted Land Uses	Viewshed Protection Requirements
MOU between the	Landfill	Grading and Height Limits.
Clemente, the Orange County Flood Control District, and the County of Orange regarding the PDL (dated July 1997)		 The final grading elevation in Zone 1 to be at a height below the level of the ridgeline behind Zone 1 as viewed from Truman Benedict Elementary School; The final grading elevation of Zone 4 to be no higher than 1,010 feet thereby minimizing the visual impact to residents of City; The boundaries of Zone 1 to be adjusted from the Zone 1 boundaries described in EIR #548 with the understanding that additional boundary adjustments may be required for Zone 1 due to geotechnical conditions, drainage, and other environmental constraints provided such adjustment does not result in a final grading plan with a height limit greater than specified above. The boundaries of Zone 4 to be adjusted as may be determined by County to be necessary to offset Landfill capacity lost due to the height limits described above provided that any additions to the Landfill resulting from the modification of the Zone 4 boundaries are not visible from within the City limits.
		Landscape Treatment
		 Trees to be planted to screen Landfill in a location which has been approved by City. The design for the treescape is described in plans and specifications dated March 23, 1997. Interim landscaping treatment to be provided consistent with erosion control measures required by current regulations. Final landscaping for Zone 1 to be installed according to regulations in effect at the time of final closure.
MOU Regarding the Prima Deshecha Landfill between the City of San Juan Capistrano and the County of Orange (dated September 1995)	Park Use Financing Plan. County agrees that the GDP will include a requirement that the ultimate recreational uses to be established on the site within City limits following closure of the landfill operations shall be mutually agreed upon by County and City. It is further agreed that the GDP will include a financial analysis and financial plan.	 <u>Ridgeline Preservation</u>. The County will install and maintain final landfill grades which result in no silhouetting above and along the General Plan- designated "major ridgeline" which forms the northern and western edges of the Landfill site boundary such that no landfill operations or placement of landfill materials will visually encroach upon the designated General Plan ridgeline or be viewed from Ortega Highway. <u>Sound/Visual Berms</u>. City and County agree to cooperate in the construction of an earthen berm to buffer residential units from noise and view of
Source: OCIWMD 19 OCIWMD 19	97. 95.	vehicles traveling to Landfill on La Pata. Said berm shall be approximately eight feet high and nine hundred feet long and designed to minimize right- of-way and to accommodate existing topography as much as possible.



R:/Projects/OCIWMD/J004/SEIR 597/Ex2.2-6_PDLRMVproperty_120605.pd

TABLE 2.2-3

PERMITTED LAND USES AND VIEWSHED PROTECTION REQUIREMENTS CONTAINED WITHIN THE COUNTY OF ORANGE/RMV SETTLEMENT AGREEMENT

Document	Permitted Land Uses	Viewshed Protection Requirements
RMV Settlement	The County is permitted to use the Restricted Area for	1. County is required to conduct a viewshed
Agreement	the following purposes:	analysis with respect to the impacts of
	1 Roads To install operate maintain repair	Zone 4's development on the RMV
	1. <u>Rodus</u> . To install, operate, maintain, repair	Benefited Property;
	maintained trails and reads in the Postricted	2. If needed, prepare and implement a
		viewshed protection and landscaping
	2 Grazing To permit livestock grazing:	plan;
	3 Utilities: Public Improvements To install	3. If required pursuant to Condition No. 5,
	operate maintain repair and/or replace (and to	the VPL [Viewshed Protection and
	grant easements or licenses to third-parties to	Landscaping Plan shall be designed to
	do the same) public utility lines and associated	provide noise, visual, litter and dust
	improvements within the Restricted Area,	screening between Landill Operations in
	including, without limitation, above-ground and	RMV Repetited Property:
	below-ground electrical transmission facilities,	A County shall confor with PMV in
	fuel and gas pipeline facilities, water, water	4. County Shall conter with Rive In developing the specific details of the VPI
	reservoirs and pipelines, sewer, and hard-wired	Plan.
	telephone, video and data transmission	5 County shall obtain RMV's approval of
	facilities, provided, however that in no event	any VPL Plan where the impacts of the
	shall any such facilities be located above ground	same reasonably affected RMV
	if the same is or would be visible from any	Benefited Property (including material.
	portion of the Benefited Property;	visual, and odor impacts).
	4. Existing Easements. To continue the use of the	
	Restricted Area for any purpose provided for in	
	any easements of record existing as of the date	
	and/or releasted any such assements:	
	5 Flood Control To install operate maintain	
	repair and/or replace flood control drainage	
	facilities and ancillary and appurtement facilities:	
	6. Hiking and Riding Trails: Other Recreational	
	Uses. Provided that any necessary permits or	
	other regulatory approvals are first obtained, to	
	install, operate, maintain, repair and/or replace	
	one or more regional and/or community hiking	
	and/or equestrian trails, parks and related	
	improvements and/or to engage in any other	
	recreational uses; provided, however, that any	
	such trails, improvements and/or uses shall be	
	located at an elevation that is at least ten feet	
	below and to the south and west of the existing	
	hadgeline located on or hear the boundary line	
	Bonofited Property:	
	7 Prescribed Fire Access To conduct prescribed	
	fires in accordance with the requirements	
	authorization or recommendation of the local fire	
	authority for purposes of the health and/or	
	safety of surrounding properties and populations	
	and otherwise to have any and all access to the	
	Restricted Area in order to fight or otherwise	
	mitigate fires;	
	8. Scientific Research. To conduct scientific	
	research;	
	9. Grading. To perform remedial grading with	
	revegetated slopes utilizing native species;	

TABLE 2.2-3 (Continued) PERMITTED LAND USES AND VIEWSHED PROTECTION REQUIREMENTS CONTAINED WITHIN THE COUNTY OF ORANGE/RMV SETTLEMENT AGREEMENT

Document	Permitted Land Uses	Viewshed Protection Requirements
	10. Mitigation. To comply with the requirements of	•
	any public agency relating to environmental,	
	geological, hydrological, archaeological,	
	paleontological, aesthetic and/or any other	
	environmental or similar land use related	
	mitigation and/or restoration measures,	
	commitments and conditions with respect to the	
	implementation of the 2001 GDP and any and	
	all other such mitigation measures and activities	
	11 Regulatory Compliance. To conduct any and all	
	activities and operations as may be necessary	
	to comply with Applicable Laws in connection	
	with the landfill's closure and post-closure	
	activities, including, but not limited to, the	
	installation, operation, maintenance, repair	
	and/or replacement of probes, monitoring wells	
	and fences; provided, however that to the extent	
	such activities and operations can be conducted	
	within the Landfill Operations Area, then such	
	activities and operations shall be conducted	
	within the Landfill Operations Area;	
	12. <u>Biological Mitigation</u> . To conduct habitat and	
	species mitigation, conservation or other similar	
	NCCP/HCP or other similar plan of	
	conservation	
	Landtill Site Grading and/or Soil Filling. To perform	
	site grading and/or soil filling to maximize capacity in	
	are not visible from the Benefited Property Such	
	activities shall not nor shall they be construed to	
	include of permit the placement of any waste within	
	the Restricted Area.	
Source: County of Ora	ange/RMV Settlement Agreement, October 2002.	

2.2.6 SUMMARY OF GENERAL DEVELOPMENT PLAN ENVIRONMENTAL IMPACTS

Following is a summary of environmental impacts from the 2001 GDP remaining at a level of significance after mitigation (as identified in FEIR 575) with subsequent changes noted as a result of supplemental documentation associated with Amendment No. 1 to the GDP.

<u>Topography</u>

FEIR 575 concluded that the landfilling component of the GDP was found to have a significant unavoidable adverse impact to topography. Implementation of the conceptual grading plan will result in significant topographic alteration of the site and the development of manufactured slopes associated with excavation and landfilling. Incorporation of mitigation measures was intended to minimize these effects by developing a grading design that reduces a manufactured appearance and is more compatible with the existing natural terrain. Implementation of Viewshed Protection Plans (VPPs) protecting surrounding viewpoints from Zone 1 and Zone 4 operations is a requirement of the MOUs and a Condition of Approval for EIR 575.

Biological Resources

FEIR 575 concluded that the landfilling, circulation, and recreational components of the project would result in: (1) long- and short-term significant impacts associated with the removal of coastal sage scrub, southern needlegrass and riparian resources, (2) potential impacts to special status plant species (such as thread-leaved brodiaea, vernal barely, paniculate tarplant, small-flowered morning glory), and potential impacts to special status wildlife species such as the least Bell's vireo and California gnatcatcher. Vegetation removal and habitat disturbance impacts of landfilling operations could affect nesting sites for listed² bird species and raptors, as well as dens for coyotes, bobcats, and mountain lions. Consultations and mitigation plans developed with the USFWS and CDFG would reduce long-term impacts to less than significant levels.

Aesthetics

FEIR 575 concluded that landfilling uses would be visible from off-site vantage points and from recreational areas around the site. Changes in topography would have the potential to impact the view from on- or off-site areas. These impacts would be reduced to a level considered less than significant via the implementation of MOU requirements (Appendix C), and viewshed protection measures designed to reduce the visibility of landfill operations to a minimum from viewpoints in adjacent housing developments.

Table 2.2-4 contains a comprehensive list of all identified potential impacts, previously adopted mitigation measures and level of significance following mitigation for the 2001 GDP and EIR 575. The following mitigation measures are currently in place for impacts as indicated below associated with the landfill component of the 2001 GDP, as identified in EIR 575 (numerical designations are from EIR 575). All mitigation commitments contained within FEIR 575 and the 2001 GDP will apply to the Proposed Project.

² Listed bird species are contained in Table 3.5-3.

		Level of Significance
AESTHETICS	I andfilling uses will be visible from off-site vantage points and from the recreational areas of the site	after Mitigation
MM 4.11-1:	Prior to final design, the IWMD, shall establish landscape standards for plantings in areas to be revegetated or screened from view. These guidelines shall illustrate all plant materials, sizes, species and quantities plus irrigation and preservation techniques. There shall be a variety of landscape types addressed including revegetating graded slopes and earthen berms, and screening of landfill operations structures and permanent landfill buildings. Roads and trail cuts will be revegetated with natural grasses, shrubs and trees to blend with the landscape character of adjacent areas. Additionally, trees selected for planting shall comply	Significant
	with the appropriate state and local regulatory requirements for the protection of groundwater.	
Impact 4.11-2:	Cut slopes and berms resulting from landfilling could be visible from on-site and off-site vantage points.	Significant
MM 4.11-2:	During final design and construction, the IWMD shall ensure that plantings will be integrated with earthen berms and cut slopes to screen undesirable views. For these situations, the landscape design guidelines shall include grading guidelines which will address issues such as the areas where berms are recommended, the sizes of such berms and recommended slope gradients to minimize soil erosion.	
Impact 4.11-3:	Permanent landfill structures could be visible from various on-site and off-site vantage points.	Significant
MM 4.11-3:	During final design, the IWMD shall incorporate design features to ensure that the design and exterior treatment of landfill operations structures and permanent recreation buildings vary in their visual character. Because of varying topography and vegetative cover, each structure and Zone will be visually unique in its apparent size and quality. Building materials shall be selected so that, in all conditions, all visible permanent structures will blend with the surrounding natural environment. A light earth tone surface color such as beige or sand is the desired exterior treatment color.	
Impact 4.11-4:	Land surfaces altered by landfilling operations in Zones 1 and 4 could be visible from various on-site and off-site vantage points.	Significant
MM 4.11-4:	As early as possible in the construction and operation of the active and closed landfill areas, the IWMD shall plant the landscape areas that will take the longest time to establish and achieve their desired visual effects. In general, rehabilitation priorities will be established based on size and visibility of the area to be landscaped. In most cases, these will be the landfilling areas in Zones 1 and 4 that are visible from adjacent land uses.	
Impact 4.11-5:	Landfilling structures located atop ridgelines on the site could result in potential visual impacts.	Significant
MM 4.11-5:	IWMD shall ensure that the design and construction of any permanent environmental control structures which occur within 200 feet of a major ridgeline are constructed in a manner which minimizes visibility off-site so as not to interrupt the natural horizon line in the existing landscape.	

AESTHETICS					Level of Significance after Mitigation
Impact 4.11-6:	Changes in topography asso the site are viewed from on-s	ciated with the landfilling may res ite or off-site areas.	ult in visual impacts	when portions of	Significant
MM 4.11-6a:	The IWMD shall ensure that the design and layout of the landfill areas includes landscaping to reduce the visual impact of the landfill surfaces following the closure of each landfill area. The IWMD shall ensure that the landscaping consists of vegetation with plantings that are consistent with the surrounding natural terrain. The IWMD shall ensure that the landscaping plantings include appropriate transitions with areas of native vegetation and areas landscaped for recreation uses. A recommended candidate plant species palette is shown in Table 4.2-1.				
	\ 200	TABLE 4.2-1 /EGETATIVE PLANTINGS 01 PRIMA DESHECHA GDP			
	Plant Species	Common Name	Pounds of Seed Per Acre		
	Artemisia californica	California sagebush	2		
	Encelia californica	California bush sunflower	3		
	Eschscholzia californica	California poppy	2		
	Lotus scoparius	deerweed	8		
	Eriogonum fasciculatum	California buckwheat	12		
	Lasthenia glabrata	goldfields	2		
	Lupinus succulentus	arroyo lupine	4		
	Collinsia heterophylla	Chinese houses	2		
	Eriophyllum confertiflorum	golden yarrow	3		
	Salvia apiana	white sage	2		
	Plantago insularis	plantain	30		
	Sisyrinchium bellum	blue-eyed grass	2		
	Diplacus longiflorus	sticky-leaved monkey-flower	2		
	Salvia mellifera	black sage	2		
	Source: Final EIR 548; November 1995				
MM 4.11-6b:	Following temporary or final closu slopes by the IWMD. Hydroseed Works Construction.	rre of landfill surfaces, hydroseeding ling shall be applied consistent witl	shall be applied to the the Standard Specif	e landfill areas and rications for Public	

		Level of Significance
AIR QUALITY		after Mitigation
Impact 4.9-1:	Potential odor nuisance could be associated with refuse brought to the landfill.	Less than Significant
MM 4.9-1:	Landfill fee station personnel and/or landfill refuse inspectors shall reject extremely odorous loads for disposal in the landfill.	
Impact 4.9-2:	A potential odor nuisance affecting recreational users could be associated with fresh refuse odor at the active face of the landfill.	Less than Significant
MM 4.9-2:	The active face of the landfill shall be covered daily. If the active face is in close proximity and upwind of on-site recreation uses, masking or neutralization agents may be added to exposed refuse to reduce the odor nuisance effects on the adjacent recreation uses.	
Impact 4.9-3:	A potential odor nuisance could be associated with LFG [landfill gas generation] emissions if they are not collected and disposed by an efficient system.	Less than Significant
MM 4.9-3:	The IWMD shall design, construct and operate new landfill areas in Zones 1 and 4 with LFG systems to maximize the collection of LFG. The LFG systems will include continuous monitoring of the LFG collection system to maximize efficient collection of LFG generated in these areas.	
Impact 4.9-4:	A potential odor nuisance could result from LFG leaks through cracks or flaws in the landfill cover in active and inactive landfill areas.	Less than Significant
MM 4.9-4:	During landfill operations, the IWMD shall continue regular visual inspections of the landfill cover and monitoring of LFG emissions throughout the entire refuse fill areas. The purpose of these inspections is to locate cracks or other defects or flaws in the landfill cover, which may allow LFG to escape. When such areas are identified, the IWMD will implement the appropriate corrective action as soon as feasible. These corrective actions may include application and compaction of additional cover material, adjustment of the existing LFG control system and/or installation of new LFG control facilities.	
Impact 4.9-5:	A potential odor nuisance could be associated with LFG generated in the active and inactive landfill areas in landfilling Zones 1 and 4.	Less than Significant
MM 4.9-5:	During landfill operations, the IWMD shall conduct periodic odor surveys on the landfill site and at various points in the area surrounding the site. The IWMD shall conduct odor surveys if any odors from the landfill are detected off site and reported by nearby residents. When the source of these odors is identified, the IWMD will implement the appropriate corrective action as soon as feasible. These corrective actions may include application and compaction of additional cover material, use of masking or neutralizing agents, adjustment of the existing LFG control system and/or installation of new LFG control facilities.	
Impact 4.9-6:	A potential dust nuisance could result to recreational users from landfilling operations.	Less than Significant
MM 4.9-6:	During landfill operations, the IWMD shall ensure that landfill operations areas that are to be left exposed temporarily, including top deck and excavation slopes, are sprayed periodically with water, as needed.	

AIR QUALITY		Level of Significance after Mitigation
Impact 4.9-7:	Exposed ground surfaces could result in a potential dust nuisance after the termination of active landfilling.	Less than Significant
MM 4.9-7:	On landfilled areas that are no longer in use, the IWMD will, as appropriate, incorporate dust control systems or vegetative covers, consistent with the Final Closure Plans and with IWMD's approved Rule 403 Compliance Plan for landfilling Zones 1 and 4.	
Impact 4.9-8:	Dusty refuse could result in a potential dust nuisance.	Less than Significant
MM 4.9-8:	During landfill operations, the landfill fee station personnel and/or landfill refuse inspectors shall refrain from accepting dusty loads of refuse for disposal in either landfilling Zone 1 or 4. Alternatively, at the discretion of landfill personnel, dusty loads of refuse may be accepted for disposal, if they are sprayed with water prior to leaving the fee station and accessing the active face of the landfill.	
Impact 4.9-9:	On-site vehicular travel on unpaved roads could result in a potential dust nuisance.	Less than Significant
MM 4.9-9a:	During landfill operations, the IWMD shall maintain water trucks on site to spray water on on-site unpaved roads as needed to minimize the generation of dust as vehicles travel on these roads, as per IWMD's approved Rule 403 Compliance Plan.	
MM 4.9-9b:	During landfill operations, the IWMD shall, to the extent feasible while still maintaining appropriate landfill operations, restrict vehicular travel on unpaved roads on the site. In the event that unpaved roads must be used, the IWMD shall spray water on these roads as needed.	
MM 4.9-9c:	As unpaved on-site roads are removed from active service, the IWMD will spray these areas with a hydromulch solution or synthetic binder.	
Impact 4.9-10:	A potential dust nuisance could be associated with graded areas or areas where the vegetation has been removed or severely disturbed.	Less than Significant
MM 4.9-10:	During landfill operations, the IWMD will use the on-site water trucks to spray water on graded areas or areas where the vegetation has been removed or severely disturbed as a result of landfilling activities, as per IWMD's approved Rule 403 Compliance Plan.	

	Level of Significance	
BIOLOGICAL RE	after Mitigation	
Impact 4.5-1:	Removal of needlegrass grassland will result from site clearing prior to construction of landfill improvements on the Prima Deshecha site.	Less than Significant
MM 4.5-1:	The restoration of needlegrass grasslands will be incorporated into the Conceptual Coastal Sage Scrub Mitigation Plan (described below in MM 4.5-2a through 2c), the IWMD will replace impacted needlegrass grassland at a 1:1 ratio.	
Impact 4.5-2:	Removal of coastal sage scrub will result from site clearing prior to construction of landfill improvements on the Prima Deshecha site.	Short-term Significant
MM 4.5-2a:	Prior to the removal of coastal sage scrub habitat resources including clearing, grubbing, mowing, discing, trenching, grading, fuel modification, or other construction related activities, the Director, IWMD or his designee shall prepare and submit, in consultation with the Planning and Development Services Department (PDSD) Director of Planning or his designee, an Interim Habitat Loss Mitigation Plan (IHLMP) to the USFWS for review and approval in compliance with the Natural Communities Conservation Plan (NCCP) and Interim Coastal Sage Scrub (CSS) Habitat Loss Process. The County remains committed to the NCCP process and intends to operate by the same procedure outlined in the Federal Endangered Species Act Section 4(d) Special Rule for Incidental Take of the coastal California gnatcatcher or other agreement as determined to be appropriate by the resource agencies.	
MM 4.5-2b:	The GDP shall be amended to include all applicable provisions of the approved Southern Subregion NCCP on its adoption by the County of Orange Board of Supervisors. The NCCP implementation programs may include, but are not limited to, requirements for the removal and mitigation replacement of lost coastal sage scrub habitat, operations restrictions, instructional signs, fencing, etc.	
MM 4.5-2c:	In accordance with an approved Conceptual Coastal Sage Scrub Mitigation Plan, the IWMD shall replace impacted coastal sage scrub at a minimum 1:1 (or as otherwise stated by USFWS) replacement ratio of in- kind habitat for onsite and offsite habitat preservation, replacement, or enhancement.	
	The IWMD shall prepare a Conceptual Coastal Sage Scrub Mitigation Plan in cooperation with the affected resource agencies (CDFG, USFWS). Guidelines for the Mitigation Plan shall be as follows:	
	 The mitigation areas/sites shall have been evaluated and selected on the basis of their suitability for use as coastal sage scrub revegetation areas. The parameters evaluated shall include but not be limited to soil condition, slope aspect, proximity to adjacent coastal sage scrub, level of difficulty of site preparation, and ownership status. 	

BIOLOGICAL RESOURCES	Level of Significance after Mitigation
 The mitigation plan shall provide procedures to prepare the soils in the mitigation area, provide detailed seeding/planting mixtures; provide seeding/planting methods; and provide any other procedures, such as supplemental irrigation, mycorrhizal inoculation, etc., that will be used for successful revegetation. 	
 Maintenance and monitoring goals shall be established. The components and implementation of the maintenance and monitoring procedures shall be consistent with the components and implementation of mitigation measure 4.5-7a. 	
In accordance with the approved Conceptual Coastal Sage Scrub Mitigation Plan, the IWMD shall develop a maintenance and monitoring program to ensure success of the revegetation effort. Maintenance shall include regular inspection of the site for excessive weed growth, erosion problems, failure of irrigation system, and/or unhealthy or dying plants. Invasion of the site by weeds in the area, especially pampas grass, artichoke thistle, castor bean, fountain grass, mustard, clover, cocklebur, and tree tobacco could be a potential maintenance problem. Maintenance crews shall be able to recognize the difference between native plant and weed seedlings. A qualified biologist will be required to instruct the maintenance crew in the identification of native plant seedlings. The maintenance program shall include procedures for regular maintenance and repair of the irrigation system.	
A system shall be developed for reporting by the maintenance crew of any unhealthy or dying plantings or failure in any of the seeded areas. This would assist the monitoring crew in the development of immediate remedial measures, such replacing plant material, to correct the problem.	
To document the success of revegetation programs, the IWMD shall ensure that the progress of the revegetated area is monitored by a qualified biologist. The maintenance and monitoring plan will address unique aspects of mitigation areas. An agreement shall be developed between the County and the USFWS and CDFG on criteria that will be used to determine successful plant establishment on a mitigation site. Success criteria will include plant cover, species diversity, habitat structure, and density, and will be based on measurements made in reference habitats near the mitigation site.	

	Level of Significance	
Impact 4.5-3:	The removal of riparian resources will occur as a result of the construction of landfill improvements on the Prima Deshecha site.	Short-term Significant
MM 4.5-3a:	Prior to grading for the landfilling activities affecting riparian resources, the IWMD, as appropriate, shall ensure that all sycamore and willow trees of four or more inches in diameter at breast height (DBH), defined as 4.5 feet from mean ground level, within the grading or construction limits of the landfilling activities, whichever is greater, and within 100 feet of grading and construction operations, shall be tagged and numbered with permanent tags under the supervision of a qualified biologist. The tag numbers of the trees to be protected and those to be removed shall be noted. Those trees adjacent to the construction areas that can be avoided will be tagged for protection. Trees that cannot be avoided during construction shall be tagged for removal and fenced off with red-orange flexible mesh fencing during grading and construction activities. Records of these numbers shall be kept by the Director, IWMD or his designee for use in mitigation, replacement, and monitoring of tree resources before, during, and after grading and construction activities. In addition, prior to grading and site preparation, the IWMD shall ensure that all trees subject to removal are marked with a red "X" on the trunk. Trees to be preserved shall be marked with yellow flagging visible from all directions and fenced-off with red-orange flexible mesh fencing during grading and construction activities.	
MM 4.5-3b:	During the process of obtaining the required 404 Permit Application and 1601 Streambed Alteration Agreement (1601/404) for encroachment into streambed areas and prior to site preparation, the IWMD shall prepare a Conceptual Riparian Mitigation Plan in cooperation with the affected resource agencies (CDFG, USFWS, and the USACE). Guidelines for the Mitigation Plan shall be as follows:	
	 The mitigation sites will be evaluated and selected on the basis of their suitability for use as riparian revegetation. The parameters evaluated shall include but not be limited to soil condition, hydrology, geology, and drainage considerations, level of difficulty of site preparation, access, contiguousness with existing habitat, and ownership status. 	
	• The mitigation plan shall include the procedures for soil preparation, provide seeding/planting mixtures; include seeding/planting methods; and include any other procedures, such as supplemental irrigation, mycorrhizal inoculation, etc., that will be used.	
	 Maintenance and monitoring goals shall be established. The components and implementation of the maintenance and monitoring assignments shall be consistent with the components and implementation of mitigation measure 4.5-3d. 	
MM 4.5-3c:	In accordance with an approved Conceptual Riparian Mitigation Plan, the IWMD shall replace impacted riparian areas at a minimum 2:1 or higher ratio of in-kind or higher quality habitat. The required replacement acreage will be approved by the resource agencies having jurisdiction over the impacted resources (i.e., CDFG, USACE, and/or USFWS), for all the GDP uses, based on jurisdictional delineations and vegetation mapping and the current 2001 GDP grading plan.	

		Level of Significance
BIOLOGICAL	after Mitigation	
MM 4.5-3d:	During the process of obtaining the 404 Permit and 1601 Streambed Alteration Agreement, and, in accordance with the approved Conceptual Riparian Mitigation Plan, the IWMD shall develop a maintenance and monitoring program to ensure success of any revegetation effort. Maintenance shall include regular inspection of the site for excessive weed growth, erosion problems, failure of irrigation system, and/or unhealthy or dying plants. Invasion of the site by weeds in the area, especially pampas grass, artichoke thistle, mustard, clover, castor bean, fountain grass, cocklebur, and tree tobacco could be a potential maintenance problem. Maintenance crews shall be able to recognize the difference between native plant and weed seedlings. A qualified biologist will be required to instruct the maintenance crew in the identification of native plant seedlings. The maintenance program shall include procedures for regular maintenance and repair of the irrigation system.	Short-term Significant
	A system shall be developed for reporting by the maintenance crew of any unhealthy or dying plantings or failure in any of the seeded areas. This would assist the monitoring crew in the development of immediate remedial measures, such as increasing the irrigation rate or replacing plant material, to correct the problem.	
	To document the success of revegetation programs, the IWMD shall ensure that the progress of the revegetated area is monitored by a qualified biologist. An agreement shall be developed between the County and the USACE, USFWS, or CDFG on criteria that will be used to determine successful plant establishment on a mitigation site. These criteria will include plant cover, and density, and will be based on measurements made in reference habitats near the mitigation site.	
	The qualified biologist shall monitor the site for five years or until the site complies with required performance standards. If the biologist determines that the mitigation site meets the conditions of the performance criteria prior to the five-year period, documentation shall be submitted to the responsible agency for approval.	
MM 4.5-3e:	Prior to grading and site preparation adjacent to riparian areas outside the limits of construction, the IWMD shall incorporate instructions in the construction documents ensuring that, in conjunction with construction activities:	
	Graded material spoils shall not be placed or stored near riparian areas outside the limits of construction.	
	The removal of streamside or bank vegetation shall be avoided wherever feasible.	
	• The amount of habitat removed shall be limited to the minimum amount required for construction.	
	• Riparian areas in the vicinity of grading or heavy recreation use, such as in Zone 1, shall be designated as Environmentally Sensitive Areas onsite preparation, grading, and construction plans, and fenced off as appropriate for protection before any of these activities begin.	
	Excess fill shall not be dumped in streams outside the limits of construction.	
	• Vehicles and equipment shall not be parked in washes or other drainages outside the limits of construction.	

		Level of Significance
BIOLOGICAL R Impact 4.5-4:	ESOURCES Special status habitats and special status species could exist within the construction limits of the landfilling improvements and could be adversely affected by the proposed landfill improvements.	after Mitigation Short-term Significant
MM 4.5-4a:	Prior to site preparation and during final design for each phase of landfill development (i.e., Phases A–D in Zone 1 and Phases A–I in Zone 4), the Director IWMD shall ensure that focused surveys are conducted by qualified biologists for the thread-leaved brodiaea, Coulter's saltbush, many-stemmed dudleya, southern tarplant, vernal barley, paniculate tarplant, and any other plant species that may warrant focused surveys in the future as determined by a qualified botanist. In addition, the Director IWMD shall ensure that focused surveys are conducted by qualified biologists for the western spadefoot toad, southwestern willow flycatcher, and other wildlife species that may warrant focused surveys in the future as determined by a qualified biologist. The results of the surveys shall be incorporated into environmental documentation for future proposed projects within the Prima Deshecha site. Identified special status species and habitats located within 300 feet of the affected area(s) shall be mapped on grading plans for each phase of development. In addition, the Director IWMD shall implement procedures approved by the appropriate resource agencies to mitigate the potential impacts to those species. In the event that landfill activities within a phase must occur prior to the completion of spring surveys, habitat for the special status plant species shall be salvaged, stored, and used in an appropriate manner as determined by a qualified biologist. The appropriate agencies will be notified prior to disturbance. All future proposed projects within the Prima Deshecha Landfill shall provide vegetation mapping on topographic maps at a scale of 1 inch equals 200 feet.	
MM 4.5-4b:	The IWMD shall ensure that, for the periods covering all site preparation, disturbance or grading of native areas, the Director, IWMD or his designee shall monitor wildlife habitat preservation. The purpose of this monitoring is to ensure that the Environmentally Sensitive Areas and Environmentally Restrictive Areas (i.e., areas outside the grading limits) will not be adversely impacted during site preparation, grading, and construction of the landfilling activities.	
	For the landfilling activities, this inspection program shall be coordinated with the Site Manager at the weekly meetings held at the Landfill to review the planned grading program for the landfilling activities. These meetings shall commence at the start of each new phase, when native ground is schedule for disturbance (e.g., grading or stockpiling, etc.). The Director, IWMD or his designee will attend these meetings and provide a status and progress report to the Operations Manager. These meetings will be held throughout the site preparation, grading and construction periods for all the landfilling activities and the monitoring reports shall continue to be prepared and submitted by the Director, IWMD or his designee until the disturbance is completed.	
	The monitor shall be on site before, during, and after the completion of site preparation, grading, and construction for all the landfilling activities.	

BIOLOGICAL R	ESOURCES	Level of Significance after Mitigation
Impact 4.5-5:	Potential vegetation removal and habitat disturbance impacts of the landfilling uses could affect nesting sites for listed bird species and raptors, and dens for coyotes, bobcats, and mountain lions.	Short-term Significant
MM 4.5-5:	During site preparation and grading for the landfill, the IWMD shall phase these operations outside significant habitat areas during the nesting and breeding season for the coastal California gnatcatcher. This measure shall be overseen and conducted by a qualified biologist.	
	During site preparation and grading for the landfill, the IWMD shall phase these operations outside significant habitat areas during the nesting and breeding season for the least Bell's vireo. This measure shall be overseen and conducted by a qualified biologist. Prior to activities that may impact potential vireo habitat, updated vireo surveys will be conducted by a qualified biologist.	
	The IWMD shall ensure that grading and construction operations for the landfilling are redirected temporarily around nesting sites for a distance of 500 feet for candidate and listed species of birds and a distance of 1,000 feet for raptors during nesting and breeding seasons between February 15 and July 15, or a distance and time period agreed upon by the USFWS. In the event that a coyote, bobcat, or mountain lion den is located, then grading and construction operations shall be redirected temporarily around the den for a distance of 1,000 feet. The nesting sites and dens should be resurveyed toward the end of the breeding seasons of these species to verify completion of the breeding cycle. Nests and dens that will be removed due to the grading and/or construction operations shall be removed only during the non-breeding season.	
Impact 4.5-6:	Potential indirect noise, air quality, and lighting disturbance impacts on biological resources could be associated with the landfilling activities.	Less than Significant.
MM 4.5-6:	The IWMD shall ensure that during final design, the landfill operation continues to incorporate regulatory agency guidelines to reduce indirect impacts associated with noise, dust, night lighting, and blowing debris. Noise shall be controlled through the proper maintenance of the construction equipment, including trucks, bulldozers, and other mobile and fixed construction equipment. Dust shall be controlled at its source with standard wetting techniques consistent with applicable Southern California Air Quality Management District (SCAQMD) requirements. Low lighting alternatives and shielded lighting shall be employed to reduce indirect impacts on surrounding habitats.	

CULTURAL/SCI	ENTIFIC RESOURCES	Level of Significance after Mitigation
Archaeological	Resources	¥
Impact 4.6-1:	Known and potential archaeological resources could be disrupted or removed as a result of the construction of the GDP landfilling improvements.	Less than Signficant
MM 4.6-1:	Prior to the initiation of any site modifications, the IWMD shall contract with a County-certified archeologist who will prepare a Testing, Monitoring and Salvage Program for Archaeological Resources for the GDP landfilling activities. The Plan shall identify the specific pre-disturbance subsurface testing program and the specific monitoring procedures, scheduling, staffing and other elements to ensure adequate testing, identification and salvage of archaeological resources prior to and during grading, site preparation, earth moving and excavation activities associated with the GDP landfilling activities. The Plan shall also identify procedures for in-place preservation of resources including the identification of typical resources that would be preserved in-place. The Plan shall also establish the authority for halting or temporarily relocating construction during preservation activities and other procedures as necessary.	
Paleontological	Resources	
Impact 4.6-2:	Known and potential paleontological resources could be disrupted or removed as a result of the construction of the GDP landfilling uses.	Less than Significant
MM 4.6-2:	Prior to the initiation of any site modifications, the IWMD shall contract with a County-certified paleontologist who will prepare a Monitoring and Salvage Plan for Paleontological Resources for the GDP landfilling activities. The Plan will identify the specific monitoring procedures, scheduling, staffing and other elements to ensure adequate identification and salvage of fossil materials during grading, site preparation, earth moving and excavation activities associated with the GDP landfilling activities. The Plan shall also identify procedures for in-place preservation of resources including the identification of typical resources that would be preserved in-place. The Plan shall also establish the authority for halting or temporarily relocating construction during the preservation activities and other procedures as necessary.	

		Level of Significance
GEOPHYSICAL		after Mitigation
Impact 4.2-1:	Hillside excavations for construction of landfill refuse capacity and daily cover soil could expose people or property to landslide or mudslide activity during the excavation period.	Less than Significant
MM 4.2-1a:	Prior to designing each phased landfill plan and specifications, the IWMD shall conduct a geotechnical investigation to determine the extent of landslide material and the soil foundation characteristics of the proposed phase. A geotechnical report of the phased site area shall be prepared which includes a landslide excavation and removal plan prepared to the satisfaction of the Director, IWMD.	
MM 4.2.1b:	For each phased grading plan, the excavation and grading plan shall ensure the stability of all cut, fill and lined slopes. Slopes shall be designed to withstand the most probable earthquake based on a return period of 100 years or as required by current regulations. Liner design plans shall be submitted to the San Diego Regional Water Quality Control Board for approval. The plans shall also be incorporated in a JTD [Joint Technical Document] and submitted to the LEA for approval and to the CIWMB for concurrence.	
Impact 4.2-2:	Seismic activity occurring along any of the active regional faults could subject the landfill property to seismic shaking which could damage landfill facilities and/or structures.	Less than Significant
MM 4.2-2a:	The IWMD shall demonstrate that landfill design plans comply with the state and federal seismic requirements in CCR Title 27, and 40 Code of Federal Regulations (CFR) §258.14 (Seismic Impact Zones) and §258.15 (Unstable Areas). These demonstrations shall be incorporated in the IWMD Operating Record prior to construction of said plans.	
MM 4.2-2b:	Prior to commencement of daily excavations for borrow material, grading plans shall be prepared, analyzed for slope stability and submitted for approval by the Director, IWMD, or his designee.	
MM 4.2-2c:	As part of a JTD, the IWMD shall present the assumptions, methods and calculations used to demonstrate seismic safety. This measure is required only if final slopes are planned to be steeper than a ratio of 3:1 (horizontal to vertical), if the site is located in an area subject to liquefaction or in unstable areas with poor foundation conditions as described in the Seismic Safety Element of the Orange County General Plan (27 CCR 17777).	
Impact 4.2-3:	Differential settlement associated with compression and decomposition of solid waste materials can be expected on the order of 30 percent of the total refuse thickness.	Less than Significant
MM 4.2.3:	As part of a JTD, the IWMD shall present the assumptions, methods and calculations used to demonstrate that differential settlement of the site will not result in future environmental impacts (27 CCR 21090).	

		Level of Significance
GEOPHYSICAL		after Mitigation
Impact 4.2-4:	Continued use of the site for landfill purposes will create the demand for soil to be used as cover material.	Less than Significant
MM 4.2-4:	When the JTD is prepared the IWMD shall identify the assumptions, methods and calculations performed to demonstrate that the excavation plans provide for sufficient quantities and sources of suitable soils or alternative cover systems for daily and intermediate cover, final cover and liner materials. This section of the JTD should also reference and summarize any borrow studies conducted to demonstrate the availability of sufficient quantities of materials. If materials are obtained on site, the description shall include which sections of the site will be excavated for each sequence of landfilling and where these materials will be stockpiled for use. Stockpile locations should not interfere with unloading, spreading, compacting, access, safety, drainage or other operations on the site. Stockpiles should be clearly shown on the fill sequencing and excavation plans prepared for construction. (27 CCR 21600).	
HAZARDS/HAZA	ARDOUS MATERIALS	Level of Significance after Mitigation
Vehicle Access	and Circulation	
Impact 4.13.1-1:	Potential conflicts could occur between refuse and recreation vehicles using the same access roads on the site.	Less than Significant
MM 4.13.1-1:	Prior to opening any recreation uses on-site, the IWMD and the RDMD/HBP shall develop and implement site operating procedures that separate refuse and recreation vehicles either by separate access routes or separate internal circulation patterns immediately after accessing the site.	
Impact 4.13.1-2:	On-site traffic controls will be necessary to regulate and direct the flow of vehicular traffic.	Less than Significant
MM 4.13.1-2:	The IWMD shall continue to implement on-site traffic operations procedures regarding on-site posted traffic speed limits and traffic controls for the landfilling operations in Zones 1 and 4.	
Impact 4.13.1-3:	Potential conflicts could arise between construction vehicles and ordinary vehicular traffic on the site during the construction of the landfilling uses.	Less than Significant
MM 4.13.1-3:	Prior to the approval of construction plans, the IWMD shall ensure that construction activities for the landfilling uses which may temporarily bring construction equipment and ordinary vehicular traffic into closer contact will continue to be mitigated by traffic control consisting of limiting access of vehicular traffic to construction areas. The traffic control plans for the 2001 GDP construction areas shall be consistent with existing RDMD/Road Programs traffic control policies and procedures.	

	RDOUS MATERIALS	Level of Significance
Hazardous Mater	ials	
Impact 4.13.2-1:	Potential impacts are associated with hazardous materials improperly brought to the landfill for disposal in the Zones 1 and 4 landfill areas.	Less than Significant
MM 4.13.2-1:	The IWMD will continue to implement its policy not to accept hazardous materials for disposal in the landfill. This policy will include, but not be limited to, visual inspection of loads at the fee booth and on the active face of the landfill during unloading; continued operation of the radiation detection systems at the fee booths; and landfill personnel recording the license plates of vehicles turned away at the entrance.	
Impact 4.13.2-2:	Potential public safety impacts could result from incomplete separation of refuse vehicles and biosolids disposal from recreational visitors.	Less than Significant
MM 4.13.2-2:	Prior to opening any recreation uses on-site, the IWMD and RDMD/HBP shall develop and implement on-site operating procedures that separate the recreation users and trash vehicles as they enter the site and that no members of the public are allowed access to the landfill areas in Zones 1 and 4 where mixing operations and disposal of biosolids with other refuse on the active face of the landfill occur.	
Impact 4.13.2-3:	Potential public safety impacts are associated with the disposal of automobile shredder waste (ASW) in the landfills in Zones 1 and 4.	Less than Significant
MM 4.13.2-3:	The IWMD shall maintain and implement operating procedures for the acceptance and disposal of non- hazardous ASW, including documentation of all ASW loads received at the landfill.	
Impact 4.13.2-4:	Hazardous wastes could be generated by the on-site maintenance of landfill vehicles.	Less than Significant
MM 4.13.2-4:	The IWMD shall continue to maintain operating procedures for the safe handling and removal of waste oil and other potentially hazardous waste materials from the on-site vehicle maintenance facility.	
Impact 4.13.2-5:	The potential exists for explosion, fire or personal exposure to hazardous materials to affect landfill workers and members of the general public during the operation of the household hazardous waste (HHW) collection station.	Less than Significant
MM 4.13.2-5:	The IWMD shall maximize protection of the public and landfill workers from accidental exposure to hazardous materials at the HHWCC, consistent with all applicable state and federal regulations. These measures shall include, but not be limited to, separation of recreation users from the HHWCC; proper handling and disposal of the HHW collected at the HHWCC; and on-site emergency response personnel and equipment.	
Fire Safety and C	Control	
Impact 4.13.3-1:	Potential fire and fire safety impacts are associated with the landfill operations in Zones 1 and 4.	Less than Significant
MM 4.13.3-1:	The IWMD shall maintain on-site operating procedures for the avoidance and control of surface fires. These practices shall include, but not be limited to, the provision of fire extinguishers and watering vehicles, posting of No Smoking signs, ground clearing and general safe operating practices.	

		Level of Significance
Impact 4.13.3-2:	Given the characteristically dry climatic conditions and native vegetation, potential fire risks could be associated with public access to the site and public access roads through the site.	Less than Significant
MM 4.13.3-2a:	Prior to the opening of public access roads on-site, the IWMD shall coordinate with the RDMD/Road Programs on the placement of fire warning signs along public roadways through the site, warning motorists of potential fire hazards, fire conditions and other relevant information.	
MM 4.13.3-2b:	The IWMD and the RDMD/Road Programs will ensure that all roads serving landfilling activities include road signs warning motorists and landfill patrons of potential fire hazards, fire conditions and other relevant information. This signing shall be consistent with the requirements of the County of Orange for roadway signing.	
MM 4.13.3-2c:	Prior to approval of construction plans, the IWMD shall ensure that all construction contractors and employees engaged in construction for the landfilling uses implement safe working practices regarding the potential for surface fires associated with construction equipment and personal vehicles. These practices, subject to the approval of the Orange County Fire Authority, shall include the installation of spark arresters on equipment that has the potential to emit sparks or glowing embers; avoiding parking vehicles in areas with high or very dry vegetation; restrictions on employee smoking; the use of open flames or fire in high hazard areas and other similar safe working practices.	
Landfill Gas Gene	eration	
Impact 4.13.4-1:	Potential impacts associated with the generation and surface migration of landfill gas could result from landfilling activities in Zones 1 and 4.	Less than Significant
MM 4.13.4-1:	The IWMD shall continue to ensure that the design and operation of the GDP landfilling activities include a LFG control system consisting of a network of collection wells, flare stacks and ERF [energy recovery facility] capacity as needed as LFG generation increases, and a monitoring program, basically expanding the existing LFG control system on-site.	
Impact 4.13.4-2:	Potential explosion hazards could result from the accumulation of landfill gas in buildings associated with the landfilling or recreation uses.	Less than Significant
MM 4.13.4-2:	Prior to the issuance of building permits and during structure siting and final design, the IWMD and RDMD/HBP shall ensure, as part of the structure siting and final design, that the construction of permanent structures with enclosed spaces on landfilled areas will not occur unless the building is designed with protection from migrating landfill gas approved by the Solid Waste Local Enforcement Agency. Such protection designs could include: gas impermeable membrane underlying the structure and/or venting of enclosed spaces in the building, particularly spaces in contact with the ground or building foundation. In addition, the building designs will incorporate an explosive gas alarm system where this would be considered to increase the overall safety of the building for occupants or users of the building.	

HYDROLOGY ANI	D WATER QUALITY	Level of Significance after Mitigation
Impact 4.2-5:	The potential exists for landfill leachate migration into groundwater through fractured and porous alluvium.	Less than Significant
MM 4.2-5a:	The IWMD shall continue to operate its existing leachate control system within the active landfill area. In addition, the IWMD shall be required to construct a corresponding leachate control and recovery system in those areas where new liners are constructed and in areas added to the active landfill area.	
MM 4.2-5b:	The site shall continue to operate under the groundwater monitoring requirements contained in Waste Discharge Requirements, Order No. 89-102, Technical Change Order (TCO) No. 1, Amended Waste Discharge Requirements contained in Order No. 93-86, and any future orders issued by the San Diego Regional Water Quality Control Board. TCO No. 1 contains the detailed Groundwater and Vadose Zone Monitoring Program for the Prima Deshecha Landfill.	
MM 4.2-5c:	As part of a revised JTD, the IWMD shall present the assumptions, methods and calculations used to predict leachate generation and sizing of the components of the leachate collection system.	
Impact 4.3-1:	Modification of the landfill surface will alter the surface hydrology of the site and could result in increases in stormwater runoff.	Less than Significant
MM 4.3-1a:	As part of a Joint Technical Document (JTD) to be prepared by IWMD, the IWMD shall present the assumptions, methods and calculations used to calculate the potential flow quantities for run-on, runoff, and sediment content of storm water flow used in sizing drainage and sediment control facilities.	
MM 4.3-1b:	As part of a JTD to be prepared by IWMD, the IWMD shall include surface drainage plans for final fill and bottom excavation plans, including any berms, down drain systems, storm drain systems, direction of flow in perimeter drainage channels, and the location of off-site discharge point for runoff water.	
MM 4.3-1c:	Detention, diversion, and drainage facilities shall be designed and constructed to accommodate the anticipated volume of precipitation and peak flows from surface runoff under the precipitation conditions specified in §20365 of Title 27 of the California Code of Regulations for each class of waste management unit (WMU). In addition, drainage facilities for WMUs shall be designed to prevent washout of the WMUs during a 100-year storm event.	
HYDROLOGY ANI	D WATER QUALITY	Level of Significance after Mitigation
Impact 4.4-1:	The potential exists for surface water quality degradation from landfilling.	Less than Significant
MM 4.4-1a:	The IWMD shall comply with its National Pollutant Discharge Elimination System (NPDES) Storm Water Pollution Prevention Plan (SWPPP) and its NPDES Monitoring and Reporting Plan for the landfilling under the GDP. This plan will ensure that the measures taken to safeguard surface water quality are effective and are being correctly employed.	
MM 4.4-1b:	The IWMD shall continue to implement the existing Surface Water Runoff Monitoring Program as described in the currently effective Waste Discharge Requirements.	

Impact 4.4-2:	Potential erosion associated with construction of the landfill-related projects could increase the silt load in surface waters.	Less than Significant
MM 4.4-2:	As part of the NPDES program and prior to approval of construction contracts, the Director, IWMD, or a designee, shall ensure that silt loading to surface waters from the construction activities will be periodically tested and controlled, where necessary, by appropriate erosion control measures, siltation basins or other settling structures.	
LAND USE AND	PLANNING	Level of Significance after Mitigation
Impact 4.7-1:	Landfill disposal areas, permanent flare stations and other temporary and permanent landfill facilities may be visible from sensitive off-site viewsheds and/or above protected ridgelines.	Less than Significant
MM 4.7-1:	During final design and implementation of the GDP landfilling activities, the IWMD shall ensure, to the extent feasible and that funding is available, that the landfill disposal areas and associated permanent and temporary landfilling facilities are sited so as to minimize visibility from beyond the site, particularly with regard to ridgelines protected by ordinances in the cities of San Clemente and San Juan Capistrano. For landfill areas and/or facilities not able to be sited below intervening protected ridgelines, options for reducing or minimizing views of operations and facilities from off-site sensitive viewsheds may include retention of natural topography, landscaping, berms and other methods as feasible and as funding is available.	
Impact 4.7-2:	Landfill activities could occur within the 200-foot ridgeline protection buffer between the landfill footprint and protected ridgelines, as established by the cities of San Clemente and San Juan Capistrano.	Less than Significant
MM 4.7.2:	Above-ground landfill facilities within the 200-foot "major ridgeline" protection zone established by the City shall be prohibited with the exception of regional, above-ground monitoring and maintenance facilities (e.g., risers, check valves, etc.) less than five (5) feet in height. Non-landfill facilities or structures shall be prohibited within the 200-foot "major ridgeline" protection zone.	
LAND USE AND	PLANNING	Level of Significance after Mitigation
Impact 4.7-3:	The flare station facilities could be visible from off-site vantage points.	Less than Significant
MM 4.7-3:	Prior to the completion and approval of construction plans, the IWMD shall ensure that the design of the flare stations needed under the GDP landfilling activities incorporates the following types of features to reduce the visual effect of these facilities:	
	 Landscaping around the flare stations will be developed to allow for a natural appearance of the area. Cut and fill areas resulting from the construction of the flare stations will be gently contoured consistent with the area topography and will be landscaped. 	
	 The flare stacks and other flare station facilities will be painted light brown colors, similar to the existing flare station facilities. 	

NOISE		Level of Significance
Impact 4.10-1:	Landfilling construction machinery and landfilling equipment could result in potential noise impacts on surrounding land uses.	Less than Significant
MM 4.10-1:	Although the construction associated with landfilling under the GDP is not anticipated to result in significant noise impacts on residential uses adjacent to the site, the IWMD shall reduce landfill operations noise impacts to the extent feasible based on available funds through the use of landscaping, berms at the face of each landfill lift, phased construction of the landfill areas and the use of buffer areas between noise sources and sensitive recreation receptors.	
		Level of Significance
Electricity		after mitigation
Impact 4.16-1:	Potential disruption of existing Southern California Edison (SCE) and San Diego Gas and Electric (SDG&E) facilities during construction of the GDP landfilling uses.	Less than Significant
MM 4.16-1:	Prior to approval of construction and grading plans, the IWMD will include, as part of the construction documents, requirements that the construction contractors coordinate with SCE and SDG&E to ensure that their facilities on the site are protected to prevent significant disruption to utility services during construction. The contractor will be required to provide written documentation of this coordination to the IWMD.	
UTILITIES AND SE	ERVICE SYSTEMS	Level of Significance after Mitigation
Santa Fe Pipeline	Company	¥
Impact 4.16-2:	The Santa Fe Pacific Pipeline Partners Inc., oil pipelines on the east side of the site could be impacted during construction of the landfilling uses.	Less than Significant
MM 4.16-2:	The IWMD will coordinate with Santa Fe Pacific Pipeline Partners Inc., during final design of the landfilling uses in Zone 4 regarding the precise location and depth of the existing pipelines on the site. The IWMD shall coordinate the landfill construction schedules with Santa Fe Pacific Pipeline Partners Inc., to allow the company to relocate its pipelines, if needed, prior to IWMD initiating construction of landfilling improvements in Zone 4 that would otherwise impact these pipeline facilities.	
Natural Gas, Pota	ble Water, Non-Potable Water, Sanitary Facilities, Storm Drains, Telephone Service and Schools	
No impacts on nation the 2001 GDP land	ural gas, potable water, non-potable water, storm drains, telephone service and schools were expected under filling activities and therefore no mitigation was required.	Less than Significant
Sanitary Facilities		
Impact 4.16-3:	No wastewater facilities are available to the site in the Santa Margarita Water District service area; therefore, all GDP landfill operations sanitary facilities would be septic tank systems, similar to the current operations at the landfill.	Less than Significant
MM 4.16-3a:	Prior to the commencement of any landfilling operations, a soils report and plans for all sewage disposal	

	systems shall be submitted to the County's Plumbing/Mechanical Plan Checking Section for review and approval.
MM 4.16-3b:	Results of percolation tests and a log of soil borings, performed and reported by a Registered Environmental Health Specialist, Registered Civil Engineer or Registered Geologist, in accordance with Environmental Health's "On-Site Sewage Disposal System Guidelines" shall be submitted to the County's Plumbing/Mechanical Plan Checking Section for review and approval. The Land Use Unit of Environmental Health shall be notified at least 48 hours prior to soil testing in order to be present during testing, if deemed necessary.
MM 4.16-3c	Each proposed individual sewage disposal system shall be designed in accordance with Environmental Health's "On-Site Disposal System Guidelines."
MM 4.16-3d	An additional soil percolation system, equal to a maximum of 100 percent of the original design capacity or as deemed necessary by the Manager, Environmental Health, shall be constructed and connected.

2.3 EXISTING OPERATIONS

The permitted environmental monitoring and control facilities located on site to support landfill operations are listed below:

- An energy recovery facility (ERF) that converts landfill gas to electricity
- A Household Hazardous Waste Collection Center (HHWCC) and a facility for the temporary storage of hazardous materials
- A landfill gas collection and flaring system for the site, which consists of horizontal collection piping and vertical extraction wells and a flaring facility
- Landfill Gas Condensate collection system
- Perimeter probes for landfill gas monitoring
- Groundwater monitoring wells located in the vicinity of the current and future refuse disposal areas
- A groundwater extraction system, located downstream of WMU1 in Zone 1, consisting of a pump station and four groundwater extraction wells within the alluvial material of the Prima Deshecha Cañada watercourse
- A leachate collection and recovery system (LCRS) for the lined areas
- Perimeter and interior drainage facilities
- A major detention/desilting basin and interim desilting/stilling basins for Zone 1
- Biological mitigation sites to the east and south of Zone 1.

2.3.1 THE 2001 GDP, EXISTING PDL PERMITS, AND SUBSEQUENT ACTIONS

Permits covering activities at the PDL site to date consist of landfill operating permits and resource agency permits that were issued for compliance with various environmental statutes and State and Federal regulations covering landfill facilities. A more detailed description of these permits is included in Section 4.3.3. Regulatory Requirements for Implementation. Proposed Project elements may trigger either new permits or a modification of these existing permits as is discussed further in Section 4.3.3.

Resource Agency Permits

The following permits have been issued to the IWMD by the state and federal regulatory agencies that have jurisdiction over the project since the 2001 GDP was approved for the Prima Deshecha Landfill Landslide Remediation Project for Zone 1:

 <u>Clean Water Act Section 404 Permit No. 980065200-ESL</u>. An Individual Permit was authorized by the USACE for the implementation of the Prima Deshecha Landfill Landslide Remediation Project based on *Riparian Habitat Mitigation Program, Prima* Deshecha Landslide Remediation Project, Modified Preferred Alternative (BonTerra Consulting 2002b) and the Coastal Sage Scrub Mitigation Program, Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative (BonTerra Consulting 2002b) and the Coastal Sage Scrub Mitigation Program, Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative (BonTerra Consulting 2002a). The intent of the project was to stabilize a large landslide immediately adjacent to the Prima Deshecha Cañada Creek and to re-align and restore the creek into an engineered bio-mitigation channel. The USFWS issued this permit on June 17, 2002.

- <u>USFWS Section 7 Consultation. Permit No. 980065200-ESL</u> includes a Section 7 Consultation with the USFWS to address impacts to the least Bell's vireo (a federally and state-listed Endangered species) and California gnatcatcher (a federally listed Threatened and state Species of Special Concern). The USFWS issued Biological Opinion (BO) 1-6-02-F-703 to the USACE on February 8, 2002.
- <u>RWQCB Section 401 Water Quality Certification Permit No. WPN 18-2001 112.02 (File No. 01C-112)</u>. On February 11, 2002, the RWQCB issued a 401 permit authorizing the Prima Deshecha Landfill Landslide Remediation Project.
- <u>CDFG Streambed Alteration Agreement (SAA) R5-2001-0301</u>. The CDFG issued an SAA for the PDL Landslide Remediation Project that authorized permanent impacts to 3.26 acres of stream and habitat. Mitigation for the habitat loss was addressed in the *Riparian Habitat Mitigation Program, Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative* (BonTerra Consulting 2002b). This permit was issued on February 19, 2001, and expires on December 31, 2004.
- <u>CDFG 2080.1 Consistency Determination</u>. The least Bell's vireo is both a state and federally listed Endangered species. Numerous meetings were conducted with the USACE, USFWS, RWQCB, and CDFG staff to avoid, minimize, and adequately mitigate for the effects of the Landslide Remediation Project on the vireo. The IWMD obtained authorizations from the USACE (Section 404 Permit No. 980065200-ESL), USFWS (Section 7 Consultation and subsequent BO No. 1-6-02-F-703), and RWQCB (Section 401 Water Quality Certification). Based on conservation measures and mitigation measures contained in these permit authorizations, IWMD requested and obtained a consistency determination from the CDFG on October 15, 2003, pursuant to the requirements of the California Endangered Species Act and as described in Section 2080.1 of the Fish and Game Code.
- Modification No. 1 to the USACE Permit No. 980065200-ESL. A modification to the original USACE Section 404 Permit was approved by the USACE on August 20, 2002, in response to the inadvertent removal of 0.18 acre of mixed sage scrub and an additional requirement for the removal of 0.08 acre of mule fat scrub for soil stockpiling in an ephemeral tributary to Prima Deshecha Cañada channel. As part of Modification No. 1, the original grading footprint was modified to avoid impacts to 0.30 acre of coastal sage scrub (CSS) and 0.04 acre of southern willow scrub habitat that the USACE permit and the associated USFWS permit previously authorized for removal in order to fully mitigate for the new impacts. In addition, the USACE requested that the USFWS re-initiate the Section 7 Consultation (Amendment No. 1 to Biological Opinion No. 1-6-02-F-703) in order to address the proposed modifications to habitat that had the potential to affect the California gnatcatcher and least Bell's vireo. The USFWS determined in their Biological Opinion issued to the USACE on August 2, 2002 (BO No. 1-6-02-F-703), that the proposed action would not likely adversely affect the gnatcatcher or vireo.
- <u>Amendment No. 2 to Biological Opinion No. 1-6-02-F-703</u>. During the landslide remediation activities, the Construction Geologist observed the landslide which underlied the cut slope that was created by the Landslide Remediation Project indicating that the landslide was still active and could threaten the newly created Prima Deshecha Cañada Bio-Mitigation Channel. IWMD prepared a Headscarp Grading Plan and determined that

approximately 0.75 acre of additional CSS habitat resources would require temporary removal. No areas under USACE jurisdiction (i.e., wetlands or waters of the U.S.) would be affected by this activity. However, since this activity was to occur within the project areas defined by the permit, the USACE took the lead and requested re-initiation of a Section 7 Consultation with the USFWS for proposed modification to 0.75 acre of CSS and to discuss potential effects on the California gnatcatcher. The USFWS issued Amendment No. 2 to BO 1-6-02-F-703 on October 31, 2003.

- <u>Extension of CDFG SAA No. R5-2001-0301</u>. An extension of CDFG SAA No. R5-2001-0301 to December 31, 2007, was granted to the IWMD on December 15, 2004.
- <u>Modification No. 2 to USACE Permit No. 980065200-ESL</u>. The USACE approved Modification No. 2 to the original permit February 22, 2005 in order to address impacts to 0.26 acre of CSS and 0.04 acre of riparian resources associated with the implementation of the Phase A2/B1 Liner and Desilting Basin project.
- <u>Amendment No 3 to USFWS Biological Opinion No. FWS-OR-703.13</u>. The USACE consulted with the USFWS as part of the request for Modification No. 2 (980065200-ESL) to implement the A2/B1 Liner and Desilting Basin project. The USFWS issued the amended BO on February 18, 2005.
- <u>Amendment No. 1 to CDFG SAA No. R5-2001-0301</u>. CDFG approved Amendment No. 1 to SAA No. R5-2001-0301 to implement the Phase A2/B1 Liner and Desilting Basin project on February 1, 2005.
- Expansion of the Section 401 Water Quality Certification for the A2/B1 Liner and Desilting Basin Project. RWQCB approved the expansion of Permit No. WPN: 18-2001 112.02haasj on February 11, 2005, for the Phase A2/B1 Liner and Desilting Basin project.

The construction portion of the Prima Deshecha Landslide Remediation Project was completed in March 2004. Implementation of the CSS mitigation program is complete. The riparian restoration construction was completed in the fall of 2004 and both programs will be monitored over the subsequent five years or until establishment has been achieved and signed off by the state and federal resource agencies, pursuant to the performance criteria contained in the approved plans.

Regulatory permits were obtained from the USACE, USFWS, CDFG, and RWQCB for the Phase A2/B1 Liner Project which included an approved mitigation program to offset the loss of 0.26 acre of CSS, 0.09 acre of mule fat scrub habitat, and 0.02 acre of U.S. jurisdictional waters. Mitigation was authorized at a ratio of 1:1 through an expansion of the existing biomitigation program within and immediately adjacent to the Prima Deshecha Cañada channel.

Landfill Operating Permits

The Prima Deshecha Landfill is a state-designated Class III facility, which is permitted for the disposal of non-hazardous municipal solid waste. No liquid or hazardous wastes are accepted or proposed for on-site disposal. The IWMD currently operates under: (1) Revised Waste Discharge Requirements (WDR) Order No. R9-2003-0306, which was issued by the San Diego Regional Water Quality Control Board (SDRWQCB) in November 2003 with Addendum No. 1 and adopted June 8, 2005; (2) General Permit to Discharge Storm Water associated with Industrial Activity (Order No. 97-03-DWQ) issued by the State Water Resources Control Board; (3) Solid Waste Facility Permit (SWFP) No. 30-AB-0019 which was issued by the County Health

Care Agency (LEAs) in March 2004, and last revised in 2005; (4) concurrence by the California Integrated Waste Management Board (CIWMB) in March 2004; (5) Permits to Construct and Permits to Operate the landfill gas collection and control systems issued by the SCAQMD; and (6) CUP 95-4 issued by the City of San Juan Capistrano, as well as other permits required for environmental monitoring and control systems.

2.3.2 IMPLEMENTATION AND STATUS OF 2001 GDP BIOLOGICAL MITIGATION MEASURES

In support of the recommendations contained within the Proposed Project's Pre-Mitigation and Regional Environmental Enhancement Opportunities Plan (REEOP), a discussion of existing biological mitigation sites at the PDL is presented below.

Exhibit 2.2-7 provides an overview of the location of each of the following biological mitigation sites on the PDL property.

<u>Site A</u>

Mitigation installation at Site A met, and in some cases, exceeded the acreage allocations specified in the May 29, 2002, *Riparian Habitat Mitigation Program* (RHMP), *Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative* (BonTerra Consulting 2002b). The site included a total of 7.52 acres of southern willow scrub, mule fat scrub, and marsh habitat plantings. Irrigation system installation was completed in the Site A re-aligned channel by February 2004. Planting of the southern willow scrub and mule fat scrub plant species was completed by March 2004.

The final installation review and walk-through was completed on March 15, 2004, at the Site A riparian mitigation site in compliance with the approved RHMP and associated project permits (referenced above). Exhibit 2.2-8 illustrates the biological successes at Site A.

Mitigation installation completion marks the initiation of the five-year maintenance and monitoring program at Site A. All maintenance tasks will be performed in compliance with the requirements listed in the project permits and in the approved RHMP. Maintenance will consist of: (1) protection of planting areas from vandalism, herbivores, etc.; (2) protection of adjacent resources during all maintenance activities; (3) erosion control; (4) irrigation and associated system maintenance for up to five years (with discontinuation preferred three years after implementation with Resource Agency approval); (5) weed species control performed on a weekly basis for the first three years of the five year program, and on a monthly basis for the remaining two years of the five year program; (6) replacement of all failed plant species as required to meet performance standards; and (7) performance of any other maintenance tasks considered necessary to facilitate compliance with the required growth performance standards.

<u>Site B</u>

The RHMP and project permits required the creation of a total of 11.84 acres of riparian habitat within the PDL site. Since Site A provided 7.52 acres of riparian planting, additional mitigation area was created in Site B, just downstream of Site A. Site B included a total of 4.32 acres of southern willow scrub, mule fat scrub, and marsh habitat planting. Mitigation installation at the Site B riparian mitigation site was completed November 1, 2003, in compliance with the approved RHMP and associated project permits. Exhibit 2.2-9 illustrates the before-and-after successes at Site B.







Prior to Site A Channel Construction 3-22-04



After Site A Channel Construction 8-2-04

Site A Mitigation Site

Prima Deshecha Landfill Supplemental Environmental Impact Report 597

Exhibit 2.2–8



 $R:\Projects\OCIWMD\J004\SEIR\ 597 PublicReviewDraft\Graphics\Ex2.2-8_siteAmitigatpics_110804.pdf$



Prior to Site B construction 3-22-00



After Site B Construction 7–13–04

Site B Mitigation Site

Prima Deshecha Landfill Supplemental Environmental Impact Report 597



Exhibit 2.2–9

 $R:\Projects\OCIWMD\J004\SEIR\597PublicReviewDraft\Graphics\Ex2.2-9_siteBmitigatpics_110804.pdf$

<u>Site C</u>

The final installation review and walk-through at the Site C coastal sage scrub mitigation site was completed on March 15, 2004, in compliance with the approved May 24, 2002, *Coastal Sage Scrub Mitigation Program* (CSSMP), *Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative* (BonTerra Consulting 2002a) and its associated project permits. The approved CSSMP, Biological Opinion, and October 31, 2003 Amendment require the creation of 13.86 acres of coastal sage scrub habitat within the PDL site as compensation for project impacts. Mitigation installation has occurred on landslide remediation cut slopes. Exhibit 2.2-10 shows the status of the mitigation as of 2004.

<u>Site D</u>

Site D was originally located within the construction limits of the landslide remediation project within Zone 1. Following the completion of the landslide remediation project, (including the installation of all of the required riparian and coastal sage scrub mitigation elements), Site D was bare and would have been treated for erosion control with the standard erosion control seed mix. IWMD determined that this barren site presented an opportunity for CSS restoration for use in offsetting future impacts within the PDL property. Mitigation installation at Site D was completed March 15, 2004 (Exhibit 2.2-11). As a site-wide mitigation program was not yet developed for the pre-mitigation areas, installation followed the implementation methodologies and guidelines described in the approved CSSMP for Site C. Pre-mitigation installation at Site D consisted of establishing a total of 18.40 acres of coastal sage scrub and grassland species on the head scarp/excavation area immediately adjacent to Site C. Approximately 0.26 acre of Site D was subsequently approved by the USACE, USFWS, CDFG, and RWQCB for use in offsetting impacts to CSS resources located within the Phase A2/B1 Liner and Desilting Basin project area.



Prior to Grading 3-22-00



After Grading 10-5-04

Site C Mitigation Site

Prima Deshecha Landfill Supplemental Environmental Impact Report 597



Exhibit 2.2-10





During Grading 10-10-03



After Grading 11-1-04

Site D Mitigation Site

Prima Deshecha Landfill Supplemental Environmental Impact Report 597

Exhibit 2.2–11



R:\Projects\OCIWMD\J004\SEIR 597PublicDraftReview\Graphics\Ex2.2-11_siteDmitigatpics_110804.pdf

SECTION 3.0 PURPOSE AND NEED FOR AMENDMENT NO. 2 TO THE 2001 GENERAL DEVELOPMENT PLAN

Amendment No. 2 and SEIR 597 further modify the 2001 GDP with the incorporation of the elements of the Proposed Project that are presented in Section 4.0. In aggregate, these project modifications address the entire property and are necessary to: (1) provide for the areal extent of future landslide remediation activities around the PDL, without which adequate slope stabilization measures could not be provided for the PDL and the site would be required to cease operations much earlier than currently planned; (2) accommodate future landfill-related features such as landfill gas generation (LFG) perimeter probes and LFG collection header lines; (3) make adjustments for Zone 4 to provide drainage and erosion control facilities that avoid sensitive biological resources; and (4) implement measures needed to ensure the long-term success of the environmental mitigation and restoration components of the overall GDP. In order to facilitate the assessment of biological impacts from future landfill operations within both zones and to coordinate pre-mitigation of these impacts, the limits of disturbance around each zone have been refined to accommodate these features.

Also included within the Proposed Project is a comprehensive Pre-mitigation Plan and Regional Environmental Enhancement Opportunities Plan (REEOP) for the site. The Pre-Mitigation Plan will allow landfill operations to continue seamlessly by providing mitigation for operational impacts well in advance of and prior to any future impacts on biological resources. Additionally, regional environmental enhancement opportunities will have been identified for offsetting environmental impacts of other County (or third party) projects.

Accordingly, this document will incorporate by reference those analyses contained in the 2001 GDP, as well as applicable and appropriate analyses from EIR 548 and EIR 575 as the baseline for SEIR 597. Specific new or updated analyses contained in this document will focus on only those aspects of the 2001 GDP that require modifications.

3.1 PHYSICAL SITE STABILITY

The PDL site is part of the Peninsular Ranges Province of southern California. Exposed bedrock materials on site consist of both marine and terrigenous sedimentary rock of the Capistrano and Monterey formations, as well as the San Onofre Breccia. These bedrock units are overlain by locally derived landslide materials and are predisposed to instability in many areas. Given the low strength of some bedrock units and the high instability potential of many of the existing landslides at the site, new slope failures and/or reactivation of existing landslides are probable. The 1998 landslide in a Zone 1 stockpile area provides evidence that this is a major concern that has significant potential to disrupt landfill operations if not proactively remediated.

Final landslide remediation measures are required to stabilize portions of Zones 1 and 4 to allow the installation of the landfill liner and leachate collection and removal system. Without the implementation of on-site landslide stabilization measures, it will not be possible to update the solid waste disposal permits needed to continue landfill operations at the PDL to the site's projected closure date. A significant amount of waste originally designated for disposal at the PDL would then need to be transported to other facilities. Landslide remediation measures are required for the full development of Zone 1 and for the majority of Zone 4.

Although final landslide remediation design plans will not be prepared until closer to the time that liner construction is scheduled (anticipated by 2008 for the last major phase in Zone 1 and by 2015 for the first phase in Zone 4), potential disturbance limits have been identified in the SEIR 597 Proposed Project based on available geotechnical information for the site.
3.2 ACCOMMODATE FUTURE LANDFILL-RELATED FEATURES

The limits of disturbance around Zones 1 and 4 have been better defined to accommodate the implementation of features needed to support future landfill operations. These features include environmental control facilities such as landfill gas perimeter probes, extraction wells and/or header lines, stockpile areas, temporary staging areas and/or other support features. These features are fairly typical support elements of landfill operations and have always been a part of ongoing operations at the PDL. EIR 575 did acknowledge the need for these facilities but was not specific on areal requirements surrounding the refuse footprint. The limits of disturbance for each zone are being refined within SEIR 597 to better document the potential acreage required for these facilities outside and adjacent to the refuse footprint and to assess the impacts to biological resources in order to coordinate and implement pre-mitigation.

3.3 <u>RE-DESIGN DESILTING SYSTEM TO REDUCE BIOLOGICAL RESOURCES</u> <u>IMPACTS</u>

3.3.1 AVOID LEAST BELL'S VIREO TERRITORIES

The 2001 GDP proposes desilting basin for Zone 4 at a location between Zones 1 and 4. However, constructing a basin at this site will have a direct impact on four least Bell's vireo territories (total removal of two territories; partial removal of two others). As the least Bell's vireo is listed as a federally Endangered species, any temporary or permanent impact on the species is considered significant and requires consultation under Section 7 of the Endangered Species Act. Accordingly, the project sponsor has a responsibility under CEQA and the Clean Water Act (CWA), as well as a number of other relevant environmental statutes, to conduct an analysis to identify the least environmentally damaging alternative to the Proposed Project. Formulation and analysis of other alternatives that met the project objectives were, therefore, undertaken to determine whether there was an alternative location for this basin that would reduce the impact to the least Bell's vireo.

3.3.2 SUSTAIN THE DOWNSTREAM RIPARIAN CORRIDOR

The Prima Deshecha Cañada stream originates in Zone 4 and provides water to biological resources downstream of the project site. Although the hydrogeologic relevancy of the spring and its underlying geologic formations require further exploration, the worst-case scenario of total depletion of surface water recharge to the spring is being considered here. Direct impact to the spring or its recharge area could alter hydrologic conditions at the site and adversely affect the biological resources and mitigation areas within and along the downstream riparian corridor. The Proposed Project provides features designed to maintain hydrologic stability for this corridor and sustain on-site biological resources once the upper portions of the Prima Deshecha Cañada stream channel are impacted within Zone 4. Mitigation for these impacts is required by environmental statute and is necessary for obtaining agency permits in order to continue landfill operation.

Landfill operations are expected to occur in Phases D and E at the east side of Zone 4 by the year 2045 (see Section 4.0). At that time, the construction of the liner system will likely reduce the potential recharge area to the spring located in Zone 4 under this worst-case scenario. The 2001 GDP proposed subdrain system for the landfill in order to collect subsurface waters that emerge underneath the site and to direct those flows to an outlet at the western edge of Prima Deshecha Cañada stream outside Zone 4. The remaining open space area on the eastern side of Zone 4 would likely contribute to sub-surface water flows into the subdrain system for a period of time; however, these flows are projected to be reduced as the area believed to recharge the spring is progressively impacted by landfill activities after 2045. *An Analysis of*

Groundwater Resources in Zone 4, Prima Deshecha Landfill, dated January 6, 2004 (see Appendix E), was conducted by GeoLogic Associates to assess the potential gross quantity of water that might be available for habitat maintenance in the Prima Deshecha Cañada stream from subdrain flows. Based on limited data and gross assumptions about aquifer characteristics, a range of 6 to 52 years was estimated for continued flows through a subdrain system for the landfill after the recharge area begins to be reduced by 2045. Although a detailed hydrogeological study is needed to confirm these spring and watershed parameters, it is assumed for the purposes of this analysis that impact to the spring's recharge area will eventually have an adverse impact on the Prima Deshecha riparian corridor.

Elements of the Proposed Project have been developed to address these concerns and to provide for long-term viability and functionality of this riparian system.

3.4 FACILITATE LONG-TERM STEWARDSHIP OF ON-SITE BIOLOGICAL RESOURCES

3.4.1 RELATIONSHIP TO OTHER REGIONAL PLANNING EFFORTS (NCCP, SAMP)

Implementation of full PDL buildout through the year 2067 (as analyzed in EIR 575 and SEIR 597) will result in impacts to federally and state-listed special status plant and wildlife species; special status habitats and wetlands; and waters of the U.S. as identified by the state and federal Endangered Species Acts, the *California Fish and Game Code*, and the federal Clean Water Act (CWA). Regulatory authorizations will be required to permit the project and to ensure appropriate mitigation in order to offset project impacts. In addition, the IWMD intends to participate in the Southern Subregion Natural Communities Conservation Plan/Master Streambed Alteration Agreement/Habitat Conservation Plan program (NCCP/MSAA/HCP), a subregional conservation planning program which is intended to provide long-term protection of natural vegetation and wildlife diversity while allowing compatible land uses and appropriate development and growth.

3.4.2 DEVELOPMENT OF A PRE-MITIGATION PROGRAM

The IWMD has prepared two habitat restoration and enhancement programs to pre-mitigate potential project impacts and to obtain long-term environmental permits to cover full project buildout. This approach ensures seamless operation of landfill activities and responsible stewardship of on-site environmental resources through the proactive development of a Pre-Mitigation Plan and a REEOP in partnership with the NCCP/MSAA/HCP program. The Pre-Mitigation Plan is intended to provide a comprehensive mitigation program to achieve self-sustaining habitats for coastal sage scrub, southern needlegrass grassland, and riparian habitats outside the current and future landfill operations. The Pre-Mitigation Plan will also include seed collection and planting of small-flowered morning glory (*Convolvulus simulans*), paniculate tarplant (*Deinandra paniculata*), vernal barley (*Hordeum intercedens*), and seed collection and transplanting of the thread-leaved brodiaea (*Brodiaea filifolia*). The objective of the Pre-Mitigation Plan is to establish these habitats and plant species prior to their removal, thus eliminating the element of temporal loss from the project site by maintaining the long-term net subregional values of these resources within the South NCCP Subregion.

3.4.3 DEVELOPMENT OF A REGIONAL ENVIRONMENTAL ENHANCEMENT OPPORTUNITIES PLAN

The REEOP is another plan that identifies areas within the PDL that are suitable for the establishment of coastal sage scrub and native grassland, as well as sensitive plant species associated with these habitat types. Once identified, these areas could be incorporated into the NCCP/MSAA/HCP Restoration and Enhancement Adaptive Program, once the

NCCP/MSAA/HCP is approved; these areas could also be subsequently used by the IWMD, other County agencies, or (potentially) other entities as off-site mitigation to enable seamless development of non-PDL projects that incur impacts to these habitat types and species. The REEOP and the Pre-Mitigation Plan would also serve as the basis for participating in the NCCP/MSAA/HCP.

SECTION 4.0 THE PROPOSED PROJECT

4.1 **PROJECT LOCATION**

The 1,530-acre Prima Deshecha Landfill (PDL) site is located in Orange County (Exhibit 4.1-1), and includes acreage within the jurisdictions of the cities of San Juan Capistrano (570 acres) and San Clemente (133 acres). The remaining 827 acres are within unincorporated Orange County. The 2001 General Development Plan (GDP) is the planning document for coordinated long-term implementation of both interim and ultimate site development uses. The most current GDP was adopted by the County in 2001 and, along with its 2002 amendment (Amendment No. 1), constitute what is referred to in this document as "the 2001 GDP."

The PDL site lies in the hills of southeastern Orange County (Exhibit 4.1-2). Ground elevations range from 230 feet above mean sea level (msl) at the southwestern boundary of the site to a maximum elevation of 1,125 feet above msl at the northeastern boundary of the site. The Prima Deshecha Cañada watercourse traverses the site from the northeast to the southwest. Two major utility easements, including a 150-foot-wide San Diego Gas and Electric (SDG&E) easement and a 200-foot-wide Southern California Edison (SCE) easement extend through the central portion of the site and separate the western (Zone 1) and eastern (Zone 4) components of the landfill property (Exhibit 2.1-1).

4.2 ENVIRONMENTAL SETTING

The PDL is situated in the western foothills of the Santa Ana Mountains, at the eastern edge of the City of San Juan Capistrano. The majority of the 1,530-acre site is located in Prima Deshecha Cañada, a hilly canyon in an upland area of south Orange County. Approximately 15 percent of the PDL site lies within the Segunda Deshecha Cañada watershed, at the southeastern corner of the PDL property. The setting in respect to each environmental issue can be found in Section 5.

The PDL is located within the following three jurisdictions: the city of San Juan Capistrano, the city of San Clemente, and in unincorporated Orange County. As such, development of the site is affected by long-range plans adopted by those municipalities. To illustrate, the Zone 1 area of the site is located in the City of San Juan Capistrano. The City of San Juan Capistrano adopted a General Plan Amendment in 1995 that changed the 2.5 (Regional Park) land use designation on the portion of the PDL located within that jurisdiction to a land use designation of "Solid Waste Facility" (SWF). The 2001 GDP for Zone 1 is consistent with this SWF land use designation in the *City of San Juan Capistrano General Plan*.

4.3 **PROJECT DESCRIPTION**

4.3.1 PROPOSED PROJECT ELEMENTS

The Proposed Project will include the following elements covered by this SEIR (Exhibit 4.3-1):

1. Changes in the limits of disturbance and other impacts associated with implementation of landslide remediation measures and/or the accommodation of landfill infrastructure and environmental control systems throughout the landfill.

The boundaries for Zones 1 and 4, as presented within EIR 575, represent the refuse footprint of each zone, with the incorporation of some additional area of cut slopes. The limits of potential disturbance proposed in SEIR 597 (which extend beyond the Zones 1 and 4 landfill boundaries) reflect, in part, the area to be graded in order to remediate unstable geologic conditions on the

property (i.e., Capistrano and Monterey formation soils). Zone 5, which accommodates the alignment as contained within the Master Plan of Arterial Highways (MPAH) for the La Pata Avenue extension, traverses the PDL site between Zones 1 and 4. The limits of disturbance for Zones 1 and 4 (as shown on Exhibit 4.3-1) encompass portions of Zone 5, which will consider site stabilization in its own project design; however, planning and design efforts for both the circulation and landfill features of the 2001 GDP are being closely coordinated to provide for project designs that are consistent and that accommodate a full range of features planned for the site. Implementation of landslide remediation measures will be needed in order to obtain permits that will fully execute both circulation and landfill project features. These measures will be implemented in a manner which minimizes both potential damage to existing and enhanced environmental resources and disruption of on-going landfill operations.

The most cost-effective method to stabilize portions of the site underlain by landslides is a buttress fill, as illustrated on Exhibit 4.3-2, Proposed Buttress Schematic. However, in order to avoid biologically sensitive terrain, remediation of the large landslide complex located within the center of Zone 4 might also be accomplished by means of a shear key (Exhibit 4.3-3, Shear Key Schematic). Other measures for unstable cut slopes in the various units of bedrock and landslide debris may include construction of low angle (2:1) or shallower cut slopes, buttress and/or stabilization fills, and structurally reinforced fills (GLA 2002). Potential landslide remediation areas around and between Zones 1 and 4 have also been incorporated within the disturbance limits based on the site's geologic map, which is presented in Section 5.0 (Exhibit 5.2-1).

Although additional geotechnical field investigations and slope-stability analyses will be conducted for future development of the PDL, an updated assessment of the potential limits of disturbance has been completed for the site based on available geotechnical information. These updated limits of disturbance are illustrated on Exhibit 4.3-1 and consist of a 278-acre increase for the entire site over the zone boundaries presented in the GDP. Although there is a possibility that further geotechnical analyses may result in the determination that additional area is required for landslide remediation in the future, an effort was made to address conservative limits of disturbance (i.e., as wide as projected to be necessary). Should the limits of disturbance change in the future (based on new information), the change will be covered by supplemental environmental documentation as appropriate; the type of document will be determined by the potential impact from a proposed change.

The updated disturbance limits, as indicated on Exhibit 4.3-1, are analyzed within SEIR 597. The 2001 GDP and EIR 575 analyzed landfill impacts for a total of approximately 800 acres, of which approximately 327 acres were located within Zone 1 and approximately 473 acres were located within Zone 4. This impact area consisted of the refuse footprint with some additional area needed for cut slopes. Geotechnical and operational considerations have changed these limits of disturbance to a total of 1,078 acres, for an incremental increase of 278 acres (or 35 percent). Of these 278 acres, approximately 110 acres are located around the perimeter of Zone 1 and 168 acres are located around the perimeter of Zone 4. This increase in acreage is associated with the need to allow for landslide remediation and other landfill-support features, as indicated in Exhibit 4.3-1. Potential stockpile and trail areas have been identified to the west and south of Zone 1, respectively, and areas along the perimeter of Zones 1 and 4 are included within the disturbance limits to accommodate future environmental protection systems. The disturbance limit to the southeast of Zone 1 has been increase of 278 acres for Zones 1 and 4 is the areal extent of the Proposed Project being analyzed within SEIR 597.

There will be no increase in the landfill prism, trash capacity, or operational life of the landfill facility as a result of the Proposed Project, nor will ongoing landfill operations be adversely







R:/Projects/OCIWMD/J004/SEIR 597PublicReview/Ex4.3-1_project_020606.pdf





affected. If future geotechnical investigations and analyses result in changes to the disturbance limit shown in SEIR 597, those changes will be addressed in a future CEQA analysis and amendment of project permits, as appropriate.

Changes in the limits of disturbance overlapping the RMV Burdened Property (Exhibit 2.2-6) include activities associated with temporary roads and infrastructure required to support landfill operations; site-grading and soil-filling associated with the implementation of landslide stabilization measures; and biological mitigation. These actions are consistent with the Settlement Agreement-designated approved uses for this property, which include landfill site-grading and/or soil-filling activities in support of landfill operations as long as they are not visible from the Benefited Property and there is no placement of waste within the Restricted Area. There will be no change in the landfill refuse prism or refuse footprint anywhere on the site from that which was approved within the 2001 GDP (as amended). Consequently, there are no such changes proposed within the Burdened Property. Accordingly, these uses are consistent with the commitments and restrictions for this area as set forth within the County of Orange/RMV Company Settlement Agreement.

2. Re-design of the desilting system for Zone 4 to minimize impacts on sensitive biological resources.

The 2001 GDP included a permanent desilting basin between Zones 1 and 4 to provide desilting capabilities for Zone 4. This location would have directly impacted four of five existing least Bell's vireo (LBV) territories (fully impacting 2 territories and partially impacting 2 others). Alternative locations for this basin were investigated to minimize or avoid impacts to these sensitive resources. Alternative design concepts proposed for this basin included both in-stream and off-stream desilting basin facilities. The preferred design alternative incorporated into the Proposed Project includes several desilting basins located around the perimeter of Zone 4. Exhibit 4.3-4 shows the revised location proposed for these facilities. One LBV territory is centered in the Zone 4 landfill and its impact is unavoidable in all alternatives including the Proposed Project.

Relocation of the function of the desilting basin to several basins around the perimeter of Zone 4, as well as other features of the Proposed Project, will result in minor modifications to the Zone 4 refuse footprint; however, these modifications would not significantly affect landfill capacity, life, or overall long-term operations.

Desilting Basins

As stated above, the Proposed Project includes up to four permanent detention/desilting basins around the perimeter of the Zone 4 2001 GDP footprint. The primary function of the desilting basins is to provide desilting and some detention capabilities for Zone 4 landfill operations. These basins can also function as part of a rainfall collection system designed to maintain flows to the downstream riparian corridor once Zone 4 operations have impacted the spring and its probable recharge area (discussed further below). Each permanent desilting basin would be concrete-lined to assist in silt-removal operations and to maintain the correct basin shape and will be located outside the refuse footprint in an effort to prevent impacts on the landfill liner system. Site-stabilization considerations will be incorporated into final desilting basin design plans.

Capacity impacts related to modifying the refuse footprint to accommodate these perimeter basins are minimal (1 to 2 million cubic yards [mcy]) as the basins are proposed for siting on the outer edges of the landfill footprint. This capacity loss can be offset through minor grading modifications. The basins would also be designed to reduce both silt content from tributary stormwater runoff areas at various stages of the landfill development and pass flows from a

100-year storm event (pursuant to County of Orange Hydrology criteria). Although the Zone 4 design plan is conceptual in nature, preliminary estimates of potential sediment yield have been made for planning purposes. The total anticipated sediment yield for the conceptual Zone 4 plan, based on County of Orange standards (Standard Plan 1327) is between 20 and 30 acrefeet (af) assuming minimal vegetation cover during the operational life of the landfill. The total preliminary capacity of the 4 permanent basins proposed for Zone 4 is about 55 af, which would meet the County's minimum standards. These basins vary in acreage from 1.3 acres to just over 3.0 acres each. Interim desilting basins will also be constructed for the various phases of development at the PDL as part of ongoing operations.

Exhibit 4.3-4, Revised Zone 4 Desilting System, illustrates the locations of potential desilting basins around Zone 4, and Table 4.3-1 illustrates the conceptual design parameters of each basin along with potential timing of construction. The first basin proposed for initial development of Zone 4 is an approximate 1.4-acre desilting basin at 480 feet above msl elevation (Basin 1) which has limited available area (potential 5 af) depending on the La Pata Avenue final alignment. Desilting Basins 1 and 2 (700 feet above msl) are anticipated to be built within Phase A of Zone 4 and are proposed to be permanent facilities that will be in place through the life of the landfill and beyond. Basin 2 would not likely be utilized until the later stages of the Phase A fill operation. This basin would be constructed after an underlying landslide is remediated as part of the Phase A development.

Desilting Basin	Elevation	Construction Timeframe ^a	Approximate Volume	Approximate Size
1	480 msl ^b	Phase A	5 af ^c	1.4 ac ^d
2	700 msl	Phase A	15+ af	2.9 ac
3	560 msl	Phase C	15+ af	2.1 ac
4 ^e	730 msl	Phase C	20+ af	3.3 ac
 ^a phasing refers to Zone 4 ^b mean sea level ^c acre-feet ^d acres ^e may not be required 				

TABLE 4.3-1PROPOSED ZONE 4 DESILTING BASINS

During the Phase C area excavation (see below), construction of a 560 feet above msl (Basin 3) elevation desilting basin is proposed to address silt generation due to additional areas of disturbed soil. The ultimate configuration for Basin 3 may be affected by the final La Pata Avenue alignment grades. This basin would also need to be constructed after an underlying landslide is remediated in the landfill Phase C area. The pad area for a 730 feet above msl (Basin 4) elevation could also be accommodated during the Phase C development. Basin 4 may not be required to meet desilting criteria alone. If it is determined to be required for desilting and/or for stormwater detention needs, Basin 4 could be constructed during the middle of Phase C fill operations. Similarly, any landslide remediation requirements for Basin 4 would be addressed with the Phase C landfill development.

In addition to providing desilting capabilities, the PDL basins can also provide detention capabilities. The 100-year storm flows calculated for the site's pre-developed (1,757 cubic feet per second [cfs]) and post-developed (1,864 cfs) conditions indicate a small increase in stormwater flows of just less than 100 cfs. Temporary detention basins can be constructed in the natural watershed area upstream of the early landfill Phases A through C to meet detention requirements. As the permanent desilting basins are constructed, the required detention can be accommodated in the northerly Basin 2 and/or Basins 3 and 4.



3. Development of features for sustaining biological resources affected by impacts to spring flows resulting from Zone 4 build-out.

The Proposed Project incorporates hydrologic features proposed for the long-term viability of on-site biological resources. These features include:

- Construction of a subsurface water storage reservoir(s) underneath the Zone 4 desilting basins (pending additional investigation confirming engineering/operations, maintenance feasibility, and regulatory permitting requirements) and/or
- Other supplemental water sources prior to, during, and after the Zone 4 operations have impacted the upper portions of the existing water source for the Prima Deshecha Cañada stream.

Surface Water Augmentation Options

When construction of Phases D and E in the eastern portion of Zone 4 commences (estimated to occur by 2045), the spring recharge area will diminish as the landfill liner system expands to prevent surface water infiltration. Therefore, another water source would be sought to supplement subdrain flows as a means of maintaining water delivery to the downstream reaches of the stream. At this time, the preferred option is construction of a subsurface reservoir under 1 or more (if needed) of the Zone 4 desilting basins (described more fully below). Should this option prove infeasible from an engineering, permitting, operations and/or maintenance standpoint, another water augmentation option may be pursued. These other options are described in Section 7.2.3.

Subsurface Reservoir

The subsurface reservoir system would function in tandem with one or more of the desilting basins (discussed above) from which stormwater would be temporarily held for sediment-control purposes. This is a potentially promising design concept that would be implemented pending engineering feasibility studies.

Concept. An underground rainfall collection system is proposed to be a series of large underground pipes for storage of de-silted water that can be collected from storm flows routed through the desilting basin. The collected rainwater would be the same surface water that would be routed in the stormwater collection channels for Zone 4 through the desilting basin prior to discharge into the Prima Deshecha Cañada natural stream channel. Rainfall runoff would be stored in water-tight pipes (i.e., large, high-density polyethelene [HDPE], gasketed reinforced concrete pipe, or sealed corrugated galvanized steel pipes), which could be placed underneath or adjacent to any of the proposed permanent desilting basins described above. The desilted water would then flow from the desilting basin through a riser pipe system into an underground series of parallel pipes (Exhibit 4.3-5). The pipe system would have manhole access but is not anticipated to require much maintenance as the water entering the system will be almost completely desilted. As an alternate option to the underground pipe system reservoir, an aboveground reservoir at a lower elevation with a connector pipe for gravity flow could also be used (Arbogast 2006).

Operation. Desilted stormwater could be stored in the underground reservoir(s) for release by gravity through a small pipe down to the streambed based on historic spring flow rate and biological mitigation permit requirements. This system would not have any manual operating requirements.

Reservoir Capacity. In order to verify that reservoir storage capacity can be achieved, a preliminary evaluation of potential requirements was performed. The spring's historic average flow rate within Zone 4 (which has not been well-documented) is estimated at 3 to 5 gallons per minute (gpm). For the purposes of this analysis, it is conservatively assumed that Zone 4 operations will eventually cut off groundwater spring contributions to channel flows completely, and a subsurface reservoir may need to replace these flows entirely. Ultimately, there could be continuing groundwater flow into the channel via geologic rock fractures or fissures. Monitoring of channel flows during Zone 4 operations will be conducted to assist in the final determination of subsurface reservoir storage capacity prior to detailed facility design.

To provide a 3 gpm flow for a period from April 15 to the end of the year requires a storage volume of approximately 1.1 million gallons or 3 af. A 5 gpm flow requires approximately 1.9 million gallons of storage capacity, or 6 af. This range of subsurface water storage capacity could be provided within the limits of the initial 480 Elevation Desilting Basin proposed for Zone 4.

Plant densities within the Prima Deshecha Cañada riparian corridor are triple those of the historic plant communities due to permit requirements associated with previous Zone 1 impacts. Accordingly, historic flows will likely need to be supplemented to sustain the increased plant densities. The subsurface reservoir concept has been developed as a contingency based on a determination of need to maintain plant densities. Using a conservative assumption that triple the plant densities will require triple the amount of water for plant maintenance, an analysis was done on the subsurface water storage required to provide a 9 to 15 gpm flow rate to the stream channel over a 9-month period each year (April 15 to December 31). The subsurface water storage capacity to support this flow rate would range from approximately 11 af (3.5 million gallons) to 18 af (5.8 million gallons). This additional water storage capacity could be accommodated underneath the 2 basins (480 and 700 Elevation Desilting Basins) proposed for Phase A of Zone 4. An aboveground water reservoir would have similar capacity requirements.

Timing. It should be noted that the spring's recharge area is not anticipated to be substantially impacted until later phases of Zone 4 development (Phases D and E). Therefore, it is anticipated that subdrain flows conveying subsurface water from the spring (underneath Phases A and B) will be discharged into the Prima Deshecha Cañada stream channel downstream of Zone 4 until approximately 2045. Flows from the subsurface water storage reservoir would then serve to augment the subdrain flows until about 2045 when it is projected that impacts would occur in the recharge area. At that time, all 4 permanent desilting basins proposed for Zone 4 can be made available for subsurface storage capacity.

Monitoring/System Design Requirements. The actual sizing and timing of construction for the subsurface storage reservoir(s) will depend on the water demand requirements for the Prima Deshecha Cañada stream habitat and the anticipated flows from storm waters. The IWMD is initiating a spring monitoring program to develop a better database of spring flows, from which a detailed analysis can be performed to design the future subsurface reservoir(s) for surface water augmentation. Performance monitoring of the Zone 1 biological mitigation channel areas will also provide data on water requirements for the stream riparian plant communities. A detailed analysis and design of the requirements for a subsurface (or aboveground) storage reservoir for surface water augmentation of spring flows will occur during engineering feasibility studies as final design of Zone 4 occurs prior to Zone 4 landfill operations, which are scheduled to begin in 2019.



D:\Projects\OCIWMD\J004\Graphics\Ex4.3-4_013006.ai





R:\Projects\OCIWMD\J004\SEIR597 Public Review Draft\graphics\Ex4.3-5_013006.pdf

Exhibit 4.3-5, Subsurface Storage Reservoir, illustrates the design concept for the subsurface storage reservoir that may be constructed underneath the Zone 4 desilting basin(s). Alternatives to this subsurface storage reservoir will include potable and/or reclaimed water sources and on-site well(s).

4. Presentation of Excavation Phasing Scenarios within Zone 4.

The Proposed Project phasing approach for Zone 4 proposes to begin excavation operations in the northwestern corner of the zone (see Phase A in Exhibit 4.3-6) upon completion and closure of Zone 1. Operations and development of the landfill would then proceed in a counterclockwise direction, with a series of excavations and refuse fills (Phases A through H) until the final grade at 1,010 feet above msl (consistent with the San Clemente MOU) is reached. As there is a major landslide complex located within the center of Zone 4, remediation of this landslide will likely take the form of excavation of the unstable material and construction of a buttress fill and/or shear key; phasing of this work may begin as early as the initiation of landfill operations in Phase A. The limits of potential disturbance due to excavation requirements for landslide remediation have been identified in the SEIR 597 Proposed Project in order to better assess environmental impacts. Future geotechnical investigations and final design plans for phase development may alter the conceptual excavation phasing limits presented in this document. However, an effort was made to address conservative overall limits of disturbance.

5. Development and implementation of a comprehensive pre-mitigation plan to mitigate for project-related impacts.

The Proposed Project is intended to address all anticipated impacts to coastal sage scrub, southern needlegrass grassland, special status plant species, and state and federal jurisdictional wetlands and waters of the United States that will likely occur at some point during the ultimate development of the Prima Deshecha Landfill between 2019 and 2067. Impacts that are scheduled to occur within Zone 4 in approximately 14 years will be mitigated at a 1:1 ratio through near-term (prior to impact) restoration, enhancement, and creation of similar habitats. Relocation of special status species will be accomplished through seed collection and distribution in appropriate areas of the landfill well before the actual impacts to these resources occur.

Based on the revised limits of disturbance (as presented in Exhibit 4.3-1), the ultimate buildout of the PDL will likely result in the following impacts:

- 122.00 acres of coastal sage scrub
- 18.05 acres of southern needlegrass grassland
- Brodiaea (Brodiaea filifolia) and special status plant species
- 9.81 acres of CDFG jurisdictional resources
- 3.42 acres of USACE jurisdictional resources

The conceptual design of the on-site Pre-mitigation Plan is illustrated on Exhibit 4.3-7, Premitigation Plan On-Site, off-site elements are illustrated on Exhibit 4.3-8, Off-Site Pre-mitigation Location (San Juan Creek Giant Reed Removal). Pre-mitigation concepts have been developed to accommodate the future expansion of Camino de los Mares through the southwestern portion of the project site, based upon the conceptual alignment shown in the MPAH and the latest design alternatives for the La Pata Avenue Extension. Pre-mitigation Plan design is based on the following goals:

- Maximize on-site restoration and enhancement mitigation opportunities. All the coastal sage scrub, southern needlegrass grassland, and special status plant species that will be impacted by the ultimate development of the PDL can be accommodated within the boundaries of the landfill property without impacting current and future landfilling operations or any potential future alignment of La Pata Avenue. Although there are some opportunities for mitigating impacts to the state and federal jurisdictional resources on site, most mitigation is proposed to occur within Caspers Wilderness Park.
- Select areas that are currently disturbed or contain non-native species. The proposed pre-mitigation opportunity areas occur within areas of the landfill property that have been disturbed by previous grazing within annual grasslands or within areas disturbed by landslide remediation activities.
- Select areas that are accessible and contiguous to provide efficiencies in *installation, maintenance and monitoring.* The proposed mitigation sites are located immediately adjacent to the newly created bio-mitigation site and existing native coastal sage scrub and riparian habitat resources and are accessible from existing landfill maintenance roads.
- Provide habitat connectivity between existing native habitat areas (including existing mitigation sites A, B, C, and existing pre-mitigation site D), and existing preserved open space in adjacent land areas. The proposed mitigation plan identifies areas of the landfill property located primarily along the southern boundaries immediately adjacent to the newly created bio-mitigation area and the existing habitat areas that contain suitable habitat for and are occupied by the least Bell's vireo (a state-and federally listed Endangered Species) and California gnatcatcher, a federally listed Threatened Species. The proposed pre-mitigation areas will also provide enhanced habitat connectivity to the adjacent, off-site Talega mitigation site in San Clemente and other permanent open space areas to the south and east.
- **Enhance sensitive species habitat.** The proposed Pre-mitigation Plan will include restoration, enhancement, and creation elements that enhance the long-term habitat conservation values for the California gnatcatcher, least Bell's vireo, and other sensitive and non-sensitive native plant and wildlife species.
- **Incorporate viewshed protection elements.** The proposed Pre-mitigation Plan will Incorporate viewshed protection requirements from City/County MOUs and agreements with adjacent landowners.

Should the actual disturbance limits impact less acreage than what is shown on Exhibit 4.3-1, surplus mitigation areas will be made available to offset the impacts of other projects as needed.

6. Development of a comprehensive conceptual plan identifying regional environmental enhancement opportunities on site.

The Regional Environmental Enhancement Opportunities Plan (REEOP) identifies restoration, enhancement, and creation opportunities on the PDL property that can be made available to satisfy potential mitigation requirements for (non-PDL) public and private projects in the region (Appendix H). The REEOP document will serve to identify these opportunities in conjunction with state and federal resource agency permit authorizations. SEIR 597 provides the CEQA



Exhibit 4.3-6

Prima Deshecha Landfill Supplemental Environmental Impact Report 597









Area of Disturbance - 1069.36 Acres

Grading Contours







documentation that would allow other pubic agencies or private landowners to use specific areas within Prima (identified by the REEOP) at the time of landfill closure for restoration mitigation purposes, if the IWMD authorizes this activity. The area available for regional enhancement opportunities can be found both outside the ultimate limits of active landfilling operations and within completed areas of phased landfill development (Exhibit 4.3-9, Environmental Opportunities). Consistent with the planning approach used for the Pre-mitigation Plan, the conceptual REEOP design has been developed to accommodate the potential future expansion of Camino de los Mares through the southwestern portion of the project site. This plan primarily provides for native grassland and coastal sage scrub opportunities based on site soils and hydrology. Please note that once a final alignment for the extension of La Pata Avenue is approved, the REEOP will be revised accordingly to accommodate required rights-of-way.

The conceptual design of the REEOP is based on the following goals (Appendix H):

- Select areas that are currently disturbed or contain non-native plant species. Enhancement of these areas will involve the overall net habitat values of the area.
- Select potential restoration opportunity areas within the landfill property limits. Selection will be based on site constraints such as landfill final cover protection, recreation uses identified by the 2001 GDP, and long-term operation and maintenance including landslide remediation and/or stabilization.
- **Select appropriate habitat restoration types.** Select appropriate habitat restoration types based on soils, slope, aspect, hydrology, and other site conditions.
- **Incorporate viewshed protection elements.** Incorporate viewshed protection requirements from City/County MOUs and agreements with adjacent landowners.
- Maintain consistency with the commitments, reservations and restrictions contained in the agreement between the County and the Rancho Mission Viejo Company. Ensure that use of the restoration areas and opportunities identified in the REEOP upholds the Right of First Refusal (after County requirements are met) for the Rancho Mission Viejo Company within the Burdened Property.

Information contained within SEIR 597 will also include the presentation of additional biological resource data and hydrology analyses, and information relevant to the acquisition of master resource agency permits (California Streambed Alteration Agreement 1602; Section 404 of the CWA; Section 401 CWA water quality certification) addressing uses through the ultimate landfill buildout (year 2067).

4.3.2 TIMING OF THE PROPOSED PROJECT

Landfill operations within Zone 1 are expected to cease in approximately 2019 when Zone 1 is filled to capacity. Prior to that time, construction activities in Zone 4 will be initiated to ready the site for landfilling according to the Fill Phasing Plan presented within FEIR 575, as summarized above in Section 2.2.2 and illustrated in Exhibit 2.2-3. Upon initiation of excavation in Phase A of Zone 4, construction of the revised desilting system could begin with construction of the basins at 480 and 700 Elevations and the associated subsurface reservoir beneath the 480 basin. Stabilization measures and landfill support features affecting the limits of disturbance will generally proceed in tandem with excavation of each phase and will be determined based on the geotechnical conditions on each portion of the site.

4.3.3 REGULATORY REQUIREMENTS FOR IMPLEMENTATION

In addition to the County of Orange, as the lead agency for CEQA, there are other agencies that have jurisdiction over site landfill operations and biological resources within the PDL who will require new or revised regulatory permits. Permits issued by biological resource agencies will be needed to authorize impacts to biological resources during long-term landfilling operations through ultimate buildout and post-closure. Landfill operating permits will need to be revised to incorporate more detailed design and operations information for Zone 4.

Resource Agency Permits and Regional Conservation Programs

Biological resources within the 1,530-acre facility are governed by several regulatory agencies and applicable statutes and guidelines for which these agencies have jurisdictional responsibility, including but not limited to: the USFWS and the Federal Endangered Species Act (FESA); the CDFG and the California Endangered Species Act (CESA); *Fish and Game Code*, Section 1602; and the USACE and Sections 401 and 404 of the Federal CWA. These agencies can use the CEQA process to ensure compliance with FESA and CESA as well as obtain mitigation for non-listed plant and wildlife species. The applicable agencies, regulations, and terminology associated with biological resource protection and management are described below.

SEIR 597 provides an analysis of impacts for those environmental resources for which the Proposed Project could potentially result in "more severe impacts" over those identified in FEIR 575. Each topical section includes the following information: description of existing conditions; identification of thresholds of significance; analysis of potential incremental effects of the Proposed Project compared to the 2001 GDP and identification of more severe impacts; identification of a mitigation program, if required, to reduce the identified impacts; and identification of unavoidable significant impacts after mitigation.

A mitigation program identified to reduce potential project impacts may consist of Project Design Features (PDFs), Standard Conditions and Requirements, and Mitigation Measures (MM). These components will be the basis of the Mitigation Monitoring and Reporting Program (MMRP).

Project Design Features (PDFs). PDFs are specific design elements that have been incorporated into the project designs to prevent the occurrence of, or reduce the significance of, potential environmental effects. PDFs have been incorporated into the project and are identified in the mitigation section for each topical issue in order to ensure that they are included in the mitigation monitoring program to be developed for, and implemented as a part of, the Proposed Project, as required by CEQA.

Standard Conditions and Requirements. Standard Conditions and Requirements are based on local, state, and/or federal regulations or laws that are required independently of CEQA review and that serve to offset or prevent specific impacts. Typical Standard Conditions and Requirements include compliance with the provisions of the SCAQMD Rules, RWQCB, and IWMD regulations.

Mitigation Measures. Where a potentially significant environmental effect has been identified that would not be reduced to a level considered less than significant through the application of PDFs and/or Standard Conditions and Requirements, project-specific Mitigation Measures have been recommended. For example, in Section 5.5 of this SEIR, Biological Resources, several measures have been recommended to reduce potentially significant construction-phase impacts (i.e., construction noise and vegetation removal) to sensitive wildlife species, such as avoiding



Environmental Opportunities

Regional Environmental Opportunities Plan Prima Deshecha Landfill

1,100 1.100 550 Feet

Exhibit 4.3–9



R:/Projects/OCIWMD/J004/Ex4.3-9_banking_opp_011906.pdf

work during the breeding and nesting season of specific bird species and pre-construction site surveys for species presence.

Federal Endangered Species Act (FESA)

The Federal ESA of 1973 protects plants and animals that are listed by the federal government as "endangered" or "threatened." The FESA is implemented by enforcement of Sections 7 and 9 of the Act.

Section 7 applies to federal agency actions (permits, funding, etc.) and covers the activities of both private parties and public agencies, such as Section 404 permits issued by the USACE for construction work in waters or wetlands. Specifically, Section 7 imposes an affirmative duty on federal agencies to ensure that their actions (including permitting) are not likely to jeopardize the continued existence of a listed species (plant or animal) or result in the destruction or modification of critical habitat (50 CFR, § 402.01[a]). Both Sections 7 and 10(a) of the FESA allow or authorize "incidental" takes in accordance with the provisions of the FESA as described above, but only with a permit which may be obtained through consultation with the USFWS.

Section 9 makes it unlawful for anyone to "take" a listed animal, which includes significantly modifying its habitat. This law applies to both public and private parties; a landowner is not allowed to "take" an endangered animal or its habitat on his/her property without first obtaining the appropriate authorization to do so in accordance with the provisions of Section 7 or 10(a) of the FESA.

California Endangered Species Act (CESA)

CESA or Section 2080 of the *Fish and Game Code* prohibits "take" of any species that the CDFG determines to be an Endangered or Threatened species. Take is defined in Section 86 of the *Fish and Game Code* as "hunt, pursue, catch, capture, or kill, or attempt to hunt, pursue, catch, capture, or kill."

CESA allows for take that is incidental to otherwise lawful development projects. CESA emphasizes early consultation to avoid potential impacts on Rare, Endangered, and Threatened species and to develop appropriate mitigation plans to offset project-induced losses of listed species populations and their essential habitats.

Through permits or Memoranda of Understanding, the CDFG may authorize individuals, public agencies, or educational institutions to import, export, take, or possess any endangered species, threatened species, or candidate species of plants and animals. Take is authorized only after the applicant demonstrates that the impacts of a project will be minimized and fully mitigated. The measures required to meet this obligation must be roughly proportional to the impact of the authorized taking of the species and must be capable of successful implementation.

Clean Water Act – Section 404

Section 404 of the CWA regulates the placement of dredged and fill material into "waters of the United States," including wetlands. The CWA authorizes the issuance of permits for such discharges as long as the proposed activity complies with environmental requirements specified in Section 404(b)(1) of the CWA. Section 404 is the primary federal program that regulates activities in wetlands. The Section 404 program is administered by both the USACE and the U.S. Environmental Protection Agency (USEPA) with the USFWS, National Marine Fisheries Service (NMFS). Several state agencies also play important advisory roles to these regulatory agencies.

The USACE has primary responsibility for the permit program and is authorized, after notice and opportunity for a public hearing, to issue Section 404 permits. In evaluating individual Section 404 permit applications, the USACE determines compliance with Section 404(b)(1) guidelines and carries out a public-interest review. This review involves balancing such public-interest factors as conservation, economics, aesthetics, wetlands protection, cultural values, navigation, fish and wildlife values, water supply, and water quality. The USACE also considers comments received from the USEPA, USFWS, NMFS, and state resource agencies.

Section 404 regulates only the discharge of dredged or fill material into "waters of the United States." Discharges of dredged and fill material are commonly associated with activities such as channel construction and maintenance, fills to create development sites, transportation improvements, and water resource projects (such as dams, jetties, and levees). Excavation activities (e.g., mechanized land clearing, ditching, channelization, runoff from disposal areas, and others) also result in at least some discharge of dredged materials, and are thus regulated.

Discharges can be authorized by either individual or general permits under Section 404. If an individual permit is required, an application form describing the proposed activity is submitted to the USACE. Once a complete application is received, the permitting agency issues a public notice that contains the information needed to evaluate the likely impact of the proposed activity. Notice is sent to all interested parties, including appropriate government agencies at the federal, state, and local level, and others as requested. Any person may request that a public hearing be held to consider the application.

The USACE is authorized to issue general permits on a nationwide, state, or regional basis for categories of activities that have minimal individual and cumulative impacts. General permits are issued for five-year periods. They allow certain activities to occur without individual federal permit approval as long as the discharger complies with standard conditions issued by the USACE. General permits eliminate individual review and thus allow certain activities to occur with little, if any, delay or paperwork. Once issued, a general permit may be modified or revoked if the permitted activities are found to have had adverse environmental impacts. On a case-by-case basis, the permitting agency may invoke discretionary authority and require a discharger that would otherwise be covered by a general permit to apply for an individual permit.

To determine impacts to actual USACE and CDFG jurisdiction versus potential jurisdiction, Regulatory Specialists from RBF Consulting conducted a project-level jurisdictional delineation between May 5 and 30, 2003, to identify and quantify the extent of areas subject to the jurisdiction of the (1) USACE, pursuant to Section 404 of the CWA and (2) CDFG, pursuant to Section 1600 of the *Fish and Game Code* within the 1,530-acre facility. The 1987 *Wetland Delineation Manual* was used to evaluate potential USACE jurisdictional areas (USACE 1987), and *A Field Guide to Lake and Streambed Alteration Agreements: Section 1600-1607 California Fish and Game Code* was used to evaluate CDFG jurisdiction (CDFG 1994). The delineation identified impacts to 4.12 acres of waters of the U.S. of which 3.42 acres were designated wetlands under the jurisdiction of the CDFG. These methods are described in detail in Appendix F, which includes the wetlands delineations in their entirety. This delineation identifies the full extent of on-site areas and their resources that are under the jurisdiction of the USACE and CDFG and establishes a baseline from which impacts for the Proposed Project can be assessed.

For the project-level delineation of CDFG jurisdiction, the document *A Field Guide to Lake and Streambed Alteration Agreements: Section 1600-1607 California Fish and Game Code* (CDFG 1994) and input from CDFG Biologists during the verification process was utilized. The methodology utilized for assessment of riparian resources in the jurisdictional delineation is based on the following:

- Where all riparian habitat was included within the bank-full stream channel (e.g., riparian herb), the outermost limits of either the bank or riparian habitat was mapped as the limits of CDFG riparian jurisdiction/habitat.
- Where riparian habitat extended beyond the bank-full channel to the active flood plain, and did not extend outside the active floodplain, the outermost limits of either the active floodplain or riparian habitat were mapped as the limits of CDFG riparian jurisdiction/ habitat.
- Where riparian habitat extended beyond the active flood plain to active terraces, the outermost limits of the riparian habitat on the terrace was mapped as the limits of CDFG riparian jurisdiction/habitat. This latter case was most typically applied to southern coast live oak riparian forest. In some cases, particularly in "U"-shaped canyons, the limits of the active terrace was not always discernible. In such cases, coast live oaks (and, in a few instances, California sycamores) were included as riparian where they either:
 (1) exhibited roots that reached the banks of the drainage, thereby benefiting from the drainage or by providing stabilization for the banks (i.e., a benefit for the stream) or
 (2) where meaningful portions of the canopy overhung the stream, thereby providing for shading or litter (nutrient cycling) which would benefit the stream. Coast live oaks (and California sycamores) located above active terraces or (where terraces were not distinct) beyond where either roots or shading provided direct benefits to the stream, were not included as riparian vegetation.

Landfill Operating Permits

Full implementation of the Second Amendment to the 2001 GDP landfill plan will also require the following landfill regulatory agency permit actions:

- Amendment to Waste Discharge Requirements (WDRs) Order No. R9-2003-0306 from the Regional Water Quality Control Board, San Diego Region.
- Revision to Solid Waste Facility Permit (SWFP) No. 30-AB-0019 from the County of Orange Health Care Agency and concurred on by the California Integrated Waste Management Department.
- Permits to Construct and Operate Landfill Gas Control System facilities for ongoing operations from the South Coast Air Quality Management District.
- Annual Update to General Permit to Discharge Storm Water Associated with Industrial Activity Water Quality Order No. 97-03-DWQ issued by the State Water Resources Control Board.

The WDR and SWFP landfill operation revisions will be approved prior to operations in Zone 4. The subsections that follow describe the regulatory agency authority for landfill operating permits that will need to be revised for full implementation of Amendment No. 2 to the 2001 GDP.

California Integrated Waste Management Board (CIWMB)

All municipal solid waste facilities in California are required to have an SWFP issued by the local enforcement agency for the CIWMB (which, for the PDL, is the County of Orange Health Care Agency), and concurred on by the CIWMB. The SWFP places conditions on the operation, general design parameters, reporting, monitoring requirements, closure, and post-closure of the

facility in accordance with the *California Code of Regulations*, Title 27 (27 CCR). The PDL operates under SWFP No. 30-AB-0019, which was last revised in November 2005.

California Regional Water Quality Control Board (RWQCB)

The State Water Resources Control Board (SWRCB) requires municipal solid waste disposal facilities to obtain WDRs in accordance with 27 CCR. The San Diego RWQCB is the local agency under the SWRCB that has jurisdiction and authority to issue site-specific WDRs for the PDL. The PDL operates under WDR Order No. R9-2003-0306 (adopted in November 2003) and Amendment No. 1 to Order No. R9-2003-0306 (adopted on June 8, 2005).

The RWQCB also regulates municipal, industrial, and construction stormwater discharge requirements under the National Pollutant Discharge Elimination System (NPDES) program. To obtain authorization for industrial stormwater discharge, the landfill must comply with a General Permit to Discharge Stormwater Associated with Industrial Activity. A Stormwater Pollution Prevention Plan (SWPPP) and a Monitoring Program and Reporting Requirements (MPRR) have been prepared for the PDL in accordance with the requirements of Water Quality Order No. 97-03-DWQ. The MPRR monitoring is performed under Order No. 97-03-DWQ.

The SWPPP is updated annually, when necessary, to include various phases of construction when there is a change in construction, operation, or maintenance procedures, which may cause the discharge of significant quantities of pollutants to surface water, groundwater, or the local agency's storm drain system. Moreover, an amendment is filed if the site is found to be in violation of any conditions of the General Permit to Discharge Stormwater Associated with Industrial Activities.

It is anticipated that the activities associated with continued landfill operations proposed in the Second Amendment to the 2001 GDP will be covered by the site's existing General Permit to Discharge Stormwater Associated with Industrial Activity and that annual updates to the SWPPP and MPRR will address the future operations of the landfill. In addition to ongoing compliance with industrial and construction NPDES permit requirements, the IWMD will coordinate with the RDMD on compliance with the NPDES permit requirements of the County's Drainage Area Master Plan and associated Water Quality Management Plan, as necessary, for full implementation of the 2001 GDP.

South Coast Air Quality Management District (SCAQMD)

The PDL falls under the jurisdiction of the SCAQMD for the monitoring and control of gas emissions and migration and dust. Facilities to collect and destroy landfill gas emitted from the landfill are installed for the existing landfill and are planned for future development in the Zones 1 and 4 areas. The SCAQMD has issued: (1) a Permit to Operate the existing landfill gas condensate collection and storage system (No. R-F22337); (2) a Permit to Operate the flare/blower to incinerate the collected landfill gas (No. F22159); (3) a Permit to Construct and Operate the landfill collection system for WMU1 and Zone 1 (No. F38717); (4) a Permit to Construct the landfill gas combustion system (No. 1 is Permit No. 322414 and No. 2 is Permit No. 322415); (5) a Permit to Construct air pollution control equipment (No. 1 is Permit No. 322416 and No. 2 is Permit No. 322417). In addition, an air quality and landfill gas monitoring program that complies with SCAQMD Rule 1150.1 has been approved for the site and the Energy Recovery Facility (ERF) is operating pursuant to a permit that the SCAQMD issued under Rules 201 and 203. The site also complies with the following:

New Source Performance Standards/Emission Guidelines (NSPS/EG): On March 12, 1996, the United States Environmental Protection Agency (USEPA) promulgated standards of performance for new municipal solid waste landfills and emission guidelines for existing

municipal solid waste landfills. These standards guidelines for active landfills are intended to limit gaseous emissions in order to prevent public nuisance and possible detriment to public health that may be caused by exposure to such emissions. In order to comply with NSPS/EG requirements, annual reports will be submitted to the USEPA with a copy to the SCAQMD.

Title V: Title V is part of the 1990 Clean Air Act Amendments and consists of a single air permit, which consolidates and replaces all the previously issued air permits for a facility. The USEPA granted interim approval of the SCAQMD Title V program in February 1997 and the program became effective March 31, 1997. The SCAQMD program is called Regulation XXX and the PDL is in compliance.

Ozone (O₂) and Particulate Matter (PM10 and PM2.5): The USEPA reviews the National Ambient Air Quality Standards every five years. The standards for O_2 , PM10, and PM 2.5 have not been updated since 1997. New methods for controlling these pollutants are constantly being discovered. SEIR 597 includes up-to-date and appropriate control strategies for reducing the proposed project's O_2 , PM10, and PM 2.5 impacts to the greatest extent feasible.

Rule 1150.1 – **Landfill Gas Emissions:** The purpose of the current Rule 1150.1 for active and inactive landfills is to prevent public nuisance and possible detriment of public health that may be caused by exposure to landfill gas emissions. The SCAQMD has rewritten these rules to meet the federal NSPS/EG requirements.

Rule 431.1 – **Sulfur Emissions:** The purpose of this rule is to reduce sulfur oxides (SOx) emissions from the burning of gaseous fuels in stationary equipment and requires a SCAQMD-issued permit to operate. The SCAQMD rewrote the Rule to raise the average daily limit of 40 parts per million (ppm) to 150 ppm effective June 12, 1998.

Rule 402 – Nuisance: This rule prohibits annoying odors from landfill operations.

Rule 403 – **Fugitive Dust Emissions:** The purpose of this rule is to reduce the amount of particulate matter entrained in the ambient air as a result of man-made fugitive dust sources by requiring actions to prevent, reduce, or mitigate fugitive dust emissions.

County of Orange Health Care Agency, Environmental Health

The County of Orange Health Care Agency, Environmental Health is the Local Enforcement Agency (LEA) having jurisdiction over the PDL. The LEA issues and enforces the terms and conditions of the SWFP, enforces all pertinent sections of 27 CCR, and conducts regular inspections of the landfill. In addition to permit revisions, the site is subject to a permit review at least once every five years. Inspections are also performed monthly by the LEA; quarterly by the SCAQMD; at least annually by the SDRWQCB, and every 18 months by the CIWMB for compliance with permit conditions and regulatory standards under each agency's jurisdiction.

- Other subsequent actions for the site include:
 - o On-going CEQA Mitigation Monitoring;
 - o Liner Construction Design Reports for Each Phase of Development;
 - Final Circulation Element Permit Approvals, Design and Improvements;
 - Needs Analysis and Plans for Interim and Ultimate Recreational Uses;

- o Construction of Interim and Ultimate Recreational Improvements; and
- Preparation of Final Closure and Post-Closure Maintenance Plans.

4.3.4 RELATIONSHIP TO OTHER REGIONAL PLANNING EFFORTS

The Natural Community Conservation Planning (NCCP) Act

In 1991, the California Legislature established the NCCP program through the enactment of the 1991 NCCP Act (*Fish and Game Code*, Section 2800–2840). The purpose of the NCCP program is to provide long-term, regional protection of natural vegetation and wildlife diversity while allowing compatible land uses and appropriate development and growth. The NCCP process was initiated to provide an alternative to "single species" conservation efforts that were relied on under existing state and federal Endangered Species Acts (CESA and FESA) prior to the NCCP Act. The shift in focus from single species, project-by-project mitigation efforts to conservation planning at the natural community level was intended to establish regional protection for a range of species that inhabit a designated natural community.

The NCCP Act:

- **Defined the NCCP Program.** Natural Community Conservation Plan means the plan is prepared pursuant to subdivision (a) of Section 2810. The plan identifies and provides for the regional or areawide protection and perpetuation of natural wildlife diversity, while allowing compatible and appropriate development and growth.
- Authorized CDFG to Enter Into Agreements. The department may enter into agreements with any person for the purpose of preparing and implementing a Natural Community Conservation Plan to provide comprehensive management and conservation of multiple wildlife species, including, but not limited to, those species listed pursuant to Article 2 (commencing with Section 2070) of Chapter 1.5. The agreement shall include cost reimbursement provisions pursuant to Section 2840.
- Authorized NCCP Planning by Local, State, and Federal Agencies. Natural Community Conservation Planning may be undertaken by local, state, and federal agencies independently or in cooperation with other persons. The Plan shall be consistent with the agreement entered into pursuant to Section 2810 and shall be approved by the department for implementation upon meeting the standards established by the department for natural community conservation.
- Authorized CDFG to Prepare Non-Regulatory Guidelines. The department may prepare non-regulatory guidelines for the development and implementation of Natural Community Conservation Plans. The guidelines are exempt from Chapter 3.5 (commencing with Section 11340) of Division 3 of Title 2 of the State Government Code. The guidelines may include, but are not limited to:
 - Defining the scope of a conservation planning area.
 - Determining conservation standards, guidelines, and objectives for the planning area.
 - Appointing one or more advisory committees to review and make recommendations regarding the preparation and implementation of natural community conservation plans. The advisory committee membership may include representation from the local community near the plan area.

- Coordinating with local, state, and federal agencies, including the Trade and Commerce Agency.
- Incorporating public input.
- Ensuring compatibility with the federal Endangered Species Act (16 U.S.C., Section 531 et seq.).
- Obtaining approval of the natural community conservation plan by the department.
- Implementing provisions of the plan.
- Monitoring and reporting on plan implementation.
- Authorized CDFG to Permit Take of Identified Species. The department may permit the taking, as provided in this code, of any identified species whose conservation and management is provided for in a department-approved natural communities conservation plan.
- **Compensate the CDFG for Authorization Compensation.** The department shall be compensated for the actual costs incurred in participating in the preparation and implementation of natural community conservation plans.

The NCCP Act was designed as a voluntary program to encourage collaborative planning programs involving landowners, local governments, state and federal agencies, environmental organizations, and interested members of the public in the formation and approval of the NCCP. The Act serves as the foundation for the development of an NCCP program. However, since 1992, two bills regarding the NCCP program have become law: Assembly Bill (AB) 3446 (California Assembly 1996) and Senate Bill (SB) 1679 (California Senate 2000). Under AB 3446, Section 2825 of the *Fish and Game Code* was amended to make compensation permissive for CDFG participation in the NCCP Plan process. The Sher Bill added Sections 2801, 2811, and 2815 which required that any future NCCP planning agreement establish a process for collection of independent scientific input and analysis in the development of the plan, as well as the appointment of independent scientists for the development of the conservation guidelines. This bill also required the CDFG to establish a process for public participation throughout the development and review of any future NCCP plan.

Southern Subregion Natural Community Conservation Plan/Habitat Conservation Plan (SSNCCP)

At this time, a separate Joint EIR/Environmental Impact Statement (EIS) is being processed for the Southern Subregional NCCP (SSNCCP) to address the potential impacts associated with the various alternative strategies that affect the region surrounding the PDL site. The Joint EIR/EIS, a separate document from this SEIR, is being prepared pursuant to existing agency guidelines and the requirements of both CEQA and the National Environmental Policy Act (NEPA). The County of Orange is the lead agency pursuant to CEQA and the USFWS would be the federal lead agency pursuant to NEPA. The EIR/EIS addresses the purpose and need for the NCCP which is a Habitat Conservation Plan, describes the affected environment, and evaluates impacts to Identified Species and associated habitats that result from the NCCP/HCP and the proposed project alternatives.

Planning Area for the NCCP

The Southern Subregion covers approximately 132,000 acres of developed, agricultural, and undeveloped natural lands (including Prima Deshecha property) and comprises about 26 percent of the land within the County of Orange (Exhibit 4.3-10). However, 30 percent of the entire Southern Subregion (about 40,000 acres) is located within the Cleveland National Forest (CNF). Since the land within the CNF is already protected, it is not being addressed in the NCCP/HCP. The Southern Subregion includes all or portions of five cities: Mission Viejo, Rancho Santa Margarita, Lake Forest, San Clemente, and San Juan Capistrano. Public agencies and operating utilities that would be affected by the NCCP/HCP include, but are not limited to, the Santa Margarita Water District (SMWD) and Transportation Corridor Agencies (TCAs).

Participants

Landowners within the Southern Subregion, including both private and public agency owners, would be affected by the NCCP/HCP. Landowners within the Southern Subregion, including both private and public agency owners, that would be affected by the NCCP/HCP program and have participated in the process to date include:

- The Santa Margarita Water District (SMWD)
- Rancho Mission Viejo Company, LLC (RMV)
- The Transportation Corridor Agencies (TCA)
- City of Mission Viejo
- City of San Clemente
- The County of Orange

The largest undeveloped private ownership in the Southern Subregion is property owned and operated by the Rancho Mission Viejo Company.

The Draft NCCP/HCP Guidelines (Draft NCCP Guidelines) and the Draft Watershed and Subbasin Planning Principles (Draft Watershed Principles) were developed by the NCCP/Special Area Management Plan (SAMP) Working Group and are intended to provide guidance for decision-makers that are keyed to local biologic, hydrologic, and geomorphic conditions. Although considered a "work in progress," both the guidelines and principles represent the most current thinking regarding protection, restoration, and management priorities for the resources within the study area, and for this reason are discussed in this SEIR. The guidelines and principles have been subject to public input during public workshops associated with the NCCP/HCP and SAMP/MSAA programs and are available for review on the County website (County of Orange 2006).

The project site is located within the planned Southern Subregion NCCP area, and the IWMD, in conjunction with the Rancho Mission Viejo Company, has prepared a Plan in cooperation with the USFWS and CDFG that includes the development of a habitat reserve system. The NCCP will be formulated so it is consistent with the development of the Draft NCCP/HCP Guidelines and Draft Watershed and Sub-basin Planning Principles.

Covered Species

The proposed Southern Orange County Subregion National Community Conservation Plan/Master Streambed Alteration Agreement/Habitat Conservation Plan (NCCP/MSAA/HCP) identifies the following species for which regulatory coverage is being requested:

- Burrowing owl (*Athene cunicularia*)
- Coastal cactus wren (Campylorhynchus brunneicapillus couesi)
- Coastal California gnatcatcher (Polioptila californica californica)
- Cooper's hawk (Accipiter cooperii)
- Grasshopper sparrow (Ammodramus savannarum)
- Least Bell's vireo (Vireo bellii pusillus)
- Long-eared owl (Asio otus)
- Southwestern willow flycatcher (Empidonx trallii extimus)
- Tricolored blackbird (Agelaius tricolor)
- White-tailed kite (Elanus leucurus)
- Yellow-breasted chat (Icteria virens)
- Yellow warbler (*Dendroica petechia*)
- Arroyo toad (*Bufo californicus*)
- Western spadefoot toad (Spea [=scaphiophus] hammondii)
- California glossy snake (Arizona elegans occidentalis)
- Coast patch-nosed snake (Salvadora hexalepis virgultea)
- Northern red-diamond rattlesnake (Crotalus ruber ruber)
- Orange-throated whiptail (Aspidoscelis hyperythra [=Cnemidophorus hyperythrus] beldingi)
- Red coachwhip (*Masticophis flagellum piceus*)
- "San Diego" coast horned lizard (Phrynosoma coronatum [blainvillei population])
- Southwestern pond turtle (*Emys* [=*clemmys*] marmorata pallida)
- Arroyo chub (Gila orcutti)
- Partially-armored threespine stickleback (Gasterosteus aculeatus ssp. microcephalus)
- Riverside fairy shrimp (Streptocephalus woottoni)
- San Diego fairy shrimp (Branchinecta sandieogonensis)
- California scrub oak (Quercus berberidifolia)
- Chaparral beargrass (Nolina cismontana)
- Coast live oak (Quercus agrifolia)
- Coulter's saltbush (Atriplex coulteri)
- Many-stemmed dudleya (Dudleya multicaulis)
- Southern tarplant (Centromadia parryi var. australis)
- Thread-leaved brodiaea (Brodiaea filifolia)

The Notices of Intent (NOI) for the initiation of the NCCP/HCP were published by the Resource Agencies in 2001, with an anticipated release of draft environmental documents sometime in 2002. However, other regional planning efforts (e.g., the Riverside County Multiple Species Habitat Conservation Program) and the major fires experienced in the fall of 2003 required the resource agencies to re-prioritize their staffing assignments to address these competing projects. The draft EIR/EIS for the NCCP/MSAA/HCP was released for public review in July 2006.

Land Use Designations and Permitted Uses in the NCCP

PDL proposes to include two NCCP/HCP designations. The landfilling-related activity areas within Zones 1 and 4 are proposed to be designated as "Development" given the long-term nature of this type of use and compliance with all the existing and future state and federal regulations associated with landfills. The NCCP/HCP does not restrict uses in areas designated "Development," such as those identified in the Landfill Activities Matrix (Table 4.3-2).

TABLE 4.3-2 LANDFILL ACTIVITIES MATRIX

Landfill liner construction and maintenance.

Waste unloading, spreading, compacting

Landfill Cover Construction & Maintenance: routine construction, inspection and repair, soil testing, soil importation and stockpile, surveying, regrading, compaction, slope repair, weed abatement, revegetation and settlement or crack repair, seeding, straw mat, erosion control blankets, geotechnical investigations, trenching, boreholing and all routine maintenance and repair of facilities that do not result in permanent loss of existing natural vegetation. Daily, intermediate, and final cover placement

Borrow site grading, earth moving, clearing and grubbing, access for grading efforts

Refuse excavation and replacement

Cleanup and remediation for unacceptable materials

Leachate and groundwater recovery, disposal and treatment system construction and installation

Landfill Water Monitoring System Construction & Maintenance: groundwater monitoring-well installation and abandonment, groundwater and stormwater monitoring, well redevelopment, dedicated pump installation

Water supply system construction and maintenance

Landfill Landscape and Irrigation System (pipeline, pump, valves, sprinklers) Construction & Maintenance: routine construction, inspection and maintenance, weed control, fertilization, rodent control, reseeding and mulching, system replacement and repair

Landfill Liquid Management System (pumps, sewer, sumps, storage tanks, pipes, power supply, electrical controls, groundwater extraction wells, leachate recovery, water, leachate or landfill gas condensate treatment unit) Construction & Maintenance: routine construction, inspection and maintenance, liquids disposal, sampling and monitoring

Landfill Drainage Facilities (bench drains, inlet structures, down drains, deck drainage, concrete channels, pipes, ditches, desilting basin, etc.) Construction & Maintenance: routine construction, inspection and repair, regrading, debris or sediment removal, erosion control, grouting, structures repair and construction, weed abatement.

Landfill Gas Recovery System (gas extraction wells, headers, laterals, valves, well heads, burners, flares, gas to energy plant, gas condensate, power supply) Construction & Maintenance: routine construction, inspection and maintenance, gas monitoring, gas well installation and repair, pipeline repair, underground fire control, well head repair, and adjustment, condensate injection into flares

Landfill Gas Control System (perimeter probes, perimeter wells, piping, pumps, power supply) Construction & Maintenance: routine construction, inspection and maintenance, gas monitoring, surface emissions monitoring, pipeline repair, probe and well installation, installation of horizontal and vertical collectors, construction of headers, weather station maintenance and repair.

Gas recovery facility installation, operations and maintenance

Dust control, fire control, vector and bird control, litter control

Remedial grading and repair for landslides and other natural occurrences

Landfill Utilities and Communications Facilities; installation, routine maintenance, repair, relocation or replacement Landfill access road, construction and maintenance.

Landfill Survey Monument Construction and Maintenance: routine construction, inspection and maintenance, survey, monument replacement or repair

Landfill Soil Excavation for cover repair, drainage and erosion control, landfill gas emission control, biological surveys, archeological and paleontological surveys and recovery.

Landfill Perimeter Fence Construction and Maintenance: routine construction, inspection and maintenance, replacement and repair.

Other landfill construction and maintenance activities required by law and regulation, including but not limited to: aerial or ground survey, landslide remediation, drilling, moisture probe installation, recycling, utility, fee booth, scales, field office and heavy equipment maintenance facility, and site security

Future uses including: mitigation, open space, regional park, etc.

Notes: Attempts will be made to undertake activities that impact vegetation outside the breeding/nesting season, including activities mandated by regulation or law affecting public health, safety, and welfare.

Activities are subject to change based upon regulation changes generated by the various solid waste regulatory agencies

Source: Orange County Integrated Waste Management Department

In addition, areas south of Zone 1; portions of Zone 3 outside Segunda Deshecha; and Segunda Deshecha (in its entirety) and the areas along the southern, eastern and northern boundaries' portions of Zone 4 (outside of any refuse disposal area) are proposed to be designated as "Supplemental Open Space" (SOS) by the NCCP program.

Function of Supplemental Open Space

Supplemental open space would:

- contribute directly to the long-term protection of Covered Species and their habitats;
- contribute to long-term subregional biological diversity;
- provide potential refugia habitat in the event of fire or other natural disturbances;
- provide supplemental connectivity between geographic elements of the Habitat Reserve; and
- contribute to the long-term protection of important abiotic (non-biological) resources and processes.

Definition of Supplemental Open Space

SOS would be protected and, if currently managed outside the Habitat Reserve Management Plan (HRMP) (i.e., the Starr Ranch Sanctuary and designated portions of the Prima Deshecha Landfill), would continue to be managed in accordance with current management practices. The management obligations for these areas are limited to current management approaches and the conditions imposed as part of prior or anticipated regulatory approvals. Where no existing habitat management is provided and/or required, no new management obligations are incurred under this NCCP/MSAA/HCP.

Take within the SOS would be authorized only as specifically called out by the NCCP/MSAA/ HCP of the Implementation Agreement (IA), primarily for infrastructure and landfill facility construction, operation, and maintenance. No take is authorized in areas designated as SOS for general residential, commercial, or industrial activities. If the affected landowners and local jurisdictions choose to seek additional take authorization for activities located within any of these areas in the future, they would be required to undertake separate review by CDFG and USFWS to determine compliance with applicable state and federal requirements under FESA and the NCCP Act.

Designated SOS lands located outside the proposed Habitat Reserve would complement the functions and values of the Habitat Reserve by: (1) protecting additional vegetation communities that currently support planning species and Covered Species and (2) preserving significant wildlife corridor and habitat linkage areas that contribute to subregional biodiversity.

Permitted Uses Within Supplemental Open Space

The SOS-designated areas within the PDL are intended to represent those portions of the landfill where no waste disposal is currently being proposed. These areas would not likely be affected by existing and future landfill operations and are intended to be preserved in a natural condition to the maximum extent possible. However, since this refuse disposal facility is expected to be in operation until 2067, and in landfill post-closure maintenance beyond 2067, it cannot be known with absolute certainty whether some of the areas designated as SOS may be needed for on-going and post-closure maintenance after landfill closure. In addition, the designated SOS will accommodate habitat restoration from landfill impacts as well as other regional restoration opportunities. Therefore, the following activities would be permitted within the areas designated as SOS within the PDL boundaries:

- Install, operate, maintain, repair and/or replace roads, public utility lines and associated improvements, and flood control devices.
- Install, operate, maintain, repair and/or replace trails, parks and related improvements and/or engage in any other recreation uses.
- Permit livestock grazing.
- Continue the use of the SOS area for any purposes provided for in any existing easements of record.
- Conduct prescribed fires allowed by the local fire authority for health and safety reasons and allow access to the SOS in order to fight or mitigate fires.
- Conduct scientific research.
- Perform remedial grading with vegetated slopes and utilize native species.
- Conduct mitigation actions as defined in the RMV Settlement Agreement with respect to the implementation of the 2001 GDP.
- Conduct any and all activities and operations as may be necessary to comply with applicable laws in connection with the landfill's closure and post-closure activities.
- Conduct biological mitigation pursuant to any approved NCCP, Habitat Conservation Plans and other similar plan of conservation.
- Perform site grading and/or soil filling (to maximize capacity) in support of landfill operations.

Any disturbances within areas designated as SOS (including the Pre-Mitigation areas and/or Regional Environmental Enhancement areas that involve the temporary removal of coastal sage scrub, southern needlegrass grassland, or riparian habitat) will be restored through the application of the appropriate hydroseed mix during the next growing season following the completion of the permitted use activities. The hydroseed application and subsequent three-year maintenance program (removal of non-native invasive plant species) shall constitute full compliance with the provisions of the NCCP/HCP. No further mitigation will be assessed against IWMD by the regulatory agencies.

Special Area Management Plans/Master Streambed Alteration Agreement (SAMP/MSAA)

A Special Area Management Plan (SAMP) is designed to be conducted in geographic areas of special sensitivity under intense development pressure. Interagency, public, and stakeholder involvement is an essential part of the SAMP. There are two main goals of the SAMP process: to establish a watershed-wide aquatic resource reserve program and to minimize individual and cumulative impacts of future projects in this watershed. The SAMP process will identify areas that will be protected and preserved, as well as areas where future activities would be allowed to occur, provided that these activities meet specific criteria developed for protection of the watersheds. As part of this process, the CDFG is preparing an MSAA in cooperation with USACE that will reflect the provisions of the final SAMP.

In consideration of the lifespan of the PDL and the need to resolve the long-term regulatory permitting requirements for this facility, IWMD intends to pursue the approval of an MSAA for this project (separate from that being prepared for the final SAMP).

As previously noted, the PDL is located outside the defined San Juan Creek and western San Mateo Creek SAMP/MSAA boundaries. The intent in describing the SAMP/MSAA planning process is to demonstrate how IWMD will contribute to the implementation of these planning efforts. Specifically, the contribution includes restoration and enhancement of a portion of San Juan Creek in the uppermost part of that watershed through the removal of exotic invasive plant species. These exotics, if allowed to continue to grow and expand into other natural areas within the drainage, would significantly reduce the long-term biological functions and values of that
watershed by displacing native habitat suitable for sensitive species such as the arroyo toad and the least Bell's vireo.

Purpose of the SAMP/MSAA. The purpose of the SAMP/MSAA (being prepared jointly by the USACE and CDFG as lead agencies) is to provide for the protection and long-term management of sensitive aquatic resources (biological and hydrological) on a landscape or watershed level. Aquatic resources in the watershed include creeks, seeps, vernal pools, alkali meadows, freshwater marshes, and riparian wetlands. To the extent practicable, state and federal waters (including wetlands) would be avoided and unavoidable impacts would be minimized and fully mitigated under the SAMP/MSAA. The SAMP/MSAA would also be designed to enable reasonable economic activities and development to be permitted within the study area portions of the San Juan Creek and western San Mateo Creek watersheds consistent with the requirements of federal and state laws (CWA, Section 404 and California Fish and Game Code, Sections 1600 et seq.).

When complete, the SAMP would provide the foundation for long-term planning and regulatory actions under Section 404 of the federal CWA by the USACE for projects under its jurisdiction within that planning boundary.

The key objectives of the SAMP/MSAA for the San Juan Creek and western San Mateo Creek watersheds are: (1) to evaluate the extent and condition of existing aquatic resources in the project area; (2) to identify and evaluate alternative land development scenarios that have been developed as part of the NCCP/HCP and SAMP/MSAA processes; (3) to address, programmatically, water quality issues raised under state and federal laws; and (4) to develop a reserve program and comprehensive management plan for the reserve areas in order to preserve and enhance existing aquatic resources. As to this last point, the SAMP/MSAA would provide a restoration plan for enhancing and protecting jurisdictional state and federal waters' aquatic resources in the watersheds.

A separate Joint EIR/EIS is in process, and will address the potential impacts associated with the SAMP/MSAA. The Joint EIR/EIS is a separate document from this Program EIR and is being prepared pursuant to existing agency guidelines and the requirements of CEQA and the NEPA. The CDFG is the lead agency pursuant to CEQA and the USACE is the federal lead agency pursuant to NEPA. The EIR/EIS addresses the purpose and need for the SAMP/MSAA, describes the affected environment, and evaluates impacts on associated habitats resulting from the SAMP/MSAA program and project alternatives. In support of the SAMP/MSAA, the USACE conducted a landscape-level delineation to identify areas of potential USACE and CDFG jurisdiction along with the mapping of areas of potential wetlands and riparian habitat within the SAMP/MSAA study area.

SAMP Study Area. The primary difference in the study areas for the NCCP and the SAMP is the San Clemente Hydrologic Unit. This unit is excluded from the SAMP study area because the unit does not drain to either the San Juan Creek or the San Mateo Creek. The study area for the SAMP covers approximately 107,350 acres within the San Juan Creek and western San Mateo watersheds, including the Cleveland National Forest (Exhibit 4.3-10). Actions within the forest are subject to a separate planning process and would not be addressed in the SAMP/MSAA, except to the extent that coordination between the two programs would be necessary to implement management actions within the San Juan watershed (e.g., Arundo control). The SAMP study area includes all or portions of the following six cities: Mission Viejo, Rancho Santa Margarita, Laguna Hills, Laguna Niguel, Dana Point, and San Juan Capistrano. Public and operating agencies affected by the SAMP include, but are not limited to, the SMWD and TCA.

The IWMD intends to address the long-term impacts to resources under the jurisdiction of the CDFG and USACE that would result from the ultimate build out of Prima through the eradication

of *Arundo donax* at a ratio of 1:1 within a portion of San Juan Creek located within Caspers Wilderness Park. This restoration will include a one-time payment to a non-profit entity to be created, that among other things, will be responsible for implementing a comprehensive exotic plant species eradication program within San Juan Creek. Prima Deshecha Landfill is not located within the San Juan Creek/San Mateo Creek SAMP boundary. In addition, opportunities to address impacts to resources under CDFG and USACE jurisdiction are limited within the project site. The study boundaries currently proposed for the NCCP/HCP and SAMP are depicted in Exhibit 4.3-10.

It is the intent of the IWMD to seek: (1) certification of SEIR 597; (2) approval of Amendment No. 2 to the 2001 GDP; (3) approval of an Individual Permit from the USACE pursuant to Section 404 of the CWA and associated Section 7 Consultation/Biological Opinion for potential effects to the least Bell's vireo and California gnatcatcher; and (4) an acquisition of an MSAA pursuant to Sections 1600–1603 of the *California Fish and Game Code* while the NCCP/HCP and SAMP programs are in process. This will enable coordination of the proposed GDP project with planning for these other regional programs. It should be noted that the NCCP/HCP and SAMP programs may not be completed by the date of the County action on the Proposed Project.

Although approval of Amendment No. 2 to the 2001 GDP will likely occur in advance of approvals and subsequent authorizations under the NCCP/HCP or SAMP, the Proposed Project has been developed consistent with the planning guidelines and conservation strategies established for these regional planning efforts. Therefore, the Proposed Project and associated SEIR can move forward without jeopardizing the preparation of the NCCP/HCP and SAMP.

La Pata Avenue Extension/Improvements

As described more fully in Section 8.0, the County's Resource Development and Management Department (RDMD), in coordination with the cities of San Clemente, San Juan Capistrano, and the IWMD, are currently conducting a Feasibility Study, Project Design Report, and EIR for the extension of La Pata Avenue from Ortega Highway (through the PDL property) to Calle Saluda in San Clemente. This work will ultimately generate a final alignment for the La Pata Avenue extension. This circulation project, although occurring in close geographic proximity to the proposed project for Amendment No. 2 at the PDL site, has a different project purpose and implementation timeframe from Amendment No. 2. Accordingly, both projects will, out of necessity, have their own separate environmental documentation and regulatory permitting requirements and will proceed on very different schedules.

Owing to the complexities and differences in implementation horizon associated with various La Pata Avenue alignment alternatives and the long-term implementation schedule for these improvements, the baseline assumption for the Prima Amendment No. 2 project to be presented in SEIR 597 will be the La Pata Avenue alignment as shown in the 2001 GDP. However, a sensitivity analysis of the current La Pata Avenue alternatives under consideration within the La Pata Avenue Feasibility Study has been included within SEIR 597 in Section 8.0, Cumulative Impacts, for information. As it is a stated goal of the 2001 GDP to accommodate adopted MPAH arterial highway alignments through the site, none of the Proposed Project alternatives presented within SEIR 597 will preclude the development of a full range of alternatives for the La Pata Avenue extension project. Close coordination between the two project efforts will continue throughout the environmental permitting process with RDMD leading and coordinating the La Pata project effort.



SECTION 5.0 ENVIRONMENTAL ANALYSIS

5.1 INTRODUCTION

This section presents the environmental impact analysis for those resource categories potentially affected by the Proposed Project. The potential effects to these resources required detailed analyses to determine whether the Proposed Project resulted in "no substantial change" over those impacts identified in FEIR 575, or if there were "more severe impacts" or new significant impacts to these resources.

All mitigation measures from the 2001 GDP EIR 575 remain as project commitments that apply to the Proposed Project, as modified by Amendment No. 2. These mitigation measures from EIR 575 are reiterated in the sections below under "Previously Adopted Mitigation." Additional mitigation measures, if any, are indicated under "Mitigation For Impacts Associated with Amendment No. 2 to the 2001 GDP." For Hydrology and Water Quality, some of the previously adopted mitigation measures were included in the Geology section of EIR 575, and therefore have slightly varying numerical designations.

Per CEQA Section 15125, an environmental setting must be presented to serve as a baseline from which to determine the significance of proposed project impacts. This discussion also provides the basis for an understanding of the regional context for the project. Existing conditions are described for each resource category below including the environmental setting.

The County of Orange, Integrated Waste Management Department (IWMD) is responsible for the implementation of all mitigation measures. Air quality mitigation measures will be implemented immediately and continuously upon certification of Amendment No. 2 and SEIR 597. All other mitigation measures will be implemented prior to either the construction of the Proposed Project or the initiation of Proposed Project impacts. As the Pre-mitigation Plan is designed to ensure no temporal loss of habitat prior to impact, the elements of this plan will be in place before the initiation of landfill operations in Zone 4.

5.2 <u>GEOPHYSICAL</u>

This section provides a summary of the information contained in Section 4.2 of EIR 575 and information associated with landslide remediation actions on site as contained within the *Prima Deshecha Landfill Geotechnical Investigations Report – Zone 4* (GeoLogic Associates, 2002). Both documents are available at the County of Orange Integrated Waste Management Department (IWMD) offices in Santa Ana, whose address is included in Section 1.1.6, and are listed as references in Section 12.0 of this report.

The information presented herein regarding impacts and potential mitigation measures for the Proposed Project buildout is based on data and mapped information contained within these reports. All requisite engineering analyses were performed per the methodology outlined in EIR 575, which includes all additional technical references pertaining to the analyses.

The methodology employed in conducting the impact analysis for geophysical resources was based on review of the references cited above and readily available updated geologic maps and accompanying data.

5.2.1 EXISTING CONDITIONS

<u>Geology</u>

The maximum change in relief across the Prima Deshecha Landfill (PDL) site is about 895 feet, with elevations that range from about 1,125 above mean sea level (msl) in the northeastern corner to about 230 feet above msl where the main channel of Prima Deshecha Cañada exits the southwestern corner of the property.

The PDL site is situated in the western foothills of the Santa Ana Mountains, which are part of the Peninsular Ranges Province of southern California. Bedrock materials exposed in the area consist of predominantly Tertiary marine sediments from the middle to late Miocene period. These sediments are comprised, from oldest to youngest, of the San Onofre Breccia, Monterey Formation, and Capistrano Formation (Exhibit 5.2-1). Collectively, these sedimentary bedrock units form a sequence of fairly impermeable breccia, sandstone, siltstone, claystone, and shale. On the surface, these bedrock materials are overlain by locally derived landslide deposits, soil debris slumps, and landslides; modern alluvial deposits in the main drainage channels; and various types of native soils (including colluvium and slopewash). These soil deposits cover much of the finer claystone and siltstone units.

The distribution, lithology, and engineering properties of the various geologic units exposed on the site are summarized below. This information has been summarized from FEIR 575.

Geologic Units

There are 7 geologic units at the PDL contributing to physical site conditions that affect the Proposed Project (Exhibit 5.2-1). Brief summaries of these units are listed below; detailed descriptions of these can be found in FEIR 575.

<u>San Onofre Breccia</u>. San Onofre Breccia is found in the north-central portion of the eastern half of the site and along the eastern border; it is a fairly resistant feature and usually forms steep cliffs and rugged terrain. Excavation of the San Onofre Breccia varies from "workable" (with some difficulty with heavy power equipment) to lesser weathered "hard" portions which likely require blasting to excavate. Slope stability is considered fair-to-poor along shear/fault zones, but generally good elsewhere. Permeability is primarily controlled by fractures and joints, but is moderately porous in highly weathered portions near the ground surface.

<u>Monterey Formation (Tm)</u>. The Monterey Formation is exposed throughout the eastern portion of the site and is largely covered by locally derived landslides. Landforms associated with areas underlain by this formation are represented by rounded, gently rolling hills with steep sloughing slopes. The majority of the Monterey Formation is easily excavated with light power equipment and the silicified portions and sandstones can generally be readily excavated with heavy power equipment. Slope stability of the formation is very poor, permeability is very low and expansivity is high.

<u>Capistrano Formation (Tc)</u>. The Capistrano Formation is exposed throughout the western twothirds of the study area and is covered by a number of large landslides composed of materials from the Capistrano Formation. Landforms associated with this formation are characterized by moderate steepness with well-rounded topography. The formation's overall slope stability is poor, as noted by the number of large landslides which are derived from it, and permeability is generally low.

Landslide Deposits (Qls). Landslides derived from the Capistrano and Monterey Formations cover at least 50 percent of the site area and vary in size from small surficial slumps to large



Geologic Features

Prima Deshecha Landfill Supplemental Environmental Impact Report 597





R:\Projects\OCIWMD\J004\SEIR 597\Graphics\ex5.2-1_geofeat_080306.pdf

landslide masses up to 120 acres in size. These landslides are very problematic relative to existing and future landfill operations.

<u>Alluvium-Slopewash (Qa)</u>. Alluvium-slopewash materials are generally restricted to the lower reaches of the larger tributaries and along the main drainage course of Prima Deshecha Cañada. The permeability is generally moderate to high except in silty clay lenses and layers, and the silty clay portions are also expansive.

<u>Man Made Fill (af)</u>. Class III landfill materials occupy two areas on the site. Waste Management Unit 1 (WMU1) occupies approximately 160 acres and is located in the westernmost portion of the site. Waste Management Unit 2 (WMU2) occupies approximately 25 acres and is located in the north-central portion of the site. The average depth of refuse in WMU1 is estimated to be approximately 150 feet (GLA 2002). WMU2 has a depth of about 100 feet and is not currently accepting landfill materials.

Geologic Structure and Seismicity

The PDL site is located within the Peninsular Ranges Province, characterized by north- to northwest-trending folds and fault systems. Ongoing seismic activity and geologically young fault features show that the Province is tectonically active. The eastern half of the site is crossed by a series of north to northwest-striking, west-dipping normal faults associated with the Cristianitos fault which is located near the eastern limit of Zone 4. Branches of the Cristianitos fault include the Forster fault which crosses through the center of the proposed Zone 4 development, and several other unnamed synthetic and antithetic faults that also cross Zone 4. No significant faulting has been mapped in the proposed Zone 1 development area.

While the majority of bedrock in the Zone 1 area is positioned on a northwest dipping homocline, 2 large-scale folds have been interpreted within the Zone 4 area. The first is a north-plunging syncline that is bordered to the south by the Forster fault. The second fold is a west-plunging syncline located in southwestern Zone 4, west of the Forster fault that appears to continue westward through the southern Zone 1 area (south of Prima Deshecha Cañada). In the Zone 4 area, the axis of this west-plunging syncline is overlain by a large, west-directed landslide complex that appears to have formed, at least in part, in response to the structure of the syncline. In addition to the large-scale structures, small-scale folding is prevalent in the Monterey Formation and less common in the Capistrano Formation.

The majority of the joint and fracture planes observed at the surface were open as much as several inches, but are believed to decrease substantially at shallow depths below the ground surface. Jointing in the San Onofre Breccia is likely to be responsible for the creation of pathways for groundwater movement and the formation of a spring located at the base of these cliff-forming rocks. The cliff-forming rocks and spring are located in the central portion of the Zone 4 landfill area on the site as shown in Exhibit 5.2-1.

There are no known Holocene (i.e., active) faults that trend toward or through the site. The seismicity of the region, based on instrument recordings of earthquakes since 1932, is relatively low compared to most of southern California. The most important faults for seismic hazards are those that are closest to the site and that are active or potentially active. These include the Newport-Inglewood (2 miles), Whittier-Elsinore (21 miles), San Jacinto (35 miles) and San Andreas (51 miles) faults.

The northwesterly trending Forster Fault and its associated branches traverse the eastern half of the site. Although the Forster Fault and its fault branches are not known to be active, a number of landslides on the site can be attributed, at least in part, to the presence of these faults. In many cases, the faults occur in close proximity to the headscarp or marginal region of several landslides.

The northwesterly trending Cristianitos Fault, located about 225 feet east of the site at its closest approach, is not considered to be active (i.e., no movement in the last 11,000 years, based on criteria established by the California Department of Mines and Geology [CDMG]).

Seismically induced seiche and tsunami are considered a remote possibility on the site due to the topographic location of the site and its distance from the ocean and large inland bodies of water.

Existing landslides may be reactivated during strong earthquake-induced ground motion, and new landslides could develop in the bedrock of the Capistrano and Monterey Formations. Thick accumulations of soils on the flanks of many of the ridges and slopes are also considered prone to landsliding during a strong seismic event in the region. Rock falls from the steep ridges in the Zone 4 development area may also occur in response to strong ground motion.

<u>Soils</u>

Soils on the site have been identified as belonging generally to the following soil series: Alo, Calleguas, Balcom, Bosanko and Botella (Wachtell, 1978). Each of these is described below.

<u>Alo Soils</u>. Alo soils overlie the Capistrano Formation. The Natural Resources Conservation Service (NRCS) classifies these soils as clay, which is relatively expansive (i.e., it has a high shrink-swell potential), easily erodible, occurs on slopes of 9 to 75 percent, exhibits very slow infiltration rates of 0.15 to 0.51 centimeters per hour and is typically up to 40 inches thick.

<u>Calleguas Soils</u>. Calleguas soils generally overlie the siltstone and sandy bedrock of the Capistrano Formation and portions of the Monterey Formation, are classified as clay loam, and are moderately expansive, easily erodible, occur on slopes of 9 to 75 percent, and exhibit very slow infiltration rates of 0.15 to 0.51 centimeters per hour.

<u>Balcom and Bosanko Soils</u>. Balcom and Bosanko soils typically overlie the Monterey Formation. These soils are classified as clay and clay loam, which are highly expansive, easily erodible, occur on slopes of 9 to 75 percent, and have very slow infiltration rates of 0.15 to 0.51 centimeters per hour.

<u>Botella Soils</u>. Botella soils occupy the major alluviated drainage on the site. These soils are classified as silty clay loam, which contains moderately expansive, fine-grained layers and lenses defined as slopewash materials. This soil is highly erodible, and has infiltration rates of 0.15 to 5.1 centimeters per hour.

In summary, all the surficial soil materials on the site are subject to erosion and mass failure through slumping on moderate-to-steep slopes. Experience has shown that on-site soils are generally subject to collapse and hydroconsolidation upon placement of overlying loads such as structural foundations and embankment fills. The surficial soil materials on the site are also prone to varying degrees of expansion. Although these soils are considered suitable for use as daily and intermediate landfill cover, they are not suitable for low permeability cover/liner materials due to the abundance of organic-rich materials (e.g., roots) and inherent low strengths.

5.2.2 SIGNIFICANCE CRITERIA

The CEQA Guidelines indicates a project will normally have a significant effect on the environment related to geology, seismicity, soils, and groundwater if it will "...expose people or structures to major geologic hazards...," or "...substantially degrade or deplete groundwater resources..."

For the purposes of this SEIR, the Proposed Project was determined to have a significant effect on the environment related to geology, seismicity, or soils if any feature of the Proposed Project created an impact which could not be designed to existing seismic standards or if any feature exceeded the thresholds as defined by the environmental checklist issued by the County of Orange.

5.2.3 POTENTIAL IMPACTS

Per the County of Orange Environmental Checklist, the Proposed Project would have a significant effect on geophysical resources if it would:

(1) Expose people or structures to potential substantial adverse effects, including the risk of loss, injury, or death involving rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault; strong seismic ground shaking; seismic-related ground failure, including liquefaction; landslides.

Proposed Project impacts from seismic activity are expected to be less than significant as the engineering design of new project features will take seismic design standards established in 27 CCR into account, resulting in remediation measures that provide stability under the design earthquake loads. Secondary seismic impacts could include differential settlement, which will also be appropriately factored in to project design. Landfill grading and cut-and-fill slopes remain generally the same as that analyzed within the 2001 GDP, and do not change with the Proposed Project; therefore there is no substantial change from previous analyses identified within the 2001 GDP.

(2) Result in substantial soil erosion or the loss of topsoil.

The Proposed Project does not result in substantial soil erosion or the loss of topsoil in significant amounts over that identified within the 2001 GDP. There is no additional material that is proposed for off-site handling or disposal due to the Proposed Project. All excavated material generated from the construction of buttress fills may be temporarily stockpiled on site and then used as backfill. This material will be handled similarly to the project features of the 2001 GDP.

Impacts to native soils are also expected to be less than significant as native soils temporarily removed due to landslide excavation will be used on site as compacted backfill material.

All mitigation measures and best management practices (BMPs) defined within the GDP will be applied to the Proposed Project, and will result in no substantial change from previous analyses within the 2001 GDP.

(3) Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse.

The preceding discussion presents the technical background for slope instability at the project site due to numerous geologic conditions. Slope instability at the project site was acknowledged within EIR 575 which states:

...given the instability of the various bedrock units and the inherent instability of the numerous landslides on the site, hazards relating to new slope failures and, perhaps more importantly, reactivation of existing landslides, could occur. In fact, portions of a number of the larger landslides in the eastern portion of the site display evidence of recent movement under natural conditions. Slope stability analyses undertaken to support the Landfill Master Plan have resulted in the development of subgrade and final landfill fill slopes which have acceptable factors of safety under static and earthquake-loading conditions. Future development of the landfill will result in total or partial removal of landslide deposits that would underlie the landfill, and engineered fills have been designed to stabilize slopes that would otherwise be prone to failure.

Geology, geologic structure, jointing and fracture planes, soil characteristics, and old landslide complexes and deposits all interact to create a significant risk of slope movement on site. As indicated above, although this information was known and incorporated into the project design for the 2001 GDP, the areal extent of the future landslide remediation activities were not fully identified. As discussed in Section 3.0, Purpose and Need for Amendment No. 2 to the 2001 GDP, one of the primary purposes of the Proposed Project is to provide for the areal extent of landslide remediation activities around the PDL so that landfill operations can be permitted in the future. Without the ability to provide for adequate slope stabilization measures, agency and operating permits for full development of the site would not be approved; this would seriously truncate the currently planned life and capacity of the landfill facility. With appropriate design and construction of landslide-stabilization features, the potential significant effects of these hazards of the Proposed Project can be avoided or substantially reduced.

Areas of potential slope stabilization within the updated limits of disturbance that are required for continuation of landfill operations are identified on Exhibit 4.3-1 as the hatched portions on the perimeter of Zones 1 and 4. These areas will likely be stabilized by buttress fills and/or a shear key, with a substantial fill and/or shear key required to stabilize the largest landslide complex in Zone 4. The Zone 4 shear key concept was a part of the project described for implementation in the 2001 GDP, but a review and update of the site's available geotechnical information has been recently conducted to better define the potential limits of disturbance required for future landslide remediation. Impacts due to known conditions from landslide features or slope failure are planned to be remediated through implementation of the Proposed Project. As these incremental remediation features will be located largely outside the refuse footprint, there will be no impacts upon landfill subsidence.

Geotechnical factors affecting the design of the newly proposed desilting system for Zone 4 and its associated desilting basins have not changed from the design considerations required by the 2001 GDP; therefore, the basins, as described in the Proposed Project, will incorporate similar measures for geotechnical stability. These perimeter basins have been sited in areas believed to be stable or in areas to be stabilized as part of the landfill design.

All mitigation measures that were incorporated into the 2001 GDP project will be applied to the Proposed Project; accordingly, there is no substantial change in effect over that which was identified in the 2001 GDP.

(4) Be located on expansive soils, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property.

The Proposed Project will not be located on expansive soils resulting in a substantial risk to life or property. Inappropriate expansive soils will be removed and used as backfill material.

(5) Have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal system where sewers are not available for the disposal of wastewater.

The Proposed Project will not require the use of septic tanks or alternative wastewater disposal systems.

Accordingly, as none of the elements of the Proposed Project will incrementally expose people or structures to major geologic hazards over those identified within EIR 575, and all elements will be designed to the existing seismic standards stated in 27 CCR. There is therefore no substantial change upon geophysical resources within the project site from the Proposed Project.

5.2.4 MITIGATION MEASURES

Previously Adopted Mitigation

The following mitigation measures are currently in place for impacts associated with the landfill component of the 2001 GDP, as identified in EIR 575 (numerical designations are from EIR 575). All mitigation commitments contained within FEIR 575 and the 2001 GDP will apply to the Proposed Project.

Impact 4.2-1: Hillside excavations for construction of landfill refuse capacity and daily cover soil could expose people or property to landslide or mudslide activity during the excavation period.

- MM 4.2-1a: Prior to designing each phased landfill plan and specifications, the IWMD shall conduct a geotechnical investigation to determine the extent of landslide material and the soil foundation characteristics of the proposed phase. A geotechnical report of the phased site area shall be prepared which includes a landslide excavation and removal plan prepared to the satisfaction of the Director, IWMD.
- MM 4.2.1b: For each phased grading plan, the excavation and grading plan shall ensure the stability of all cut, fill and lined slopes. Slopes shall be designed to withstand the most probable earthquake based on a return period of 100 years or as required by current regulations. Liner design plans shall be submitted to the San Diego Regional Water Quality Control Board for approval. The plans shall also be incorporated in an JTD and submitted to the LEA for approval and to the CIWMB for concurrence.

Impact 4.2-2: Seismic activity occurring along any of the active regional faults could subject the landfill property to seismic shaking which could damage landfill facilities and/or structures.

MM 4.2-2a: The IWMD shall demonstrate that landfill design plans comply with the state and federal seismic requirements in CCR Title 27, and 40 Code of Federal Regulations (CFR) §258.14 (Seismic Impact Zones) and §258.15 (Unstable Areas). These demonstrations shall be incorporated in the IWMD Operating Record prior to construction of said plans.

- MM 4.2-2b: Prior to commencement of daily excavations for borrow material, grading plans shall be prepared, analyzed for slope stability and submitted for approval by the Director, IWMD, or his designee.
- MM 4.2-2c: As part of a JTD, the IWMD shall present the assumptions, methods and calculations used to demonstrate seismic safety. This measure is required only if final slopes are planned to be steeper than a ratio of 3:1 (horizontal to vertical), if the site is located in an area subject to liquefaction or in unstable areas with poor foundation conditions as described in the Seismic Safety Element of the Orange County General Plan (27 CCR 17777).

Impact 4.2-3: Differential settlement associated with compression and decomposition of solid waste materials can be expected on the order of 30 percent of the total refuse thickness.

MM 4.2.3: As part of a JTD, the IWMD shall present the assumptions, methods and calculations used to demonstrate that differential settlement of the site will not result in future environmental impacts (27 CCR 21090).

Impact 4.2-4: Continued use of the site for landfill purposes will create the demand for soil to be used as cover material.

MM 4.2-4: When the JTD is prepared, the IWMD shall identify the assumptions, methods and calculations performed to demonstrate that the excavation plans provide for sufficient quantities and sources of suitable soils or alternative cover systems for daily and intermediate cover, final cover and liner materials. This section of the JTD should also reference and summarize any borrow studies conducted to demonstrate the availability of sufficient quantities of materials. If materials are obtained on site, the description shall include which sections of the site will be excavated for each sequence of landfilling and where these materials will be stockpiled for use. Stockpile locations should not interfere with unloading, spreading, compacting, access, safety, drainage or other operations on the site. Stockpiles should be clearly shown on the fill sequencing and excavation plans prepared for construction. (27 CCR 21600).

Mitigation For Impacts Associated with Amendment No. 2 to the 2001 GDP

No additional mitigation measures are required.

5.2.5 LEVEL OF SIGNIFICANCE AFTER MITIGATION

The Proposed Project has been determined to have no substantial change from previous analyses on geophysical resources.

5.3 HYDROLOGY AND WATER QUALITY

This section provides a summary of relevant information from FEIR 575 and is updated with more recent information that was provided by supplemental studies and analyses.

5.3.1 EXISTING CONDITIONS

The PDL site is located within the 2 major watersheds of Prima Deshecha Cañada and Segunda Deshecha Cañada, as shown on Exhibit 5.3-1.



Pre-Development Site Hydrology

Prima Deshecha Landfill Supplemental Environmental Impact Report 597



(3.3-1 072505.ai) D:\Projec



The following presents a summary of the hydrologic information contained within the 2001 GDP and EIR 575, *Prima Deshecha Landfill Geotechnical Investigations Report – Zone 4* (GeoLogic Associates 2002), and an *Analysis of Groundwater Resources in Zone 4, Prima Deshecha Landfill, Orange County, California* (GeoLogic Associates 2004, located in Appendix E).

Prima Deshecha Cañada

The Prima Deshecha Cañada watershed covers approximately 1,298.5 acres, or roughly 84 percent of the project site. As shown on Exhibit 5.3-1, the Prima Deshecha Cañada channel is the major hydrologic feature on the site and extends from the northeastern corner to the southwestern corner of the property. Several small tributary streams extend into the canyon and flow into the main southwesterly trending channel. The surface flows exit the site at the southwestern corner and enter the upstream inlet of the M01-reinforced concrete box (RCB) storm drain in the City of San Clemente. This RCB parallels Camino de los Mares to the Shorecliffs Golf Course, where the flows are discharged into an RCB that traverses the golf course. At the southern end of the golf course, the flows enter a double RCB that crosses Avenida Vaquero and Pacific Coast Highway where it is discharged into the Pacific Ocean.

As it qualifies as a "water of the United States," the Prima Deshecha Cañada watercourse is under the jurisdiction of Section 404 of the Clean Water Act, administered by the USACE (see Section 4.3.3 for more detailed information on the 404 permit program). California Department of Fish and Game (CDFG) jurisdictional areas are also present on the site pursuant to Division 2, Chapter 6, Sections 1600–1603 of the California *Fish and Game Code* (see also Section 4.3.3). A Jurisdictional Delineation for the site was completed in 2003 and can be found in Appendix F. The jurisdictional limits of the CDFG and USACE referenced in this SEIR are the result of this detailed field delineation.

Water Sources for Prima Deshecha Cañada Stream

Flows within Prima Deshecha Cañada stream occur predominantly in a southwesterly direction. A perennial spring is located in the northernmost reaches of Prima Deshecha Cañada stream close to the center of Zone 4 (Exhibit 5.3-1), within an outcrop of the San Onofre Breccia near the head of the canyon. Currently, the primary sources of water for the stream are believed to be precipitation runoff and spring flows. Average rainfall precipitation at the PDL is approximately 10.2 inches per year (GLA 2004), 90 percent of which typically falls between December and March. Between March and December, the only source of surface water to the stream is believed to be spring flow, estimated at approximately two to five gallons per minute (GLA 2004). Surface flow continues down the channel to a point where it eventually submerges into the alluvium of the streambed.

As referenced in EIR 575, the source of the groundwater emanating from the spring may be an artesian aquifer in the formation. It is also possible the groundwater may be derived from underlying geologic units such as the Topanga Formation, with groundwater moving upwards (towards the surface) along fractures in the San Onofre Breccia and/or along the fault which is in close proximity to the spring. In addition, some portion of the spring flow may be derived from downward percolation of surface water in a structural depression (probably a syncline) east of the spring. More recent field observations and analytical geological investigations indicate that the source of spring flows is groundwater stored in the Monterey Formation and uppermost San Onofre Breccia (GLA 2002) (Exhibit 5.3-2). It is likely that this spring daylights in this area due to a splay of the Forster Fault, which juxtaposes the low permeability claystone of the Capistrano Formation against the higher permeability breccia and conglomerate of the San Onofre Breccia. Without a detailed study of on-site groundwater resources, there is some uncertainty as to specific contribution made by each of the geologic units to overall spring flow rates. However, owing to its geologic characteristics relative to the spring, the Monterey Formation is assumed

to provide the major component of spring flows into the Prima Deshecha Cañada stream channel. Accordingly, the easterly portion of Zone 4 would likely function as the recharge area for the Monterey Sandstone aquifer that feeds the spring.

Segunda Deshecha Cañada

The Segunda Deshecha Cañada watershed (referred to as "Segunda Deshecha" on site) covers approximately 230 acres or roughly 15 percent of the site. It drains to the southeastern corner of the site and continues approximately 1.5 miles as a natural channel to Avenida Pico (Exhibit 5.3-1). Upon reaching Avenida Pico, flows from the Segunda Deshecha Cañada stream channel enter an RCB for about 3,400 feet. Flows then continue in a natural channel until entering another RCB approximately 2,700 feet north of the Interstate 5 freeway (I-5). That RCB opens into a concrete-lined trapezoidal channel for a short distance before it transitions to an RCB that flows under the I-5 and a shopping center to the south. Flows continue south through a concrete-lined trapezoidal channel with culverts at Calle de los Molinos, El Camino Real, and Avenida Pico. The channel becomes a double RCB as it crosses under Pacific Coast Highway, a restaurant, and a parking lot before it opens into a concrete-lined tracks and discharges into the ocean at the north beach parking lot near Avenida Pico and Pacific Coast Highway.

The portion of the PDL site within the Segunda Deshecha Cañada watershed is incised by four small channels. The 100-year storm flows (Q_{100}) for these 4 streams range from 110 cubic feet per second (cfs) to 214 cfs prior to their confluence south of the site where the Q_{100} has been calculated to be 903 cfs (EIR 575). The Segunda Deshecha Cañada channel continues in a southerly direction beyond the confluence of the four tributary channels.

A small area along the northern and western periphery of the site, covering approximately 17 acres or about 1 percent of the site, drains north through natural channels to San Juan Creek. This peripheral area is characterized by minor surface flows which drain to the north and west after leaving the site. The 100-year storm flows from this entire portion would total approximately 200 cfs. This area of the site contributes only a very small amount of the surface flow within that adjacent drainage area due to its location at the extreme outer reaches of the watershed.

Groundwater

The PDL site lies in the San Clemente Subunit of the San Juan Groundwater Basin. EIR 575 classifies regional groundwater resources as follows:

- 1. The <u>shallow aquifer</u> (of depth ranging from 10 to 20 feet) exists within the erosional channels (alluvium) incised into the Capistrano siltstone formation. Surface water runoff and mineral springs in the area intermittently recharge the shallow aquifer. It has been documented that two cattle-water wells in the shallow aquifer were destroyed because of the low yield and the high ion concentration.
- 2. The <u>deep aquifer</u> (of known beneficial uses in the region) is primarily found in the San Onofre Breccia formation at a depth of approximately 1,000 feet.

Regional Groundwater Resources

Groundwater from alluvial deposits in the San Clemente subunit is only known to have been produced from two shallow wells in Segunda Deshecha. These wells, previously used to water cattle by the Rancho Mission Viejo Company, are no longer in existence. The direction of groundwater movement in Segunda Deshecha, based on historic water levels in these two



wells, is towards the Pacific Ocean to the south. Groundwater resources in this area are not anticipated to be affected by the Proposed Project, as there will be no disruption to Segunda Deshecha. The only uses proposed for this area are passive recreation, open space, and native habitat. Based on these hydrologic conditions, small channels within Segunda Deshecha were identified for potential riparian restoration in the proposed Pre-mitigation Plan. Also, some portions of the upland areas within Segunda Deshecha were identified for coastal sage scrub and southern needlegrass grassland restoration in fulfillment of the mitigation requirements for the ultimate buildout of Prima Deshecha Landfill.

The closest wells to the site with groundwater from alluvial deposits are located in the San Juan subunit, about two miles north of the site (EIR 575). These wells, which are hydrologically upgradient of the site, are in San Juan Creek which drains in a westerly direction into Oso Creek. There are no known wells which produce groundwater from alluvial aquifers downgradient of the site (EIR 575).

Relative to the geologic formations comprising the project site, the Capistrano Formation is believed to have a low potential for groundwater. The principal usable reserves of groundwater in the area surrounding the site occur at a depth of approximately 400 to 900 feet below ground surface (bgs) in the Capistrano Formation. Water is pumped from this depth at a small well field in San Clemente (EIR 575). Groundwater in this unit is produced from poorly consolidated sandstones and conglomeratic sandstone lenses. The groundwater flow direction is generally seaward, towards the west. Groundwater in these aquifers is believed to exist under confined conditions (Exhibit 5.3-3)

Regionally, the Monterey Formation is considered to have minor potential for groundwater due to the abundance of low permeability siltstone, shale, and mudstone and the fact that no economic groundwater resources are known to occur in this unit in the vicinity of the site (EIR 575).

The San Onofre Breccia may contain economic quantities of groundwater based on the exposed lithologies in this formation and the presence of the spring emanating from surface outcrops at the toe of an approximately 200 foot high bluff/cliff area in the eastern central half of the site (Exhibit 5.3-1).

On-Site Groundwater Occurrences

EIR 575 has identified groundwater resources at the site in 4 discrete environments occurring:

- Unconfined within alluvial sediments;
- Perched above low-permeability concretions or clay beds;
- Unconfined or confined within bedrock joints or fractures; and
- Unconfined or confined within sandstone beds.

Groundwater in the first environment (i.e., unconfined within alluvial sediments) occurs in alluvial sediments along Prima Deshecha Cañada stream and its tributaries and eventually infiltrates the underlying bedrock along fractures and joints (thereby contributing to remaining environments identified above). Alluvial groundwater is monitored by six monitoring wells located along Prima Deshecha Cañada stream.

The remaining groundwater environments (i.e., the second, third, and fourth bullets identified above) occur in the landslide deposits and bedrock units throughout the site, and are monitored by monitoring wells MW-1 through MW-11, S-1, and the spring (Exhibit 5.3-3). Hydraulic conductivities are variable in the bedrock environments, and based on well development and sampling observations, it appears that the hydraulic conductivity is highest within the Monterey

Formation sandstone and lowest within the Capistrano Formation claystone. Previously measured groundwater elevations indicate that groundwater flows from the northeast to the southwest at hydraulic gradients ranging from approximately 0.12 feet (horizontally) per foot (topographically) in the northeastern portion of the site, to about 0.03 feet per foot near the southwestern portion of the site (GeoSyntec 1999). The Forster fault zone transects the eastern half of the site, and these faults may act as impedances to groundwater flow, accounting for the steeper gradient within the northeastern portion of the site.

Beneath Zone 4, groundwater was encountered in 31 of 37 boreholes at depths ranging from 18 to 336.5 feet bgs (GLA 2002). Eight groundwater monitoring wells were installed as part of a geotechnical investigation that was commissioned in 1996, the findings of which were published in January 2002 (GLA 2002). Wells were screened again in each of the 3 geologic formations exposed in Zone 4 (the Capistrano formation, the Monterey formation, and the San Onofre Breccia). Each well was constructed to evaluate uppermost groundwater conditions in the bedrock adjacent to the proposed Zone 4 development area. Comparison of the firstencountered groundwater elevation to the static water elevation suggests that groundwater in the bedrock exists under semi-confined to confined conditions. There may be areas that would require dewatering during construction prior to the development of the Zone 4 liner system (i.e., within the landslide remediation areas), even though the majority of the proposed refuse fill area will be separated from the uppermost groundwater by a minimum of 5 feet. Accordingly, a dendritic subdrain system was incorporated into the composite liner system to prevent buildup of hydrostatic stress beneath the liner and to maintain a five foot separation between refuse deposits and groundwater. Associated discharges from this system will be evaluated prior to the development of Zone 4 in order to determine the permit requirements in effect at that time for non-stormwater discharges and to check for compliance with amended waste discharge requirements for the site.

Groundwater Quality

The results of water quality analyses indicate that groundwater in each formation contains elevated concentrations of numerous inorganic constituents. Table 5.3-1 presents water quality data collected between 1994 and 1997 at the spring located within Zone 4. Comparison of sampling results with the state and federal Applicable or Relevant and Appropriate Requirements (ARARs) and the Basin Objectives that were established by the San Diego Regional Water Quality Control Board (SDRWQCB) indicate that background water quality is very poor and exceeds the various concentration limits for chloride, sulfate, specific conductance, total dissolved solids, boron, and several other metals. Furthermore, a groundwater chemistry assessment of samples collected from wells screened in bedrock formations indicates that bedrock groundwater is of poor quality and is characterized by dissolved solids, chloride, and sulfate concentrations that exceed the State of California drinking water standards. However, it should be noted that the water quality characteristics of the spring feeding the Prima Deshecha Cañada channel have not adversely affected riparian resources in the downstream corridor.

Comparison of the on-site studies with the regional studies suggests that the regionally defined "shallow aquifer" corresponds with the on-site observations of alluvial groundwater. However, the regionally-defined "deep aquifer" has not been intercepted by on-site studies.



Source: GeoLogic Associates, 1999

Exhibit 5.3-3

Groundwater Monitoring Wells and Gradients

Prima Deshecha Landfill Supplemental Environmental Impact Report 597



 $\label{eq:resonance} R:\Projects\OCIWMD\J004\SEIR597\Public\Review\Draft\graphics\Ex5.3-3_groundwtrmonitwellsgrad_080205.pdf$

TABLE 5.3-1 HISTORICAL SUMMARY DATA – GROUNDWATER SPRING **GEOTECHNICAL INVESTIGATION**

Analyte	Units	Mar 1991	May 1991	Feb 1992	Jun 1992	Mar 1993	May 1993	Aug 1993	Nov 1993	Feb 1994	May 1994	Oct 1994	Apr 1996	Jul 1996	Apr 1997	Jul 1997	MED.	AVG.	STD. DEV.	MIN.	MAX.
GENERAL CHEMISTRY													·								
Alkalinity	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	273	245	250	267	259	259	12	245	273
Bicarbonate Alkalinity	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	273	245	250	267	259	259	12	245	273
Carbonate	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	0.7	0.7	0.7	0.7	NC	NC	NC	NC	NC
Chloride	mg/L	1,257	1,212	1,165	1,167	795	NA	768	800	771	787	770	944	900	888	858	798	949	218	768	1,257
Hardness	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	1,883	1,782	1,859	1,817	1,838	1,835	45	1,782	1,883
lon Balance	%	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	0.008	0.87	2.1	1.52	1.20	1.12	0.90	0.008	2.1
Nitrate (as N)	mg/L	2.2	2.2	2.2	2.2	7	NA	3.1	1.8	1.3	0.6	1.7	0.3	0.4	0.4	0.44	10	1.70	2.1	0.3	7
Nitrogen, Total	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
рН	units	7.70	7.30	8.00	7.20	8.20	8.37	8.25	8.18	8.25	7.99	8.31	7.86	7.92	8.04	7.92	8.00	7.97	0.35	7.2	8.37
Phosphate, Total	mg/L	NA	NA	NA	NA	0.1	NA	NC	NC	NC	NC	NC									
Specific Conductance	µmhos/cm	NA	NA	NA	NA	3,630	NA	NA	NA	NA	NA	NA	4,340	4,340	4,570	4,210	4,210	4,018	386	4,570	4,340
Sulfate	mg/L	1,491	1,552	1,519	1,496	1,190	NA	847	860	923	980	888	1,051	1,003	1,038	982	1,021	1,130	267	847	1,552
Total Dissolved Solids	mg/L	4,422	4,574	4,746	4,804	3,100	2,700	3,040	3,350	2,740	2,980	3,250	3,550	3,124	3,520	3,382	3,350	3,552	723	2,700	4,804
Turbidity	NTU	0.94	23	49	31	1.1	4	1.6	0.65	3.1	1.7	73.5	6.7	3.5	NA	NA	3.5	15.4	22.9	0.65	73.5
METALS	•		1	1			•	-	-	•	1			-		•		1			
Aluminum	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Antimony	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Arsenic	mg/L	0.006	0.004	0.005	0.005	0.01	NA	0.014	0.01	0.01	0.01	0.01	NA	NA	NA	NA	0.005	0.005	0.001	0.004	0.006
Barium	mg/L	0.05	0.07	0.07	0.06	0.06	NA	0.02	0.06	0.05	0.05	0.04	NA	NA	NA	NA	0.06	0.06	0.01	0.04	0.08
Beryllium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Cadmium	mg/L	0.01	0.01	0.01	0.01	0.01	NA	0.01	0.01	0.01	0.01	0.01	NA	NA	NA	NA	NC	NC	NC	NC	NC
Calcium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	412	388	409	407	408	404	11	388	412
Chromium	mg/L	0.01	0.01	0.01	0.01	0.02	NA	0.02	0.02	0.02	0.02	0.02	NA	NA	NA	NA	NC	NC	NC	NC	NC
Cobalt	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Copper	mg/L	0.01	0.01	0.02	0.01	0.02	NA	0.08	0.02	0.02	0.02	0.02	NA	NA	NA	NA	0.02	0.04	0.04	0.01	0.08
Iron	mg/L	0.19	1.54	0.07	0.13	0.1	NA	0.1	0.1	0.1	0.1	0.0	NA	NA	NA	NA	0.16	0.48	0.71	0.07	1.54
Lead	mg/L	0.01	0.01	0.01	0.01	0.02	NA	0.02	0.02	0.02	0.02	0.02	NA	NA	NA	NA	NC	NC	NC	NC	NC
Magnesium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	208	198	204	195	201	201	6	195	208
Mercury	mg/L	0.001	0.001	0.001	0.001	0.002	NA	0.002	0.002	0.002	0.002	0.002	NA	NA	NA	NA	NC	NC	NC	NC	NC
Molybdenum	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Nickel	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Potassium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	7.82	8.14	8.31	9.39	8.23	8.42	0.68	7.82	9.39
Selenium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Silver	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Sodium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	386	332	379	344	361.5	360.3	26.3	332	386
Thallium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Tin	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Vanadium	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
Zinc	mg/L	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NC	NC	NC	NC	NC
VOLATILE AND SEMI-VOLA	ATILE ORGAN	OLATILE AND SEMI-VOLATILE ORGANIC COMPOUNDS – None Detected																			

Notes:

Compound or analyte not detected; value reported is the laboratory method detection limit (MDL). Compound or analyte is measured at a trace concentration between the laboratory MDL and the practical quantitation limit (PQL)

Quantified concentration or value. Constituent or analyte not analyzed during that monitoring.

Source: GLA 2002

5.3.2 SIGNIFICANCE CRITERIA

The CEQA Guidelines indicate that a project will normally have a significant effect on the environment related to hydrology and groundwater if it will "...substantially degrade or deplete ground water resources...interfere substantially with ground water recharge...[or] cause substantial flooding, erosion, or siltation..."

For the purposes of this SEIR, the GDP was determined to have a significant effect on the environment related to hydrology and groundwater if a GDP impact met the language of the CEQA Guidelines or exceeded the thresholds set by the Environmental Checklist issued by the County of Orange.

5.3.3 POTENTIAL IMPACTS

Per the County of Orange Environmental Checklist, the Proposed Project would have a significant effect on hydrology and water quality if it would:

(1) Violate any water quality standards or waste discharge requirements.

The Proposed Project will not create either runoff or water discharge that will exceed water quality standards or violate existing waste discharge requirements (WDRs). Operations proposed in Amendment No. 2 to the 2001 GDP will be consistent with the operations currently covered under the site's WDRs and its General Permit to Discharge Stormwater Associated With Industrial Activity (including stormwater collection, leachate collection, and disposal and groundwater subdrain flow collection and disposal). WDRs at the PDL were updated in 2004 and 2005 for ongoing landfill operations, and all applicable measures and BMPs will be in place before and during project construction so that the site will comply with water quality standards during the construction of project features.

In addition to ongoing compliance with National Pollutant Discharge Elimination System (NPDES) requirements for industrial discharges, the proposed project will comply with Section 7 of Orange County's Drainage Area Management Plan (DAMP), which requires a Water Quality Management Plan. Compliance with the DAMP is a project mitigation requirement, and the IWMD will coordinate compliance with the County of Orange Watershed Coastal Resources Division accordingly.

Any flows from the proposed subsurface reservoir that are released into Prima Deshecha Cañada stream channel are not expected to leave the property boundary but are intended to percolate into the alluvium within the length of the riparian corridor. However, should any flows actually cross the property boundary, they will discharge into improved stormwater channels that flow into the ocean and will therefore not impact any downstream water bodies. Permitting requirements for the subsurface reservoir will be addressed closer to the development of Zone 4 (prior to facility construction) in order to ensure that water quality standards and waste discharge requirements are met.

(2) Substantially deplete groundwater supplies or interfere substantially with groundwater recharge such that there would be a net deficit in aquifer volume or a lowering of the local groundwater table level (e.g., the production rate of preexisting nearby wells would drop to a level which would not support existing land uses or planned uses for which permits have been granted).

Full project buildout within Zone 4 will significantly affect the spring providing flows into the Prima Deshecha Cañada stream. In addition, the area believed to provide a recharge function for this spring will be largely impacted by landfill operations within Phases D, E and G, which are

currently estimated to begin in the year 2045. At that time, the spring head waters would be removed by landfill grading operations, and the recharge area would be reduced by the construction of the liner system that is required for landfill operations.

This recharge area has not been shown to be significant relative to regional groundwater resources; however, on-site groundwater occurrences appear to be related to quantities of flow through the Monterey Sandstone, which is exposed throughout the eastern portion of the site within Zone 4. Consequently, impact to on-site groundwater resources may occur as this recharge area is reduced.

Most importantly, the biological resources occupying the riparian zone along the Prima Deshecha Cañada stream are largely dependent upon maintenance of these flows within the stream channel. A series of alternatives have been developed for augmenting surface water flows, as needed, to offset potential project impact to stream flow. The Proposed Project analyzed in SEIR 597 considers a potentially promising concept involving modification to Zone 4's desilting system through the incorporation of a subsurface rainfall storage reservoir(s). This (these) reservoir(s) would serve to offset the impacts of removal of the spring and its recharge area on local groundwater resources by storing stormwater collected in planned desilting basins, and slowly discharging these flows downstream through an outlet pipe. The reservoirs would be located adjacent to or beneath one or more of the future permanent desilting basins proposed for Zone 4, and would consist of a series of large underground pipes (Exhibit 4.3-5). These pipes will be used for storage of desilted water that can be collected from stormflows routed from Zone 4. Rainfall runoff would be stored in water-tight pipes (i.e., large high density polyethelene [HDPE], gasketed reinforced concrete pipes, or sealed corrugated alvanized steel pipes). As water is collected in the western basin(s), a portion of the desilted water will be directed into the underground storage system where it will be released via gravity flow into the downstream channel. If the underground storage system is full, remaining storm water will be released directly into the channel downstream of the western desilting basin. Pending engineering feasibility studies and operational and permit requirements, this system is proposed to mitigate potential negative local effects from groundwater recharge reduction. Should this option prove infeasible, other surface water augmentation options would be pursued.

(3) Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river, in a manner, which would result in substantial erosion or siltation on site or off site.

As indicated above, landfill operations within Zone 4 will eventually remove the Prima Deshecha Cañada stream channel and its associated spring. This impact constitutes a substantial alteration of the existing drainage pattern of the area in the short term, as identified in EIR 575. Exhibit 5.3-4 presents post-project hydrologic conditions at the site. The construction of a subsurface rainfall storage system (or other surface water augmentation measures) will serve to mimic the natural drainage pattern of the site by maintaining flows within the remaining downstream portions of the channel for sustenance of the biological resources within the riparian corridor. Proposed Project Zone 4 desilting system revisions will not substantially alter the post-project hydrologic conditions at the site, as analyzed in the EIR 575. The revised desilting system primarily functions to better utilize desilted water from the site and minimizes direct biological impacts from desilting basin construction.

Per existing commitments in EIR 575, sediment- and erosion-control plans will be prepared and implemented on an annual basis to reduce sediment and to control erosion on the landfill site. Silt generated on the active landfill surface will be reduced further with the desilting facilities required by 27 CCR. These facilities include permanent and interim desilting basins that are intended to prevent substantial on- or off-site erosion or siltation.

(4) Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river, or substantially increase the rate or amount of surface runoff in a manner, which would result in flooding on site or off site.

Although there will be permanent alteration of the existing drainage pattern on the site (Exhibit 5.3-4), there will be no increase in the rate or amount of surface water that is generated by any element of the Proposed Project, and there will be no induced flooding on site or off site from the Proposed Project over that which was analyzed in EIR 575.

(5) Create or contribute runoff water which would exceed the capacity of existing or planned storm water drainage systems or provide substantial additional sources of polluted runoff.

Project drainage features, as incorporated into EIR 575, will be designed to regulate 100-year flows off the project site. The Proposed Project will have no negative incremental effect on these flows.

(6) Have a significant adverse impact on groundwater quality or otherwise substantially degrade water quality.

Groundwater quality will not be affected by construction of the desilting basins, the rainfall storage reservoir, or the landslide remediation features included in the Proposed Project. Adequate and appropriate desilting features will accompany project construction and operation.

(7) Place housing within a 100-year flood hazard area as mapped on a federal Flood Hazard Boundary or Flood Insurance Rate Map or other flood hazard delineation map.

The Proposed Project will not increase flood flows off the project site or impact mapped flood hazard areas. No housing development is associated with the Proposed Project.

(8) Place within a 100-year flood hazard area structures, which would impede or redirect flood flows.

The Proposed Project will not place any structures within a 100-year floodplain that would impede or redirect flows.

(9) Expose people or structures to a significant risk of loss, injury or death involving flooding, including flooding as a result of the failure of a levee or dam.

The Proposed Project will not alter the risk of either people or structures to loss, injury or death associated with flooding or damage to flood-control facilities.

(10) Inundation by seiche, tsunami, or mudflow.

Seiches and tsunamis are not potential issues at the project site due to the lack of proximity to a major body of water. The Proposed Project will not incrementally affect the occurrence or probability of mudflows on the site.





Prima Deshecha Landfill Supplemental Environmental Impact Report 597



D:\Projec





5.3.4 MITIGATION MEASURES

Previously Adopted Mitigation

The following mitigation measures are currently in place for impacts associated with the landfill component of the 2001 GDP, as identified in EIR 575 (numerical designations are from EIR 575). All mitigation commitments contained within EIR 575 and the 2001 GDP will apply to the Proposed Project.

Impact 4.2-5: The potential exists for landfill leachate migration into groundwater through fractured and porous alluvium.

- MM 4.2-5a: The IWMD shall continue to operate its existing leachate control system within the active landfill area. In addition, the IWMD shall be required to construct a corresponding leachate control and recovery system in those areas where new liners are constructed and in areas added to the active landfill area.
- MM 4.2-5b: The site shall continue to operate under the groundwater monitoring requirements contained in Waste Discharge Requirements, Order No. 89-102, Technical Change Order (TCO) No. 1, Amended Waste Discharge Requirements contained in Order No. 93-86, and any future orders issued by the San Diego Regional Water Quality Control Board. TCO No. 1 contains the detailed Groundwater and Vadose Zone Monitoring Program for the Prima Deshecha Landfill.
- MM 4.2-5c: As part of a revised JTD, the IWMD shall present the assumptions, methods and calculations used to predict leachate generation and sizing of the components of the leachate collection system.

Impact 4.3-1: Modification of the landfill surface will alter the surface hydrology of the site and could result in increases in stormwater runoff.

- MM 4.3-1a: As part of a Joint Technical Document (JTD) to be prepared by IWMD, the IWMD shall present the assumptions, methods and calculations used to calculate the potential flow quantities for run-on, runoff, and sediment content of storm water flow used in sizing drainage and sediment control facilities.
- MM 4.3-1b: As part of a JTD to be prepared by IWMD, the IWMD shall include surface drainage plans for final fill and bottom excavation plans, including any berms, down drain systems, storm drain systems, direction of flow in perimeter drainage channels, and the location of off-site discharge point for runoff water.
- MM 4.3-1c: Detention, diversion, and drainage facilities shall be designed and constructed to accommodate the anticipated volume of precipitation and peak flows from surface runoff under the precipitation conditions specified in §20365 of Title 27 of the California Code of Regulations for each class of waste management unit (WMU). In addition, drainage facilities for WMUs shall be designed to prevent washout of the WMUs during a 100-year storm event.

Impact 4.4-1: The potential exists for surface water quality degradation from landfilling.

MM 4.4-1a: The IWMD shall comply with its National Pollutant Discharge Elimination System (NPDES) Storm Water Pollution Prevention Plan (SWPPP) and its NPDES Monitoring and Reporting Plan for the landfilling under the GDP. This plan will

ensure that the measures taken to safeguard surface water quality are effective and are being correctly employed.

MM 4.4-1b: The IWMD shall continue to implement the existing Surface Water Runoff Monitoring Program as described in the currently effective Waste Discharge Requirements.

Impact 4.4-2: Potential erosion associated with construction of the landfill-related projects could increase the silt load in surface waters.

MM 4.4-2: As part of the NPDES program and prior to approval of construction contracts, the Director, IWMD, or a designee, shall ensure that silt loading to surface waters from the construction activities will be periodically tested and controlled, where necessary, by appropriate erosion control measures, siltation basins or other settling structures.

Mitigation for Impacts Associated with Amendment No. 2 to the 2001 GDP

Although several project impacts have been identified in the above analyses relative to permanent impacts on groundwater recharge and drainage patterns within Zone 4, the Proposed Project has been designed to offset these impacts through the creation of a rainfall collection system and subsurface reservoir to augment surface water flows (or other water augmentation measures deemed to be feasible and approved by the regulatory agencies). These design features are intended to mimic natural hydrologic conditions and serve to maintain biological resources within Prima Deshecha Cañada stream channel. Consequently, with these project elements in place in addition to the mitigation measures identified above for the 2001 GDP, there will be no effect to on- or off-site resources. However, compliance with current requirements set forth in the DAMP will be required.

MM 5.3-1 The Proposed Project will comply with Section 7 of the Drainage Area Management Plan (DAMP) for Orange County through the development of a Water Quality Management Plan.

5.3.5 LEVEL OF SIGNIFICANCE AFTER MITIGATION

With the incorporation of the mitigation measures above and implementation of the Proposed Project as presented, there will be no significant impacts to Hydrology and Water Quality from the Proposed Project.

5.4 <u>AIR QUALITY</u>

This section provides a summary of the information contained in Section 4.9 of FEIR 575, supplemented with updated information on existing air quality conditions and revised air quality rules and regulations that affect operations at the PDL site. A copy of FEIR 575 is available at the IWMD offices in Santa Ana. Please refer to Section 1.1.6 for the location of the IWMD offices.

The Proposed Project would not require additional equipment, additional soil disturbance, or additional vehicle miles traveled (VMT) over and above the project as it was analyzed in FEIR 575 due to the fact that landslide remediation and related activities are ongoing events at the landfill that were part of the analyzed project. Although the area of disturbance that was analyzed in FEIR 575 would expand as a result of the Proposed Project, the following issues that are of concern in air quality analyses would not change and, therefore, air quality emissions would not change:

- The types and quantities of equipment used for daily operations and landslide remediation activities at the project site would be consistent with the types and quantities of equipment that are used for ongoing landfill operations covered under FEIR 575. No additional equipment is anticipated over that used for landslide remediation activities covered under FEIR 575.
- The quantity of soil that would be disturbed on a daily basis during landslide remediation activities for the Proposed Project is not anticipated to be greater than the quantity of soil excavated for previous landslide-remediation activities covered under FEIR 575. Even if activity occurs in landslide-remediation areas and active portions of the landfill disposal area simultaneously, total quantities would not exceed those of the project analyzed in FEIR 575.
- Daily VMT would not increase as a result of the Proposed Project. The number of vehicle trips associated with landfill operations and landslide-remediation activities are not anticipated to be greater than VMTs for previous landslide-remediation activities and landfill operations covered under FEIR 575. No additional trips are anticipated due to implementation of the proposed project.

5.4.1 EXISTING CONDITIONS

Air Quality Management Framework

The Prima Deshecha Landfill is located in the South Coast Air Basin (SCAB) which is within the jurisdiction of the South Coast Air Quality Management District (SCAQMD). The SCAQMD prepares the Air Quality Management Plan (AQMP) and adopts and enforces rules and regulations for stationary sources in the Basin. The 2003 AQMP is the SCAB's current operative air plan.

Monitored Air Pollutants in the SCAB

State and federal ambient air quality standards have been established for carbon monoxide (CO), ozone (O_2), nitrogen dioxide (NO_2), sulfur dioxide (SO_2), particulate matter (PM10 and PM2.5), lead (Pb), and sulfate(SO_4), as illustrated in Table 5.4-1.

Carbon Monoxide

CO is a colorless, odorless gas that results from the incomplete combustion of fossil fuels. Over 80 percent of the CO emitted in urban areas is contributed by motor vehicles. High levels of CO commonly occur near major roadways and freeways. CO interferes with the blood's ability to carry oxygen to the body's tissues and results in numerous adverse health effects. CO is a criteria air pollutant. The federal and state standards for CO are presented in Table 5.4-1, State and Federal Ambient Air Quality Standards.

Nitrogen Dioxide

Nitrogen dioxide (NO₂) is a criteria air pollutant and may result in numerous adverse health effects. It absorbs blue light, resulting in a brownish-red cast to the atmosphere and reduced visibility.

Nitrogen Oxides

"Nitrogen oxides" is a general term pertaining to compounds of nitric acid (NOx), NO₂, and other oxides of nitrogen. Nitrogen oxides are typically created during combustion processes and are major contributors to smog formation and acid deposition.

<u>Ozone</u>

Ozone is a strong-smelling, pale blue, reactive toxic chemical gas that consists of three oxygen atoms. Ozone is a product of the photochemical process that involves the sun's energy. Ozone exists in the upper atmosphere ozone layer and at the earth's surface. Ozone at the earth's surface causes numerous adverse health effects and is a criteria air pollutant. It is a major component of smog.

Ozone is a secondary pollutant; it is not directly emitted. Ozone is the result of chemical reactions between other pollutants, most importantly hydrocarbons and NO₂, which occur only in the presence of bright sunlight. Pollutants emitted from upwind cities react during transport downwind to produce the oxidant concentrations that are experienced in the area. Many areas of the SCAQMD contribute to the ozone levels experienced at the monitoring stations, with the more significant areas being those directly upwind.

		National Standard					
Air Pollutant	State Standard	Primary	Secondary				
Ozone (O ₃)	0.09 ppm, 1-hr avg. 0.07 ppm, 8-hr avg.	0.08 ppm, 8-hr avg.	0.08 ppm, 8-hr avg.				
Respirable Particulate Matter (PM10)	50 μg/m³, 24-hr avg. 20 μg/m³ AGM	150 μg/m ³ , 24-hr avg. 50 μg/m ³ AAM	150 μg/m ³ , 24-hr avg. 50 μg/m ³ AAM				
Fine Particulate Matter (PM2.5)	No 24-hr., State std. 12 μg/m³ AGM	65 μg/m ³ , 24-hr avg. 15 μg/m ³ AAM	65 μg/m ³ , 24-hr avg. 15 μg/m ³ AAM				
Carbon Monoxide (CO)	9.0 ppm, 8-hr avg. 20 ppm, 1-hr avg.	9 ppm, 8-hr avg. 35 ppm, 1-hr avg.	None				
Nitrogen Dioxide (NO ₂)	0.25 ppm, 1-hr avg.	0.053 ppm, annual avg.	0.053 ppm, annual avg.				
Sulfur Dioxide (SO ₂)	0.25 ppm, 1-hr 0.04 ppm, 24-hr avg.	0.03 ppm, annual avg. 0.14 ppm, 24-hr avg.	0.5 ppm, 3-hr avg.				
Lead (Pb)	1.5 µg/m ³ , monthly avg.	1.5 µg/m ³ , calendar quarter	1.5 <i>µ</i> g/m ³				
Visibility-Reducing Particles	Extinction coefficient of 0.23 per km, visibility of 10 miles at relative humidity less than 70%, 1 observation	_	_				
Sulfates (SO ₄)	25 µg/m³, 24-hr avg.	—	—				
Hydrogen Sulfide (H ₂ S)	0.03 ppm, 1-hr avg.		—				
Vinyl Chloride	0.01 ppm, 24-hr avg.	_	—				
ppm = parts per million by volume μ g/m ³ = micrograms per cubic meter AAM = annual arithmetic mean AGM = annual geometric mean Source: SCAOMD 2003	r						

TABLE 5.4-1STATE AND FEDERAL AMBIENT AIR QUALITY STANDARDS

Particulate Matter

Particulate matter (PM10 and PM2.5) is a major air pollutant that consists of tiny solid or liquid particles of soot, dust, smoke, fumes, and aerosols. The size of the particles allows them to easily enter the air sacs in the lungs where they may be deposited, resulting in adverse health effects. Particulate matter also causes visibility reduction. PM10 and PM2.5 are criteria air pollutants. PM10 refers to particulate matter ten microns or smaller; PM2.5 are two and one-half microns and smaller.

Particulate levels in the area are generally due to natural sources, grading operations, and motor vehicles. According to the EPA, some people are much more sensitive than others to breathing fine particles (PM10 and PM2.5). People with influenza, chronic respiratory and cardiovascular diseases, and the elderly may suffer worsening illness and premature death due to breathing these fine particles. People with bronchitis can expect aggravated symptoms from breathing in fine particles. Children may experience decline in lung function due to breathing in PM10 and PM2.5. Other groups considered sensitive are smokers and people who cannot breathe well through their noses. Exercising athletes are also considered sensitive because many breathe through their mouths.

Reactive Organic Gas (ROG)

Reactive organic gas (ROG) is a reactive chemical gas, composed of hydrocarbons that may contribute to the formation of smog (also see volatile organic compounds [VOC]).

Volatile Organic Compounds (VOC)

VOCs are hydrocarbon compounds that exist in the ambient air. VOCs contribute to the formation of smog and/or may, themselves, be toxic. VOCs often have an odor. Some examples of VOCs include gasoline, alcohol, and the solvents used in paints.

SCAB Attainment Status

The EPA has designated the SCAB as an extreme non-attainment area for 1-hour ozone; a "severe 17" non-attainment area for 8-hour ozone; a serious non-attainment area for PM10 and CO; and a non-attainment area for PM2.5. The Basin must achieve the federal 1-hour ozone standard by November 15, 2010. The deadline for achieving the federal PM10 and 8-hour CO standards is December 31, 2006. The 2007 South Coast AQMP, which is currently under preparation, will address issues related to the Basin's current attainment status and include measures to reduce emissions and comply with federal and state air quality standards.

Air Quality Mitigation Approaches in the SCAB

Consistent with its overall goal to meet applicable state and federal requirements and to demonstrate attainment with ambient air quality standards, the 2003 AQMP, which builds on the 1997 AQMP, uses 2 tiers of emission reduction measures: (1) short- and intermediate-term measures and (2) long-term measures.

Short- and intermediate-term measures propose the application of available technologies and management practices by the year 2010. These measures rely on known technologies and proposed actions to be taken by several agencies that currently have statutory authority to implement such measures. These measures rely on both traditional command and control and alternative approaches for implementation.

To ultimately achieve ambient air quality standards, additional emission reductions will be necessary. Long-term measures rely on the advancement of technologies and control methods that can reasonably be expected to occur between 2010 and 2030. These long-term measures rely on further development and refinement of known low- and zero-emission control technologies for both mobile and stationary sources, along with technological breakthroughs.

Meteorology and Climate

Meteorological conditions are an important factor in air quality. The SCAB lies within a large coastal plain that contains broad valleys and low hills. It is bordered by the Pacific Ocean on the southwest side and surrounded by mountains on all other sides. The local climate is typically mild, with cool summers, mild winters, and infrequent rainfall. While these conditions result in poor air quality throughout much of the SCAB, south Orange County enjoys relatively healthful air quality most of the time.

The average annual temperature in south Orange County is 62°F with limited diurnal and seasonal variations. The San Juan Capistrano area averages slightly more than 10 inches of rainfall annually. Seasonal rainfall amounts are strongly dependent upon large-scale global weather patterns such as El Niño or La Niña conditions.

Winds at the Prima Deshecha site are characterized by onshore flows from the southwest and west at seven to nine miles per hour during the day, especially in the summer. At night, particularly in the winter, a weak offshore flow develops and the wind direction rotates through the southeast and south until the onshore flow becomes dominant in the early afternoon. Santa Ana winds occasionally result in strong, hot, dry, gusty winds from the northeast and east across the south Orange County area.

Temperature inversions that control the vertical depth through which pollutants can be mixed contribute to poor air quality conditions throughout much of southern California. The daytime onshore flow of marine air is capped by a massive dome of warm air that acts like a giant lid over the SCAB. As the clean ocean air moves inland, pollutants are continually added from below without any dilution from above and unhealthful levels of smog, consisting mainly of ozone (O_3), are formed. A second inversion forms at night as cool air pools in low elevations while the air aloft remains warm. Shallow radiation inversions are formed, especially in winter. These tend to trap pollutants near intensive traffic sources such as freeways and shopping centers, forming localized violations of carbon monoxide (CO), or CO hot spots.

Site Micrometeorology

Pre-monitoring efforts conducted during preparation of EIR 575 formed the basis for selecting the optimal sites in developing monitoring plans for the Prima Deshecha Landfill, as required by several mandatory air monitoring programs. The measurement programs showed that the general pattern of offshore flows at night and onshore flows during the day is observed across much of the project site.

Nocturnal winds follow the canyon terrain from northeast to southwest. Because this light flow occurs in conjunction with low-level trapping inversions, any air emissions (such as odorous landfill gas generation [LFG]) are concentrated near the desilting basin at the south end of WMU1 and near the homes south of the landfill beyond the site boundary. Historical odor complaints have come primarily from the small group of residents near the nocturnal landfill outflow.

Daytime winds are predominantly south-to-north because of canyon trending. The net daytime transport route is northward and slightly eastward toward undeveloped parcels that are currently used for cattle grazing.

Existing Air Quality

Air quality at any site is affected by regional air quality and local pollutant sources. Regional air quality is determined by the release of pollutants throughout the air basin. Estimates for existing emissions in the SCAB are included in the *1997 Air Quality Management Plan* (SCAQMD 1996). The data indicate that mobile sources (i.e., on-road motor vehicles) are the major contributor to regional emissions, accounting for approximately 51 percent of VOC, 63 percent of NOx emissions, and approximately 78 percent of CO emissions.

The Prima Deshecha Landfill is located near the Saddleback Valley and Central Orange County air quality monitoring stations. The data collected at these stations are considered representative of the air quality experienced in the vicinity of the landfill.

Table 5.4-2 presents the federal and state air quality standards as well as the monitored pollutant levels from the Saddleback Valley monitoring station (for CO, ozone, PM10, and PM2.5) and the Central Orange County monitoring station (for NOx).

As Table 5.4-2 illustrates, ozone and particulate matter (both PM10 and PM2.5) are the air pollutants of primary concern in the vicinity of the Prima Deshecha Landfill.

<u>Ozone</u>

Both the state and the federal 1-hour ozone standards were exceeded several days during each of the past three years at the Saddleback Valley monitoring station. Likewise, the federal 8-hour ozone standard was exceeded a few days.

Particulate Matter

State PM10 standards were exceeded on only two occasions in the past three years at the Saddleback Valley station. Federal PM2.5 standards which were not available.

Carbon Monoxide

Currently, CO levels in the project region are in compliance with the state and federal 1-hour and 8-hour standards.

Nitrogen Dioxide

NO₂ levels in the project region are in compliance with state and federal standards.

It should be noted that air quality has improved in the vicinity of the Prima Deshecha Landfill, as it has throughout the entire SCAB, during the past five years (i.e., since preparation of FEIR 575). Consequently, exceedances of state and federal air quality standards are on the decline. The SCABs air quality improvements are attributable to several factors, including the availability and use of cleaner fuels and cleaner vehicles, as well as implementation of more stringent pollution controls.

TABLE 5.4-2AIR QUALITY LEVELS MEASURED AT THE SADDLEBACK VALLEY
MONITORING STATION

	California	National	-	%	Max.	Days State Standard	Days National Standard		
Pollutant	Standard	Standard	Year	Meas. ^a	Level	Exceeded ^D	Exceeded		
Ozone	0.09 ppm	0.12 ppm	2005	100	0.125	3	1		
	for 1 hr.	for 1 hr.	2004	100	0.116	11	0		
			2003	99	0.153	16	4		
Ozone	No	0.08 ppm	2005	100	0.085	6	1		
	Standard	for 8 hr.	2004	100	0.089	20	2		
			2003	99	0.105	n/a	8		
CO	20 ppm	35 ppm	2005	100	2.0	0	0		
	for 1 hr.	for 1 hr.	2004	100	2.0	0	0		
			2003	99	3.0	0	0		
CO	9.0 ppm	9 ppm	2005	97	1.6	0	0		
	for 8 hr.	for 8 hr.	2004	99	1.6	0	0		
			2003	99	1.8	0	0		
NOx	0.25 ppm		2005	100	0.09	0	0		
	for 1 hr.		2004	99	0.12	0	0		
			2003	99	0.13	0	0		
Particulates	50 µg/m³	150 µg/m³	2005	15	41	0	0		
PM10	for 24 hr.	for 24 hr.	2004	16	47	0	0		
(24 Hour)			2003	16	64	2	0		
Particulates	20 µg/m³	50 µg/m³	2005	16	19.0	n/a	n/a		
PM10			2004	16	23.7	n/a	n/a		
(Annual)			2003	17	31.8	n/a	n/a		
Particulates	No State	65 µg/m³	2005	31	35.4	n/a	n/a		
PM2.5	Standard	For 24 hr.	2004	30	49.4	n/a	n/a		
(24 Hour)			2003	30	95	n/a	n/a		
Particulates	12 µg/m³	15 µg/m³	2005	31	10.7	n/a	n/a		
PM 2.5	AAM ³	AAM ^c	2004	30	12.1	n/a	n/a		
Annual			2003	30	13.1	n/a	n/a		
n/a Data not a	vailable.								
^a Percent of	^a Percent of year that measurements were made.								
c Appuel Arith	TU and PIVI 2.5 2	4-nour standards	s, daily monit	oring is not pe	normea.				
Annual Arithmetic Mean									

Note: Central Orange County Monitoring Station data accessed for nitrogen dioxide.

Source: SCAQMD 2006

Existing Site Pollution Sources

Air pollutants are emitted in limited amounts from a variety of activities at the Prima Deshecha Landfill site. Existing sources of air pollutants include:

- Exhaust emissions from loaded packer trucks and public vehicles traveling from the landfill gate to the working face of the landfill and the return trips of the empty vehicles back to the site exit.
- Exhaust pollution from scrapers, dozers, compactors, water trucks, and other operations equipment.
- Combustion pollutants created during the combustion LFG in the energy recovery facility (ERF). The ERF combusts LFG to generate electricity. To further reduce emissions from unburned hydrocarbons, each ERF unit contains an afterburner that destroys 80 percent of the ROGs that are not destroyed in the engine-generator set.

- Combustion of LFG in a waste gas flare. The flare system is a back-up system to the ERF, and used only if the ERF is inoperative for maintenance or repair.
- Surface emission of LFG containing ROGs and trace amounts of toxic air contaminants from the fraction of LFG not captured by the control system.
- Fugitive dust from vehicle travel on unpaved surfaces, from the extraction and transport of cover material, from the placement of daily cover, and from dust in certain types of refuse such as demolition debris or dirty scrap lumber. The amount of dust generated is highly variable and does not lend itself to precise quantification. Because such dust generation is largely determined by the amount of dust control being implemented at the landfill site, the focus on fugitive dust analysis is less on emissions estimates and more on the implementation of conscientious and effective housekeeping procedures.

Existing Site Emissions

The information in Table 5.4-3 is from FEIR 575 Section 4.9, Air Quality. Because daily emissions would not change as a result of the Proposed Project, this information is considered to be current and correct for the Proposed Project.

Toxic Air Contaminants

The ERFs operating at the Prima Deshecha Landfill have the potential to emit toxic air contaminants. However, SCAQMD's Rule 1401 prohibits the air district from issuing an authority to construct or a permit to operate to any facility that would create an unacceptable public health risk from the emission of toxic air contaminants. Unacceptable individual cancer risk from a permitted source is one change in a million. If Toxics–Best Available Control Technology (T-BACT) is employed, the allowable risk is increased to ten in one million. According to the analysis contained in FEIR 575, the ERF at the Prima Deshecha Landfill underwent a Tier 4 (i.e., full health risk) Assessment as part of its permitting process. The calculated risk for the existing ERF/afterburner system was determined to be less than one in one million at full capacity: an acceptable level.

	Pollutant Emissions (Ibs/day)							
	СО	ROG	NOx	PM10	SOx			
Mobile Source Emissions	336	50	545	34	39			
Energy Recovery Facility	860	152	300	286	15			
Uncaptured LFG Surface Emissions		1,456			—			
TOTAL	1,196	1,658	845	320	54			
SCAQMD Significance Threshold	550	55	55	150	150			
Significant?	YES	YES	YES	YES	NO			
Source: Giroux & Associates 1999 (FEIRS	575, Table 4.9-1	3)		<u> </u>				

TABLE 5.4-3EXISTING (2005) DAILY POLLUTANT EMISSIONS INVENTORY AT PRIMA
DESHECHA LANDFILL (4,000 TPD)

TABLE 5.4-4 FORECAST (2020) DAILY POLLUTANT EMISSIONS INVENTORY AT PRIMA DESHECHA LANDFILL (4,000 TPD)

	Pollutant Emissions (Ibs/day)							
	СО	ROG	NOx	PM10	SOx			
Mobile Source Emissions	310	37	404	34	39			
Energy Recovery Facility	979	173	341	325	17			
LFG Combustion	82	13	99	22	14			
Uncaptured LFG Surface Emissions		2,803						
TOTAL	1,371	3,026	844	381	70			
SCAQMD Significance Threshold	550	55	55	150	150			
Significant?	YES	YES	YES	YES	NO			
Source: Giroux & Associates 1999 (FEIRS 575, Table 4.9-13)								

Odors

Odors at the Prima Deshecha Landfill result from refuse as well as LFG operations. Under worst-case conditions, fresh trash odor is detectable up to one-half mile from landfill sites (County of Orange 1991). With prevailing daytime southwest-to-northeast winds at the Prima Deshecha Landfill, occasional fresh trash odor detection would be confined to on-site locations away from any existing or planned homes. Therefore, SEIR 575 determined that daytime odors from landfilling would not have a significant odor impact on any sensitive receptor population.

According to FEIR 575, odor complaints at the Prima Deshecha Landfill have almost always been in the evening, after disposal activity has ceased and all fresh trash has been buried, and originate less than one mile from the center of Landfill Zone 1.

Existing On-Site Odor Control

As described in detail in Section 5.4.4, the IWMD has implemented several measures to reduce the potential for odor nuisance from the Prima Deshecha Landfill.

Dust

Dust impacts can result from the excavation and placement of cover, from travel on unpaved roadways, and from dusty loaded refuse containing materials such as demolition debris. Dust generated during these types of landfilling activities may range from very small particles (PM2.5) to larger diameter particles. Dust generation and associated impacts are almost exclusively a daytime phenomenon at landfills.

The SCAQMD requires dust control from surface disturbance through Rule 403. Rule 403 contains a performance standard that limits the difference in downwind versus upwind dust levels.

Existing On-Site Dust Control

As described in detail in Section 5.4.4 below, IWMD has implemented several measures to reduce the potential for dust nuisance from the Prima Deshecha Landfill.

5.4.2 SIGNIFICANCE CRITERIA

For purposes of analysis, the Proposed Project's air quality impacts are separated into shortterm impacts due to construction and long-term permanent impacts from operations. The County of Orange, as the lead agency for the Prima Deshecha Landfill project, is responsible for making determinations regarding the existence of significant air quality impacts.

5.4.3 POTENTIAL IMPACTS

The criteria used to determine the significance of potential project-related air quality impacts are taken from the Initial Study checklist form in Appendix G of the State CEQA Guidelines. Based on these thresholds, the project would result in a significant impact related to air quality if it would:

(1) Conflict with or obstruct implementation of the applicable air quality plan.

(2) Violate any air quality standard or contribute substantially to an existing or projected air quality violation.

The significance thresholds recommended by the SCAQMD in its *CEQA Air Quality Handbook* are presented in Table 5.4-5. Construction and operational emissions are considered by the SCAQMD to be significant if they exceed these thresholds.

TABLE 5.4-5								
SCAQMD REGIONAL POLLUTANT EMISSION								
THRESHOLDS OF SIGNIFICANCE								

	Pollutant Emissions (lbs/day)								
	СО	ROG	NOx	PM10	SOx				
Construction	550	75	100	150	150				
Operation	550	55	55	150	150				
Source: SCAQMD 1993									

In addition, project impacts would be significant if they exceed the following California standards for localized CO:

- 1-hour CO standard of 20.0 parts per million (ppm)
- 8-hour CO standard of 9.0 ppm.

(3) Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or state ambient air quality standard (including releasing emissions, which exceed quantitative thresholds for ozone precursors).

(4) Expose sensitive receptors to substantial pollutant concentrations.

(5) Create objectionable odors affecting a substantial number of people.

As discussed above, the proposed project would not result in a change in air quality emissions beyond those identified in FEIR 575, which analyzed the impacts associated with daily operations of up to 4,000 tpd and future recreational and roadway uses. No new impacts are anticipated. Consequently, no additional mitigation measures are required.

5.4.4 MITIGATION MEASURES

Notwithstanding the fact that the Proposed Project would be consistent with FEIR 575 (and would not, therefore, require additional mitigation), the SCAQMD has recently adopted substantial modifications to Rule 403 – Fugitive Dust, which is applicable to operations at the Prima Deshecha Landfill. Section 5.4.4.1, Previously Adopted Mitigation, describes the mitigation measures already in place at the Prima Deshecha Landfill. Section 5.4.4.2, Recommended Mitigation Measure Updates, provides recommendations for bringing the landfill up-to-date with current provisions of SCAQMD Rule 403.

5.4.4.1 Previously Adopted Mitigation

Note: The numbering in this section corresponds with the numbering in FEIR 575.

- Impact 4.9-1: Potential odor nuisance could be associated with refuse brought to the landfill.
- MM 4.9-1: Landfill fee station personnel and/or landfill refuse inspectors shall reject extremely odorous loads for disposal in the landfill.

Impact 4.9-2: A potential odor nuisance affecting recreational users could be associated with fresh refuse odor at the active face of the landfill.

MM 4.9-2: The active face of the landfill shall be covered daily. If the active face is in close proximity and upwind of on-site recreation uses, masking or neutralization agents may be added to exposed refuse to reduce the odor nuisance effects on the adjacent recreation uses.

Impact 4.9-3: A potential odor nuisance could be associated with LFG emissions if they are not collected and disposed by an efficient system.

MM 4.9-3: The IWMD shall design, construct and operate new landfill areas in Zones 1 and 4 with LFG systems to maximize the collection of LFG. The LFG systems will include continuous monitoring of the LFG collection system to maximize efficient collection of LFG generated in these areas.

Impact 4.9-4: A potential odor nuisance could result from LFG leaks through cracks or flaws in the landfill cover in active and inactive landfill areas.

MM 4.9-4: During landfill operations, the IWMD shall continue regular visual inspections of the landfill cover and monitoring of LFG emissions throughout the entire refuse fill areas. The purpose of these inspections is to locate cracks or other defects or flaws in the landfill cover, which may allow LFG to escape. When such areas are identified, the IWMD will implement the appropriate corrective action as soon as feasible. These corrective actions may include application and compaction of additional cover material, adjustment of the existing LFG control system and/or installation of new LFG control facilities.

Impact 4.9-5: A potential odor nuisance could be associated with LFG generated in the active and inactive landfill areas in landfilling Zones 1 and 4.

MM 4.9-5: During landfill operations, the IWMD shall conduct periodic odor surveys on the landfill site and at various points in the area surrounding the site. The IWMD shall conduct odor surveys if any odors from the landfill are detected off site
and reported by nearby residents. When the source of these odors is identified, the IWMD will implement the appropriate corrective action as soon as feasible. These corrective actions may include application and compaction of additional cover material, use of masking or neutralizing agents, adjustment of the existing LFG control system and/or installation of new LFG control facilities.

Impact 4.9-6: A potential dust nuisance could result to recreational users from landfilling operations.

MM 4.9-6: During landfill operations, the IWMD shall ensure that landfill operations areas that are to be left exposed temporarily, including top deck and excavation slopes, are sprayed periodically with water, as needed.

Impact 4.9-7: Exposed ground surfaces could result in a potential dust nuisance after the termination of active landfilling.

MM 4.9-7: On landfilled areas that are no longer in use, the IWMD will, as appropriate, incorporate dust control systems or vegetative covers, consistent with the Final Closure Plans and with IWMD's approved Rule 403 Compliance Plan for landfilling Zones 1 and 4.

Impact 4.9-8: Dusty refuse could result in a potential dust nuisance.

MM 4.9-8: During landfill operations, the landfill fee station personnel and/or landfill refuse inspectors shall refrain from accepting dusty loads of refuse for disposal in either landfilling Zone 1 or 4. Alternatively, at the discretion of landfill personnel, dusty loads of refuse may be accepted for disposal, if they are sprayed with water prior to leaving the fee station and accessing the active face of the landfill.

Impact 4.9-9: On-site vehicular travel on unpaved roads could result in a potential dust nuisance.

- MM 4.9-9a: During landfill operations, the IWMD shall maintain water trucks on site to spray water on on-site unpaved roads as needed to minimize the generation of dust as vehicles travel on these roads, as per IWMD's approved Rule 403 Compliance Plan.
- MM 4.9-9b: During landfill operations, the IWMD shall, to the extent feasible while still maintaining appropriate landfill operations, restrict vehicular travel on unpaved roads on the site. In the event that unpaved roads must be used, the IWMD shall spray water on these roads as needed.
- MM 4.9-9c: As unpaved on-site roads are removed from active service, the IWMD will spray these areas with a hydromulch solution or synthetic binder.

Impact 4.9-10: A potential dust nuisance could be associated with graded areas or areas where the vegetation has been removed or severely disturbed.

MM 4.9-10: During landfill operations, the IWMD will use the on-site water trucks to spray water on graded areas or areas where the vegetation has been removed or severely disturbed as a result of landfilling activities, as per IWMD's approved Rule 403 Compliance Plan.

5.4.4.2 Recommended Mitigation Measure Updates

Particulate Emission (PM10) Control

MM 5.4-1: IWMD and its contractors shall be required to comply with regional rules to reduce air pollutant emissions. SCAQMD Rule 401 sets limits on the opacity of visible plumes of dust resulting from activities at the landfill. SCAQMD Rule 402 requires that air pollutant emissions generated at the landfill not be a nuisance off site. SCAQMD Rule 403 requires that fugitive dust be controlled with the best available control measures so that the presence of such dust does not remain visible in the atmosphere beyond the property line of the emission source. Two options are presented in Rule 403: monitoring of particulate concentrations, or active control. Monitoring involves a sampling network around the project with no additional control measures unless specified concentrations are exceeded. The active control option does not require any monitoring, but requires that a list of measures be implemented on a daily basis.

SCAQMD Rule 403 requires that "best available control measures" be utilized whenever a dust-generating activity occurs in the Air Basin. These measures are listed in Table 1 of Rule 403 and called out in Table 5.4-6 below. It is important to note that all applicable measures from Table 5.4-6 should be implemented to achieve the required PM10 emissions reductions.

Rule 403 requires that "Large Projects" implement additional measures. A Large Project is defined as "any active operations on property which contains 50 or more acres of disturbed surface area; or any earth-moving operation with a daily earth-moving or throughput volume of 3,850 cubic meters (5,000 cubic yards) or more than three times during the most recent 365 day period. The Prima Deshecha Landfill would be considered a Large Project under Rule 403. Therefore, the landfill is required to implement the applicable actions specified in Table 2 of the Rule. Table 2 from Rule 403 is presented below as Table 5.4-7.

As a Large Operation, the landfill will also be required to:

- Submit a fully executed Large Operation Notification (SCAQMD Form 403N) to the SCAQMD Executive Officer within 7 days of qualifying as a large operation;
- Include, as part of the notification, the name(s), address(es), and phone number(s) of the person(s) responsible for the submittal, and a description of the operation(s), including a map depicting the location of the site;
- Maintain daily records to document the specific dust-control actions taken, maintain such records for a period of not less than three years; and make such records available to the Executive Officer upon request;
- Install and maintain project signage with project contact signage that meets the minimum standards of the *Rule 403 Implementation Handbook*, prior to initiating any earthmoving activities;

TABLE 5.4-6 REQUIRED BEST AVAILABLE CONTROL MEASURES (SCAQMD RULE 403, TABLE 1)

	Control Measure	Guidance		
Backf	illing			
01-1	Stabilize backfill material when not actively bandling: and	Mix backfill soil with water prior to moving		
01-2	Stabilize backfill material during handling; and	 Dedicate water truck or high capacity hose to backfilling equipment 		
01-3	Stabilize soil at completion of activity.	 Empty loader bucket slowly so that no dust plumes are generated 		
		Minimize drop height from loader bucket		
Cleari	ng and Grubbing			
02-1	Maintain stability of soil through pre-watering of site prior to clearing and grubbing; and	Maintain live perennial vegetation where possible Apply water in sufficient quantity to prevent		
02-2	Stabilize soil during clearing and grubbing activities; and	generation of dust plumes		
02-3	Stabilize soil immediately after clearing and grubbing activities.			
Cleari	ng Forms			
03-1	Use water spray to clear forms; or	Use of high pressure air to clear forms may cause		
03-2	Use sweeping and water spray to clear forms; or	exceedance of Rule requirements		
03-3	Use vacuum system to clear forms.			
Crush	ing			
04-1	Stabilize surface soils prior to operation of support	 Follow permit conditions for crushing equipment 		
04.2	Stabilize meterial offer cruching	 Pre-water material prior to loading into crusher 		
04-2	Stabilize material after crushing.	 Monitor crusher emissions opacity 		
		 Apply water to crushed material to prevent dust plumes 		
Cut ar	nd Fill			
05-1 05-2	Stabilize soil during and after cut and fill activities.	 For large sites, pre-water with sprinklers or water trucks and allow time for penetration 		
	, i i i i i i i i i i i i i i i i i i i	 Use water trucks/pulls to water soils to depth of cut prior to subsequent cuts 		
Demo	lition – Mechanical/Manual			
06-1	Stabilize wind erodible surfaces to reduce dust; and	 Apply water in sufficient quantities to prevent the generation of visible dust plumes 		
06-2	Stabilize surface soil where support equipment and vehicles will operate; and			
06-3	Stabilize loose soil and demolition debris; and			
06-4	Comply with AQMD Rule 1403.			
Distur	bed Soil			
07-1	Stabilize disturbed soil throughout the construction site; and	 Limit vehicular traffic and disturbances on soils where possible 		
07-02	Stabilize disturbed soil between structures	 If interior block walls are planned, install as early as possible 		
		 Apply water or a stabilizing agent in sufficient quantities to prevent the generation of visible dust plumes 		
Earth-	Moving Activities			
08-1 08-2	Pre-apply water to depth of proposed cuts; and Re-apply water as necessary to maintain soils in a	Grade each project phase separately, timed to coincide with construction phase		
002	damp condition and to ensure that visible	Upwind fencing can prevent material movement on		
	emissions do not exceed 100 feet in any direction; and	site		
08-3	Stabilize soils once earth-moving activities are	Apply water or a stabilizing agent in sufficient guantities to prevent the generation of visible dust		
	complete.	plumes		

TABLE 5.4-6 (Continued) REQUIRED BEST AVAILABLE CONTROL MEASURES (SCAQMD RULE 403, TABLE 1)

	Control Measure	Guidance
Impor	ting/Exporting of Bulk Materials	
09-1	Stabilize material while loading to reduce fugitive dust emissions; and	Use tarps or other suitable enclosures on haul trucks
09-2	Maintain at least six inches of freeboard on haul vehicles; and	Check belly-dump truck seals regularly and remove any trapped rocks to prevent spillage
09-3	Stabilize material while transporting to reduce fugitive dust emissions; and	Comply with track-out prevention/mitigation requirements
09-4	Stabilize material while unloading to reduce fugitive dust emissions; and	 Provide water while loading and unloading to reduce visible dust plumes
09-5	Comply with Vehicle Code Section 23114.	
Lands	scaping	
10-1	Stabilize soils, materials, slopes	Apply water to materials to stabilize Maintain materials in a crusted condition
		Maintain effective cover over materials
		• Stabilize sloping surfaces using soil binders until vegetation or ground cover can effectively stabilize the slopes
		Hydroseed prior to rain season
Road	Shoulder Maintenance	
11-1	Apply water to unpaved shoulders prior to clearing; and	Installation of curbing and/or paving of road shoulders can reduce recurring maintenance costs
11-2	Apply chemical dust suppressants and/or washed gravel to maintain a stabilized surface after completing road shoulder maintenance.	Use of chemical dust suppressants can inhibit vegetation growth and reduce future road shoulder maintenance costs
Scree	ning	
12-1 12-2	Pre-water material prior to screening; and Limit fugitive dust emissions to opacity and plume	Dedicate water truck or high capacity hose to screening operation
12-3	length standards; and Stabilize material immediately after screening	 Drop material through the screen slowly and minimize drop height
		 Install wind barrier with a porosity of no more than 50% upwind of screen to the height of the drop point
Stagir	ng Areas	· · · · · · · · · · · · · · · · · · ·
13-1	Stabilize staging areas during use; and	 Limit size of staging area
13-2	Stabilize staging area soils at project completion.	Limit vehicle speeds to 15 miles per hour
		 Limit number and size of staging area entrances/exists
Stock	piles/Bulk Material Handling	
14-1	Stabilize stockpiled materials.	Add or remove material from the downwind portion
14-2	Stockpiles within 100 yards of off-site occupied	of the storage pile
	buildings must not be greater than eight feet in	Maintain storage piles to avoid steep sides or faces
	height; or must have a road bladed to the top to	
	operational water irrigation system that is capable	
	of complete stockpile coverage.	
Traffic	c Areas for Construction Activities	
15-1	Stabilize all off-road traffic and parking areas; and	Apply gravel/paving to all haul routes as soon as
15-2	Stabilize all haul routes; and	possible to all future roadway areas
15-3	Direct construction traffic over established haul	Barriers can be used to ensure vehicles are only used on established parking areas/haul routes

TABLE 5.4-6 (Continued) REQUIRED BEST AVAILABLE CONTROL MEASURES (SCAQMD RULE 403, TABLE 1)

	Control Measure	Guidance		
Trenc	hing			
16-1	Stabilize surface soils where trencher or excavator and support equipment will operate; and	 Pre-watering of soils prior to trenching is an effective preventive measure. 		
16.2	Stabilize soils at the completion of trenching activities.	 For deep trenching activities, pre-trench to 18 inches soak soils via the pre-trench and resuming trenching 		
		 Washing mud and soils from equipment at the conclusion of trenching activities can prevent crusting and drying of soil on equipment 		
Truck	Loading			
17-1 17.2	Pre-water material prior to loading; and Ensure that freeboard exceeds six inches (CVC	 Empty loader bucket such that no visible dust plumes are created 		
	23114)	 Ensure that the loader bucket is close to the truck to minimize drop height while loading 		
Turf C	Overseeding			
18-1	Apply sufficient water immediately prior to conducting turf vacuuming activities to meet opacity and plume length standards; and	 Haul waste material immediately off site 		
18-2	Cover haul vehicles prior to exiting the site.			
Unpay	ved Roads/Parking Lots			
19-1	Stabilize soils to meet the applicable performance standards; and	 Restricting vehicular access to established unpav travel paths and parking lots can reduce 		
19-2	Limit vehicular travel to established unpaved roads (haul routes) and unpaved parking lots.	stabilization requirements		
Vacar	it Land			
20-1	In instances where vacant lots are 0.10 acre or larger and have a cumulative area of 500 square feet or more that are driven over and/or used by motor vehicles and/or off-road vehicles, prevent motor vehicle and/or off-road vehicle trespassing, parking and/or access by installing barriers, curbs, fences, gates, posts, signs, shrubs, trees or other effective control measures.			

- Identify a dust control supervisor that is employed by or contracted with the property owner or developer, is on the site or available on-site within 30 minutes during working hours, has the authority to expeditiously employ sufficient dust mitigation measures to ensure compliance with all Rule requirements, and has completed the AQMD Fugitive Dust Control Class and has been issued a valid Certificate of Completion for the class; and
- Notify the SCAQMD Executive Officer in writing within 30 days after the site no longer qualifies as a large operation.

TABLE 5.4-7 FUGITIVE DUST CONTROL ACTIONS (SCAQMD RULE 403, TABLE 1)

Fugitive Dust Source Category			
	Control Actions		
Earth-r	moving (except construction cutting and filling areas, and mining operations)		
(1a)	Maintain soil moisture content at a minimum of 12 percent, as determined by the ASTM [American Society for Testing and Materials] method D2216, or other equivalent method approved by the Executive Officer, the		
	the first three hours of active operations during a calendar day, and two such evaluations each subsequent		
	four-hour period of active operations; OR		
(1a-1)	For any earth-moving which is more than 100 feet from all property lines, conduct watering as necessary to prevent visible dust emissions from exceeding 100 feet in length in any direction.		
Earth-r	moving: Construction fill areas		
(1b)	Maintain soil moisture content at a minimum of 12 percent, as determined by ASTM method D2216, or other equivalent method approved by the Executive Officer, the California Air Resources Board, and the U.S. EPA.		
	For areas which have an optimum moisture content for compaction of less than 12 percent, as determined by ASTM Method 1557 or other equivalent method approved by the Executive Officer and the California Air		
	Resources Board and the U.S. EPA, complete the compaction process as expeditiously as possible after		
	achieving at least 70 percent of the optimum soil moisture content. Two soil moisture evaluations must be conducted during the first three hours of active operations during a calendar day, and two such evaluations		
	during each subsequent four-hour period of active operations.		
Earth-r	moving: Construction cut areas and mining operations		
(1c)	Conduct watering as necessary to prevent visible emissions from extending more than 100 feet beyond the		
	active cut or mining area unless the area is inaccessible to watering vehicles due to slope conditions or other		
	safety factors.		
Distur	bed surface areas (except completed grading areas)		
(2a/b)	Apply dust suppression in sufficient quantity and frequency to maintain a stabilized surface. Any areas which cannot be stabilized, as evidenced by wind driven fugitive dust must have an application of water at least		
Dist	twice per day to at least 80 percent of the unstabilized area.		
Disture	bed surface areas: Completed grading areas		
(2c)	Apply chemical stabilizers within five working days of grading completion; OR		
(2d)	Take actions (3a) or (3c) specified for inactive disturbed surface areas.		
Inactiv	e disturbed surface areas		
(3a)	Apply water to at least 80 percent of all inactive disturbed surface areas on a daily basis when there is evidence of wind driven fugitive dust, excluding any areas which are inaccessible to watering vehicles due to excessive slope or other safety conditions; OR		
(3b)	Apply dust suppressants in sufficient quantity and frequency to maintain a stabilized surface; OR		
(3c)	Establish a vegetative ground cover within 21 days after active operations have ceased. Ground cover must be of sufficient density to expose less than 30 percent of unstabilized ground within 90 days of planting, and at all times thereafter;		
(3d)	Utilize any combination of control actions (3a), (3b), and (3c) such that, in total, these actions apply to all inactive disturbed surface areas		
Unnav	ed Roads		
(4a)	Water all roads used for any vehicular traffic at least once per every two hours of active operations [3 times]		
(10)	per normal 8 hour work day]; OR		
(4b)	Water all roads used for any vehicular traffic once daily and restrict vehicle speeds to 15 miles per hour; OR		
(4c)	Apply a chemical stabilizer to all unpaved road surfaces in sufficient quantity and frequency to maintain a stabilized surface.		

TABLE 5.4-7 (Continued) FUGITIVE DUST CONTROL ACTIONS (SCAQMD RULE 403, TABLE 1)

Fugitive Dust Source Category			
	Control Actions		
Open s	storage piles		
(5a)	Apply chemical stabilizers;		
	OR		
(5b)	Apply water to at least 80 percent of the surface area of all open storage piles on a daily basis when there is evidence of wind driven fugitive dust;		
	OR		
(5c)	Install temporary coverings;		
	OR		
(5d)	Install a three-sided enclosure with walls with no more than 50 percent porosity which extend, at a minimum, to the top of the pile. This option may only be used at aggregate-related plants or at cement manufacturing facilities.		
All Cat	regories		
(6a)	Any other control measures approved by the Executive Officer and the U.S. EPA as equivalent to the methods specified in Table 2 may be used.		

Rule 403 also requires that the construction activities "shall not cause or allow PM10 levels to exceed 50 micrograms per cubic meter when determined by simultaneous sampling, as the difference between upwind and down wind sample." Large Projects that cannot meet this performance standard are required to implement the applicable actions specified in Table 3 of Rule 403. Table 3 from Rule 403 is presented below as Table 5.4-8.

Further, Rule 403 requires that that the project shall not "allow track-out to extend 25 feet or more in cumulative length from the point of origin from an active operation." All track-out from an active operation is required to be removed at the conclusion of each workday or evening shift. Any active operation with a disturbed surface area of five or more acres, or with a daily import or export of 100 cubic yards or more of bulk materials must utilize at least one of the measures listed in Table 5.4-9 at each vehicle egress from the site to a paved public road.

TABLE 5.4-8 CONTINGENCY CONTROL MEASURES FOR LARGE OPERATIONS (SCAQMD RULE 403, TABLE 3)

	Fugitive Dust Source Category Control Actions				
Earth-I	moving				
(1A)	Cease all active operations; OR				
(2A)	Apply water to soil not more than 15 minutes prior to moving such soil.				
Distur	bed surface areas				
(0B)	On the last day of active operations prior to a weekend, holiday, or any other period when active operations will not occur for not more than four consecutive days: apply water with a mixture of chemical stabilizer diluted to not less than 1/20 of the concentration required to maintain a stabilized surface for a period of six months; OR				
(1B)	Apply chemical stabilizers prior to wind event;				
(2B)	Apply water to all unstabilized disturbed areas 3 times per day. If there is any evidence of wind driven fugitive dust, watering frequency is increased to a minimum of four times per day; OR				
(3B)	Take the actions specified in Table 2, Item (3c); OR				
(4B)	Utilize any combination of control actions (1B), (2B), and (3B) such that, in total, these actions apply to all disturbed surface areas.				
Unpav	ed Roads				
(1C)	Apply chemical stabilizers prior to wind event; OR				
(2C)	Apply water twice per hour during active operation; OR				
(3C)	Stop all vehicular traffic.				
Open S	Storage Piles				
(1D)	Apply water twice per hour; OR				
(2D)	Install temporary coverings.				
Paved	Road Track-Out				
(1E)	Cover all haul vehicles; OR				
(2E)	Comply with the vehicle freeboard requirements of Section 23114 of the California Vehicle Code for both public and private roads.				
All Cat	egories				
(1F)	Any other control measures approved by the Executive Officer and the U.S. EPA as equivalent to the methods specified in Table 3 may be used.				

TABLE 5.4-9TRACK OUT CONTROL OPTIONS

- (A) Install a pad consisting of washed gravel (minimum-size: one inch) maintained in a clean condition to a depth of at least six inches and extending at least 20 feet wide and 50 feet long.
- (B) Pave the surface extending at least 100 feet and a width of at least 20 feet wide.
- (C) Utilize a wheel shaker/wheel spreading device consisting of raised dividers (rails, pipe, or grates) at least 24 feet long and 10 feet wide to remove bulk material from tires and vehicle under carriages before vehicles exit the site.
- (D) Install and utilize a wheel washing system to remove bulk material from tires and vehicle undercarriages before vehicles exit the site.
- (E) Any other control measures approved by the Executive Officer and the U.S. EPA as equivalent to the methods specified items (A) through (D) above.

Mobile Equipment Emission Control

- MM 5.4-2: To reduce equipment emissions, the following measures shall be implemented when feasible.
 - Use low emission mobile construction equipment. "CARB Certified" heavy construction equipment conforms to the latest off-road CARB emission standards and is the lowest polluting equipment available. The use of this equipment would reduce heavy equipment NO_x emissions by approximately 30 percent and heavy equipment PM10 emissions by approximately 50 percent from the emissions levels shown in Tables 5.4-3 through 5.4-5. This is a substantial reduction but will not reduce emissions to less than the significance thresholds.
 - Maintain construction equipment engines by keeping them tuned.
 - Use low sulfur fuel for stationary construction equipment. This is required by SCAQMD Rules 431.1 and 431.2.
 - Utilize existing power sources (i.e., power poles) when feasible. This measure would minimize the use of higher polluting gas or diesel generators.
 - Use aqueous diesel fuel where feasible and reasonably commercially available.
 - Use cooled exhaust gas recirculation (EGR) where feasible and reasonably commercially available.

Several of the mitigation measures listed above are advanced emission control technologies that are currently not commercially available. For example, aqueous diesel fuel reduces NOx formation by reducing combustion temperatures, resulting in lower NOx emissions. According to the SCAQMD, the current availability of this fuel technology is limited, and it may not be available for use at the landfill. In addition, with EGR diesel engines, a small amount of hot exhaust gas is routed through a cooler and is mixed with fresh air entering the engine. The exhaust gas helps reduce the temperature during combustion, which lowers the formation of thermal NOx. EGR technology is in the development phase, and has not been fully commercialized. To the extent that the advanced emissions-control technologies become reasonably commercially available, or are required by the CARB from grading contractors, then such advanced emissions-control technologies will be used.

Furthermore, a requirement to install diesel particulate filters on construction equipment used at the landfill was considered to further reduce emissions. However, the availability of construction equipment retrofitted with diesel particulate filters is limited. This is a result of operational problems in diesel engines equipped with these filters. Therefore, this potential mitigation measure for construction is considered infeasible.

5.4.5 LEVEL OF SIGNIFICANCE AFTER MITIGATION

The Prima Deshecha Landfill is currently implementing several mitigation measures to reduce potential air quality impacts. Implementation of the mitigation measures described in Section 5.4.4 would help to further reduce air quality impacts that result from operations at the Prima Deshecha Landfill. However, even with implementation of all existing and recommended mitigation measures, operations at the Prima Deshecha Landfill would result in significant and unavoidable air quality impacts.

5.5 BIOLOGICAL RESOURCES

5.5.1 EXISTING CONDITIONS

Vegetation Types

Vegetation mapping for the project site was completed by P&D Consultants in 1999 (County of Orange 2004). Site-wide vegetation surveys were conducted by a qualified Biologist in October 2000 using a 400-scale color aerial photograph, which was later transferred by clear acetate to a 400-scale topographic base and provided the basis for the creation of Exhibit 4.5-1 of EIR 575.

In 2003 and 2005, BonTerra Consulting reviewed, refined, and updated the vegetation mapping data to reflect the current landfill operation and associated limits of disturbance, as shown in Exhibit 5.5-1. The project site has undergone some changes since the certification of EIR 575 including ongoing landfilling operations addressed by this document, the Site B Landslide Remediation Project and associated creation of the Bio-Mitigation program, and the additional headscarp grading expansion of the landslide remediation project within Zone 1.

Grassland, coastal sage scrub, coastal sage scrub-grassland ecotones, chaparral, riparian, woodlands, wet meadows and marshes, and other areas are described in detail. The updates to the existing vegetation types and acreage from January 31, 2001, to present are identified in Table 5.5-1.

As previously indicated, the 2001 vegetative database was originally mapped using large-scale color aerial photography with some limited ground-truthing. BonTerra Consulting updated the vegetation database in 2003 and 2005 to reflect disturbances associated with the ongoing landfill operations and management. In addition, areas identified as containing coastal sage scrub habitat were evaluated in the field by a qualified Biologist as construction occurred to determine if the habitat met the standard under the Orange County Habitat Classification System (HCS) for coastal sage scrub. In some instances where the standard was not met, these areas were re-mapped as ruderal or annual grassland. Some areas containing coastal sage scrub were authorized for removal as part of the Landslide Remediation Project. The vegetation database also includes the habitat enhancement efforts associated with the landslide remediation project. In addition, BonTerra Consulting updated all vegetation mapping for the site in 2005 using HCS and mapping to a scale of 1 inch equals 200 feet.



Note: Areas within Zone 4 designated as Buckwheat Scrub vegetation (2.3.7-R) were hydroseeded for erosion control purposes and are not part of any mitigation program at Prima. For these reasons the impacts associated with project construction do not include these areas.

Biological Resources Prima Deshecha Landfill V S 1,100 550 0 1,100 Feet





Prima Deshecha Landfill Supplemental Environmental Impact Report

) - E	1,100	550	0	1,100
				Feet



R:/Projects/OCIWMD/J004/Ex5.5-2_sos_dist_061506.pdf

TABLE 5.5-1 VEGETATION TYPES AND OTHER AREAS ACREAGE IN 2001, 2003, AND 2005

Vegetation Classification ^a	Vegetation Types and Other Areas	January 31, 2001 Habitat Acres ^b	2003 Updated Habitat Acres	2005 Updated Habitat
Grasslands				
4.1	Annual Grassland	910.10	804.40	783.80
4.2	Elymus Grassland			5.13
4.3	Southern Coastal Needlegrass Grassland	15.00	14.30	21.75
4.6	Ruderal	83.00	27.60	18.43
4.7	Mixed Perennial Grassland			2.41
4.9	Castor Bean/Fennel			2.43
Coastal Sage S	crub			
2.3.2	California Sagebrush-Orange Bush Monkeyflower			68.98
2.3.4	Black Sage Scrub	22.50	21.40	5.10
2.3.6	Sagebrush Scrub	78.90	77.60	23.20
2.3.7-R	Buckwheat Scrub (Revegetation)			3.53
2.3.9	Coyote Brush	58.80	54.70	26.31
2.3.10	Mixed Sage Scrub			25.56
2.3.10-R	Mixed Sage Scrub (Revegetation)			31.91
Coastal Sage S	crub-Grassland Ecotone			•
2.8.1	Sagebrush-Grassland	2.50	2.50	1.91
2.8.2	Buckwheat-Grassland	6.10	4.90	0.67
2.8.5	Mixed Sage Scrub-Grassland	34.20	29.80	56.87
Chaparral				•
3.12	Toyon-Sumac Chaparral	63.80	63.10	71.41
Riparian				
7.1	Riparian Herb	2.10	2.10	2.56
7.2	Southern Willow Scrub	8.60	5.40	3.83
7.3	Mule Fat Scrub	10.50	7.70	1.76
7.4	Southern Sycamore Riparian Woodland	3.30	3.30	4.09
7.12	Mexican Elderberry Riparian Scrub			3.37
Woodland				
8.4	Mexican Elderberry Woodland	2.50	2.40	2.19
Wet Meadows a	ind Marshes			
5.2	Alkali Meadow	0.20	0.20	0.00
6.3	Freshwater & Alkali Marsh ^c	0.40	0.00	3.90
Other Areas				
15.3	Non-Urban Commercial/Industrial/Institutional	16.30	0.40	17.64
15.5	Parks and Ornamental Plantings	10.60	4.30	3.87
16.1	Cleared or Graded	210.90	404.20	303.46
16.2	Other Developed Areas (Erosion Control Plantings)			5.78
	Total	1530.30	1530.30	1530.06

The habitat codes and names represent the Orange County Habitat Classification System. Vegetation types and acreage were obtained from EIR 575 Table 4.5-1. b

с

This vegetation type is not described below because it was removed as part of the landslide remediation project during re-location of the Prima Deshecha Cañada channel. The habitat type is being recreated as part of the Bio-Mitigation program.

Grasslands

The project site is generally characterized by areas of annual grassland, dominated by naturalized annual grasses and forbs. Large portions of the grasslands on the project site are extremely degraded by infestation of large stands of cardoon (artichoke thistle, *Cynara cardunculus*) and black mustard (*Brassica nigra*). Other grassland vegetation types on the project site include patches of southern coastal needlegrass grassland, Elymus grassland, mixed perennial grassland, castor bean/fennel, and ruderal (disturbed) areas.

Annual Grassland (4.1)

The majority of the project site is generally characterized by large areas of annual grassland. This vegetation type consists of naturalized annual grasses and forbs, along with a minor amount of subshrub cover. Common grasses found in this vegetation type included ripgut brome (*Bromus diandrus*), soft chess (*Bromus hordeaceus*), foxtail barley (*Hordeum murinum ssp. leporinum*), foxtail chess (*Bromus madritensis ssp. rubens*), slender wild oat (*Avena barbata*), perennial ryegrass (*Lolium perenne*), wild oat (*Avena fatua*), and little-seed canary grass (*Phalaris minor*).

Many of these grasslands have an open cover or dense patches of black mustard and cardoon. Other characteristic forbs include bur clover (*Medicago polymorpha*), red-stemmed filaree (*Erodium cicutarium*), shortpod mustard (*Hirschfeldia incana*), arroyo lupine (*Lupinus succulentus*), common sow thistle (*Sonchus oleraceus*), tocalote (*Centaurea melitensis*), dove weed (*Eremocarpus setigerus*), narrow-leaved milkweed (*Asclepias fascicularis*), common horseweed (*Conyza canadensis*), tall wreath plant (*Stephanomeria virgata*), coyote melon (*Cucurbita foetidissima*), cheese weed (*Malva parviflora*), Italian thistle (*Carduus pycnocephalus*), common fiddleneck (*Amsinckia menziesii*), Australian saltbush (*Atriplex semibaccata*), prickly lettuce (*Lactuca serriola*), telegraph weed (*Heterotheca grandiflora*), rose clover (*Trifolium hirtum*), fascicled tarweed (*Deiandra fasciculata*), and vinegar weed (*Trichostema lanceolatum*).

Many moist areas in these grasslands contain curly dock (*Rumex crispus*), annual sunflower (*Helianthus annuus*), bristly ox tongue (*Picris echioides*), western ragweed (*Ambrosia psilostachya*), white sweet clover (*Melilotus alba*), horehound (*Marrubium vulgare*), little-seed canary grass, perennial rye grass, soft chess, and cocklebur (*Xanthium strumarium*).

Occasionally, scattered shrub or subshrub cover is found in these grassland vegetation types. Common shrub species included coastal goldenbush (*Isocoma menziesii*), coyote brush (*Baccharis pilularis*), tree tobacco (*Nicotiana glauca*), cudweed aster (*Lessingia filaginifolia*), and coastal prickly pear (*Opuntia littoralis*).

Elymus Grassland (4.2)

The Elymus grassland vegetation type occurs on mesic slopes and is dominated by a nearmonotypic stand of giant wild rye (*Leymus* [=*Elymus*] condensatus). Occasional individuals of Mexican elderberry (*Sambucus mexicana*) and coyote brush (*Baccharis pilularis*) are also present in this vegetation type.

Southern Coastal Needlegrass Grassland (4.3)

A grassland containing at least five percent cover of purple needlegrass (*Nassella pulchra*) is found on the project site. Other annual grasses found in this vegetation type consist of soft chess, ripgut brome, nit grass (*Gastridium ventricosum*), foxtail fescue (*Vulpia myuros*), and slender wild oat.

Common forbs in this vegetation type consist of shortpod mustard, dove weed, blue-eyed grass (*Sisyrinchium bellum*), common fiddleneck, sliver puffs (*Uropappus lindleyi*), coast jepsonia (*Jepsonia parryi*), arroyo lupine, black mustard, coyote melon, and cardoon.

Shrubs and subshrubs occasionally found in this vegetation type consist of coastal goldenbush, cudweed aster, coastal prickly pear, long-stemmed buckwheat (*Eriogonum elongatum*), coyote brush, and California sagebrush (*Artemisia californica*).

Ruderal (4.6)

The ruderal vegetation type includes disturbed areas on the project site, often found near the active landfill or other previously graded localities. Some areas consist of stands of Russian thistle (*Salsola tragus*) associated with Australian saltbush, black mustard, bur clover, bristly ox tongue, tall wreath plant, Crete hedypnois (*Hedypnois cretica*), common horseweed, ripgut brome, crystal ice plant (*Mesembryanthemum crystallinum*), Persian knotweed (*Polygonum argyrocoleon*), suberect saltbush (*Atriplex suberecta*), foxtail barley, tumbling pigweed (*Amaranthus albus*), Bermuda grass (*Cynodon dactylon*), Italian thistle, puncture vine (*Tribulus terrestris*), prickly lettuce, dove weed, common knotweed (*Polygonum arenastrum*), fennel (*Foeniculum vulgare*), soft chess, cheese weed, and horehound.

Other areas consist of dense stands of black mustard or cardoon. These stands were often found with ripgut brome, annual sunflower, bristly ox tongue, bur clover, horehound, rose clover, Australian saltbush, white sweet clover, tree tobacco, common horseweed, Russian thistle, common sow thistle, and perennial rye grass.

Mixed Perennial Grassland (4.7)

The mixed perennial grassland vegetation type is similar to southern coastal needlegrass grassland, but is dominated by a mix of native perennial grass species including California brome (*Bromus carinatus*), needle grasses (*Nassella* spp.), giant wild rye, and bent grass (*Agrostis pallens*).

In addition to the dominant perennial grasses, widely scattered woody shrubs are found in these grasslands including coyote brush, coastal goldenbush, and California sagebrush. Common forb species found in this vegetation type include common fiddleneck, common goldenstar (*Bloomeria crocea*), goose grass (*Galium aparine*), and western verbena (*Verbena lasiostachys*).

Castor Bean/Fennel (4.9)

The castor bean/fennel vegetation type category was created to differentiate large stands of invasive castor bean (*Ricinus communis*) and fennel from other common grassland and ruderal vegetation types present on the project site. Patches of castor bean/fennel vegetation are present in disturbed areas just south of the recycling center on the site. Other species that may be present within the castor bean/fennel vegetation type include non-native grasses such as wild oat (*Avena* sp.) and annual bromes (*Bromus* sp.), and occasional weedy forbs including black mustard.

Coastal Sage Scrub

California Sagebrush-Orange Bush Monkeyflower Scrub (2.3.2)

The California sagebrush-orange bush monkeyflower vegetation type is a mesic expression of coastal sage brush vegetation that is dominated by California sagebrush and orange bush

monkeyflower (*Mimulus aurantiacus*), with abundant coyote brush, Mexican elderberry, and giant wild rye present. Other species that may be present in this vegetation type include coastal goldenbush, black sage (*Salvia mellifera*), white sage (*Salvia apiana*), and poison oak (*Toxicodendron diversilobum*). This vegetation type occurs on mesic slopes in several areas of the project site.

Black Sage Scrub (2.3.4)

Black sage scrub is found on south facing slopes in association with the sumac chaparral. This scrub is dominated by stands of black sage. Other species occurring in this vegetation type include California bush sunflower (*Encelia californica*), interior flat-topped buckwheat (*Eriogonum fasciculatum* var. *foliolosum*), coastal goldenbush, California brickellbush (*Brickellia californica*), coastal prickly pear, and deer weed (*Lotus scoparius*). Laurel sumac (*Malosma* laurina) and lemonadeberry (*Rhus integrifolia*) are often associated with this scrub.

The understory is composed of foxtail chess, shortpod mustard, finger-leaved morning glory (*Calystegia macrostegia* ssp. *arida*), foxtail fescue, bicolored everlasting (*Gnaphalium bicolor*), caterpillar phacelia (*Phacelia cicutaria*), fascicled tarweed, black mustard, slender wild oat, tocalote, red-stemmed filaree, and foothill needlegrass (*Nassella lepida*).

Sagebrush Scrub (2.3.6)

A coastal sage scrub dominated by California sagebrush is found on several localities within the project site. More mesic stands of this vegetation type are associated with orange bush monkeyflower, giant wild rye, poison oak, coyote brush, Mexican elderberry, coastal goldenbush, and lemonadeberry. The understory in the mesic stands of this vegetation type consists of ripgut brome, Italian thistle, black mustard, California everlasting (*Gnaphalium californicum*), nit grass, soft chess, bur clover, common golden star, rigid hedge nettle (*Stachysrigida*), foothill needlegrass, western verbena, slender wild oat, tall wreath plant, and foxtail fescue.

The xeric aspects of this vegetation type are associated with white sage, interior flat-topped buckwheat, black sage, laurel sumac, coastal prickly pear, cudweed aster, coastal goldenbush, orange bush monkeyflower, deerweed, and long-stemmed buckwheat. The understory found in the xeric aspects of this vegetation type consists of foxtail chess, purple needlegrass, soft chess, tall wreath plant, fascicled tarweed, shortpod mustard, common fiddleneck, blue-eyed grass, ripgut brome, red-stemmed filaree, and bicolored everlasting.

Buckwheat Scrub (2.3.7-R)

Buckwheat scrub is characterized by nearly monotypic stands of California buckwheat. Other species may be present in low densities in this vegetation type including black sage, California sagebrush, big saltbush (*Atriplex lentiformis*), and fourwing saltbush (*Atriplex canescens*). Nonnative forbs and grasses are also common in this vegetation type including wild oats (*Avena sp.*), black mustard, and sourclover (*Melilotus indica*). All the buckwheat scrub areas mapped on the project site were coastal sage scrub revegetation sites dominated by California buckwheat.

Coyote Brush Scrub (2.3.9)

A coastal sage scrub dominated by coyote brush is found on the steep sides of drainages and other mesic areas of the project site. This scrub is characterized by coyote brush, along with giant wild rye, California sagebrush, orange bush monkeyflower, Mexican elderberry, coastal goldenbush, poison oak, and tree tobacco.

The understory consists of a variety of grasses and forbs. Common species in this vegetation type include black mustard, ripgut brome, Italian thistle, foxtail fescue, soft chess, western verbena, California everlasting, scarlet pimpernel (*Anagallis arvensis*), common golden star, purple needlegrass, bur clover, sharp-tooth sanicle (*Sanicula arguta*), rattlesnake weed (*Daucus pusillus*), nit grass, and rigid hedge nettle.

Mixed Sage Scrub (2.3.10 and 2.3.10-R)

Mixed sage scrub is dominated by an even mix of four or more species of sage scrub species including black sage, California sage brush, California buckwheat, laurel sumac, California bush sunflower, and deerweed. In more mesic expressions of this vegetation type, coastal goldenbush, white sage, orange bush monkeyflower, and coyote brush may occur. Species found in more xeric stands include coastal prickly pear and wishbone bush (*Mirabilis californica*).

Coastal Sage Scrub-Grassland Ecotone

Sagebrush-Grassland Ecotone (2.8.1)

This ecotonal sage scrub vegetation type consists of open stands of California sagebrush, coastal goldenbush, and coyote brush in the annual grassland. The species composition of the annual grassland is similar to that described above for the annual grassland vegetation type.

Buckwheat-Grassland Ecotone (2.8.2)

This open scrub vegetation type is found on previously disturbed slopes within the active landfill or near adjacent facilities. Interior flat-topped buckwheat is the dominant plant species, which is associated with coastal goldenbush, white sage, four-wing saltbush, black sage, California bush sunflower, big saltbush, and tree tobacco.

The grassland in this ecotonal vegetation type is comprised of black mustard, ripgut brome, rose clover, slender wild oat, Australian saltbush, common sow thistle, foxtail chess, white sweet clover, telegraph weed, shortpod mustard, arroyo lupine, common horseweed, bur clover, tall wreath plant, sweet alyssum (*Lobularia maritima*), tocalote, sea-lavender (*Limonium* sp.), and Russian thistle.

Mixed Sage Scrub-Grassland Ecotone (2.8.5)

Mixed sage scrub-grassland ecotone is a common vegetation type on the project site and consists of small-to-large stands of coyote brush in the grassland. Other shrubs found scattered in this vegetation type consist of giant wild rye, coastal goldenbush, and California sagebrush.

The grassland component of this vegetation type is generally composed of ripgut brome, soft chess, black mustard, curly dock, Italian thistle, dove weed, western ragweed, cardoon, shortpod mustard, bristly ox tongue, bur clover, white sweet clover, and narrow-leaved milkweed.

Chaparral

Toyon-Sumac Chaparral (3.1.2)

Several of the mesic slopes above the steep canyons contain chaparral vegetation characterized by stands of lemonadeberry and laurel sumac. This vegetation type often contains large areas of coastal sage scrub between the large evergreen sumac shrubs. Other

characteristic shrubs found in this chaparral included black sage, interior flat-topped buckwheat, coastal prickly pear, California sagebrush, coastal goldenbush, California brickellbush, orange bush monkeyflower, deerweed, giant wild rye, and coyote brush.

The understory in this chaparral consists of foxtail chess, shortpod mustard, ripgut brome, caterpillar phacelia, black mustard, bicolored lupine, foothill needlegrass, finger-leaved morning glory, tall wreath plant, tocalote, and white everlasting (*Gnaphalium canescens*).

Riparian

Riparian Herb (7.1)

Several of the tributaries to Prima Deshecha Cañada, some of the retention basins, and a former cattle pond in the northeast portion of the site contain herbaceous species commonly found in moist vegetation types. Common species in these vegetation types include western ragweed, cocklebur, common celery (*Apium graveolens*), fennel, Mexican tea (*Chenopodium ambrosioides*), narrow-leaved cattail (*Typha angustifolia*), bristly ox tongue, curly dock, Bermuda grass, ripgut brome, soft chess, annual beard grass (*Polypogon monspeliensis*), Johnsongrass (*Sorghum halepense*), common plantain (*Plantago major*), white sweet clover, saltgrass (*Distichlis spicata*), Persian knotweed (*Polygonum argyrocoleon*), Mexican sprangletop (*Leptochloa uninervia*), common horseweed, dwarf nettle (*Urtica urens*), and tree tobacco.

Southern Willow Scrub (7.2 and 7.2-R)

Southern willow scrub is found in Prima Deshecha Cañada and generally consists of dense stands of arroyo willow (*Salix lasiolepis*) in the drainage. Other willows occasionally found in this vegetation type include red willow (*Salix laevigata*) and black willow (*Salix gooddingii*). Other species in this scrub include mule fat (*Baccharis salicifolia*), coyote brush, giant wild rye, and California sagebrush.

The understory in this dense scrub is comprised of western ragweed, saltgrass, common plantain, ripgut brome, cocklebur, narrow-leaved cattail, bristly ox tongue, pampas grass (*Cortaderia sellonana*), soft chess, common celery, white sweet clover, annual beard grass, alkali bulrush (*Scirpus maritimus*), yerba mansa (*Anemopsis californica*), Bermuda grass, nettle (*Urtica dioica*), rigid hedge-nettle, mugwort (*Artemisia douglasiana*), and curly dock.

Mule Fat Scrub (7.3)

Several of the feeder drainages and portions of Prima Deshecha Cañada contain a riparian scrub dominated by mule fat. Other, less common shrubs include coyote brush, arroyo willow, black willow, coastal goldenbush, and giant wild rye. Uncommon weedy shrubs in this vegetation type consist of tree tobacco, Mediterranean tamarisk (*Tamarix ramosissima*), and giant reed (*Arundo donax*).

The understory is often comprised of dense stands of western ragweed along with ripgut brome, soft chess, cocklebur, alkali heliotrope (*Heliotropium curassavicum*), mugwort, western verbena, bristly ox tongue, white sweet clover, curly dock, Bermuda grass, Italian thistle, milk thistle (*Silybum marianum*), nettle-leaved goosefoot (*Chenopodium murale*), annual beard grass, Mexican tea, common horseweed, common celery, saltgrass, fennel, Persian knotweed, and common plantain.

Southern Sycamore Riparian Woodland (7.4)

This vegetation type is found in the upper drainage of Prima Deshecha Cañada. It consists of an overstory of western sycamore (*Platanus racemosa*), along with Mexican elderberry. A lower shrub layer is comprised of mule fat, lemonadeberry, laurel sumac, coyote brush, giant wild rye, poison oak, California wild rose (*Rosa californica*), and coastal prickly pear.

The understory is generally characterized by dense stands of ripgut brome, western ragweed, Italian thistle, black mustard, western verbena, poison hemlock (*Conium maculatum*), coyote melon, rattlesnake weed, Douglas' nightshade (*Solanum douglasii*), and wild blackberry (*Rubus ursinus*).

Species found in or adjacent to the channel include water cress (*Rorippa nasturtium aquaticum*), nettle, saltgrass, common celery, common plantain, alkali bulrush, ripgut brome, Italian thistle, western ragweed, cocklebur, and bristly ox tongue.

Mexican Elderberry Riparian Scrub (7.12)

Mexican elderberry riparian scrub is similar to the Mexican elderberry woodland vegetation type found on the project site; however, this riparian scrub vegetation type is restricted to drainages and contains understory species typically associated with riparian habitats. Common understory species found in the Mexican elderberry riparian scrub vegetation type include scattered coyote brush and occasional Emory's baccharis (*Baccharis emoryi*). Dense stands of nettle (*Urtica dioica*) and giant wild rye are also common in this vegetation type, along with scattered western verbena and western ragweed.

Woodland

Mexican Elderberry Woodland (8.4)

Small stands of Mexican elderberry are found in isolated areas on the project site. These small islands are composed of Mexican elderberry associated with lemonadeberry, tree tobacco, giant wild rye, coyote brush, California brickellbush, coastal goldenbush, California sagebrush, coastal prickly pear, and laurel sumac.

Wet Meadows and Marshes

Alkali Meadow (5.2)

Alkali meadows have been historically found at a few localities along Prima Deshecha Cañada and on some of the tributaries to this channel. This vegetation type consists of dense mats of saltgrass. Other species found in this vegetation type include rabbit's foot grass, western ragweed, cocklebur, bristly ox tongue, soft chess, pitseed goosefoot (*Chenopodium berlandieri*), white sweet clover, alkali bulrush, and common celery.

None of the alkali meadow vegetation type areas were found during the 2005 field mapping surveys. Many of the areas previously mapped as alkali meadow were found to support mule fat or coyote brush scrub.

Other Areas

Non-Urban Commercial/Industrial/Institutional (15.3)

The various facility buildings for the landfill operation were mapped as non-urban institutional.

Parks and Ornamental Plantings (15.5)

There is fairly large area of eucalyptus woodland in the southwestern corner of the project site, which is composed of river red gum trees (*Eucalyptus camadulensis*) and other species of eucalyptus (*Eucalyptus* spp.). Other ornamental plantings are found near the landfill entrance and around the existing facility buildings. These include plants of Peruvian pepper (*Schinus molle*), eucalyptus, golden wattle (*Acacia longifolia*), wattle (*Acacia spp.*), fan palm (*Chamaeropis fortunei*), flowering plum (*Prunus sp.*), Japanese yew (*Podocarpus macrophyllus*), myoporum (*Myoporum laetum*), oleander (*Nerium oleander*), melaleuca (*Melaleuca sp.*), hottentot fig (*Carpobrotus edulis*), floss silk tree (*Chorisia speciosa*), western sycamore, Fraser's photinia (*Photinia fraseri*), African daisy (*Osteospernum ecklonis*), Spanish bayonet (*Yucca alifolia*), pittosporum (*Pittosporum tobria*), and several species of pines (*Pinus sp.*).

Cleared or Graded (16.1)

The actively graded areas within the landfill, the landslide areas, and other localities within the project site that have been cleared of any vegetation are placed in this mapping unit.

Other Developed Areas (Erosion-Control Plantings) (16.2)

The erosion-control landscaping mapping unit was created to describe several cut-slope areas that appear to have been treated with an erosion-control seed mix that contains both native and non-native species. The dominant species in these areas are brittlebush (*Encelia farinosa*) and rose clover. Occasional phacelia (*Phacelia* sp.) and deerweed are also present on these slopes. This vegetation type is found on manufactured slopes immediately adjacent to the active landfill area.

<u>Wildlife</u>

Fish

Given the hydrologic and geologic conditions of the Prima Deshecha Cañada stream channel, this drainage course and its associated tributaries are considered ephemeral and support a very narrow non-contiguous riparian corridor containing southern willow scrub, mule fat scrub, and freshwater marsh. No fish species have been observed during previous surveys or during any subsequent field investigations. The only exception is in Basin B described below (Exhibit 2.2-7).

The Basin B component of the Bio-mitigation Program is located at the terminus of Prima Deshecha Cañada channel and was designed to capture and retain storm flows and intermittent flows to create the alluvial groundwater necessary for establishing a southern willow scrub and alkali and freshwater marsh habitats. As a result, Basin B may be inundated or ponded for up to four months each year depending on seasonal rainfall. As such, Orange County Vector Control (COVC) releases mosquito fish into the basin to control potential mosquito populations.

Amphibians

Amphibians require moisture for at least a portion of their life cycle, and many require standing or flowing water for reproduction. Terrestrial species may or may not require standing water for reproduction. These species are able to survive in dry areas by aestivating (i.e., remaining beneath the soil in burrows or under logs and leaf litter, emerging only when temperatures are low and humidity is high). Many of these species' habitats are associated with water and they emerge to breed once the rainy season begins. Soil moisture conditions can remain high throughout the year in some habitat types, depending on factors such as the amount of vegetation cover, elevation, and slope aspect.

Common native species observed and/or expected to occur within the project area include the garden slender salamander (*Batrachoseps major*), Pacific tree frog (*Pseudacris regilla*), and western toad (*Bufa boreas*).

Reptiles

Reptile diversity and abundance typically varies with vegetation type and character. Many species prefer only one or two vegetation types; however, most will forage in a variety of habitats. Most species occurring in open areas use rodent burrows for cover, protection from predators, and refuge during extreme weather conditions.

A variety of lizards and snakes have been observed during surveys on the project site. The lizards include side-blotched lizard (*Uta stansburiana*), western fence lizard (*Sceloporus occidentalis*), western whiptail (*Aspidoscelis tigris*), western skink (*Eumeces* skiltonianus), and southern alligator lizard (*Elgaria multicarinata*). Snakes observed during the surveys include ringneck snake (*Diadophis punctatus*), common kingsnake (*Lampropeltis getulus*), racer (*Coluber constrictor*), coachwhip (*Masticophis flagellum*), striped racer (*Masticophis lateralis*), gopher snake (*Pituophis catenifer*), western rattlesnake (*Crotalus viridis*), and red diamond rattlesnake (*Crotalus ruber*).

Birds

A variety of bird species reside in habitats or utilize portions of the project area throughout the year. Other species are present only during certain seasons. For example, thousands of gulls visit the project area during the winter season in an attempt to forage within the refuse disposal area, but migrate out of the region during the summer breeding season. The list of bird species that have been observed or have the potential to occur on site has been updated from those provided in EIR 575 based on regular site visits to the project site from 2000 to present by Brian Daniels, BonTerra Consulting Senior Biologist.

Resident bird species that occur in the project area include the turkey vulture (*Cathartes aura*), mallard (*Anas platyrhynchos*), red-tailed hawk (*Buteo jamaicensis*), American kestrel (*Falco sparverius*), California quail (*Callipepla californica*), killdeer (*Charadrius vociferous*), mourning dove (*Zenaida macroura*), greater roadrunner (*Geococcyx californianus*), barn owl (*Tyto alba*), great horned owl (*Bubo virginianus*), Anna's hummingbird (*Calypte anna*), Nuttall's woodpecker (*Picoides nuttallii*), Say's phoebe (*Sayornis saya*), black phoebe (*Sayornis nigricans*), western scrub-jay (*Aphelocoma californica*), American crow (*Corvus brachyrhynchos*), common raven (*Corvus corax*), bushtit (*Psaltriparus minimus*), Bewick's wren (*Thryomanes bewickii*), house wren (*Troglodyes aedon*), wrentit (*Chamaea fasciata*), northern mockingbird (*Mimus polyglottos*), California thrasher (*Toxostoma redivivum*), common yellowthroat (*Geothlypis trichas*), spotted towhee (*Pipilo maculates*), California towhee (*Pipilo crissalis*), rufous-crowned sparrow (*Aimophila ruficeps*), song sparrow (*Melospiza melodia*), red-winged blackbird (*Agelaius phoeniceus*), western meadowlark (*Sturnella neglecta*), Brewer's blackbird (*Euphagus cyanocephalus*), house finch (*Carquelis tristis*).

Migratory bird species that breed during the spring and summer in the project area include Costa's hummingbird (*Calypte costae*), black-chinned hummingbird (*Archilochus alexandri*), ash-throated flycatcher (*Myiarchus cinerascens*), cliff swallow (*Petrochelidon pyrrhonota*), yellow-breasted chat (*Icteria virens*), grasshopper sparrow (*Ammodramus savannarum*), black-

headed grosbeak (*Pheucticus melanocephalus*), blue grosbeak (*Guiraca caerulea*), hooded oriole (*Icterus cucullatus*), and Bullock's oriole (*Icterus bullockii*).

Wintering bird species that occur in the project area include western gull (*Larus occidentalis*), glaucous-winged gull (*Larus glaucescens*), herring gull (*Larus argentatus*), California gull (*Larus californicus*), northern flicker (*Colaptes auratus*), ruby-crowned kinglet (*Regulus calendula*), blue-gray gnatcatcher (*Polioptila caerulea*), hermit thrush (*Catharus guttatus*), American pipit (*Anthus rubescens*), cedar waxwing (*Bombycilla cedrorum*), yellow-rumped warbler (*Dendroica coronata*), savannah sparrow (*Passerculus sandwichensis*), vesper sparrow (*Pooecetes gramineus*), Lincoln's sparrow (*Melospiza lincolnii*), white-crowned sparrow (*Zonotrichia leucophrys*), golden-crowned sparrow (*Zonotrichia atricapilla*), and dark-eyed junco (*Junco hyemalis*).

Migrant bird species that occur in the project area during spring and fall migration include the spotted sandpiper (*Actitis macularia*), greater yellowlegs (*Tringa* melanoleuca), rufous hummingbird (*Selasphorus rufus*), western wood-pewee (*Contopus sordidulus*), western kingbird (*Tyrannus verticalis*), Cassin's vireo (*Vireo cassinii*), warbling vireo (*Vireo gilvus*), marsh wren (*Cistothorus palustris*), Nashville warbler (*Vermivora ruficapilla*), black-throated gray warbler (*Dendroica nigrescens*), Townsend's warbler (*Dendroica townsendii*), MacGillivray's warbler (*Oporornis tolmiei*), Wilson's warbler (*Wilsonia pusilla*), chipping sparrow (*Spizella passerina*), and lazuli bunting (*Passerina amoena*).

Mammals

A variety of common small mammal species are expected to occur on the project site and are expected to include most of the following species: ornate shrew (*Sorex ornatus*), broad-footed mole (*Scapanus latimanus*), desert cottontail (*Sylvilagus audubonii*), California ground squirrel (*Spermophilus beecheyi*), Botta's pocket gopher (*Thomomys bottae*), San Diego pocket mouse (*Chaetodipus fallax*), California vole (*Microtus californicus*), house mouse (*Mus musculus*), dusky-footed woodrat (*Neotoma fuscipes*), California mouse (*Peromyscus californicus*), cactus mouse (*Peromyscus eremicus*), deer mouse (*Peromyscus maniculatus*), black rat (*Rattus rattus*), and western harvest mouse (*Reithrodonomys megalotis*).

Common bat species expected to occur on the project site include the big brown bat (*Eptesicus fuscus*), hoary bat (*Lasiurus cinereus*), California myotis (*Myotis californicus*), yuma myotis (*Myotis yumanensis*), western pipistrelle (*Pipistrellus hesperus*), and Brazilian free-tailed bat (*Tadarida brasiliensis*).

Larger mammal species known to occur or expected to occur on the project site include the Virginia opossum (*Didalphis virginiana*), coyote (*Canis latrans*), common raccoon (*Procyon lotor*), striped skunk (*Mephitis mephitis*), long-tailed weasel (*Mustela frenata*), mountain lion (*Felis concolor*), bobcat (*Lynx rufus*), and mule deer (*Odocoileus hemionus*).

Wildlife Movement

A fundamental concept and central tenet of conservation biology theory is that habitat fragmentation and isolation leads to extinction of local populations as a result of two processes: (1) reduction in total habitat area which reduces effective population sizes and (2) insularization of local populations which affects dispersal and immigration rates (Wilcox and Murphy 1985; Wilcove et al. 1986). Wilcox and Murphy further point out that immigration may be impeded by conversion of natural habitat between occupied or potential habitat patches, thus increasing the probability of extinction. It is this latter point that is the crux of the habitat linkage problem. That is, isolation of habitat patches accompanied by intervening inhospitable land cover (e.g., urban development, roadways) is thought to increase the probability of permanent extinction of local

populations. Because of complex community-level interactions (e.g., mutualistic species, habitat guilds, keystone species), the loss of one or a few species from a habitat patch as a direct result of habitat fragmentation (primary extinctions) also may result in multiple "secondary" extinctions within the habitat patch (Wilcox and Murphy 1985).

Wildlife movement activities usually fall into one of three movement categories: (1) dispersal (e.g., juvenile animals from natal areas or individuals extending range distributions); (2) seasonal migration; and (3) movements related to home range activities (foraging for food or water, defending territories, searching for mates, breeding areas, or cover). A number of terms have been used in various wildlife movement studies (such as "wildlife corridor," "travel route," "habitat linkage," and "wildlife crossing") to refer to areas in which wildlife move from one area to another. To clarify the meaning of these terms and to facilitate the discussion on wildlife movement in this analysis, these terms are defined as follows:

Travel route: A landscape feature (such as a ridgeline, drainage, canyon, or riparian strip) within a larger natural habitat area that is used frequently by animals to facilitate movement and to provide access to necessary resources (e.g., water, food, cover, den sites). The travel route is generally preferred because it provides the least amount of topographic resistance in moving from one area to another. It contains adequate food, water, and/or cover while moving between habitat areas and provides a relatively direct link between target habitat areas.

Wildlife corridors: A piece of habitat, usually linear in nature, that connects two or more large blocks of habitat that would otherwise be fragmented or isolated from one another. A corridor generally contains suitable cover, food, and/or water to support species and facilitate movement while in the corridor. Urban areas or other areas unsuitable for wildlife usually bound wildlife corridors. Larger, landscape-level corridors (often referred to as "habitat or landscape linkages") can provide both transitory and resident habitat for a variety of species. Wildlife corridors may also contain "choke-points" (e.g., hourglass or funnel shapes) that are most often man-made structures such as culverts and flood control channels that wildlife quickly move through.

Wildlife crossing: A small, narrow area, relatively short in length and generally constricted in nature, that allows wildlife to pass under or through an obstacle or barrier that otherwise hinders or prevents movement. Crossings typically are man-made and include culverts, underpasses, drainage pipes, and tunnels to provide access across or under roads, highways, pipelines, or other physical obstacles. These often represent "choke points" along a wildlife corridor.

Habitat linkage: Habitat or landscape linkages are areas of natural habitat that function to join two larger blocks of habitat. They serve as connections between habitat blocks and help reduce the adverse effects of habitat fragmentation by providing a potential route for gene flow and long-term dispersal. Habitat linkages may serve both as "live-in" habitat and avenues of gene flow for small animals such as reptiles, amphibians, and rodents. Habitat linkages also provide for the transit of larger species, but as opposed to wildlife corridors (as defined above) they also may be "live-in" habitat for larger species (i.e., support breeding sites, frequent use areas). Habitat linkages also may be represented by continuous habitat or by closely spaced habitat "islands" that function as stepping stones for dispersal and movement (especially for birds and flying insects).

Habitat linkages and wildlife corridors facilitate the dispersal by smaller, less mobile species and frequent movement (e.g., daily, weekly) by large mammal species such as mountain lion, mule deer, coyote, and bobcat. The identified species only highlight a much broader suite of species served by the habitat linkages and corridors. Accordingly, the species identified should not be

interpreted as the only species that benefit from the linkages and corridors. It can be reasonably assumed that linkages and corridors that function for large mammals also function for many other species.

The project site is located within the upper Prima Deshecha Cañada watershed that borders a mix of urbanized and open space areas. Urbanized areas border the project site to the west and south (San Clemente) and to the north (San Juan Capistrano). New development areas are to the north (Whispering Hills) and south (Talega) of the project site. Protected open space areas close to the project site include Forster Ranch to the northwest and the Talega Nature Preserve to the east. Although there have been disturbances due to landfilling operations within Zone 1 of the landfill facility, there remains a substantial amount of open space within the project site. These open space areas generally continue to facilitate wildlife movement in multiple directions and provide "live-in habitat" for many species. As such, the project site serves as an important habitat linkage between the open spaces of Forster Ranch and the Talega Nature Preserve.

Special Status Species – Federal- and State-Listed Species

Special Status Plant Species

Special status plant species include those species listed by the state or federal governments as Endangered, Threatened or Rare and species which are candidates for future listing. It also encompasses the species determined by the CNDDB to meet the CEQA (Section 15380) criteria as "rare and endangered," even though they have not been officially listed by any agency (CDFG 2004b). Finally, the list considers those species noted by the California Native Plant Society (CNPS 2001), considered within the Natural Communities Conservation Plan guidelines for coastal sage scrub (CDFG 1992), considered by the County of Orange as "rare or endangered," or considered to be a species of special interest that requires consideration in CEQA or planning studies in the region (Bramlet 2003, Gray and Bramlet 1994a).

To determine the known localities of plant species of special interest in the study region, the CNDDB (CDFG 2002b) element reports were developed for the San Clemente, Dana Point, San Juan Capistrano and Cañada Gobernadora USGS 7.5' quadrangles. To supplement this information, recent EIRs and/or biological surveys in the region were also examined (Bramlet 1998; BonTerra Consulting 2000a, 2000b; MBA 1998; P&D 2002).

Plant Species of Special Interest

Table 5.5-2 presents 35 special status plant species potentially occurring on the project site.

TABLE 5.5-2 SPECIAL STATUS PLANT SPECIES POTENTIALLY OCCURRING ON THE PROJECT SITE

Creation	Federal/	CNPS/	Commente
Species	State	Other	Comments
Coulter's saltbush Atriplex coulteri		CNPS 1B	Potential to occur on site. Not observed during previous surveys. Occurs in alkaline habitats in valley grassland and coastal sage scrub vegetation types.
South Coast saltscale Atriplex pacifica		CNPS 1B	Not expected to occur on site. Occurs on sea bluffs in coastal sage scrub vegetation types.
Parish's brittlescale Atriplex parishii		CNPS 1B	Not expected to occur on site. Occurs on alkali flats and in valley grassland vegetation types.
thread-leaved brodiaea Brodiaea filifolia	FT SE	CNPS 1B	Observed on site. Occurs in native and annual grassland vegetation types. The population on the project site consists of seven individuals and is located within the Phase C3 area (2005).
Catalina mariposa lily Calochortus catalinae		CNPS 4	Potential to occur on site. Not observed during previous surveys. Found in heavy soils in grassland and chaparral vegetation types.
intermediate mariposa lily Calochortus weedii var. intermedius		CNPS 1B	Potential to occur on site. Not observed during previous surveys. Occurs in coastal sage scrub, chaparral, and grassland habitats.
Santa Barbara morning-glory Calystegia sepium ssp. binghamiae		CNPS 1A	Not expected to occur on site. Occurs in coastal salt marsh vegetation types.
Buck's jewel-flower Caulanthus heterophyllus var. pseudosimulans		LC	Potential to occur on site. Not observed during previous surveys. Occurs in open scrub and chaparral vegetation types on granitic substrates.
southern tarplant Centromadia [Hemizonia] parryi ssp. australis		CNPS 1B	Potential to occur on site. Not observed during previous surveys. Occurs in moist, saline habitats in grassland vegetation types.
San Diego mountain mahogany Cercocarpus minutiflorus		LC	Not expected to occur on site. Occurs in chaparral and southern maritime chaparral vegetation types.
Orcutt's pincushion Chaenactis glabriuscula var. orcuttiana		CNPS 1B	Not expected to occur on site. Occurs on coastal dunes in coastal sage scrub vegetation types.
summer holly Comarostaphylis diversifolia ssp. diversifolia		CNPS 1B	Not expected to occur on site. Occurs on dry coastal slopes in chaparral vegetation types.
small-flowered morning-glory Convolvulus simulans		CNPS 4	Observed on site in grassland vegetation in the southwestern portion of the project site.
paniculate tarplant Deinandra [Hemizonia] paniculata		CNPS 4	Observed on site. Occurs in grassland vegetation types in the eastern and northwestern portions of the project site.
western dichondra Dichondra occidentalis		CNPS 4	Potential to occur onsite. Not observed during previous surveys. Occurs in the understory of coastal sage scrub and in needlegrass grassland.
Blochman's dudleya Dudleya blochmaniae ssp. blochmaniae		CNPS 1B	Not expected to occur on site. Occurs on dry, clay substrates in coastal sage scrub vegetation types directly on the coast.
many-stemmed dudleya Dudleya multicaulis		CNPS 1B	Potential to occur on site. Not observed during previous surveys. Occurs on dry, clay substrates in coastal sage scrub and chaparral vegetation types.
Laguna Beach dudleya Dudleya stolonifera	FT ST	CNPS 1B	Not expected to occur on site. Occurs on cliffs in coastal sage scrub vegetation types near Laguna Beach.
sticky dudleya Dudleya viscida		CNPS 1B	Not expected to occur on site. Not observed during previous surveys. Occurs on rocky cliffs in coastal sage scrub vegetation types.
Pendleton button-celery Eryngium pendletonensis		CNPS 1B	Not expected to occur on site. Narrow endemic occurring in clay, vernally mesic coastal bluff scrub, valley and foothill grassland, and vernal pool vegetation types.

TABLE 5.5-2 (Continued) SPECIAL STATUS PLANT SPECIES POTENTIALLY OCCURRING ON THE **PROJECT SITE**

Species	Federal/ State	CNPS/ Other	Comments
cliff spurge Euphorbia misera		CNPS 2	Not expected to occur on site. Occurs on sea bluffs in coastal sage scrub vegetation types.
Palmer's grapplinghook Harpagonella palmeri		CNPS 4	Potential to occur on site. Not observed during previous surveys. Occurs on dry slopes and mesas in chaparral vegetation types.
vernal barley Hordeum intercedens		CNPS 3	Observed on site. Occurs in alkali/saline conditions in vernal pool and grassland vegetation types
Robinson's peppergrass Lepidium virginicum var. robinsonii		CNPS 1B	Potential to occur on site. Not observed during previous surveys. Occurs in coastal sage scrub and chaparral vegetation types.
small-flowered microseris Microseris douglasii var. platycarpha		CNPS 4	Observed on site. Occurs on clay soils in grassland and alkali meadow vegetation types.
prostrate navarretia Navarretia prostrata		CNPS 1B	Not expected to occur on site. Occurs in vernal pool and alkali floodplain habitats.
chaparral nolina Nolina cismontane		CNPS 1B	Not expected to occur on site. Associated with sandstone or gabbro soils in coastal sage scrub and chaparral vegetation types.
golden-rayed pentachaeta Pentachaeta aurea		CNPS 4	Potential to occur on site. Not observed during previous surveys. Occurs in dry, open places in grassland, coastal sage scrub, and yellow pine forest vegetation types.
Nuttall's scrub oak Quercus dumosa		CNPS 1B	Not expected to occur on site. Associated with sandstone substrates in chaparral and foothill woodland vegetation types near the coast.
Coulter's matilija poppy Romneya coulteri		CNPS 4	Potential to occur on site. Not observed during previous surveys. Occurs in dry washes and canyons in chaparral and coastal sage scrub vegetation types.
San Miguel savory Satureja chandleri		CNPS 1B	Not expected to occur on site. Associated with rocky, gabbroic, or metavolcanic soils in chaparral, woodland, coastal scrub, riparian woodland, and grassland vegetation types.
rayless ragwort Senecio aphanactis		CNPS 2	Potential to occur on site. Not observed during previous surveys. Occurs in dry, open places in coastal sage scrub and chaparral vegetation types.
Salt Spring checkerbloom Sidalcea neomexicana		CNPS 2	Potential to occur on site. Not observed during previous surveys. Occurs in alkali habitats and in mesic areas in coastal sage scrub, chaparral, lower montane coniferous forest, Mojavean desert scrub, and playas.
Parry's tetracoccus Tetracoccus dioicus		CNPS 1B	Not expected to occur on site. Occurs on dry, stony slopes in chaparral vegetation types.
crownbeard Verbesina dissita	FT ST	CNPS 1B	Not expected to occur on site. Occurs on coastal slopes in maritime chaparral and coastal scrub vegetation types.

Federal DesignationsFE= Listed by the Federal government as an endangered species.

= Listed by the Federal government as a threatened species.

 State Designations

 SE
 = Listed by the State of California as an endangered species.

ST = Listed by the State of California as a threatened species.

<u>California Native Plant Society (CNPS)</u> CNPS 1A = Plants presumed extinct in California.

- CNPS 1B = Plants considered rare, threatened or endangered in California and elsewhere.
- CNPS 2 = Plants rare, threatened or endangered in California but more common elsewhere.
- CNPS 3 = Plants about which we need more information A review list.
- CNPS 4 = Plants of limited distribution A watch list.
- <u>Other</u>

FΤ

- LR = Locally rare species.
- LC = Species of local concern.

Coulter's Saltbush (*Atriplex coulteri*)

Coulter's saltbush is a CNPS List 1B species. This perennial herb typically blooms between March and October (CNPS 2001). It is found in somewhat alkaline low places in valley grassland and coastal sage scrub vegetation types (Munz 1974). It occurs in open sites in coastal shrubland from sea level to approximately 160 feet above mean sea level (msl) (Hickman 1993). This species is known from Los Angeles, Orange, Riverside, Santa Barbara, San Bernardino, and San Diego counties, and from the Channel Islands and Baja California (CNPS 2001). In Orange County this species has been recorded in Laguna Beach, Pelican Hill, Signal Hill, Newport Beach, Trabuco Canyon, and Cristianitos Canyon. This species was documented from the Whispering Hills project site (Bramlet 1998), and from Cristianitos Canyon (MBA 1998). Coulter's saltbush has the potential to occur in the project site but was not observed during previous surveys.

South Coast Saltscale (*Atriplex pacifica*)

South Coast saltscale is a CNPS List 1B species. This annual herb typically blooms between March and October (CNPS 2001). It is found largely on sea bluffs in coastal sage scrub vegetation types (Munz 1974). It occurs from sea level to approximately 330 feet above msl (Hickman 1993). This species is known from Los Angeles, Orange, Riverside, San Diego, and Ventura counties, and from the Channel Islands and Baja California (CNPS 2001). This species is known to occur at San Clemente State Beach in the project vicinity (CDFG 2001). South Coast saltscale is not expected to occur in the project site.

Parish's Brittlescale (Atriplex parishii)

Parish's brittlescale is a CNPS List 1B species. This annual herb typically blooms between June and October (CNPS 2001). It is found in alkali flats and valley grassland vegetation types, from cismontane southern California to the desert edges and to the Central Valley (Munz 1974). It occurs from sea level to approximately 6,230 feet above msl (Hickman 1993). This species is historically known from Los Angeles, Orange, Riverside, San Bernardino, and San Diego counties, and from Baja California (CNPS 2001). It is considered extirpated in Los Angeles, Orange, San Bernardino, and San Diego counties; this species has only been collected once in California since 1974, but is believed to be extant in Baja California (CNPS 2004). Parish's brittlescale is not expected to occur on the project site.

Thread-leaved Brodiaea (Brodiaea filifolia)

Thread-leaved brodiaea is a federally listed Threatened species, a state-listed Endangered species, and a CNPS List 1B species. This bulbiferous perennial herb typically blooms between March and June (CNPS 2001). It is found in grassland and vernal pool habitats from approximately 200 to 980 feet above msl (Hickman 1993). This species typically occurs on clay soils in vernal pools, coastal scrub, chaparral openings, cismontane woodlands, playas, and valley and foothill grasslands (CNPS 2001). This species is known to occur in San Diego, Riverside, Los Angeles, San Bernardino, and Orange counties (CNPS 2001).

This species was only recently located in Orange County and is currently known to occur in Aliso Canyon, Casper's Regional Park, Talega Reserve, Segunda Deshecha, Forster Ranch, Camp Pendleton, and Cristianitos Canyon (CDFG 2001). The Forster Ranch population is the most significant population, containing over 4,000 plants; this site is located approximately 3.8 miles southwest of the project site. Other populations in Segunda Deshecha are found some 2.3 miles southeast of the project site.

Surveys along the foothill corridor alignment have located approximately eight localities of this species extending from north of the Ortega Highway in Cañada Gobernadora, to sites west of Cristianitos road extending from areas just south of the Ortega Highway to areas near the existing clay pits. Localities were also found in south Cristianitos Canyon on the west side of Cristianitos road and just south of the reserve boundaries. Finally, there is another group of localities in the lower portion of Cristianitos Canyon, just south of Pico. A total of 4,400 plants have been recorded for all of these localities.

During recent surveys at the Prima Deshecha landfill, a small locality (9 plants) of this species was found approximately 2,200 feet southwest of the landfill entrance, and south of the ridgeline access road (Bramlet 2004). The plants were found in a disturbed annual grassland that was dominated by artichoke thistle (*Cynara cardunculus*) on Alo clay soils on project site.

On December 8, 2004, the USFWS proposed to designate 9,403 acres of land as critical habitat for the federally Threatened thread-leaved brodiaea. The proposed areas include land in Los Angeles, San Bernardino, Orange, and San Diego counties. There are 23 known occurrences in Orange County, 13 of which are being proposed for critical habitat.

The proposed Critical Habitat Designation Unit 4d is located within the property boundaries of the Prima Deshecha Landfill. Proposed Critical Habitat Designation Unit 4d is described in the proposed Rule, on page 71292 as consisting of 119 acres of privately owned land northeast of San Clemente in western Orange County. However, the entire 119 acres of Unit 4d falls within the boundaries of the Prima Deshecha Landfill, which is publicly owned land.

As previously indicated, the Prima Deshecha Landfill has an approved General Development Plan for all future development phases of the landfill operation. Based on the Prima Deshecha Landfill remaining airspace capacity, the landfill is not anticipated to close until the year 2067.

As part of the amendment process for the 2001 General Development Plan focused biological surveys have been conducted on the entire Prima Deshecha property in order to determine the presence of the thread-leaved brodiaea. These focused biological surveys have determined that the only area of the site where the thread-leaved brodiaea is known to exist is less than 1 acre and is located within future landfilling area Zone 1, Phase C3. Seven thread-leaved brodiaea plants have been documented in this area based on surveys conducted in 2005 (BonTerra Consulting 2006c). This isolated brodiaea population is located in an area of the landfill site that contains invasive plant species such as artichoke thistle and black mustard that will affect the long-term viability of this small, isolated group of plants.

It is anticipated that landfill operations will begin to enter the Zone 1, Phase C3 area (Exhibit 2.2-3) within the next two years. Development of the landfill within this area will result in disturbance to the thread-leaved brodiaea. Therefore, the IWMD will work with the USFWS and the CDFG to develop a Thread-leaved Brodiaea Pre-mitigation Program for the site which may include transplanting of individual plants and soils to an area within the PDL outside any existing or future landfilling operations. The plan would also include seed collection, nursery propagation, and long-term monitoring.

In addition, the IWMD has submitted a letter to the USFWS requesting that the critical habitat designation for the thread-leaved brodiaea within the Prima Deshecha Landfill property boundary be eliminated.

Catalina Mariposa Lily (*Calochortus catalinae*)

Catalina mariposa lily is a CNPS List 4 species due to the decline of known populations throughout southern California and the continued loss of habitat for this species. This

bulbiferous perennial herb typically blooms between February and May (CNPS 2001). It prefers heavy soil on open grassy slopes and openings in brush, in valley grassland, and in chaparral vegetation types (Munz 1974). It is found from sea level to approximately 2,300 feet above msl (Hickman 1993). This species is known from Los Angeles, Santa Barbara, San Diego, San Luis Obispo, Ventura, and Orange counties, and from the Channel Islands (CNPS 2001). This species is rather common in the open grasslands of the Santa Monica Mountains and is also found in Orange County, mostly in the northeastern portion of the County, but this species also ranges from the San Joaquin Hills and south to the Ortega Highway (Highway 74). South of Highway 74, this mariposa lily appears to be very uncommon. However, recent studies (P&D 2002) have noted four localities southeast of the landfill boundary. Catalina mariposa lily has the potential to occur on the project site but was not observed during previous surveys.

Intermediate Mariposa Lily (Calochortus weedii var. intermedius)

Intermediate mariposa lily is a CNPS List 1B species. This bulbiferous perennial herb typically blooms between May and July (CNPS 2001). It is found on hilly coastal sage scrub and valley grassland vegetation types (Munz 1974). It prefers dry, rocky, open slopes from sea level to approximately 2,230 feet above msl (Hickman 1993). This species is known from Los Angeles, Riverside, and Orange counties (CNPS 2001). This species is known to occur in the Chino Hills and Carbon Canyon, and in the Santa Ana Mountains to the Gypsum Canyon area. In Orange County, this species has been recorded from Chino Hills to the San Joaquin Hills. In the foothill areas, it is generally found on exposed sandstone substrate and extends from Gypsum Canyon south to Cristianitos Canyon. Intermediate mariposa lily has potential to occur on the project site but was not observed during previous surveys.

Santa Barbara Morning-glory (Calystegia sepium ssp. binghamiae)

Santa Barbara morning-glory is a CNPS List 1A species; List 1A species are those species that are presumed to be extinct. This species was probably extirpated by wetland modification and urbanization (CNPS 2001). This rhizomatous perennial herb typically bloomed between April and May (CNPS 2001). This species historically occurred in coastal salt marshes in Santa Barbara and Orange counties (Munz 1974). It was found from sea level to approximately 70 feet above msl (Hickman 1993). The only known historic locations in Orange County are in the Wintersberg Channel and a marsh along Pacific Coast Highway in Huntington Beach (CDFG 1998). Santa Barbara morning-glory is not expected to occur within the project site.

Buck's Jewel-flower (Caulanthus heterophyllus var. pseudosimulans)

Buck's jewel-flower is not formally listed by the resource agencies or CNPS, but is considered a species of local concern. It typically blooms between March and May (Munz 1974). This species is found in dry, open scrub and chaparral vegetation types, often following a fire or other disturbance, from sea level to approximately 4,270 feet above msl (Hickman 1993). It prefers granitic substrates. Although mostly known from the inland Riverside Valley, the Buck's jewelflower also occurs in the Santa Monica and Santa Ana mountains. It has also been recorded from western San Bernardino County, including the lower elevations of the San Bernardino Mountains, and northern San Diego County. Although specimens have been annotated as this variety in various local herbaria, the nomenclature for this variety has not been formally published. Therefore, although this name is not currently valid, it has been recognized by CDFG (2003) and other agencies working on plant conservation (Reiser 2001). The distribution of the Buck's jewel-flower is not well known in Orange County. It has been recorded in the Santa Ana Mountains. It is also known occur in Blind Canyon, Fremont Canyon, Limestone Canyon, the San Joaquin Hills, south Laguna, and San Juan Hot Springs. It has recently been recorded from a single locality within the Donna O'Neill Reserve. Buck's jewel-flower has the potential to occur on the project site but was not observed during previous surveys.

Southern Tarplant (Centromadia [Hemizonia] parryi ssp. australis)

Southern tarplant is a CNPS List 1B species. This annual herb typically blooms between May and November (CNPS 2001). It is found in coastal lowlands in valley grassland vegetation types (Munz 1974). It prefers seasonally moist, saline grasslands from sea level to approximately 660 feet above msl (Hickman 1993). This species is also found in vernal pools and around the margins of marshes and swamps; it is known from Los Angeles, Orange, Santa Barbara, San Diego, and Ventura counties, and from Baja California (CNPS 2001). This species occurs in a number of highly fragmented populations from Santa Barbara County to Baja California. In Orange County it is known from a number of scattered populations occurring at the Wintersburg channel (Bolsa Chica area), Fairview park area, Upper Newport Bay, San Joaquin Marsh, UCI campus, and localities along Peters Canyon channel in Mission Viejo and at the Corner of Laguna Canyon and Laguna Canyon Road. In southern Orange County, this species has been recorded for the Cañada Chiquita region. Along the southern Foothill Transportation Corridor some 18 localities have been identified. Southern tarplant has the potential to occur on the project site but was not observed during previous surveys.

San Diego Mountain Mahogany (Cercocarpus minutiflorus)

San Diego mountain mahogany is not formally listed by the resource agencies or CNPS, but is considered a species of local concern. It typically blooms between March and May and is found on dry slopes in chaparral vegetation types (Munz 1974). It occurs from sea level to approximately 4,270 feet above msl (Hickman 1993). It is generally known from San Diego County and from Baja California (Hickman 1993). In San Diego County, this species is the principal mountain mahogany found at lower elevations. In contrast, the San Diego mountain mahogany has a very limited distribution in Orange County and has been located only from Niguel Hill in the Aliso Woods regional park. A single shrub has recently been recorded from the Donna O'Neill reserve. San Diego mountain mahogany is not expected to occur on the project site.

Orcutt's Pincushion (Chaenactis glabriuscula var. orcuttiana)

Orcutt's pincushion is a CNPS List 1B species. This annual herb typically blooms between January and August (CNPS 2001). It occurs on coastal dunes in coastal strand vegetation types (Munz 1974). It is found from sea level to approximately 330 feet above msl (Hickman 1993). This species is known from Los Angeles, Orange, San Diego, and Ventura counties, and from Baja California; it is considered extirpated from Orange County (CNPS 2001). Orcutt's pincushion is not expected to occur on the project site.

Summer Holly (Comarostaphylis diversifolia ssp. diversifolia)

Summer holly is a CNPS List 1B species. This evergreen shrub typically blooms between April and June (CNPS 2001). It occurs on dry coastal slopes at low elevations in chaparral vegetation types (Munz 1974). It is found from approximately 330 to 1,800 feet above msl (Hickman 1993). This species is known from Orange, Riverside, and San Diego counties, and from Baja California (CNPS 2001). It is known to occur in Laguna Niguel (CDFG 2001) in Orange County. Summer holly is not expected to occur on the project site.

Small-flowered Morning-glory (Convolvulus simulans)

Small-flowered morning-glory is a CNPS List 4 species. This annual herb typically blooms between March and July (CNPS 2001). It is found in grassy and rocky places, in valley grassland and coastal sage scrub vegetation types (Munz 1974). It is associated with wet clay and serpentine ridges, and occurs from approximately 100 to 2,300 feet above msl (Hickman

1993). It is known from Contra Costa, Kern, Los Angeles, Orange, Riverside, Santa Barbara, San Benito, San Diego, San Joaquin, San Luis Obispo, and Stanislaus counties, and from the Channel Islands and Baja California (CNPS 2001). This species is not well known in Orange County and currently has only been located in Cristianitos Canyon, Aliso Canyon, Yorba Linda, Brea, Dana Point, Prima Deshecha, Bell Creek, and Cristianitos Canyon. Three localities of small-flowered morning glory were observed on the project site in 2003 (Bramlet 2004), two of which consisted only of a single plant. A third locality, found in the southwest portion of the project site in the 2005 survey, contained approximately 1,200 individuals of small-flowered morning glory (BonTerra Consulting 2006c).

Paniculate Tarplant (*Deinandra* [*Hemizonia*] *paniculata*)

Paniculate tarplant is a CNPS List 4 species. This annual herb typically blooms between April and November (CNPS 2001). It occurs on low elevation dry hills and mesas, in valley grassland vegetation types (Munz 1974). It is found from sea level to approximately 980 feet above msl (Hickman 1993). This species also occurs in coastal scrub and foothill grasslands, and is usually vernally mesic (CNPS 2001). It is known from Riverside, Orange, San Bernardino, and San Diego counties, and from Baja California (CNPS 2001). Surveys conducted in 2003 located approximately 6,000 plants in 9 locations within the eastern portion of the project site (Bramlet 2004). Paniculate tarplant has also been observed in grassland habitat on the northwest portion of the project site (Bramlet 2000).

Western Dichondra (Dichondra occidentalis)

Western dichondra is a CNPS List 4 species. This rhizomatous perennial herb typically blooms between March and July (CNPS 2001). It occurs on mostly dry sandy banks, in brush or under trees, in coastal sage scrub, chaparral, and southern oak woodland vegetation types (Munz 1974). It is found on slopes and headlands from approximately 160 to 1,640 feet above msl (Hickman 1993); it may also occur in valley and foothill grasslands (CNPS 2001). This species is known from Orange, Santa Barbara, San Diego, and Ventura counties, and from the Channel Islands and Baja California (CNPS 2001). In Orange County, this species is known to occur along the coast at Dana Point, San Joaquin Hills, Chiquita Ridge, and in Cristianitos Canyon (Bramlet 2004). Western dichondra has the potential to occur on the project site but was not observed during previous surveys.

Blochman's Dudleya (*Dudleya blochmaniae* ssp. *blochmaniae*)

Blochman's dudleya is a CNPS List 1B species. This perennial herb typically blooms between April and June (CNPS 2001). It occurs in dry stony places, often on serpentine substrates, in coastal sage scrub vegetation types located near the coast (Munz 1974). It is also associated with clay-dominated soils, and is found from sea level to approximately 1,480 feet above msl (Hickman 1993). This species is known from Los Angeles, Orange, Santa Barbara, San Diego, San Luis Obispo, and Ventura counties, and from Baja California (CNPS 2001). Blochman's dudleya has been reported in Orange County at the Dana Point Headlands and bluffs at San Clemente (CDFG 2004b). Blochman's dudleya is not expected to occur on the project site.

Many-stemmed Dudleya (Dudleya multicaulis)

Many-stemmed dudleya is a CNPS List 1B species. This perennial herb typically blooms between April and July (CNPS 2001). It is found in dry stony places in coastal sage scrub and chaparral vegetation types (Munz 1974). It prefers heavy, often clay soils on the coastal plain, and is found from sea level to approximately 1,970 feet above msl (Hickman 1993). This species also occurs in valley and foothill grasslands, and is known from Los Angeles, San Bernardino, Riverside, Orange, and San Diego counties (CNPS 2001). Known populations of

this species in Orange County include Cañada Chiquita, Christianitos Canyon, and Segunda Deshecha (Bramlet 2004). Many-stemmed dudleya has the potential to occur on the project site but was not observed during previous surveys.

Laguna Beach Dudleya (Dudleya stolonifera)

Laguna Beach dudleya is a federally and state-listed Threatened species and a CNPS List 1B species. This stoloniferous perennial herb typically blooms between May and July (CNPS 2001). It occurs on cliffs in coastal sage scrub vegetation types in canyons near Laguna Beach (Munz 1974). It is found on north-facing cliffs and outcrops, from sea level to approximately 820 feet above msl (Hickman 1993). This species is known from approximately nine occurrences near Laguna Beach in Orange County (CNPS 2004). Laguna Beach dudleya is not expected to occur on the project site.

Sticky Dudleya (Dudleya viscida)

Sticky dudleya is a CNPS List 1B species. This perennial herb typically blooms between May and June (CNPS 2001). It occurs in rocky cliffs in coastal sage scrub vegetation types. It is found from sea level to approximately 1,480 feet above msl (Hickman 1993). This species is known from Orange, Riverside, and San Diego counties (CNPS 2001). This species is known to occur in San Juan Canyon east of San Juan Hot Springs (CDFG 2004a). Sticky dudleya is not expected to occur on the project site.

Pendleton Button-celery (Eryngium pendletonensis)

Pendleton button-celery is a CNPS List 1B species. This perennial herb typically blooms between April and June (CNPS 2001). It occurs in clay, vernally mesic coastal bluff scrub, valley and foothill grassland and vernal pools, and is found from approximately 50 to 360 feet above msl (CNPS 2001). This is a new species that was described in 1999 (Marsden and Simpson 1999). The Pendleton button-celery is a narrow endemic to San Diego County that occurs on exposed coastal bluffs and grasslands (Marsden and Simpson 1999). Pendleton button-celery is not expected to occur on the project site.

Cliff Spurge (Euphorbia miseria)

Cliff spurge is a CNPS List 2 species. This shrub typically blooms between December and August (CNPS 2001). It occurs on sea bluffs in coastal sage scrub vegetation types (Munz 1974). It is found on rocky slopes from sea level to approximately 1,640 feet above msl (Hickman 1993). This species is known from Orange, Riverside, and San Diego counties, and from the Channel Islands and Baja California (CNPS 2001). A historic location was also recorded in Riverside County, but this location was heavily damaged by frost. This species is threatened by development. This species is known to occur in Dana Point, Laguna Beach, and Corona del Mar in Orange County. Cliff spurge is not expected to occur on the project site.

Palmer's Grapplinghook (Harpagonella palmeri)

Palmer's grapplinghook is a CNPS List 4 species. This annual herb typically blooms between March and May (CNPS 2001). It occurs on dry slopes and mesas in chaparral vegetation types (Munz 1974). It is also found in coastal scrub and grassland from sea level to approximately 1,480 feet above msl (Hickman 1993). This species is associated with clay soils, and is known from Los Angeles, Orange, Riverside, and San Diego counties, and from Baja California (CNPS 2001). In Orange County, Palmer's grapplinghook has generally been found on clay soils in the southern part of the County. Populations have been located at Dana Point, Casper's Regional Park, Gabino Canyon, Cristianitos Canyon, and the ridgeline east of Cañada Chiquita (Bramlet

2004). Palmer's grapplinghook has the potential to occur on the project site but was not observed during previous surveys.

Vernal Barley (Hordeum intercedens)

Vernal barley is a CNPS List 3 species. This annual herb typically blooms between March and June (CNPS 2001). It occurs in vernal pools; in dry, saline streambeds; and in alkaline flats from sea level to approximately 3,280 feet above msl (Hickman 1993). It is known from Kings, Los Angeles, Mono, Orange, Riverside, Santa Barbara, San Benito, San Diego, San Mateo, and Ventura counties, and from the Channel Islands and Baja California (CNPS 2001). In Orange County, this species is known to occur at Fairview Park, the UCI Ecological Reserve, Dana Point, and San Clemente State Beach. Vernal barley was documented in nine locations throughout the project site during focused surveys conducted in 2003 (Bramlet 2004).

Robinson's Peppergrass (Lepidium virginicum var. robinsonii)

Robinson's peppergrass is a CNPS List 1B species. This annual herb typically blooms between January and July (CNPS 2001). It occurs in coastal sage scrub and chaparral vegetation types (Munz 1974). It is found on dry soils from sea level to approximately 1,640 feet above msl (Hickman 1993). This species is known from Los Angeles, Riverside, Orange, Santa Barbara (extirpated), San Bernardino, and San Diego counties, and from Baja California (CNPS 2001). Its distribution in Orange County is poorly understood. The only recent report in Orange County is from Modjeska Canyon in the Santa Ana Mountains (CDFG 2004b). It has not been reported in Rancho Mission Viejo, but suitable habitat is found throughout the South County area (Bramlet 2004). Robinson's peppergrass has the potential to occur on the project site but was not observed during previous surveys.

Small-flowered Microseris (*Microseris douglasii* var. *platycarpha*)

Small-flowered microseris is a CNPS List 4 species. This annual herb typically blooms between March and May (CNPS 2001). It occurs in grassy places (Munz 1974). It is found on inland clay soils often near vernal pools or serpentine outcrops, from sea level to approximately 3,280 feet above msl (Hickman 1993). It is known from Los Angeles, Riverside, Orange, and San Diego counties, and from the Channel Islands and Baja California (CNPS 2001). In Orange County, this species has been found in Chino Hills, Irvine Ranch area, Fremont Canyon, Blind Canyon, near Quail Hill, UCI Ecological Reserve, Fairview Park, and Saddleback meadow (Bramlet 2004). Surveys for the Foothill Transportation Corridor (FTC) have located the small-flowered microseris in 19 localities within the project site (Bramlet 2004). Small-flowered microseris has the potential to occur on the project but was not observed during previous surveys.

Prostrate Navarretia (*Navarretia prostrata*)

Prostrate navarretia is a CNPS List 1B species. This annual herb typically blooms between April and July (CNPS 2001). It occurs in vernal pools and moist places (Munz 1974). It is also found in alkaline floodplains, and grows from sea level to approximately 2,300 feet above msl (Hickman 1993). This species is known from Alameda (extirpated), Merced, Monterey, Los Angeles, Orange, Riverside, San Bernardino (possibly extirpated), and San Diego counties (CNPS 2001). Prostrate navarretia is not expected to occur on the project site.

Chaparral Nolina (Nolina cismontana)

Chaparral nolina is a CNPS List 1B species. This evergreen shrub typically blooms from May to July (CNPS 2001). It is generally associated with sandstone or gabbro soils in chaparral and coastal scrub vegetation types, and is found from approximately 460 to 4,180 feet above msl

(CNPS 2001). This species is known from Los Angeles, Orange, San Diego, and Ventura counties (CNPS 2001). Chaparral nolina is not expected to occur on the project site.

Golden-rayed Pentachaeta (Pentachaeta aurea)

Golden-rayed pentachaeta is a CNPS List 4 species. This annual herb typically blooms between March and July (CNPS 2001). It occurs in dry open places in valley grassland, coastal sage scrub, and yellow pine forest vegetation types (Munz 1974). It is found in grassy areas, from sea level to approximately 6,070 feet above msl (Hickman 1993). This species is known from Los Angeles, Orange, Riverside, San Bernardino, and San Diego counties, and from Baja California (CNPS 2001). Golden-rayed pentachaeta has the potential to occur in the project site but was not observed during previous surveys.

Nuttall's Scrub Oak (Quercus dumosa)

Nuttall's scrub oak is a CNPS List 1B species. This evergreen shrub typically blooms between February and April (CNPS 2001). It occurs on dry slopes in chaparral and foothill woodland vegetation types (Munz 1974). It is found on generally sandy soils and sandstone substrates near the coast, from sea level to approximately 660 feet above msl (Hickman 1993). This species is known from Santa Barbara, Orange, and San Diego counties, and from Baja California (CNPS 2001). In Orange County, it is found in Aliso Woods Regional Park, Laguna Niguel, and Dana Point (CDFG 2004a). Nuttall's scrub oak is not expected to occur on the project site.

Coulter's Matilija Poppy (*Romneya coulteri*)

Coulter's matilija poppy is a CNPS List 4 species. This rhizomatous perennial shrub typically blooms between March and July (CNPS 2001). It occurs in dry washes and canyons in chaparral and coastal sage scrub vegetation types, away from the immediate coast (Munz 1974). It has the largest flowers of any plant native to California, and is found from sea level to approximately 3,940 feet above msl (Hickman 1993). It is known from Los Angeles, Riverside, Orange, and San Diego counties (CNPS 2001). This species occurs in scattered localities throughout Orange County, including Caspers Regional Park (Bramlet 2004). Coulter's matilija poppy has the potential to occur project site, but was not observed during previous surveys.

San Miguel Savory (Satureja chandleri)

San Miguel savory is a CNPS List 1B species. This perennial herb typically blooms between March and July (CNPS 2001). It occurs in rocky canyons in chaparral vegetation types (Munz 1974). It is found from approximately 1,700 to 2,260 feet above msl (Hickman 1993). This species is generally associated with rocky, gabbroic, or metavolcanic soils in chaparral, cismontane woodland, coastal scrub, riparian woodland, and valley and foothill grassland vegetation types (CNPS 2001). It is known from Orange, Riverside, and San Diego counties, and from Baja California (CNPS 2001). This species occurs near San Juan Hot Springs (CDFG 2001). San Miguel savory is not expected to occur on the project site.

Rayless Ragwort (Senecio aphanactis)

Rayless ragwort is a CNPS List 2 species. This annual herb typically blooms between January and April (CNPS 2001). It occurs in dry open places in coastal sage scrub and chaparral vegetation types located near the coast (Munz 1974). It is also found in drying alkaline flats, and grows from sea level to approximately 1,310 feet above msl (Hickman 1993). This species is known from Alameda, Contra Costa, Fresno, Los Angeles, Merced, Orange, Riverside, Santa Barbara, Santa Clara, San Diego, San Luis Obispo, Solano, and Ventura counties, and from Baja California (CNPS 2001). In Orange County, this species is known from Dana Point Headlands and the UC Irvine Ecological Reserve (CDFG 2004b). Rayless ragwort has potential to occur on the project site but was not observed during previous surveys.

Salt Spring Checkerbloom (Sidalcea neomexicana)

Salt Spring checkerbloom is a CNPS List 2 species. This perennial herb typically blooms between March and June (CNPS 2001). It is found in alkaline springs and marshes, from sea level to approximately 4,920 feet above msl (Hickman 1993). It also occurs on mesic sites in coastal scrub, chaparral, lower montane coniferous forest, Mojavean desert scrub, and playas (CNPS 2001). This species is known from Los Angeles (extirpated), Ventura, Orange, Santa Barbara, San Bernardino, and Riverside counties, and from Baja California (CNPS 2001). In Orange County, this species has been recorded in Los Alamitos and east of Cañada Chiquita. Salt Spring checkerbloom has the potential to occur on the project site but was not observed during previous surveys.

Parry's Tetracoccus (*Tetracoccus dioicus*)

Parry's tetracoccus is a CNPS List 1B species. This deciduous shrub typically blooms between April and May (CNPS 2001). It occurs on dry stony slopes in chaparral vegetation types (Munz 1974). It is found from sea level to approximately 3,280 feet above msl (Hickman 1993). This species also occurs in coastal scrub, and is known from Orange, Riverside, and San Diego counties, and from Baja California (CNPS 2001). In Orange County, this species is known to occur at the San Juan Campground, which is near the headwaters of San Juan Creek in the Cleveland National Forest. Parry's tetracoccus is not expected to occur on the project site.

Crownbeard (Verbesina dissita)

Crownbeard is a federally and state-listed Threatened species and a CNPS List 1B species. This perennial herb typically blooms between April and July (CNPS 2001). It occurs on shrubby coastal slopes, from sea level to approximately 330 feet above msl (Hickman 1993). It grows in maritime chaparral and coastal scrub vegetation types (CNPS 2001). It is known in California from only two occurrences near southern Laguna Beach in Orange County; it also is found in Baja California (CNPS 2001). This species is known to occur in Aliso Canyon (CDFG 2001). The majority of populations of this species occur in southern maritime chaparral, with about ten percent occurring in coastal sage scrub. Crownbeard is not expected to occur on the project site.

Special Status Wildlife Species

Table 5.5-3 below presents 44 special status wildlife species that potentially occur on the project site.

Invertebrates

San Diego Fairy Shrimp (Branchinecta sandiegonensis)

San Diego fairy shrimp is a federally Endangered species. This species typically occupies depressional topography in pools with a depth of 3 to 12 inches in San Diego and Orange counties. In the vicinity of the study area, this species is known from Cañada Chiquita and San Onofre Bluff (CDFG 2001). The abandoned cattle pond in the eastern portion of the project site provides potentially suitable habitat for this species. Therefore, there is potential for the San Diego fairy shrimp to occur on the project site.

TABLE 5.5-3 SPECIAL STATUS WILDLIFE SPECIES POTENTIALLY OCCURRING ON THE PROJECT SITE

Species	USFWS	CDFG	Potential for Occurrence
Invertebrates			
San Diego fairy shrimp Branchinecta sandiegonensis	FE	CSC	Potential to occur on site. The cattle pond in eastern portion of project site provides potentially suitable habitat for this species.
Riverside fairy shrimp Streptocephalus woottoni	FE	CSC	Potential to occur on site. The cattle pond in eastern portion of project site provides potentially suitable habitat for this species.
Amphibians			
western spadefoot toad Spea [Scaphiopus] hammondii	_	CSC	Observed on site. Grasslands and other sparsely vegetated habitats provide suitable habitat for this species. Ponds, including vernal pools and cattle ponds, are required for breeding.
arroyo toad Buffo californicus	FE	CSC	Not expected to occur on site. No suitable habitat, as the species requires rivers with shallow, gravelly pools adjacent to sandy terraces.
Reptiles			
southwestern pond turtle Emys [Clemmys] marmorata pallida	_	CSC	species requires wetlands with more substantial ponds than are present on the project site. Also, surveys have not detected this species.
coast horned lizard [blainvillei population] Phrynosoma cornatum	_	CSC	Potential to occur on site. Generally requires dry sandy openings within coastal sage scrub and grassland habitats. Potentially suitable habitat exists on the project site.
Coronado skink Eumeces skiltonianus interparietalis		CSC	Observed on site (subspecies undetermined). Generally frequents grassland, open chaparral, and sage scrub habitats.
Belding's orange-throated whiptail Aspidoscelis [Cnemidophorus] hyperythra beldingi		CSC	Potential to occur on site. Generally occurs in openings within coastal sage scrub and chaparral habitats. Potentially suitable habitat exists on the project site.
silvery legless lizard Anniella pulchra pulchra		CSC	Potential to occur on site. Inhabits moist sandy soil within woodlands, riparian, and scrub habitats. Potentially suitable habitat exists on the project site.
coast patch-nosed snake Salvadora hexalepis virgultea	_	CSC	Potential to occur on site. Occurs in a variety of habitats, including coastal sage scrub and chaparral. Potentially suitable habitat exists on the project site.
northern red-diamond rattlesnake Crotalus [Salvadora] ruber ruber		CSC	Observed on site. Generally found in coastal sage scrub and grassland habitats.
Birds		r	Observed on site Derwines woodlands for nesting
Cooper's Hawk Accipiter cooperii		CSC	Present every breeding season since 2000, but nesting not confirmed.
sharp shinned hawk Accipiter striatus	—	CSC	Observed on site. Only a winter resident to Orange County.
golden eagle Aquila chrysaetos	_	CSC	Observed on site. The project site provides suitable foraging habitat for this species as indicated by several observations of different individuals since 2000.
ferruginous hawk Buteo regalis		CSC	Potential to occur on site. Rare but regular winter visitor to open habitats, such as grasslands and agricultural fields. The project site provides suitable foraging habitat for this raptor.
Swainson's hawk Buteo swainsoni	_	ST	Observed on site. A very rare migrant in Orange County. This raptor is typically observed over open habitats such as grasslands and agricultural fields. A spring migrant was observed flying over the project site during 1992 spring surveys (April 10) and another individual was observed at the project site on December 9, 2002.
TABLE 5.5-3 (Continued) SPECIAL STATUS WILDLIFE SPECIES POTENTIALLY OCCURRING ON THE PROJECT SITE

Species	USFWS	CDFG	Potential for Occurrence
northern harrier <i>Circus cyaneus</i>		CSC	Observed on site. Uncommon winter visitor in Orange County, with a few pairs remaining through summer. Forages over open habitats such as grasslands, marshlands, or agricultural areas. One breeding pair has been present during recent surveys (2000–2005), with nesting occurring in coastal sage scrub habitat just west of the project site.
white-tailed kite Elanus leucurus	_	FP	Observed on site. Uncommon-to-fairly common local winter resident and rare-to-uncommon summer resident in Orange County. This raptor prefers broken woodland with extensive grasslands for breeding. Suitable habitat for nesting is found in the sycamore-lined portions of Prima Deshecha Cañada stream channel.
prairie falcon Falco mexicanus		CSC	Observed on site. Rare but regular winter resident to Orange County. This raptor formerly nested in the County. Requires open habitats, such as grasslands, agricultural fields, and marshes for foraging. Suitable foraging habitat exists on site.
American peregrine falcon Falco peregrinus anatum		SE/FP	Potential to occur on site. Currently, an uncommon winter resident and rare summer breeding resident in Orange County. Prefers coastal habitats including bays, lagoons, marshes and beaches, wherever birds congregate. The project site provides suitable foraging habitat for this falcon, but no nesting habitat.
long-eared owl Asio otus		CSC	Potential to occur on site. Currently, a rare breeding resident in Orange County. This species requires grasslands and scrub habitats for foraging and dense woodlands for nesting. The project site provides suitable foraging, but marginal nesting habitat.
burrowing owl Athene cunicularia	_	CSC	Observed on site. Rare and local summer and winter resident in Orange County. Requires open fields in relatively level terrain, generally with the presence of ground squirrel burrows for roosting and nesting. One individual observed in November 2004 and April 2005.
southwestern willow flycatcher Empidonas traillii extimus	FE	SE	Only migrants observed on site. A very rare, local summer resident in southern California that requires well-developed riparian woodland. Very few recent breeding records for Orange County. The species (subspecies other than <i>extimus</i>) is an uncommon migrant throughout Orange and other southern California counties. No nesting or territorial birds were observed on site during surveys.
loggerhead shrike Lanius ludovicianus		CSC	Observed on site. Fairly common winter visitor and an uncommon summer resident in Orange County. Requires open-to-partially-open habitats. Suitable habitat occurs throughout the site. One breeding pair has been present on the project site during recent surveys (2000–2004).
least Bell's vireo Vireo bellii pusillus	FE	SE	Observed on site. A rare, local summer resident in southern California, and typically prefers well-developed riparian woodland for breeding habitat. Directed surveys have been conducted in 1998 (4 pairs), 2000 (4 pairs), 2001 (9 pairs), 2002 (5 pairs), 2003 (4 pairs), 2004 (6 pairs), and 2005 (mid-season results indicate 10 territories and most with pairs).

TABLE 5.5-3 (Continued) SPECIAL STATUS WILDLIFE SPECIES POTENTIALLY OCCURRING ON THE PROJECT SITE

Species	USFWS	CDFG	Potential for Occurrence
California horned lark Eremophilia alpestris actia		CSC	Observed on site. Rare-to-uncommon breeding resident in Orange County. Essentially requires bare ground to short grass habitats in relatively level terrain. This species is most common around the active landfill portion of the project site.
San Diego cactus wren Campylorhynchus brunneicapillus sandiegensis		CSC	Observed on site. Uncommon, year-round resident in Orange County. Requires coastal cholla and prickly pear cactus patches within coastal sage scrub habitat. Six pairs of cactus wrens were located on the project site during 1992 field surveys, but none were detected during the 1998, 2003, 2004, and 2005 surveys.
coastal California gnatcatcher Polioptila californica californica	FT	CSC	Observed on site. Uncommon, year-round resident in Orange County. Requires coastal sage scrub habitat, generally in more coastal or lower elevation areas. Directed surveys covering the whole project site have been conducted in 1992 (10 pairs), 1998 (6 pairs), 2003 (4 to 5 pairs), 2004 (2 pairs), 2005 (2 pairs), and 2006 (7 pairs and 1 individual). The survey results indicate a natural cycle in the species population on the project site as well as within the region.
yellow warbler Drendroica petechia brewsteri		CSC	Observed on site. Uncommon breeder in Orange County. Requires riparian habitat for breeding, but widespread in all habitats during migration. Several pairs have nested in Prima Deshecha Cañada during recent surveys (2000–2005).
yellow-breasted chat <i>Icteria virens</i>		CSC	Observed on site. Locally, rare-to-uncommon summer resident in Orange County. Requires riparian habitats with dense brushy vegetation with or without an overstory of trees. Several pairs have nested in Prima Deshecha Cañada during recent surveys (2000–2005).
southern California rufous-crowned sparrow <i>Aimophia ruficeps canescens</i>	_	CSC	Observed on site. Uncommon-to-fairly common year- round resident in Orange County. Favors steep and sometimes rocky slopes in sparsely vegetated, often grassy, habitats. This sparrow is present at several locations on the project site, with as many as 12 to 15 total pairs estimated to be present during previous surveys (1992–1998).
Bell's sage sparrow Amphispiza belli belli		CSC	Limited potential to occur on site. Rare year-round resident in Orange County. Found in coastal sage scrub or sparse chaparral habitats. Known from only a few areas in the County.
tricolored blackbird Agelaius tricolor		CSC	Observed on site. Uncommon year-round resident in Orange County. Primary breeding habitat consists of bodies of water with dense reed beds and adjacent grasslands. The project site provides suitable foraging habitat and potentially suitable nesting habitat. A flock of 125 was observed foraging at the active landfill portion of the project site on November 14, 2002.
Mammals			
pallid bat Antrozous pallidus		CSC	Potential to occur on site. Occurs most often in mixed oak and grassland habitats. Roosting habitat consists of rock crevices.
pale big-eared bat Corynorhinus townsendii pallescens		CSC	Potential to occur on site. Occurs in a variety of habitats. Roosting habitat consists of mines, caves, and buildings.
Yuma myotis <i>Myotis yumanensis</i>	_	CSC	Potential to occur on site. Shows strong affinity for wooded canyon bottoms and water (i.e., ponds, lakes). Preferred roosting habitat consists of caves and old buildings.

TABLE 5.5-3 (Continued) SPECIAL STATUS WILDLIFE SPECIES POTENTIALLY OCCURRING ON THE PROJECT SITE

Species	USFWS	CDFG	Potential for Occurrence					
western mastiff bat Eumops perotis		CSC	Limited potential to occur on site. This is a rare resident in Orange County. These are large bats with specific roosting requirements: vertical cliffs with deep crevices about 15 or 20 feet above the ground.					
pocketed free-tailed bat Nyctinomops fermorosaccus		CSC	Potential to occur on site. Occurs in a variety of habitats. Roosting habitat consists of rock crevices, caverns, and buildings.					
Big free-tailed bat Nyctinomops macrotis		CSC	Limited potential to occur on site. Forages over water habitats and roosts in rocky crevices on high cliffs.					
San Diego black-tailed jackrabbit Lepus californicus bennettii	_	CSC	Not expected to occur on site. Rare-to-locally-uncommon resident in Orange County. Prefers open habitats of relatively level terrain, such as alluvial scrub and agricultural fields.					
northwestern San Diego pocket mouse <i>Chaetodipus fallax fallax</i>		CSC	Potential to occur on site. The population status of this species is poorly known, although it probably occurs uncommonly in south Orange County. It inhabits mixed chaparral and coastal sage scrub, especially in sandy or rocky areas.					
Pacific pocket mouse Perognahnus longimembris paxificus	FE	CSC	Not expected to occur on site. Rare and local year-round resident in Orange County. This species prefers coastal grassland and sage scrub associations in areas of relatively loose soils. There is limited data on this species, but it has been recorded recently as close as Cristianitos Road and I-5 (known as the San Mateo population). Project site is outside the known range for the species.					
San Diego desert woodrat Neotoma lepida intermedia	San Diego desert woodrat		Potential to occur on site. Occupies sparse habitats such as coastal sage scrub. They build distinctive stick home in understory of vegetation.					
southern grasshopper mouse Onychomys torridus Ramona		CSC	Potential to occur on site. Occupies grassland and sparse scrub habitats with sandy soils.					
Federal Designations: FC = Federal Candidate Species (form FE = Federally listed Endangered FT = Federally listed Threatened FPT = Proposed for listing as Federally State Designations: CSC CSC = California Special Concern Spec FP = California Department of Fish an SE = State-listed Endangered ST = State-listed Threatened SA = Special Animal	nerly Categor Threatened ies d Game Fully	y 1 candidat	Species					

On October 23, 2000, the USFWS published a final critical habitat for the San Diego fairy shrimp. A total of 4,025 acres in Orange and San Diego counties were designated as critical habitat for the San Diego fairy shrimp. The project site is not located in any areas designated as San Diego fairy shrimp critical habitat.

Riverside Fairy Shrimp (Streptocephalus woottoni)

Riverside fairy shrimp is a federally listed Endangered species. Riverside fairy shrimp are restricted to deep, seasonal vernal pools, ephemeral ponds, and stock ponds (i.e., pools with a depth of 12 to 18 inches or more). This species prefers warm-water pools that have low-to-moderate dissolved solids and that remain filled for extended periods of time. All known habitat for the species lies within annual grasslands, which may be interspersed throughout chaparral or coastal sage scrub vegetation (Dudek & Associates 2000). This species ranges from

southern California to northern Baja California (USFWS 2000). In the vicinity of the project site, this species is known from Cañada Gobernadora and Cristianitos Creek (CDFG 2001). The abandoned cattle pond in the eastern portion of the project site provides potentially suitable habitat for this species. Therefore, there is the potential for the Riverside fairy shrimp to occur on the project site.

On September 21, 2000, the USFWS published a proposed critical habitat for the Riverside fairy shrimp designating 12,060 acres of land in Los Angeles, Orange, Riverside, San Diego, and Ventura counties, California. The project site is outside the critical habitat area designated for this species.

Amphibians

Western Spadefoot Toad (Spea [Scaphiopus] hammondii)

Western spadefoot toad is a California Species of Special Concern and an NCCP/HCP Planning Species. From the Santa Clara River Valley in Los Angeles and Ventura counties southward, an estimated 80 percent of habitat for this species has been lost (Stebbins 2003). This species inhabits grassland, coastal sage scrub, and other habitats with open, sandy, gravelly soils. The western spadefoot breeds in quiet streams, vernal pools, and temporary ponds and is rarely seen outside the breeding season. For reproduction and successful metamorphosis, western spadefoot require rain-filled pools that hold standing water for more than three weeks (Feaver 1971; Brown 1966, 1967). The species is known to occur from the project site vicinity (such as San Juan Creek near Antonio Parkway Bridge) to the confluence with Verdugo Canyon. The abandoned cattle pond and surrounding grasslands in the eastern portion of the project site provide suitable habitat for this species. The western spadefoot was observed on the project site when an estimated several hundred tadpoles were identified in this pond on May 3, 2005.

Arroyo Toad (Bufo californicus)

Arroyo toad is a federal Endangered species, a California Species of Special Concern, and a CDFG protected species. This species is restricted to rivers with shallow, gravelly pools adjacent to sandy terraces. It forages on sandy terraces with complete canopy coverage by cottonwoods or willows. Adults excavate shallow burrows on terraces where they shelter during the day and during the dry season. This species historically occurred from San Luis Obispo to San Diego counties along most major rivers. Most of the remaining populations occur in the National Forests. In the project region, this species is known to occur in San Juan Creek, San Mateo Creek, Cristianitos Creek, Talega Canyon, Gabino Canyon, and on Camp Pendleton (CDFG 2001). The project site does not provide suitable habitat for this species. Therefore, the arroyo toad is not expected to occur on the project site.

On February 7, 2001, the USFWS published the final designation of 182,360 acres of land as critical habitat for the Endangered arroyo toad. These lands encompass portions of Monterey, Santa Barbara, Ventura, Los Angeles, San Bernardino, Orange, Riverside, and San Diego counties in California. The project site is outside the critical habitat area designated for this species.

Reptiles

Southwestern Pond Turtle (Emys [Clemmys] marmorata pallida)

Southwestern pond turtle is a California Species of Special Concern and an NCCP/HCP Planning Species. This subspecies occurs from approximately the San Francisco Bay area south through the Coast Ranges to northern Baja California, Mexico (Stebbins 2003). The

southwestern pond turtle is estimated to be in decline throughout 75 to 80 percent of its range (Stebbins 2003). The current range is similar to the historic range, but populations have become fragmented by agriculture and urban development. The southwestern pond turtle occurs primarily in freshwater rivers, streams, lakes, ponds, vernal pools, and seasonal wetlands and require water depths in excess of six feet and basking sites such as logs, banks, or other suitable areas above water level. The species is known from the project site vicinity such as in San Juan Creek. The project site does not provide suitable habitat for this species. Therefore, the southwestern pond turtle is not expected to occur on the project site.

Coast Horned Lizard [blainvillei population] (Phrynosoma coronatum)

Coast horned lizard (blainvillei population) is a California Species of Special Concern and an NCCP/HCP Planning Species. The two former subspecies of the coast horned lizard, (*P. c. blainvillei* and *P. c. frontale*) have recently been eliminated in current scientific literature, such as Stebbins (2003), based on current scientific studies on this species. The coast horned lizard occurs throughout much of California, west of the desert and Cascade-Sierra highlands south to Baja California, Mexico (Stebbins 2003). However, many of the populations in lowland areas have been reduced or eliminated due to urbanization and agricultural expansion (Stebbins 2003). It is a small, spiny, somewhat rounded lizard that prefers open areas for basking and loose, friable soil for burrowing (Stebbins 2003). Three factors have contributed to its decline: loss of habitat, over collecting, and the introduction of exotic ants. In some places, especially adjacent to urban areas, the introduced ants have displaced the native species upon which the lizard feeds (Hix 1990). The species is known from the project site vicinity such as in upper Cristianitos Canyon. The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the coast horned lizard (blainvillei population) to occur on the project site.

Coronado Skink (Eumeces skitonianus interparietalis)

Coronado skink is a California Species of Special Concern. It is one of four subspecies of the western skink and inhabits open, rocky habitats within scrub, chaparral, and grassland vegetation types. There are differing accounts for the range of this subspecies. Stebbins (1985) includes lower southern California, approximately from Orange County south along the coast into northern Baja California, Mexico, in this subspecies range. Tanner (1988) states that the Coronado Island skink is restricted to the mountains of extreme southern San Diego County and northern Baja California, Mexico, with small populations on Coronado Island and Todos Santos Island. However, Tanner's range map for the species includes Orange County within the area of intergradation between the Coronado Island skink and Skilton skink (*Eumeces skiltonianus*). Stebbins (2003) shows that the Coronado Island skink has been combined with Skilton's skink. Although the subspecies that occurs on the project site is undetermined, the western skink has been observed on the project site.

Belding's Orange-Throated Whiptail (Aspidoscelis [Cnemidophorus] hyperythra beldingi)

Belding's orange-throated whiptail is a California Species of Special Concern and an NCCP/HCP Planning Species. The two former subspecies of the orange-throated whiptail, (*C. c. hyperythrus* and *C. c. beldingi*) have recently been eliminated in current scientific literature, such as Stebbins (2003), based on current scientific studies on this species. This species occurs below 2,000 feet above msl in the western Peninsular Ranges from Orange and San Bernardino counties south to Baja California, Mexico (Stebbins 2003). Approximately 75 percent of the former range has been lost to development and the remaining populations are highly fragmented (Stebbins 2003). The orange-throated whiptail occurs in washes and in open areas of sage scrub and chaparral with gravelly soils, often with rocks. The orange-throated whiptail prefers the well-drained friable soil on slopes with a southern exposure that are barren or only

sparsely covered with vegetation. The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the Belding's orange-throated whiptail to occur on the project site.

Silvery Legless Lizard (Anniella pulchra pulchra)

The silvery legless lizard is a federal Species of Concern and a California Species of Special Concern. It is a small, secretive lizard that spends most of its life beneath the soil, under stones, logs, debris, or within leaf litter. The silvery legless lizard inhabits areas with moist sandy soil, including dry washes, woodlands, riparian, and scrub communities at elevations ranging from sea level to about 5,000 feet above msl (Stebbins 1985). The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the silvery legless lizard to occur on the project site.

Coast Patch-Nosed Snake (Salvadora hexalepis virgultea)

Coast patch-nosed snake is a California Species of Special Concern. It inhabits open sandy areas with rocky outcrops in scrub, grassland, and woodland vegetation types. This species, one of five subspecies of the patch-nosed snake, is a moderately sized, active snake. The coast patch-nosed snake ranges along the coast of California from San Luis Obispo County south into Baja California, Mexico. It occurs from sea level to about 7,000 feet above msl (Stebbins 1985). Threats to this species' continued survival include development, grazing, and fire control (Jennings and Hayes 1994). The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the coast patch-nosed snake to occur on the project site.

Northern Red-Diamond Rattlesnake (Crotalus [Salvadora] ruber ruber)

Northern red-diamond rattlesnake is a California Species of Special Concern. It inhabits open scrub, chaparral, and grassland vegetation types. This species ranges from southern San Bernardino County, south into Baja California, Mexico at elevations from sea level to about 5,000 feet above msl (Stebbins 1985). This species is threatened by development and human disturbance (Jennings and Hayes 1994). The project site provides suitable habitat for this species. The northern red-diamond rattlesnake has been observed on the project site.

Birds

Cooper's Hawk (Accipiter cooperii)

The Cooper's hawk is a California Species of Special Concern. Both resident and migratory populations exist in Orange County. Wintering Cooper's hawks are often seen in wooded urban areas and native woodland vegetation types. Preferred nesting habitats are oak and riparian woodlands dominated by sycamores and willows. Cooper's hawks in the region prey on small birds and rodents that live in woodland and occasionally scrub and chaparral vegetation types. The project site provides suitable foraging and nesting habitat for this species. The Cooper's hawk has been observed on the project site every breeding season since 2000, although nesting has not been confirmed.

Sharp-Shinned Hawk (Accipiter striatus)

The sharp-shinned hawk is a California Species of Special Concern. It is a relatively uncommon species that breeds to the north of the region and only occurs in Orange County during the winter season and migration. This raptor prefers woodland vegetation types, but can also be found in virtually any habitat as it passes through an area during migration. Oak and riparian

areas are preferred habitats. Some individuals probably winter in the county, while others continue to northern South America. The project site provides suitable foraging habitat, but not nesting habitat for this species. The sharp-shinned hawk has been observed on the project site.

Golden Eagle (Aquila chrysaetos)

The golden eagle is a California Fully Protected species, a California Species of Special Concern, and is also protected by the federal Bald Eagle Act. Habitat for this species generally consists of grasslands, deserts, savannahs, and early successional stages of forest and shrub habitats. Broad expanses of open country are required for foraging while nesting is primarily restricted to rugged mountainous areas in large trees or on cliffs (Johnsgard 1990). The golden eagle is an uncommon resident throughout southern California, except in the Colorado Desert and Colorado River where it is a casual winter visitor (Garret and Dunn 1981). The golden eagle is known to nest at Starr Ranch (CDFG 2001). The project site provides suitable foraging habitat and potentially suitable, but limited nesting habitat for this species. There have been several observations of different individuals of this species foraging on the project site since 2000.

Ferruginous Hawk (Buteo regalis)

Ferruginous hawk is a California Species of Special Concern. This raptor breeds north of the region but occurs regularly every winter season in small numbers. Ferruginous hawks occur in Orange County from mid-fall through early spring, and forage over agricultural areas, grasslands, and the ecotone between coastal sage scrub and grasslands. The distribution of the ferruginous hawk in Orange County has been greatly reduced as a result of the loss of wintering grounds. This species is also threatened by shooting (Ehrlich et al. 1988). The project site provides potentially suitable foraging habitat, but not nesting habitat for this species. Therefore, there is the potential for the ferruginous hawk to occur on the project site.

Swainson's Hawk (Buteo swainsoni)

Swainson's hawk is a state-listed Threatened species. This species formerly nested in Orange County, but has since been extirpated as a breeding species. Although rare, there is the potential for the Swainson's hawk to forage in the county as it migrates to and from South America. This species is threatened by loss of habitat, habitat deterioration on the South American wintering grounds, human disturbance at nest sites, shooting, and possibly pesticides (Remsen 1978). The project site provides potentially suitable foraging habitat, but is not expected to provide nesting habitat for this species. A spring migrant was observed flying over the project site during the 1992 surveys (April 10) and another individual was observed foraging on the project site on December 9, 2002.

Northern Harrier (Circus cyaneus)

Northern harrier is a California Species of Special Concern. It is a regular winter migrant in marshes and fields throughout southern California, but is very scarce as a local breeder (Garrett and Dunn 1981). Breeding habitat includes prairie, savannah, slough, wet meadow, and marsh vegetation types. Northern harrier can be expected at any month of the year and can be seen foraging in grassland, scrub, and riparian vegetation types. While once a relatively common species during fall, winter, and spring in undeveloped areas of Orange County, the northern harrier population is now greatly reduced and localized in distribution. This species is threatened by loss of habitat, pesticides (Ehrlich et al. 1988), and loss of suitable breeding habitat (Grinnell and Miller 1944). The project site provides suitable foraging and nesting habitat for this species. The northern harrier has been observed annually on the project site since 2000, including a breeding pair. Nesting by this pair has been confirmed, although no nest has yet been located on the project site.

White-Tailed Kite (Elanus leucurus)

White-tailed kite is a California Fully Protected species. This species is an uncommon-to-locallyfairly-common resident in coastal southern California, and a rare visitor and local nester on the western edge of the deserts (Garrett and Dunn 1981). White-tailed kites, while readily observable in undeveloped portions of Orange County, have begun to decline sharply in the region in the last decade. Reasons for this decline have been identified as loss of foraging habitat, roost sites, and nesting habitat (Bloom 1996). Kites typically nest in oaks, willows, and sycamores, and forage in grassland and scrub vegetation types. White-tailed kites show strong site fidelity to nest groves and trees. The project site provides suitable foraging and nesting habitat for this species. The white-tailed kite has been observed foraging on the project site.

Prairie Falcon (Falco mexicanus)

Prairie falcon is a California Species of Special Concern. Because of winter and nesting habitat loss, few areas remain in Orange County where prairie falcons can be consistently observed; no nest sites have been documented in the county in over 50 years. Preferred foraging habitat in Orange County includes grasslands, scrub vegetation types, estuaries, and typically dry environments where there are cliffs and bluffs for nests. This species is threatened by human disturbance at nest sites, shooting, and pesticides (Remsen 1978). The project site provides suitable foraging, but marginal nesting habitat for this species. The prairie falcon has been observed foraging on the project site during the winter season on a few occasions since 2000.

American Peregrine Falcon (Falco peregrinus anatum)

American peregrine falcon is a state-listed Endangered species and a CDFG fully protected species that, due to recent population gains, has been recently delisted from the USFWS's Endangered list. No such delisting has been proposed by the state. Peregrine falcons prey almost exclusively on birds and use a variety of habitats, particularly in wetlands and coastal areas. This species prefers to nest in cliffs or high structures. The project site provides suitable foraging, but no nesting habitat for this species. Therefore, there is the potential for the American peregrine falcon to occur on the project site for foraging, but is not expected to occur for nesting.

Long-Eared Owl (Asio otus)

Long-eared owl is a California Species of Special Concern. This species is a rare resident of Orange County with only 20 extant breeding territories. In Orange County this species nests in oak and willow woodlands and forages in scrub and grassland vegetation types. Long-eared owls have declined throughout California, but the most pronounced reductions have occurred in the southwestern part of the state where a minimum 55 percent decline has been documented (Bloom 1996). This species is known to occur in Bell and Wagon Wheel canyons (CDFG 2001). The project site provides suitable foraging, but marginally suitable nesting habitat for this species. Therefore, there is the potential for an individual long-eared owl (i.e., dispersing young) to occur for foraging on the project site, but this owl is not expected to nest on the project site.

Burrowing Owl (Athene cunicularia)

Burrowing owl is a California Species of Special Concern. In Orange County, burrowing owls breed and forage in grasslands and prefer flat-to-low-rolling hills in treeless terrain. They are small owls that nest in burrows, typically in open habitats most often along banks and roadsides. The burrowing owl is a widespread species throughout the western United States, but has declined in many other areas due to habitat modification, poisoning of its prey items, shooting, and human disturbance (Remsen 1978). Burrowing owls have undergone a severe decline in

Orange County (Bloom 1996). In 1999, the breeding population in Orange County was estimated to be four pairs (Bloom 1999). This species is known to occur in Newport Back Bay, UCI, and at the Seal Beach Naval Weapons Station (CDFG 2002a). The project site provides suitable foraging and nesting habitat for this species. The burrowing owl was observed on the project site during November 2004 and April 2005.

Southwestern Willow Flycatcher (Empidonax traillii extimus)

Southwestern willow flycatcher is a federally and state-listed Endangered species. This subspecies was once considered a common breeder in coastal southern California. However, this subspecies has declined drastically due to a loss of breeding habitat and nest parasitism by the brown-headed cowbird (*Molothrus ater*). This subspecies breeds in southern California, southern Nevada, southern Utah, Arizona, New Mexico, and western Texas (USFWS 1995). This species occurs in riparian habitats along rivers, streams, or other wetlands where dense growths of willows, arrowweed (*Pluchea* sp.), tamarisk (*Tamarix* sp.), or other plants are present, often with a scattered overstory of cottonwood (*Populus* sp.) (USFWS 1995). The project site provides potentially suitable nesting habitat for this species. Although migrants have been observed on the project site, no nesting or territorial willow flycatchers have been observed on the project site.

On July 22, 1997, the USFWS published a final critical habitat for this species. Approximately 160.7 river km (99.8 river miles) in Kern, Riverside, San Bernardino, and San Diego counties were designated for the southwestern willow flycatcher. The study area is not located in the designated critical habitat area for this species.

Loggerhead Shrike (Lanius Iudovicianus)

Loggerhead shrike is a California Species of Special Concern. This species is a fairly common resident of lowlands and foothills in southern California. Shrikes inhabit grasslands and other dry, open habitats. They can often be found perched on fences and posts from which prey items (large insects, small mammals, lizards) can be seen. This species is threatened by habitat loss and pesticides (Ehrlich et al. 1988). This species was observed on UCI in 1995 (UCI Office of Campus and Environmental Planning 1995). The project site provides suitable foraging and nesting habitat for this species. The loggerhead shrike has been observed on the project site with one breeding pair present since 2000.

Least Bell's Vireo (Vireo bellii pusillus)

Least Bell's vireo is a federally and state-listed Endangered species. This vireo was formerly considered to be a common breeder in riparian habitats throughout the Central Valley and other low-elevation river systems in California and Baja California, Mexico (Franzreb 1989). It is now considered to be a rare and local summer resident (Garrett and Dunn 1981), although there have been some regional population increases (Hamilton and Willick 1996). The least Bell's vireo breeds primarily in riparian habitats dominated by willows with dense understory vegetation (USFWS 1986). A dense shrub layer two-to-ten feet above ground is the most important habitat characteristic for this species (Goldwasser 1981; Franzreb 1989). The least Bell's vireo was first detected on the project site in 1998 (four pairs) and has been present during BonTerra Consulting surveys conducted in 2000 (four pairs), 2001 (nine pairs), 2002 (five pairs), 2003 (four pairs), 2004 (six pairs), and 2005 (mid-season results indicate ten territories and most with pairs).

On February 2, 1994, the USFWS published a final critical habitat for the least Bell's vireo, designating approximately 15,200 hectares (37,560 acres) of land in Santa Barbara, Ventura,

Los Angeles, San Bernardino, Riverside, and San Diego counties, California (USFWS 1994). The study area is located in the designated critical habitat area for this species.

California Horned Lark (Eremophila alpestris actia)

California horned lark is a California Species of Special Concern. The California horned lark is found along the coast of northern California, in the San Joaquin Valley, in the coast ranges south of San Francisco Bay, and in southern California west of the deserts. In southern California, this subspecies is a fairly common breeding resident in grasslands and other dry, open habitats. During the winter season, other subspecies occur in southern California and the horned lark, including all subspecies, can be locally common in the region. This species is threatened by loss of habitat due to agriculture and development. California horned lark has been observed on the project site.

San Diego Cactus Wren (Campylorhynchus brunneicapillus sandiegensis)

San Diego cactus wren is a California Species of Special Concern. This subspecies was proposed by Rea and Weaver (1990) and apparently has a coastal range that extends through San Diego County to southern Orange County. Some authorities consider the taxonomic status of cactus wrens in the southwestern U.S. to be uncertain (Proudfoot et al. 2000). Coastal populations of the cactus wren are found in southern California from San Diego County north to Ventura County (Garrett and Dunn 1981) and are declining due to loss of habitat. Except for the Banning Pass area west of Palm Springs, the coastal populations of cactus wren appear to be isolated from interior populations. On the coastal slope of southern California, cactus wrens inhabit coastal sage scrub and alluvial sage scrub habitats that have sufficient amounts of prickly pear cactus and/or cholla. The project site provides suitable habitat for this species. Although six pairs were identified during the 1992 surveys, this species were not observed during 1998, 2000, 2002, 2003, 2004, and 2005 surveys.

Coastal California Gnatcatcher (Polioptila californica californica)

Coastal California gnatcatcher is a federally listed Threatened species and a California Species of Special Concern. This species occurs in most of Baja California, Mexico's arid regions, but is extremely localized in the United States where it predominantly occurs in coastal regions of highly urbanized Los Angeles, Orange, Riverside, and San Diego counties (Atwood 1992). In California, this species is an obligate resident of several distinct subassociations of the coastal sage scrub vegetation type. Brood parasitism by brown-headed cowbirds and loss of habitat to urban development have been cited as causes of the coastal California gnatcatcher's population decline (Unitt 1984; Atwood 1990). The coastal California gnatcatcher has been observed on the project site during previous focused surveys in 1992 (ten pairs) and 1998 (six pairs) and during BonTerra Consulting surveys in 2003 (four to five pairs), 2004 (two pairs), 2005 (two pairs), and 2006 (7 pairs and 1 individual). The survey results indicate a natural cycle in the species population on the project site and within the region.

On April 24, 2003, the USFWS published a proposed rule to designate 495,795 acres of land as critical habitat for the coastal California gnatcatcher. Until the final rule is published, the previous critical habitat designation (USFWS 2000) is in effect. These lands encompass portions of Ventura, Los Angeles, San Bernardino, Orange, Riverside, and San Diego counties in California. The study area is located within areas designated as critical habitat of both the 2000 critical habitat and the 2003 proposed critical habitat, although the 2003 proposed critical habitat covers a reduced area.

Yellow Warbler (Dendroica petechia brewsteri)

Yellow warbler is a California Species of Special Concern. The yellow warbler is one of the most widespread and abundant warblers in North America, but populations in the west have been affected by overgrazing of riparian habitats (Lowther et al. 1999). The breeding range of the yellow warbler is along the coast from northwestern Washington south to western Baja California (Dunn and Garrett 1997). Breeding habitat for this subspecies consists of riparian woodlands dominated by willows (Dunn and Garrett 1997). The breeding population in California has declined due to loss of habitat and brood parasitism by the brown-headed cowbird (Remsen 1978). The project site provides suitable nesting habitat for this species. Several pairs have nested on the project site during recent surveys (2000 to 2004).

Yellow-Breasted Chat (Icteria virens)

Yellow-breasted chat is a California Species of Special Concern. This species occurs as an uncommon and local summer resident in southern California along the coast and in the deserts (Garret and Dunn 1981). This large warbler was once a fairly common summer resident in riparian woodlands throughout California, but is now much reduced in numbers, especially in southern California (Remsen 1978). For nesting, this species requires dense, brushy tangles near water and riparian woodlands supporting a thick understory. This species is threatened by loss of habitat and possibly nest parasitism by the brown-headed cowbird (Remsen 1978). The project site provides suitable nesting habitat for this species. Several pairs have nested on the project site during recent surveys (2000 to 2004).

Southern California Rufous-Crowned Sparrow (Aimophila ruficeps canescens)

Southern California rufous-crowned sparrow is a California Species of Special Concern. This sparrow is a year-round resident on the coastal slopes of the Transverse and Peninsular Mountain Ranges from northwestern Los Angeles County south to northwestern Baja California (Collins 1999). Garrett and Dunn (1981) considered this sparrow to be fairly common in suitable habitat. The southern California rufous-crowned sparrow prefers slopes, often steep and rocky, with sparse brush, especially coastal sage scrub species, intermixed with grasses (Garrett and Dunn 1981). It is a difficult bird to observe due to its shy, secretive habits and the habitat it occupies (Collins 1999). The project site provides suitable habitat for this species. The southern California rufous-crowned sparrow has been observed in small numbers on the project site during recent surveys (2000 to 2004). Approximately 12 to 15 pairs were estimated to be present on the project site during previous surveys (1992 to 1998).

Bell's Sage Sparrow (Amphispiza belli belli)

Bell's sage sparrow is a California Species of Special Concern. This coastal subspecies *A. b. bellii* is an uncommon-to-fairly-common local resident in the interior foothills of coastal southern California. The Bell's sage sparrow breeds in low, dense chamise chaparral and in dry scrub vegetation types, often with stands of cactus (Garrett and Dunn 1981). This species is threatened by loss of habitat due to development and likely nest parasitism by the brownheaded cowbird (Ehrlich et al. 1988). This species has not been observed on the project site during surveys where its detection is considered likely. The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the Bell's sage sparrow to occur on the project site, but it is considered limited due to the lack of detection during surveys.

Tricolored Blackbird (Agelaius tricolor)

Tricolored blackbird is a California Species of Special Concern. Primarily a California species, the tricolored blackbird is highly gregarious and nests in dense colonies that have been

estimated to be more than 200,000 birds (Beedy and Hamilton 1999). Typical breeding colonies have been located in freshwater marshes dominated by cattails or bulrushes, although breeding habitat can also include diverse upland habitats and agricultural areas (Beedy and Hamilton 1999). Recent breeding colonies have been located in a variety of vegetation types including Himalayan blackberries (*Rubus discolor*), giant cane (*Arundo donax*), safflower (*Carthamus tinctorius*), black mustard (*Brassica nigra*), stinging nettles (*Urtica dioica*), tamarisk (*Tamarix* spp.), grainfields, riparian scrublands, and various forests (Beedy and Hamilton 1999). This is an increasingly rare and local breeder in Orange County. The project site provides suitable foraging and potentially suitable nesting habitat for this species. The tricolored blackbird was observed during surveys, including a flock of 125 that were observed foraging at the active landfill portion of the project site on November 14, 2002.

Mammals

Pallid Bat (Antrozous pallidus)

Pallid bat is a California Species of Special Concern that most commonly occurs in mixed oak and grassland habitats. This large bat roosts in rock crevices. The pallid bat is very sensitive to disturbance at its roosting sites (CDFG 2001). This species is known to occur near Cañada Gobernadora and San Juan Creek (CDFG 2001). The project site provides suitable foraging habitat but a very limited amount of potentially suitable roosting habitat for this species. Therefore, there is the potential for the pallid bat to forage on the project site, but it is not expected to roost on the project site.

Pale Big-Eared Bat (Corynorhinus townsendii pallescens)

Pale big-eared bat occurs throughout California and is a federal and California Species of Special Concern. In the southern portion of the state, the subspecies *C. t. pallescens* occupies a variety of vegetation types, including oak woodlands, arid deserts, grasslands, and high-elevation forests and meadows (Hall 1981). Known roosting sites in California include mines, caves, and buildings. The project site provides suitable foraging habitat but a limited amount of potentially suitable roosting habitat for this species. Therefore, there is the potential for the pale big-eared bat to forage on the project site, but it is not expected to roost on the project site.

Yuma Myotis (Myotis yumanensis)

Yuma myotis is a federal Species of Concern and a California Species of Special Concern. This relatively small bat occurs statewide and is closely associated with water and wooded canyon bottoms throughout its range. Caves and old buildings are preferred roosting habitats, with roosts numbering up to 2,000 individuals. The project site provides suitable foraging habitat, but a limited amount of potentially suitable roosting habitat for this species. Therefore, there is the potential for the Yuma myotis to forage on the project site, but it is not expected to roost on the project site.

Western Mastiff Bat (Eumops perotis)

Western mastiff bat is a California Species of Special Concern. The subspecies that occurs in southern California is California mastiff bat (*Eumops perotis californicus*). Western mastiff bat is a very wide ranging and high flying insectivore that typically forages in open areas with high cliffs. This species roosts in small colonies in crevices on cliff faces. It occurs in the southeastern San Joaquin Valley and Coastal Ranges from Monterey County southward through southern California, and from the coast eastward to the Colorado Desert (Zeiner et al. 1990). Western mastiff bat is found in many open semi-arid to arid habitats including conifer and deciduous woodlands, coastal scrub, grasslands, palm oases, chaparral, desert scrub, and

urban (Zeiner et al. 1990). Threats to this species include loss of habitat due to development, drainage of marshes, and conversion of land to agriculture (Williams 1986). The project site provides potentially suitable foraging habitat, but no suitable roosting habitat for this species. Therefore, there is the potential for the western mastiff bat to forage on the project site, but it is considered limited due to the scarcity of suitable roosting habitat in the region.

Pocketed Free-Tailed Bat (Nyctinomops fermorosaccus)

Pocketed free-tailed bat is a California Species of Special Concern. This species is known to occur in areas with ponds, streams, or arid deserts that provide suitable foraging habitats for this species. This species roosts in rock crevices, caverns, or buildings. This species occurs primarily in pinyon-juniper woodlands, desert scrub, desert succulent scrub, desert riparian, desert wash, alkali desert scrub, Joshua tree, and palm oases. The project site provides suitable foraging habitat, but a limited amount of potentially suitable roosting habitat for this species. Therefore, there is the potential for the pocketed free-tailed bat to forage on the project site, but it is not expected to roost on the project site.

Big Free-Tailed Bat (Nyctinomops macrotis)

Big free-tailed bat is a California Species of Special Concern. This species feeds primarily on moths caught while flying over water sources in suitable habitat in the southwestern United States. This species prefers rugged, rocky terrain, but wanders widely during autumn, and roosts in small colonies in rocky crevices of high cliffs. Suitable foraging habitat for this species is present in the study area. The project site provides potentially suitable foraging habitat, but no suitable roosting habitat for this species. Therefore, there is the potential for the big free-tailed bat to occur on the project site, but it is considered limited due to the limited amount of potentially suitable roosting habitat in the region.

San Diego Black-Tailed Jackrabbit (Lepus californicus bennettii)

San Diego black-tailed jackrabbit is a California Species of Special Concern. The San Diego black-tailed jackrabbit is restricted to the coastal slopes of southern California from Santa Barbara County to northwestern Baja California, Mexico (Hall and Kelson 1959). This species prefers relatively open areas with sparse shrub cover. Although this species is considered widespread and common, the subspecies has declined due to loss of habitat. This species is diurnal but generally more active at dawn and dusk or under low-light conditions (e.g., cloudy) (Whitaker 1980, Zeiner et al. 1990). The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the San Diego black-tailed jackrabbit to occur on the project site, but it is considered limited due to the lack of detection during surveys.

Northwestern San Diego Pocket Mouse (Chaetodipus fallax fallax)

Northwestern San Diego pocket mouse is a California Species of Special Concern. This species is a common resident of sandy herbaceous areas, usually in association with rocks or coarse gravel. It occurs mainly in arid coastal and desert border areas in San Diego, Riverside, and San Bernardino counties in coastal scrub, chamise-redshank chaparral, mixed chaparral, sagebrush, desert wash, desert scrub, desert succulent shrub, pinyon-juniper, and annual grassland (Zeiner et al. 1990). The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the Northwestern San Diego pocket mouse to occur on the project site.

Pacific Pocket Mouse (Perognathus longimembris)

Pacific pocket mouse is a federally listed Endangered species and a California Species of Special Concern. This species was historically known to inhabit the narrow coastal plains from the Mexican border north to El Segundo, Los Angeles County (CDFG 2003). This species seems to prefer soils of fine alluvial sands near the ocean (CDFG 2003). In the project region, this species is known to occur at the Dana Point Headlands and at two populations near San Mateo Creek on Camp Pendleton (CDFG 2001). All historic locations of this species are known to occur within 2.5 miles of the coast. The project site is outside the known range for the species, and therefore the Pacific pocket mouse is not expected to occur on the project site.

San Diego Desert Woodrat (Neotoma lepida intermedia)

San Diego desert woodrat is a California Species of Special Concern. This subspecies occurs along the coastal slopes of southern California from San Luis Obispo County to northwestern Baja California (Hall and Kelson 1959). This species occupies areas with sparse vegetation, especially areas comprised of cactus and other thorny plants. They build and occupy stick homes that are situated over a burrow, rock crevice, or base of tree or bush (Zeiner et al. 1990). These distinctive homes provide protection and food storage (Whitaker 1980). The project site provides suitable habitat for this species. Therefore, there is the potential for the San Diego desert woodrat to occur on the project site.

Southern Grasshopper Mouse (Onychomys torridus ramona)

Southern grasshopper mouse is a federal Species of Concern and a state Species of Special Concern. It is a territorial, predatory rodent of grassland and sparse scrub vegetation types that prefers sandy soils and has been found to occur from Los Angeles County to northwestern Baja California. The project site provides potentially suitable habitat for this species. Therefore, there is the potential for the southern grasshopper mouse to occur on the project site.

5.5.2 SIGNIFICANCE CRITERIA

The potential significance of environmental impacts on biological resources has been assessed using impact significance criteria that mirror the policy contained in CEQA, Section 21001(c) of the California *Public Resources Code*. Accordingly, the State Legislature has established it to be the policy of the state to:

Prevent the elimination of fish or wildlife species due to man's activities, ensure that fish and wildlife populations do not drop below self-perpetuating levels, and preserve for future generations representations of all plant and animal communities...

Determining whether a project may have a significant effect or impact plays a critical role in the CEQA process. According to CEQA Section 15064.7, Thresholds of Significance, each public agency is encouraged to develop and adopt (by ordinance, resolution, rule, or regulation) thresholds of significance that the agency uses in the determination of the significance of environmental effects. A Significance Threshold is a quantitative, qualitative, or performance level of a particular environmental effect that would normally be determined to be significant by the agency if the threshold is exceeded.

Appendix G of the CEQA Guidelines is specific in addressing biological resources and encompasses a broader range of resources to be considered, including: candidate, sensitive, or special status species; riparian habitat or other sensitive natural communities; federally protected wetlands; fish and wildlife movement corridors; local policies or ordinances protecting biological resources; and adopted habitat conservation plans. These factors are considered through the checklist of questions answered during the Initial Study process that is used to determine the appropriate type of environmental documentation for a project (Negative Declaration, Mitigated Negative Declaration, or EIR). Because these questions are derived from standards in other laws, regulations, and other commonly used thresholds, these standards have been used as the basis for defining significance thresholds in this SEIR.

For the purpose of this analysis, impacts on biological resources are considered significant (before consideration of offsetting mitigation measures) if one or more of the following conditions would result from implementation of the proposed project:

- If the project has a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations, or by the CDFG or USFWS (CEQA Guidelines, Appendix G, IV[a]),
- If the project has a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the CDFG or USFWS (CEQA Guidelines, Appendix G, IV[b]),
- If the project has a substantial adverse effect on federally protected wetlands as defined by Section 404 of the Clean Water Act (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means (CEQA Guidelines, Appendix G, IV[c]),
- If the project interferes substantially with the movement of any native or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impedes the use of native wildlife nursery sites (CEQA Guidelines, Appendix G, IV[d]),
- If the project conflicts with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance (CEQA Guidelines, Appendix G, IV[e]),
- If the project conflicts with the provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan (CEQA Guidelines, Appendix G, IV[f]).

An evaluation of whether an impact on biological resources would result in a "substantial adverse effect" must consider both the resource itself and how that resource fits into a regional context. For the proposed project, the regional setting of the project includes the following United States Geological Survey (USGS) quads that cover the San Diego Creek Watershed and that were queried in the records search: Cañada Gobernadora and San Clemente.

5.5.3 POTENTIAL IMPACTS

Potential Impacts

This section will assess project impacts based on the proposed limits of disturbance associated with the ultimate buildout of the Prima Deshecha Landfill above and beyond those limits identified in EIR 575 to vegetation, special status plant and wildlife species, and state and federal jurisdictional resources (i.e., wetlands and waters) and compliance with the NCCP/HCP and SAMP programs.

Refuse disposal activities are currently underway in Zone 1 and will continue until approximately year 2019 at which time these activities will shift to Zone 4 and continue until landfill closure in approximately year 2067. It should be noted that site preparation including landslide remediation/stabilization necessary for refuse disposal within Zone 4 will begin prior to 2019. In addition, the project will be implemented in phases within each zone. Therefore, impacts to biological and jurisdictional resources will likely occur throughout the life of the landfill over an approximate 60-year period.

Vegetation Types Direct Impacts

Table 5.5-4 below identifies the acres of impact in the 2001 GDP. Based upon the limits of disturbance associated with the ultimate buildout of the landfill that are intended to better represent the actual effects of landslide remediation and stabilization and ongoing operation and management of the landfill, the proposed project will result in increases in impacts to grasslands, coastal sage scrub, chaparral, riparian, wet meadows and marshes, and other areas.

Table 5.5-5 provides a comparison of impacts by vegetation type between the 2001 Prima Deshecha GDP (as shown in EIR No. 575) and the Proposed Project for Zones 1 and 4.

Impacts to vegetation resources by zone, phase, and scheduled implementation by fiscal year are depicted in Table 5.5-45. The most significant vegetation resources removals will occur within Zone 4 during Phases A through D as part of the initial landslide remediation and stabilization activities necessary for refuse disposal and operation and management of the landfill.

Grasslands

Grasslands including annual grassland, needlegrass grassland, and ruderal grassland will be impacted by implementation of the Proposed Project as shown in Table 5.5-4. Impacts on needlegrass grassland are considered significant due to the relative scarcity of this vegetation type in southern California. This impact would be mitigated to a level considered less than significant by implementation of previously adopted MM 4.5-1.

Impacts on annual grassland and needlegrass grassland would be considered adverse and significant. However, impacts to needlegrass grassland will be mitigated to a less-than-significant level through the implementation of the proposed Pre-mitigation Plan.

Coastal Sage Scrub

Any impact on coastal sage scrub including coyote brush, sagebrush scrub, black sage scrub, sagebrush-grassland, buckwheat-grassland, and mixed sage scrub grassland (as shown in Table 5.5-4) would be considered significant because these vegetation types are considered to be special status due to their decline in the region and their potential to support special status plant and wildlife species, such as the coastal California gnatcatcher. The project could also involve temporary impacts to created coastal sage scrub resulting from landslide stabilization and subsequent remediation and/or installation of landfill-related facilities. In these cases, the disturbed portion of the site will be hydroseeded with the appropriate native plant species and will be maintained for a period not to exceed three years in accordance with the Coastal Sage Scrub (CSS) and Native Grassland (NG) Mitigation Plan (CSS/NG Mitigation Plan) approved June 17, 2002, by the USFWS and CDFG (BonTerra Consulting 2002a). In addition, a Premitigation Plan has been prepared to address impacts to CSS, native grassland, and riparian resources associated with the ultimate buildout of Prima. Implementation of the Pre-mitigation

TABLE 5.5-4
VEGETATION IMPACTS BY PHASE FOR THE PROPOSED PROJECT
(AS OF 2004) ^a

						ZON	E 1			ZONE 4									
		FY	03/04	05/06	05/06	06/07	08/10	10/13	18/19	07/10	15/19	20/25	27/37	39/44	45/47	46/52	55/57	58/60	
Code	Vegetative Types	Phase	A2	B1	C2	C3	D1	D2	WMU1	SH	А	В	С	D	Е	F	G	Н	Total
12.1	Open Water								0.22										0.22
15.3	Non-Urban Commercial/Industrial/Institutional			1.45	0.55		1.22	4.14	4.21		5.66				0.32		0.08		17.63
15.5	Parks and Ornamental Plantings							0.73	1.71		0.16								2.60
15.5/7.2	Parks and Ornamental Plantings/Southern Willow Sc	rub							0.21										0.21
16.1	Cleared or Graded		0.30	2.03	1.16	4.25	6.10	5.91	237.91		20.91	2.43	3.74	0.91	1.40	1.10	0.51	1.18	289.84
16.2	Other Developed Areas (Erosion-Control Plantings)							0.02	5.75										5.77
2.3.10	Mixed Sage Scrub								0.54		4.72	4.25		6.30		6.87	0.01		22.69
2.3.10-R	Mixed Sage Scrub (Revegetation)								<29.34>										<29.34>
2.3.12	Sagebrush-Covote Brush							0.63			1.36		0.28						2.27
2.3.2	California Sagebrush-Orange Bush Monkeyflower Sc	crub							0.59			0.08	18.65						19.32
2.3.4	Black Sage Scrub										1.61	1.06				2.43			5.10
2.3.6	Sagebrush Scrub								0.21		-	3.91		0.20					4.32
2.3.7-R	Buckwheat Scrub (Revegetation)										3.53							\rightarrow	3.53
2.3.9	Covote Brush						0.25	0.66	1.71		1.07	3.60	7.59	2.16	0.06				17.10
2.3.9/7.3	Covote Brush/Mule Fat Scrub												2.87						2.87
2.8.1	Sagebrush-Grassland																	0.11	0.11
2.8.2	Buckwheat-Grassland								0.67										0.67
285	Mixed Sage Scrub-Grassland		0 13				0.33	2 60	1 53		4 02	7 16	12 79	4 67	0 15	0.30			33.68
3 12	Toyon-Sumac Chaparral		0.10				0.00	0.31			31.82	2 56	0.35	0.46	00	9.57			45.07
4 1	Annual Grassland		2 67			9 23	12 01	22 21	36 80		65.51	60.34	107 74	61 65	56 69	23 76	21.38	15 76	495.75
42	Elymus Grassland					0.20			00.00		00.01	4 28	0.85	000	00.00				5.13
4.3	Southern Coastal Needlegrass Grassland							0.05	0 24		0.37	0.54	0.81	11 76	1 21	0 75	0.65	0.15	16.53
1.0	Southern Coastal Needlegrass Grassland/Mexican F	Iderberry						0.00	0.21		0.07	0.01	0.01	11.70		0.10	0.00	0.10	
4.3/8.4	Woodland	liaonooniy													1.52			ļ	1.52
46	Buderal		0.09			0.20		4 47	5 20		6 65		0.81			0.09			17.51
47	Mixed Perennial Grassland		0.00			0.20			0.20		0.00		0.01	0 72	1 37	0.00			2.09
49	Castor Bean/Fennel							0.57			1 74			0					2.31
6.3/7.2-R	Riparian Herb/Southern Willow Scrub (Revegetation)							0.07	<3.90>		1.7 1								<3.90>
7 1	Riparian Herb	/						0.05	0.56		2 49	0.24		0.04					3.38
7 12	Mexican Elderberry Rinarian Scrub							0.00	0.00			0	1 05	0.01					1.05
7 1-R	Riparian Herb (Revegetation)							<0.01>	<1 82>				1.00						<1.83>
72	Southern Willow Scrub							0.08	0.11		0.27	0 44							0.90
72/239	Southern Willow Scrub/Covote Brush							0.00	0.50		0.27	0.10							0.00
7.2-R	Southern Willow Scrub (Revegetation)								<7 23>			0.10							<7 23>
73	Mule Fat Scrub								0.07		0.82								0.89
73/239	Mule Fat Scrub/Covote Brush								0.07		0.02		0.76						0.05
7 3/2 3 9-R	Mule Fat Scrub/Covote Brush (Revegetation)								<0.00>		0.15		0.70						<0.00
7 4	Southern Sycamore Rinarian Woodland								-0.002		3 31	0.78							4 00
84	Mexican Elderberry Woodland										0.18	0.70	0.00	0 00			0.20	0.65	1 30
U.7		Total	3 10	3 4 9	1 71	13 68	10 01	42 43	208 74		156 33	91 77	158 39	80.09	62 72	44 87	22 92	17.85	1 026 04
During the 2	2005 vegetation surveys, areas that were identified as "R	Revegetation	" or "R	include	d slope	s hydros	eeded wi	th CSS sr	ecies for e	rosion-co	ntrol purr	oses o	nlv. These	e reveae	etated a	reas we	re not ir	istalled	as habitat

During the 2005 vegetation surveys, areas that were identified as "Revegetation" or "R" included slopes hydroseeded with CSS species for erosion-control purposes only. These revegetated areas were not installed as habita restoration to offset impacts associated with any landfill-related impacts. Therefore, loss of these revegetated areas will not be counted in the CSS impact analysis contained in this document. SH – Scale House

¹ Based on preliminary, conceptual phasing limits subject to change with more detailed design.

TABLE 5.5-5 VEGETATION IMPACT DIFFERENCES

Vegetation Type	2001 GDP EIR 575	Amendment No. 2 2001 GDP SEIR 597	Difference
California Sagebrush-Orange Bush Monkeyflower Scrub (2.3.2)	0.00	19.32	19.32
Black Sage Scrub (2.3.4)	22.47	5.10	-17.37
Sagebrush Scrub (2.3.6)	23.23	4.32	-18.91
Buckwheat Scrub (2.3.7-R)	0.00	3.53	3.53
Covote Bush Scrub (2.3.9)	10.57	17.10	6.53
Covote Bush /Mule Fat Scrub (2.3.9/7.3)	0.00	2 87	2.87
Mixed Sage Scrub (2.3.10)	0.00	22.69	22.69
Mixed Sage Scrub (2.3.10-R)	0.00	<29.34>	0.00
Sage Scrub-Covote Bush (2.3.12)	0.00	2.27	2.27
Sagebrush-Grassland (2.8.1)	0.00	0.11	0.11
Buckwheat–Grassland (2.8.2)	2.55	0.67	-1.88
Mixed Sage Scrub-Grassland (2.8.5)	24 23	33.68	9.45
Coastal Sage Scrub Subtotal	83.05	111.66	28.61
Toyon Sumac Chaparral (3.12)	34.95	45.07	10.12
Chaparral Subtotal	34.95	45.07	10.12
Annual Grasslands (4.1)	400 73	495 75	95.02
Flymus Grassland (4.2)	0.00	5 13	5 13
Southern Needlegrass Grasslands (4.3)	5.61	16.53	10.92
Southern Needlegrass Grasslands/Mexican Elderberry (4.3/8.4)	0.00	1.52	1.52
Ruderal Grasslands (4.6)	45.48	17.51	-27.97
Mixed Perenial Grassland (4.7)	0.00	2 09	2 09
Caster bean/Fennel (4.9)	0.00	2.31	2.31
Grasslands Subtotal	451.82	540.84	89.02
Alkali Meadows (5.2)	0.00	0.00	0.00
Seasonal Wetlands Subtotal	0.00	0.00	0.00
Freshwater and Alkali Marsh (6.3/6.4)	0.41	0.00	-0.41
Riparian Herb/Southern Willow Scrub Revegetation (6.3/7.2-R)	0.00	<3.90>	0.00
Marsh Subtotal	0.41	0.00	-0.41
Herbaceous Riparian (7.1)	2.06	3.38	1.32
Riparian Herb (Revegetation) (7.1-R)	0.00	<1.83>	0.00
Mexican Elderberry Riparian Scrub (7.12)	0.00	1.05	1.05
Southern Willow Scrub (7.2)	3.15	0.90	-2.25
Southern Willow Scrub/Coyote Bush (7.2/2.3.9)	0.00	0.60	0.60
Southern Willow Scrub/Coyote Bush (Revegetation) (7.2/2.3.9)	0.00	<0.09>	0.00
Southern Willow Scrub (Revegetation) (7.2-R)	0.00	<7.23>	0.00
Mule Fat Scrub (7.3)	5.22	0.89	-4.33
Mule Fat Scrub/Coyote Bush (7.3/2.3.9)	0.00	0.89	0.89
Southern Sycamore Riparian Woodland (7.4)	0.00	4.09	4.09
Riparian Subtotal	10.43	11.80	1.37
Mexican Elderberry Woodland (8.4)	0.47	1.30	0.83
Woodland Subtotal	0.47	1.30	0.83
Open Water (12.1)	0.00	0.22	0.22
Non-Urban Institutional (15.3)	5.11	17.63	12.52
Ornamental Plantings (15.5)	4.94	2.81	-2.13
Graded (16.1)	199.86	289.84	89.98
Cleared (16.2)	0.00	5.77	5.77
Disturbed/Urban Subtotal	209.91	316.27	106.36
TOTAL IMPACTS	791.04	1,026.94	235.90

Vegetation types identified as "Revegetation" or "R" that are located in Zone 1 were created as part of the landslide remediation project and will not be impacted by this project. Other areas designated as "Revegetation" or "R" (such as mixed sage scrub) within Zone 4 were created as part of erosion control measures and will not be counted as loss of CSS.

Chaparral and Woodland

Toyon-sumac chaparral and Mexican elderberry woodland will be impacted by project implementation as shown in Table 5.5-4. Impacts on these vegetation types would be considered adverse but not significant because they are abundant and considered to have a low biological value.

Riparian and Wet Meadows and Marshes

Impacts on riparian vegetation types and wet meadows and marshes (including riparian herb, southern willow scrub, mule fat scrub, sycamore riparian woodland, Mexican elderberry riparian scrub, and alkali meadow) are shown in Table 5.5-4. Any grading or filling of these areas may require authorization from the USACE and the CDFG. A Jurisdictional Delineation was conducted for the Proposed Project and is included as Appendix F. The delineation states that the Proposed Project would permanently impact approximately 3.42 acres of USACE jurisdictional areas. A total of 9.81 acres of CDFG jurisdictional areas would be impacted. These impacts would be considered significant. Implementation of previously adopted MM 4.5-3 would reduce the impacts to less-than-significant levels.

Other Areas

Other areas (including parks and ornamental plantings, non-urban commercial/industrial/ institutional, and cleared or graded areas) will be impacted by project implementation as shown in Table 5.5-4. Impacts on these areas would be not be considered adverse due to their low biological value.

Wildlife Direct Impacts

To assess impacts on wildlife, the total impact on vegetation types that provide habitat for wildlife was evaluated. Exhibit 5.5-1 illustrates the vegetation types (i.e., wildlife habitat) that would be impacted as a result of project buildout. The following discussion of wildlife impacts focuses on the common species occurring on the project site.

General Habitat Loss, Wildlife Loss, and Wildlife Displacement

Implementation of the proposed project would result in the loss of native and non-native habitats that provide nesting, roosting, foraging, and denning opportunities for a variety of wildlife species. Removing or altering habitats within the project site would result in the loss of small mammals, reptiles, amphibians, and other wildlife of slow mobility that live within the project's direct impact area. More mobile wildlife species now using the project site would be forced to move into remaining areas of open space, consequently increasing competition for available resources in those areas. This situation would result in losing individuals of the wildlife population that cannot successfully compete. These impacts on wildlife species as a result of habitat loss, direct wildlife loss, and wildlife displacement would be considered adverse, though less than significant, because they would not substantially reduce wildlife populations in the region.

Special Status Species Direct Impacts

Special Status Plant Species

Proposed landfill operations within undeveloped portions of Zones 1 and 4 will result in the removal/disturbance of areas within the landfill containing thread-leaved brodiaea, vernal barley, small-flowered morning glory, and paniculate tarplant. See Exhibit 5.5-1 for locations of special

status plant species. The Proposed Project Pre-mitigation Plan (Exhibit 4.3-7) contains proposed and potential restoration sites for these species within areas of appropriate future habitats. These areas will be located outside existing and future landfill operations, but within the PDL (project) boundary.

Thread-Leaved Brodiaea (Brodiaea filifolia)

Thread-leaved brodiaea is listed as federally and state-Endangered. A small population of this plant species is located in the north-central portion of Zone 1 immediately adjacent to and south of the most northerly landfill maintenance road and boundary of the landfill. Disturbance to this portion of the landfill and subsequent impacts to this species are required to stabilize landslide conditions and provide the appropriate drainage for refuse disposal and long-term landfill operation and management. Landslide remediation activities are scheduled to occur through Fiscal Year (FY) 2005–2006. Impacts to this species were addressed in EIR 575 and provisions for mitigation were included in MM 4.5-4a of that document and additional provisions in Mitigation Measure MM 3.5-1 of this document. In compliance with this mitigation measure, the IWMD formally submitted a letter to the CDFG requesting authorization (in accordance with the provisions of Section 2081[b] and California Fish and Game Code) to collect seed and propagate this plant species and transplant plants and soils containing these plants to mitigation sites located outside any existing and/or future landfill operations area. The IWMD has also transmitted a letter to the USFWS commenting on the critical habitat designation for this species and requesting that critical habitat designations be eliminated within the Prima Deshecha Landfill. In response, the USFWS removed the designation within the Prima Deshecha Landfill in the December 13, 2005, Thread-leaved Brodiaea Critical Habitat approved (Federal Register 70-FR7380).

Vernal Barley (Hordeum intercedens)

Vernal barley is a CNPS List 3 species and was found in 9 locations within the landfill totaling approximately 1,700 individual plants. Impacts to this species were addressed in EIR 575 and provisions for mitigation were identified in MM 4.5-4a of that document. In addition, SEIR 597 contains a Pre-mitigation Plan as a project component that will include provisions for on-site seed collection for incorporation into proposed restoration areas that are outside existing and future landfilling operation areas but that are within the PDL (project) boundary.

Small-Flowered Morning Glory (Convolvulus simulans)

Small-flowered morning glory is a CNPS List 4 species and was observed in 3 locations within the project site containing approximately 1,200 individual plants.

Impacts to this species were not addressed in EIR 575. SEIR 597 contains a Pre-mitigation Plan as a project component or Project Design Feature that will include provisions for seed collection or purchase of a seed mix containing this species for incorporation into proposed restoration areas that are outside existing and future landfilling operation areas but that are within the PDL (project) boundary.

Paniculate Tarplant (Deinandra paniculata)

Paniculate tarplant is a CNPS List 4 plant species and was observed in 6 locations on the project site, primarily in the eastern portions of Zone 4, containing approximately 6,000 individual plants. Impacts to this species were addressed in EIR 575 and mitigation is identified in MM 4.5-9a. Seeds of these plant species will either be collected on site and/or purchased as part of a seed mix to be incorporated into the restoration and enhancement program located in areas that are outside existing and future landfilling operations but that are

within the PDL (project) boundary in accordance with the provisions of the Pre-mitigation Plan that was approved by the USFWS and the CDFG.

Special Status Wildlife Species

Invertebrates

San Diego fairy shrimp and Riverside fairy shrimp potentially occur in the abandoned cattle pond located in the northeastern portion of the project site. Although the occurrence of either species is considered unlikely, the required focused surveys would determine their absence or presence. If either fairy shrimp species is present, then landfill activities in this part of the project site may result in impacts that would be considered significant. These potentially significant impacts would be reduced to less-than-significant levels with implementation of additional Mitigation Measure MM 5.5-2.

Amphibians

Arroyo southwestern toad is not expected to occur on the project site due to lack of suitable habitat. Therefore, the proposed project would not impact this species, and no mitigation would be required.

The western spadefoot toad occurs in the abandoned cattle pond located in the northeastern portion of the project site within Zone 4. Proposed landfill activities in this part of the project site may result in the loss of this pond. Without appropriate breeding ponds, the western spadefoot would not be able to persist within the Prima Deshecha Cañada watershed. Although the western spadefoot toad is not listed as Endangered or Threatened, these potential impacts would meet the significance criteria in Section 15380 of CEQA. Potential impacts to the western spadefoot toad would be reduced to less-than-significant levels with implementation of additional Mitigation Measure MM 5.5-3.

Reptiles

Southwestern pond turtle is not expected to occur on the project site due to lack of suitable habitat. Therefore, the proposed project would not impact this species, and no mitigation would be required.

Coast horned lizard, Coronado skink, Belding's orange-throated whiptail, silvery legless lizard, coast patch-nosed snake, and northern red-diamond rattlesnake occur or potentially occur on the project site. These species primarily use the coastal sage scrub and riparian habitats on the site. As the Proposed Project impacts a small area of habitat relative to the availability of habitat in the region, impacts on these species would be considered adverse but not significant, and no mitigation would be required.

<u>Birds</u>

Swainson's hawk, American peregrine falcon, and southwestern willow flycatcher are Threatened and/or Endangered species that occur or potentially occur at the project site. Swainson's hawk has been observed twice on the project site. Since this hawk only occurs as a rare migrant in the region, the loss of potentially suitable foraging habitat is considered adverse but not significant, and no mitigation is required. American peregrine falcon has the potential to occur on the project site for foraging only, as the project site does not support suitable nesting habitat. The landfill activities attract lots of birds including gulls to the trash and, as a result, may occasionally attract this falcon to the project site as it preys on other birds. Therefore, the proposed project is not expected to adversely affect the American peregrine falcon and no mitigation is required. No nesting or territorial southwestern willow flycatchers have been detected during focused surveys for this species on the project site, although migrant willow flycatchers (subspecies undetermined) have been observed. Habitat supporting migrant willow flycatchers will not be impacted by the Proposed Project. Accordingly, the Proposed Project is not expected to adversely affect either migrant willow flycatcher or southwestern willow flycatcher and no mitigation is required.

A variety of other special status passerine bird species not listed as Threatened or Endangered by state or federal resources agencies have been observed or potentially occur on the project site. Potential impacts on these species would primarily consist of habitat loss. The impacts on Cooper's hawk, sharp-shinned hawk, golden eagle, ferruginous hawk, northern harrier, whitetailed kite, prairie falcon, long-eared owl, burrowing owl, loggerhead shrike, California horned lark, San Diego cactus wren, yellow warbler, yellow-breasted chat, southern California rufouscrowned sparrow, Bell's sage sparrow, and tricolored blackbird would be adverse but not substantial enough to be considered significant. Therefore, no mitigation would be required.

Least Bell's vireo is a summer breeding resident in the Prima Deshecha Cañada stream channel on the project site. Focused surveys have found four pairs in 1998, four pairs in 2000, nine pairs in 2001, five pairs in 2002, four pairs in 2003, six pairs in 2004, and mid-season survey results for 2005 indicate ten territories with most containing pairs. The landslide remediation project was permitted and riparian habitat was cleared on March 4–6, 2002, in preparation of construction. The channel has since been realigned and riparian revegetation replanted. In addition, the Basin B mitigation site was created to support riparian and marsh habitats. Future landfill activities within Zones 1 and 4 have the potential to impact up to four least Bell's vireo territories. These impacts would be considered significant, but would be reduced to a less-than-significant level by implementation of MM 4.5-3a through 3e, MM 4.5-5a, MM 4.5-8a through 8e, and MM 4.5-10a.

Coastal California gnatcatcher is a year-round resident of coastal sage scrub habitats on the project site. The survey results from 1992 to 2006 show that this species is showing natural fluctuation in population on the project site similar to the region. Implementation of the project has the potential to directly impact coastal California gnatcatcher territories through the loss of habitat. These impacts would be considered significant, but would be reduced to less-than-significant levels with implementation of MM 4.5-2a through 2c, MM 4.5-5a, MM 4.5-7a through 7c, and MM 4.5-10a through 10b.

Several special status and common raptor species have the potential to use the project site for foraging. The loss of approximately 540 acres of native and non-native vegetation on the project site would cumulatively contribute to the ongoing regional and local loss of foraging habitat for these species. This is considered an adverse, though not significant, impact. In addition, the white-tailed kite, northern harrier, and burrowing owl, as well as common raptor species (i.e., red-tailed hawk) have the potential to nest on the project site. Any impacts on an active raptor nest or burrow would be considered significant due to the highly sensitive nature of these species with the resource agencies and public. In addition, the loss of an active raptor nest would also be considered a violation of the California *Fish and Game Code*, Sections 3503, 3503.5, and 3513. This impact will be mitigated to a level considered less than significant by implementation of MMs 4.5-5b and 4.5-10a through 10b.

<u>Mammals</u>

Pacific pocket mouse and San Diego black-tailed jackrabbit are not expected to occur on the project site. The Pacific pocket mouse is not expected to occur because the project site is outside the known range for this species. San Diego black-tailed jackrabbit is not expected to

occur because it has not been observed during any surveys. Therefore, there would be no impacts on these two species as a result of the proposed project and no mitigation would be required.

Special status mammal species potentially present in the study area include pallid bat, pale bigeared bat, Yuma myotis, western mastiff bat, pocketed free-tailed bat, big free-tailed bat, northwestern San Diego pocket mouse, San Diego desert woodrat, and southern grasshopper mouse. The proposed project would result in the loss of potential habitat for these species. This impact would be adverse, but not substantial enough to be considered significant. Therefore, no mitigation would be required.

Wildlife Movement

Implementation of the proposed project would further fragment existing wildlife habitat and wildlife travel routes on the project site and the surrounding open space areas. This would result in reduced opportunities for genetic exchange between wildlife populations, especially those of limited mobility such as amphibians, reptiles, and small mammals. Most birds and larger mammal species that are more capable of crossing larger areas of habitat would be affected to a lesser extent. These impacts on wildlife due to habitat fragmentation would be considered adverse, though less than significant because they would not substantially reduce wildlife populations in the region.

The project site provides an important habitat linkage between the Talega Nature Preserve to the east and the open spaces of Forster Ranch to the west. Implementation of the proposed project has the potential to adversely affect movement to and from the native habitats located in these nearby open space areas. However, the open spaces set aside in Zone 3 of the project site will serve to facilitate continued wildlife movement on the south side of the project site. As a result, open spaces to the west of the project site would still be connected to open spaces to the east and south of the project site via this preserved habitat. Therefore, the long-term impacts to regional and local wildlife movement would be adverse, but not considered significant.

San Juan Creek/Western San Mateo Creek Special Area Management Plan (SAMP)

As previously noted, the project site is not located within the boundaries of the San Juan Creek/Western San Mateo Creek SAMP and will not have a direct impact on the implementation of that planning effort once approved. However, the IWMD is proposing mitigation solutions within the San Juan Creek portion of Ronald W. Caspers Wilderness Park to address impacts to the least Bell's vireo within the PDL. As such, proposals to perform habitat restoration (Avando [giant reed] eradication) within San Juan Creek would constitute a furtherance of the SAMP program.

Southern Subregion NCCP Guidelines

As previously noted, the project is located within the Southern Subregion NCCP/HCP study area. However, the guidelines are designed almost exclusively to address the physical features and biological resources found within the Rancho Mission Viejo properties. The IWMD intends to participate in the NCCP/HCP through the designation "Supplemental Open Space." The remainder of the site will be proposed to be designated "Development" (Exhibit 5.5-2).

5.5.4 MITIGATION MEASURES

Mitigation Measures

Previously Adopted Mitigation

The following mitigation measures are currently in place associated with the landfill component of the 2001 GDP, identified in EIR 575 (numerical designations are from EIR 575). All mitigation commitments contained within FEIR 575 and the 2001 GDP will apply to the Proposed Project.

Impact 4.5-1: Removal of needlegrass grassland will result from site clearing prior to construction of landfill improvements on the Prima Deshecha site.

MM 4.5-1: The restoration of needlegrass grasslands will be incorporated into the Conceptual Coastal Sage Scrub Mitigation Plan (described below in MM 4.5-2a through 2c), the IWMD will replace impacted needlegrass grassland at a 1:1 ratio.

Impact 4.5-2: Removal of coastal sage scrub will result from site clearing prior to construction of landfill improvements on the Prima Deshecha site.

- MM 4.5-2a: Prior to the removal of coastal sage scrub habitat resources including clearing, grubbing, mowing, discing, trenching, grading, fuel modification, or other construction related activities, the Director, IWMD or his designee shall prepare and submit, in consultation with the Planning and Development Services Department (PDSD) Director of Planning or his designee, an Interim Habitat Loss Mitigation Plan (IHLMP) to the USFWS for review and approval in compliance with the Natural Communities Conservation Plan (NCCP) and Interim Coastal Sage Scrub (CSS) Habitat Loss Process. The County remains committed to the NCCP process and intends to operate by the same procedure outlined in the Federal Endangered Species Act Section 4(d) Special Rule for Incidental Take of the coastal California gnatcatcher or other agreement as determined to be appropriate by the resource agencies.
- MM 4.5-2b: The GDP shall be amended to include all applicable provisions of the approved Southern Subregion NCCP on its adoption by the County of Orange Board of Supervisors. The NCCP implementation programs may include, but are not limited to, requirements for the removal and mitigation replacement of lost coastal sage scrub habitat, operations restrictions, instructional signs, fencing, etc.
- MM 4.5-2c: In accordance with an approved Conceptual Coastal Sage Scrub Mitigation Plan, the IWMD shall replace impacted coastal sage scrub at a minimum 1:1 (or as otherwise stated by USFWS) replacement ratio of in-kind habitat for onsite and offsite habitat preservation, replacement, or enhancement.

The IWMD shall prepare a Conceptual Coastal Sage Scrub Mitigation Plan in cooperation with the affected resource agencies (CDFG, USFWS). Guidelines for the Mitigation Plan shall be as follows:

• The mitigation areas/sites shall have been evaluated and selected on the basis of their suitability for use as coastal sage scrub revegetation areas. The parameters evaluated shall include but not be limited to soil condition, slope

aspect, proximity to adjacent coastal sage scrub, level of difficulty of site preparation, and ownership status.

- The mitigation plan shall provide procedures to prepare the soils in the mitigation area, provide detailed seeding/planting mixtures; provide seeding/ planting methods; and provide any other procedures, such as supplemental irrigation, mycorrhizal inoculation, etc., that will be used for successful revegetation.
- Maintenance and monitoring goals shall be established. The components and implementation of the maintenance and monitoring procedures shall be consistent with the components and implementation of mitigation measure 4.5-7a.

In accordance with the approved Conceptual Coastal Sage Scrub Mitigation Plan, the IWMD shall develop a maintenance and monitoring program to ensure success of the revegetation effort. Maintenance shall include regular inspection of the site for excessive weed growth, erosion problems, failure of irrigation system, and/or unhealthy or dying plants. Invasion of the site by weeds in the area, especially pampas grass, artichoke thistle, castor bean, fountain grass, mustard, clover, cocklebur, and tree tobacco could be a potential maintenance problem. Maintenance crews shall be able to recognize the difference between native plant and weed seedlings. A qualified biologist will be required to instruct the maintenance crew in the identification of native plant seedlings. The maintenance program shall include procedures for regular maintenance and repair of the irrigation system.

A system shall be developed for reporting by the maintenance crew of any unhealthy or dying plantings or failure in any of the seeded areas. This would assist the monitoring crew in the development of immediate remedial measures, such replacing plant material, to correct the problem.

To document the success of revegetation programs, the IWMD shall ensure that the progress of the revegetated area is monitored by a qualified biologist. The maintenance and monitoring plan will address unique aspects of mitigation areas. An agreement shall be developed between the County and the USFWS and CDFG on criteria that will be used to determine successful plant establishment on a mitigation site. Success criteria will include plant cover, species diversity, habitat structure, and density, and will be based on measurements made in reference habitats near the mitigation site.

Impact 4.5-3: The removal of riparian resources will occur as a result of the construction of landfill improvements on the Prima Deshecha site.

MM 4.5-3a: Prior to grading for the landfilling activities affecting riparian resources, the IWMD, as appropriate, shall ensure that all sycamore and willow trees of four or more inches in diameter at breast height (DBH), defined as 4.5 feet from mean ground level, within the grading or construction limits of the landfilling activities, whichever is greater, and within 100 feet of grading and construction operations, shall be tagged and numbered with permanent tags under the supervision of a qualified biologist. The tag numbers of the trees to be protected and those to be removed shall be noted. Those trees adjacent to the construction areas that can be avoided will be tagged for protection. Trees that cannot be avoided during

construction shall be tagged for removal and fenced off with red-orange flexible mesh fencing during grading and construction activities. Records of these numbers shall be kept by the Director, IWMD or his designee for use in mitigation, replacement, and monitoring of tree resources before, during, and after grading and construction activities. In addition, prior to grading and site preparation, the IWMD shall ensure that all trees subject to removal are marked with a red "X" on the trunk. Trees to be preserved shall be marked with yellow flagging visible from all directions and fenced-off with red-orange flexible mesh fencing during grading and construction activities.

- MM 4.5-3b: During the process of obtaining the required 404 Permit Application and 1601 Streambed Alteration Agreement (1601/404) for encroachment into streambed areas and prior to site preparation, the IWMD shall prepare a Conceptual Riparian Mitigation Plan in cooperation with the affected resource agencies (CDFG, USFWS, and the USACE). Guidelines for the Mitigation Plan shall be as follows:
 - The mitigation sites will be evaluated and selected on the basis of their suitability for use as riparian revegetation. The parameters evaluated shall include but not be limited to soil condition, hydrology, geology, and drainage considerations, level of difficulty of site preparation, access, contiguousness with existing habitat, and ownership status.
 - The mitigation plan shall include the procedures for soil preparation, provide seeding/planting mixtures; include seeding/planting methods; and include any other procedures, such as supplemental irrigation, mycorrhizal inoculation, etc., that will be used.
 - Maintenance and monitoring goals shall be established. The components and implementation of the maintenance and monitoring assignments shall be consistent with the components and implementation of mitigation measure 4.5-3d.
- MM 4.5-3c: In accordance with an approved Conceptual Riparian Mitigation Plan, the IWMD shall replace impacted riparian areas at a minimum 2:1 or higher ratio of in-kind or higher quality habitat. The required replacement acreage will be approved by the resource agencies having jurisdiction over the impacted resources (i.e., CDFG, USACE, and/or USFWS), for all the GDP uses, based on jurisdictional delineations and vegetation mapping and the current 2001 GDP grading plan.
- MM 4.5-3d: During the process of obtaining the 404 Permit and 1601 Streambed Alteration Agreement, and, in accordance with the approved Conceptual Riparian Mitigation Plan, the IWMD shall develop a maintenance and monitoring program to ensure success of any revegetation effort. Maintenance shall include regular inspection of the site for excessive weed growth, erosion problems, failure of irrigation system, and/or unhealthy or dying plants. Invasion of the site by weeds in the area, especially pampas grass, artichoke thistle, mustard, clover, castor bean, fountain grass, cocklebur, and tree tobacco could be a potential maintenance problem. Maintenance crews shall be able to recognize the difference between native plant and weed seedlings. A qualified biologist will be required to instruct the maintenance crew in the identification of native plant seedlings. The

maintenance program shall include procedures for regular maintenance and repair of the irrigation system.

A system shall be developed for reporting by the maintenance crew of any unhealthy or dying plantings or failure in any of the seeded areas. This would assist the monitoring crew in the development of immediate remedial measures, such as increasing the irrigation rate or replacing plant material, to correct the problem.

To document the success of revegetation programs, the IWMD shall ensure that the progress of the revegetated area is monitored by a qualified biologist. An agreement shall be developed between the County and the USACE, USFWS, or CDFG on criteria that will be used to determine successful plant establishment on a mitigation site. These criteria will include plant cover, and density, and will be based on measurements made in reference habitats near the mitigation site.

The qualified biologist shall monitor the site for five years or until the site complies with required performance standards. If the biologist determines that the mitigation site meets the conditions of the performance criteria prior to the five-year period, documentation shall be submitted to the responsible agency for approval.

- MM 4.5-3e: Prior to grading and site preparation adjacent to riparian areas outside the limits of construction, the IWMD shall incorporate instructions in the construction documents ensuring that, in conjunction with construction activities:
 - Graded material spoils shall not be placed or stored near riparian areas outside the limits of construction.
 - The removal of streamside or bank vegetation shall be avoided wherever feasible.
 - The amount of habitat removed shall be limited to the minimum amount required for construction.
 - Riparian areas in the vicinity of grading or heavy recreation use, such as in Zone 1, shall be designated as Environmentally Sensitive Areas onsite preparation, grading, and construction plans, and fenced off as appropriate for protection before any of these activities begin.
 - Excess fill shall not be dumped in streams outside the limits of construction.
 - Vehicles and equipment shall not be parked in washes or other drainages outside the limits of construction.

Impact 4.5-4: Special status habitats and special status species could exist within the construction limits of the landfilling improvements and could be adversely affected by the proposed landfill improvements.

MM 4.5-4a: Prior to site preparation and during final design for each phase of landfill development (i.e., Phases A–D in Zone 1 and Phases A–I in Zone 4), the Director IWMD shall ensure that focused surveys are conducted by qualified biologists for the thread-leaved brodiaea, Coulter's saltbush, many-stemmed dudleya, southern tarplant, vernal barley, paniculate tarplant, and any other plant

species that may warrant focused surveys in the future as determined by a qualified botanist. In addition, the Director, IWMD shall ensure that focused surveys are conducted by qualified biologists for the western spadefoot toad, southwestern willow flycatcher, and other wildlife species that may warrant focused surveys in the future as determined by a gualified biologist. The results of the surveys shall be incorporated into environmental documentation for future proposed projects within the Prima Deshecha site. Identified special status species and habitats located within 300 feet of the affected area(s) shall be mapped on grading plans for each phase of development. In addition, the Director IWMD shall implement procedures approved by the appropriate resource agencies to mitigate the potential impacts to those species. In the event that landfill activities within a phase must occur prior to the completion of spring surveys, habitat for the special status plant species shall be salvaged, stored and used in an appropriate manner as determined by a gualified biologist. The appropriate agencies will be notified prior to disturbance. All future proposed projects within the Prima Deshecha Landfill shall provide vegetation mapping on topographic maps at a scale of 1 inch equals 200 feet.

Impact 4.5-4b: The IWMD shall ensure that, for the periods covering all site preparation, disturbance or grading of native areas, the Director, IWMD or his designee shall monitor wildlife habitat preservation. The purpose of this monitoring is to ensure that the Environmentally Sensitive Areas and Environmentally Restrictive Areas (i.e., areas outside the grading limits) will not be adversely impacted during site preparation, grading, and construction of the landfilling activities.

For the landfilling activities, this inspection program shall be coordinated with the Site Manager at the weekly meetings held at the Landfill to review the planned grading program for the landfilling activities. These meetings shall commence at the start of each new phase, when native ground is schedule for disturbance (e.g., grading or stockpiling, etc.). The Director, IWMD or his designee will attend these meetings and provide a status and progress report to the Operations Manager. These meetings will be held throughout the site preparation, grading and construction periods for all the landfilling activities and the monitoring reports shall continue to be prepared and submitted by the Director, IWMD or his designee until the disturbance is completed.

The monitor shall be onsite before, during, and after the completion of site preparation, grading, and construction for all the landfilling activities.

Impact 4.5-5: Potential vegetation removal and habitat disturbance impacts of the landfilling uses could affect nesting sites for listed bird species and raptors, and dens for coyotes, bobcats, and mountain lions.

MM 4.5-5: During site preparation and grading for the landfill, the IWMD shall phase these operations outside significant habitat areas during the nesting and breeding season for the coastal California gnatcatcher. This measure shall be overseen and conducted by a qualified biologist.

During site preparation and grading for the landfill, the IWMD shall phase these operations outside significant habitat areas during the nesting and breeding season for the least Bell's vireo. This measure shall be overseen and conducted by a qualified biologist. Prior to activities that may impact potential vireo habitat, updated vireo surveys will be conducted by a qualified biologist.

The IWMD shall ensure that grading and construction operations for the landfilling are redirected temporarily around nesting sites for a distance of 500 feet for candidate and listed species of birds and a distance of 1,000 feet for raptors during nesting and breeding seasons between February 15 and July 15, or a distance and time period agreed upon by the USFWS. In the event that a coyote, bobcat, or mountain lion den is located, then grading and construction operations shall be redirected temporarily around the den for a distance of 1,000 feet. The nesting sites and dens should be resurveyed toward the end of the breeding seasons of these species to verify completion of the breeding cycle. Nests and dens that will be removed due to the grading and/or construction operations shall be removed only during the non-breeding season.

Impact 4.5-6: Potential indirect noise, air quality, and lighting disturbance impacts on biological resources could be associated with the landfilling activities.

MM 4.5-6: The IWMD shall ensure that during final design, the landfill operation continues to incorporate regulatory agency guidelines to reduce indirect impacts associated with noise, dust, night lighting, and blowing debris. Noise shall be controlled through the proper maintenance of the construction equipment, including trucks, bulldozers, and other mobile and fixed construction equipment. Dust shall be controlled at its source with standard wetting techniques consistent with applicable Southern California Air Quality Management District (SCAQMD) requirements. Low lighting alternatives and shielded lighting shall be employed to reduce indirect impacts on surrounding habitats.

Additional Mitigation Measures

Mitigation Measure MM 5.5-1 – Additional Provisions for Thread-Leaved Brodiaea

Prior to the Initiation of construction within Phase C3, OCIWMD will obtain authorization to take the thread-leaved brodiaea may be obtained from CDFG through the provisions of Section 2081(b) of the California Fish and Game Code if no federal nexus is present such as a USACE Section 404.

If a USACE Section 404 Permit is being pursued, IWMD would request consultation with the USFWS under Section 7 of the FESA. Consultation is required between the USFWS and a federal agency (such as the USACE) whenever a federal action is likely to adversely affect species listed as Threatened or Endangered, such as thread-leaved brodiaea. The anticipated federal action is the issuance/amendment of a 404 permit that will affect the thread-leaved brodiaea.

At the conclusion of the consultation, the USFWS will prepare a Biological Opinion based upon its review of the information provided herein. The final Biological Opinion may include an incidental take statement.

As part of the consultation process under Section 7 of the FESA, the California Department of Fish and Game (CDFG) will be consulted pursuant to Section 2080.1 of the California *Fish and Game Code*. Because the Project will affect a state-listed species, the thread-leaved brodiaea, CDFG concurrence with the Project conservation measures is required. The mitigation for the thread-leaved brodiaea will include the following requirements:

• A pre-construction survey during the peak flowering period, approximately March through June, will be conducted by a qualified biologist. The limits of each brodiaea

location within the impact area will be clearly delineated with lath and brightly colored flagging.

- The loss of thread-leaved brodiaea will be mitigated by seed and bulb collection, and revegetation into suitable mitigation site(s). A qualified biologist shall prepare a mitigation plan for review/approval by the United States Fish and Wildlife Service and oversee its implementation. The detailed mitigation plan shall include the following requirements:
 - The known populations of thread-leaved brodiaea on the project site shall be determined and mapped as the "collection area." The collection area shall include only areas within the impact footprint.
 - The existing locations of thread-leaved brodiaea shall be monitored every two weeks by a qualified biologist to determine when the seeds are ready for collection. A qualified seed collector shall collect all of the seeds from the plants within the collection area when the seeds are ripe. The seeds will be cleaned and stored by a qualified nursery or institution with appropriate storage facilities.
 - Following the seed collection, the bulbs shall be removed by an approved method (e.g., bulb collection or block transplantation). The bulbs shall either be transplanted directly or stored by a qualified nursery or institution with appropriate storage facilities. If the bulbs are collected and the block transplantation method is not used, then the top 12 inches of topsoil from the thread-leaved brodiaea locations shall be scraped, stockpiled, and used at the selected mitigation site.
 - The mitigation site(s) shall be located in open space. The site(s) shall not attempt to enhance existing populations and shall be located so as not to be impacted by any pesticides or herbicides used on adjacent properties.
 - The thread-leaved brodiaea mitigation site(s) will be prepared for seeding as described in a conceptual restoration plan.
 - The topsoil shall be re-spread in the selected location as approved by the project biologist. Approximately 60 percent of the seeds and bulbs collected shall be spread/placed in the fall following soil preparation. Forty percent of the seed and bulbs shall be kept in storage for subsequent seeding, if necessary.
 - A detailed maintenance and monitoring plan shall be developed by a qualified biologist. The plan shall include detailed descriptions of maintenance appropriate for the site, monitoring requirements, and annual report requirements, and shall have the full authority to suspend any operation in the study area which is, in the qualified biologist's opinion, not consistent with the restoration plan. Any disputes regarding the consistency of an action with the restoration plan will be resolved by the appropriate Project Applicant and the biologist.
 - The performance criteria shall be developed in the maintenance and monitoring plan and approved by a qualified biologist. The performance criteria shall also include percent cover, density, and seed production requirements. These criteria shall be developed by a qualified biologist following habitat analysis of an existing high-quality thread-leaved brodiaea population. This information will be recorded by a qualified biologist.

- If the germination goal is not achieved following the first season, remediation measures shall be implemented prior to seeding with the remaining 40 percent of seed and bulbs. Remedial measures shall include at a minimum: soils testing, control of invasive species, soil amendments, and physical disturbance (to provide scarification of the seed) of the planted areas by raking or similar actions. Additional mitigation measures may be suggested as determined appropriate by the project biologist.
- Potential seed sources from additional donor sites shall also be identified in case it becomes necessary to collect additional seed for use on the site following performance of remedial measures.

IWMD is currently pursuing authorization to collect seed and propagate the brodiaea as well as transplantation of the plants and soils containing plants from CDFG under Section 2081(b).

Mitigation Measure MM 5.5-2 – Fairy Shrimp Surveys

Prior to the initiation of construction activities that involve the removal of any pond within Zone 4, the IWMD shall have focused surveys conducted for the San Diego fairy shrimp and Riverside fairy shrimp by a biologist possessing the necessary resource agency permits. The surveys will be performed during the winter season prior to any construction activities on the site that may impact appropriate habitat for the fairy shrimp (i.e., ponds). The surveys will follow the protocol developed by the USFWS for these species. If it is determined that either or both fairy shrimp species are not present, then no further mitigation is necessary. However, if one or both fairy shrimp species are present, then consultation with the USFWS will be necessary in order to obtain a take authorization prior to any construction activities that may impact the species. The permitting process would require the preparation of a Biological Assessment which would include a mitigation plan to avoid or minimize impacts on this species.

Mitigation Measure MM 5.5-3 – Western Spadefoot Toad Surveys

Prior to the initiation of construction activities that involve the removal of habitat that is known and/or has the potential to support the western spadefoot toad, the IWMD shall have a focused survey conducted, where appropriate, on the project site prior to any potential impacts and during the breeding season for this species (February through May). The survey results will be submitted within 30 days after completion of the last survey to the CDFG for concurrence. Based on the May 3, 2005 survey results, a relocation program will be developed for western spadefoot on the project site. The relocation program will include a detailed methodology for locating, capturing, and relocating individuals prior to construction. The program will identify a suitable location for relocation of the western spadefoot prior to capture. The relocation program will require a biologist with the necessary permits for handling the western spadefoot. Prior to implementation of the relocation program, the program and the biologist(s) implementing the program will be subject to approval of the CDFG.

Mitigation Measure MM 5.5-4 – Existing Mitigation and Future Pre-Mitigation:

Any disturbance to existing or future mitigation areas, including those created by the Pre-Mitigation Plan or the Regional Environmental Enhancement Plan contained herein shall be restored by the IWMD at the completion of the landfilling activity during the next growing season using a hydroseed mix consistent with the appropriate approved mitigation plan. All restored areas will be maintained to remove non-native invasive plant species for a maximum of three years. Implementation of this mitigation measure shall constitute full compliance with the provisions of SEIR 597 and the approved CSS/NG Mitigation Plan. No further mitigation will be assessed against IWMD by the resource agencies.

5.5.5 LEVEL OF SIGNIFICANCE AFTER MITIGATION

The Proposed Project will affect approximately 255.02 additional acres of biological resources to address potential landslide remediation activities necessary to stabilize the existing landslide complexes within Zones 1 and 4. Of this amount, approximately 55.24 acres contain sensitive habitats. With the implementation of the Pre-mitigation Plan requiring the installation and establishment of CSS and southern needlegrass grassland prior to any impacts to these habitat types, the long-term net habitat values associated with these habitat types within the region will be sustained through project buildout with no cumulative loss. Therefore, impacts associated with the removal of this additional habitat will be reduced to less-than-significant level through the implementation of this plan.

In addition, SEIR 597 identifies potential impacts to the increased least Bell's vireo numbers as well as the identification of potential impacts to San Diego fairy shrimp, Riverside fairy shrimp, and western spadefoot toad, which were not previously addressed in EIR 575. Implementation of mitigation measures included in the SEIR will reduce the affects of this project to less-than-significant levels.

5.6 UTILITIES AND SERVICE SYSTEMS

5.6.1 EXISTING CONDITIONS

Electricity

Electrical service in the area and to the landfill site is provided by the San Diego Gas and Electric Company (SDG&E). Twelve-thousand-volt distribution lines are located underground in Camino de los Mares to the southwest of the site and overhead along La Pata Avenue to the north. The primary power source is the Capistrano Substation located in San Juan Capistrano.

Two parallel easements (Exhibit 5.6-1) with elevated transmission lines held by SDG&E and Southern California Edison (SCE) traverse the site along the Orange County/San Juan Capistrano border. The SDG&E pole and pipeline easement is 150-feet-wide and contains two 138-kilovolt (kV) lines on one set of towers and one 69-kV line on double wood poles. The SCE power line easement, which contains two 230-kV power lines, is 200 feet wide and is parallel and adjacent to the SDG&E easement. Also, a 12-foot-wide SDG&E easement at the southwestern portion of the site serves the leachate pump station.

Natural Gas

Natural gas service is provided to the site by the Southern California Gas Company (SCGC). According to SCGC facility maps, two 8-inch gas lines contained within two separate 12-foot easements, are located in the area: one in Camino de los Mares to the southwest and one in Ortega Highway to the north.

Potable Water

Potable water to the PDL is provided by the Capistrano Valley Water District (CVWD) via a water supply line located along La Pata Avenue. This source is used as the primary source of potable water to the site, but is also used as a secondary source for non-potable uses as necessary. Prior to construction of this water line, potable water was provided by the City of San Clemente Water Department (SCWD) through a fire hydrant located at the end of Camino de los

Mares at the south end of the PDL site. At this time, this hydrant is only used on a short-term basis when the primary source of water from the CVWD is temporarily unavailable or when water is required for a specific activity in an area inaccessible to the CVWD water supply line.

Non-Potable Water

The only existing non-potable water source to the PDL is provided by the Santa Margarita Water District (SMWD), and is used for purposes such as on-site dust control. According to the SMWD main office, this non-potable water is chlorinated effluent (reclaimed water) from the Chiquita Treatment Plant on Ortega Highway, in unincorporated Orange County.

Sanitation Facilities

Currently, the site is served by portable toilets and a septic tank system. No sewage lines currently serve the site. Sanitary sewer facilities in the area are owned and operated by the cities of San Juan Capistrano and San Clemente. According to the San Clemente Public Works Department (SCPWD), the only existing sewer line in the area is a line located in Camino de los Mares which is an eight-inch diameter line immediately south of the site boundary, expanding to an 18-inch diameter line further south in Camino de los Mares. This line currently terminates within 200 feet of the southwest boundary of the site.

Storm Drains

Section 5.3.1 (Hydrology, Existing Conditions) provides a general description of the M01 drainage system used to evacuate storm flows from the project site. This system is essentially a reinforced concrete box (RCB) storm drain which was designed to accommodate a 25-year storm. To the north of the site, storm water flows through natural channels flowing into San Juan Creek.

Telephone Service

Telephone service in the area is provided by Pacific Bell of California, Inc. (PacBell). According to its Tustin office, existing telephone service to the site is provided by an overhead line along La Pata Avenue to the site boundary. From that point, the line is underground for approximately 785 feet, where it connects to a pedestal or junction box at the fee station at the landfill entrance. Other PacBell facilities are in underground conduits to the southwest of the site in Camino de los Mares.

In 1972 a 15-foot telephone line easement was granted to the Southern California Telephone Company on the northeastern part of the site. This easement was later assumed by PacBell, and subsequently quit-claimed.

<u>Schools</u>

The site is in the Capistrano Unified School District. Recent state legislation (specifically Assembly Bills 2926, 1600 and 181 and Senate Bill 1287) provides guidelines, procedures and restrictions related to the levy of development fees for school facilities, especially with regard to commercial and industrial development. The structures associated with recreation uses that generate employment, such as the clubhouses associated with the golf courses, may impact the Capistrano Unified School District. Fees may be assessed upon the implementation of these types of recreational facilities.

Santa Fe Pacific Pipelines Partners Inc. Pipelines

A ten-foot-wide easement was issued to the San Diego Pipeline Company in 1966 on the northeastern portion of the PDL property. Although this is still an active easement, there is no evidence indicating that a gasoline or natural gas pipeline was ever actually installed on the Prima property (Arnau 2005).

5.6.2 SIGNIFICANCE CRITERIA

A project will normally have a significant effect on the environment related to public utilities if it will "...substantially degrade water quality...contaminate a public water supply...[or] extend a sewer trunk line with capacity to serve new development..." For purposes of this SEIR, public utility impacts were considered to be significant if they met these defined criteria or if they resulted in significant increased need for equipment, facilities, or staff in any of the jurisdictions affected by the GDP.

5.6.3 POTENTIAL IMPACTS

Electricity

Exhibit 5.6-1 illustrates the relative locations of utility easements on the PDL property and the proposed limits of disturbance. Implementation of landslide stabilization measures in the vicinity of the Zones 1-4-5 overlap will impact the SCE and SDG&E easements traversing the center of the PDL property. Existing transmission lines will have to be temporarily relocated or re-routed in order to avoid service disruption during construction. Once construction is complete, transmission lines through the site will be replaced. The IWMD will coordinate closely with SCE and SDG&E in the development of a plan to provide for uninterrupted electrical transmission during construction.

The Proposed Project will not require the use of supplemental lighting or operating equipment beyond that already accommodated at the site. Electrical demand for the Proposed Project elements is anticipated to be minimal and temporary in nature while landslide remediation activities are underway.

Natural Gas

Proposed Project elements are not anticipated to affect SCGC facilities. In addition, the landslide remediation work and Zone 4 desilting system construction will not require natural gas and would not result in a substantial increase in the demand or require substantial new or expanded natural gas facilities on the site.

<u>Water</u>

Potable water at the site will continue to be supplied through the water delivery line along La Pata Avenue. Proposed project elements will not result in an increase in potable water demand on site.

The demand for non-potable water at the PDL is not expected to be affected by the Proposed Project. Water required to sustain the biomitigation areas is proposed to be derived from the project desilting system, which mimics existing hydrologic conditions by capturing local storm flows and releasing them at a controlled rate into the Prima Deshecha Canada stream channel. Upland biomitigation areas are not anticipated to require a long-term supplemental water source.



 ε
 2,000
 1,000
 0
 2,000

 Feet
 Feet
 Feet
 Feet

P:P

R:\Projects\OCIWMD\J004\SEIR597 Public Review Draft\graphics\Ex5.6-1_ease_012406.pdf

Accordingly, no impacts are anticipated to water utilities as a result of the Proposed Project.

Sanitation

The Proposed Project will not affect the number of permanent employees located at the PDL, and there are no plans by the IWMD to extend pipelines to any of the existing sewer trunk systems in the vicinity of the site. All existing GDP landfill sanitary facilities will be maintained as is, and will be unaffected by the Proposed Project. Therefore, no impacts to sanitation facilities are anticipated as a result of the Proposed Project.

Storm Drains

As indicated above in Section 5.3, on-site flows and drainage areas will be redirected per the redesigned Zone 4 desilting system. These basins will serve to reduce suspended sediment and floating debris before the water is discharged to the watercourse, and will handle the 100-year runoff from the project site. Accordingly, there will be no impact on local area storm drains.

Telephone Service

The Proposed Project elements will not affect the existing PacBell facilities on the northern side of the site in the vicinity of the site entrance and the fee station area. There are no planned permanent improvements as part of the Proposed Project that would require a change in telephone service at the site. Therefore, the Proposed Project will not have an impact on telephone services.

<u>Schools</u>

As no residential uses are proposed under the Proposed Project and there will be no increase in permanent employees on staff at the PDL, the Proposed Project will not have an impact on school facilities in the Capistrano Unified School District.

5.6.4 MITIGATION MEASURES

Previously Adopted Mitigation Measures

The following mitigation measures are currently in place for impacts associated with the landfill component of the 2001 GDP, as identified in EIR 575 (numerical designations are from EIR 575). All mitigation commitments contained within FEIR 575 and the 2001 GDP will apply to the Proposed Project.

Electricity

Impact 4.16-1: Potential disruption of existing Southern California Edison (SCE) and San Diego Gas and Electric (SDG&E) facilities during construction of the GDP landfilling uses.

MM 4.16-1: Prior to approval of construction and grading plans, the IWMD will include, as part of the construction documents, requirements that the construction contractors coordinate with SCE and SDG&E to ensure that their facilities on the site are protected to prevent significant disruption to utility services during construction. The contractor will be required to provide written documentation of this coordination to the IWMD.
Santa Fe Pipeline Company

Impact 4.16-2: The Santa Fe Pacific Pipeline Partners Inc., oil pipelines on the east side of the site could be impacted during construction of the landfilling uses.

MM 4.16-2: The IWMD will coordinate with Santa Fe Pacific Pipeline Partners Inc., during final design of the landfilling uses in Zone 4 regarding the precise location and depth of the existing pipelines on the site. The IWMD shall coordinate the landfill construction schedules with Santa Fe Pacific Pipeline Partners Inc., to allow the company to relocate its pipelines, if needed, prior to IWMD initiating construction of landfilling improvements in Zone 4 that would otherwise impact these pipeline facilities.

Natural Gas, Potable Water, Non-Potable Water, Sanitary Facilities, Storm Drains, Telephone Service and Schools

No impacts on natural gas, potable water, non-potable water, storm drains, telephone service and schools are expected under the 2001 GDP landfilling activities and therefore no mitigation is required.

Sanitary Facilities

Impact 4.16-3: No wastewater facilities are available to the site in the Santa Margarita Water District service area; therefore, all GDP landfill operations sanitary facilities would be septic tank systems, similar to the current operations at the landfill.

- MM 4.16-3a: Prior to the commencement of any landfilling operations, a soils report and plans for all sewage disposal systems shall be submitted to the County's Plumbing/Mechanical Plan Checking Section for review and approval.
- MM 4.16-3b: Results of percolation tests and a log of soil borings, performed and reported by a Registered Environmental Health Specialist, Registered Civil Engineer or Registered Geologist, in accordance with Environmental Health's "On-Site Sewage Disposal System Guidelines" shall be submitted to the County's Plumbing/Mechanical Plan Checking Section for review and approval. The Land Use Unit of Environmental Health shall be notified at least 48 hours prior to soil testing in order to be present during testing, if deemed necessary.
- MM 4.16-3c: Each proposed individual sewage disposal system shall be designed in accordance with Environmental Health's "On-Site Disposal System Guidelines."
- MM 4.16-3d: An additional soil percolation system, equal to a maximum of 100 percent of the original design capacity or as deemed necessary by the Manager, Environmental Health, shall be constructed and connected.

Mitigation for Impacts Associated with Amendment No. 2 to the 2001 GDP

PDF 5.6-1: SCE and SDG&E electrical transmission facilities will be relocated or rerouted, if necessary, in order to avoid service interruptions during construction of landslide remediation measures through the center of the site. IWMD will coordinate closely with SCE and SDG&E in the development of a plan to ensure cost-effective and efficient temporary facility relocation and post-construction re-establishment of transmission lines through the site.

5.6.5 LEVEL OF SIGNIFICANCE AFTER MITIGATION

Less than significant.

SECTION 6.0 ENVIRONMENTAL ISSUES NOT REQUIRING SUBSTANTIAL ADDITIONAL ANALYSIS

6.1 INTRODUCTION

As noted in Section 1.0, CEQA Guidelines, Section 15163(a)(2) indicates that an SEIR is only required to address the information "necessary to make the previous EIR adequate for the project as revised." Accordingly, this section of SEIR 597 provides a brief summary of the environmental issues for which implementation of the proposed project have resulted in a determination of "no substantial change" over those impacts identified in EIR 575.

All mitigation commitments contained within FEIR 575 and the 2001 GDP will apply to the Proposed Project and are located in Section 2.2, Overview of the General Development Plan. These mitigation measures (contained within Table 2.2-4) are currently in place for impacts associated with the landfill component of the 2001 GDP, as identified in EIR 575 (numerical designations are from EIR 575).

6.2 <u>SUMMARY OF ISSUES NOT REQUIRING ADDITIONAL ANALYSIS</u>

6.2.1 LAND USE AND PLANNING

The Proposed Project does not involve any changes to the existing land uses that are outlined in the 2001 GDP. The PDL will maintain its current primary use as a landfill, and will not change the daily maximum refuse being accepted or permitted at the site. Although there is additional grading considered in association with implementation of landslide stabilization measures, it is not considered significant as all refuse fill slopes will be constructed consistent with the 2001 GDP once these stabilization measures are in place.

The only element with a minor effect on land use is associated with the Pre-mitigation and Regional Enhancement plans. These plans identify mitigation and enhancement opportunities within Zone 4 that could potentially encourage a more passive recreational use of the area in the post-closure period. However, EIR 575 indicates that the future designated use for Zone 4 will be a regional park; design and features of this park are to be determined by a recreational needs analysis conducted in partnership with the surrounding communities in the post-closure timeframe. The IWMD will retain discretionary authority over the appropriate mix of long-term uses for the property.

Accordingly, the Proposed Project does not result in substantial change from the previous analyses contained within EIR 575 and the analyses and mitigation measures outlined in EIR 575 are adequate to support the Proposed Project. Therefore, no further analysis or additional mitigation is required.

6.2.2 AGRICULTURE

Subsequent to certification of EIR 575, the project site has not been subject to a new agricultural use (other than grazing) and the state Important Farmland designations have not changed. The Proposed Project elements will have no effect on agricultural uses other than potentially removing the possibility of grazing as an acceptable land use in Zone 4 over the post-closure time period once the Pre-mitigation and Regional Environmental Enhancement plans are approved. As implementation of the Proposed Project results in no substantial change in effect over that identified in the 2001 GDP, the analysis of agricultural resources provided in EIR 575 is adequate for SEIR 597. Therefore, no further analysis is warranted.

6.2.3 POPULATION AND HOUSING

The Proposed Project has no elements to it that could be considered growth-inducing, and no significant incremental impacts are expected related to population and housing. As the Proposed Project will result in no substantial change over the effects identified in EIR 575, the analyses contained within EIR 575 are deemed adequate for SEIR 597.

6.2.4 TRANSPORTATION AND CIRCULATION

The Proposed Project elements will not affect short- or long-range traffic conditions, as described in EIR 575, as daily refuse tonnages into the landfill, overall landfill capacity, and land uses on the project site have not changed. Construction of the elements of the Proposed Project will also occur entirely within the boundaries of the PDL, and therefore no change to the traffic patterns in the surrounding intersections is anticipated. It is anticipated that construction of the rainfall collection system basins and landslide remediation measures will not require the use of additional equipment considered in the 2001 GDP and that staging for this construction will also be on site. Material volumes generated from construction of the Proposed Project are not expected to be incrementally significant, and will be handled on site as well. Accordingly, the Proposed Project is expected to have no substantial change from the effects identified in the previous analysis.

6.2.5 NOISE

Although there will be an incremental change to construction activities at the landfill as a result of proposed landslide stabilization measures, it is not expected to be greater than the noise levels associated with landslide stabilization construction previously covered in EIR 575 and is not expected to contribute significantly to noise levels because of ongoing disposal operations at the landfill. Further, much of this landslide remediation activity is to take place in and around Zone 4, which is farther from sensitive receptor sites than current operations within Zone 1. The Pre-mitigation and Regional Environmental Enhancement plans support the open space quality of the area and will reduce noise emissions from the project post-closure over potential postclosure uses contained within the 2001 GDP. Therefore, no further analysis is warranted.

6.2.6 AESTHETICS

Pre-mitigation and Regional Enhancement Elements to the Proposed Project will add to the undeveloped and open space character of the property in the post-closure time period. Implementation of these elements will not create any negative aesthetic impact during establishment of the environmental habitats on site.

Although there will be an incremental change to the landscape as a result of proposed landslide stabilization measures, it will not significantly change final surface grading or fill slopes and is not expected to contribute significantly to the aesthetic impacts that were analyzed in EIR 575. It is anticipated that landslide remediation within the revised limits of disturbance and construction of the revised desilting system will occur below ridgelines and not pose an aesthetic impact.

6.2.7 CULTURAL AND SCIENTIFIC RESOURCES

Due to the static nature of cultural resources in the landscape, the archaeological and paleontological conditions of the proposed project would be consistent with those identified in EIR 575. Increasing the limits of disturbance for the project may expose some unknown paleontological or archaeological resource to impacts from grading; however, implementation of the mitigation measures in EIR 575 will reduce this potential impact to below the threshold of

significance. There are no known historic resources located within the incremental area of disturbance associated with the Proposed Project. No additional analyses are therefore required.

6.2.8 RECREATION

As the proposed project does not contain any elements that would be considered growthinducing, no impact would be expected to local or regional recreational resources. The Proposed Project does not alter the 2001 GDP's commitment to incorporate several trails around the perimeter and through the PDL property, consistent with both County and City (San Juan Capistrano and San Clemente) trail plans and safety considerations associated with landfill operations. Subsequent to issuance of the 2001 GDP, the County approved a minor modification to the 'Proposed County Connector Trail,' shown on Exhibit 6.2-1. This connector trail alignment, approved in March 2004, replaced an existing trail alignment adjacent to the landfill property boundary that was too steep in grade for use. Other future routing modifications of on-site trails within Zones 3 and 4 may be required by the specific functional requirements of the newly restored habitat areas designated by the Pre-mitigation and Regional Environmental Enhancement Opportunity plans; however, it is anticipated that the site will still be able to accommodate these trails in aggregate.

Identification of the pre-mitigation plan and enhancement opportunities may affect the desired mix of recreational uses for Zone 4 in the post-closure period. The proposed Pre-mitigation and the Regional Environmental Enhancement Opportunity plans on the post-closure Zone 4 landfill is intended to occur within a mix of potential post-closure uses that may include appropriate recreational uses. Consistent with the approach contained within the 2001 GDP, these recreational uses are to be developed by a needs analysis as Zone 4 closure nears. The GDP notes that a future recreational needs analysis will likely consider proximity to environmental enhancement features, landfill-related regulatory and technical restrictions, and the demands and priorities of surrounding communities at that time in determining which post-closure uses are appropriate and desirable. As the implementation of environmental enhancement opportunities is discretionary and can be flexible in execution, it is anticipated that these mixed uses can be blended over time in a satisfactory manner.

6.2.9 MINERAL RESOURCES

The Proposed Project for Amendment No. 2 does not contain any element that would affect or alter the findings of EIR 575 with respect to Energy and Mineral Resources. Therefore, no additional analyses are required.

6.2.10 HAZARDS

The proposed project does not involve any changes to the GDP that would contribute to the creation of a public health hazard, the transport of disease, or increase on-site vectors. Although the revised desilting system does incorporate the addition of several perimeter basins, existing mitigation measures require that there is no incremental effect associated with water-related vector issues such as mosquitoes. In addition, the PDL will continue to operate in accordance with all existing regulations and permitting requirements, and will also continue to be monitored by all agencies having jurisdiction. Therefore, no further analysis is warranted.

6.2.11 PUBLIC SERVICES

The proposed project does not include any elements that would contribute to any change in demand to public services. No further analysis is necessary.



Map Not to Scale

R:\OCIWMD\J004\SEIR Public Review Draft\graphics\Ex6.2-1_012406.pdf

SECTION 7.0 ALTERNATIVES TO THE PROPOSED PROJECT

7.1 <u>OVERVIEW</u>

Section 15126 of the CEQA Guidelines requires that an EIR evaluate alternatives to the Proposed Project which could feasibly attain the basic objectives of the project, but would avoid or substantially lessen any of the significant environmental effects of the project. The comparative merits of these alternatives must then be evaluated. A No Action Alternative must be included in this comparative evaluation.

As discussed in Section 3.0, the purpose and need for the Proposed Project is to: (1) provide for physical site stability within the PDL site to ensure continued operation of landfill activities; (2) accommodate future landfill-related features; (3) redesign the desilting system to reduce biological resource impacts; and (4) facilitate long-term stewardship of on-site biological resources.

Section 2.2.3 summarizes the previously reviewed alternatives within EIRs 548 and 575. These alternatives are presented as "considered but eliminated from further analysis," or "analyzed in detail." Following is a discussion of the alternatives to the Proposed Project for Amendment No. 2 to the GDP that were considered but eliminated from detailed analysis based upon feasibility issues. Lastly, Section 7.2 of this document presents the alternatives to the Proposed Project that address the project's purpose and needs, and analyzes them in greater detail.

7.1.1 ALTERNATIVES CONSIDERED BUT NOT EVALUATED IN SEIR 597

Several alternatives were considered for detailed analysis in SEIR 597 but were eliminated due to infeasibility. These alternatives are listed below.

Reduce the Zone 4 Footprint and Deepen to Maintain Capacity. This alternative sought to reduce the impact of Zone 4 on both the spring recharge area in the eastern portion of the zone and the least Bell's vireo territories on the western perimeter of the zone. In accordance with the project objectives, maintenance of Zone 4 capacity under this alternative would require a significant deepening of the fill area. This alternative landfill design would require approval of landfill liner installation below historic groundwater levels. Significant uncertainties relative to the stability of existing subterranean landslide complexes and the potential impacts to local and regional groundwater resources would require extensive technical studies over an extended period of time in order to obtain the data needed for feasibility determination. This alternative was eliminated from further analysis due to likely significant impacts and feasibility issues.

Shift the Zone 4 footprint over the ridge into Segunda Deshecha. Moving the Zone 4 footprint south into Segunda Deshecha was considered as a project alternative for analysis. This would require impact to an area currently designated as open space for sensitive environmental resources; its value as an open space preserve and conflict between active landfilling uses and the permitted uses for the Supplemental Open Space proposed under the South Orange County NCCP (under development) resulted in the elimination of this alternative from further consideration. In addition, landfill operations within Segunda Deshecha violate established viewshed commitments with the cities of San Juan Capistrano and San Clemente and contribute to making this alternative infeasible.

Shift the Zone 4 footprint north to or past the property boundary. This alternative would shift the Zone 4 footprint outside the area of impact for the least Bell's vireo territories and would reduce the impacts on the spring recharge area. Landfill capacity is maintained through extension of the landfill refuse footprint north of the property. Significant impacts accrue under this planning scenario to utility line rights-of-way through the project site; existing development to the north precludes impact to the property boundary. Accordingly, this alternative was eliminated from further consideration.

7.2 <u>ALTERNATIVES TO THE PROPOSED PROJECT FOR AMENDMENT NO. 2 TO THE</u> <u>GDP EVALUATED IN SEIR 597</u>

Table 7.2-1 provides a comparative analysis of potentially feasible alternatives to the Proposed Project.

			Permits	Reduces				
	Meets	Maintains	Project	Bio	Reduces			Maintenance
	Purpose	Landfill	Through	Impacts	Impact	Cost	Engineering	Channel
Alternative	and Need	Capacity	Build-Out	from GDP	to LBV	Increase	Feasibility	Flows
Proposed Project	•	•	•	•	•		•	•
No Action Alternative		•					•	
Alternative 1	•	•	•				•	
Alternative 2	•	•	•				•	•
Alternative 3			•	•	•	•		
Alternative 4			•	•		•		

TABLE 7.2-1 ALTERNATIVES MATRIX

7.2.1 NO ACTION ALTERNATIVE

Description of the No Action Alternative

The No Action Alternative consists of the approved project in the 2001 GDP, as revised by Amendment No. 1, the MOU between the County of Orange and the cities of San Juan Capistrano and San Clemente, and agreements with the Rancho Mission Viejo Company, LLC (RMV) (Exhibit 2.1-1). These documents can be found in Appendices C and D. The project in the 2001 GDP (as amended) represents the most recently approved decision document guiding landfill, circulation, and recreation uses of the project site through the ultimate project buildout in the year 2067. An overview of the plan as contained in the 2001 GDP can be found in Section 2.0.

Under the No Action Alternative, neither the areal extent of landslide nor slope-stabilization measures that are required for the implementation of the Zone 4 portion of the project site were specifically presented in EIR 575 (Exhibit 2.1-1). Based upon the geotechnical characteristics of the area and the recent history of landslides at the site (May 1998), landslide-remediation measures that involve excavation buttress fill activities for Zones 1 and 4 will extend beyond the original zone boundaries. Without approval to implement these measures, it is unlikely that resource agency or landfill operations permits would be issued for initiation of landfill operations in Zone 4. Under this scenario, the reduction of capacity at the PDL caused by a reduction in the area available for stable landfill operations would require the County of Orange to identify feasible off-site disposal alternatives to offset this loss in capacity.

The No Action Alternative does not include development of the rainfall storage system/ subsurface reservoir. Even under a modified Zone 4 footprint to avoid unstable areas, impacts

to the spring recharge area within Zone 4 would occur requiring the implementation of mitigation measures similar to those being evaluated in the Proposed Project.

Without a Pre-mitigation Plan in place, project mitigation would have to be developed without the benefit of a regional view toward biological resource restoration and enhancement. The benefits to the environment from a regional planning approach would be much more limited without effective partnering in the South Orange County NCCP or SAMP programs.

As previously discussed, active landfilling operations are currently occurring in Zone 1. The analyses contained in this document are consequently focused on the environmental effects, permitting activities, and subsequent initiation of landfill operations within Zone 4 that are anticipated to begin in the year 2019. Accordingly, the No Action Alternative description and project alternatives analyses, as detailed below, apply primarily to upcoming operations specifically within Zone 4

Impacts of the No Action Alternative

Geotechnical Considerations

Geotechnical considerations of the 2001 GDP are documented in EIR 575. Mitigation measures incorporated into EIR 575 would remain in place, but with no landslide stabilization measures occurring outside the existing limits of disturbance. Although EIR 575 geotechnical mitigation measures could be implemented within existing zone boundaries, landfill capacity would be significantly reduced in areas of significant geotechnical hazard.

Hydrology and Water Quality

The No Action Alternative would not alter the existing limits of disturbance around the PDL. However, the recharge area for the spring in Zone 4 would be impacted sooner since operations would move into the eastern portion of the site sooner if the IWMD cannot stabilize the landslide complex within the central and southwestern portions of Zone 4 under a modified Zone 4 footprint. This would cut off most of the flows into Prima Deshecha Cañada channel and would therefore result in impacts to riparian resources. The Proposed Project includes a rainfall storage system (that is intended to maintain consistent flows to downstream riparian resources that are dependent upon these flows) and/or the surface water augmentation options.

Air Quality

Under the No Action Alternative, air quality commitments from EIR 575 would continue to be maintained and would include updates to field operating permits, as required, to ensure compliance with air quality regulations. However, it should be noted that several applicable regulations have been amended since the certification of EIR 575. The recent updates are discussed in detail in Section 5.4. Where previous measures have been superseded, it is appropriate to implement more recent requirements. This applies to the No Action Scenario.

Biological Resources

Under implementation of an existing Zone 4 refuse footprint, biological resources within the PDL under the No Action Alternative would eventually be impacted by a reduction in the recharge area for the spring that feeds Prima Deshecha Cañada channel. This indirect impact would occur through a gradual decrease in the frequency and volume of flows within the channel itself. The eventual direct impact to the spring would reduce these flows to a much lower volume flow that would be provided by direct runoff into the stream channel itself. Once flows are significantly impacted within this channel habitat, a supplemental water source would be needed

to maintain the riparian corridor. Impacts from this supplemental water source will vary based upon the source and method of delivery.

The No Action Alternative does not provide for a Pre-mitigation Plan or Regional Environmental Enhancement Plan. Without these elements, regulatory permits for buildout of the PDL would likely occur on a phase-by-phase basis over the period of modified project buildout. Phase-by-phase regulatory permitting could impact the IWMD's ability to provide uninterrupted, long-term refuse disposal services. The Pre-mitigation Plan contained within the Proposed Project (in concert with the Regional Environmental Enhancement Opportunities Plan) assures continued seamless landfill operations while also ensuring that stewardship of on-site biological resources is maximized through a site-wide mitigation program and integration with regional environmental planning efforts. Implementation and establishment of a Pre-mitigation Program years in advance of project impacts will enhance the long-term biological resource conservation values and functions for coastal scrub, needlegrass, and riparian-dependent wildlife species well beyond that which would occur with incremental implementation of project mitigation over the build-out timeframe.

Other

If the No Action Alternative resulted in the loss of landfill capacity at the PDL, numerous significant impacts could occur from the need to transport refuse out of the region to a remote location. These impacts would accrue to traffic, air quality, hazards, and noise associated with the transporting of refuse out of the county or state via truck or train. Ultimate impacts could also accrue to population and housing (if refuse disposal becomes a significant regional development constraint).

Summary of the No Action Alternative

The No Action Alternative is based on landfill operations at the site continuing in accordance with the 2001 GDP. However, that landfill capacity (as estimated in the 2001 GDP) will not be achieved if the area required for landslide remediation is not available (or permitted for disturbance). If landfill capacity is reduced, the need for the County to look elsewhere for refuse disposal would be considered significant, as documented within EIR 575. If the County can continue operations in Zone 4 (either in whole or in part) then biological resources will be affected incrementally over time by both indirect and direct impacts to the spring that feeds Prima Deshecha Cañada. Once this impact occurs, a supplemental water source will need to be identified to maintain flows in the channel. Impacts associated with this water source will vary depending upon the source and method of delivery.

7.2.2 ALTERNATIVE 1: MAINTAIN 2001 GDP ZONE 4 FOOTPRINT AND DETENTION/DESILTING BASIN BETWEEN ZONES 1 AND 4

Description of Alternative 1

Alternative 1 (Exhibit 7.2-1) consists of the 2001 GDP design for Zone 4 including a detention/ desilting basin located between Zones 1 and 4, but proposes that this basin be situated north of and outside the Prima Deshecha Cañada stream. Although the basin is proposed to be located outside the Prima Deshecha Cañada streambed, there would be temporary impacts to the streambed during construction associated with remediation requirements for a landslide complex underneath the area proposed for the basin. These impacts are discussed in more detail below. Therefore, Alternative 1 would involve an expansion of the disturbance limits from the Proposed Project in the central portion of the site between Zones 1 and 4.







Impacts of Alternative 1

Geotechnical Considerations

Preliminary geotechnical investigations indicate that the proposed basin location is in an area of a large landslide complex, which will require extensive earthwork removal and replacement to provide for a stable subgrade for basin construction.

Biological Resources

The landslide remediation will require removal and recompaction of substantial alluvial materials in and around the streambed which will result in impacts to five least Bell's vireo (LBV) territories. The impacts would be temporary until the streambed is restored. Additional mitigation approved by the USFWS and CDFG would have to be implemented to offset impacts associated with the temporary loss of riparian habitat.

Alternative 1 would have a permanent impact on one LBV territory and temporary impacts to four LBV territories.

Other

Landfill Capacity. Alternative 1 maintains a balanced landfill with the same capacity (59 million tons), as indicated within the 2001 GDP.

7.2.3 ALTERNATIVE 2: MAINTAIN 2001 GDP ZONE 4 FOOTPRINT AND DETENTION/DESILTING BASIN BETWEEN ZONES 1 AND 4 WITH SURFACE WATER AUGMENTATION

Description of Alternative 2

Alternative 2 (Exhibit 7.2-2) consists of the 2001 GDP design for Zone 4, including a detention/ desilting basin located between Zones 1 and 4 north of the Prima Deshecha Cañada channel, with surface water augmentation of spring flows. The augmentation approach is proposed to be implemented when subdrain flows are not sufficient to sustain riparian habitat within the downstream portions of Prima Deshecha Cañada channel. Subdrain flows will be utilized to the extent possible to recharge the Prima Deshecha stream on the landfill property and will not be discharged off site except during significant storm events. During these events, storm water will be subject to NPDES monitoring requirements. Groundwater discharges may need to comply with the San Diego RWQCB, Order No. 2001-96, NPDES No. CAG909002, which contains specific requirements for groundwater extraction discharges to surface waters.

It should be noted that if future biological monitoring data indicates that surface water augmentation is required as part of the Proposed Project, one or more of the water augmentation options described below may be incorporated into the Proposed Project plan if the subsurface reservoir proves to be infeasible.

Water Augmentation Options

As discussed above for the project, when construction of the eastern portion of Zone 4 by Phases D and E commences (estimated to occur by 2045), the spring recharge area will diminish as the landfill liner system expands and prevents surface water infiltration. Therefore, another water source would be sought to supplement subdrain flows as a means of maintaining delivery of water to the downstream reaches of the stream.

Off-Site Water Source

A potable water main is located along the ridgeline north and west of Zone 1. However, the supplier, Capistrano Valley Water District (CVWD), has in the recent past indicated that its current usage allocation would not meet the site's entire operations needs due to a shortage of water storage capacity in the service area.

Another water line has been constructed for the field offices and ancillary facilities at the entrance to the landfill, which is a source of non-potable water. This water is provided by an offsite well operated by the Santa Margarita Water District (SMWD). The existing non-potable water supply line at the landfill is capable of supplying twice the amount of flow currently used so that additional water could be provided with the existing supply line capacity for stream flow replenishment. A connection to the new non-potable water supply line would be needed to provide surface water augmentation to the stream. The feasibility of this alternative depends on the availability of the off-site well water for the site's use beyond 2045.

On-Site Water Reservoir

Off-site water piped into the site can be stored in aboveground water reservoirs for gradual release into the Prima Deshecha Cañada channel. Siting of these aboveground storage facilities will need to consider stability factors, the bearing capacity of foundation soils, and compatibility with future operational areas.

Another opportunity for providing on-site storage capacity for Alternative 2 is from storm water flows by diverting storm water from the Zone 4 area into a storage facility underneath the proposed detention/desilting basin(s) (see Proposed Project design in Section 4.3). Landslide stabilization also needs to be considered for this alternative.

Irrigation Water Source(s)

This alternative proposes the use of irrigation water from the post-closure regional park end use proposed for Zone 1. The timing for Zone 1 end-use development is projected to be about five to ten years after Zone 1 is closed (in 2019) to allow for sufficient settlement of the landfill in support of an active park use. An analysis of the specific park uses, funding sources, and anticipated settlement will be conducted closer to closure of Zone 1. To comply with landfill regulations, flows from the existing Zone 1 area are currently collected in a separate drainage channel from the natural Prima Deshecha Cañada bio-mitigation channel flows for testing prior to discharge. There is the potential to redirect Zone 1 post-closure surface water irrigation flows from the park end use from Zone 1 directly into the Prima Deshecha Cañada bio-mitigation channel recharge area, thereby augmenting natural water flows. Direct discharge of the surface water flows from Zone 1 into the bio-mitigation channel would require approval from the Regional Water Quality Control Board (RWQCB).

Other opportunities for the use of irrigation water for stream flow augmentation could include embankment irrigation for the future La Pata Road and irrigation from the proposed park end use for Zone 4.

On-Site/Off-Site Groundwater Extraction

Groundwater extraction options were evaluated from both on and off the site (piped to the site). Exhibit 5.3-3 illustrates the location of groundwater monitoring wells and gradients within the project area.







Water would be pumped from local groundwater wells and discharged into the stream channel to sustain biological riparian resources.

Impacts of Alternative 2

Geotechnical Considerations

Geotechnical complexities within Zone 4 are discussed in Section 3.1 (Purpose and Need). Remediation measures would be implemented for the Zone 4 basin upon initiation of excavation within Zone 4.

Hydrology and Water Quality

Potential water quality impacts will likely vary based upon the option selected for surface water augmentation.

Off-Site Water Source

Off-site potable water sources are treated to drinking water standards and will be more than adequate, in terms of a consistent water supply and water quality, for sustaining biological resources along Prima Deshecha Cañada.

On-Site Water Reservoir

An on-site water reservoir, whether surface or subsurface in design, will utilize storm water runoff from the surrounding area. This runoff currently provides a portion of the stream channel flows to Prima Deshecha Cañada; accordingly, water quality from flows stored within an on-site reservoir are not anticipated to be substantively different from existing conditions.

Irrigation Water Source(s)

Runoff associated with irrigated water from Zone 1 post-closure will be treated with Best Management Practices (BMPs), as necessary, before it enters the Prima Deshecha stream channel. Consequently, water quality issues, if any, will be addressed prior to discharge.

Groundwater

Local well water quality data indicates a failure to meet EPA drinking water standards for certain constituents, but is not anticipated to have a negative effect relative to the primary purpose of sustaining existing biological resources along the riparian corridor (GLA 2002).

Gross initial estimates of between 3 and 5 gallons per minute (gpm) would draw down local well water levels, but this is not expected to have a significant impact on local or regional groundwater levels or gradients (GLA 2004). There is some potential for groundwater extraction from the deeper portion of the bedrock underneath Zone 4; however, that potential is dependent on the pumping characteristics of the bedrock and the groundwater availability. The quantity and potential flow rate of the bedrock cannot be determined at this time based on current available data.

To assess the potential availability of deeper groundwater resources, several wells would need to be drilled to depths of up to 1,000 feet below ground surface (bgs). Water availability would be assessed by conducting a series of pumping tests in strata identified as having water production potential. Data obtained from the pumping well and nearby observation wells would be used to determine the quantity of available water, the approximate geometry of the

groundwater resource (to determine if deep extraction would affect other users of this resource), the permeability of the water-bearing strata, and the extractability of the groundwater. This field information would then be used to develop a detailed hydrogeologic model to predict the sustainability of the resource based on pumping rates that would be required to provide sufficient water to support the habitat mitigation areas.

However, as flow rates from the existing spring are very modest in volume, it is anticipated that adequate quantities of groundwater would be available to replace these spring flows.

Biological Resources

Alternative 2 will have a permanent impact to five LBV territories and all the special status plant species identified by Alternative 1. This alternative also requires re-establishment of the Prima Deshecha Cañada spring after landslide stabilization measures are implemented, which may provide habitat suitable for LBV occupation.

Groundwater resources are likely to be subject to RWQCB Order No. 2001-96 requirements, and extraction capabilities from the site are not clearly understood without additional data collection efforts. Irrigation and other off-site water sources are not considered 'natural' flows and are not favored for environmental restoration or mitigation purposes. However, it should be noted that the existing natural hydrology is not considered adequate for existing habitat resources.

Other

Landfill Capacity. There are no negative impacts to landfill capacity with this alternative.

Cost Considerations. Mitigation costs would be higher for Alternative 2 than those for the Proposed Project based upon greater impacts to the LBV, the Prima Deshecha Cañada channel, and the channel's associated biological resources. In addition, installation of groundwater wells, pipelines to convey irrigation water, reservoirs, and irrigation lines will raise project costs. Data collection for groundwater extraction efforts would also be costly.

7.2.4 ALTERNATIVE 3: MODIFY ZONE 4 FOOTPRINT TO AVOID PERMANENT IMPACT TO THREE LEAST BELL'S VIREO TERRITORIES

Description of Alternative 3

Alternative 3 proposes a design to shift the Zone 4 grading plan east in order to place the graded slopes outside portions of the Prima Deshecha Cañada stream channel. The Zones 1 and 4 desilting basin is also moved to an off-line location outside the existing streambed and riparian area. This alternative (Exhibit 7.2-3) is proposed to avoid direct impact to three occupied LBV territories located between Zones 1 and 4. It should be noted that the western territory (shown on Exhibit 7.2-3) will be temporarily impacted with the future development of Zone 1 and the associated landslide remediation measures. This temporary impact is anticipated to occur by approximately 2012. Alternative 3 significantly reduces landfill capacity by approximately 24 percent; accordingly, Zone 4 bottom grades are deeper to restore this lost capacity.



W - E	1,100	550	0	1
V				
9				

Impacts of Alternative 3

Geotechnical Considerations

The deeper landfill bottom required to reclaim capacity impacted by this alternative may require a larger shear key design in Zone 4 in order to stabilize the additional refuse mass over the existing landslide area. Significant uncertainties relative to stability of existing subterranean landslide complexes and potential impacts on local and regional groundwater resources would require extensive technical studies over an extended period of time to obtain the data needed for feasibility determination.

Hydrology and Water Quality

The deeper landfill bottom required to reclaim the capacity impacted by this alternative will require regulatory approval of a landfill liner system design below historic groundwater elevations. The landfill bottom grading proposed for the current Zone 4 landfill meets the 27 CCR requirements for separation of groundwater from the liner system. A deeper landfill will require a larger subdrain system to collect groundwater and regulatory approval for an alternative liner system. Additional construction costs would also be incurred for this alternative due to complications associated with the generation of additional groundwater during excavation.

Biological Resources

Alternative 3 will have a permanent impact to two LBV territories and a temporary impact during basin construction and landslide remediation on a portion of one territory. It should be noted that even though this alternative avoids a direct permanent impact to the three existing LBV territories, a portion of one of the three would still be indirectly impacted by changed sunlight exposure caused by modification of adjacent landfill slopes and progressive dewatering of the alluvium supporting the habitat associated with subdrain flows. Supplemental water would still be required to sustain the habitat that supports the remaining LBV territories.

Other

Landfill Stockpile Operations. There are no soil stockpile areas available on site that are large enough, provide a stable foundation area, or have not already been allocated for specific use to accommodate the sizable volume of soil that would be generated with this alternative (by going deeper) to offset the projected capacity loss. Transportation of this large volume of soil to currently undeveloped off-site disposal or stockpile locations would result in potentially significant adverse environmental effects to current and future traffic conditions; air quality associated with truck traffic; potential biological impacts on undeveloped, off-site areas; and economic effects.

Landfill Capacity. Alternative 3 will significantly decrease the refuse capacity of the landfill, resulting in a loss of an estimated 20.8 million cubic yards (mcy) of refuse capacity unless bottom grades are substantially deepened. With an average density of 1,333 pounds per cubic yard (cy), or 0.67 ton per cy, the loss is estimated to be approximately 14 million tons, or about 11 years of landfill capacity. This would be a 24 percent reduction in refuse capacity.

Cost Considerations. The economic impact from the refuse capacity reduction associated with this alternative is the additional cost to the rate payer for alternate means of refuse disposal. The cost in today's dollars due to the lost capacity at the Prima Deshecha Landfill could be the difference between the current cost for disposal (\$22/ton) and the cost for alternate disposal. It is assumed that the only alternate disposal option that would be available when the Prima

Deshecha Landfill reaches the reduced capacity limit associated with this alternative would be rail haul, which is estimated to cost \$75/ton in today's dollars (Stirrat 2006). Based on these assumptions, the additional cost that could be passed on to the rate payer would be approximately \$742 million.

An evaluation was also made of offsetting all the capacity loss with a deeper excavation. Based on a landfill design for Zone 4, which balances the air space capacity with the soil requirements for the landfill operation, a deeper excavation to offset capacity loss would result in approximately 15 to 20 mcy of excess soil to be managed on site or hauled off site. There are no areas available on the site that: (1) are large enough to provide a stable foundation and (2) have not already been allocated for specific use to accommodate the significant volume of excess soil generated by this alternative.

The estimated cost, in today's dollars, for transporting 15 to 20 mcy of stockpile material to an off-site location is approximately \$10/cy for a total of \$150 to \$200 million. This cost estimate does not consider additional costs associated with property acquisition, compaction, or site cleaning costs. These estimates include the costs of handling, loading, and trucking this material to a location within a 10-mile radius of the site, assuming that a suitable location can be found.

7.2.5 ALTERNATIVE 4: SHIFT ZONE 4 FOOTPRINT SOUTHWEST FOR RECHARGE PURPOSES

Alternative 4 (Exhibit 7.2-4) was developed to reduce the impacts of the landfill on the eastern portion of Zone 4, which functions as the spring recharge area for the spring that feeds the Prima Deshecha Cañada stream.

Description of Alternative 4

Alternative 4 proposes a design to shift the Zone 4 landfill footprint southwest approximately 300 feet. In order to maintain capacity, the revised design plan would entail excavating deeper, and filling higher along La Pata Avenue without exceeding the top elevation limit of 1,010 above mean sea level (msl). The depth of the landfill is determined by the amount of soil excavation required for landfill cover operations. Any soil in excess of what will be used during landfill operations needs to be stockpiled somewhere on the site, outside the limits of the landfill area. Areas available for stockpiling on site are limited and the cost and potential environmental impacts of transporting soil off site are high, as discussed in Alternative 3. The deck area for the landfill in Alternative 4 is smaller and narrower than the 2001 GDP Zone 4 design and may also impact landfill operations, viewshed, and end use potential.

Impacts of Alternative 4

Geotechnical Considerations

Shear Key Construction. By shifting the landfill to the southwest, more mass is placed over the upper two thirds of a massive landside complex where a substantial shear key or buttress fill is currently proposed for the 2001 GDP Zone 4 design. A larger shear key or buttress may be required under this alternative in order to accommodate the additional refuse load and deeper excavation. An analysis of potential landslide remediation design requirements can be conducted if Alternative 4 is deemed the least environmentally damaging alternative (GLA 2004).

Detention/Desilting Requirements. The northeasterly edge of Zone 4 would be shifted approximately 1,000 feet southwest which would reduce direct impacts to the drainage area of



N			ai inipaoti i	
W- C	1,100	550	0	1,1
s				

the spring. An outlet channel would be constructed along the entire southern edge of the landfill, eventually turning north to connect with the Zone 4 desilting basin. Accordingly, the southwestern edge of the landfill (southwest) shifts 300 feet and the northwestern edge shifts (southwest) 1,000 feet. The channel would have to convey larger flows than the current Zone 4 2001 GDP design. With Alternative 4, more of the watershed in the northeastern potion of the site would remain in its natural state resulting in an increase in storm water flows to be conveyed around the southerly perimeter of the landfill. It is estimated that an additional 120 cubic feet per second (cfs) of surface flows would need to be managed in this area. Surface detention/desilting basins could be constructed outside the perimeter of the refuse footprint to better manage these flows. The basin(s) would need to be lined to prevent impacts on the landfill liner system.

Alternative 4 avoids direct impacts to approximately 120 acres of natural watershed to the east of the refuse area and could therefore continue to provide groundwater recharge. This area consists of six natural canyons that would remain undisturbed. However, drainage from the canyons would be blocked with the alternative landfill configuration. Since ponding of water behind the Zone 4 landfill liner containment system does not comply with landfill regulatory requirements, interceptor drains or other drainage improvements would need to be designed and constructed for the canyons to collect and remove ponded water from behind the landfill liner system in order to maintain the integrity of this system.

Accordingly, groundwater that accumulates within five feet of the refuse prism will need to be collected and transported in the subdrain system to the toe of the landfill. The estimated flow rate and quantity of recharge water available for transport through a subdrain system cannot be determined based on existing information. It is likely, however, that coupled with a collection of surface water flows, the additional subdrain flows from this portion of the drainage basin would approximate or exceed existing stream flows. Higher flows through the subdrain system would be problematic as discussed below.

Design Requirements for the Sub-drain System. The current design of the landfill subdrain system must address conveyance of groundwater underneath the landfill. A larger number of subdrains, more efficient drainage media, and design modifications to include cleanouts would be necessary to ensure that a free-draining condition is maintained over an extended period. The potential for leakage from the subdrain system is higher with this alternative than the current design which could negatively affect the shear key stabilization fill directly under the southwesterly end of the landfill. This could significantly increase system design complexity and reduce the factor of safety to below an acceptable level.

Significant subdrain enhancements will be required to maintain, over an indefinite period, a 5-foot separation between the groundwater and the refuse prism at the bottom of the proposed landfill, as required by 27 CCR. Normally, a subdrain system is designed to remove foreseeable, short-term groundwater flows in order to maintain a five-foot separation between groundwater and refuse. Typically, the amount of water conveyed in a subdrain system decreases over time as the expanding refuse footprint decreases the recharge area and groundwater elevations decrease. A subdrain system is not intended to be used as an aqueduct. Large water flows moving through a subdrain system on a continual basis have the potential to erode or damage system components, damage the overlying composite liner system, and compromise the stability of the landfill. Water conveyance for the purpose of habitat maintenance or other uses not involved in the protection of groundwater quality would entail less design complexity and technical risk if provided in aboveground channels where the system can be easily monitored and maintained.

Biological Resources

Alternative 4 will have a permanent impact to 2 full LBV territories and a very small portion of 2 other LBV territories during the first few phases of the Zone 4 implementation. This alternative involves a shift in the Zone 4 footprint to the southwest to preserve the recharge area that provides hydrology to support the riparian habitats located downstream within Prima Deshecha Cañada channel. Although this alternative is intended to reduce potential impacts to the channel, the project will interrupt the natural stream flow and subsequently affect the hydrology within Prima Deshecha Cañada channel, which is known to support ten pairs of LBV. Therefore, supplemental water would likely be required to sustain the habitats located within the remaining natural and created riparian habitats located immediately downstream. Also, impact to special status species would be the same as Alternatives 1, 2, and 3.

Other

Landfill Capacity. The Alternative 4 design plan results in a reduction of approximately 6 mcy of refuse capacity from the 2001 GDP Zone 4 landfill design, even with a deeper excavation (approximately 70 feet) and higher fill slopes along the westerly edge. Assuming an average density of 1,333 pounds per cy or 0.67 ton per cy, the refuse capacity loss is equivalent to approximately 4 million tons. This translates to 7 percent of the total refuse volume, or over 3 years of landfill capacity.

The economic impact from the refuse capacity reduction associated with this alternative would involve additional costs to county rate payers for alternate means of refuse disposal. The cost in today's dollars to the rate payers due to the lost capacity at the Prima Deshecha Landfill could be the difference between the current cost (\$22/ton) and the cost for alternate disposal. It is assumed that the only alternate disposal option that would be available when the Prima Deshecha Landfill reaches the reduced capacity associated with this alternative would be rail haul, which is estimated to cost \$75/ton in today's dollars (Stirrat 2006) reference Los Angeles County Sanitation Districts). Using these assumptions, the additional cost that would be passed on to the rate payer is approximately \$212 million. In addition to economic impacts of reduced refuse capacity at the site, there will be numerous environmental impacts associated with transport and disposal at an out-of-county facility. This loss of capacity is also inconsistent with one of the project's primary objectives.

Cost Considerations. Additional costs would be incurred through mitigation of temporary impacts associated with construction and would be significantly higher than the implementation of a Pre-mitigation Plan that would establish habitats which would be lost to landfill operations well in advance of the loss. Pre-mitigation would allow for mitigation at a 1:1 ratio as opposed to mitigation at higher costs and on a phase-by-phase basis at higher ratios. In addition, geotechnical investigations would be required to refine design elements. Also, costs would likely be higher to address the increased shear key or buttress fill requirements for slope stability. Costs would also be incurred for an engineered liner and more complex subdrain system.

7.3 <u>ENVIRONMENTALLY SUPERIOR ALTERNATIVE/LEAST ENVIRONMENTALLY</u> <u>DAMAGING PRACTICABLE ALTERNATIVE</u>

Analysis of feasible alternatives to the Proposed Project indicates that the Proposed Project is the Environmentally Superior Alternative, as considered under CEQA, and is the Least Environmentally Damaging Alternative, as considered under Section 404 of the Clean Water Act. The project purpose and need reflects the objectives of reducing impacts to biological resources, enhancing site stabilization, and reducing potential future negative effects to landfill operations and on-site environmental resources. Formulation of the Proposed Project was guided by the desire to minimize environmental impacts of landfilling activities and to maximize environmental enhancement and protection outputs at the project site. This analysis of the alternatives developed for the Proposed Project concludes that the alternatives provide reduced benefits to local and regional biological resources and/or impacts to other resource categories. As the Proposed Project has been expressly formulated to maximize environmental benefits through proactive pre-mitigation planning and to minimize impacts to on-site biological resources associated with landfill operations through re-design of project features, it has been identified as both the Environmentally Superior Alternative and the Least Environmentally Damaging Practicable Alternative.

SECTION 8.0 CUMULATIVE IMPACTS

Section 15130 of the CEQA Guidelines requires that a project's cumulative impacts be discussed when the incremental effect is "cumulatively considerable." According to CEQA Guideline §15065(c), the term "cumulatively considerable" means "...that the incremental effects of an individual project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects...". Specifically, CEQA Guidelines, Section 15355 defines cumulative impacts as "...two or more individual effects which, when considered together, are considerable or which compound or increase other environmental impacts."

SEIR 597 is a supplement to EIR 575, which contained a thorough cumulative impact analysis for the 2001 GDP, as amended. As has been demonstrated throughout this SEIR, the impacts from Amendment No. 2 will not incrementally increase any environmental impacts that cannot be mitigated to a less-than-significant level, other than air quality (please refer to Section 5.4 for a detailed discussion on this resource). Impacts from implementation of Amendment No. 2 to the GDP are relatively minor, and will not induce any cumulative effects beyond those identified in EIR 575 for the 2001 GDP.

La Pata Avenue Gap Closure Project: Sensitivity Analysis

This roadway project would construct an extension of La Pata Avenue from Ortega Highway to Calle Saluda in San Clemente. The County of Orange Resources Development and Management Department (RDMD) is in the process of preparing an EIR for this project. In April 2005, the County of Orange completed a Feasibility Study to evaluate two potential roadway alignments through the PDL property, herein designated as "western" and "eastern" alignments. The "western" alignment shown in the Feasibility Study has since been modified in response to technical and neighborhood constraints and is shown in Exhibit 8.1-1. Subsequently, the County intends to prepare a Project Report to refine project construction elements (such as grading and drainage) as the basis for environmental analysis.

In an effort to ensure close coordination between the La Pata Avenue Gap Closure Project and the Proposed Project for Amendment No. 2 to the 2001 GDP at the PDL, the County RDMD and the IWMD have met with representatives of the USACE, USFWS, CDFG, and RWQCB to review site conditions and discuss issues concerning resources under their respective jurisdictions. Based on these regulatory agency discussions (which are considered preliminary assessments for both conceptual alignments and associated grading for landslide stabilization/remediation), the following potential impacts have been identified that will be addressed in the La Pata Avenue Gap Closure Project EIR. It should be noted that the construction of the La Pata Avenue Gap Closure Project is currently projected to occur around 2013, and may therefore occur in advance of any landfilling-related activities within Zone 4 of the Proposed Project. Some of the areas requiring landslide remediation by RDMD for road construction will overlap areas requiring this same or similar treatment by the IWMD in preparation for landfilling operations in Zone 4. As such, only areas that are outside these common landslide remediation/stabilization areas required for the construction of La Pata Avenue and that were not previously identified by the SEIR for Prima are identified below:

• <u>Biological</u>: Implementation of either the eastern or western alignments may involve impacts to approximately 38.0 acres of existing coastal sage scrub (CSS) outside the areas previously addressed by this SEIR. Also, project implementation would result in impacts to approximately 13.0 acres of recently planted CSS and approximately 0.45 acre of freshwater marsh within the Site A portion of the Phase B Landslide Remediation Bio-Mitigation areas of the Prima Deshecha Landslide Remediation Project. The project

has the potential to also impact approximately 6.1 acres under the jurisdiction of the CDFG, including approximately 3.05 acres under USACE jurisdiction within the Prima Deshecha Cañada channel and its associated tributaries (in addition to that acreage impacted by Amendment No. 2). Project implementation may also impact three least Bell's vireo (LBV) territories, one California gnatcatcher territory, and an area containing the small-flowered morning glory (Convuvulus simulans). Portions of the roadway may affect existing wildlife movement and use along: (1) the southern boundary of the landfill which connects existing open space areas east of the project with the Talega and Forster Ranch developments; designated open space areas within Segunda Deshecha; (2) the southern portions of Zones 1 and 4; and (3) San Clemente open space areas located south and west of the project site. In addition, the roadway project may potentially affect the hydrological conditions/spring flows that support the existing and newly created riparian habitat resources within the Prima Deshecha Cañada channel. Lastly, the roadway project may impact mitigation restoration sites within the project site (e.g., CSS and freshwater marsh) and at both the Talega and Forster Ranch developments.

Mitigation for the La Pata Avenue project is being coordinated between the RDMD and the IWMD. CSS mitigation opportunities for this project have been identified on site within Segunda Deshecha, and riparian and jurisdictional mitigation opportunities have been identified off site within the San Juan Creek portions of Ronald M. Caspers Wilderness Park (opportunities which are in addition to those contained within Amendment No. 2's Proposed Project Pre-Mitigation Plan).

- <u>Geological</u>: Existing landforms would be altered and the roadway would traverse an area with extensive landslides.
- <u>Air Quality</u>: Short-term air quality impacts related to temporary construction emissions would occur; however, the project may result in long-term air quality benefits by reducing the long-term emissions associated with congestion.
- <u>Noise</u>: There would be both short-term and long-term noise impacts, though the significance of these impacts would be dependent upon the alignment selected and its proximity to noise-sensitive receptors. These impacts would also include potential effects on LBV and California gnatcatcher breeding and nesting activities in habitat areas located immediately adjacent to the newly constructed roadway.
- <u>Aesthetics</u>: The project would potentially result in a substantial amount of grading, with potential aesthetic impacts.

Alternate La Pata Avenue alignments were considered in formulating the Proposed Project and its alternatives. The RDMD and the IWMD have conducted frequent coordination meetings to discuss the alignments that will be considered in the La Pata Avenue Feasibility Study in an effort to resolve any potential conflicts between the two projects. These meetings also provided an opportunity to formulate on-site and off-site mitigation programs for both the Proposed Project and the La Pata Avenue Gap Closure Project. Mitigation program discussions considered wildlife corridor connections from open space areas east and south of the Proposed Project site (including Talega and Forster Ranch) to the southern edges of the PDL property, and the San Clemente open space to the west of the Proposed Project. These alignments are illustrated on Exhibit 8.1-1.

Close coordination of these 2 projects have indicated that implementation of the Proposed Project for Amendment No. 2 will neither preclude the design, construction, or operation and



La Pata Avenue Potential Roadway Alignments

Prima Deshecha Landfill Supplemental Environmental Impact Report 597

Exhibit 8.1–1



R:/Projects/OCIWMD/J004/SEIR597 Public Review Draft\graphics\ex8.1-1_la_pata_083006.pdf

maintenance requirements for either potential roadway alignment nor will it impact implementation of the Proposed Project Pre-mitigation Plan.

SECTION 9.0 GROWTH-INDUCING IMPACTS, UNAVOIDABLE ADVERSE IMPACTS, AND SIGNIFICANT IRREVERSIBLE ENVIRONMENTAL CHANGES

9.1 GROWTH-INDUCING IMPACTS

Section 15126.2(d) of the California Environmental Quality Act (CEQA) Guidelines requires that an Environmental Impact Report (EIR) describe the potential growth-inducing impacts of a proposed project. Specifically, Section 15126.2(d) state that a project must:

Discuss the ways in which the proposed project could foster economic development or population growth, or the construction of additional housing, either directly or indirectly, in the surrounding environment.... [It should] also discuss the characteristics of some projects which may encourage and facilitate other activities that could substantially affect the environment, either individually or cumulatively. It must not be assumed that growth in any area is necessarily beneficial, detrimental or of little significance to the environment.

Implementation of the Proposed Project will not involve an increase in landfill capacity or significant changes in landfill operations over the long-term; as a result, the number of employees at the PDL will not change substantially with implementation of the Proposed Project. Employees will continue to perform similar landfill operations including administration, landfill cover operations, and other landfill-related operations in order to support operations covered under EIR 575. The numbers and types of equipment used at the PDL will not change substantially as a result of the Proposed Project, nor will the days of operation or schedule of the facility change substantially.

The Proposed Project will not, in itself, be an inducement to growth, as the improvements under the Proposed Project would not entail new residences or the extension of major infrastructure facilities (i.e., sewer or water lines, roads) that would result in secondary or indirect growth in and around the area. In addition, waste disposal needs for the region have been considered for the project based on the County of Orange in the *Countywide Integrated Waste Management Plan* (CIWMP), which assesses existing and approved development and ultimate General Plan buildout for southern Orange County. As the Proposed Project does not include elements that alter refuse capacity at the site; create permanent employment opportunities; result in the extension of major infrastructure facilities; or affect regional housing trends, there will be no incremental growth-inducing effects as a result. This issue was adequately addressed in EIR 575 for the 2001 GDP as amended.

9.2 UNAVOIDABLE ADVERSE IMPACTS

Section 15126(b) of the CEQA Guidelines requires that an EIR "describe any significant impacts, including those which can be mitigated, but not reduced to a level of insignificance. Where there are impacts that cannot be alleviated without imposing an alternative design, their implications and the reasons why the project is being proposed, notwithstanding their effect, should be described."

All project impacts can be mitigated to a level that is less than significant except those for air quality. Implementation of updated mitigation measures and best available control technology will minimize this impact to the maximum extent practicable, but not below significance thresholds.

9-1

As described in Section 5.4, Air Quality, the Prima Deshecha Landfill is currently implementing several mitigation measures to reduce potential air quality impacts. Implementation of the mitigation measures described in Section 5.4.4 will help to further reduce air quality impacts that result from operations at the Prima Deshecha Landfill. However, even with implementation of all existing and recommended mitigation measures, operations at the Prima Deshecha Landfill would continue to result in significant and unavoidable air quality impacts.

9.3 SIGNIFICANT IRREVERSIBLE ENVIRONMENTAL CHANGES

Section 15126.2(c) of the CEQA Guidelines requires that an EIR discuss significant adverse irreversible environmental changes that would be caused by implementation of the proposed project. In addition, irretrievable commitments of resources should be evaluated. Implementation of the proposed project will result in both short- and long-term commitments of natural resources.

Implementation of the Proposed Project will not result in significant changes in quantities of building materials over those assessed within the 2001 GDP and EIR 575. Proposed project elements consist largely of incremental modifications to the 2001 GDP that will be accomplished concurrently with approved site-preparation activities and will not consume significant quantities of additional resources. Those materials that are utilized for implementation of proposed landslide remediation measures and the Zone 4 desilting system are anticipated to be in adequate supply into the foreseeable future. In addition, several elements of the Proposed Project are oriented toward environmental enhancement or restoration and will not consume resources for construction. The Proposed Project will not result in the wasteful, inefficient, or unnecessary consumption of energy during construction. Therefore, no significant impacts from energy consumption would occur.

The Proposed Project for Amendment No. 2 will not result in any significant impacts to local or regional energy supplies; will not impact peak or base energy standards; and will not violate existing energy standards or result in significant impacts to energy resources. The Proposed Project will result in the irretrievable and irreversible commitment of energy resources in the form of diesel fuel, gasoline, and electricity. However, these uses are part of normal operations and are not considered a substantial or a wasteful use of resources. In addition, these types of resources are anticipated to be in adequate supply into the foreseeable future. Therefore, impacts are not considered significant.

Water will be utilized for dust-control and watering activities that will occur as a result of the Proposed Project. These uses are currently in effect at the site and are required for sitemitigation responsibilities; therefore, the Proposed Project is not anticipated to incrementally increase these needs to a significant level. These are not considered to be wasteful, inefficient, or unnecessary uses of water. As sources of water for the Proposed Project are available and anticipated to be in adequate supply into the foreseeable future, impacts due to this irretrievable and irreversible commitment of resources are not considered significant.

SECTION 10.0 PERSONS AND ORGANIZATION CONSULTED/LIST OF PREPARERS

10.1 PERSONS AND AGENCIES CONSULTED

FEDERAL GOVERNMENT

U.S. Army Corps of Engineers

Cori Ferrar, Environmental Scientist, Project Manager Jason Lambert, Biologist, Project Manager

U.S. Fish and Wildlife Service

Ken Corey, NCCP/HCP Conservation Planning Jill Terp, Fish and Wildlife Biologist

STATE OF CALIFORNIA

California Department of Fish and Game

Don Chadwick, Supervisor, Streambed Group Donna Cobb, Biologist, Streambed Group Warren Wong, Biologist, Planning Conservation

LOCAL GOVERNMENT

County of Orange

Resources and Development Management Department (RDMD)

Tim Neely, Director of Planning Harry Persaud, Senior Planner, Transportation Section

Integrated Waste Management Department (IWMD)

Suzanne McClanahan, Manager, Office of Public Affairs Mike Giancola, Deputy Director, Landfill Operations, Prima Deshecha Landfill John Tzeng, Site Engineer Jim Pfaff, Administrative Analyst, Office of Public Affairs Rochelle Carpenter, Project Manager Linda Hagthrop, Public Information Officer George Ker, A-E Project Manager, Prima Deshecha Landfill Oscar Velastegui, Senior Civil Engineer, Prima Deshecha Landfill Kevin Kondru, Manager, Regulatory Compliance Section, Environmental Services John Arnau, Planner, Regulatory Compliance Section, Environmental Services

COMPANIES AND INDIVIDUALS

Bryan A. Stirrat & Associates, Inc.

Christine Arbogast, Senior Vice President R. J. Wilson, Civil Engineer

GeoLogic Associates, Inc.

Gary Lass, Principal John Hower, Senior Geologist

10.2 PREPARERS AND CONTRIBUTORS

CONSULTANTS

BonTerra Consulting

Joan Patronite Kelly, Managing Principal Pam Castens, Senior Project Manager Gary Medeiros, Associate Principal, Regulatory Services Ann Johnston, Principal, Biological Services Amber Oneal, Project Manager, Biological Services Brian Daniels, Senior Biologist Sandy Leatherman, Senior Biologist Melissa Howe, Associate Principal, Restoration Ecologist Richard Lewis, Senior Restoration Ecologist Cindy Krebs, Senior Project Manager, Air Quality Diane Barrett, Environmental Planner Jenni Gómez, GIS Specialist Chris Starbird, GIS Specialist Johnnie Garcia, GIS Technician Julia York, Technical Writer Kathy Linklater, Word Processor Susan Canino, Administrative Assistant Heidi Hollstein, Administrative Assistant

SECTION 11.0 LIST OF REFERENCES

- Arbogast, C.M. 2006 (August 24). Personal communication. Email from C. Arbogast (Bryan A. Stirrat & Associates) to P. Castens (BonTerra Consulting).
- Arnau, J. 2005 (July 12). Personal communication. Telephone conversation between J. Arnau (IWMD) P. Castens (BonTerra Consulting).
- Atwood, J.L. 1992. Rare, Local, Little-Known, and Declining North American Breeders A Closer Look. *Birding* 25: 228-233. Colorado Springs: American Birding Association.
- —. 1990. *Status Review of the California Gnatcatcher (*Polioptila californica). Manomet, MA: Manomet Bird Observatory.
- Beedy, E.C., and W.J. Hamilton III. 1999. Tricolored Blackbird (Agelaius tricolor). The Birds of North America, No. 423 (A. Poole and F. Gill, eds.). Philadelphia: The Birds of North America, Inc.
- Bloom, P.H. 1996. Personal communication regarding raptor data on Rancho Mission Viejo.
- BonTerra Consulting. 2006a (September). *Proposed Pre-Mitigation for Prima Desecha Landfill*. Costa Mesa, CA: BonTerra Consulting.
- _____. 2006b (September). Regional Environmental Enhancement Opportunities Plan for Prima Desecha Landfill. Costa Mesa: BonTerra Consulting.
- 2006c (September). Results of Special Status Plant Surveys for the Prima Deshecha Landfill, San Juan Capistrano, San Clemente, and Unincorporated Orange County, California (A Letter Report to J. Tzeng, Prima Desecha Landfill, from G.A. Medeiros and M.A. Howe, BonTerra Consulting). Costa Mesa: BonTerra Consulting.
- _____. 2002a (May). Coastal Sage Scrub Mitigation Program, Prima Desecha Landslide Remediation Project, Modified Preferred Alternative. Costa Mesa: BonTerra Consulting.
- _____. May 29, 2002b. *Riparian Habitat Mitigation Program, Prima Deshecha Landslide Remediation Project, Modified Preferred Alternative.* Costa Mesa: BonTerra Consulting.
- _____. 2000a. *Biological Survey for Prima Deshecha Landfill.* Costa Mesa: BonTerra Consulting.
- _____. 2000b. Capistrano Valley Water District SC-04 south connector pipeline project: Biological technical report (prepared for David Evans and Associates). Costa Mesa: BonTerra Consulting.
- Bramlet, D.E. 2004. *Botanical Survey for Special Status Plant Species at the Prima Deshecha Landfill.* (prepared for BonTerra Consulting) Pomona, CA: California Polytechnic University.
- _____. (Compiler). 2003. Plant species in the 6th edition of the CNPS Inventory, and Species of Local Concern in Orange County. Pomona: California Polytechnic University.
- _____. 2000. Plant *communities of the Prima Deshecha landfill* prepared for BonTerra Consulting). Pomona: California Polytechnic University.

- ____. 1998. *Botanical resources of the Whispering Hills project site* (prepared for Campbell BioConsulting). Pomona: California Polytechnic University.
- Brown, H.A. 1967. Embryonic temperature adaptations and genetic compatibility of two allopatric populations of the spadefoot toad, *Scaphiopus hammondii. Evolution* 21(4): 742–761. Lawrence, KS: Allen Press for the Society for the Study of Evolution.
- _____. 1966. Temperature adaptation and evolutionary divergence in allopatric populations of the spadefoot toad, *Scaphiopus hammondii*. PhD Dissertation, University of California, Riverside, California.
- California Department of Fish and Game (CDFG). 2004a. *California Natural Diversity (RareFind) Database.* Sacramento: CDFG, Natural Heritage Division.
- _____. 2004b (January). List of Special Plants. Sacramento: CDFG, Natural Heritage Division.
- _____. 2003. California Natural Diversity (RareFind) Database. Sacramento: CDFG, Natural Heritage Division.
- . 2002a. <u>California Natural Diversity Database.</u> Records of Occurrence for Dana Point, San Juan Capistrano, San Clemente, Dana Point and Cañada Gobenadora 7.5 Minute Quadrangles. Sacramento: CDFG, Natural Heritage Division.
- _____. 2002b. *California Natural Diversity (RareFind) Database.* Sacramento: CDFG, Natural Heritage Division.
- _____. 2001. *California Natural Diversity (RareFind) Database.* Sacramento: CDFG, Natural Heritage Division.
- _____. 1998. *California Natural Diversity (RareFind) Database.* Sacramento: CDFG, Natural Heritage Division.
- _____. 1994. A Field Guide to Lake and Streambed Alteration Agreements Sections 1600–1607. Sacramento: CDFG, Environmental Services Division.
- _____. 1992. *Coastal Sage Scrub Survey Guidelines* (prepared by the Southern Coastal Sage Scrub Scientific Review Panel). Sacramento: CDFG, Natural Heritage Division.
- California Governor's Office of Planning and Research. 1996. California Environmental Quality Act. Sacramento: California Resource Agency. http://ceres.ca.gov/topic/env_law/ceqa/guidelines/art11.html.
- California Native Plant Society (CNPS). 2004. Inventory of Rare and Endangered Plants (online edition, v6-04c). California Native Plant Society. Sacramento: CNPS. Accessed from http://www.cnps.org/inventory.
- _____. 2001. Inventory of rare and endangered plants of California (Sixth Edition). Rare plant scientific advisory committee (D.P. Tibor, Ed.). Sacramento: CNPS.
- California, State of. 2006. *California Fish and Game Code.* Sacramento: the State. http://www.leginfo.ca.gov/cgi-bin/calawquery?codesection=fgc&codebody=&hits=20.

- California, State of., Assembly. 1996 (September 17). Assembly Bill (AB) No. 3446: Natural community conservation planning: compensation (to amend Sections 2810 and 2840 of the California Fish and Game Code). Sacramento: California Assembly.
- California, State of, Senate. 2000 (July 5). Senate Bill (SB) No. 1679: Environmental protection: conflict of interest of public officials: natural community conservation plans: Cache Creek reclamation: delta protection: flood control (to add Sections 2801, 2811, and 2815 to the Fish and Game Code, to amend Section 1091.5 of, and to add Sections 11361 and 12805.1 to the Government Code, relating to the environment.
- Collins, P.W. 1999. Rufous-crowned Sparrow (*Aimophila ruficeps*). The Birds of North America, No. 472 (A. Poole and F. Gill, Eds.). Philadelphia: The Birds of North America, Inc..
- Dudek & Associates. 2000. Species Accounts for the Western Riverside Multiple Species Habitat Conservation Plan (prepared for Riverside County Integrated Project). Riverside, CA: Riverside Blueprint. http://rcip.org/conservation.htm.
- Dunn, J.L., and K.L. Garrett. 1997. A Field Guide to Warblers of North America. *Peterson's Field Guide Series: 49.* New York: Houghton Mifflin Company.
- Ehrlich, P. R., D.S. Dobkin, and D. Wheye. 1988. The *Birder's Handbook: A Field Guide to the Natural History of North American Birds*. New York: Simon and Schuster.
- Feaver, P.E. 1971. Breeding pool selection and larval mortality of three California amphibians: *Ambystoma tigrinum californiense* Gray, *Hyla regilla* Baird and Girard, and *Scaphiopus hammondii* Girard. MA Thesis, Fresno, CA: Fresno State College.
- Franzreb, K.E. 1989. Ecology and conservation of the endangered least Bell's vireo. Biological Report 89(1). Washington D.C.: USFWS.
- Garrett, K.L., and J.L. Dunn. 1981. *Birds of Southern California: Status and Distribution*. Los Angeles: Audubon Press.
- GeoLogic Associates, Inc (GLA). 2004. An Analysis of Groundwater Resources in Zone 4, Prima Desecha Landfill. Diamond Bar, CA: GLA.
- _____. 2002. Prima Desecha Landfill Geotechnical Investigations Report Zone 4. Diamond Bar: GLA.
- GeoSyntec Consultants. 1999 (April). Semi-Annual Detection Monitoring Report—October 1998–March 1999 (prepared for County of Orange IWMD). Huntington Beach, CA: GeoSyntec Consultants.
- Giroux & Associates. 1999 (July). *Air Quality Impact Assessment Methodology for Air Quality.* Irvine, CA: Giroux & Associates.
- Goldwasser, S. 1981. Habitat requirements of the least Bell's vireo. Sacramento: CDFG.
- Gray, J. and D.E. Bramlet. 1994 (as revised). *Species of special interest* (prepared for County of Orange EMA). Orange County Natural Resources GIS Project. Orange: the County.
- _____. 1992. *Habitat classification system* (prepared for OCEMA). Orange County Natural Resources GIS Project. Orange: the County.

- Grinnell, J., and A.H. Miller. 1944. The distribution of the birds of California. *Pacific Coast Avifauna No. 27.* Waco, TX: Cooper Ornithological Club.
- Hall, E.R. 1981. The Mammals of North America (2nd ed.). New York: John Wiley & Sons.
- Hall, E.R., and K.R. Kelson. 1959. *The Mammals of North America* (Vols. I & II). New York: The Ronald Press Co.
- Hamilton, Robert A. and Douglas R. Willick. 1996. *The Birds of Orange County, California: Status and Distribution*. Irvine: Sea and Sage Audubon Society.
- Hickman, J.C. (Ed.). 1993. *The Jepson Manual: Higher Plants of California*. Berkeley, CA: University of California Press.
- Hix, A.B. 1990. Sensitivity of San Diego's Biological Resources: An Informational Report (prepared for the City of San Diego Planning Department, Development and Environmental Planning Division). San Diego: A.B. Hix.
- Jennings, M.R. and Hayes, M.P. 1994. *Amphibian and Reptile Species of Special Concern in California* (prepared for the CDFG). Rancho Cordova, CA: Jennings and Hayes.
- Johnsgard, P.A. 1990. *Hawks, Eagles and Falcons of North America.* Washington D.C.: Smithsonian Institution Press.
- Lowther, P. E., C. Celada, N.K. Klein, C.C. Rimmer, and D.A. Spector. 1999. Yellow Warbler (*Dendroica petechia*). *The Birds of North America, No. 454* (A. Poole and F. Gill, eds.). Philadelphia: The Birds of North America, Inc.
- Marsden, K.L. and M.G. Simpson. 1999. Eryongonum Pendletonensis (*Apiaceae*), A New Species from Southern California. *Madroño* 46(1): 61–64. Berkeley: California Botanical Society.
- Michael Brandman Associates (MBA). 1998. *Final Natural Environment Study for Foothill Transportation Corridor South* (prepared for Foothill Transportation Agency). Irvine: MBA.
- Munz, P.A. 1974. A Flora of Southern California. Berkeley: University of California Press.
- Orange, County of. 2006. Southern Subregion Natural Community Conservation Plan. Orange: the County. http://www.ocplanning.net/ssnccp/nccp_hcp.aspx.
- Orange, County of, Environmental Management Agency (OCEMA). 1995 (November). Final (Revised) Environmental Impact Report No. 548: Prima Deshecha Landfill General Development Plan. Orange: the County.
- Orange, County of, Integrated Waste Management Department (IWMD). 2004. Supplemental Environmental Impact Report No. 575. Orange: the County.
- _____. 2001 (November). Final Program Environmental Impact Report, 2001 Prima Deshecha General Development Plan. Orange: the County.
- _____. 2001 (January). Draft 2001 General Development Plan: Prima Deshecha Landfill. Orange: the County.

- _____. 1997. Memorandum of Understanding between the City of San Clemente, Orange County Flood Control District, and County of Orange Regarding Prima Deshecha Landfill. Orange: the County.
- _____. 1995. *Memorandum of Understanding between the City of San Juan Capistrano and the County of Orange Regarding Prima Deshecha Landfill.* Orange: the County.
- _____. 1991. North Orange County Landfill and Alternative Technologies Study (NOCLATS). Orange: the County.
- P&D Consultants. 2002. Natural environment study for the south Orange County transportation infrastructure improvement project (prepared for Foothill/Eastern Transportation Corridor Agency). Orange: P&D Consultants.
- Proudfoot, G.A., D.A. Sherry, and S. Johnson. 2000. Cactus Wren (*Campylorhynchus brunneicapillus*). *The Birds of North America, No. 558* (A. Poole and F. Gills, eds.). Philadelphia: The Birds of North America, Inc..
- Rancho Mission Viejo, LLC (RMV). 2004 (October 14). Covenant and Declaration of Restrictions between RMV and County of Orange, Integrated Waste Management Department (RMV Agreements). Orange: the County.
- RBF Consulting. 2004 (June). *Delineation of Jurisdictional Waters, Prima Deshecha Landfill* (prepared for BonTerra Consulting and County of Orange, Integrated Waste Management Department). Irvine: RBF Consulting.
- Rea, A.M. and K.L. Weaver. 1990. The Taxonomy, Distribution, and Status of the Coastal California Cactus Wren. *Western Birds* 21:81–126. Los Angeles: Western Field Ornithologists.
- Reiser, G.H. 2001. Rare plants of San Diego. San Diego: Aquifer Press.
- Remsen, J. V., Jr. 1978. Bird Species of Special Concern in California: An Annotated List of Declining or Vulnerable Bird Species (Administrative Report No. 78-1). Sacramento: CDFG, Wildlife Management Branch.
- South Coast Air Quality Management Board. 2003. 2003 Air Quality Management Plan. Diamond Bar, CA: SCAQMD. http://www.aqmd.gov/aqmp/AQMD03AQMP.htm.
- Stebbins, R.C. 1985. A Field Guide to Western Reptiles and Amphibians (2nd ed.). Boston: Houghton-Mifflin Company.
- _____. 2003. A Field Guide to Western Reptiles and Amphibians (3nd ed.). Boston: Houghton-Mifflin Company.
- Stirrat, B. 2006. Personal Communication. Telephone conversation between B. Stirrat (Brian A. Stirrat & Associates) and A. Mallick (LA County Sanitation Districts).
- Tanner, W.W. 1988. Eumeces skitonianus. Catalogue of American Amphibians and Reptiles (447):1–4. Salt Lake City: Society for the Study of Amphibians and Reptiles.
- Unitt, P. 1984. *The Birds of San Diego County (Memoir 13).* San Diego: San Diego Society of Natural History.
- U.S. Army Corps of Engineers (USACE). 1987. Army Corps of Engineers Wetlands Delineation Manual (Technical Report Y–87-1). Vicksburg, MS: U.S. Army Engineer Waterways Experiment Station, Environmental Laboratory.
- U.S. Congress. 1978 (as amended). *16 United States Code 668–668d, 54 Stat. 250* (The Bald Eagle Protection Act of 1940). Washington D.C.: U.S. Government Printing Office.
- ———. 1977. 33 United States Code (Section 1344 et seq., U.S. Clean Water Act). Washington D.C.: U.S. Government Printing Office.
- U.S. Fish and Wildlife Service (USFWS). Designation of Critical Habitat for the Coastal California Gnatcatcher (*Polioptila californica californica*) In Los Angeles, Orange, Riverside, San Bernardino, San Diego, and Ventura Counties, and Determination of Distinct Vertibrate Population Segment for the California Gnatcatcher (*Polioptila californica*) – Proposed Rule. *Federal Register* 68(79): 20227–20312.
- . 2000a (October 24). Endangered and Threatened Wildlife and Plants: Final Determination of Critical Habitat for the Coastal California Gnatcatcher- Final Rule. *Federal Register* 65 (206): 63679–63743. Washington D.C.: USFWS.
- 2000b (September 21). Endangered and Threatened Wildlife and Plants; Proposed Designation of Critical Habitat for the Riverside Fairy Shrimp. *Federal Register* 65(184): 57136–57159. Washington D.C.: USFWS.
- . 1997 (July 22). Endangered and Threatened Wildlife and Plants: Final Determination of Critical Habitat for the Southwestern Willow Flycatcher. *Federal Register* 62(140): 39129–39147.
- _____. 1995 (February 27). Endangered and Threatened Wildlife and Plants: Final Rule, Determining Endangered Status for the Southwestern Willow Flycatcher. *Federal Register* 60(38): 10693–10715.
- _____. 1994 (February 2). Endangered and Threatened Wildlife and Plants; Designation of Critical Habitat for the Least Bell's Vireo. *Federal Register* 59 (22): 4845–4867.
- _____. 1986 (May 2). Endangered and Threatened Wildlife and Plants; Determination of Endangered Status for the Least Bell's Vireo. *Federal Register* 51: 16474–16481.
- Wachtell, J.K. 1978. Soil survey of Orange County and western part of Riverside County, California. Washington, D.C.:USDA, Soil Conservation Service and Forest Service.
- Whitaker, Jr., J.O. 1980. *The Audubon Society Field Guide to North American Mammals.* New York: Alfred A. Knopf, Inc.
- Wilcox, B.A. & D.D. Murphy. 1985. Conservation Strategy: the Effects of Fragmentation on Extinction. *American Naturalist* 125:879–887. Chicago: University of Chicago Press.
- Williams, D.F. 1986. *Mammalian Species of Special Concern in California* (Administrative Report 86-1). Sacramento: CDFG.
- Zeiner, D.C., W.F. Laudenslayer Jr., K.E. Mayer, M. White, eds. 1990. *California's Wildlife Vol. 3: Mammals.* Sacramento: CDFG, The Resources Agency.

DRAFT SUPPLEMENTAL ENVIRONMENTAL IMPACT REPORT 597 SECOND AMENDMENT TO THE 2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN

VOLUME 2

Prepared for:

Orange County Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, CA 92703

> Contact: Rochelle Carpenter (714) 834-4060

> > Prepared by:

BonTerra Consulting 151 Kalmus Drive, Suite E-200 Costa Mesa, CA 92626

Contact: Joan Patronite Kelly, AICP (714) 444-9199





APPENDIX A

Notice of Preparation, February 9, 2004 Notice of Preparation Comment Letters



NOTICE OF PREPARATION

February 9, 2004 Date: Notice of Intent to Prepare Draft Supplemental Environmental Impact Subject: Report # 597 Project Title: Second Amendment to 2001 Prima Deshecha General Development Plan County of Orange Integrated Waste Management Department Applicant:

The County of Orange has determined that a Supplemental Environmental Impact Report (SEIR) is necessary for the subject project. The County will be the Lead Agency for the subject project and will prepare the SEIR. In order for the concerns of your agency to be incorporated into the Draft SEIR, we need to know the views of your agency as to the scope and content of the environmental information relevant to your agency's statutory responsibilities in connection with the proposed project. Your agency must consider the SEIR when considering your permit or approval for the project. The project description, location, and an analysis indicating the probable environmental effects of the proposed action are contained in the attached materials.

Pursuant to Section 21080.4 of CEQA, your response must be sent not later than 30 days after receipt of this notice. Please mail your response to Robert Richmond at the following address:

> Integrated Waste Management Department Fax: (714) 834-4002 320 N. Flower Street, Suite 400 Santa Ana, CA 92703

Public Information Officer: Linda Hagthrop Phone: (714) 834-4176

Attachment: Project Description/ **Environmental Effects** Submitted by:

Abut Rialman

G:\Env Svcs\GCR_RC\Prima 2nd Amendment\02-06-04 NOP Cover.doc







NOTICE OF PREPARATION FOR SUPPLEMENTAL DRAFT EIR 597

SECOND AMENDMENT TO THE 2001 PRIMA DESHECHA GENERAL DEVELOPMENT PLAN

I. PROJECT SUMMARY

The County of Orange Integrated Waste Management Department is proposing minor changes to the adopted 2001 Prima Deshecha General Development Plan (GDP). The changes to the GDP primarily relate to project mitigation features associated with obtaining state and federal resource agency permits and authorizations needed for implementation of the approved 2001 GDP, actions associated with landslide remediation, and some additional elements required for the success of the environmental mitigation and restoration components of the 2001 GDP.

This Notice of Preparation (NOP) is being circulated pursuant to California Public Resources Code Section 21153(a) and California Environmental Quality Act (CEQA) Guidelines Section 15082. Public agencies and the public are invited to comment on the proposed scope and content of the environmental information to be included in the Supplemental Environmental Impact Report (SEIR). The format of the NOP is as follows: Section II describes the project location, Section III discusses the background and purpose of the proposed project and previous applicable plans/environmental documentation, Section IV provides a detailed description of the proposed project. Section V presents alternatives to the Proposed Project, Section VI discusses compatibility with adopted plans and Section VI describes responsible agencies for the Proposed Project. This is followed by the Environmental Analysis Checklist, and the Environmental Analysis and Checklist Responses.

II. PROJECT LOCATION

The 1,530-acre Prima Deshecha Landfill (PDL) site is located in Orange County (see Exhibit 1), and includes acreage within the jurisdictions of the cities of San Juan Capistrano (570 acres) and San Clemente (133 acres). The remaining 827 acres are within unincorporated Orange County. The 2001 GDP is the planning document to guide coordinated long-term implementation of both interim and ultimate site development uses. The GDP was approved in November 2001 and, along with its first amendment (approved in October 2002), constitute what is referred to herein as the "2001 GDP."

The PDL site lies in the hills of southeastern Orange County. Ground elevations range from 230 feet above mean sea level (amsl) at the southwest boundary of the site to a maximum elevation of 1,125 feet amsl at the northeast boundary of the site. The Prima Deshecha Cañada watercourse traverses the site from the northeast to the southwest. Two major utility easements, including a 150-foot-wide San Diego Gas and Electric (SDG&E) easement and a 200-foot-wide Southern California Edison (SCE) easement, extend through the central portion of the site and separate the western (Zone 1) and eastern (Zone 4) landfill components of the landfill property (see Exhibit 2).

1

Conservation Plan (NCCP/HCP) led by the U.S. Fish and Wildlife Service, the California Department of Fish and Game, and the County of Orange, and the San Juan/San Mateo Creeks Special Area Management Plan (SAMP) led by the U.S. Army Corps of Engineers and the California Department of Fish and Game. This SEIR will include a discussion of the relationship of the Prima Deshecha GDP and associated Amendment No. 2 to these regional planning efforts.

IV. DESCRIPTION OF THE PROPOSED PROJECT

General. The 2001 GDP divides the 1,530-acre site into five zones for planning purposes, as shown in Exhibit 2 and briefly described below, with proposed design changes, zone boundary adjustments, and landfill grading and height limits based upon subsequent discussions with regional partners and site requirements. As indicated above, the Amendment No. 2/SEIR 597 will address project-level impacts for those actions nearing preliminary or final design, project mitigation features associated with State and Federal resource agency permitting and regulatory authorizations, elements needed to address physical site stability required for continued landfill operations, and elements needed to provide for the long term success of the environmental mitigation and restoration components of the 2001 GDP. Specific new analyses contained herein will focus on those aspects of the 2001 GDP that serve as the basis for Amendment No. 2.

The project objectives approved in the 2001 GDP will continue to guide project planning at the site.

Proposed Project Elements. The proposed project will include the following elements covered by this SEIR (Exhibit 3):

- 1. The limits of potential disturbance which extend beyond the landfill footprint accurately reflect the area of potential landslide remediation needed to address unstable geologic conditions on the property (i.e., Capistrano formation soils) and implement the existing approved 2001 GDP. As evidenced by the landslide which occurred at the site in 1998, slope instability is problematic throughout the PDL site. Landslide remediation measures will be necessary for future development of the site and will be implemented to minimize potential damage to existing and enhanced environmental resources and disruption of on-going landfill operations. Although additional geotechnical field investigations and slope stability analyses will be conducted for future development of the PDL; an updated assessment of potential limits of disturbance has been completed for the site based on the available geotechnical information (Exhibit 3). These updated potential disturbance limits will be analyzed within SEIR 597. There will be no increase in landfill prism, trash capacity, or operational life of the landfill facility as a result of the Proposed Project, nor will ongoing landfill operations be adversely affected. If future geotechnical investigations and analysis result in changes to the development limit shown in SEIR 597, those changes will be addressed in a future CEQA analysis.
- 2. The location and preliminary design of a desilting basin proposed in the GDP between Zones 1 and 4. Design concepts proposed for this basin will include both in-stream and off-stream storage facilities. Exhibit 3 shows the general location proposed for this facility. The desilting basin location is in an area of known landslide deposits and will require landslide stabilization. This stabilization is proposed to occur within the development limits shown on Exhibit 3.
- 3. Long-term maintenance and preservation of the Zone 1 landslide remediation biomitigation sites and existing riparian resources to be permanently preserved

3

- Zone 3: Zone 3 occurs within Segunda Deshecha and the southern portions of the PDL property. This zone is dominated by non-native grassland resources and ruderal species such as artichoke thistle and black mustard. This area of the site also contains small patches of native vegetation, including coastal sage scrub habitat used by the California gnatcatcher, and mixed chaparral. Amendment No. 2 to the 2001 GDP will include the identification of candidate restoration and enhancement opportunities within a portion of Zone 3 to be used as pre-mitigation to offset impacts associated with the ultimate build-out of the landfill.
- Zone 4: This zone comprises a 473-acre area (including non-refuse cut acreage) of which 409 acres is to be used for refuse fill operations (Exhibit 2). The total area of 409 acres is a design change from the 2001 GDP associated with discussions with RMV, which resulted in a reduction in the Zone 4 refuse footprint (from 412 to 409 acres). The RMV agreements also identify RMV's Benefited Property, and place conditions on a No-Build Area within the Benefited Property adjacent to and east of the Prima Deshecha property.
- Zone 5: This zone encompasses the area of disturbance for construction of La Pata Avenue. The boundaries of Zone 5 presented in Exhibit 1 were defined based upon a conceptual alignment design proposed by the County in a 1996 Interim Project Design Report. Accordingly, there are no elements of the Proposed Project that occur within Zone 5.

TIMING OF PROPOSED PROJECT ELEMENTS: The timing for implementation of measures for slope stabilization will be determined by additional analyses and will be presented in the SEIR. The existing desilting basin at the south portion of Zone 1 is scheduled to be enlarged and improved in 2004/2005 to support the future development of Zone 1, and the on-line detention/desilting basin between Zones 1 and 4 will be implemented in approximately 2017 (prior to initiation of refuse filling in Zone 4). Requirements for maintenance of the biomitigation areas will be analyzed and implemented once the results of additional studies are available to assist in specifying the needs of the site. Zone 4 is planned for future refuse disposal activities after Zone 1 is filled to capacity, which is estimated to take place in 2019, and would be in active operation for approximately 48 years.

V. ALTERNATIVES TO THE PROPOSED PROJECT

Alternatives to the Proposed Project that will be considered are listed below in accordance with their feasibility:

Proposed for Detailed Alternative Analysis:

- 1. No Project. This alternative maintains the project actions as approved in the 2001 GDP. The elements covered within the 2001 GDP include solid waste disposal, various regional park and recreational uses, and implementation of a key arterial highway and road extension included in the Master Plan of Arterial Highway (MPAH), Orange County Circulation Plan (OCCP), and Circulation Elements of the cities of San Juan Capistrano and San Clemente. The 2001 GDP incorporates three primary elements including a Landfill Plan covering build-out of Zones 1 and 4, a Circulation Plan, and a Recreation Plan.
- 2. *Revisions to Zone 4 Development*. Several alternatives will be evaluated that involve redesign and relocation of the Zone 4 footprint and associated limits of disturbance

- California Regional Water Quality Control Board, San Diego
- California Integrated Waste Management Board
- California Department of Fish and Game
- Caltrans
- United States Army Corps of Engineers
- United States Fish and Wildlife Service

The approval process for Second Amendment and SEIR will include a hearing at the County of Orange Planning Commission with a recommendation for approval; the project will then proceed to the Board of Supervisors. Once the SEIR for the project is certified by the Board, local and state responsible agencies will be able to rely on the environmental clearance for issuing permits.

Exhibits:

- 1. Regional Location Map
- 2. Local Vicinity Map
- 3. Site Location and Proposed Project Elements Map

		ISSUES & SUPPORTING DATA SOURCES	New Significant Impact	More Severe Impacts	No Substantial Change From Previous Analysis
	b)	Displace substantial numbers of existing housing, necessitating the construction of replacement housing elsewhere?			\boxtimes
	c)	Displace substantial numbers of people, necessitating the construction of replacement housing elsewhere?			\boxtimes
4.	GH pe	COPHYSICAL. Would the project result in or expose ople to impacts involving:			
	a)	Expose people or structures to potential substantial adverse effects, including the risk of loss, injury, or death involving:			\boxtimes
		i) Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault? Refer to Division of Mines and Geology Special Publication 42.			
		ii) Strong seismic ground shaking?			\boxtimes
		iii) Seismic-related ground failure, including liquefaction?			\boxtimes
		iv) Landslides?			\boxtimes
	b)	Result in substantial soil erosion or the loss of topsoil?			\boxtimes
	c)	Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse?			
	d)	Be located on expansive soils, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property?			
	e)	Have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal system where sewers are not available for the disposal of wastewater?			\boxtimes

		ISSUES & SUPPORTING DATA SOURCES	New Significant Impact	More Severe Impacts	No Substantial Change From Previous Analysis
6	. T	RANSPORTATION/CIRCULATION. Would the project	•		Allalybis
	re	sult in:			
	a)	Increase in traffic, which is substantial in relation to the existing traffic load and capacity of the street system (i.e., result in a substantial increase in either the number of vehicle trips, the volume to capacity ratio on roads, or congestion at intersections)?			
	b)	Exceed, either individually or cumulatively, a level of service standard established by the county congestion management agency for designated roads or highways?			
	c)	Result in a change in air traffic patterns, including either an increase in traffic levels or a change in location that result in substantial safety risks?			
	ď)	Substantially increase hazards due to a design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment)?			\boxtimes
	e)	Result in inadequate emergency access?			\boxtimes
	f)	Result in inadequate parking capacity?			\boxtimes
	g)	Conflict with adopted policies, plan or programs supporting alternative transportation (e.g. bus turnouts, bicycle racks)?			
7.	AD	R QUALITY. Would the project:			
	a)	Conflict with or obstruct implementation of the applicable air quality plan?			\boxtimes
	b)	Violate any air quality standard or contribute substantially to an existing or projected air quality violation?			\boxtimes
	c)	Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non- attainment under an applicable federal or state ambient air quality standard (including releasing emissions, which exceed quantitative thresholds for ozone precursors)?			\boxtimes

۰,

P

		ISSUES & SUPPORTING DATA SOURCES	New Significant Impact	More Severe Impacts	No Substantial Change From Previous Analysis
	b) Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the California Department of Fish and Game or U.S. Fish and Wildlife Services?			
	c)	Have a substantial adverse effect on Federally protected wetlands as defined by Section 404 of the Clean Water Act (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means?			
	d)	Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites?			
	c)	Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance?			
	f)	Conflict with provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan?			\boxtimes
10. AESTHETICS. Would the project:					
	a)	Have a substantial adverse effect a scenic vista?			\boxtimes
	b)	Substantially damage scenic resources, including, but not limited to, trees, rock outcroppings, and historic buildings within a state scenic highway?			\boxtimes
	c)	Substantially degrade the existing visual character or quality of the site and its surroundings?			\boxtimes
	đ)	Create a new source of substantial light or glare, which would adversely affect day or nighttime views in the area?			\boxtimes
11.	CU pro	LTURAL/SCIENTIFIC RESOURCES, Would the ject:			
	a)	Cause a substantial adverse change in the significance of a historical resource as defined in Section 15064.5?			\boxtimes

	ISSUES & SUPPORTING DATA SOURCES	New Significant Impact	More Severe Impacts	No Substantial Change From Previous Analysis	
d)	Be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would it create a significant hazard to the public or the environment?				
e)	For a project located within an airport land use plan or, where such plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard for people residing or working in the project area?				
ſ)	For a project within the vicinity of private airstrip, would the project result in a safety hazard for people residing or working in the project area?				
g)	Impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan?				
h)	Expose people or structures to a significant risk or loss, injury or death involving wildland fires, including where wildlands are adjacent to urbanized areas or where residences are intermixed with wildlands?				
i)	Include a new or retrofitted storm water treatment control Best Management Practice (BMP), (e.g. water quality treatment basin, constructed treatment wetlands), the operation of which could result in significant environmental effects (e.g. increased vectors and odors)?				
15. PUBLIC SERVICES. Would the project:					
a)	Result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for any of the public services:				
	i) Fire protection?			\boxtimes	
	ii) Police protection?			\boxtimes	

	ISSUES & SUPPORTING DATA SOURCES	New Significant Impact	More Severe Impacts	No Substantial Change From Previous Analysis
Ъ)	Does the project have possible environmental effects, which are individually limited but cumulatively considerable? ("cumulatively considerable" means that the incremental effects of an individual project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects.)			
c)	Does project have environmental effects which will cause substantial adverse effects on human beings, either directly or			\boxtimes

DETERMINATION:

indirectly

Based upon the evidence in light of the whole record documented in the attached environmental checklist explanation, cited incorporations and attachments, I find that the proposed project:

Has been previously analyzed as part of an earlier CEQA document (which either mitigated the project or adopted impacts pursuant to findings) adopted/certified pursuant to State and County CEQA Guidelines. The proposed project is a component of the whole action analyzed in the previously adopted/certified CEQA document.

 \boxtimes

Has been previously analyzed as part of an earlier CEQA document (which either mitigated the project or adopted impacts pursuant to findings) adopted/certified pursuant to State and County CEQA Guidelines. Minor additions and/or clarifications are needed to make the previous documentation adequate to cover the project, which are documented in this addendum to the earlier CEQA document (CEQA §15164).

Has been previously analyzed as part of an earlier CEQA document (which either mitigated the project or adopted impacts pursuant to findings) adopted/certified pursuant to State and County CEQA Guidelines. However, there is important new information and/or substantial changes have occurred requiring the preparation of an additional CEQA document (ND or EIR) pursuant to CEQA Guidelines Sections 15162 through 15163.

Signature: Lin hunore

Planner: Robert Richmond Environmental Services Fax: (714) 834-4002

Revised 2-13-03

NOTE: All referenced and/or incorporated documents may be reviewed by appointment only, at the County of Orange Integrated Waste Management Department, 320 N. Flower Street, Santa Ana, California, unless otherwise specified. An appointment can be made by contacting the Public Information Officer at (714) 834-4176. primary use is being contemplated for the proposed project. Project implementation will not result in the displacement of existing market-rate or affordable housing. Therefore, no significant impacts related to housing availability are anticipated.

4. GEOPHYSICAL

- a) Expose people or structures to potential substantial adverse effects, including the risk of loss, Injury, or death involving:
 - Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault? Refer to Division of Mines and Geology Special Publication 42.
 - ii) Strong seismic ground shaking?
 - iii) Seismic-related ground failure, including liquefaction?
 - iv) Landslides?
- b) Result in substantial soil erosion or the loss of topsoil?
- d) Be located on expansive soils, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property?
- e) Have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal system where sewers are not available for the disposal of wastewater?

No Substantial Change from Previous Analysis. The Proposed Project at Prima Deshecha Landfill remains consistent in concept with activities approved in the 2001 GDP EIR and, therefore, will have no substantial change from the previous analysis. Prior mitigation commitments to ensure the stability of all cut, fill, and lined slopes will maintain the effects of grading operations below significance, and all landfill design plans will comply with state and federal seismic stability requirements. Erosion control measures implemented to comply with landfill regulations include vegetation of cut slopes, silt control, and soil stabilizers to minimize impacts from soil erosion. No significant change is anticipated from previous analyses on land subsidence, expansive soils, erosion, or loss of topsoil.

c) Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse?

No Substantial Change from Previous Analysis. The 2001 GDP and EIR 575 Identified the potential for activating existing landslides during future landfill development if development areas are not stabilized. Future landslide remediation design and construction in the expanded disturbance footprint, similar to that proposed in the 2001 GDP, will serve to stabilize existing soil conditions.

5. HYDROLOGY AND WATER QUALITY

- a) Violate any water quality standards or waste discharge requirements?
- f) Have a significant adverse impact on groundwater quality or otherwise substantially degrade water quality?

No Substantial Change from Previous Analysis. The Proposed Project will not violate water quality standards or discharge requirements or otherwise substantially degrade water quality. In addition, PDL continues to monitor water quality on-site in accordance with the San Diego Regional Water Quality Control Board (RWQCB) waste discharge requirements. PDL also operates in compliance with the National Pollutant Discharge Elimination System (NPDES) program under the Clean Water Act.

- i) Expose people or structures to a significant risk of loss, injury or death involving flooding, including flooding as a result of the failure of a levee or dam?
- j) Inundation by seiche, tsunami, or mudflow?

No Substantial Change from Previous Analysis. As noted above, landfill operations are an existing condition and it is not anticipated that the Proposed Project will result in more severe impacts from runoff than indicated in the previous analysis. On-site detention, diversion and drainage facilities are designed and constructed to accommodate the volume of peak flows in compliance with the California Code of Regulations for landfill operations. No portion of the landfill operations is located within the 100-year floodplain, and there is no potential for flooding as a result of the failure of a levee or dam, or inundation by seiche or tsunami. Since the Proposed Project includes landslide remediation, the potential impacts of mudflow, as analyzed in EIR 575, are substantially reduced.

6. TRANSPORTATION/CIRCULATION

- a) Increase in traffic, which is substantial in relation to the existing traffic load and capacity of the street system (i.e., result in a substantial increase in either the number of vehicle trips, the volume to capacity ratio on roads, or congestion at intersections)?
- b) Exceed, either individually or cumulatively, a level of service standard established by the county congestion management agency for designated roads or highways?
- c) Result in a change in air traffic patterns, including either an increase in traffic levels or a change in location that result in substantial safety risks?
- d) Substantially increase hazards due to a design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment)?
- e) Result in inadequate emergency access?
- f) Result in inadequate parking capacity?
- g) Conflict with adopted policies, plan or programs supporting alternative transportation (e.g. bus turnouts, bicycle racks)?

No Substantial Change from Previous Analysis. Project implementation will not significantly affect roadway operations and/or existing levels of service as the landfill is currently in operation and supplemental project features as proposed herein will not alter traffic volumes or patterns onto or off the site, or in the vicinity.

The Proposed Project is being coordinated with local entities concerned with regional transportation systems. The 2001 GDP was developed with a view toward incorporation of an extension of La Pata Avenue through the site, and allowance has also been made for the extension of Camino de los Mares through the southwestern corner of the site. The Proposed Project maintains the approach of the 2001 GDP regarding these elements. The IWMD is committed to continued close coordination with the cities of San Clemente and San Juan Capistrano on regional transportation needs. In addition, transportation features needed to support landfill operations have been provided for on-site, and there will be no additional impact to transportation corridors or Level of Service impacts off the PDL property due to the supplemental project features proposed herein.

The Proposed Project does not conflict with adopted policies supporting alternative modes of transportation or have the potential to impact parking capacity. The landfill is served by vehicles hauling municipal solid waste and sewage sludge (i.e., biosolids), including passenger cars, single unit trucks, and combination vehicles. No change in this condition is proposed and no impacts are anticipated. The Proposed Project will have no effect on air traffic patterns.

be developed that includes salvage of seed for use within a designated restoration area within the site as determined by a restoration ecologist in consultation with the CDFG.

The proposed action may also involve impacts to at least four previously detected LBV territories requiring consultation with and subsequent authorizations from the USFWS and CDFG prior to removal of any habitat supporting this species including the development of a mitigation program that may occur both on- and off-site. Sensitive biological resources on-site that could be impacted by noise generated from short-term construction activities, such as the Least Bell's Vireo (LBV) and the California gnatcatcher (gnatcatcher), will be accommodated through the incorporation of mitigation measures such as installation of berms or phasing and/or management of construction activities, as needed, to reduce the impacts to less than significant.

The 2003 gnatcatcher surveys indicate that this species was detected in the Zone 3 and in an undeveloped area northwest of Zone 1. LBV surveys indicate that four pairs and one solitary male were detected in 2003, within Prima Deshecha Cañada Channel up and downstream of the newly realigned Prima Deshecha Cañada Channel. The proposed action may impact one known location of California gnatcatcher between Zones 1 and 4. However, a mitigation program involving the creation of coastal sage scrub habitat will be developed in consultation with the USFWS and CDFG and implemented prior to the initiation of landfilling operations within Zone 4.

- b) Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the California Department of Fish and Game or U.S. Fish and Wildlife Services?
- d) Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites?
- e) Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance?
- f) Conflict with provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan?

The biological resources database for the entire Prima Deshecha Landfill was updated in 2002-2003 in anticipation of IWMD's participation in the Southern NCCP program and to provide the necessary biological and jurisdictional resource data to obtain state and federal regulatory permits and agreement authorizations. This effort included revisions to the vegetation database to reflect proposed limits of disturbance, and verification of the extent of coastal sage scrub, chaparral, riparian and native grassland resources; special status plant species surveys; and California gnatcatcher and LBV surveys. These efforts also included the delineation of state and federal jurisdiction wetlands and waters of the U.S. and the identification of potential habitat restoration. Based on these efforts, the site can be characterized as follows:

Prima is a 1,530-acre facility with 408 acres of disturbed area primarily in Zone 1, the site access roads, weigh station, and administration buildings; 804.4 acres of non-native annual grassland dominated by ruderal species, such as artichoke thistle and black mustard covering most of the site; 191.0 acres of coastal sage scrub distributed in moderate to small patches in the southern and eastern portions of the site in Zones 3 and 4; 63.1 acres of chaparral in the northern portion of Zone 4 and in Segunda Deshecha; 20.9 acres of riparian primarily within Prima Deshecha Cañada drainage and the main side drainage between Zones 1 and 4; 14.3 acres of southern needlegrass grassland near the eastern boundaries of Zone 4; and 27.6 acres of ruderal vegetation in small patches throughout the site. The proposed action to stabilize existing landslide conditions and prepare areas of the site for landfilling operations and the installation of associated facilities will involve the removal

residents of the City of San Clemente. IWMD and the City of San Clemente have agreed on the elements of a Viewshed Protection Plan (VPP), to be prepared by IWMD prior to development of Zone 4. The Zone 4 VPP will provide assurance that landfill operations are maintained at an elevation no greater than the natural ground surface elevation of the surrounding topography, keeping the visibility of landfill operations to a minimum from adjacent housing developments. In addition, the VPP will provide for a landscaping plan and contour grading plan elements that are to be accepted by the local municipalities.

The Proposed Project will maintain these commitments in the design and evaluation of all project features. Accordingly, no impact is anticipated to aesthetics from the Proposed Project.

11. CULTURAL/SCIENTIFIC RESOURCES

- a) Cause a substantial adverse change in the significance of a historical resource as defined in Section 15064.5?
- b) Cause a substantial adverse changed in the significance of an archaeological resource pursuant to Section 15064.5?
- c) Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?

No Substantial Change from Previous Analysis. The proposed project elements are located in areas previously studied for the 2001 GDP and EIR 575. Although enlargement of the disturbance footprint due to remediation activities is expected to result in significant earth movement, cultural/scientific resources consultants under contract with the County will continue to monitor operations at the landfill to ensure that identified resources are not adversely impacted.

12. RECREATION

- a) Would the project increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated?
- b) Does the project include recreational facilities or require the construction or expansion of recreational facilities, which might have an adverse physical effect on the environment?

No Substantial Change from Previous Analysis. An expressed purpose of the 2001 GDP is to create both interim and long-term recreational opportunities on the project site. These opportunities, as listed in the 2001 GDP, will not be subject to alteration or effect from the Proposed Project. In addition, implementation of the Proposed Project will not result in an increased demand for local or regional parks or other recreational facilities. Several recreational facilities in the form of trails (hiking, riding, biking) are located in the vicinity and around the perimeter of the project site. Access to these trails will be maintained through the period of project implementation, with temporary detours as needed based upon localized construction of landslide remediation measures. No significant impacts are anticipated as a result of project implementation.

13. MINERAL RESOURCES

- a) Result in the loss of availability of a known mineral resource that would be of value to the region and the residents of the state?
- b) Result in the loss of availability of a locally important mineral resource recovery site delineated on a local general plan, specific plan, or other land use plan?

15. PUBLIC SERVICES

a) Result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for any of the public services:

No Substantial Change from Previous Analysis. The landfill use is an existing condition and implementation of the Proposed Project would not result in the need for increased fire or police services. Project elements being proposed will not create any impact on schools in the area as no new students will be generated by either the implementation of landslide stabilization measures or other supplemental on-site project elements.

The project itself will not create the need for new roads. The extension of La Pata Avenue accommodated by the GDP is a regional transportation element and is not part of the Proposed Project.

The project is not expected to result in any impacts on public facilities such as libraries, city halls, post offices, or hospitals. No regional facilities such as other regional parks are expected to be impacted.

16. UTILITIES AND SERVICE SYSTEMS

- a) Exceed wastewater treatment requirements of the applicable Regional Water Quality Control Board?
- c) Require or result in the construction of new storm water drainage facilities or expansion of existing facilities, the construction of which would cause significant environmental effects?
- e) Result in a determination by the wastewater treatment provider, which serves or may serve the project that it has adequate capacity to serve the project's projected demand in addition to the provider's existing commitments?
- f) Be served by a landfill with sufficient permitted capacity to accommodate the project's solid waste disposal needs?
- g) Comply with federal, state and local statures and regulations related to solid waste?

No Substantial Change from Previous Analysis. As the Proposed Project remains consistent with existing landfill activities, wastewater treatment requirements and determinations, storm drain facilities and landfill service will not be affected beyond that which has already been identified in prior documents. All landfill activities comply with federal, state and local statutes and regulations related to solid waste.

- b) Require or result in the construction of new water or wastewater treatment facilities or expansion of existing facilities, the construction of which could cause significant environmental impacts?
- d) Have sufficient water supplies available to serve the project from existing entitlements and resources, or are new or expanded entitlements needed?

More Severe Impacts. The Proposed Project may result in an increased need for potable or reclaimed water on the site for long-term biomitigation maintenance. Currently, it has not been determined whether additional water sources are required for this purpose, or how much water will be necessary to maintain the biomitigation areas. A thorough evaluation of the alternatives will be conducted with a view toward the minimization of impacts to utilities.

GNEnv Svcs\GCR_RCIPrima 2nd Amendment\02-06-04 NOP Analysis FINAL.DOC 10



ORANGE COUNTY FIRE AUTHORITY P. O. Box 86, Orange, CA 92856-9086 • 145 South Water St., Orange, CA 92866-2123

Chip Prather, Fire Chief

(714) 744-0400

www.ocfa.org

February 18, 2004

Integrated Waste Management Robert Richmond 320 N Flower St #400 Santa Ana, CA 92703

Ref: Second Ammendment to 2001 Prima Deshecha GDP

Dear Mr. Richmond:

The Orange County Fire Authority (OCFA) serves the project site, including the city and unincorporated county areas. We have no comment on the amendment. Please ensure that the OCFA Hazardous Materials Services Section is provided updated information for the Emergency Business Plan. While no additional public safety resources are needed as a result of this project, all standard conditions and guidelines will be applied to the project during the normal review process.

Please contact me if you have any additional needs: michelehernandez@ocfa.org or 714-744-0420.

Michele Hernandez Management Analyst Strategic Services Section

Serving the Cities of: Aliso Viejo + Buena Park + Cypress + Dana Point + Irvine + Laguna Hills + Laguna Niguel + Laguna Woods + Lake Forest + La Palma Las Alamitos - Mission Viejo - Placentia - Rancho Santa Margarita - San Clemente - San Juan Capistrano - Scal Beach - Stanton - Tustin - Villa Park Westminster + Yorba Linda + and Unincorporated Arcus of Orange County

South Coast Air Quality Management District 21805 E. Copley Drive, Diamond Bar, CA 91765-4182 (909) 396-2000 · www.aqmd.gov

February 19, 2004

Mr. Robert Richmond Integrated Waste Management Dept. 320 N. Flower Street, Suite 400 Santa Ana, CA 92703

Dear Mr. Richmond:

Notice of Preparation of a Draft Supplemental Environmental Impact Report for Second Amendment to 2001 Prima Deshecha General <u>Development Plan</u>

The South Coast Air Quality Management District (SCAQMD) appreciates the opportunity to comment on the above-mentioned document. The SCAQMD's comments are recommendations regarding the analysis of potential air quality impacts from the proposed project that should be included in the Draft Environmental Impact Report (EIR).

Air Quality Analysis

The SCAQMD adopted its California Environmental Quality Act (CEQA) Air Quality Handbook in 1993 to assist other public agencies with the preparation of air quality analyses. The SCAQMD recommends that the Lead Agency use this Handbook as guidance when preparing its air quality analysis. Copies of the Handbook are available from the SCAQMD's Subscription Services Department by calling (909) 396-3720. Alternatively, lead agency may wish to consider using the California Air Resources Board (CARB) approved URBEMIS 2002 Model. This model is available on the CARB Website at: www.arb.ca.gov.

The Lead Agency should identify any potential adverse air quality impacts that could occur from all phases of the project and all air pollutant sources related to the project. Air quality impacts from both construction and operations should be calculated. Construction-related air quality impacts typically include, but are not limited to, emissions from the use of heavy-duty equipment from grading, earth-loading/unloading, paving, architectural coatings, off-road mobile sources (e.g., heavy-duty construction equipment) and on-road mobile sources (e.g., construction worker vehicle trips, material transport trips). Operation-related air quality impacts may include, but are not limited to, emissions from stationary sources (e.g., boilers), area sources (e.g., solvents and coatings), and vehicular trips (e.g., on- and off-road tailpipe emissions and entrained dust). Air quality impacts from indirect sources, that is, sources that generate or attract vehicular trips should be included in the analysis. An analysis of all toxic air contaminant impacts due to the -2-

February 19, 2003

decommissioning or use of equipment potentially generating such air pollutants should also be included.

Mitigation Measures

In the event that the project generates significant adverse air quality impacts, CEQA requires that all feasible mitigation measures be utilized during project construction and operation to minimize or eliminate significant adverse air quality impacts. To assist the Lead Agency with identifying possible mitigation measures for the project, please refer to Chapter 11 of the SCAQMD CEQA Air Quality Handbook for sample air quality mitigation measures. Additionally, SCAQMD's Rule 403 – Fugitive Dust, and the Implementation Handbook contain numerous measures for controlling construction-related emissions that should be considered for use as CEQA mitigation if not otherwise required. Pursuant to state CEQA Guidelines §15126.4 (a)(1)(D), any impacts resulting from mitigation measures must also be discussed.

Data Sources

SCAQMD rules and relevant air quality reports and data are available by calling the SCAQMD's Public Information Center at (909) 396-2039. Much of the information available through the Public Information Center is also available via the SCAQMD's World Wide Web Homepage (http://www.aqmd.gov).

The SCAQMD is willing to work with the Lead Agency to ensure that project-related emissions are accurately identified, categorized, and evaluated. Please call Charles Blankson, Ph.D., Air Quality Specialist, CEQA Section, at (909) 396-3304 if you have any questions regarding this letter.

Sincerely,

Steve Smith

Steve Smith, Ph.D. Program Supervisor, CEQA Section Planning, Rule Development and Area Sources

SS:CB:li

ORC040211-02LI Control Number

ccellence

March 4, 2004

Robert Richmond Environmental Services County of Orange IWMD 320 North Flower Street, Suite 400 Santa Ana, CA 92703

Subject: Notice of Preparation of Draft Supplemental Environmental Impact Report # 597 Prima Deshecha Landfill (SWIS No. 30-AB 0019), San Juan Capistrano

COUNTY OF ORANGE

HEALTH CARE AGENCY

REGULATORY HEALTH SERVICES ENVIRONMENTAL HEALTH

Dear Mr. Richmond:

On February 10, 2004, the Solid Waste Local Enforcement Agency (LEA) received a Notice of Preparation (NOP) for Draft SEIR 597. This CEQA document will address project elements needed for landslide stabilization and maintenance of environmental mitigation and restoration areas at Prima Deshecha Landfill. Orange County Integrated Waste Management Department (OCIWMD) prepared this NOP.

- 1. Section IV DESCRIPTION OF THE PROPOSED PROJECT, Proposed Project Elements: "The limits of potential disturbance which extend beyond the landfill footprint accurately reflect the area of potential landslide....". For clarity purposes, the LEA suggests replacing "landfill footprint" with "waste footprint" in the draft SEIR since the landfill encompasses waste fill areas as well as native buffer zones.
- Section IV DESCRIPTION OF THE PROPOSED PROJECT, Zone Descriptions: Not clear if any elements of the proposed project to be analyzed by SEIR 507 are within Zone 4. Please clarify.

If you have any questions, please contact me at (714) 667-2026.

Sincerely,

Ossama "Sam" Abu-Shaban, PE, DEE Senior Civil Engineer Solid Waste Local Enforcement Agency Environmental Health

JULIETTE A. POULSON, RN, MN DIRECTOR

MIKE SPURGEON DEPUTY AGENCY DIRECTOR REGULATORY HEALTH SERVICES

STEVEN K. WONG, REHS, MPH DIRECTOR ENVIRONMENTAL HEALTH

> MAILING ADDRESS: 2009 EAST EDINGER AVENUE SANTA ANA, CA 92705-4720

I ELEPHONE: (714) 007-3000 FAX: (714) 972-0749 E-MAIL: environhealth@hca.co.orange.ca.us San Joaquin Hills Corridor Agency

Chairwoman: Linda Lindholm Laguna Niguel

March 8, 2004



Foothill/Eastern Corridor Agency

Chairman: Peter Herzog Lake Forest

Integrated Waste Management Department 320 N. Flower Street, Suite 400 Santa Ana, CA 92703

Subject: Notice of Intent to Prepare Draft Supplemental Environmental Impact Report #597

To Whom It May Concern:

Thank you for forwarding the Notice of Intent for the Draft Supplemental Environmental Impact Report No. 597 for the Prima Deshecha Landfill. The Transportation Corridor Agencies (TCA) has reviewed the above-subject notice and would like to offer the following comments for your review:

- The TCA is in the process of evaluating several alternatives for the extension of the Foothill Transportation Corridor – South (FTC-S). Some of the alternatives under consideration have the potential to impact landfill operations. The draft environmental document for the FTC-S Project will be released soon and available for review. The TCA requests that the County continue to coordinate with us on this project.
- The TCA would like to review any future environmental documents and/or technical appendices related to this Notice of Intent.

Should you have any questions regarding this letter, please contact me at (949) 754-3483.

Sincerely,

Macie Cleary-Milan

Macie Cleary-Milan Deputy Director Environmental Planning

cc: Valarie McFall

Walter D. Kreutzen, Chief Executive Officer

125 PACIFICA, SUITE 100, IRVINE CA 92618-3304 • P.O. BOX 53770, IRVINE CA 92619-3770 • 949/754-3400 FAX 949/754-3467 www.thetoliroads.com

Members: Aliso Viejo • Anaheim • Costa Mesa • County of Orange • Dana Point • Irvine • Laguna Hills • Laguna Niguel • Laguna Woods • Lake Forest Mission Viejo • Newport Beach • Orange • Rancho Santa Margarita • Santa Ana • San Clemente • San Juan Capistrano • Tustin • Yorba Linda



California Integrated Waste Management Board

Linda Moulton-Patterson, Chair 1001 I Street • Sacramento, California 95814 • (916) 341-6000 Mailing Address: P. O. Box 4025, Sacramento, CA 95812-4025 www.ciwmb.ca.gov



Arnold Schwarzenegger Governor

Terry Tamminen Secretary for Environmental Protection

March 9, 2004

Mr. Robert Richmond Integrated Waste Management Department Environmental Services 320 North Flower Street Santa Ana, CA 92703-5000



Subject: SCH No. 1999041035: Notice of Preparation for a Draft Supplemental Environmental Impact Report for the Second Amendment to the 2001 Prima Deshecha General Development Plan for Prima Deshecha Landfill - Solid Waste Facilities Permit No. 30-AB-0019, San Juan Capistrano, Orange County

Dear Mr. Richmond:

Thank you for allowing the California Integrated Waste Management Board's (Board) staff to provide comments for this proposed project and for your agency's consideration of these comments as part of the California Environmental Quality Act (CEQA) process.

Board staff has reviewed the environmental document cited above and offer the following project description, analysis, and our recommendations for the proposed project based on Board staff's understanding of the project. If Board staff's project description varies substantially from the project as understood by the Lead Agency, Board staff requests notification of any significant differences prior to preparation of the Draft Supplemental Environmental Impact Report.

PROJECT DESCRIPTION

The County of Orange Integrated Waste Management Department, acting as Lead Agency, has prepared and circulated a Notice of Preparation of a Draft Supplemental Environmental Impact Report in order to comply with the CEQA and to provide information to, and solicit consultation with, Responsible Agencies in the approval of proposed changes in design and operations at the Prima Deshecha Landfill. The proposed project may require revision of SWFP No.30-AB-0019, as well as other federal, state and local approvals.

California Environmental Protection Agency

2 Printed on Recycled Paper

The energy challenge facing California is real. Every Californian needs to take immediate action to reduce energy consumption. For a list of simple ways you can reduce demand and cut your energy costs, see our Web site at http://www.ciwmb.ca.gov/ NOP DSEIR Prima Deshecha Landfill

RECEIVED

MAR 15 2004

March 9, 2004

The Prima Deshecha Landfill is located in Orange County and in the deages within the jurisdictions of the cities of San Juan Capistrano and San Clemente. The landfill has been operating since the mid-1970s.

The Proposed Project will address project elements needed for landslide stabilization of the site, project features required for maintenance of environmental mitigation measures and restoration areas, both of which constitute incremental additions or changes to the approved project as defined in the 2001 General Development Plan. The Proposed Project description for Amendment No. 2 will not increase the landfill prism, trash capacity or the operating life of the facility and will not adversely affect ongoing landfill operation.

BOARD COMMENTS AND QUESTIONS

Since the Board may be one of the Responsible Agencies involved in the discretionary approval process for the SWFP revision, Board staff will perform an environmental analysis for this project, using the environmental document developed by the Lead Agency as required in CEQA Guidelines, Title 14, California Code of Regulations (14 CCR), Section 15096.

To help the Lead Agency prepare a CEQA document that will be as complete as possible, and to identify environmental impacts for consideration in the Draft Supplemental Environmental Impact Report for the purposes of the Board's environmental evaluation, Board staff offer the following comments and questions:

Supplemental Environmental Impact Report

If the proposed project remains as it is described in this Notice of Preparation as circulated through the Office of Planning and Research, State Clearinghouse, Board staff has no substantive comments at this time.

If the proposed project changes so as to change the operation of the landfill by changing such things as the landfill prism, trash capacity, the operating life of the facility, disposal footprint, peak tonnage, peak vehicle count, landfill operation, elevations, etc. Board staff would at that point have comments on the Draft Supplemental Environmental Impact Report.

BOARD CEQA REVIEW

As a Responsible Agency under CEQA, Board staff's comments on environmental documents are intended to assist the Lead Agency in developing an environmental document that will be as complete and adequate as possible for use by the Lead Agency and all Responsible Agencies.

Board staff's comments are intended to help decision-makers 1) identify potential impacts from proposed projects; 2) determine whether any such impacts are significant; and 3) ascertain whether significant impacts can be mitigated to a level of insignificance in compliance with the CEQA statutes and guidelines.

NOP DSEIR Prima Deshecha Landfill

When performing the initial review of a CEQA document such as a Draft Environmental Impact Report or Negative Declaration during the circulation process, the first analysis Board staff must make is to evaluate whether or not the proposed CEQA document clearly describes all phases of the project and assesses all potential primary and secondary impacts to the environment and/or public health and safety that could occur if the proposed project is implemented.

When evaluating the adequacy of an environmental document for purposes of SWFP concurrence, Board staff must compare the design and operation of the facility as described in the proposed SWFP with the project as described and evaluated in the environmental document cited for CEQA compliance in the proposed SWFP.

In order for Board staff to evaluate and recommend whether or not the environmental document is adequate for use in the Board's permitting process, the proposed project must be described in sufficient detail for the Environmental Review Section staff to understand and evaluate the proposed project, the potential environmental impacts, proposed mitigation measures, and findings as presented by the Lead Agency.

When the proposed SWFP is received by the Board along with the citation of evidence of CEQA compliance by the LEA, the second analysis performed by the Board staff is to evaluate whether or not the CEQA evaluation in the cited environmental document supports the requested specifications, revisions, and/or conditions of the proposed SWFP. For instance, does the environmental document clearly describe and assess the potential air quality, water quality, geological impacts, traffic, noise, dust, vector and other health and safety impacts that can be associated with the proposed solid waste facility or changes in design and/or operation? When this type of information is included and addressed in the environmental document, the SWFP concurrence process is greatly facilitated.

After comparison of the cited CEQA document with the proposed SWFP, Board staff makes a recommendation to the Board regarding the adequacy of the CEQA document for the Board's SWFP concurrence purposes. The Board members make the final determination of the adequacy of the CEQA document for SWFP concurrence as well as whether or not to concur in issuance of the SWFP.

Summary

Board staff thanks the Lead Agency for the opportunity to review and comment on the Notice of Preparation and hopes that this comment letter will be useful to the Lead Agency in carrying out their responsibilities in the CEQA process.

Board staff requests copies of any subsequent environmental documents including, the Draft Supplemental Environmental Impact Report, the Final Environmental Impact Report, the RFI, any Statements of Overriding Considerations, copies of public notices and any Notices of Determination for this project. Refer to 14 CCR, Section 15094(c) that states: "<u>If the project</u> requires a discretionary approval from a state agency, the notice of determination shall also file with OPR [Office of Planning and Research, State Clearinghouse]." NOP DSEIR Prima Deshecha Landfill

Board staff requests that the Lead Agency provide a copy of its responses to comments at least ten days prior to certifying the Final Environmental Impact Report. Refer to PRC Section 21092.5(a).

If the document is certified during a public hearing, the Board staff request ten days advance notice of this hearing. If the document is certified without a public hearing, Board staff requests ten days advance notification of the date of the certification and project approval by the decisionmaking body.

If you have any questions regarding these comments, please contact me at 916.341.6728 or email at <u>rseamans@ciwmb.ca.gov</u>. Additional information regarding Environmental Impact Reports for landfills can be found on the Board's website at http://www.ciwmb.ca.gov/LEACentral/CEQA/disposal.htm.

Sincerely,

Raymond M. Seamans Permitting and Inspection Branch, Region 4 Environmental Review Permitting and Enforcement Division California Integrated Waste Management Board

cc: Tadese Gebre-Hawariat Permitting and Inspection Branch, Region 4 Permitting and Enforcement Division California Integrated Waste Management Board

> Suzanne Hambleton, Supervisor Permitting and Inspection Branch, Region 4 Permitting and Enforcement Division California Integrated Waste Management Board

Patty Henshaw Ossama Abu-Shaban County of Orange Health Care Agency Environmental Health Division 2009 East Edinger Avenue Santa Ana, CA 92705

State of California - The Resources Agency



DEPARTMENT OF FISH AND GAME http://www.dfg.ca.gov 4949 Viewridge Avenue San Diego, CA 92123 (858) 467-4201



March 12, 2004

Robert Richmond County of Orange Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, California 92703

Comments on the Notice of Preparation of a Draft Supplemental Environmental Impact Report for the Second Amendment to the 2001 Prima Deshecha General Development Plan Project in the County of Orange (SCH# 1999041035)

Dear Mr. Richmond:

The Department of Fish and Game (Department) appreciates this opportunity to comment on the above-referenced project, relative to impacts to biological resources. The Department is a Trustee Agency and a Responsible Agency pursuant to the California Environmental Quality Act (CEQA), Sections 15386 and 15381 respectively. As a Trustee Agency, the Department must be consulted by the Lead Agency during the preparation and public review for project-specific CEQA documents. As a Trustee Agency, the Department reviews proposed projects, comments on their impacts, and determines whether the mitigation measures or alternatives proposed are feasible and appropriate. Pursuant to Section 1802 of the Fish and Game Code, the Department has jurisdiction over the conservation, protection, and management of fish, wildlife, native plants and habitat necessary for biologically sustainable populations of those species. The Department also administers the Natural Community Conservation Planning Program (NCCP).

Project Description

The project proposes a second amendment to the adopted 2001 Prima Deshecha General Development Plan (GDP) to accommodate mitigation features associated with landslide and subsequent remediation activities that were included in the first amendment. The active 1,530-acre Prima Deshecha Landfill is located in southern Orange County within the planning boundaries of Orange County's Southern Subregion NCCP. The Draft Supplemental Environmental Impact Report (DSEIR) will provide project-level analysis for actions described and adopted at the programmatic level by the 2001 GDP. According to the Notice of Preparation (NOP), the proposed project will not increase the volume or operating life of the landfill or affect ongoing activities.

Project elements proposed include updated geotechnical investigations and analyses; location and preliminary design of a desilting basin; hydrologic design and analysis to ensure existing riparian mitigation sites associated with Zone 1 landslide remediation work can be Mr. Robert Richmond March 12, 2004 Page 2

maintained; and development and implementation of a pre-mitigation plan to address impacts associated with the eventual build-out of Zone 4.

Specific Comments

Portions of the landfill will eventually be restored to native habitats and will provide connectivity for wildlife as part of the Southern Subregion NCCP Reserve system. The project should endeavor to minimize impacts to existing biological resources as well as include measures to ensure that implementation of the proposed work will not preclude connectivity for wildlife after landfill closure and during the operational period.

Although construction of La Pata Avenue is not included as part of the project, the DSEIR should include analysis of alternative alignments of the road and potential effects on wildlife movement and design of preserved portions of the landfill. Specifically, the project should ensure that alternative locations of La Pata Avenue are not precluded and that subsequent construction and operation of the road does not adversely affect the preparation and implementation of Orange County's Southern Subregion NCCP.

We appreciate and encourage the County's efforts to implement restoration plans prior to project impacts. These efforts will minimize temporal loss of habitat and possibly avoid them altogether. We look forward to working with County's staff and consultants to develop a restoration plan.

General Comments

To enable the Department to adequately review and comment on the proposed project from the standpoint of the protection of plants, fish and wildlife, we recommend the following information be included in the DEIR:

- 1. A complete discussion of the purpose and need for, and description of, the proposed project, including all staging areas and access routes to the construction and staging areas.
- 2. A complete list and assessment of the flora and fauna within and next to the project area, with particular emphasis upon identifying State or federally listed rare, threatened, endangered, or proposed candidate species, California Species of Special Concern and/or State Protected or Fully Protected species, and any locally unique species and sensitive habitats. Specifically, the DEIR should include:
 - a. A thorough assessment of Rare Natural Communities on site and within the area of impact, following the Department's Guidelines for Assessing Impacts to Rare Plants and Rare Natural Communities (Attachment 1; revised May 8, 2000).
 - b. A current inventory of the biological resources associated with each habitat type on site and within the area of impact. The Department's California Natural Diversity Data

Mr. Robert Richmond March 12, 2004 Page 3

> Base in Sacramento should be contacted at (916) 327-5960 to obtain current information on any previously reported sensitive species and habitat, including Significant Natural Areas identified under Chapter 12 of the Fish and Game Code.

- c. An inventory of rare, threatened, and endangered species on site and within the area of impact. Species to be addressed should include all those which meet the CEQA definition (see CEQA Guidelines, Section 15380).
- d. Discussions regarding seasonal variations in use by sensitive species of the project site as well as the area of impact on those species, using acceptable species-specific survey procedures as determined through consultation with the Department. Focused speciesspecific surveys, conducted in conformance with established protocols at the appropriate time of year and time of day when the sensitive species are active or otherwise identifiable, are required.
- 3. A thorough discussion of direct, indirect, and cumulative impacts expected to adversely affect biological resources. All facets of the project should be included in this assessment. Specifically, the DEIR should provide:
 - a. Specific acreage and descriptions of the types of wetlands, coastal sage scrub, and other sensitive habitats that will or may be affected by the proposed project or project alternatives. Maps and tables should be used to summarize such information.
 - b. Discussions regarding the regional setting, pursuant to the CEQA Guidelines, Section 15125(a), with special emphasis on resources that are rare or unique to the region that would be affected by the project. This discussion is critical to an assessment of environmental impacts.
 - c. Detailed discussions, including both qualitative and quantitative analyses, of the potentially affected listed and sensitive species (fish, wildlife, plants), and their habitats on the proposed project site, area of impact, and alternative sites, including information pertaining to their local status and distribution. The anticipated or real impacts of the project on these species and habitats should be fully addressed.
 - d. Discussions regarding indirect project impacts on biological resources, including resources in nearby public lands, open space, adjacent natural habitats, riparian ecosystems, and any designated and/or proposed NCCP reserve lands. Impacts on, and maintenance of, wildlife corridor/movement areas, including access to undisturbed habitats in adjacent areas, should be fully evaluated and provided. A discussion of potential adverse impacts from lighting, noise, human activity, exotic species, and drainage. The latter subject should address: project-related changes on drainage patterns on and downstream of the project site; the volume, velocity, and frequency of existing

Mr. Robert Richmond March 12, 2004 Page 4

and post-project surface flows; polluted runoff; soil erosion and/or sedimentation in streams and water bodies; and post-project fate of runoff from the project site.

- e. Discussions regarding possible conflicts resulting from wildlife-human interactions at the interface between the development project and natural habitats. The zoning of areas for development projects or other uses that are nearby or adjacent to natural areas may inadvertently contribute to wildlife-human interactions.
- f. An analysis of cumulative effects, as described under CEQA Guidelines, Section 15130. General and specific plans, and past, present, and anticipated future projects, should be analyzed concerning their impacts on similar plant communities and wildlife habitats.
- g. If applicable, an analysis of the effect that the project may have on completion and implementation of regional and/or subregional conservation programs. Under Section 2800 through Section 2840 of the Fish and Game Code, the Department, through the NCCP program, is coordinating with local jurisdictions, landowners, and the Federal Government to preserve local and regional biological diversity. Coastal sage scrub is the first natural community to be planned for under the NCCP program. The Department recommends that the Lead Agency ensure that the development of this and other proposed projects do not preclude long-term preserve planning options and that projects conform with other requirements of the NCCP program. Jurisdictions participating in the NCCP program should assess specific projects for consistency with the NCCP Conservation Guidelines. Additionally, the jurisdictions should quantify and qualify: 1) the amount of coastal sage scrub within their boundaries; 2) the acreage of coastal sage scrub habitat removed by individual projects; and 3) any acreage set aside for mitigation. This information should be kept in an updated ledger system.
- 4. Mitigation measures for adverse project-related impacts on sensitive plants, animals, and habitats. Measures to fully avoid and otherwise protect Rare Natural Communities (Attachment 2) from project-related impacts. The Department considers these communities as threatened habitats having both regional and local significance.

Mitigation measures should emphasize avoidance, and where avoidance is infeasible, reduction of project impacts. For unavoidable impacts, off-site mitigation through acquisition and preservation in perpetuity of the affected habitats should be addressed. The Department generally does not support the use of relocation, salvage, and/or transplantation as mitigation for impacts on rare, threatened, or endangered species. Studies have shown that these efforts are experimental in nature and largely unsuccessful.

This discussion should include measures to perpetually protect the targeted habitat values where preservation and/or restoration is proposed. The objective should be to offset the project-induced qualitative and quantitative losses of wildlife habitat values. Issues that should be addressed include restrictions on access, proposed land dedications, monitoring

Mr. Robert Richmond March 12, 2004

Page 5

and management programs, control of illegal dumping, water pollution, increased human intrusion, etc. Plans for restoration and revegetation should be prepared by persons with expertise in southern California ecosystems and native plant revegetation techniques. Each plan should include, at a minimum: (a) the location of the mitigation site; (b) the plant species to be used; (c) a schematic depicting the mitigation area; (d) time of year that planting will occur; (e) a description of the irrigation methodology; (f) measures to control exotic vegetation on site; (g) success criteria; (h) a detailed monitoring program; (i) contingency measures should the success criteria not be met; and (j) identification of the entity(ies) that will guarantee achieving the success criteria and provide for conservation of the mitigation site in perpetuity.

Mitigation measures to alleviate indirect project impacts on biological resources must be included, including measures to minimize changes in the hydrologic regimes on site, and means to convey runoff without damaging biological resources, including the morphology of on-site and downstream habitats.

- 5. A California Endangered Species Act (CESA) Permit must be obtained, if the project has the potential to result in "take" of species of plants or animals listed under CESA, either during construction or over the life of the project. CESA Permits are issued to conserve, protect, enhance, and restore State-listed threatened or endangered species and their habitats. Early consultation is encouraged, as significant modification to a project and mitigation measures may be required in order to obtain a CESA Permit. Revisions to the Fish and Game Code, effective January 1998, may require that the Department issue a separate CEQA document for the issuance of a 2081 permit unless the project CEQA document addresses all project impacts to listed species and specifies a mitigation monitoring and reporting program that will meet the requirements of a 2081 permit. For these reasons, the following information is requested:
 - a. Biological mitigation monitoring and reporting proposals should be of sufficient detail and resolution to satisfy the requirements for a CESA Permit.
 - b. A Department-approved Mitigation Agreement and Mitigation Plan are required for plants listed as rare under the Native Plant Protection Act.
- 6. Descriptions and analyses of a range of alternatives to ensure that alternatives to the proposed project are fully considered and evaluated. The analyses must include alternatives that avoid or otherwise reduce impacts to sensitive biological resources. Specific alternative locations should be evaluated in areas of lower resource sensitivity where appropriate.
- 7. The Department has responsibility for wetland and riparian habitats. It is the policy of the Department to strongly discourage development in wetlands or conversion of wetlands to uplands. We oppose any development or conversion which would result in a reduction of wetland acreage or wetland habitat values, unless, at a minimum, project mitigation assures there will be "no net loss" of either wetland habitat values or acreage. Development and conversion include but are not limited to conversion to subsurface drains, placement of fill

Mr. Robert Richmond March 12, 2004

Page 6

or building of structures within the wetland, and channelization or removal of materials from the streambed. All wetlands and watercourses, whether intermittent or perennial, should be retained and provided with substantial setbacks which preserve the riparian and aquatic values and maintain their value to on-site and off-site wildlife populations.

- a. If the site has the potential to support aquatic, riparian, or wetland habitat, a jurisdictional delineation of lakes, streams, and associated riparian habitats should be included in the DEIR, including a delineation of wetlands pursuant to the U. S. Fish and Wildlife Service wetland definition adopted by the Department¹. Please note that some wetland and riparian habitats subject to the Department's authority may extend beyond the jurisdictional limits of the U.S. Army Corps of Engineers.
- b. The project may require a Lake or Streambed Alteration Agreement, pursuant to Section 1600 et seq. of the Fish and Game Code, with the applicant prior to the applicant's commencement of any activity that will substantially divert or obstruct the natural flow or substantially change the bed, channel, or bank (which may include associated riparian resources) of a river, stream or lake, or use material from a streambed. The Department's issuance of a Lake or Streambed Alteration Agreement for a project that is subject to CEQA will require CEQA compliance actions by the Department as a responsible agency. The Department as a responsible agency under CEQA may consider the local jurisdiction's (lead agency) Negative Declaration or Environmental Impact Report for the project. To minimize additional requirements by the Department pursuant to Section 1600 et seq. and/or under CEQA, the document should fully identify the potential impacts to the lake, stream or riparian resources and provide adequate avoidance, mitigation, monitoring and reporting commitments for issuance of the agreement².

The Department appreciates the opportunity to comment on your project and look forward to working with the County and their consultants. Please contact Warren Wong of the Department at (858) 467-4249 if you have any questions or comments concerning this letter.

KA Clandards Sincerely,

William E. Tippets Environmental Program Manager

Attachments (2)

¹ Cowardin, Lewis M., et al. 1979. <u>Classification of Wetlands and Deepwater Habitats of the United</u> <u>States</u>. U.S. Department of the Interior, Fish and Wildlife Service.

² A Streambed Alteration Agreement form may be obtained by writing to: Department of Fish and Game, 4949 Viewridge Avenue, San Diego, CA 92123, by calling (858) 636-3160, or by accessing the Department's web site at www.dfg.ca.gov/1600.
Guidelines for Assessing the Effects of Proposed Projects on Rare, Threatened, and Endangered Plants and Natural Communities

State of California THE RESOURCES AGENCY Department of Fish and Game December 9, 1983 Revised May 8, 2000

The following recommendations are intended to help those who prepare and review environmental documents determine when a botanical survey is needed, who should be considered qualified to conduct such surveys, how field surveys should be conducted, and what information should be contained in the survey report. The Department may recommend that lead agencies not accept the results of surveys that are not conducted according to these guidelines.

1. Botanical surveys are conducted in order to determine the environmental effects of proposed projects on all rare, threatened, and endangered plants and plant communities. Rare, threatened, and endangered plants are not necessarily limited to those species which have been "listed" by state and federal agencies but should include any species that, based on all available data, can be shown to be rare, threatened, and/or endangered under the following definitions:

A species, subspecies, or variety of plant is "endangered" when the prospects of its survival and reproduction are in immediate jeopardy from one or more causes, including loss of habitat, change in habitat, over-exploitation, predation, competition, or disease. A plant is "threatened" when it is likely to become endangered in the foreseeable future in the absence of protection measures. A plant is "rare" when, although not presently threatened with extinction, the species, subspecies, or variety is found in such small numbers throughout its range that it may be endangered if its environment worsens.

Rare natural communities are those communities that are of highly limited distribution. These communities may or may not contain rare, threatened, or endangered species. The most current version of the California Natural Diversity Database's List of California Terrestrial Natural Communities may be used as a guide to the names and status of communities.

2. It is appropriate to conduct a botanical field survey to determine if, or to the extent that, rare, threatened, or endangered plants will be affected by a proposed project when:

- a. Natural vegetation occurs on the site, it is unknown if rare, threatened, or endangered plants or habitats occur on the site, and the project has the potential for direct or indirect effects on vegetation; or
- b. Rare plants have historically been identified on the project site, but adequate information for impact assessment is lacking.
- 3. Botanical consultants should possess the following qualifications:
 - a. Experience conducting floristic field surveys;
 - b. Knowledge of plant taxonomy and plant community ecology;
 - c. Familiarity with the plants of the area, including rare, threatened, and endangered species;
 - d. Familiarity with the appropriate state and federal statutes related to plants and plant collecting; and,
 - e. Experience with analyzing impacts of development on native plant species and communities.
- 4. Field surveys should be conducted in a manner that will locate any rare, threatened, or endangered species that may be present. Specifically, rare, threatened, or endangered plant surveys should be:
 - a. Conducted in the field at the proper time of year when rare, threatened, or endangered species are both evident and identifiable. Usually, this is when the plants are flowering.

When rare, threatened, or endangered plants are known to occur in the type(s) of habitat present in the project area, nearby accessible occurrences of the plants (reference sites) should be observed to determine that the species are identifiable at the time of the survey.

- b. Floristic in nature. A floristic survey requires that every plant observed be identified to the extent necessary to determine its rarity and listing status. In addition, a sufficient number of visits spaced throughout the growing season are necessary to accurately determine what plants exist on the site. In order to properly characterize the site and document the completeness of the survey, a complete list of plants observed on the site should be included in every botanical survey report.
- c. Conducted in a manner that is consistent with conservation ethics. Collections (voucher specimens) of rare, threatened, or endangered species, or suspected rare, threatened, or endangered species should be made only when such actions would not jeopardize the continued existence of the population and in accordance with applicable state and federal permit requirements. A collecting permit from the Habitat Conservation Planning Branch of DFG is required for collection of state-listed plant species. Voucher specimens should be deposited at recognized public herbaria for future reference. Photography should be used to document plant identification and habitat whenever possible, but especially when the population cannot withstand collection of voucher specimens.
- d. Conducted using systematic field techniques in all habitats of the site to ensure a thorough coverage of potential impact areas.
- e. Well documented. When a rare, threatened, or endangered plant (or rare plant community) is located, a California Native Species (or Community) Field Survey Form or equivalent written form, accompanied by a copy of the appropriate portion of a 7.5 minute topographic map with the occurrence mapped, should be completed and submitted to the Natural Diversity Database. Locations may be best documented using global positioning systems (GPS) and presented in map and digital forms as these tools become more accessible.
- 5. Reports of botanical field surveys should be included in or with environmental assessments, negative declarations and mitigated negative declarations, Timber Harvesting Plans (THPs), EIR's, and EIS's, and should contain the following information:
 - a. Project description, including a detailed map of the project location and study area.
 - b. A written description of biological setting referencing the community nomenclature used and a vegetation map.
 - c. Detailed description of survey methodology.
 - d. Dates of field surveys and total person-hours spent on field surveys.
 - e. Results of field survey including detailed maps and specific location data for each plant population found. Investigators are encouraged to provide GPS data and maps documenting population boundaries.
 - f. An assessment of potential impacts. This should include a map showing the distribution of plants in relation to proposed activities.
 - g. Discussion of the significance of rare, threatened, or endangered plant populations in the project area considering nearby populations and total species distribution.
 - h. Recommended measures to avoid impacts.
 - i. A list of all plants observed on the project area. Plants should be identified to the taxonomic level necessary to determine whether or not they are rare, threatened or endangered.
 - j. Description of reference site(s) visited and phenological development of rare, threatened, or endangered plant(s).
 - k. Copies of all California Native Species Field Survey Forms or Natural Community Field Survey Forms.
 - 1. Name of field investigator(s).

1

j. References cited, persons contacted, herbaria visited, and the location of voucher specimens.

ATTACHMENT 2

Sensitivity of Top Priority Rare Natural Communities in Southern California

Sensitivity rankings are determined by the Department of Fish and Game, California Natural Diversity Data Base and based on either number of known occurrences (locations) and/or amount of habitat remaining (acreage). The three rankings used for these top priority rare natural communities are as follows:

S1.# Less than 6 known locations and/or on less than 2,000 acres of habitat remaining.

S2.# Occurs in 6-20 known locations and/or 2,000-10,000 acres of habitat remaining.

S3.# Occurs in 21-100-known locations and/or 10,000-50,000 acres of habitat remaining.

The number to the right of the decimal point after the ranking refers to the degree of threat posed to that natural community regardless of the ranking. For example:

S1.<u>1</u> = <u>very threatened</u> S2.<u>2</u> = <u>threatened</u> S3.<u>3</u> = no current threats known

Sensitivity Rankings (February 1992)

<u>Rank</u>

Community Name

S1.1

Mojave Riparian Forest Sonoran Cottonwood Willow Riparian Mesquite Bosque Elephant Tree Woodland Crucifixion Thorn Woodland Alithorn Woodland Arizonan Woodland Southern California Walnut Forest Mainland Cherry Forest Southern Bishop Pine Forest **Torrey Pine Forest Desert Mountain White Fir Forest** Southern Dune Scrub Southern Coastal Bluff Scrub Maritime Succulent Scrub Riversidean Alluvial Fan Sage Scrub Southern Maritime Chaparral Valley Needlegrass Grassland Great Basin Grassland Mojave Desert Grassland Pebble Plains Southern Sedge Bog Cismontane Alkali Marsh

Southern Foredunes Mono Pumice Flat Southern Interior Basalt Flow Vernal Pool

Venturan Coastal Sage Scrub Diegan Coastal Sage Scrub Riversidean Upland Coastal Sage Scrub Riversidean Desert Sage Scrub Sagebrush Steppe Desert Sink Scrub Mafic Southern Mixed Chaparral San Diego Mesa Hardpan Vernal Pool San Diego Mesa Claypan Vernal Pool Alkali Meadow Southern Coastal Salt Marsh Coastal Brackish Marsh Transmontane Alkali Marsh Coastal and Valley Freshwater Marsh Southern Arroyo Willow Riparian Forest Southern Willow Scrub Modoc-Great Basin Cottonwood Willow Riparian Modoc-Great Basin Riparian Scrub Mojave Desert Wash Scrub Engelmann Oak Woodland Open Engelmann Oak Woodland Closed Engelmann Oak Woodland Island Oak Woodland California Walnut Woodland Island Ironwood Forest Island Cherry Forest Southern Interior Cypress Forest Bigcone Spruce-Canyon Oak Forest

Active Coastal Dunes Active Desert Dunes Stabilized and Partially Stabilized Desert Dunes Stabilized and Partially Stabilized Desert Sandfield Mojave Mixed Steppe Transmontane Freshwater Marsh Coulter Pine Forest Southern California Fellfield White Mountains Fellfield

S2.3

Bristlecone Pine Forest Limber Pine Forest

CDFG Attachment 2 for NOP Comment Letters

S2.1

S1.2

S2.2

STATE OF CALIFORNIA -- BUSINESS, TRANSPORTATION AND HOUSING AGENCY

ARNOLD SCHWARZENECGER, Governor

DEPARTMENT OF TRANSPORTATION

DISTRICT 12 3337 MICHELSON DRIVE SUITE 380 IRVINE, CA 92612-8894 PHONE (949) 724-2255 FAX (949) 724-2592 TTY (949) 756-7813



Flax your powert Be energy efficient!

FAX AND MAIL

March 15, 2004

Mr. Kobert Richmond County Of Orange IWMD 320 N. Flower Street, Suite 400 Santa Ana, CA 92703

File: IGR/CEQA SCH#: 1999041035 Log #: 1373 SR: 74, I-5

Subject: NOP for the 2001 Prima Deshecha General Development Plan (aka SEIR 597)

Dear Mr. Richmond,

Thank you for the opportunity to comment on the NOP for the 2001 Prima Deshecha General Development Plan (aka SEIR 597) received February 25, 2004. The project is located in the County of Orange area of San Juan Capistrano off Ortega Highway (SR-74) and La Pata. The project consists of landslide stabilization of the site, and maintenance of environmental mitigation and restoration areas. This project is Amendment No. 2 to the 2001 General Development Plan.

Caltrans District 12 is a reviewing agency on this project and has the following comments:

- 1. It is proposed to incorporate additional landfill design considerations into a previously drafted General Development Plan (GDP). This is a 5-zone concept with Zone 5 heing reserved for the development of La Pata Avenue as identified in a County of Orange 1996 Interim Project Design Report. As reported, no work will be done at this time in this Zone. Because Zone 5 is an important element in future arterial development, any future work to Zone 5 should include a complete traffic study to address traffic impacts in the region, the change in LOS from existing conditions and a discussion of the use of public traffic and landfill vehicles utilizing the road.
- 2. The Lead Agency should include the Department of Transportation in its close coordination on regional transportation needs with regards to La Pata Avenue.
- 3. All care should be taken to contain loads resulting from construction and operation of the landfill, from blowing over or onto State Right of Way or facilities.

"Catrans improves mubility across California"

Please continue to keep us informed of this project and other future developments, which could potentially impact our transportation facilities. If you have any questions or need to contact us, please do not besitate to call Maureen El Harake at (949) 724-2086.

Sincerely,

Zen Robert Jos

District 12 IGR/Community Planning Branch

C: Terry Pencovic, HQ IGR/CEQA Terry Roberts, OPR Raouf Moussa, Traffic Operations - South

"Caltrans improves mobility across California"

Richmond, Bob

From: Hagthrop, Linda Sent: Monday, March 15, 2004 2:08 PM To: Richmond, Bob Cc: info Subject: FW: SEIR 597 Bob, here's your response.

-----Original Message-----From: Adrian Peters [mailto:Apeters@Talega.com] Sent: Monday, March 15, 2004 10:31 AM To: 'info@iwmd.ocgov.com' Subject: SEIR 597

Bob,

We appreciate the opportunity to review the Notice of Preparation (NOP) for SEIR 597. At this time Talega Associates, LLC does not have any comments on the NOP; however, we reserve the right to comment on the SEIR once it becomes available for public review.

We would be grateful if you could keep us informed of issues involved with the Prima Deshecha landfill. In the meantime if you have any questions or would like to discuss any elements of this document with me in further detail please feel free to call me at (949) 498-1366.

Sincerely,

Adrian J. Peters, AICP Senior Project Manager

APPENDIX B

2001 GDP Amendment 2

Amendment No. 2 2001 General Development Plan

Prima Deshecha Landfill County of Orange, California September 1, 2006 Amendment No. 1: October 2002 Amendment No. 2: August 2006

Prepared For:

COUNTY OF ORANGE Integrated Waste Management Department 320 North Flower Street, Suite 400 Santa Ana, California 92703



Prepared By:

BRYAN A. STIRRAT & ASSOCIATES 1360 Valley Vista Drive Diamond Bar, California 91765 (909) 860-7777



BONTERRA CONSULTING 151 Kalmus Drive, Suite E-200 Costa Mesa, California 92626 (714) 444-9199 NOTE: THE FOLLOWING (DRAFT) AMENDMENT NO. 2 TO THE 2001 GENERAL DEVELOPMENT PLAN (GDP) IS PRESENTED IN STRIKE-AND-ADD FORMAT IN ORDER TO FACILITATE REVIEW OF THE PROPOSED GDP MODIFICATIONS

PRIMA DESHECHA LANDFILL AMENDED 2001 GENERAL DEVELOPMENT PLAN

TABLE OF CONTENTS

GLOSSARY OF ACRONYMS

EXECUTIVE SUMMARY ES					
1.0	INTRODUCTION				
	1.1 1.2 1.3 1.4 1.5 1.6	Site Development History Site Location Current Site Status Summary of the 2001 GDP Purpose and Need for the Project Project Objectives of the General Development Plan			
		1.6.1 1.6.2 1.6.3	Solid Waste Management Objectives Circulation Objectives Recreation Objectives	1-10 1-10 1-10	
2.0	GDP ELEMENTS				
	2.1	Landfill Plan			
		2.1.1 2.1.2 2.1.3 2.1.4 2.1.5	Introduction Design Criteria Environmental Protection Elements Interim and Final Revegetation/Landscaping Final Closure and Post-Closure Maintenance Funding	2-1 2-2 2-8 2-12 2-13	
	2.2	2 Circulation and Roadway Plan		2-15	
		2.2.1 2.2.2 2.2.3	Introduction Arterial Extensions Internal Circulation System	2-15 2-16 2-18	
	2.3	2.3 Recreation Activities Plan		2-19	
		2.3.1 2.3.2 2.3.3 2.3.4 2.3.5	Introduction Recreational Use Potential Phasing for Interim and Ultimate Recreational Uses Natural Open Space Regional Park Financing Plan	2-19 2-20 2-20 2-22 2-22	

PRIMA DESHECHA LANDFILL AMENDED 2001 GENERAL DEVELOPMENT PLAN

TABLE OF CONTENTS (Continued)

3.0	GDP	3-1				
	3.1 3.2	Overview Landfill Plan Phasing		3-1 3-1		
		3.2.1 3.2.2	Zone 1 - Landfilling Zone 4 - Landfilling	3-2 3-7		
	3.3	Recreational Plan Phasing		3-7		
		3.3.1 3.3.2 3.3.3	Phasing of Recreational Uses in Zone 1 Phasing of Trail Uses Phasing of Recreational Uses in Zone 4	3-8 3-8 3-9		
	3.4 3.5	Circulation Plan Phasing Factors Affecting Phasing		3-9 3-9		
		3.5.1 3.5.2 3.5.3	Factors Affecting Landfill Phasing Factors Affecting Circulation Phasing Factors Affecting Recreation Phasing	3-10 3-10 3-10		
	3.6	Subsequent Actions		3-11		
4.0	LAN	DFILL R	4-1			
5.0	TEC	TECHNICAL REFERENCES 5				

PRIMA DESHECHA LANDFILL AMENDED 2001 GENERAL DEVELOPMENT PLAN

TABLE OF CONTENTS (Continued)

LIST OF TABLES

- Table 1:Summary of Remaining Capacity and Life (as of June, 2006)
- Table 2:Summary of Capacity, Excavation Volume and Life for Zone 1(By Phase)
and Zone 4 (as of June, 2006)

LIST OF FIGURES

- Figure 1: Regional Location
- Figure 2: Site Location
- Figure 3: Zone Uses
- Figure 3A: Zone Uses with zone 5 Constructed Prior to Zone 4
- Figure 3B: Zone Uses with zone 4 Constructed Prior to Zone 5
- Figure 4: Proposed Final Grading Plans
- Figure 5: Line of Sight Cross-Section Location Map
- Figure 6: Line of Sight Cross-Sections for San Juan Capistrano
- Figure 7: Line of Sight Cross-Section for San Clemente
- Figure 8: Typical Cross-Section of Fill Slopes
- Figure 9: Typical Cross-Sections of Liner
- Figure 10: Conceptual Golf Course Plan Zone 1
- Figure 11: On-Site Conceptual Circulation Plan
- Figure 12: Excavation Phasing Plan Zone 1
- Figure 13: Fill Phasing Plan Zone 1
- Figure 14: Excavation Phasing Plan Zone 4
- Figure 15: Fill Phasing Plan Zone 4

SECTION 1.0

٠

÷

.

,

1.0 INTRODUCTION

1.1 SITE DEVELOPMENT HISTORY

In February, 1973, the Orange County Board of Supervisors (Board of Supervisors) directed the adoption of a multi-use concept of refuse disposal/recreational development to establish the Prima Deshecha Landfill. At that time, it was determined that a master park plan was not necessary for the site. The disposal of municipal solid waste was initiated in 1976, in an area now known as Waste Management Unit 2 (WMU2). In December, 1976, a planning consultant was retained to prepare a General Development Plan (GDP) to combine both recreational and refuse disposal plans for the site.

An Interim Project Report/Environmental Impact Analysis for the Prima Deshecha site was submitted in August, 1978 to the County of Orange (County) Harbors Beaches and Parks (HBP) Commission. The report contained an Interim Plan and two ultimate Alternative Schematic Plans. Alternative 2 (an 81 million cubic yard refuse plan covering 800 acres of landfill area and 200 acres of borrow area for a total of 1,000 acres) was recommended by the Commission and subsequently adopted by the Board of Supervisors in December, 1978. That Alternative Schematic Plan was further refined and provided the basis for the 1979 Prima Deshecha GDP as well as the initial and current-Solid Waste Facilities Permit (SWFP) No. 30-A-0019 for the site. In 1980, the disposal operations were moved to a second active area known as Waste Management Unit 1 (WMU1).

In 1994, an updated draft GDP was prepared and described in a Program Environmental Impact Report (EIR 548) which was certified in November, 1995. The draft 19945 GDP itself was not approved pending additional landfill design considerations to be negotiated with the City of San Clemente. Negotiations with the City of San Clemente were <u>consequently</u> completed and the Board of Supervisors approved a Memorandum of Understanding (MOU) with the City of San Clemente on July 1, 1997, which included design features to be implemented at the landfill. Conditions governing landfill design and operations within the jurisdiction of the City of San Juan Capistrano were also subsequently negotiated and documented in a MOU approved by the Board of Supervisors on September 12, 1995, and a Conditional Use Permit (CUP) approved by the City of San Juan Capistrano on December 5, 1995. The previously proposed 1994 GDP was then modified in 2001 in an effort to provide compliance with the design and operational conditions set for both Cities. The MOU-approved-design features contained within both approved MOUs were have been incorporated into this-the 2001 GDP, which replacesd the 1979-1994 GDP and currently serves as the future planning guide for the Prima Deshecha site. Also incorporated into theis 2001 GDP are related_design requirements for design and operations conditions for development of the landfill within the jurisdiction of the City of San Juan Capistrano. These are documented in a MOU approved by the Board of Supervisors on September 12, 1995 and a Conditional Use Permit (CUP) approved by the City of San Juan Capistrano on December 5, 1995. The previously proposed 1995 GDP has been modified in an effort to ensure compliance with the design and operational conditions for both Cities while also remediating a landslide which occurred in May, 1998 in a stockpile area south of the Prima Deshecha Cañada streamchannel. This modification has resulted in enlarging the limit of the Zone 1 landfill plan and re-routing to the south a portion of the stream Prima Deshecha Cañada channel impacted by the landslide. This results in a total landfill area for the entire site of approximately 800 acres, as compared to 1,000 acres in the approved 1979 GDP.

The 2001 Prima Deshecha GDP was further modified as an outcome of negotiations between the County and Rancho Mission Viejo, LLC (RMV), the adjacent landowner to the north and east of the Prima Deshecha Landfill property. These negotiations resulted in a Settlement Agreement and Covenant and Declaration of Restrictions (collectively referred to herein as RMV agreements) on 945 acres comprising the eastern portion of the Prima Deshecha property. The RMV agreements identifiedy a Landfill Operations Area and Restricted Area on the eastern 945 acres of the Prima Deshecha Landfill property (referred to in the RMV agreements as the Burdened Property) and specifiedy conditions and restrictions for each of those areas. One design change to the 2001 GDP due to the RMV agreements wasis a reduction in the Zone 4 refuse footprint previously established at 412 acres to 409 acres. The total Zone 4 area (including some cut areas outside refuse limits) remaineds the same at 473 acres. The agreements also identifiedy RMV's Benefited Property and placed conditions on a No-Build Area within the Benefited Property adjacent to and east of the Prima Deshecha property. These agreements also contain requirements for La Pata Avenue funding, maintenance and litter control. Collectively, the RMV agreements constitute Amendment No. 1 to the 2001 GDP. Reference is made to the RMV agreements and conditions placed on the Prima Deshecha Landfill property in this Amendment No. 2 to the 2001this GDP, where appropriate.

EIR 575 addresses environmental impacts resulting from the 2001 GDP for the property, which includes replacing the existing long term plan for the landfill,

recreational uses and traffic circulation. EIR 575 also addresses near-term activities planned for the next phases of landfill development.

Supplemental Environmental Impact Report 597 (SEIR 597) addresses environmental impacts of the project proposed in Amendment No. 2 to the 2001 GDP, herein referred to as "Amendment No. 2." The Amendment No. 2 Proposed Project considers a conservative measure in the areal extent of disturbance anticipated for potential landslide remediation to achieve slope stability within Zones 1 and 4, potential stockpile and trail areas and additional area for ancillary facilities. Amendment No. 2 incorporates biological mitigation measures for landfill project impacts within the revised potential disturbance areas. Landfill phasing is also updated (within previously approved landfill limits) and a regional environmental enhancement program detailing restoration and mitigation opportunities on site is included in the Proposed Project description. These regional environmental enhancement program opportunities would generally be made available to non-IWMD public and private project applicants.

The Proposed Project as proposed in Amendment No. 2 will not change the overall landfill airspace capacity, maximum daily waste intake, footprint of the refuse prism, or final landfill post-closure elevations.

1.2 SITE LOCATION

The 1,530-acre Prima Deshecha Landfill site is located in south Orange County (see Figure 1). The County-owned site includes acreage within the jurisdictions of the cities of San Juan Capistrano (570 acres) and San Clemente (133 acres). The remaining 827 acres are within unincorporated Orange County. The operator of the site, County of Orange Integrated Waste Management Department (IWMD), has prepared a-this GDP Amendment No. 2 for the site. The GDP is a planning document to guide coordinated long-term implementation of both interim and ultimate site development uses.

The Prima Deshecha site lies in the hills of southeastern Orange County. Ground elevations on the site range from 230 feet above mean sea level (amsl) at the southwest boundary of the site to a maximum elevation of 1,125 feet amsl at the northeast boundary of the site. The Prima Deshecha Cañada watercourse traverses the site from the northeast to the southwest. Two major utility easements, including a 150-foot wide San Diego Gas and Electric (SDG&E) easement and a 200-foot wide Southern California Edison (SCE) easement, extend through the central portion of the site which separate the western (Zone 1) and eastern (Zone 4) components of the landfill property. These features are illustrated in Figure 2.

1.3 CURRENT SITE STATUS

WMU1 (see Figure 2), as well as<u>and newly</u> lined waste disposal areas, Phases A, and A1, <u>A2</u>, <u>B</u>, <u>B1</u> and <u>C1</u> located directly <u>north</u>, east and <u>north</u> <u>southeast</u> of WMU1, respectively, overliey <u>170</u> <u>approximately 193</u> acres of the western portion of the site and is <u>are</u> located entirely within the City limits of San Juan Capistrano. Landfilling of municipal solid waste with some biosolids (i.e., digested sewage sludge) continues at the site, with a total in-place refuse <u>air space</u> volume (as of January, 2004<u>6</u>) of <u>13.1</u> <u>29.7</u> million cubic yards (mcy) (see Table 1). This includes Rrefuse that has also been landfilled in WMU2 (see Figure 2), a 33-acre portion of the site located east of the current entrance facilities in unincorporated Orange County. The in-place refuse volume (as of January, 2001) in WMU2 is approximately <u>1.5</u> mcy. The first lined waste disposal area for the site, referred to as Phase A, was developed in 1998 and is located east of WMU1. Landfilling has proceeded easterly into Phase A, and-north into Phases A1 and A2, east into Phase C1 and southeast into Phases B and B1. WMU1 is currently used as the wet weather disposal area.

The Prima Deshecha Landfill is a state-designated Class III facility which is permitted for the disposal of non-hazardous municipal solid waste and biosolids. No liquid or hazardous wastes are accepted or proposed for on-site disposal. The IWMD currently operates under Waste Discharge Requirements (WDR 93-86 and WDR 89-102R9-2003-0306) issued by the San Diego Regional Water Quality Control Board (SDRWQCB), a <u>Solid Waste Facility Permit (SWFP)</u> (No. 30-AB-0019) issued by the County Health Care Agency, Environmental Health Division which is the Local Enforcement Agency (LEA) and concurred on by the California Integrated Waste Management Board (CIWMB), and CUP 95-4 issued by the City of San Juan Capistrano, as well as other permits required for environmental monitoring and control systems.

In addition to the landfilling operations, related facilities and activities at the site include:

- Personnel facilities, site office, and equipment maintenance facilities.
- An energy recovery facility (ERF) that converts landfill gas to electricity.

- A Household Hazardous Waste Collection Center (HHWCC) and a facility for the temporary storage of hazardous materials.
- One fee booth and two scales. <u>A third scale is planned for 2007, pending</u> <u>approval of easement encroachments by San Diego Gas & Electric</u> <u>Company.</u>
- A landfill gas collection and flaring system for the site, which consists of vertical and horizontal extraction wells, collection piping and a flaring facility.
- Groundwater monitoring wells located in the vicinity of the current and future refuse disposal areas.
- A groundwater extraction system located downstream of WMU1, consisting of a pump station and four groundwater extraction wells within the alluvial material of the Prima Deshecha Cañada watercourse.

«Temporary bridge/underpass for landfill operations-

- A leachate collection and recovery system (LCRS) for the lined areas, including a-<u>two_10,00015,000-</u>gallon collection tanks.
- Perimeter and interior drainage facilities.
- Biological mitigation sites to the east and south of Zone 1.
- The main concrete-lined/desilting basin for Zone 1.

Since the site began operations in 1976, there has been substantial residential development in the area, particularly to the south within the City of San Clemente. The 2001 GDP and subsequent amendments takes the current and projected proximity of urban development into account.

1.4 SUMMARY OF THE 2001 GDP

The Prima Deshecha 2001 GDP provides for the effective management of multiple uses on the site, including solid waste disposal, various regional park and recreational uses, and implementation of a key arterial highway and road extentsion included identified in the Master Plan of Arterial Highway (MPAH),

Orange County Circulation Plan (OCCP), and Circulation Elements of the Cities of San Juan Capistrano and San Clemente.

The GDP divides the total 1,530-acre site into five zones for planning purposes as shown on Figure 3 and briefly described below:

Zone Descriptions

- Zone 1: This zone includes the currently active refuse disposal area. In approximately 1<u>3</u>8 years, as of January, 2004<u>6</u>, or the year 2019, the Zone 1 landfill would-<u>is projected to</u> be completely filled. After closure activities have been completed, satisfactory access established, and sufficient settlement has occurred, the ultimate recreational uses <u>in</u> <u>Zone 1</u> as identified in a needs analysis could be implemented.
- Zone 2: This zone identifies all of the recreational trails that traverse the property. On-site city trails around Zone 1 can be used throughout the development of Zones 1 and 4 as long as the protection of public health and safety can be provided. Trails depicted along the perimeter of Zone 4 will be available during filling operations of Zone 1, but will be closed to the public during the filling operations in Zone 4 for the protection of public safety. The trails along the perimeter of Zone 4 are restricted by the RMV agreements to ten (10) feet below and to the south and west of the existing ridgeline between the Prima Deshecha property and adjacent RMV property. The GDP proposes to eventually connect the County trail along Zone 4 with on-site City trails proposed along Zone 1 to provide a complete loop for trail users. These on-site trail connections would not occur until Zone 4 is closed and a trail crossing under future La Pata Avenue is provided to the south. Ongoing Ddiscussions commencing in January, 2001 with representatives of the Cities of San Clemente and San Juan Capistrano are focused on identifying specific trail alignments around Zone 1. These on-site trail connections would not occur until Zone 4 is closed and a trail crossing under future La Pata Avenue is provided to the south.
- Zone 3: This zone contains native vegetation, including coastal sage scrub habitat used by the California gnatcatcher, and mixed chaparral. The intent of the GDP is to retain the majority of Zone 3 in a native state. Some habitat enhancement could be implemented, where portions of these areas have been disturbed in the past or to compensate for lost

habitat associated with the development of the GDP or with other development in Orange County. The Proposed Project as contained within this Amendment No. 2 incorporates much of Zone 3 into a premitigation area for landfill impacts or regional environmental enhancement area.

- Zone 4: This zone is planned for future refuse disposal after Zone 1 is filled to capacity <u>currently estimated to occur in the year 2019</u>. Two-The western boundariesy for this zone have been defined. The boundary ultimately established depends on whether the Zone 4 landfill or<u>was</u> established to provide for the future extension of La Pata Avenue is constructed first (see Figures 3A and 3B)through the site. This zone is to remain undisturbed during development of Zone-1, and following its closure, serves as the final refuse disposal site on the property <u>after Zone 1 is closed</u>. Following closure of Zone 4, planned for the year 2067 and after sufficient settlement has occurred, implementation of the ultimate recreational activities can begin. These activities would be determined through a needs analysis and park plan undertaken near the time of closure. The current post-closure <u>general plan</u>-designated land use for Zone 4 is a regional park.
- Zone 5: This zone encompasses the area of disturbance for construction of La Pata Avenue. The boundaries of Zone 5 in the 2001 GDP were defined based on a conceptual alignment design and the assumption that La Pata <u>Avenue</u> would be constructed prior to the Zone 4 landfill (see Figure 3A). Should the Zone 4 landfill be constructed first (Figure 3B) and based on the final design for La Pata <u>Avenue</u>, the area of disturbance for La Pata <u>Avenue</u> would be modified. <u>The County</u> <u>Resources and Development Management Department (RDMD) is</u> currently conducting a feasibility study to determine the best alignment for the La Pata Avenue extension. Should this study indicate an <u>alternative alignment to that shown in the 2001 GDP is the preferred</u> project. the boundaries of Zone 5 in the GDP will be revised upon approval of the preferred alignment.

The GDP does not specify a defined set of uses for the remaining property outside the boundaries of the five zones. This <u>remaining</u> property is currently used for ancillary landfill operations (i.e., landfill gas flare facility, <u>energy recovery</u> <u>facility [ERF]</u>), landfill infrastructure (i.e., scalehouse, field offices) and viewshed protection. It is anticipated that in the future it could accommodate additional uses such as <u>landslide remediation</u>, temporary stockpile, trails, biological

mitigation, flood control facilities, recreational trail staging area(s) and open space buffer. It is the intent of Amendment No. 2 to define a conservative increase in the temporary limits of disturbance around Zones 1 and 4 that would accommodate these features.

The landfill plan for Zones 1 and 4 provides for approximately 116.8 mcy of remaining refuse capacity (as of January, 2001). The <u>a</u> total <u>airspace</u> site capacity, including an <u>remaining airspace</u> in <u>place refuse</u> volume of <u>142.5</u> <u>14.6</u> mcy in <u>Zone 1 and WMU2</u> (as of <u>JanuaryJune 30</u>, 20016), is <u>of</u> <u>131.4</u> <u>171.6</u> mcy (see Table 1, page T-1). Features accommodated by the GDP for continued development of the landfill include a liner and leachate collection and removal system (LCRS) in future areas of the landfill zones, relocation of the HHWCC (completed in 2005), continued expansion of the landfill gas control system, modifications to the landfill gas control flare station, an ERF and potential acquisition of potable water (which will be the subject of subsequent environmental analysis)potable and non-potable water lines. The GDP also identifies provides locations for detention desilting basins and permanent LCRS facilities. The desilting basin system for Zone 4 is proposed to be modified in Amendment No. 2 to minimize impacts to sensitive biological resources.

Biological mitigation requirements for the landslide remediation element of the 2001 GDP resulted in the establishment of 70 acres of on-site mitigation at the PDL. Figure 4 illustrates the location of these sites, which offset project impacts to riparian, wetland, coastal sage scrub and upland habitats.

Recreational uses would ultimately be provided in two of five zones on the site. The GDP also accommodates the extension of Camino de los Mares and La Pata Avenue through the site, consistent with approved alignments shown on the MPAH. <u>As indicated above, the final alignment for the La Pata Avenue</u> <u>extension will be determined through the completion of a detailed feasibility study</u> <u>and final design</u>.

1.5 PURPOSE AND NEED FOR THE PROJECT

The shortage of landfill space in the urban areas of Southern California is well documented and the value of the Prima Deshecha site <u>to the Southern California</u> region as a permitted landfill the Southern California region is <u>to</u>-be preserved and protected maintained.

A GDP is a planning document to guide coordinated long-term implementation of both interim and ultimate site development uses. The GDP for the Prima Deshecha site also provides for effective management of multiple uses on the site which include solid waste management, regional park and recreational development and completion of a major <u>roadway</u> link <u>identified</u> in the MPAH. The GDP enables concurrent implementation of these activities through a phasing program which allows multiple uses to be adequately separated or buffered during site development.

The GDP describes numerous operational needs, planning issues, opportunities and constraints, which have influenced the configuration and phasing of the GDP. It should be recognized that meeting solid waste disposal needs is the most important function on the site and will take precedence over other possible uses. To that end, the general development concept is for the site to function primarily as a solid waste disposal facility and, secondly, to provide interim and ultimate recreational opportunities for the general public. No priority issue is foreseen with implementation of the MPAH, which is accommodated by the two landfill zones.

Amendment No. 2 and SEIR 597 further modify the GDP with the incorporation of the elements of the Proposed Project listed below. In aggregate, these project modifications address the entire property and are necessary to (1) provide for slope stability measures, without which the PDL would be required to cease operations; (2) accommodate future landfill-related features such as LFG perimeter probes, LFG collection header lines; (3) modify the desilting basin location for Zone 4 to avoid sensitive biological resources and unstable areas; and (4) implement measures needed to provide for the long-term success of the environmental mitigation and restoration components of the overall GDP. In order to facilitate the assessment of biological impacts from future landfill operations within both zones and coordinate pre-mitigation of these impacts, the limits of disturbance around each zone have been refined to accommodate these features.

Also included within Amendment No. 2 is the development of a comprehensive pre-mitigation plan for the site, which will allow landfill operations to continue seamlessly, as mitigation of operational impacts on biological resources will have occurred before the impact happens. Additionally, on-site regional enhancement opportunities will have been identified for offsetting environmental impacts of other County (or third party) projects. Proposed project elements in Amendment No.2 will not alter landfill airspace capacity, maximum daily waste intake, footprint of the refuse prism, or final landfill post-closure elevations.

1.6 PROJECT OBJECTIVES OF THE GENERAL DEVELOPMENT PLAN

Implementation of the 2001 Prima Deshecha GDP is intended to achieve several solid waste management, circulation and recreation objectives. The objectives identified below were utilized in the preparation of the GDP, particularly with regard to the landfill design and operations. <u>These objectives continue to apply</u> to the 2001 GDP, as amended, herein.

1.6.1 SOLID WASTE MANAGEMENT OBJECTIVES

- Optimize the use of the site as a long-term waste disposal facility.
- Provide for consistency with the County of Orange Integrated Waste Management Plan (CIWMP), adopted County and applicable City General Plans, zoning regulations and compliance with City MOU design and operational conditions.
- Provide a long-term, regional solid waste management facility with appropriate safeguards to protect public health and safety as well as water, air, soil and other important resources which exist on-site and on surrounding property.

1.6.2 CIRCULATION OBJECTIVES

- Provide for regional as well as local access to landfill operations and recreational activities on the site.
- Accommodate adopted MPAH arterial highway alignments through the site.

1.6.3 RECREATION OBJECTIVES

- Identify preferred activities that include a variety of passive and limited active recreational uses which respond to the changing recreational needs in the region.
- Provide a phased recreation concept for implementation of both interim and ultimate recreational uses as solid waste management activities allow.
- Consider recreation goals and objectives of the Orange County Master Plan of Regional Parks as well as with those identified in the San Juan Capistrano and San Clemente General Plans.

- Provide opportunities for the benefit of the public to develop and operate recreation facilities within the regional park.
- Preserve regionally significant habitat on the site which will be set aside as natural reserves and which can be utilized throughout the region for educational purposes.
- Provide essential linkages to the existing multiple use trails in the area which will also serve the recreation elements of the GDP.

SECTION 2.0

2.0 GDP ELEMENTS

The 2001 Prima Deshecha GDP encompasses the following three elements:

- Landfill Plan
- Circulation Plan
- Recreation Plan

The 2001 Prima Deshecha GDP provides for the effective management of multiple uses on the site, including solid waste disposal, implementation of key arterial highway and road extensions included in the MPAH, OCCP, and Circulation Elements of the Cities of San Juan Capistrano and San Clemente and various regional park and recreational uses.

The three elements are considered together in the 2001 GDP in order to <u>ensure allow</u> <u>for</u> compatibl<u>eility of the existing</u>, interim and ultimate uses on the site as well as to achieve the goals and objectives of approved local and regional plans and policies. It is important to note that unless stated otherwise, references made to the GDP refer to all three of the elements listed above. <u>However</u>, <u>Amendment No. 2 to the GDP amends</u> the Landfill Plan and Recreation Plan elements of the 2001 GDP only.

2.1 LANDFILL PLAN

2.1.1 INTRODUCTION

The 2001 GDP divides the total 1,530-acre site into five zones for planning purposes as shown on Figure 3. Two zones are designated for landfilling. The 125-acre WMU1-landfill is located within Zone 1 which refuse disposal area ultimately provides for 271 total acres (see Figure 4) to be filled over approximately 18.113 years (from January, 20046), and the 2001 GDP identified a total disturbance area (including cut slopes) around Zone 1 of 327 acres. The total updated acreage for Zone 1 including non-refuse cutpotential disturbance areas identified in Amendment No. 2 is 319 437 acres (see Figure 43). Zone 4 is designated for the development of a future landfill area in the east central portion of the site (see Figure 4). Zone 4 consists of 409 acres (including the <u>14</u> acres of the original 33-acre WMU2 disposal area, WMU2) to be filled with refuse and would be in active operation for approximately 48-3 years following the closure of Zone 1. The total disturbance acreage for Zone 4 identified in the 2001 GDP (including cut slopes) is 473 acres. This area has been updated in Amendment No. 2 to a non-refuse cut areaspotential disturbance area of 641 acres. (Figure 4). The potential disturbance area is based on acreage that may

be needed for future landslide remediation, stockpiling, trails and other ancillary facilities.

The Prima Deshecha landfill is permitted to accept up to 4,000 tons per day (tpd). The site life for Zone 1 is based on an incremental increase in the disposal rate of $\underline{up \ to \ 2,500}$ tpd in 20006 to 4,000 tpd by year 2014, and 4,000 tpd thereafter. The site life for Zone 4 is based on a daily refuse inflow rate of 4,000 tpd. Details on these assumptions are presented in Section 2.1.2.

The entrance facilities, field offices, ERF and landfill gas flare station are located in the north central portion of the site just west of Zone 4. The HHWCC, which is utilized to collect household hazardous waste (HHW) generated by households within the County, <u>wasis currently</u> located in Zone 1 and is planned to be but has <u>been</u> relocated near the field offices. The collected HHW is temporarily stored on-site and disposed off-site or recycled appropriately. The HHW is not disposed in the landfill.

2.1.2 DESIGN CRITERIA

Prior to preparing refined plans for the future landfill operations in Zones 1 and 4, a number of landfill design criteria were developed. The criteria balanced applicable regulatory standards with surrounding land use compatibility and onsite environmental considerations. Although these criteria reduce the potential capacity of the site for landfilling, the GDP does provide for a substantial landfill life of approximately 66-61 years, as of January, 20064. The established criteria are considered critical to creating an optimal relationship between waste disposal operations and other site uses.

The landfill development criteria consider grading and height limits, site capacity, and design issues, as described in the following sections.

Landfill Grading and Height Limits

The first step in preparing plans for landfilling operations was to determine the boundaries of areas that could be <u>made</u> available for landfilling. Establishment of the landfill footprints was primarily driven by geotechnical recommendations for slope stability. Consideration was also given to avoiding minimizing impacts <u>on</u> environmentally sensitive areas, ridgelines, areas that had high visibility from current and future development, transmission line corridors and future roadway easements. The Southern California Edison Company (SCE), San Diego Gas and Electric Company (SDG&E) and existing and future public roadway easements through the site were to be avoided in establishing refuse

boundaries. Grading and height limits imposed by the Cities of San Clemente and San Juan Capistrano, as stated in MOU's with these Cities and by the RMV agreements discussed in Section 1.1, are reflected below:

City of San Clemente Requirements

Grading and Height Limits:

- The final grading elevation in Zone 1 is to be at a height below the level of the ridgeline behind Zone 1 as viewed from Truman Benedict Elementary School located at 1251 Sarmentoso, San Clemente, California 92673.
- The final grading elevation of Zone 4 is to be no higher than 1,010 feet, thereby minimizing the visual impact to residents of the City.
- It is understood that the side slopes of the landfill may be steeper than what is reflected in EIR No. 548 as may be determined by the County to be necessary to offset landfill capacity lost due to the height limits described above and in the MOU.

Zone Boundary Adjustments:

- The boundaries of Zone 1 to be adjusted from the Zone 1 boundaries described in EIR No. 548, with the understanding that additional boundary adjustments may be required for Zone 1 due to geotechnical conditions, drainage, and other environmental constraints provided such adjustment does not result in a final grading plan with a height limit greater than that specified above and in the MOU.
- The boundaries of Zone 4 to be adjusted as may be determined by the County to be necessary to offset landfill capacity lost due to the height limits described above and in the MOU, provided that any additions to the landfill resulting from the modification of the Zone 4 boundaries are not visible from within the City limits.

City of San Juan Capistrano Requirements

For that portion of the site within the boundaries of the City of San Juan Capistrano, the following conditions apply:

- According to the CUP, the permittee shall not extend any portion of the landfilling operation above the plane or outside the surface area of the fill design as established by the GDP, Prima Deshecha Landfill and the MOU executed between the City and County. "Landfilling operation" refers to that portion of the subject property in which waste is to be permanently placed and then buried under daily, interim and final cover material.
- According to the CUP, with the exception of previously established on-site borrow areas, the permittee shall excavate cover material only within the refuse disposal limits established by the GDP. This condition does not prohibit excavations outside those areas for the purposes of constructing drainage structures or noise abatement devices; performing work necessary to abate hazards to public or private property; or assuring slope stability.
- According to the MOU, the County will install and maintain final landfill whichlandfill grades which result in no silhouetting above and along the General Plan - designated "major ridgeline" which forms the northern and western edges of the landfill site boundary such that no landfill operations or placement of landfill materials will visually encroach upon the designated General Plan ridgeline or be viewed from Ortega Highway.

Rancho Mission Viejo, LLC (RMV) Requirements

The RMV agreements contain grading and fill restrictions on a Landfill Operations Area and a Restricted Area within 945 acres of the east portion of the Prima Deshecha Landfill to minimize effects of the landfill operation on RMV's Benefited Property to the north and east of the landfill property (see Section 1.1 for a description of those areas).

A Covenant and Declaration of Restrictions:

- Height Restriction The height of any waste or refuse placed within the Landfill Operations Area shall not exceed one thousand and ten (1,010) feet above mean sea level unless and until appropriate measures (including any measures required by Applicable Laws) are taken to screen from view any portion of such waste or refuse that is visible from RMV's Benefited Property.
- Design Adjustments The County shall adjust the design of any landfill within the Burdened Property and/or take such other steps as may be necessary to

prevent or mitigate any landfill-related costs and impacts on the Benefited Property.

- Site Grading and/or Soil Filling The County may perform site grading and/or soil filling (to maximize capacity) within the Restricted Area in support of Landfill Operations so long as the same are not visible from the Benefited Property.
- Ridgeline Buffer So as to maintain a buffer zone and natural barrier to minimize viewing, noise, dust, litter, and other effects, if any, of Landfill Operations on the Benefited Property, the height and natural contour of the existing ridge lines most immediately contiguous to the boundary line between the Prima Deshecha Landfill property and the RMV Benefited Property shall not be materially modified by the County; provided, nothing in this provision shall preclude the installation and maintenance of landscaping along said ridge line.

Settlement Agreement:

- Zone 4 Phasing of Landfill Activities County anticipates that landfill operations within the Landfill Operations Area will occur in phases moving in a west to east progression, and that the placement of refuse within the easterly portion of the Landfill Operations Area will not commence until after the year 2025. County, shall in good faith, consider any alternatives or suggestions tendered by RMV prior to materially altering the general west-toeast phasing scheme.
- Viewshed Analysis County is required to conduct a viewshed analysis with respect to the impacts of Zone 4's development on the RMV Benefited Property, and, if needed, prepare and implement a viewshed protection and landscaping plan ("VPL Plan").

Based on these limitations, two landfill zones have been established which represent approximately 800-1.078 total acres of the 1,530-acre site (see Figure 4). It should be noted that not all of the 1078 acres may be actually needed for landfill activities, or disturbed. Zone 1 consists of a total of 319-437 acres in the western portion of the site, of which 271 acres will be refuse fill areas. The final landfill grades for Zone 1 are below the major ridgelines which form the northern and western edges of the landfill site boundary as viewed from Ortega Highway, the valleys of San Juan Capistrano and the Truman Benedict Elementary School

in San Clemente. Line of sight cross-sections for the Zone 1 final <u>landfill</u> grades are presented in Figures 5, 6 and 7 from a viewpoint in San Juan Capistrano and from the Truman Benedict Elementary School in San Clemente.

Zone 4 provides-consists of a total of 473-<u>641</u> acres in the east central portion of the site for future landfill development. The Zone 4 refuse fill area, including that area<u>14</u> acres of the previously filled in WMU2, is 409 acres. The Zone 4 landfill has a maximum elevation of 1,010 feet as previously described in EIR 548 and in accordance with MOU requirements of the City of San Clemente, and the Settlement Agreement with RMV. The Zone 4 footprint has been established to minimize the need for landslide stabilization mitigation<u>avoid impacts to</u> Segunda Deshecha Cañada and to minimize visual impacts from the south. itIt is anticipated that landslide remediation activities included in Amendment No. 2 to the GDP will temporarily affect the Zone 4 Landfill Operations Area and Restricted Area as defined by the RMV agreements; however, these activities will be consistent with the approved uses for the property as summarized above-Zone 4 final slopes and deck area have been modified to provide a more natural, undulating appearance.

Criteria Used to Determine Site Capacity

The following site capacity criteria was developed for planning purposes to estimate the associated site life and potential impacts of landfilling on regional traffic, air quality, etc.:

- <u>Zone 1</u> Incremental incremental increase in average daily refuse input of up to 3,0002.500 tpd from in January 1, 20046 to December 31, 2008; 3,500 tpd from January 1, 2009 to December 31, 2013; and 4,000 tpd from January 1,by year 2014 and thereafter to the end of the Zone 1 life.
- Zone 4 average daily refuse input assumed to be 4,000 tpd.
- Average daily refuse input includes up to 350 tpd of biosolids, over a six-day week. The biosolids input meets or exceeds the current and projected needs of the South Orange County Wastewater Authority (SOCWA) and other agencies.
- Average effective density of waste in-place for Zones 1 and 4 of 1,333 lbs/cy; no net export or import of soil for cover material.

- Soil cover material usage based on a volume ratio of disposed refuse to cover soil of 4:1, including the use of tarps, greenwaste, or other alternative daily covers.
- The capacity and life of the site could be increased or decreased based on changes in landfill design standards and regulations, changes in daily cover use and final cover requirements, <u>changes in refuse density</u>, changes to the assumed refuse inflow rates and other similar changes.

Another provision to protect the future solid waste disposal capacity of landfills in California is the California Integrated Waste Management Act of 1989 (AB 939) requirement to maximize the diversion of recyclable materials from landfillings. Materials such as plastics, paper, aluminum and vegetative matter, if not landfilled, can result in a substantial reduction in the amount of refuse deposited in landfills. In consideration of the policy to use landfill capacity for revenue by importing refuse from adjacent counties, the current estimates for solid waste disposal under the GDP represent an estimated maximum disposal rate for the near future, notwithstanding implementation of the AB 939 requirements for the diversion of recyclable materials from landfilling.

Criteria Related to Site Design

The following principal design criteria for development of the landfill are based on State minimum standards:

- Minimum top deck slope of three (3) percent.
- Finished surface slopes of <u>a maximum 3</u>:1 between benches, 3.5:1 gross (see Figure 8).
- Interior interim surface slopes <u>maximum of 3.5:1</u>.
- Access roadway width 50 feet.
- Maximum access roadway slope seven (7) percent.
- Cut slopes inclined as permitted by geology and liner stability. The maximum depth will be used under these constraints.

- Bottom slope <u>approximately</u> two (2) percent minimum and maximum as permitted by fill stability and constructability.
- Maintenance and drainage benches 15 feet wide measured level from the flow line. Final maintenance and drainage benches – spaced at 4<u>5</u>0-foot <u>maximum</u> vertical intervals, assuming 20-foot lifts.

Geotechnical Design Criteria

The Prima Deshecha site is part of the Peninsular Ranges Province of Southern California. Exposed bedrock materials consist predominantly of marine and nonmarine sedimentary rock of the Capistrano and Monterey formations and the San Onofre Breccia which are overlain by bedrock-derived landslides, modern alluvial deposits in the main drainage channels and various types of native soils. Economically useful geologic resources do not occur on-site, with the exception of materials which may be suitable for daily and final cover or liner construction in further development of the landfill. Given the low strength of some bedrock units and the potential instability of numerous landslides on the site, new slope failures and reactivation of existing landslides are possible. In fact, a landslide occurred in 1998 in a Zone 1 stockpile area and portions of a number of the larger landslides on the east portion of the site display evidence of recent movement under existing conditions. The footprints for Zones 1 and 4 have been configured to enhance stabilization of refuse fills. Other stabilization measures for unstable cut slopes in the various units of bedrock and landslide debris could include construction of low angle (2:1) or shallower cut slopes, buttress and/or stabilization fills, shear keys and structurally reinforced fills. The GDP, as amended, provides for these types of stabilization measures within Zones 1 and 4 by increasing the potential limits of disturbance associated with these features based upon available geotechnical information. Although there is a possibility that further geotechnical analyses and detailed engineering design may result in the determination that additional area is required for landslide stabilization in the future, an effort was made to include conservative limits of disturbance in the GDP Amendment No. 2.

2.1.3 ENVIRONMENTAL PROTECTION ELEMENTS

The design for landfill operations under the GDP includes a number of environmental protection elements which respond to the established GDP goals and applicable local, state and federal regulations. These elements include compliance with surface and groundwater monitoring requirements and air and gas monitoring requirements. These controls are described below:

Groundwater Protection Systems

Leachate is liquid which passes through the landfill, coming in contact with disposed wastes and possibly absorbing contaminants. The sources of moisture in a landfill may include: (1) rainfall which infiltrates the surface cover; (2) moisture in the refuse; and (3) moisture generated by decomposition.

Landfill regulations minimize the production of leachate by preventing infiltration. Infiltration reduction is accomplished by prohibiting disposal of liquid wastes in the landfill, effective drainage management which diverts <u>off-site</u> surface water flows away from the landfill, separating the bottom of the landfill from groundwater by means of a liner, <u>placing cover over waste on a daily basis</u> and placing a low-permeability final cover.

Drainage improvements for the site include perimeter storm drain channels around the fill areas, downdrains and terrace drains on the slopes and desilting basins. Final storm drain improvements are designed to accommodate flows from a 24-hour, 100-year storm event. The existing desilting basin at the south portion of Zone 1 will has been be enlarged and improved (in 2005) for future development of Zone 1. A detention/desilting basin between Zones 1 and 4 is was conceptually proposed in the 2001 GDP to meet stormwater detention/desilting requirements for ultimate development of the landfill in Zone 4. Amendment No. 2 proposes desilting basins around the perimeter of Zone 4 in lieu of a basin downstream of Zone 4 (between Zones 1 and 4) in order to avoid sensitive biological resources. Interim desilting basins will also be constructed as part of on-going landfill operations.

WMU1 contains no liner or underlying LCRS, since landfill operations in this area were initiated in 1980, before the 1984 adoption of the California Code of Regulations (CCR) Title 23, Chapter 15 (now Title 27) which established standards for leachate control. The existing groundwater extraction system was constructed as a condition for acceptance of bio-solids at the landfill. The system consists of four groundwater extraction wells, which are situated in a line roughly perpendicular to the Prima Deshecha Cañada streambed and slightly downstream of the toe of WMU1. The purpose of these wells is to extract groundwater which is flowing down_gradient, away from the landfill, in the alluvium which fillswithin the canyon bottom. The wells are situated so that groundwater flowing downstream down gradient under WMU1 through the alluvial aquifer can be captured, if leachate is detected in the groundwater.

The collected leachate, if any, is tested and initially-disposed off-site or applied to the surface of the lined portion of the landfill as a dust control measure. The final methods of disposal of landfill leachate must beare approved by the SDRWQCB and the LEA. Eventually, depending on the quantities and composition of the leachate, including the degree of contamination, an on-site leachate treatment facility may be installed with the effluent being discharged to the public sanitary sewer system or used as on-site dust control. To date, the extracted groundwater has not been found to be affected by leachate and is used for landfill operational purposes.

For that portion of Zone 1 which is lined, leachate may be collected and disposed. In accordance with Title 27, new areas to be landfilled under the GDP will be underlain by a liner and LCRS. For the development of Zone 1, an alternative liner design petition was prepared and approved by the SDRWQCB in WDR 93-86, Addendum No. 1 and was further modified in WDR R9-2003-0306, Addendum No. 1. The liner system design meets the requirements for alternative designs provided in Section III.A.1.b and Section III.A.3 of the State Water Resources Control Board Resolution No. 93-62. A typical cross-section of the currently approved alternative liner design is included in Figure 9.

Any variations of this design approach for the liner and LCRS will be submitted to the SDRWQCB for approval. Any-Liquid percolating through the landfill would will be collected and will flow by gravity through the above-described systemLCRS to storage tanks that will be are located at the base of the landfill. Collected liquids will be Leachate collected in storage tanks may be transported off-site for treatment and disposal by a licensed hauler or used on-site for dust control in lined areas only as approved by the SDRWQCB. Alternate uses of the collected leachate will need approval by the SDRWQCB-tested prior to final disposition.

Air Quality Protection Systems

Landfill gas in the active fill area is currently collected by an active gas extraction system of horizontal collection piping and vertical wells. The gas is piped to the existing flare station and beyond that facility to the newan eEnergy rRecovery fFacility (ERF). The IWMD has provided for energy recovery as an alternative to continued flaring of the landfill gas. An ERF has been designed, built and is currently being operated by Algonquin Power Systems under a lease with the County. Algonquin Power Systems will own the rights to all the landfill gas from the landfill for at least 20 years. The ERF is located north of WMU2 and northeast of the current scale house. The ERF accomplishes two objectives:
- The ERF provides additional capacity for the destruction of increasing guantities of landfill gas, a by-product of the decomposition of buried refuse.
- The ERF converts the gas to green power or electricity, which is sold to
 SDG&E. Sufficient power may be generated to supply approximately 3,000
 homes.

As the landfill continues to receive refuse, the system will be expanded through the installation of both horizontal collection piping and vertical wells. Collected landfill gas will continue to be converted to electricity with additional flares installed as back-up as capacity requirements dictate. Some minor grading of the area may be necessary to create pads for additional flares and piping. Additional above-grade piping would be required to transport landfill gas from newly developed areas to the existing flare station and on to the energy recovery facilityERF. At closure, the site will still require a flare station and/or a gas utilization facility until landfill gas is no longer produced by the landfilled waste.

As gas flows through the landfill gas collection system, it cools and moisture condenses, resulting in a liquid called condensate. Condensate is separated from the landfill gas and is currently collected in tanks for off-site disposal. The condensate may eventually be treated in a leachate/condensate treatment system prior to discharge or it may be piped to the flare station for combustion. The final disposal methods of landfill gas condensate must be approved by the SDRWQCB, AQMD and the LEA.

The IWMD has provided for energy recovery as an alternative to continued flaring of the landfill gas. An Energy Recovery Facility (ERF) has been designed, built and is currently being operated by NEO California LLC (NEO) under a lease with the County. NEO will own the rights to all the landfill gas from the landfill for at least 20 years. The ERF is located north of WMU2 and northeast of the current scale house. The ERF accomplishes two objectives:

- The ERF provides additional capacity for the destruction of increasing quantities of landfill gas, a by product of the decomposition of buried refuse.
- The ERF converts the gas to electricity, which is sold to SDG&E. Sufficient power may be generated to supply approximately 3,000 homes.

At closure, the site will still require a flare station and/or a gas utilization facility until landfill gas is no longer produced by the landfilled waste.

2.1.4 INTERIM AND FINAL REVEGETATION/LANDSCAPING

An important consideration of the landfill development, for erosion control as well as visual enhancement, is revegetation and landscaping of completed surfaces. Requirements for revegetation and landscaping for the Prima Deshecha Landfill are imposed by State requirements and by the Cities of San Clemente and San Juan Capistrano. Interim revegetation and landscaping requirements are included in the CUP with the City of San Juan Capistrano and the MOU with the City of San Clemente as summarized below:

CUP with City of San Juan Capistrano Requirements

- The final landfill slopes shall be concurrently reclaimed and revegetated within 90 days of completion. If directed by the LEA, the permittee shall install a temporary vegetation cover on all slopes and other areas that are to remain inactive for a period longer than 90 days.
- To the extent possible, revegetation shall blend with species indigenous to the area and be drought tolerant and shall be capable of rapid establishment. Plant selection shall not include exotic, invasive species as determined by the United States Fish and Wildlife Service (USFWS) and the California Department of Fish and Game (DFG).

In addition to the above specific requirements of the CUP, the City of San Juan Capistrano has also requested that the use of Native California plants for revegetation be consistent with the existing plant palette bordering the landfill.

MOU with San Clemente Requirements

- Trees have been planted to screen the landfill in a location which has been approved by the City. The design of the treescape is described in plans and specifications titled "Prima Deshecha Landfill Mass Excavation Grading Plans for Zone 1 - Phase A, "approved by the Board of Supervisors on May 13, 1997 and shall be in accordance with the "Tree Planting Plan" contained in those plans and specifications dated March 23, 1997.
- Interim landscaping treatment is to be provided consistent with erosion control measures required by current regulations.

In addition to the above requirements of the MOU, a Viewshed Protection Plan (VPP) is to be prepared to mitigate views of Zone 4 from San Clemente (to the south) as a condition of approval for EIR 575. The VPP is to be cooperatively prepared between the City of San Clemente and IWMD prior to development of Zone 4.

As required by CCR, Title 27, final landscape plans are included in the existing Preliminary Closure and Post-Closure Maintenance Plan (19932003). Thise document will be updated to reflectvegetation proposed in the Preliminary Closure and Post-Closure Maintenance Plan is a selected plant species at closure which includes native seasonal grasses. Landfill landscaping will not include exotic plant species that may be invasive to native habitats including species on lists A1, A2 and B of the California Exotic Pest Plant Council's list of "Exotic Pest Plants of Greatest Ecological Concern in California as of October 1999" as required in the site's Section 1601 Streambed Alteration Agreement the closure design elements of the approved 2001 GDP, as part of the landfill SWFP revision process.

The Final Closure Plan for each zone will be prepared two years prior to closure of each landfill zone and will include existing or modified landscape plans. The proposed pre-mitigation program in Amendment No.2 will be completed years in advance of closure, and there will be no other landscaping obligations associated with other regional planning efforts. A discussion of funding for final closure and post-closure improvements including landscaping is included in Section 2.1.5.

2.1.5 FINAL CLOSURE AND POST-CLOSURE MAINTENANCE FUNDING

In order to ensure that landfill operators are able to fund final closure improvements for their sites, Title 27 of the CCR requires that operators establish financial assurance mechanisms for both closure and post-closure.

The following describes IWMD's funding for landfill closure/post-closure costs in accordance with the CCR, Title 27.

Closure

Title 27 of the CCR requires that IWMD establish a Closure Escrow Account for each active landfill and to fund each escrow annually so that sufficient funds are set aside for estimated closure costs at the time of closure. The amount required for closure is based on a Preliminary Closure Plan with cost estimates developed by IWMD and approved by the CIWMB/SDRWQCB in accordance with State regulations. The IWMD sets aside \$1.15 per ton of refuse disposed at each of its three active landfills for closure funding. The funds are deposited into a trust account and transferred annually to each landfill's Closure Escrow

Account. The escrow amounts are calculated by dividing annual refuse tonnage disposed by remaining permitted landfill capacity and multiplying the result by the remaining closure cost amount. The Preliminary Closure Plan cost for the entire Prima Deshecha site approved by the <u>SDRWQCB/CIWMB in February</u>, <u>19932003 and 2004, respectively</u>, is \$44,725,31846,435,450. As of January, 2004<u>5</u>, Zone 1 closure is anticipated in approximately <u>18-13</u> years and Zone 4 in approximately <u>66-61</u> years. As of December <u>31, 2000August 2006</u>, the Prima Deshecha Landfill Closure Escrow Account balance is \$4,648,959 <u>\$15,920,921.33</u>.

Post Closure Maintenance Care

Under Title 27, Chapter 6, Article 2 Financial Assurance for Post Closure Maintenance, Section 22210 allows the Integrated Waste Management Department (IWMD) to guarantee post-closure funding for the landfills through an approved financial assurance mechanism known as a Pledge of Revenue. The Pledge of Revenue financial assurance mechanism allows the IWMD to forego funding of the post-closure costs in advance and pay the costs as they are incurred, beginning when closure construction is complete and continuing annually through a minimum 30-year post-closure maintenance period.

The IWMD elected to use a Pledge of Revenue financial assurance mechanism and has entered into a formal agreement with CIWMB, in accordance with provisions under Title 27. There is not a need or statutory requirement to include IWMD's current funding for post closure maintenance, because the decision to pre-fund post closure maintenance care is an IWMD administrative policy decision.

Title 27 allows the IWMD to guarantee post-closure funding for the landfills through an approved financial assurance mechanism known as a Pledge of Revenue. The Pledge of Revenue financial mechanism allows the IWMD to forego funding of the post-closure costs in advance and pay the costs as they occur, beginning when closure construction is complete and continuing annually through the 30-year post-closure period. The IWMD has elected to use the <u>a</u> Pledge of Revenue financial assurance mechanism, and currently has no funds set aside for post-closure maintenance of the landfills at this time. However, under the IWMD Fifteen-Year Financial Plan, beginning in Fiscal Year 2003/04, IWMD will establish and begin funding a post-closure account within its Environmental Liability Reserve Fund. Funding for post-closure will be achieved by setting aside \$0.38 of the \$1.00 collected per ton for the Environmental Liability Reserve Fund.

Integrated Waste Management System Financial Stability

IWMD has developed amaintains an ongoing Fifteen-Year Financial Plan (Plan). Under the Plan, IWMD has developed tonnage and revenue assumptions and projections that incorporate both operations and regulatory compliance cost obligations, which include funds set aside for closure and post-closure activities as described above. An Independent Engineer has validated the assumptions and projections and concluded that IWMD cashflow and net operating income support the operation and maintenance of the disposal system, under its current per ton tipping fee.

2.2 CIRCULATION AND ROADWAY PLAN

2.2.1 INTRODUCTION

A key element of the GDP is the identification of the circulation and roadway improvements necessary to support the landfilling and recreation uses and to accommodate the arterial highway needs detailed in the currently approved MPAH, OCCP and City Circulation Plans. Internal, local circulation networks will be developed, as necessary, for the phased landfill operations and interim recreation use activities on site. Provisions for the future extension of La Pata Avenue through the site (Figure 11) as a major regional arterial link of the MPAH (Figure 1) are also included in this <u>Amendment No. 2-GDP</u>, as presented in the <u>2001 GDP</u>. The currently approved MPAH and OCCP identify the approved ultimate capacity and conceptual alignment for this roadway.

Other roadways identified in the GDP include the extension of Camino de los Mares through the southwest corner of the site (see Figure 11). The circulation plans for the Cities of San Clemente and San Juan Capistrano identify the alignment for Camino de los Mares, which is proposed to connect to Camino Las Ramblas in San Juan Capistrano. As currently reflected on the MPAH, Camino Las Ramblas would be extended from its current terminus in a northeasterly alignment to the west and north of the landfill boundaries, eventually joining La Pata Avenue.

None of the arterial highway extensions will be constructed by the County IWMD as part of the landfill development. The County's construction of La Pata Avenue is neither contingent nor dependent on the development of the site in accordance with the GDP. Although not all of these circulation improvements will be implemented in conjunction with the GDP, it is important that the GDP not preclude the future development of these approved roadway extensions.

2.2.2 ARTERIAL EXTENSIONS

The following section describes each component of the arterial circulation system (see Figure 11) including the alignment for the connection of La Pata Avenue through the site and the ability to extend Camino de los Mares and Camino Las Ramblas, given the configuration of the GDP. Although these roadways would not be developed as part of the GDP, this analysis was conducted to ensure that the GDP landfilling and recreational uses would not ultimately preclude their extension have accommodated the conceptually proposed alignments for these roadways as depicted in the MPAH, OCCP and City Circulation Plans.

La Pata Avenue

The northern portion of La Pata Avenue currently consists of a 1.5-mile long, three-lane road providing access to the landfill between the northerly limits of the landfill and Ortega Highway. It provides two southbound (uphill) lanes and one northbound (downhill) lane. An extension of La Pata Avenue through the landfill is proposed in the current MPAH and OCCP which would be a new arterial highway accommodating regional transportation demands and could-will also serve current and future development of the landfill. This proposed roadway extension would also include some realignment and widening of the existing La Pata Avenue. Within the aforementioned limits, the MPAH classifies La Pata Avenue as a Primary Arterial Highway from Ortega Highway to just south of the southerly landfill property where it is then classified as a Major Arterial Highway to Avenida Pico. The San Juan Capistrano Circulation Element classifies La Pata Avenue between the landfill's southerly limits and Ortega Highway as a Secondary Arterial Highway which provides equivalent traffic capacity as the County's Primary Arterial Highway. The San Clemente Circulation Element classifies La Pata Avenue as a Primary Arterial Highway.

Ultimate MPAH and OCCP improvements would upgrade existing La Pata Avenue to its current designation of Primary Arterial Highway. The project would include a new four-lane extension of La Pata Avenue through the landfill to join the <u>planned-existing</u> Avenida La Pata in San Clemente. The completion of La Pata Avenue between Ortega Highway and Avenida Pico would complete a major segment of the MPAH and OCCP and improve north-south circulation in the area. Various improvement alternatives for the La Pata Avenue alignment through the site have been previously identified and compared, including alternatives analyzed in Final EIR 548. The alignment, shown in Figure 11, from Ortega Highway through the site is <u>the a</u> County Preferred Alignment as-illustrated in the <u>a</u> La Pata Avenue Project Report (November, 1996), which <u>is-was a</u> County staffrecommended alignment and has not been officially adopted. <u>The County is</u> <u>currently conducting a feasibility study (the La Pata Avenue Gap Closure Study)</u> which will identify the preferred alignment based upon current project objectives and site conditions. Amendment No. 2 of the GDP is being closely coordinated with studies on the La Pata Avenue extension to provide for consistency and compatibility is maintained between both projects. Should the La Pata Avenue Gap Closure Study identify a different alignment that is ultimately be approved, additional amendment ofjustments in the GDP might be required.

Camino de los Mares

Thise existing Camino de los Mares roadway located within the City of San Clemente consists of a four-lane roadway terminating at the southern property line of the site. The extension of Camino de los Mares (north of its existing terminus in Forster Ranch at the southwest landfill boundary) to Camino Las Ramblas would create a secondary arterial highway as designated on the Cities' Circulation Plans. This road extension is shown on Figure 11 in the southwest corner of the property. There is sufficient property available within the GDP boundaries, unaffected by landfill operations, to allow extension of this arterial highway. However, a biological mitigation site and associated conservation easement are in close proximity to the conceptual alignment. The conservation easement contains provisions/permissions that allow the roadway to cross through the easement area. Should mitigation areas be affected, these resources will need to be replaced in a manner approved by the resource agencies. If this extension is constructed, a connection for alternate (southern) access to Zone 1, after closure of the landfill, could be developed. However, in accordance with the MOU with the City of San Clemente, in no event will haulers utilizing the landfill for the disposal of solid waste be permitted to use Camino de los Mares as an access route.

Camino Las Ramblas

The MPAH indicates that Camino Las Ramblas will continue in a northeasterly direction from its intersection with the proposed Camino de los Mares extension and proceed adjacent to the westerly and northerly landfill boundaries, eventually joining La Pata Avenue (see Figure 1). The City of San Juan Capistrano passed a resolution on December 14, 1999 that stipulates the City's intention to pursue deletion of the Camino Las Ramblas extention to La Pata Avenue. Therefore, the City must submit a request to the OCTA to amend the MPAH. Prior to any action taken by OCTA, the City of San Juan Capistrano would be required to

prepare and process a General Plan Amendment (i.e., Circulation Element) Amendment and appropriate CEQA documentation. If the amendment is approved by the City of San Juan Capistrano, the amendment would then be forwarded to the OCTA Board for action. The MPAH Amendment Process could take up to a year or longer to process through the OCTA.

An amendment to the MPAH as a result of actions taken by the City of San Juan Capistrano and/or San Clemente regarding the alignment of these arterial roadways may also necessitate a revision to the 2001 GDP to ensure consistency between the 2001 GDP Circulation Component and the MPAH.

The County and the City of San Juan Capistrano through the GMA process will analyze the need for this facility in the near future.

2.2.3 INTERNAL CIRCULATION SYSTEM

The general development concept is for the site to function primarily as a solid waste disposal facility and, secondly, to provide interim and ultimate recreational opportunities for the general public. To this end, the following criteria has been established for the internal roadway circulation plan:

- Landfill operations shall remain uninterrupted.
- Landfill operations traffic shall be separated from on-site recreational traffic.
- Landfill operations must be allowed to cross under the SDG&E and SCE electrical transmission lines without interrupting traffic flow on La Pata Avenue, if constructed.
- Refuse truck traffic will be approaching the landfill from the north via Ortega Highway and La Pata Avenue.
- If La Pata Avenue is eventually connected to the City of San Clemente roadway system, some trucks collecting refuse in San Clemente may approach the landfill from the south. (However, in accordance with the MOU with the City of San Clemente, haulers utilizing the landfill for refuse disposal cannot use Camino de los Mares as an access route. Also, in the event that the County's existing contracts with haulers to import waste to the landfill from outside Orange County, and require access to the landfill via San Clemente City streets, is required, approval of a designated haul route shall

first be obtained from the City of San Clemente, which shall be compensated for that access in accordance with the MOU.)

Other issues considered in developing the internal circulation system include protection of sensitive biological resources, aesthetic considerations regarding ridgeline impacts, geotechnical constraints and physical constraints associated with the existing utility easements.

The conceptual circulation plan, shown in Figure 11, also-identifies individual onsite access roads, provides for the connection of conceptual alignments for La Pata Avenue and for the extension of Camino de los Mares through the site.

The proposed site access and circulation system is intended to segregate existing and future landfill-related traffic from traffic generated by interim recreational uses. The goal of segregation is complicated by the dynamic nature of landfill-related circulation patterns which will continue to change as the fill area elevation increases and as operations shift from Zone 1 to Zone 4. Currently, landfill access is provided by La Pata Avenue which extends south from Ortega Highway and terminates at the northern boundary of the landfill site.

With the extension of La Pata Avenue through the site, landfill access and traffic circulation will need to be integrated into the preferred alignment plan. After the Zone 1 landfill is closed, an intersection at the landfill entrance is proposed to separate Zone 4 landfill traffic from recreational traffic accessing the regional park or golf course in Zone 1 and multiple-use trail staging areas proposed directly to the west of the entrance facilities will need to be accomodated. After entering the site, landfill traffic for Zone 4 would initially use a northerly perimeter road. Eventually, ILandfill traffic may_access future Zone 4 from a new scalehouse location to the south fromoff the La Pata Avenue extension (see Figure 11). These new entrance facilities in the middle of the property could ultimately could be used for Zone 4 recreational uses following closure.

2.3 RECREATION ACTIVITIES PLAN

2.3.1 INTRODUCTION

Proposed recreational uses on the Prima Deshecha <u>Landfill</u> site should respond to the needs of south Orange County residents. Various agency groups representing the Cities of San Juan Capistrano and San Clemente, the Talega Valley Reserve and the County of Orange have previously recommended that the proposed recreational uses on the site take into consideration the existing recreational activities adjacent to or near the project site (i.e., multiple use trails). These uses should also be consistent with the County of Orange Master Plan of Regional Recreational Facilities which identifies proposed future recreational facilities.

2.3.2 RECREATIONAL USE POTENTIAL

Given the variety of possible recreational uses on the site, research was previously conducted to determine whether the recreational uses would be compatible with adjacent land uses designated on the General Plans of the adjacent cities and the County of Orange. Specifically, the Orange County Public Facilities and Resources Department/Harbors, Beaches and Parks Resources and Development Management Department/Harbors, Beaches and Parks (PF&RDRDMD/HBP) and the Cities of San Juan Capistrano and San Clemente, through the public review process for EIR 548, developed a list of possible recreational uses including a regional staging area and multiple use trails which are currently were appropriate to meet the demands of county-wide residents at the time. However, these demands may change over time by the time the site closes. They will, therefore, be re-evaluated in a needs analysis prior to the time of landfill closure, when the recreational needs can be assessed with greater certainty. EIR 548 was prepared to serve as the environmental documentation for the 1995 Prima Deshecha GDP. The 1995 proposed GDP also included Landfill, Circulation and Recreation Components and was intended to be the long-range planning program for landfill, circulation and recreation programs occurring on the subject property. Although the proposed 1995 GDP was not approved by the Orange County Board of Supervisors, Final EIR 548 was certified in November 1995.

2.3.3 PHASING FOR INTERIM AND ULTIMATE RECREATIONAL USES

The purpose of the recreation plan-component of the GDP is to identify opportunities and locate sites for various interim and ultimate recreational uses in the context of pre- and post-closure landfill operations. Figure 3 shows areas designated for recreational development and landfill operations. The GDP is divided into five zones which delineate current and future landfill operations, and potential interim and ultimate recreational areas. <u>Ongoing Dd</u>iscussions commencing in January, 2001 with representatives of the Cities of San Clemente and San Juan Capistrano are focused on identifying specific trail alignments around Zone 1. Interim uses are those recreational activities which can occur during landfilling operations on the site. The primary use of the site is as a landfill. While waste management operations are occurring, however, limited interim and ultimate recreational activities can occur on other locations at the site depending on the status of landfill <u>closureactivities</u>, satisfactory access and protection of public health and safety.

As previously illustrated, Figure 3 depicts alignments for regional and local riding and hiking trails throughout the site in areas designated as Zone 2. Although some sections of these trails have been constructed, the majority are not yet built. None of the trails on or in the immediate vicinity of the site are yet constructed. For the majority of these trails, final alignments have not yet been determined. Therefore, it is not possible to predict when an individualeach trail would be completed and open for use by the public. Trails depicted along the perimeter of the Zone 4 landfill area will be available as interim recreational use only during filling operations of the Zone 1 landfill. Once landfill operations are moved to Zone 4, it is proposed that this perimeter trail be closed to the public based on protection of public health and safety. PF&RDRDMD/HBP may evaluate the possibility of relocating this trail away from landfill operations. This would depend upon its use and importance as a regional trail and the existence of a through trail along the western perimeter of the site at that time.

A regional park or golf course (see Figure 10) for Zone 1 is a viable use which can be accommodated on implemented as soon as the deck of the Zone 1 landfill<u>once it</u> is closed, sufficient settlement has occurred and concessionaires (in the case of a golf course) are identified. However, a recreation needs analysis will be performed just prior to Zone 1 closure before a final use plan is adopted. As mentioned above, trails may also be implemented in those areas which are unaffected by landfill operations or where landfilling has been completed and formally closed. This allows for staging of recreational uses over an extended time frame when landfill operations would also be occurring on the site.

<u>The 2001 GDP proposed a An ultimate regional park recreational use, (</u>-the proposed <u>`Prima Deshecha Regional Park'</u> as depicted on the Master Plan of Regional Recreational Facilities of the Recreation Element), <u>has been</u> designated for Zone 4. Although the PDL site has been designated as "Regional Park" by the County on the adopted Land Use Element, ultimate regional park recreational uses have not been identified for Zone 4. However, Zone 4 will not be available for regional park use until after the closure of the Zone 4 landfill, which is not anticipated until about 2067. Consequently, a commitment to implement specific uses is not included in the 2001 GDP since recreational demand for this park might be quite different by that time. A needs analysis and park plan reflecting the recreational needs of south Orange County residents will be more appropriately developed nearer to the time of closure of Zone 4. <u>In</u>

addition, biological mitigation and environmental enhancement actions within and around Zone 4 may require a more passive recreational post-closure use.

2.3.4 NATURAL OPEN SPACE

In addition to recreational uses, two areas of the site will be retained as natural open space. Zone 3 contains two natural areas on the site (see Figure 3) which will not be impacted by landfill refuse filling operations and should be protected and retained in their natural state in concert with the Natural Community Conservation Plan (NCCP) program being developed by the County of Orange, landowners, environmental groups and resource agencies. These natural areas include habitat used by the California gnatcatcher which is an avian species protected by the Federal Endangered Species Act. SEIR597 for Amendment No. 2 to the GDP contains an extensive biological pre-mitigation program as well as a regional environmental enhancement program that identifies biological resource enhancement opportunities onsite consistent with the NCCP. These programs will assist in facilitating streamlined regulatory authorizations and continuation of landfill activities through project build-out, while providing for the long-term protection and enhancement of biological resources at the site in a manner consistent with local, State and Federal regional planning objectives. Some habitat enhancement may be implemented on site subject to the provisions of an approved NCCP, where habitat areas were disturbed in the past or to compensate for habitat lost as a result of implementation of the GDP landfilling, circulation and recreational uses.

2.3.5 REGIONAL PARK FINANCING PLAN

The PF&RDRDMD/HBP provides administrative, planning, and operational services for the County regional park system. Funding for PF&RDRDMD/HBP is provided from a percentage of County property tax revenues dedicated to the regional park system. PF&RDRDMD/HBP Capital Project funds are allocated based upon rating and ranking criteria specified within their Five-Year Capital Plan.

The Five-Year Capital Plan is updated annually. County regional park programs and construction of other potential recreational improvements are identified and implemented in conjunction with this Five-Year Capital Plan. Prima Deshecha Landfill is currently designated on the County Master Plan of Regional Recreational Facilities as a proposed regional park. The Five-Year Capital Plan is reviewed by the PF&RDRDMD/HBP Commission, and presented to the Board of Supervisors for approval as part of the County's annual budget process.

County regional parks are designed for passive, open space use; in contrast, urban community parks provide for active recreational uses. If the needs assessment for a regional park indicates that active recreational programs are needed over and above those provided by the County regional park system, those programs become the responsibility of the local municipality park and recreational planning process. The goal of County Regional Recreational Park programs is to accommodate Orange County's regional recreation needs. However, County parks have provided rent-free-leased space for active community uses within regional parklands (i.e., Mile Square Park in Fountain Valley and Yorba Regional Park in Yorba Linda), with the local municipality providing for the programming and operations of these facilities.

Zones 1 and 4 Regional Park Financing

The IWMD will begin preparation of a Final Closure and Post-Closure Maintenance Plan approximately five years prior to the cessation of waste acceptance in Zones 1 and 4. These documents will be submitted to the CIWMB two years prior to the planned closure as required per CCR, Title 27. The closure plan, including final end use, must be approved by regulatory agencies prior to initiation of closure activities. During the five-year period prior to the last date of waste acceptanceclosure, the PF&RDRDMD/HBP will_include the Prima Deshecha Regional Park in their Five-Year Capital Plan. The process will involve a needs analysis for regional, and as appropriate, local uses undertaken in cooperation with adjacent cities and interest groups. A definitive cost study will also be conducted as part of this process once the proposed uses are established.

Zone 2 Trail Financing

The Recreational Element of the County General Plan includes a Master Plan of Regional Riding and Hiking Trails Component. County trail development, maintenance and operations are funded as part of the PF&RDRDMD/HBP Five-Year Capital Program (Fund No. 405 of the County Service Area No. 26 annual budget). Other funding sources include new, private developments and the cities. The City trails proposed in Zone 2 of the GDP are funded by the individual cities.

SECTION 3.0

3.0 GDP PHASING

3.1 OVERVIEW

The implementation of the 2001 GDP landfill plan is proposed to be phased over a span of approximately <u>66-61</u> years from January, <u>20012006</u>. Active landfilling will continue in Zone 1 concurrently with the development of trails for recreational use around the perimeter of <u>the siteZone 1</u>. When landfilling in Zone 1 has ceased, closure activities have been completed, satisfactory access has been established, sufficient settlement has occurred, and landfilling has begun in Zone 4, the ultimate recreational use(<u>s</u>) in Zone 1, a regional park or <u>18-hole golf</u> coursefor Zone 1, as identified in a needs analysis, can be developed. When the landfilling in Zone 4 is complete, the ultimate recreational uses can be developed for that site after closure activities have been completed and sufficient settlement has occurred. The only long-term landfill related activities that will occur on the site after Zone 4 is filled will be associated with the continued collection and disposal of leachate and landfill gas, ongoing maintenance of the landfill final cover, <u>drainage</u> and groundwater monitoring.

The proposed phasing and factors affecting phasing for the landfill, recreation and circulation uses on the site are described in Sections 3.2 through 3.5. In Section 3.6, actions subsequent to approval of the 2001 GDP to plan, design and implement the GDP uses are identified.

3.2 LANDFILL PLAN PHASING

Zone 1 includes the lateral (eastward) and vertical development of the previous WMU1 refuse area from 425-100 acres to 271 acres (see Figure 4) over a period of approximately 48.413 years from 20046. This time span is based on assumptions presented in Section 2.1.2 and Table 2. Zone 4 encompasses a landfill refuse footprint of 409 acres in the east portion of the site, which would last for approximately 48.3 years based on a disposal rate of 4,000 tpd. The total life of the site for landfill purposes, as of JanuaryJune 30, 20064_{7} is estimated to be approximately 616 additional years or to 2067. The total inplace and remaining refuse capacity for the site is summarized in Table 2-(see page T-2). The site is life could be extended if less refuse is accepted for disposal than the projected amount and/or if new technologies are developed which have the effect of increasing the landfill capacity. One such technology is the use of tarps, currently utilized at the landfill, as well as greenwaste or other alternative approaches to daily cover on the active face of the landfill. However, soil for daily cover will continue to be necessary.

As the owner/operator of the landfill, the County of Orange is required to provide amendments to operating permit documentation to the LEA at least every five years, or more frequently, to discuss any changes in the site design, operations plan and/or the remaining life of the landfill.

For operational guidance and to allow a closer examination of the environmental impacts of the GDP over time, phasing scenarios were developed for the landfill which include a four major phases plan for sequential of development of for Zone 1 (see Figures 12 and 13 Phases A through D) and a nine- major phase plan for the development of Zone 4 (see Figure 14Phases A through I). Smaller sub-phases of development are implemented based on operational needs and capital budgeting requirements. Table 2 presents a summary of information on refuse capacity, excavation volumes and projected life for Zones 1 (by phase) and 4. Figures 12 through 14-15 illustrate the sequential progress major phases of the landfilling activities at the site from existing conditions (20056) to the end of landfilling operations under the GDP. These major phases will be developed in sub-phases as operations progress. Figures 12 and 13 present the phasing limits of excavation and refuse filling for Zone 1, respectively. Figures 14 and 15 present the phasing limits of excavation and refuse filling for Zone 4, respectively. The excavation phasing limits have been modified for both Zones 1 and 4 in Amendment No. 2 to reflect a greater potential disturbance limit for landslide remediation. The design is intended to minimize the need for stockpiling and double handling of cover material (soil), although substantial stockpiling will be necessary. In general, soil is excavated in new development areas on the site and is stockpiled in future disposal areas and/or is used as daily cover for ongoing operations. It is intended that all excavated soil be used on-site, and that no exporting or importing of soil will be necessary. Should there be a need to import or export soil in the future, that plan would be subject to separate CEQA documentation.

3.2.1 ZONE 1 - LANDFILLING

As indicated above, four phases of excavation are proposed for the development of Zone 1 (refer to Figure 12). Filling has occurred in the landfill areas, WMU1, Phases A and A1 with future development continuing to the south and east of the present active footprint of the landfill. The ultimate development of Zone 1 will provide a total area of 271 acres for refuse fill. Eight phases of excavation within major Phases A through D are currently proposed for the full development of Zone 1 (refer to Figure 13). Filling has occurred in landfill areas designated as WMU1 (Summer 1980), Phases A (February 1999), A1 (November 2000), C1 (July 2002), and B (July 2004), A2 (September 2005), B1 (December 2005), and <u>C2 (expected October 2006) with future development continuing to the east of</u> <u>these active landfill areas.</u> The ultimate development of Zone 1 will provide a <u>total area of 271 acres for refuse fill.</u> The following describes the lined phases of <u>development for Zone 1 (as of 2006).</u>

Phase A and A1

The first phase (Phase A) of lateral development in Zone 1 extends to the east of WMU1 and was the first lined cell in Zone 1. Development of Phase A began in July, 1997 and a <u>A</u> liner and LCRS was were installed prior to refuse filling in Phase Athat area, which began in February, 1999.

The Phase A development required the excavation of approximately 4.1 mcy of soil, the majority of which (2.4 mcy) was stockpiled to the north and west of WMU1 (Stockpile 2). Approximately 1.7 mcy of the material excavated from Phase A was stockpiled in a 17-acre area to the south of the landfill (Stockpile No. 1), adjacent to the Prima Deshecha Cañada water course running through the site. As a result of unusually heavy rains during early 1998, a landslide developed in Stockpile No. 1.

It is estimated that the Stockpile 1 Landslide is more than 2.0 mcy (which is greater than the 1.7 mcy of stockpiled material). It extends over 2,500 feet adjacent to the Prima Deshecha Cañada stream much of this length shows areas of distress to the stream. At the present time, flows through the stream have not been completely blocked although movement of the slide mass has narrowed the stream along the toe of the failure. Recent monitoring has indicated that the slide mass is moving in a northeasterly direction (directly into the stream) and it is anticipated that the landslide will eventually block flows through the channel. Remediation of this landslide is proposed as part of the Phase B development.

As a result of the landslide, a Phase A1 area was developed to ensure ongoing operations during the approval process for the landslide remediation plan. Phase A1 is a lined area and is located north of WMU1 (previously Stockpile Area No. 2) and provides<u>d</u> an additional 1.8 years of airspace capacity.

Phase A2

The Phase A2 area was recently constructed (2005) and encompasses approximately 7.5 acres of grading and 3.1 acres of liner area and requires the excavation of approximately 60,000 cy of soil material, 5,000 cy of fill and subsequent lining of the subgrade. Phase A2 was developed to eliminate a low area created from the adjacent refuse fill. This will ease operational access and drainage. The Phase A2 development area will provide a small increment of additional airspace (estimated at 680,000 cy) for the area between Phases A1 and C1 (see Figure 13). The primary purpose of lining this area is to improve operations and drainage with additional refuse fill. A majority of the cut slope excavation for the Phase A2 liner subgrade was completed concurrently with Phase A1, as the majority of grading proposed for Phase A2 (cut and fill) is required to tie in adjacent bench grades and to accommodate a perimeter drainage system.

<u>The additional liner to be placed for Phase A2 will require realignment of two</u> <u>existing concrete v-ditches to provide perimeter drainage control for the</u> <u>expanded liner. The interim perimeter drainage control system for Phase A2 will</u> <u>be designed to accommodate a 100-year, 24-hour storm event and to minimize</u> <u>erosion.</u>

Phase B

The excavation for the next phase (Phase B) of development will be occurred along the southern boundary of Phase A and future Phases C and DZone 1 (see Figure 12). The design basis for the excavation of Phase B is-was the remediation of the landslide in Stockpile No. 1 which includes<u>d</u> realignment of the Prima Deshecha Cañada stream-channel to the south. The remediation project includes<u>d</u> the removal of alluvial and stockpile material and recompaction of soil fill to stabilize the landslide, and the realignment of the natural open channel so that it would carrycarries water to the east and south of the existing stockpile. To provide a positive grade required to establish this new stream alignment (a minimum 1% grade), some ponding would be required and compacted fill would need to be placed in the channel and surrounding area upstream of the stockpileZone 1.

This project would involve the complete removal of approximately 1.7 mcy of stockpiled material and recompaction of approximately 1.2 mcy to stabilize the landslide, achieve channel grades, and provide liner subgrade. The excavation of approximately 1.4 mcy of hillside, comprised of non-native grasslands, would also be required to achieve the realigned channel grade. The realigned natural stream channel would cover in excess of 6 acres, would vary in width from 50 to 85 feet and would be approximately 3,100 feet long. The approximate 85-foot width of the majority of new channel corridor would be adjacent to a 25-foot wide landfill perimeter maintenance road, improved drainage channel and a setback for a final cover keyway. The realigned stream would be outside the area of

landfill operations, thus minimizing indirect impacts on streambed biological resources from daily landfill operations and would provide opportunity for additional riparian habitat.Phase B excavation involved the complete removal of approximately 1.7 mcy of stockpiled material and recompaction of approximately 1.0 mcy to stabilize the landslide, achieve realigned channel grades, and provide the liner subgrade. The realigned natural stream channel is in excess of six acres, varying in width from 50 to 101 feet and is approximately 3,100 feet long. The approximate 85 to 101 foot width of the majority of new channel corridor is adjacent to a 25-foot wide landfill perimeter maintenance road, improved drainage channel and a setback for a final cover. The realigned stream is located outside the area of landfill operations, thus minimizing indirect impacts on streambed biological resources from daily landfill operations and providing opportunity for additional riparian habitat.

The entirePart of the area to the south of Phase A and future Phases C and D will bewas excavated for Phase B; however, only the west portion of the Phase B will beexcavation was lined initially for refuse filling (see Figure 13). The remaining area of Phase B will be being used as a stockpile area and will be lined as part of Phase D for refuse fillingfuture phases of development (future major Phases C and D. Future Phase B filling will extend north into the WMU1 fill area. Phase B filling will eventually extend north into the WMU1 fill area.

<u>Phase B1</u>

The Phase B1 area (construction completed in (2005), encompasses approximately 10.1 acres of excavation and 7.2 acres of liner area, and requireds the excavation of approximately 450,000 cubic yards (cy) of soil material and subsequent lining of the subgrade. The Phase B1 liner area unifies filling operations between Phase B to the south and Phase C1 to the north (see Figure 13). This area could not be lined previously with either Phase C1 or Phase B due to a delay in relocating the site's HHWCC located within B1. The HHWCC was recently relocated to the site entrance facility area and Phase B1 is intended to fill in the gap left between Phases B and C1, thereby, facilitating refuse fill operations in that area. The Phase B1 development area provides approximately 4.0 mcy of airspace capacity resulting in two years of life (based on 3,000 tons per day refuse inflow rate, 4:1 refuse-to-soil ratio and 307 operating days per year).

Phase C

The third phase (Phase C) will extend to the east of Phase A. The construction of the liner system in Phase C will likely occur in two stages due to the projected life of Phase C. Phase C will extend to the east of Phase A (see Figures 13) and is currently comprised of sub-phases C1 and C2. Due to delays in obtaining the landslide remediation and Phase B development project permits and approvals, an interim solution of developing Phase C1 to provide refuse airspace was previously necessary. Phase C1 was developed in July, 2002 ahead of Phase B, as it did not require any permit action other than CEQA compliance. Phase B was, therefore, developed after Phase C1. Phase C1 comprises 34 acres including 21 acres of lined area.

The Phase C2 liner area is proposed to provide additional airspace and time for IWMD excavation of the northerly cut slopes of Zone 1. The Phase C2 development area will provide up to 2.8 million cubic yards (mcy) of airspace capacity resulting in approximately 1.6 years of life (based on 3.000 tons per day refuse inflow rate, 4:1 refuse-to-soil ratio and 307 operating days per year). The Zone 1-Phase C2 development area will extend operations from Phase B and B1 to the east in an area previously occupied by a temporary desilting basin constructed during Phase B1/A2. The eastern edge of the Phase C2 excavation abuts the Phase B stockpile. The northern and western limits of Phase C2 will join the existing landfill subgrade elevations and will connect with the existing liner systems in Phases B and B1. Refuse filling in Phase C2 will also join the refuse fill gradients in Phases B and B1. The Phase C2 liner system is currently under construction (completion expected in October 2006).

Phase D

The fourth and final phase <u>of Zone 1</u> (Phase D) will extend to the east of Phases C-<u>B</u> and <u>BC</u>. The construction of the liner system in Phase D will also likely occur in two <u>stages-sub-phases</u> based on the projected <u>capacity and</u> life for Phase D.

Phases C and D will involve the excavation and stockpile of materials for daily cover in the west portion of Phase B and over previous fill areas (for Phase D). Filling in Phase D will reach the remaining final grades in Zone 1.

The Zone 1 final grades are based on MOUs with the Cities of San Clemente and San Juan Capistrano and a CUP with the City of San Juan Capistrano. Excavation of the first phase in Zone 4 will commence toward before the end of filling Phase D of Zone 1. In addition to the four <u>major</u> phases of liner development in Zone 1, the existing desilting basin at the toe of Zone 1 will be was enlarged in 2005 and improved <u>concrete lined</u>. and a <u>A</u> permanent LCRS facility is proposed was also completed south of Zone 1 as shown in Figure 4.

3.2.2 ZONE 4 - LANDFILLING

The current phasing scenario for Zone 4 proposes to begin operations in the northwestern corner (see Phase A in Figure 14<u>15) upon completion of Zone 1</u>. Operations and development of the landfill would then proceed in a counterclockwisewest to east direction, with a series of excavations (see Figure 15) and refuse fills (Phase A through I) until the final grade at 1,010 feet amsl (consistent with the San Clemente MOU) is reached. A more detailed phasing scenario will be developed prior to filling in Zone 4 based on geotechnical stability analyses for each incremental phase. Figure 4 presents the final grades of the completed landfill.

Excavated material in Phase A will be used for daily cover and compacted fills proposed for future phases in Zone 4. Excess excavation material from Phase A will can be stockpiled in Phases B and Fsubsequent phase areas. Phases B, C, D, E and F will not are not anticipated to require any soil stockpiling for their development. Once fill operations reach Phases G and H, soil material excavated to develop these phases will be stockpiled on the previously filled Phase D.

3.3 RECREATIONAL PLAN PHASING

The phasing and implementation of recreational uses of the site are constrained in that the landfilling activities and uses will always take precedence over the recreation and circulation improvements on the site and will always govern the timing of uses proposed to occur in areas that were formerly used for landfilling. In addition, the decision to proceed with an interim or ultimate recreational use must be supported by evidence that these uses will not impact ongoing landfilling operations on the site and that public health and safety can be protected. When the active disposal of solid waste on the site is completed, the ultimate recreational uses proposed to be implemented cannot interfere with or adversely affect long-term landfill management activities, including cover maintenance, landfill gas collection and disposal, leachate collection and recovery, groundwater well monitoring and other ongoing landfill maintenance and postclosure activities. Active recreational uses are proposed for Zones 1 and 4 with trail uses proposed in Zone 2 (including a trail crossing across Zone 5) as described below.

3.3.1 PHASING OF RECREATIONAL USES IN ZONE 1

Zone 1 is an area where current landfilling takes place. The area now being filled will gradually be moved eastward to include all of the 271 acres designated for <u>landfilling refuse</u> in Zone 1. In approximately <u>18-13</u> years, as of <u>January</u>, 2004<u>6</u>, or the year 2019 at the current GDP projected rate of disposal, Zone 1 would be completely filled. After closure activities have been completed, satisfactory access established, and sufficient settlement has occurred, the ultimate recreational uses as identified in a needs analysis could be implemented.

3.3.2 PHASING OF TRAIL USES

The PF&RDRDMD/HBP will coordinate the development of trails on the Prima Deshecha site with the development of trails outside the site proposed by the Cities of San Clemente and San Juan Capistrano. The County PF&RDRDMD/HBP's proposed Regional Riding and Hiking TrailsPrima Deshecha Trail traverses the Prima Deshecha site along the northeastern perimeter of the property around Zone 4 as shown on Figure 3. The County trail will connect directly with the Cristianitos, University of California and the Regional Riding and Hiking TrailsSan Juan Creek trails, which themselves provide connections to other trails in the area to the south and north, respectively. The off-site connections for the County trails are shown on Figure 3.

As discussed in Section 2.3.4, trails depicted along the perimeter of Zone 4 will be available as interim recreational use only during filling operations of Zone 1. <u>Ongoing</u> Ddiscussions commencing in January, 2001 with representatives of the Cities of San Clemente and San Juan Capistrano are focused on identifying specific trail alignments around Zone 1. Upon commencement of filling operations in Zone 4, these trails will be closed to the public for the protection of <u>public safety</u>. <u>PF&RDRDMD</u>/HBP may evaluate the possibility of relocating these trails away from landfill operations. This would depend upon their use and importance as a regional trail and the existence of a through trail along the western perimeter of the site at that time. Trail heads proposed directly west of the current entrance facilities will be developed after Zone 1 is closed and directly east of the current entrance facilities after Zone 4 is closed. On-site City trails around Zone 1 are shown on Figure 3 and can be used throughout the development of Zones 1 and 4. Also identified in Figure 3 are the off-site connections to City trails. The GDP proposes to eventually connect the County trail along Zone 4 with on-site City trails proposed along Zone 1 to provide a complete loop for trail users. These on-site trail connections would not occur until Zone 4 is closed (upon cessation of landfilling) and a trail crossing under future La Pata Avenue in Zone 5 is provided to the south.

3.3.3 PHASING OF RECREATIONAL USES IN ZONE 4

Zone 4 is the largest zone on the site, covering approximately 409 acres of refuse area. No interim uses are proposed for Zone 4, although trail uses are proposed during the operational life of Zone 1. After all landfilling operations have been completed in Zone 4, satisfactory access is established and sufficient settlement has occurred, the implementation of the ultimate recreational activities could begin. The ultimate recreational uses for Zone 4 include a wide range of possibilities, and at this time, the designated use for this zone is a regional park. Depending on the demand for recreational uses at the time of implementation, a needs analysis would be conducted and a park plan developed consistent with these demands and adjacent land uses in the area. In addition, biological mitigation and environmental enhancement actions within and around Zone 4 may result in a more passive recreational post-closure use. The park plan would be prepared concurrently with development of the final closure plan.

3.4 CIRCULATION PLAN PHASING

Two road fee programs were developed to provide funding for construction of La Pata Avenue. The first is the San Clemente Regional Circulation Financing and Phasing Program from which Avenida La Pata was recently deleted by the City of San Clemente. The second is the Avenida La Pata Supplementary Road Fee Program which was formulated by the Board of Supervisors on November 12, 1991, to provide an additional revenue of \$8.5 million for the construction of La Pata Avenue. The Area of Benefit of the latter program consists only of the unincorporated Talega Planned Community. The County has no immediate plans to construct this portion of the roadway; therefore, road fund availability is not being considered at this time. The County RDMD is undertaking a Feasibility Study and will subsequently prepare an Environmental Impact Report (EIR) for a preferred alignment of the La Pata Avenue Gap Closure project. The EIR is anticipated to be publicly distributed in 2006 and the road extension constructed by 2013. Construction of the extension of Camino de los Mares through the southwest corner of the site would be initiated by the Cities of San Clemente and/or San Juan Capistrano. Construction of any roadways through the site would be coordinated among the cities and the IWMD.

3.5 FACTORS AFFECTING PHASING

The implementation and phasing of the GDP landfilling, recreational uses and circulation may be affected by factors that could either change in the future or that are otherwise outside the control of the IWMD, <u>PF&RDRDMD</u>/HBP and other interested agencies and parties. Some of these potential factors are discussed in this section.

3.5.1 FACTORS AFFECTING LANDFILL PHASING

The phasing and staging of the landfilling uses under the GDP could be affected by the following factors:

- Increase or reduction in the rate of disposal could result in landfilling operations occurring for a shorter or longer period on the site.
- The capacity and life of the site could be increased or decreased based on <u>detailed geotechnical analysis and final design</u>, changes in landfill design standards and regulations, changes in daily cover use and final cover requirements, changes to the permitted refuse inflow rate and other similar changes.
- Increased recycling activities could result in a reduction in the volume of solid waste disposal in the landfill.

3.5.2 FACTORS AFFECTING CIRCULATION PHASING

The phasing of the circulation and roadway improvements under the GDP could be affected by the following factors:

- Timing and construction of other arterial or freeway improvements in the area.
- Availability of funding to construct the extension of La Pata Avenue through the site, to extend Camino de los Mares and Camino Las Ramblas, and to construct the widened segment of La Pata Avenue north of the site.

- Phasing of the recreational uses and other demand for the extension of La Pata Avenue through the site.
- Amendments to arterial highway extensions in the MPAH and changes to City general plan circulation elements.

3.5.3 FACTORS AFFECTING RECREATION PHASING

The phasing of the implementation of the recreational uses under the GDP could be affected by the following factors:

- Changes in the life of Zones 1 and 4 due to factors discussed in Section 3.5.1.
- Phasing of the on-site circulation improvements and the access roads to the individual recreational zones.
- Changes in the existing and future demand for recreational resources in south Orange County, including changes in the demand for the types of recreational uses proposed for the site.
- Receipt of a satisfactory proposal by a concessionaire to develop and operate a golf course on Zone 1, if that is the final selected use for the area.

3.6 SUBSEQUENT ACTIONS

Upon certification of <u>SEIR 575-597</u> and approval of <u>Amendment No. 2 to</u> the 2001 GDP, permit revision applications will be submitted to the LEA and SDRWQCB to reflect the new landfill plan. Also, an application for an amended Conditional Use Permit with the City of San Juan Capistrano will be submitted the following permits will be required in order to implement a biological premitigation program prior to future landfill impacts on biological resources:

- Individual 404 Permit from the U.S. Army Corps of Engineers including a Biological Opinion from the United States Fish and Wildlife Service.
- 401 Water Quality Certification Permit from the Regional Water Quality Control Board, San Diego Region.
- 1602 Streambed Alteration Agreement from the State Department of Fish and Game. This permit will not be required upon approval of the SNCCP/MSAA/HCP.
- 2080.1 Consistency Determination for Federally listed species. This determination will not be required upon approval of the SNCCP/MSAA/HCP.

The two primary support documents for the application are the Joint Technical Document and the Preliminary Closure and Post-Closure Maintenance Plans (PCPCMP). The Joint Technical Document includes:

... Disposal Site Improvements.

*Disposal Site Operations.

Disposal Site Controls.

»Disposal Site Design.

Disposal Site Characteristics.

The PCPCMP includes:

∗Final Cover Design.

Final Grading and Drainage Plans.

«Landfill Gas Control System Plans.

«Landscaping and Irrigation Plans.

«Site Security Plan.

*Closure Plan Implementation Schedule.

*Post-Closure Maintenance, Monitoring and Operational Procedures.

«Closure and Post-Closure Maintenance Costs.

Full implementation of the 2001 GDP landfill plan will also require the following landfill regulatory agency permit actions:

- Amended Waste Discharge Requirements from the Regional Water Quality Control Board, San Diego Region.
- Revised Solid Waste Facility Permit from the County of Orange Health Care
 Agency and concurred on by the California Integrated Waste Management
 Department.

- Permits to Construct/Operate Landfill Gas Control System facilities for ongoing operations from the South Coast Air Quality Management District.
- Annual Update to the General Permit to Discharge Stormwater Associated
 with Industrial Activity Water Quality Order No. 97-03-DWQ. Issued by the
 State Water Resources Control Board.

These landfill operation permit revisions need to be approved prior to operations in Zone 4.

Once revised<u>In addition to permits are issuedrevisions</u>, the site <u>will beis</u> subject to a permit review at least once every five years. In addition, the LEA inspects the site monthly, the SCAQMD quarterly, the SDRWQCB at least annually, and the CIWMB every 18 months for compliance with permit conditions and regulatory standards under each agency's jurisdiction.

Other subsequent actions for the site include:

- 404 Permit from the U.S. Army Corps of Engineers.

- 401 Permit from the Regional Water Quality Control Board, San Diego Region.
- 1601 Streambed Alteration Agreement from the State Department of Fish and Game.
- On-going CEQA Mitigation Monitoring.
- Needs Analysis and Plans for Interim and Ultimate Recreational Uses.
- <u>Liner Construction Phasing PlansDesign Reports</u> for <u>Landfilling Activitieseach</u> phase of development.
- Final Circulation Alignments and Element Permit Approvals, Design, and Improvements.
- Needs Analysis and Plans for Interim and Ultimate Recreational Uses.
- Construction of Interim and Ultimate Recreational Improvements.
- Preparation of Final Closure and Post-Closure Plans.

SECTION 4.0

4.0 LANDFILL REGULATIONS

The Prima Deshecha Landfill is a Class III landfill which is permitted for the disposal of non-hazardous municipal solid waste and digested sewage sludge (biosolids). State law requires that landfills operate under the authority of the CIWMB which exercises its authority through approval of a SWFP issued by a LEA. Organizationally, the LEA for Orange County landfills is a department within the County of Orange Health Care Agency, Environmental Health Division. The SDRWQCB also regulates landfill operations and design to ensure protection of surface and groundwater. The SDRWQCB exercises its authority through issuance of WDRs. The South Coast Air Quality Management District (SCAQMD) also regulates landfill operations related to landfill emissions control and monitoring and fugitive dust control.

The basis for the issuance of a SWFP is continuously reviewed during the life of the landfill and the permit may be modified, revised or revoked at any time. The system of landfill operation review (and imposed adjustment, as necessary) is accomplished through a reporting and monitoring procedure established by state statute. The basic elements of this reporting and monitoring system are:

- Joint Technical Document (JTD) (Title 27, California Code of Regulations [27 CCR] 21585 and 21590). The JTD is a document which fully describes the landfill site operations, design and future plans and is the primary regulatory support document for a SWFP and WDRs. The information contained in the JTD is used by the LEA and RWQCB to monitor compliance with a SWFP or WDR, respectively. In order to maintain a valid permit, the operator must <u>continuously updatekeep</u> the JTD<u>current</u>, through the filing of amendments to keep the information current.
- <u>Review of Permits (27 CCR 21675)</u>. The SWFP shall be reviewed and, if necessary, revised from the date of last issuance at least once every five years. A permit review report is prepared by the LEA to determine if any permit changes have occurred at the site which require <u>a permit that action to be taken by the operator</u>.
- <u>Recordkeeping Requirements (27 CCR 20515)</u>. The disposal site operator must maintain operational records which are open to inspection by the LEA and any other authorized regulatory or enforcement agency during normal business hours.
- <u>State Minimum Standards (27 CCR)</u>. Title 27 CCR minimum standards are administered by the CIWMB, LEA and the local RWQCB. These regulations are oriented toward refuse disposal operations and site design including provisions for odor and litter control, hazardous waste exclusion, protection of surface water and groundwater quality, landfill gas control, and closure and post-closure care.

Although the CIWMB has primary oversight and regulatory responsibilities for the site and has designated the County of Orange Environmental Health Care Agency, Environmental Health Division as its LEA, the site is regulated at other federal, state and local levels. The site must also comply with regulatory and administrative requirements set forth by the U.S. Environmental Protection Agency (USEPA), the USFWS, the United States Army Corps of Engineers (ACOEUSACE), the DFG, CEQA, the SDRWQCB, the SCAQMD, the Orange County Fire Authority, the County of Orange Resources and Development Management Department, City MOUs and Land Use Permits. The following are descriptions of these agencies and the regulations or requirements they are responsible for at the site.

FEDERAL LEVEL

United States Environmental Protection Agency

On October 9, 1991, the USEPA promulgated Subtitle D changes to the Resource Conservation and Recovery Act providing for nationwide minimum standards for landfilling municipal solid waste which became effective October 9, 1993. The regulations include requirements relating to daily cover, liners, landfill gas control, recordkeeping, groundwater monitoring, and closure and post-closure maintenance. After the USEPA approves a state plan, the regulations allow discretion on the part of state regulators to grant some flexibility to landfill operators in implementing Subtitle D regulations. California has been designated an "Approved State".

In addition, the USEPA is responsible for implementation of the Federal Clean Air Act (CAA). The CAA was first enacted in 1955 and has been amended numerous times in subsequent years (1963, 1965, 1967, 1970, 1977, 1990, and 1997). Under the authority granted by the CAA, USEPA has established National Ambient Air Quality Standards (NAAQS) for the following criteria pollutants: carbon monoxide (CO), lead (Pb), nitrogen dioxide (NO₂), ozone (O₃), particulate matter (PM10 and PM2.5), and sulfur dioxide (SO₂).

The CAA also specifies future dates for achieving compliance with the NAAQS and mandates that states submit and implement a State Implementation Plan (SIP) for local areas not meeting these standards. These plans must include pollution control measures that demonstrate how the standards will be met. The 1990 amendments to the CAA identify specific emission reduction goals for air basins not meeting the NAAQS. These amendments require both a demonstration of reasonable further progress toward attainment and incorporation of additional sanctions for failure to attain or meet interim milestones.

U.S. Army Corps of Engineers

The ACOEUSACE Regulatory Branch is responsible for assuring compliance with Section 404 of the Clean Water Act with respect to wetlands resources. Most activities involving wetland impacts require the approval of an individual 404 permit by the ACOEUSACE.

U.S. Fish and Wildlife Service

The USFWS reviews and comments on all federal actions that affect wetlands and waters of the United States, including all 404 permitting applications submitted to the ACOEUSACE to assure compliance with the Federal Endangered Species Act (FESA) which concerns activities that affect plant or animal species listed in the FESA. USFWS implements the FESA through various mechanisms: the Natural Community Conservation Plans, Interim Habitat Loss Mitigation Plans, and Habitat Conservation Plans.

STATE AND LOCAL LEVEL

California Air Act/California Air Resources Board (CARB)

The California Clean Air Act (CCAA), signed into law in 1988, requires all areas of the State to achieve and maintain the California Ambient Air Quality Standards (CAAQS) by the earliest practical date. The CAAQS are at least as stringent, and often more stringent, than the NAAQS.

The CARB has been granted jurisdiction over a number of air pollutant emission sources that operate in the State. Specifically, CARB has the authority to develop emission standards for on-road motor vehicles, as well as for stationary sources and some off-road mobile sources. In turn, CARB has granted authority to the regional air pollution control and air quality management districts to develop stationary source emission standards, issue air quality permits, and enforce permit conditions.

California Department of Fish and Game

Any activity that affects a river, stream, or lake may require a Lake or Streambed Alteration Agreement in accordance with the provisions of Sections 1600-1607 of the CDFG Code.

Section 2080.1 of the California Fish and Game Code authorizes the Director, California Department of Fish and Game to consider a request for incidental take of species listed as endangered by both the California Department of Fish and Game (CDFG) and the Department of the Interior for which an incidental take permit has been issued pursuant to Section 1539 of Title 16 of the United States Code authorizing the taking of an endangered species or a threatened species that is listed pursuant to Section 1533 of Title 16 of the United States Code that is an endangered species, threatened species, or candidate species pursuant to this chapter. It is IWMD's intent to obtain incidental take authorizations from the Department of the Interior through the provisions of Section 7 of the Federal Endangered Species Act in association with the U.S. Army Corps of Engineer's Section 404 permit required to address impacts to wetlands and Waters of the United States that are under the jurisdiction of this federal agency.

Once the Section 7 Consultation is completed between the USACE and the U.S. Fish and Wildlife Service (USFWS) and a biological opinion (BO) has been issued to the USACE by the USFWS, IWMD will prepare and transmit a request for a Consistency Determination with this BO to the Director, CDFG. This submittal will notify the Director in writing that IWMD has received an incidental take permit issued pursuant to the federal Endangered Species Act of 1973 (16 U.S.C.A. Sec. 1531 et seq.).

California Environmental Quality Act

County landfill projects are required to be in compliance with CEQA. When approving landfill projects, the County is required to adopt adequate environmental documentation in order to comply with this law.

South Coast Air Quality Management District

The SCAQMD has jurisdiction over an area of 10,743 square miles consisting of Orange County, the non-desert portions of Los Angeles, Riverside and San Bernardino counties, and the Riverside County portions of the Salton Sea Air Basin and Mojave Desert Air Basin. Southern California Air Basin (SCAB) is a subregion of the SCAQMD's jurisdiction, which covers an area of 6,745 square miles and includes all of Orange County and the nondesert portions of Los Angeles, Riverside, and San Bernardino counties. While air quality in this area has improved, the basin requires continued diligence to meet air quality standards.

The SCAQMD has adopted a series of Air Quality Management Plans (AQMPs) to meet the CAAQS and NAAQS. These plans require, among other emissions-reducing activities, control technology for existing sources; control programs for area sources and indirect sources; a permitting system designed to ensure no net increase in emissions from any new or modified permitted sources of emissions; transportation control measures; sufficient control strategies to achieve a five percent or more annual reduction in emissions (or 15 percent or more in a three-year period) for Reactive <u>Organic Compounds (ROC)</u>, 1 NO_X, CO, and PM10; and demonstration of compliance with the CARB's established reporting periods for compliance with air quality goals.

The 1997 AQMP, as amended in 1999, was approved by the USEPA in April 2000. The 1997/1999 AQMP represents the SIP for CO, O₃, NO₂, and PM10 that is currently applicable to sources in the SoCAB.

On August 1, 2003, the SCAQMD adopted a comprehensive update, the 2003 AQMP for the basin. The 2003 AQMP outlines the air pollution control measures needed to meet now superceded federal 1-hour standard for O_3 by 2010,² and to meet the federal PM₁₀ standard by 2006. It also demonstrates how the federal standard for CO, achieved for the first time at the end of 2002, will be maintained. Lastly, the plan takes a preliminary look at what will be needed to achieve new and more stringent health standards for O_3 and PM2.5. The 2003 AQMP was approved by CARB and submitted to USEPA for its final approval on January 9, 2004.

The SCAQMD also adopts rules to implement portions of the AQMP. Various rules apply to landfill operations, including landfill emissions control and monitoring, sulfur emissions monitoring, and fugitive dust control. The SCAQMD conducts periodic inspections of the site and, similar to the RWQCB, may impose civil liabilities for permit violations.

Certain stationary sources of air pollution at the Prima Deshecha Landfill require permits from the SCAQMD. Emission increases related to those sources may also be subject to SCAQMD Regulation XIII or Regulation XXX (Title V).

In addition to the AQMP and its rules and regulations, the SCAQMD published a handbook (*CEQA Air Quality Handbook*; most recent version: November 1993) that is intended to provide local governments with guidance for analyzing and mitigating project-specific air quality impacts for both land use and permitting projects. The Handbook provides standards, methodologies and procedures for conducting air quality analyses in EIRs.

The California Clean Air Act, Federal Clean Air Act and the Lewis-Presley Air Quality Management Act authorize the adoption of rules and regulations for air quality permits and govern the enforcement of those permits and rules. These acts are all administered and enforced by the SCAQMD. Various rules apply to landfill operations,

Reactive organic compounds (ROC) and volatile organic compounds (VOC) are designations made by CARB and USEPA, respectively, for organic compounds that can react with NO_X in the presence of sunlight to form O3. Slight variations exist between the two designations; for example, the CARB definition of ROC includes ethane while the USEPA definition of VOC does not.

² In 1997, the USEPA adopted a new 8-hour O3 NAAQS, and on June 15, 2005, the previous 1-hour O3 NAAQS was revoked.

including landfill emissions control and monitoring, sulfur emissions monitoring, and fugitive dust control. The SCAQMD conducts periodic inspections of the site and, similar to the RWQCB, may impose civil liabilities for permit violations.

<u>New Source Performance Standards/Emission Guidelines (NSPS/EG)</u>: On March 12, 1996, the USEPA promulgated standards of performance for new municipal solid waste landfills and emission guidelines for existing municipal solid waste landfills. These standards/guidelines for active landfills are intended to limit gaseous emissions to prevent public nuisance and possible detriment to public health caused by exposure to such emissions.</u>

Regulation XXX – Title V Permits: This regulation prohibits construction, modification, relocation, or operation of a Title V facility, or equipment located at a Title V facility, without first obtaining a Title V permit or permit revision that allows such construction, modification, relocation or operation. Title V is part of the 1990 Clean Air Act Amendments and consists of a single air permit, which consolidates and replaces all the previously issued air permits for a facility. The USEPA granted interim approval of the SCAQMD Title V program in February, 1997 and the program became effective March 31, 1997. The SCAQMD program is called Regulation XXX.

Rule 1402 – Control of Taxic Air Contaminants from Existing Sources: The purpose of this rule is to reduce the health risk associated with emissions of toxic air contaminants from existing sources by specifying limits for maximum individual cancer risk (MICR), cancer burden, and noncancer acute and chronic hazard index (HI) applicable to total facility emissions and by requiring facilities to implement risk reduction plans to achieve specified risk limits, as required by the Hot Spots Act and this rule. The rule also specifies public notification and inventory requirements.

Rule 1401 – New Source Review of Toxic Air Contaminants: This rule prohibits the air district from issuing an authority to construct or a permit to operate to any facility that would create an unacceptable public health risk from the emissions of toxic air contaminants. Unacceptable individual cancer risk from a permitted source is one change in a million. If Toxics-Best Available Control Technology (T-BACT) is employed, the allowable risk is increased to 10 in one million.

Regulation XIII – New Source Review: This regulation requires that Best Available Control Technology (BACT) be utilized to reduce pollutants from stationary sources.

<u>Ozone and Particulate Matter (PM_{10} and $PM_{2.5}$)</u>: The USEPA promulgated new clean air standards for ozone and fine particulates on July 18, 1997 based on the most recent medical studies regarding public health. The USEPA will be finalizing policies to implement the new air quality standards. In addition, the USEPA will require new

controls after the year 2002. The SCAQMD is soliciting preliminary public comments on possible control approaches and the best means to monitor the progress towards attainment of the new air quality standards.

<u>Rule 1150.1 – Landfill Gas Emissions</u>: The purpose of the current Rule 1150.1 for active and inactive landfills is to prevent public nuisance and possible detriment of public health caused by exposure to landfill gas emissions. The SCAQMD rewrote these rules to meet the federal NSPS/EG requirements.

<u>Rule 431.1 – Sulfur Emissions</u>: The purpose of this rule is to reduce sulfur oxides (SOx) emissions from the burning of gaseous fuels in stationary equipment and requires a permit to operate from the SCAQMD. The SCAQMD rewrote the Rule to raise the average daily limit of 40 parts per million (ppm) to 150 ppm effective June 12, 1998.

<u>Rule 403 – Fugitive Dust Emissions</u>: The purpose of this rule is to reduce the amount of particulate matter entrained in the ambient air as a result of man-made fugitive dust sources by requiring actions to prevent, reduce or mitigate fugitive dust emissions.

Rule 402 – Nuisance: This rule prohibits annoying odors from landfill operations.

San Diego Regional Water Quality Control Board

Under the Porter-Cologne Water Quality Act (California Water Code Section 13000 et. seq.) (Porter-Cologne), the County is required to report waste discharges that could affect water quality. Porter-Cologne is administered and enforced by the State of California Water Resources Control Board and Regional Water Quality Control Boards. The SDRWQCB regulates the Prima Deshecha Landfill.

<u>Waste Discharge Requirements</u>: Pursuant to Porter-Cologne, the RWQCBs issue WDRs containing terms and conditions of permitted discharges for landfills. The WDRs typically mandate a regular self-monitoring program to detect pollutants. In the event of a violation of a WDR, the RWQCB may issue either a cease and desist order or a cleanup and abatement order which mandate deadlines for remedial action. A landfill operator's failure to comply with a RWQCB order or reporting requirements may result in administrative or judicial civil liabilities ranging up to \$25,000 a day.

<u>National Pollutant Discharge Elimination System</u>: National Pollutant Discharge Elimination System (NPDES) is a federal program of the Clean Water Act, administered by the RWQCB which regulates non-point storm water pollution. In 1997, the USEPA reauthorized the program and the County filed new Notices of Intent in compliance with the program. <u>Solid Waste Assessment Testing</u>: Porter-Cologne also instituted the water quality Solid Waste Assessment Testing (SWAT) program, which requires an analysis of surface and groundwater under and within a one mile radius of a designated landfill for leakage of hazardous waste. If leakage outside of the landfill occurs, operators of the landfill must notify the State Department of Health Services and the CIWMB. These agencies may impose remedial action upon the landfill depending on the nature and extent of the release.

Orange County Fire Authority

The Orange County Fire Authority regulates the storage and use of flammable or combustible liquids and the adequacy of fire break roads at the site.

Orange County Resources and Development Management Department

The County of Orange RDMD regulates the construction of structures at the site and issues building, mechanical, electrical, and plumbing permits for certain landfill projects.

Cities of San Clemente and San Juan Capistrano

MOUs between the County of Orange and the City of San Juan Capistrano (approved in September, 1995) and between the County of Orange and the City of San Clemente (approved in July, 1997) have been issued which specify requirements, guidelines and conditions for the operations and development of the site.

The MOU with the City of San Juan Capistrano identifies mitigation agreed to by the County for impacts on the City due to the acceptance of out-of-County waste at the landfill. This MOU addresses road improvements, ridgeline preservation, sound/visual berms, park planning and traffic circulation as they relate to the General Plan Amendment for the site and the GDP.

The MOU with the City of San Clemente establishes GDP guidelines (including grading and height limits, zone boundary adjustments and landscape treatment), provides for the development of a joint feasibility study to determine appropriate flood control infrastructure for the Prima Deshecha Cañada watershed from the landfill to Interstate 5 and addresses certain water quality issues.

A CUP (No. 95-4) has also been issued by the City of San Juan Capistrano for that portion of the site within the City boundaries. The permit places conditions on the landfill use and landfill-related uses for Zone 1.
Rancho Mission Viejo, LLC (RMV)

The RMV agreements between the County and Rancho Mission Viejo, LLC place restrictions on the eastern 945 acres of the Prima Deshecha property including Zone 4. These restrictions are contained in a Settlement Agreement and Covenant and Declaration of Restrictions.

The Settlement Agreement contains requirements for County's use of restricted area on Prima Deshecha Landfill (PDL) property and RMV use of a no-build area adjacent to the PDL property on RMV lands.

The Covenant and Declaration of Restrictions contains restrictions on the landfill operations area and restricted area on PDL property as well as requirements for maintenance, dust and litter controls for La Pata Avenue.

The project described in the GDP as amended can be implemented under the requirements in the Settlement Agreement and Covenant & Declaration of Restrictions.

SECTION 5.0

5.0 TECHNICAL REFERENCES

- BonTerra Consulting, Screencheck Draft Supplemental Environmental Impact Report 597, Second Amendment to the 2001 Prima Deshecha General Development Plan, prepared for Orange County Integrated Waste Management Department (August 5, 2005)
- Bryan A. Stirrat & Associates, Inc., *Preliminary Closure Plan_Joint Technical Document Prima Deshecha Sanitary Landfill Volumes 1<u>1</u>, and 2<u>1</u><u>1</u>, and 1<u>11</u> prepared for County of Orange Integrated Waste Management Department (July, 1990, Revised February, 1993 November 2003, Amendment No. 1: April 2004, Amendment No. 2: May 2005).*
- Bryan A. Stirrat & Associates, Inc., *Prima Deshecha Cañada <u>Landfill Amended 2001</u> General Development Plan, prepared for County of Orange (August, 1994, Revised February, 1998January 2001 – Amended October 2002).*
- City of San Clemente, Planning Department, telephone conversation with Matt Everling, Planner, January 15, 1998 (714) 498-2533.
- City of San Juan Capistrano General Plan (December 18, 1974).
- Converse Consultants, *Geology and Soils Environmental Impact Report Prima* Deshecha Landfill – County of Orange, California, prepared for P & D Technologies (June 17, 1992).
- County of Orange, *Revised Draft Environmental Impact Report No. 548*, Prima Deshecha Landfill General Development Plan, September, 1995.
- GeoLogic Associates, *Alternative Liner Petition for the Zone 1 Expansion*, prepared for County of Orange, Integrated Waste Management Department (1997).
- <u>GeoLogic Associates, Analysis of Groundwater Resources in Zone 4, Prima Deshecha</u> <u>Landfill, Orange County, California, prepared for Bryan A. Stirrat & Associates,</u> (January 6, 2004).
- GeoLogic Associates, *Geotechnical Investigation Report Zone 1 Master Plan*, prepared for County of Orange, Integrated Waste Management Department (October, 1999).

<u>GeoLogic Associates, Geotechnical Investigation Report – Zone 4 Master Plan,</u> prepared for County of Orange, Integrated Waste Management Department (January, 2002).

Keeton Kreitzer Consulting, *Draft-<u>Final</u> Program Environmental Impact Report No.* 575, SCH No. 99041035, prepared for Orange County Integrated Waste Management Department (September 25, 2000)November 6, 2001)-

San Diego Regional Water Quality Control Board, *Water Quality Control Plan* for the San Diego Basin (September 8, 1994).

Wesnousky, S.R., *Earthquakes, Quaternary Faults, and Seismic Hazard in California*, in Journal of Geophysical Research, Vol. 91, No. B12, pp 12,57-12,631 (November 10, 1986).

TABLES

TABLE 1

PRIMA DESHECHA LANDFILL SUMMARY OF TOTAL AIRSPACE (AS OF JUNE 30, 2006)

Landfill Area	Airspace Filled ⁽¹⁾ (mcy)	Remaining Total Airspace ⁽¹⁾ (or Capacity) (mcy)	Total Airspace (or Capacity) (mcy) ^{(2) (3)}
Zone 1	26.8	26.3	53.1
Zone 4	2.3 ⁽⁴⁾	116.2 ⁽⁵⁾	118.5
Total	29.1	142.5	171.6

The following assumptions are the basis for Table 1 and are subject to change as operations progress:

mcy = million cubic yards.

- (1) Airspace filled based on IWMD Landfill Capacity Data As of June 30, 2006 table.
- (2) Total airspace for Zones 1 and 4 based on JTD Amendment No. 2, Appendix C, dated May 2005. Includes airspace available for refuse, daily/intermediate cover (assuming 4:1 refuse-to-soil ratio) and liner/protective cover/LCRS (0.4 mcy for Zone 1 and 5.2 mcy for Zone 4).
- (3) Final cover not accounted for in Total Airspace until landfill is within 5 years of closure. Natural settlement of landfill mass should create enough additional airspace to accommodate final cover volume.
- (4) Based on IWMD tonnage records for WMU2 (JTD Amendment No. 2, Appendix C, dated May 2005).
- (5) Based on JTD Amendment No. 2, Table 4, dated May 2005.

TABLE 2

PRIMA DESHECHA LANDFILL AMENDMENT NO. 2 TO 2001 GENERAL DEVELOPMENT PLAN SUMMARY OF REMAINING CAPACITY AND LIFE (AS OF JUNE 30, 2006)

Landfill Area	Remaining Total Airspace (or Capacity) (mcy) ⁽¹⁾	Remaining Refuse Capacity (mcy) ⁽²⁾	Remaining Refuse Tonnage (million tons) ⁽³⁾	Remaining Life (years)
Zone 1	26.3	20.7	13.8	13.4 ⁽⁴⁾
Zone 4 116.2		88.8	59.2	48.3 ⁽⁵⁾
Total	142.5	109.5	73.0	61.7

The following assumptions are the basis for Table 2 and are subject to change as operations progress:

mcy = million cubic yards

- ⁽¹⁾ See Table 1.
- ⁽²⁾ Conversion from net airspace (Total Airspace Liner/Protective Cover/LCRS) based on 4:1 refuse-to-soil ratio. Liner/Protective Cover/LCRS volume for Zone 1 = 0.4 mcy and Zone 4 = 5.2 mcy).
- ⁽³⁾ Assumes unit weight of waste in-place is 1,333 lbs/cy for Zones 1 and 4.
- ⁽⁴⁾ Assumes refuse inflow rate of 2,500 tpd from 6/30/06 to 12/31/08; 3,000 tpd from 1/1/09 to 12/31/13; and 4,000 tpd from 1/1/14, thereafter.
- ⁽⁵⁾ Assumes a daily refuse inflow rate of 4,000 tpd. Zone 4 life after closure of Zone 1.

FIGURES































(909) 860-7777

PRIMA DESHECHA LANDFILL

BRYAN A. STIRRAT & ASSOCIATES CIVIL AND ENVIRONMENTAL ENGINEERS 1360 VALLEY VISTA DRIVE DIAMOND BAR, CA 91765 GENERAL DEVELOPMENT PLAN FILL PHASING LIMITS - ZONE 1



LEGEND

620	

3:1	
-	

 PROPOSED DRAINAGE IMPROVEMENTS
 PROPOSED CONTOURS
 EXISTING CONTOURS
 DAYLIGHT LINE
 PROPERTY BOUNDARY
 EXISTING WATER COURSE
 DIRECTION AND RATE OF SLOPE
 PHASE LIMITS

FIGURE 13

	JOB NO.
	9645-91
	DATE
	9-2000
	DRAWN BY
	м.т.в.
	FILE NAME:
	17887DB.DWG





APPENDIX C

Memorandum of Understanding between the City of San Clemente, the Orange County Flood Control District and the County of Orange regarding the Prima Deshecha Landfill

and the

Memorandum of Understanding regarding the Prima Deshecha Landfill between the City of San Juan Capistrano and the County of Orange

MEMORANDUM OF UNDERSTANDING BETWEEN THE CITY OF SAN CLEMENTE THE ORANGE COUNTY FLOOD CONTROL DISTRICT AND THE COUNTY OF ORANGE

REGARDING THE PRIMA DESHECHA LANDFILL

THIS MEMORANDUM OF UNDERSTANDING (MOU) is entered into on this /At dav fully . 1997, between the County of Orange herein after referred to as "COUNTY." and the of Orange County Flood Control District, hereinafter referred to as "OCFCD," and the City of San Clemente, hereinafter referred to as the "CITY," through their respective legislative bodies.

RECITALS

The Prima Deshecha Landfill ("LANDFILL") is an active landfill owned and operated by the COUNTY, Integrated Waste Management Department (IWMD) as part of its landfill system.

COUNTY'S proposed General Development Plan (GDP) for the LANDFILL identifies two areas 17 for refuse disposal on the LANDFILL property. They are designated as Zone 1 and Zone 4 (see Exhibit 1). COUNTY currently conducts refuse fill operations in the first canyon in Zone 1 and plans to continue operations into the second canyon in Zone 1. Proposed phasing would allow for the 20 commencement of refuse filling in Zone 4, in the eastern portion of the LANDFILL, following the completion of refuse fill operations in Zone 1.

The LANDFILL has served since 1976 as the municipal solid waste disposal site for the City of 23 San Clemente. The LANDFILL is visible primarily from areas developed since opening of the 24 LANDFILL in 1976. The landfill will also be visible from areas planned for future development within 25 the CITY. CITY is concerned with the preservation of ridgeline views from these areas. COUNTY is 26 willing to balance the visual and other environmental concerns of CITY with COUNTY requirements 27 to: 1) protect air, water, and land; 2) maximize landfill capacity; and, 3) charge comparable rates. 28

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

18

19

21

22

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

1.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

A. Preferred GDP Guidelines

CITY and COUNTY agree that the following guidelines constitute design elements of a preferred alternative General Development Plan, and that COUNTY will present a project alternative including these guidelines, together with necessary environmental documentation pursuant to the California Environmental Quality Act, to the Orange County Board of Supervisors for consideration:

Grading and Height Limits

(a) The final grading elevation in Zone 1 to be at a height below the level of the ridgeline behind Zone 1 as viewed from Truman Benedict Elementary School located at 1251
 Sarmentoso, San Clemente, California 92673, as reflected in Exhibits 2-3, originals of which have been provided to the CITY.

(b) The final grading elevation of Zone 4 to be no higher than 1010 feet, thereby minimizing the visual impact to residents of CITY.

(c) It is understood that the side slopes of the LANDFILL may be steeper than what is reflected in EIR #548 as may be determined by COUNTY to be necessary to offset LANDFILL capacity lost due to the height limits described in this Paragraph A-1.

2. Zone Boundary Adjustments

(a) The boundaries of Zone 1 to be adjusted from the Zone 1 boundaries described in EIR #548, as reflected in Exhibit 2, with the understanding that additional boundary adjustments may be required for Zone 1 due to geotechnical conditions, drainage, and other environmental constraints provided such adjustment does not result in a final grading plan with a height limit greater than that specified in Paragraph A.1.(a).

(b) The boundaries of Zone 4 to be adjusted as may be determined by COUNTY to be necessary to offset LANDFILL capacity lost due to the height limits described in Paragraph A.1, provided that any additions to the LANDFILL resulting from the modification of the Zone 4 boundaries are not visible from within the CITY limits.

27 //

28 //

3. Landscape Treatment

1

2

3

4

5

6

7

8

9

10

11

(a) Trees to be planted to screen LANDFILL in a location which has been approved by CITY. The design for the tree scape is described in plans and specifications titled "Prima Deshecha Landfill Mass Excavation Grading Plans for Zone 1 - Phase A, approved by COUNTY's Board of Supervisors on May 13, 1997. A copy of the "Tree Planting Plan contained in those plans and specifications dated March 23, 1997 is contained in Exhibit 4. A full size copy of the landscaping plan sheet has been provided to CITY.

(b) Interim landscaping treatment to be provided consistent with erosion control measures required by current regulations.

(c) Final landscaping for Zone 1 to be installed according to regulations in effect at the time of final closure.

B. COUNTY and CITY Commitments in Event COUNTY Approves GDP Consistent With Preferred GDP Guidelines

If after completion of and consideration of environmental documentation. COUNTY approves 14 a GDP consistent with the preferred GDP guidelines set forth in Paragraph A of this MOU and the other 15 provisions of this MOU, CITY will not object to the GDP, to such subsequent actions of COUNTY as 16 may be necessary to implement the GDP (including without limitation the development of zone specific 17 plans for Zone 1 and 4, the master plan for the LANDFILL, the application for a Solid Waste Facility 18 Permit, and the final grading plan), or to such modification of EIR #548 as COUNTY may determine 19 to be necessary in order to adequately describe the environmental impacts of the project. In addition, 20 at such time that COUNTY approves a GDP consistent with the preferred GDP guidelines set forth in 21 Paragraph A of this MOU and the other provisions of this MOU, COUNTY commits that it shall comply 22 with such preferred GDP guidelines and the other provisions of this MOU in its future development and 23 use of the LANDFILL. Notwithstanding the foregoing, it is understood that CITY may continue to 24 comment on COUNTY's development of more specific grading and landscaping plans consistent with 25 the intent of this MOU. 26

- 27 //
- 28 //

1 C.

Flood Control/Drainage

COUNTY will design all on-site drainage features for the LANDFILL to meet 100-year
 flood control standards as contained in federal, state, and local laws and regulations.

CITY, COUNTY and OCFCD agree to cooperate in the development of a joint feasibility
 study to determine the appropriate flood control infrastructure for the Prima Deshecha Canyon
 Watershed (MO1) from the LANDFILL to the Interstate 5 Freeway.

3. CITY, COUNTY and OCFCD agree to equally share the cost of this study (one-third
each) with a not to exceed total of \$60.980. Any increase in this cost must receive written approval of
all parties.

4. CITY, COUNTY, and OCFCD agree to collaboratively explore alternatives for the appropriate flood control infrastructure for the MO1 as a result of the joint feasibility study described in Section C (2, 3).

13 D. Water Quality Issues

1. Potable Water

In the future, if COUNTY wishes to purchase potable water from CITY, and if CITY
wishes to sell potable water to COUNTY. COUNTY shall pay water connection fees at CITY's normal
rate. CITY shall process COUNTY'S application and take final action on the application in a timely
manner, in no event to exceed two (2) months after the date of request.

19

2.

3.

14

Reclaimed Water

In the future, if COUNTY wishes CITY to extend a reclaimed (non-domestic) water line to the LANDFILL, and if CITY wishes to complete the reclaimed water line construction. CITY will pay for all construction costs up to the LANDFILL boundary, and COUNTY will pay for any costs associated with on-site construction improvements, and CITY agrees to waive the connection fee for the reclaimed water line.

25

Leachate Collection and Control

COUNTY agrees to operate the LANDFILL in conformity with all applicable federal and
 state laws governing ground water protection including, but not limited to, those laws which prescribe
 liner, leachate collection, and disposal requirements for municipal solid waste landfills as determined

by the San Diego Regional Water Quality Control Board or such other regulatory agency exercising jurisdiction over the LANDFILL.

4.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

Bio-solids (Sewage Sludge)

 (a) COUNTY agrees to accept CITY's digested, dewatered, secondary treated biosolids at the LANDFILL in quantities permitted by the Solid Waste Facility Permit issued by the Local Enforcement Agency of the California Integrated Waste Management Board, and in accordance with applicable laws and regulations.

(b) During the LANDFILL five-year Solid Waste Facility Permit Review process the COUNTY agrees to consult with the Aliso Water Management Agency/South East Regional Reclamation Authority (AWMA/SERRA) to determine South County jurisdiction needs for disposal of digested, dewatered, secondary treated bio-solids and to accommodate the disposal of South County bio-solids in accordance with applicable laws and regulations.

E. Landfill Access

1. COUNTY agrees that Camino de los Mares will not be used for access to or from the LANDFILL for refuse disposal vehicles.

Any contract COUNTY enters into permitting haulers to utilize the LANDFILL for the
 disposal of solid waste originating outside of Orange County (Out-of-COUNTY Waste) will require the
 hauler to utilize a designated access route to the LANDFILL. Any such access route utilizing San
 Clemente streets will be approved in advance by the CITY. In no event will said access route include
 Camino de los Mares.

If any contracts for Out-of-County Waste provide for an access route over CITY streets.
 then COUNTY will pay CITY no less than eighty-one cents (\$.81) per ton ("Host Fee") for each ton of
 Out-of-County Waste conveyed over CITY streets and deposited in the LANDFILL for as long as such
 Out-of-County Waste continues to be deposited in the LANDFILL. The Host Fee is mitigation for
 LANDFILL access only. It is not compensation for acceptance of solid waste and will be paid only to
 the city through which the waste is transported.

27 Payment of the Host Fee will be made quarterly within one (1) month of the close of collection
28 for the quarter for all loads which are composed of Out-of-County Waste. A subsequent payment will

be made within twenty (20) days of receipt of origin of waste data from the hauler of any mixed loads composed of both Out-of-County Waste and In-County Waste. Mixed loads shall be prorated. Should the County's gate fee be changed after COUNTY begins providing disposal services for Out-of-County Waste, the Host Fee will be adjusted. The adjustment to the Host Fee will be made according to the following formula utilizing a floating decimal. The new adjusted Host Fee will be rounded to the nearest whole cent.

N = C x (A/B)

7

8

9

10

11

A = The new adjusted gate fee for Out-of-County Waste

B = The current gate fee for Out-of-County Waste

C = The current Host Fee for the CITY

N = The new Host Fee for the CITY

12 F. County Expenditure of Funds

To the extent this agreement obligates expenditures of COUNTY funds, other than OCFCD funds, those expenditures are limited to the County of Orange. Integrated Waste Management Department Enterprise Fund, which must be used only for LANDFILL related purposes.

16 G. Enforcement

COUNTY will comply with all applicable regulations, restrictions, and statutes at the
 Federal, State, and local level, as well as all provisions in this MOU.

Any discretionary actions by COUNTY set forth in this MOU which are not covered by 19 2. EIR #548 are subject to future California Environmental Quality Act (CEQA) compliance. By entering 20 into this MOU COUNTY does not commit itself to adopt any specific project except as set forth in 21 Paragraph B after the GDP is approved. If the GDP approved by COUNTY is inconsistent with the 22 terms of this MOU, CITY and COUNTY may terminate this MOU with all parties bearing any costs 23 incurred by them up to such date of termination. CITY reserves its rights to provide further input and 24 comments on any areas not specifically covered in this MOU including final landscaping and future 25 recreational uses for the site. 26

3. If the ownership of operating responsibilities of the LANDFILL are transferred or
assigned to any other entity or agency, public or private, COUNTY shall ensure that the obligations

identified in this MOU will be assigned to and assumed by the transferee/assignee so that the terms of
 this MOU shall continue to be met.

H. Notices

3

4

5

6

7

8

1. Any notice required or permitted by this MOU shall be in writing and sufficiently given if delivered in person or sent by certified or registered mail, postage prepaid, to the notice address of the respective parties set forth in this MOU. Changes in the respective addresses to which such notices may be directed may be made from time to time by any party by notice to the other party. The present addresses of the parties are:

9 COUNTY, IWMD: County of Orange Integrated Waste Management Department 10 Attention: Director, IWMD 11 12 Mailing Address: 320 North Flower Street. Suite 400 13 Santa Ana, CA 92703 14 15 OCFCD: Orange County Flood Control District 16 Attention: Director, Public Facilities and Resources Department 17 18 Mailing Address: P.O. Box 4048 19 300 North Flower Street 20 Santa Ana, CA 92702-4048 21 22 CITY: 23 City of San Clemente Attention: City Manager 24 25 Mailing Address: 100 Avenida Presidio 26 27 San Clemente, CA 92672 28 11

	1	
1		I.

3

4

5

6

7

8

9

10

11

12

13

14

15

16

19

23

24

25

Amendments

Amendments to this MOU may be considered at any time by mutual consent of CITY, OCFCD and COUNTY.

IN WITNESS WHEREOF, the parties hereto have executed this Memorandum of Understanding on the dates opposite their respective signatures:

COUNTY OF ORANGE, political subdivision of the State of California ORANGE COUNTY FLOOD CONTROL DISTRICT. A body corporate and politic in the State of California

JUL - 1 1997 Dated:

vielan G. Filen

Chairman of the Board of Supervisors of Orange County, California and Orange County Flood Control District

ATTEST edson for DARLENE J. BLOOM

TO THE CHAIRMAN OF THE BOARD

SIGNED AND CERTIFIED THAT A COPY

OF THIS DOCUMENT HAS BEEN DELIVERED

 Clerk of the Board of Supervisors of Orange County, California and
 Orange County Flood Control District

20 21 APPROVED AS TO FORM: LAURENCE M WATSON, COUNTY COUNSEL 22

____ Dated: _____ 25/47 Hunt, Deputy By: CONCUR:

COUNTY, IWMD 26 L. Will ____ Dated: _____/11/97 27 By: Vicki L. Wilson, Director

County of Orange, Integrated Waste Management Department 28

CONCUR: 1 ORANGE COUNTY FLOOD CONTROL DISTRICT 2 6/11/97 q DATE: 3 Public Facilities and Resources Department By: John W. Sibley, Director, 4 CITY OF SAN CLEMENTE 5 6 Dated: 7 By: Mayor 8 ATTEST: 9 Elinary Dated: 6/4/97 10 City Clerk 11 12 13 APPROVED AS TO FORM: 14 Dated: 6/4/97 15 Cht Attorney 16 17 18 19 20 21 22 23 24 25 26 27 28 9. REV. 5/25/97




LINE OF SIGHT STUDY CROSS-SECTION (LOOKING WEST)

PRIMA DESHECHA LANDFILL





MEMORANDUM OF UNDERSTANDING REGARDING THE PRIMA DESHECHA LANDFILL

between

THE CITY OF SAN JUAN CAPISTRANO

and

THE COUNTY OF ORANGE

TABLE OF CONTENTS

PREAMBLE
SECTION 1. GENERAL PLAN AMENDMENT/GENERAL DEVELOPMENT PLAN 1 A. Road Improvements 2 B. Ridgeline Preservation 3 C. Sound/Visual Berms 3 D. Park Use Financing Plan 3 E. Traffic Circulation 3
SECTION 2. SOLAG RELOCATION
SECTION 3. HOST FEE
SECTION 4. TONNAGE LIMITATIONS
SECTION 5. RECIPROCAL COOPERATION 5 A. Processing General Development Plan 5 B. Cooperation on Antonio Parkway 5 C. Cooperation on Solid Waste Facilities Permit 5
SECTION 6. STATUS OF CITY AND INDEMNIFICATION
SECTION 7. REPORTS AND INFORMATION
SECTION 8. BINDING EFFECT
SECTION 9. EXHIBITS INCORPORATED
SECTION 10. FORCE MAJEURE
SECTION 11. NOTICES
SECTION 12. AMENDMENTS
EXHIBIT A. MAP OF LANDFILL AREA
EXHIBIT B. PROJECT SCHEDULE

i

MEMORANDUM OF UNDERSTANDING BETWEEN THE CITY OF SAN JUAN CAPISTRANO AND THE COUNTY OF ORANGE REGARDING THE PRIMA DESHECHA LANDFILL

This Memorandum of Understanding is made and entered into this _____ day of September, 1995, by and between the County of Orange, hereinafter referred to as the "COUNTY" and the City of San Juan Capistrano, hereinafter referred to as the "CITY."

PREAMBLE

The Prima Deshecha Landfill "LANDFILL" is one of the three active landfills owned and operated by the COUNTY as part of its landfill system. By COUNTY Ordinance, COUNTY may contract to provide solid waste disposal services to governmental entities and/or solid waste haulers for municipal solid waste originating outside of the COUNTY ("Out-of-County Waste") provided said contracts meet certain conditions specified in Orange County Codified Ordinance ("OCCO") § 4-3-116. The COUNTY is considering entering into contracts to accept Out-of-County Waste at the LANDFILL. It is the intent of the COUNTY to participate in the mitigation of any potential impacts on the CITY.

The purpose of this Memorandum of Understanding is to identify the understanding of the parties as to how the COUNTY will reasonably mitigate these impacts on the CITY.

The COUNTY is currently under a Stipulated Order of Compliance and Agreement in operating the LANDFILL. The present Solid Waste Facility Permit (SWFP) was authorized at a time when the LANDFILL was operating at a level of approximately seven hundred and fifty (750) tons per day. The COUNTY is currently accepting approximately eleven hundred (1,100) tons per day of the municipal solid waste at the LANDFILL and is presently submitting a request for a revised SWFP for authorization to operate the LANDFILL at a proposed four thousand (4,000) tons per day. Any contracts entered into for disposal services of Out-of-County Waste shall not cause the tonnage to exceed four thousand (4,000) tons per day.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

SECTION 1. GENERAL PLAN AMENDMENT/GENERAL DEVELOPMENT PLAN.

CITY has received a payment of twenty thousand dollars (\$20,000) from COUNTY which represents the deposit for processing the COUNTY's application for General Plan Amendment. Any unused balance shall be refunded to COUNTY.

A. Road Improvements - If COUNTY begins accepting Out-of-County Waste at the LANDFILL, COUNTY shall be lead agency for design and construction of Antonio Parkway, a minimum four-lane roadway between Oso Parkway and Ortega Highway, by May 31, 1999. COUNTY shall be responsible for certification of an Environmental Impact Report ("EIR") by the Board of Supervisors, contribution of right-of-way from the property owners, setting aside sufficient funds for its design, construction, maintenance, drainage, slopes and environmental mitigation, including the cost of clearing any title obligations.

It is further agreed that in order to accept Out-of-County Waste at the LANDFILL that the following schedule of events must be met:

i. COUNTY Board of Supervisors shall set aside all funding for the design and construction of Antonio Parkway in accordance with the following schedule:

COUNTY shall accumulate gas tax funds over two (2) successive fiscal years beginning in 1995-96. COUNTY shall include with its adopting resolutions for its budget, a finding that the funding, design and expedited construction of Antonio Parkway between Oso Parkway and Ortega Highway is its highest priority capital project. In addition, this agreement, including this budget priority, shall be included as an executory contract to be assumed under any "Plan of Adjustment" to be filled by the Bankruptcy Court.

ii. COUNTY Board of Supervisors shall "Award Contract" for construction of Antonio Parkway by December 31, 1997.

iii. Start of construction to commence within 90 days of award of contract and be maintained as a continuous operation until completion, except for delays caused by acts of nature.

iv. Antonio Parkway to be completed not as a toll road and open for public use no later than May 31, 1999.

If the above stated schedule is not met, the COUNTY will cease acceptance of all Outof-County Waste at the LANDFILL until said milestones have been met. The COUNTY shall hold CITY harmless from any claims that may arise due to other contractual agreements that COUNTY has with separate third parties related to the construction of Antonio Parkway. CITY agrees not to interfere with any contracts between COUNTY and third parties. Changes to the above schedule may be discussed as part of a thirty (30) day meet and confer period prior to the expiration of the above dates. Any extension of the above performance dates can only be authorized upon agreement of both COUNTY and CITY. COUNTY will not submit an application for revision of SWFP No. 30-AB-0019 subsequent to the application for revision of SWFP No. 30-AB-0019 currently being reviewed by the staff of the Local Enforcement Agency ("LEA") and California Integrated Waste Management Board ("CIWMB") until a contract has been awarded for the construction of Antonio Parkway.

COUNTY agrees to incorporate the above construction schedule for Antonio Parkway as a mitigation measure in the EIR currently being prepared for the CITY's General Plan Amendment and the COUNTY's General Development Plan ("GDP") for the LANDFILL. Said mitigation measure shall be in full force and effect in accordance with the provision of the California Environmental Quality Act ("CEQA").

B. Ridgeline Preservation - The COUNTY will install and maintain final landfill grades which result in no silhouetting above and along the General Plan - designated "major ridgeline" which forms the northern and western edges of the LANDFILL site boundary such that no landfill operations or placement of landfill materials will visually encroach upon the designated General Plan ridgeline or be viewed from Ortega Highway.

C. Sound/Visual Berms - COUNTY agrees to construct an earthen berm per Exhibit A in order to buffer residential units from noise generated by vehicles travelling to LANDFILL on La Pata. Said berm shall be a minimum four to five (4 to 5) feet high, and nine hundred (900) feet long and be designed such that it may be constructed so as to not require the acquisition of any additional right-of-way by the COUNTY. Further, berm shall be designed to be accommodated by existing topography, without requiring retaining structures or significant grading. Construction should include hydroseeding or other method of erosion control. COUNTY agrees to repair berm as necessary. The berm is to be constructed prior to importation of Out-of-County Waste to LANDFILL in 1997.

D. Park Use Financing Plan - COUNTY agrees that the GDP will include a requirement that the ultimate recreational uses to be established on the site within CITY limits following closure of the landfill operations shall be mutually agreed upon by COUNTY and CITY. It is further agreed that the GDP will include a financial analysis and financial plan.

E. Traffic Circulation - Any agreement for Out-of-County Waste will require that Out-of-County Waste be delivered via I-5 Freeway to Ortega Highway to La Pata Avenue. Any other alternate route must be approved by the Director of Engineering and Building or his designee. CITY's approval shall not be unreasonably withheld.

Any agreement for Out-of-County Waste will require that Out-of-County Waste be delivered to the LANDFILL in transfer trailer vehicles, approximately twenty (20) tons of cargo, so as to reduce adverse traffic impacts unless approved in writing by City Manager or his designee.

SECTION 2. SOLAG RELOCATION.

When an application is submitted, COUNTY will expedite review of conditional use permits within ninety (90) days excepting appeals for the relocation of Solag Disposal Company to property owned by the San Juan Company, a California limited partnership, and San Juan Partnership No. I, a California limited partnership, or any property whereby COUNTY has jurisdiction over the land use permit process or construction permit process, and COUNTY will not unreasonably withhold permits for such relocation.

SECTION 3. HOST FEE.

The COUNTY will pay the CITY no less than eighty-one cents (\$.81) per ton for each ton of Out-of-County Waste deposited in the LANDFILL, regardless of whatever tip fee is finally negotiated, for as long as such Out-of-County Waste continues to be deposited in the LANDFILL ("Host Fee"). Said Host Fee is mitigation for LANDFILL access only and is not compensation for acceptance of solid waste.

Payment of the Host Fee will be made quarterly within one (1) month of the close of collection for the quarter for all loads which are composed of Out-of-County Waste. A subsequent payment will be made within twenty (20) days of receipt of origin of waste data from the hauler of any mixed loads composed of both Out-of-County Waste and In-County Waste. Mixed loads shall be prorated. Should the COUNTY's gate fee be increased after the COUNTY begins providing disposal services for Out-of-County Waste, the Host Fee will be adjusted. The adjustment to the Host Fee will be made according to the following formula utilizing a floating decimal. The new adjusted Host Fee will be rounded to the nearest whole cent.

N = C x (A/B)

A = The new adjusted gate fee for Out-of-County Waste

B = The current gate fee for Out-of-County Waste

C = The current Host Fee for the CITY

N = The new Host Fee for the CITY

SECTION 4. TONNAGE LIMITATIONS.

Landfill operation will be limited to a maximum of four thousand (4,000) tons per day of municipal solid waste for the length of time the LANDFILL is in operation.

Any SWFP issued by the State of California or LEA shall specifically stipulate a maximum daily tonnage limitation of four thousand (4,000) tons per day and an annual tonnage limitation (based

on three hundred and seven [307] operating days per year) of one million, two hundred and twentyeight thousand (1,228,000) tons of municipal solid waste.

Importation shall begin January of 1997, and shall not exceed two thousand (2,000) tons per day for any day prior to completion of Antonio Parkway. Antonio Parkway shall not be constructed as a toll road. If construction of Antonio Parkway is not completed by May 31, 1999, COUNTY will discontinue use of LANDFILL for the disposal of Out-of-County Waste, and shall cease all LANDFILL operation above two thousand (2,000) tons per day of municipal solid waste of any kind, including In-County Waste. Said limitations shall remain in effect until Antonio Parkway is completed and open to public traffic.

SECTION 5. RECIPROCAL COOPERATION.

A. Processing General Development Plan - The CITY will proceed with processing the application submitted by the COUNTY for an amendment to its General Plan reflecting the interim landfill use of the site occupied by the LANDFILL and will present that amendment to the City Council no later than November 21, 1995. In order to meet this obligation, the COUNTY shall submit a draft EIR to the CITY no later than September 22, 1995. The CITY and COUNTY agree to the schedule and milestones for processing the General Plan Amendment contained in Exhibit B, attached and incorporated by this reference excepting appeals. If CITY fails to meet its milestones for processing General Plan Amendments, CITY shall not object to COUNTY obtaining extensions of Senate Bill (SB) 17 (Craven)(Stats. 1995, Chapter 4 uXX).

B. Cooperation on Antonio Parkway - CITY shall cooperate with COUNTY in providing necessary information and other actions required by the COUNTY in the preparation and certification of the EIR for both Antonio Parkway and the LANDFILL GDP and process any necessary permits, right of entry agreement, and applications involved in the funding and construction of the road.

<u>C.</u> Cooperation on Solid Waste Facilities Permit. The CITY will cooperate with the COUNTY by providing the COUNTY with any necessary documents or clearances required by the LEA or the CIWMB so as to enable the COUNTY to complete its pending permit revision application, including, but not limited to information required by Public Resources Code Section 50000.5. Once CITY has acted on the General Plan Amendment, CITY shall provide LEA and CIWMB with records of action.

SECTION 6. STATUS OF CITY AND INDEMNIFICATION.

The CITY and COUNTY desire to leave no doubt as to their respective roles, and that by entering into this Memorandum of Understanding, CITY is not thereby becoming an "Arranger" as that term is used in CERCLA § 107 (a)(3), and that it is the COUNTY, not the CITY, which is "Arranging For" the disposal of Out-of-County Waste in LANDFILL which may contain hazardous, toxic, harmful or corrosive substances.

COUNTY agrees to defend, indemnify and hold CITY, and its officers, employees and agents harmless for all claims, injury, death or damage to property resulting solely from the acts or omissions of the COUNTY, its officers, employees, agents for liability arising from any activity of the COUNTY, its officers, employees, agents taken pursuant to the provisions of this Memorandum of Understanding including liability arising due to delivery of Out-of-County Waste to the LANDFILL. CITY agrees to defend, indemnify and hold COUNTY, and its officers, employees and agents harmless for all claims, injury, death or damage to property resulting solely from the acts or omissions of the CITY, its officers, employees, agents for liability arising from any activity of the CITY, its officers, employees and agents taken pursuant to the provisions of this Memorandum of Understanding. If the CITY's negligence combines with the COUNTY's negligence to cause injury, the parties agree that liability will be apportioned as determined by a court of competent jurisdiction. Neither party shall request a jury apportionment.

SECTION 7. REPORTS AND INFORMATION.

The CITY reserves the right to inspect the records of the COUNTY and the right to audit and to recompute any amounts deemed payable under this Memorandum of Understanding. Any such audit must be at CITY expense and take place within thirty-six (36) months of the close of each of the fiscal years. Any additional amount due must be paid within thirty (30) days following written notice to the COUNTY by the CITY. This notice shall include a copy of the audit report.

SECTION 8. BINDING EFFECT.

Each covenant, obligation, and condition contained in this Memorandum of Understanding shall inure to the benefit of and be binding upon the parties to this Memorandum of Understanding. If LANDFILL is transferred by COUNTY, COUNTY will provide in any such agreement that subsequent operator will comply with all requirements of this Memorandum of Understanding, which apply to the operation of the LANDFILL.

SECTION 9. EXHIBIT INCORPORATED.

Exhibits "A" through "B" are attached to and incorporated in this Memorandum of Understanding by reference.

SECTION 10. FORCE MAJEURE.

Should the performance of the obligations of either party under this Memorandum of Understanding be interrupted or delayed by any occurrence not occasioned by the conduct of either party to this Memorandum of Understanding, whether that occurrence is an act of God, war, civil insurrection, fire, flood, storm, strikes, lockouts, or by any law, regulation, or order of any federal or state court, that party's performance under this Memorandum of Understanding shall be excused for whatever period of time after the occurrence is reasonably necessary to remedy the effects of that occurrence.

SECTION 11. NOTICES.

Any notice required or permitted by this Memorandum of Understanding shall be in writing and sufficiently given if delivered in person or sent by certified or registered mail, postage prepaid, to the notice address of the respective parties set forth in this Memorandum of Understanding. Changes in the respective addresses to which such notices may be directed may be made from time to time by any party by notice to the other party. The present addresses of the parties are:

COUNTY: County of Orange

EMA/Integrated Waste Management Department Attn: Director

Location for Direct Deliveries and Certified Mail: 320 North Flower Street, Suite 400 Santa Ana, CA 92703

Mailing Address: P.O. Box 4048 Santa Ana, CA 92702-4048

CITY : City of San Juan Capistrano Attn: City Manager

> Mailing Address: 32400 Paseo Adelanto San Juan Capistrano, CA 92675

SECTION 12. AMENDMENTS.

Neither this Memorandum of Understanding nor any provision hereof may be changed, modified, amended or waived except by written agreement duly authorized and executed by both parties.

IN WITNESS WHEREOF, the parties hereto have executed this Memorandum of Understanding on the dates opposite their respective signatures:

COUNTY OF ORANGE, a political subdivision of the State of California

DATED: 9 - 12 - 95

By: C Gaddi Vasquez, Chairman of its Board of Supervisors

SIGNED AND CERTIFIED THAT A COPY OF THIS DOCUMENT HAS BEEN DELIVERED TO THE CHAIRMAN OF THE BOARD

athlen E. Groons

Kathleen E. Goodno, Acting Clerk of the Board of Supervisors of Orange County, California

APPROVED AS TO FORM: LAURENCE M. WATSON CHIEF ASSISTANT COUNTY COUNSEL

DATED: Geoffrey K. Hunt, Deputy

,

CITY OF SAN JUAN CAPISTRANO

DATED <u>September 19, 199</u>5

By: Carolyn Nash

Carolyn Nash, Mayor

ATTEST: Cheryl Johnson, Citv

APPROVED AS TO FORM.* DATED: _ Richard Denhalter, City Attorney of the

City of San Juan Capistrano, California



EXHIBIT B

PRIMA DESHECHA LANDFILL MOU & EIR 548 PROJECT SCHEDULE

SEPTEMBER 5, 1995

Α. Prepare Revised Draft EIR -August 24 through September 15, 1995 Β. City Council Execute MOU -September 8, 1995 Board of Supervisors Execute MOU -September 12, 1995 С. Print Revised Draft EIR -September 15 through September 17, 1995 D. Distribution of Revised Draft EIR -September 18, 1995 E. 45-Day Public Review Period¹ -September 19 through November 2, 1995 F. Prepare Response to Comments -November 2 through November 7, 1995 G. City/County Commission(s) -Week of November 13th H. City Council/Certifies EIR/Adopts GPA -November 21, 1995 B/S Certifies EIR -November 21, 1995 (Special Meeting needed) I. Permit Approval by CIWMB -December 1995

¹ The following items will begin during this period: staff reports, response to comments received early, mitigation monitoring and reporting program, findings of fact and statements of overriding consideration.

SECTION 11. NOTICES.

Any notice required or permitted by this Memorandum of Understanding shall be in writing and sufficiently given if delivered in person or sent by certified or registered mail, postage prepaid, to the notice address of the respective parties set forth in this Memorandum of Understanding. Changes in the respective addresses to which such notices may be directed may be made from time to time by any party by notice to the other party. The present addresses of the parties are:

COUNTY: County of Orange EMA/Integrated Waste Management Department Attn: Director

> Location for Direct Deliveries and Certified Mail: 320 North Flower Street, Suite 400 Santa Ana, CA 92703

Mailing Address: P.O. Box 4048 Santa Ana, CA 92702-4048

CITY : City of San Juan Capistrano Attn: City Manager

> Mailing Address: 32400 Paseo Adelanto San Juan Capistrano, CA 92675

SECTION 12. AMENDMENTS.

Neither this Memorandum of Understanding nor any provision hereof may be changed, modified, amended or waived except by written agreement duly authorized and executed by both parties.

IN WITNESS WHEREOF, the parties hereto have executed this Memorandum of Understanding on the dates opposite their respective signatures:

> COUNTY OF ORANGE, a political subdivision of the State of California

DATED: 9 - 12 - 95

By: C Gaddi Vasquez, Chairman of its Board of Supervisors

SIGNED AND CERTIFIED THAT A COPY OF THIS DOCUMENT HAS BEEN DELIVERED TO THE CHAIRMAN OF THE BOARD

athleen E Groons

Kathleen E. Goodno, Acting Clerk of the Board of Supervisors of Orange County, California

APPROVED AS TO FORM: LAURENCE M. WATSON CHIEF ASSISTANT COUNTY COUNSEL DATED:

Geoffrey K. Hunt, Deputy

CITY OF SAN JUAN CAPISTRANO

DATED: Septender 19, 1945

By: Carolyn Nash, Mayor

ATTEST:

Cheryl Johnson, Cit

APPROVED AS TO FORM:* DATED: Ridhard Denhalter, City Attorney of the

City of San Juan Capistrano, California

FIRST AMENDMENT TO MEMORANDUM OF UNDERSTANDING BETWEEN THE CITY OF SAN JUAN CAPISTRANO AND THE COUNTY OF ORANGE REGARDING THE PRIMA DESHECHA LANDFILL

This First amendment is made and entered into this $\frac{2}{2}$ day of $\frac{2}{2}$. 1995, by and between the County of Orange, hereinafter referred to as "COUNTY", and the City of San Juan Capistrano, hereinafter referred to as "CITY", and is made to the Mernorandum of Understanding between the parties dated September 12, 1995, hereafter collectively referred to as the MOU.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

1. The Sound/Visual Berms Section, is amended by deleting Paragraph. Section 1. C. Sound/Visual Berms in its entirety, and substituting the following:

Paragraph I. C. - Sound/Visual Berm

CITY and COUNTY agree to cooperate in the construction of an earthen berm per Exhibit A in order to buffer residential units from noise and view of vehicles traveling to LANDFILL on La Pata. Said berm shall be approximately eight (8) feet high, and nine hundred (900) feet long. Berm shall be designed in a manner to minimize the need for extra right-of-way and to accommodate existing topography as much as possible. CITY and COUNTY shall agree on technical specifications of berm design prior to its construction. Should additional right-of-way be needed, CITY and COUNTY agree to cooperate in negotiations with appropriate landowner (s).

If the California Integrated Waste Management Board receives a letter from the City of San Juan Capistrano by November 27, 1995, which satisfies the General Plan Consistency requirements of Public Resources Code Section 50000, COUNTY shall pay total cost of said berm, including but not limited to, engineering and design, right of way acquisition and construction costs. Construction shall include hydroseeding or similar method of erosion control. COUNTY agrees to repair berm as necessary. Construction shall be completed within one (1) year of permit issuance. If General Plan Consistency finding persuant to Section 50000 is not sent by City to the CIWMB by November 27, 1995, COUNTY's share of total cost shall be \$45,000 and construction shall begin at a date mutually agreed to by CITY and COUNTY.

All other terms and conditions of the agreement remain unchanged.

IN WITNESS WHEREOF, the parties hereto have executed this First Amendment to the Memorandum of Understanding on the date's opposite their respective signatures:

COUNTY OF ORANGE, a political subdivision of the State of California

By:

DATED: NOV 2 1 1995

Roger Stanton. Chairman of the Board of Supervisors

SIGNED AND CERTIFIED THAT A COPY OF THIS DOCUMENT HAS BEEN DELIVERED TO THE CHAIRMAN OF THE BOARD

(xalin E Cootis

Kathleen E. Goodno. Acting Clerk of the Board of Supervisors of Orange County, California

APPROVED AS TO FORM: LAURENCE My WATSON. CHIEF ASSISTANT COUNTY COUNSEL

DATED: 11/22/25

Georfrey R. Hunt, Deputy

.:

CITY OF SAN JUAN CAPISTRANO

DATED: January 2, 1996

By: Uliciti Hart Wyatt Hart, Mayor

ATTEST:

ervi Johnson.

APPROVED AS TO FORM:

Richard Denhalter, City Attorney of the City of San Juan Capistrano, California

DATED: 14195



Attachment A

APPENDIX D

Amendment No. 1 to the GDP

ORANGE COUNTY BOARD OF SUPERVISORS MINUTE ORDER October 22, 2002

1140044400

T ULTUL

F 11 P

Submitting Agency/Department: INTEGRATED WASTE MANAGEMENT DEPARTMENT

Approve settlement agreement and covenant & declaration of restrictions; approve amended 2001 Prima Deshecha General Development Plan; and make California Environmental Quality Act and other findings - District 5

The following is action taken by the Board of Supervisors: APPROVED AS RECOMMENDED I OTHER I

, I WHI WEBLICK OF THE DURING

Unanimous [] (1) SMITH: X (2) SILVA: Y (3) SPITZER: Y (4) COAD: X (5) WILSON: Y

Vote Key: Y=Yes; N=No; A=Abstain; X=Excused; B.O.=Board Order

Documents accompanying this matter:

Resolution(s)Ordinances(s)

.....

Contract(s)

Item No. 14

Special Notes:

Copies sent to:

CEO

IWMD – Janice V. Goss Barbara Love Chris Cruz

Auditor

11/01/02



I certify that the foregoing is a true and correct copy of the Minute Order adopted by the Board of Supervisors, Orange County, State of California, DARLENE J. BLOOM, Clerk of the Board

By:__ Deputy

l

OFTIMAL OLERA OF THE BOARD OF MELE COUNTY

SETTLEMENT AGREEMENT

This Settlement Agreement ("Agreement") is made effective as of this 2 day of 2002 ("Effective Date"), by and between the County of Orange ("County") and DMB San Juan Investment North, LLC, a Delaware limited liability company ("DMBSJIN"), San Juan Partnership No. I, a California limited partnership ("SJP I"), and San Juan Partnership No. IV, a California limited partnership ("SJP IV"). DMBSJIN, SJP I and SJP IV are referred to collectively herein as "RMV." County and RMV are referred to collectively as the "Parties" and individually as a "Party."

RECITALS

A. The 1,530 acre Prima Deshecha Landfill site ("PDL") is located in south Orange County and is operated by County's Integrated Waste Management Department ("IWMD").

B. RMV owns certain real property situated adjacent to the eastern and northern boundaries of the PDL, which is part of a larger block of land commonly referred to as "Rancho Mission Viejo." Rancho Mission Viejo is comprised of approximately 25,000 contiguous acres of currently undeveloped land that RMV and its affiliated entities continue to own in south Orange County.

C. In 1979, the County approved the first General Development Plan ("1979 GDP") for the PDL. The 1979 GDP authorized the County to engage in certain landfilling and ancillary activities on approximately 1,000 acres of the PDL.

D. In 2001, the County proposed a new General Development Plan ("2001 GDP") to replace the 1979 GDP. The 2001 GDP divides the PDL into five zones and designates certain uses for each zone. Specifically, Zones 1 and 4 are designated for landfilling, Zone 5 is

designated for regional circulation, and Zones 2 and 3 are designated for recreational trails/open space. In addition, the 2001 GDP anticipates that Zones 1 and 4 will be converted into a regional park and/or used for recreational purposes, once these areas have reached landfill capacity and landfilling activities have ceased.

E. In conjunction with the 2001 GDP, the County prepared draft Environmental Impact Report No. 575 ("DEIR No. 575"), which assessed the potential environmental impacts associated with the new development plan. DEIR No. 575 was circulated for public review and comment on February 5, 2001.

F. Following the close of the written comment period, the County prepared final Environmental Impact Report No. 575 ("FEIR No. 575") and held two public hearings to solicit additional public input concerning the adequacy of FEIR No. 575 and the proposed adoption of the 2001 GDP. The first hearing was held before the Orange County Planning Commission on October 24, 2001; the second hearing was held before the Orange County Board of Supervisors on November 6, 2001. At the close of the second hearing, the Board of Supervisors passed and adopted Resolution No. 01-380 ("Resolution No. 01-380," a copy of which is attached hereto as Attachment 5) whereby the Board of Supervisors (i) certified FEIR No. 575 as having been completed in accordance with the provisions of the California Environmental Quality Act ("CEQA") and (ii) approved and adopted the 2001 GDP. As used herein, "EIR No. 575" shall mean FEIR No. 575, together with any mitigation measures, amendments, or conditions required in connection therewith, as certified by the Board of Supervisors on November 6, 2001.

G. During the written comment period and at both public hearings, RMV submitted comments in which it asserted that FEIR No. 575 was inadequate and objected to approval of the 2001 GDP. Among other things, RMV asserted that FEIR No. 575 failed to adequately consider

the potential impacts of the 2001 GDP on RMV's anticipated future use of its property adjacent to the PDL.

H. Subsequent to approval of the 2001 GDP by the Board of Supervisors, RMV also asserted that the new development plan was inconsistent with the terms and conditions of the Prior Agreement (defined below).

I. There currently is a controversy and dispute between the County and RMV concerning (i) the adequacy of EIR No. 575, (ii) the impact of the 2001 GDP on RMV's anticipated future use of its adjacent property, and (iii) the consistency of the 2001 GDP with the Prior Agreement. As set forth more fully below, the Parties desire to fully settle, compromise, and resolve this controversy and dispute, as well as any and all related issues and concerns.

AGREEMENT

NOW, THEREFORE, in consideration of the mutual terms, covenants, conditions, promises, and benefits contained herein, and for other good and valuable consideration, the Parties agree as follows:

1. **Definitions.** For the purposes of this Agreement, the following terms shall have the meanings set forth below:

1.1. <u>Burdened Property</u>. The term "Burdened Property" shall mean the portion of the PDL defined as the Burdened Property in the County Restrictive Covenant (defined below) and more particularly described and depicted on Exhibits "A" and "B" of the County Restrictive Covenant.

1.2. <u>County Restrictive Covenant</u>. The term "County Restrictive Covenant" shall mean the Covenant and Declaration of Restrictions entered into concurrently herewith

pursuant to Section 3 hereof between the Parties in the form of <u>Attachment "1</u>" to this Agreement.

1.3. <u>County's Successors-in-Interest</u>. The term "County's Successors-in-Interest" shall mean any person or entity that acquires all or a portion of the Burdened Property.

1.4. Landfill Operations. The term "Landfill Operations" shall mean: (i) the storing, collecting, transporting, treating, processing, recycling, reusing, converting, disposing, transferring, placing, burying, relocating or otherwise handling in any manner whatsoever any solid, semi-solid, liquid or gas waste of any type; (ii) all activities and operations related to closure and/or post-closure of the landfill; (iii) the construction, use, operation, maintenance, repair and/or replacement of buildings, structures, improvements, environmental monitoring and control systems, property and equipment in connection with the activities and operations identified in subsections (i) and (ii) above; (iv) all site grading, cutting, filling and other earth movement work performed in connection with the activities and operations identified in subsections (i), (ii) or (iii) above; and (v) the construction, use, operation, maintenance, repair and/or replacement of roads, paths, trails and accessways (whether temporary or permanent) used in connection with the activities and operations (i), (ii), (iii) and (iv) above.

1.5. <u>Landfill Operations Area</u>. The term "Landfill Operations Area" shall mean that portion of the Burdened Property defined as the Landfill Operations Area in the County Restrictive Covenant and more particularly delineated on Exhibit "B" of the County Restrictive Covenant.

1.6. <u>No-Build Area</u>. The term "No-Build Area" shall mean that portion of the RMV Benefitted Property more particularly delineated on <u>Attachment "2</u>" to this Agreement.

1.7. <u>Prior Agreement</u>. The term "Prior Agreement" shall mean the Purchase and Option Agreement executed between the County and RMV's predecessors-in-interest on June 13, 1973, a copy which is attached to this Agreement as <u>Attachment "3</u>," pursuant to which the County acquired the Burdened Property from RMV's predecessors-in-interest.

1.8. <u>Restricted Area</u>. The term "Restricted Area" shall mean that portion of the Burdened Property defined as the Restricted Area in the County Restrictive Covenant and more particularly delineated on Exhibit "B" of the County Restrictive Covenant.

1.9. <u>RMV Benefitted Property</u>. The term "RMV Benefitted Property" shall mean that real property adjacent to the eastern and northern boundaries of the Burdened Property and more particularly described and depicted on <u>Attachment "4</u>" to this Agreement.

1.10. <u>RMV's Successors-in-Interest</u>. The term "RMV's Successors-in-Interest" shall mean: (i) any person or entity that acquires ten percent (10%) or more of the assets of DMBSJIN, SJP I, or SJP IV, provided that such assets include all or a portion of the Benefitted Property; (ii) any division, subsidiary, operating company or wholly-owned entity of DMBSJIN, SJP I, or SJP IV; (iii) any person or entity that acquires ten percent (10%) or more of DMBSJIN's, SJP I's, or SJP IV's interest in the Benefitted Property; and (iv) lineal descendents of Marguerite O'Neill (and/or each of their estate planning trusts or entities) provided such persons acquire fee interests in all or a portion of the Benefitted Property. Except for any person expressly identified in the preceding sentence, RMV's Successors-in-Interest shall not include any individual consumer or purchaser of one or more residential units within the RMV Benefitted Property or any association (including any homeowners' association), as such term is defined in California Civil Code Section 1351.

2. <u>Amendment of Prior Agreement</u>. As of the Effective Date of this Agreement, the provisions of Section I.C.5 of the Prior Agreement shall no longer be of any force and effect, it being the intention of the Parties hereto that this Agreement shall supercede those provisions of the Prior Agreement. Notwithstanding the preceding sentence, the Parties agree that in the event this Agreement is held by a court of competent jurisdiction to be invalid, void or unenforceable, then upon recordation of a certified copy of such holding (or a short form memorandum of the same) in the Official Records of County by either Party or by the court (provided such recorded document shall refer to the title, instrument number and recording date of the County Restrictive Covenant): (i) the County Restrictive Covenant shall automatically become invalid, void or unenforceable and (ii) the provisions of Section I.C.5 of the Prior Agreement shall again become effective.

3. <u>Execution of County Restrictive Covenant</u>. As a material inducement for RMV to enter into this Agreement, the County agrees to execute, record and comply with the County Restrictive Covenant, the terms of which are hereby incorporated into this Agreement as if set forth in full herein. Any violation of the County Restrictive Covenant shall be a violation of this Agreement. The County shall cause the County Restrictive Covenant to be recorded in the official records of Orange County within five (5) days of the date hereof.

4. <u>RMV's Use of Restricted Area</u>.

4.1. <u>Grazing</u>. RMV has requested a license to enter and use the Restricted Area to conduct livestock grazing for a period of 15 years. The County agrees to grant RMV such a license, subject to such terms and conditions as are reasonably acceptable to the County and, if applicable, the completion of any necessary CEQA documentation. Any such license shall be set forth in a separate written agreement that is approved and executed by both Parties.

Any such license shall be free of charge to RMV except that County may require that RMV reimburse County for any direct, out-of-pocket costs incurred by County in connection therewith.

4.2. NCCP Mitigation.

4.2.1 Right of First Refusal. The Parties acknowledge that the Restricted Area may be used to conduct habitat mitigation or other similar operations (including remedial grading and revegetation) pursuant to any approved Natural Community Conservation Plan, Habitat Conservation Plan or other similar plan of conservation (collectively "NCCP"), and that the County may seek to sell or otherwise transfer its rights to do the same either through a sale of the land or through sales of mitigation units or credits (as used herein, any such land, units, credits or other mitigation rights are referred to as "NCCP Credits"). County hereby grants to RMV a right of first refusal for any NCCP Credits on the following terms and conditions. Before County may sell or transfer all or any NCCP Credits to any third party, County must first offer to sell or transfer the same to RMV by giving written notice of the specific price, terms and conditions on which County is willing to sell or transfer (and the proposed transferee is willing to purchase) the NCCP Credits. RMV will have thirty (30) days after receiving County's notice within which to notify County in writing that RMV elects to purchase the subject NCCP Credits on the same terms and conditions contained in County's notice; in the event RMV fails to notify County in the above time period, RMV shall be deemed to have not elected to purchase the subject NCCP Credits. If RMV timely elects to purchase the subject NCCP Credits, then County shall sell the subject NCCP Credits to RMV pursuant to the terms of the County's notice, provided (i) in no event shall the closing occur in less than thirty (30) days from the date of RMV's election to purchase and

(ii) RMV shall have the right to pay money in lieu of any non-cash consideration described in County's notice. If RMV elects not to purchase the subject NCCP Credits (whether actively or passively), County may sell or transfer the same at the price and upon the terms contained in County's notice. The obligations of this Section shall survive throughout the term of this Agreement and shall be binding upon County and County's Successors-in-Interest.

4.2.2 <u>Cooperation</u>. The Parties acknowledge that due to the size and adjacency of the No-Build Area and the Restricted Area, certain benefits may be realized in jointly coordinating, planning and/or implementing an NCCP on the No-Build Area and/or the Restricted Area. In order to realize such benefits of any NCCP, the Parties agree to reasonably cooperate with respect to the creation and/or implementation of any NCCP on the Restricted Area and/or the No-Build Area; provided, however, that neither Party shall be obligated to subject their respective lands to an NCCP.

5. **RMV's Use of No-Build Area.**

5.1. <u>Restriction on Habitable Buildings</u>. RMV hereby agrees not to construct or allow construction of any habitable buildings within the No-Build Area for a period of thirty (30) years following closure of the Landfill Operations Area.

5.2. <u>Further Documentation</u>. If, in the future, RMV transfers all or any portion of the No Build Area to another entity or third-party, then RMV shall include in the deed or other document effectuating such transfer a restrictive covenant against the No-Build Area consistent with the restrictions set forth in Section 5.1 or shall otherwise obligate such transferee to comply with Section 5.1 pursuant to a written agreement reasonably acceptable to the County. Similarly, if, in the future, RMV seeks to obtain any development or entitlement rights for the No-Build

Area, then the County shall have the right to require RMV to execute and record a restrictive covenant against the No-Build Area consistent with the restrictions set forth in Section 5.1 or to otherwise make such rights conditioned upon RMV's compliance with Section 5.1.

5.3. Further Restrictions. RMV agrees not to protest or otherwise object to the imposition and/or recordation of a restrictive covenant against the No-Build Area pursuant to Sections 5.1 and 5.2. Further, RMV agrees not to assert any action or claim (including, but not limited to, an action or claim for inverse condemnation) against the County based on the imposition of the restrictions against the No-Build Area set forth in Section 5.1. County agrees not to seek any restrictions on RMV's use of the No Build Area for any of the purposes set forth in Section 5.4 in connection with the impacts or effects of County's Landfill Operations on the Burdened Property.

5.4. <u>RMV's Retained Rights</u>. Apart from the specific restrictions set forth in this Agreement, RMV shall retain all other rights and privileges relating to the development and use of the RMV Benefitted Property (including the No-Build Area). Except as provided in Section 5.5 and without limiting the generality of the preceding sentence, nothing in this Agreement prohibits RMV from: (i) constructing water, sewer, gas, telecommunication, electric or other utilities and appurtenances within the No-Build Area; (ii) constructing non-habitable structures, trails or paths within the No-Build Area; (iii) conducting grading activities or otherwise altering the elevations of the No-Build Area (provided such grading activities do not diminish the soil stability of the Burdened Property, and provided further, that in the event any such activities alter the existing elevations as identified on Attachment 2, then any provisions herein respecting visual impacts shall be based upon the existing elevations and not such altered elevations); (iv) seeking entitlements and development approvals (including those associated

with residential units) for the No-Build Area prior to expiration of the thirty (30)-year period specified in Section 5.1; (v) using the No-Build Area in connection with, or for the benefit of, an NCCP or otherwise in furtherance of habitat and/or species protection and/or preservation; (vi) constructing a habitable structure on those portions of a lot that are outside of the No-Build Area, where such lot is only partially within the No-Build Area; (vii) using the No-Build Area for agriculture and/or grazing purposes; and (viii) using any and all land within the No-Build Area: (a) to satisfy the requirements of any public agency (including the County) in regards to general open space requirements or environmental or land use related mitigation and/or restoration measures, commitments or conditions with respect to development within the Rancho Mission Viejo or any portion thereof; (b) for the purpose of transferring any and all development rights pertaining to the No-Build Area for use in other portions of Rancho Mission Viejo; and/or (c) for the purpose of receiving tax benefits, for example, by placing a conservation easement over the No-Build Area in favor of, or transferring the title in the No-Build Area to, an appropriate non-profit entity. This Agreement shall not, nor shall it be construed to, apply to or in any way restrict RMV's use of or activities in areas outside the boundaries of the No-Build Area.

5.5. <u>Compliance with Laws</u>. RMV agrees to obtain all necessary permits and approvals, and to comply with all applicable laws (including, without limitation, compliance with health and safety laws, regulations and permit conditions which require certain construction materials, methods or processes be utilized on account of the proximity to the Landfill Operations), in conjunction with any use of the No Build Area for the purposes set forth in Section 5.4 and any other uses not otherwise restricted by this Agreement but allowed on and within the No-Build Area. Furthermore, RMV shall not engage in any such use of the No Build

Area if doing so would materially conflict or otherwise interfere with the restrictions set forth in Section 5.1 or the County's rights pursuant to Sections 5.2 and 5.3.

5.6. <u>Obligations Run with Land</u>. The restrictions and obligations set forth in this Section 5 shall run with and be binding upon the No-Build Area for the benefit of the Burdened Property.

6. <u>County's Use of Landfill Operations Area</u>. The County shall conduct its Landfill Operations within the Landfill Operations Area in accordance with: (i) the restrictions and provisions set forth in the County Restrictive Covenant; and (ii) the provisions set forth Sections 6.1 through 6.6 below.

6.1. <u>Phasing of Landfill Activities</u>. As of the date hereof, the County anticipates that Landfill Operations within the Landfill Operations Area will occur in phases moving in a west to east progression, and that the placement of refuse within the easterly portion of the Landfill Operations Area will not commence until after the year 2025. Beginning two (2) years prior to commencement of Landfill Operations in the Landfill Operations Area and continuing until such time as the RMV Benefitted Property is fully developed, the County shall provide annual reports to RMV that describe the County's then current plans for the phasing of its Landfill Operations within the Landfill Operations Area. In the event that the County anticipates materially altering the general west-to-east phasing scheme described above, the County shall provide written notice of the same to RMV and shall meet and confer with RMV about such alterations. County shall, in good faith, consider any alternatives or suggestions tendered by RMV prior to materially altering the general west-to-east phasing scheme.

6.2. <u>Sampling and Monitoring of Landfill Gas</u>. The County shall conduct gas sampling and monitoring as needed to confirm that no landfill gas migrates from the Landfill

Operations Area onto the RMV Benefitted Property. Such sampling and monitoring shall commence no later than the first day upon which refuse is placed within 1,000 feet of the RMV Benefitted Property (or such greater distance if required by applicable laws) and shall be conducted at least semi-annually thereafter for a period of at least 30 years following closure of the PDL. The results of such sampling and monitoring shall be transmitted at no cost to RMV upon its request.

6.3. Implementation of Mitigation Measures. The County shall conduct its Landfill Operations in the Landfill Operations Area in accordance with the 2001 GDP and EIR No. 575, and any adopted mitigation measures specified in or required by the 2001 GDP and EIR No. 575. However, if any subsequent, supplemental or additional development plans and/or environmental documents are prepared and approved for the PDL after the date of execution of this Agreement, then such plans and documents, and any adopted mitigation measures specified in or required by such plans and documents, that are in conflict or inconsistent with those established by the 2001 GDP or EIR No. 575 shall take precedence over the 2001 GDP and EIR No. 575. Notwithstanding the above, the County shall not be permitted to implement such plans or documents or any adopted mitigation measures specified in or required by such plans or documents if such implementation would result in materially greater (or materially less effective mitigation of the) impacts of the Landfill Operations on the RMV Benefitted Property (including, without limitation, any visual impacts on account of the height of the landfill) than specified in or required by the 2001 GDP and EIR No. 575.

6.4. <u>Implementation of Viewshed Protection and Landscaping Plan</u>. Pursuant to Condition No. 5 of Resolution No. 01-380 (which condition is set forth on page 5 thereof), County is required to conduct a viewshed analysis with respect to the impacts of Zone 4's

development on the RMV Benefitted Property, and, if needed, prepare and implement a viewshed protection and landscaping plan ("VPL Plan"). If required pursuant to Condition No. 5 of Resolution No. 01-380, the VPL Plan shall be designed to provide noise, visual, litter and dust screening between Landfill Operations in the Landfill Operations Area and the RMV Benefitted Property. The County shall confer with RMV in developing the specific details of the VPL Plan; further, the County shall obtain RMV's approval of any VPL Plan where the impacts of the same reasonably affect RMV Benefitted Property (including material visual, noise and odor impacts).

6.5. <u>Control of Vectors.</u> The County shall adopt and implement policies and procedures designed to control, to the extent practicable, the growth and spread of non-native invasive plants and animals both within the Landfill Operations Area and from the Landfill Operations Area onto the RMV Benefitted Property. Notwithstanding the above, the County may use non-native plants and trees for screening purposes and erosion control as part of any VPL Plan required pursuant to Section 6.4. Records of any vector inspections conducted at any location within the Landfill Operations Area and any actions resulting from such inspections shall be made available at no cost to RMV upon its request. In addition to any other vector control efforts, the County shall investigate the feasibility of using falcons to reduce the potential impact of birds on off-site property.

6.6. <u>County's Retained Rights</u>. Apart from the specific restrictions set forth in this Agreement (including, without limitation, those set forth in the County Restrictive Covenant), the County shall retain all other rights and privileges relating to the development and use of the Burdened Property (including the Restricted Area) and the implementation of the 2001 GDP, including but not limited to the right to use soil fill or other engineering methods to minimize the capacity loss resulting from any design adjustments required by the restrictions,

covenants and limitations set forth in this Agreement (including, without limitation, those set forth in the County Restrictive Covenant). This Agreement shall not, nor shall it be construed to, apply to or in any way restrict the County's use of or activities in areas outside the boundaries of the Burdened Property.

7. La Pata Avenue.

7.1. Nomenclature. The road commonly known as La Pata Avenue currently consists of an approximately 1.5-mile long, three-lane road that provides access to the PDL between the northerly limits of the PDL and Ortega Highway. The Orange County Transportation Authority ("OCTA") Master Plan of Arterial Highways ("MPAH") and the County's Circulation Plan identify a proposed future primary arterial highway with a conceptual alignment from Ortega Highway to Avenida Pico in the City of San Clemente, with the portion to the north of the PDL to Ortega Highway generally following the existing access road for the PDL from Ortega Highway. This conceptual alignment is referred to in various documents as "Avenida La Pata." For purposes of this Agreement, "La Pata Avenue" shall mean only those portions of said proposed future primary arterial highway that extend from Ortega Highway to the southerly limits of the PDL at the San Clemente/Orange County border.

7.2. <u>Construction/Reconstruction Costs</u>. The County believes that La Pata Avenue meets landfill access needs to the PDL. Notwithstanding, if the County, the OCTA, or any other party chooses to improve, widen, realign, reconstruct or extend La Pata Avenue (or any portion thereof to meet primary arterial highway standards), then County will be responsible for seven and two-tenths percent (7.2%) of the applicable project costs. The Parties agree that said percentage constitutes the County's fair share of such costs in that the same represents the parties' estimate of the percentage of estimated passenger car-equivalent, average daily trips
generated by or associated with Landfill Operations at the PDL (i.e., at current maximum permitted daily tonnage) relative to the total estimated passenger car-equivalent, average daily trips of build-out traffic on La Pata Avenue. If, subsequent to the Effective Date of this Agreement, County obtains approval to increase the maximum permitted daily tonnage, then County's fair share percentage shall be recalculated based on the increased daily tonnage.

7.3. <u>Maintenance and Repair Costs</u>. County is and shall continue to be responsible for all maintenance and repair of La Pata Avenue.

7.4. Disclaimer of Intent. This Agreement is not intended to be, nor shall it be construed as, a commitment by the County to make any improvements or changes to Existing La Pata as currently constructed. Notwithstanding anything to the contrary expressed or implied in this Agreement, County's obligations to fund its fair share of costs set forth in Section 7.2 above relate to County's obligations regarding La Pata solely with respect to the Landfill Operations at the PDL; this Agreement is not intended nor shall it be construed to limit or preclude: (i) other agreements between the County, RMV and/or other parties that may further define or delineate the sharing of costs associated with the improvement, widening, realignment, reconstruction, construction, extension, use, maintenance and repair of La Pata Avenue not related to use of La Pata Avenue for access to the PDL, or (ii) County's, RMV's or any other person's obligations to contribute to the construction costs of La Pata Avenue.

7.5. <u>License to Use La Pata Avenue</u>. RMV and certain other persons currently use La Pata Avenue to access portions of the Rancho Mission Viejo for agricultural and other purposes. RMV is hereby granted an irrevocable license to continue using La Pata Avenue for these purposes, subject to the County's continuing right to specify and/or approve of the locations of and reasonable conditions for such access. This license is granted to and inures

solely to the benefit of RMV, RMV's Successor-in-Interests and/or RMV's affiliates and may not be transferred to any other party without the County's prior written approval, which may be withheld at the County's sole discretion. Further and except for those users identified in the following sentence, RMV shall not license, permit or in any other manner allow any other party under its authority, direction and/or control to use La Pata Avenue for any reason without the County's prior written approval, which may be withheld at the County's sole discretion. The parties acknowledge that in addition to RMV the following persons currently gain access to Rancho Mission Viejo via La Pata Avenue: (i) Tierra Verde Industries (dba La Pata Greenwaste): (ii) Freddie Love's stables; (iii) Orange County Fire Authority; (iv) public utility companies servicing existing facilities (including Santa Margarita Water District, Southern California Edison, San Diego Gas and Electric and Kinder Morgan (aka SFPP)); (v) Sierra Soils; and (vi) Rancho Mission Viejo, LLC and its affiliated entities (including, without limitation, those affiliates for whom Rancho Mission Viejo, LLC acts or serves as agent and/or manager).

8. Release of Claims.

8.1. RMV's Release of Claims Relating to EIR 575 and Approval of 2001

<u>GDP</u>. Except for the obligations provided herein, RMV hereby unconditionally releases, remises, acquits and forever discharges the County from claims or causes of action, including any and all administrative or judicial hearings or appeals, or any other litigation in a court of law, either at law or in equity, of any kind, nature and description, presently known or unknown and whether presently existent or nonexistent, relating to: (i) the County's certification of EIR No. 575 and/or the approvals relating to the 2001 GDP granted on or about November 6, 2001; and (ii) any subsequent actions taken by the County to implement the 2001 GDP, so long as said actions are consistent with the terms of this Agreement.

8.2. <u>Waiver of California Civil Code Section 1542</u>. RMV agrees that the release contained in Section 8.1 extends to all claims of any kind or nature relating to the County's certification of EIR No. 575 and/or the approvals relating to the 2001 GDP granted on or about November 6, 2001, whether known or unknown, suspected or unsuspected, and in that regard RMV acknowledges that it has read, been advised by counsel concerning, and considered and understands the full nature, extent and import of the provisions of Section 1542 of the Civil Code of California, which reads as follows:

A general release does not extend to claims which the creditor does not know or suspect to exist in his favor at the time of executing the release, which if known by him must have materially affected his settlement with the debtor.

RMV further declares that it knowingly and willingly enters into this Agreement notwithstanding the provisions of Section 1542 of the Civil Code of California. To the fullest extent allowed by law, RMV hereby expressly, knowingly, and voluntarily waives and relinquishes any and all rights and benefits that it now has or may have in the future under California Civil Code Section 1542 with respect to the actions identified in Section 8.1 above.

8.3. County's Release of Claims Relating to EIR 575 and Approval of 2001

<u>GDP</u>. Except for the obligations provided herein, County hereby unconditionally releases, remises, acquits and forever discharges RMV from claims or causes of action, including any and all administrative or judicial hearings or appeals, or any other litigation in a court of law, either at law or in equity, of any kind, nature and description, presently known or unknown and whether presently existent or nonexistent, relating to: (i) RMV's assertions that the County's certification of EIR No. 575 and/or the approvals relating to the 2001 GDP granted on or about November 6, 2001 were inadequate; and (ii) any subsequent actions taken by the RMV in connections with such assertions, so long as said actions are consistent with the terms of this Agreement.

9. <u>Indemnification</u>.

9.1. <u>County's Obligations</u>. The County shall save, protect, defend, indemnify and hold harmless RMV, and all of their officers, directors, shareholders, partners, members, agents, representatives, managers and employees from and against any actions, causes of actions, claims, liabilities, losses, costs, damages, fees and expenses of whatever kind (including, without limitation, attorneys' fees and costs) caused by, relating to or arising from the County's breach of this Agreement, including, without limitation, any breach of the County Restrictive Covenant.

9.2. <u>RMV's Obligations</u>. In addition to any indemnification required pursuant to Section 10 below, each constituent entity of RMV (as such entities are identified in the introductory paragraph hereof) (the "Indemnifying Entity") shall save, protect, defend, indemnify and hold harmless the County, the other constituent entities of RMV, and each and all of their respective elected and appointed officials, officers, directors, agents, representatives, consultants, managers, staff and employees from and against any actions, causes of actions, claims, liabilities, losses, costs, damages, fees and expenses of whatever kind (including, without limitation, attorneys' fees and costs) caused by, relating to or arising from the Indemnifying Entity's breach of this Agreement.

10. <u>**Right of Inspection**</u>. RMV shall, at all reasonable times, have the right to enter the Burdened Property to inspect the Landfill Operations in order to confirm the County's compliance with this Agreement. In connection with any such entry, RMV shall comply with County's reasonable notification requirements and other rules and regulations concerning entry

on the Burdened Property. Furthermore, the Indemnifying Entity (as defined in Section 9.2 above) agrees to save, protect, defend, indemnify and hold harmless the County, the other constituent entities of RMV, and each and all of their respective elected and appointed officials, officers, directors, agents, representatives, consultants, managers, staff and employees from and against any and all actions, causes of action, claims, liabilities, losses, costs, damages and expenses of whatever kind (including, without limitation, attorneys' fees and costs) ("Claims"), caused by, relating to or arising from the Indemnifying Entity's entry onto the Burdened Property in connection with RMV's inspection of landfill operations; provided, however, that RMV shall have no obligation under this section to indemnify County or any other person to the extent such Claims result from the negligence or misconduct of County or any of its elected officials, officers, directors, agents, representatives, consultants, managers, staff and employees.

11. Right of Enforcement.

11.1. Notice and Cure. In the case of breach of any obligation hereunder, a Party shall not be deemed to be in default of this Agreement if such Party cures such a breach within ninety (90) days of receiving notice from the other Party of such breach; provided, however, that if such breach is not capable of being cured within said period, no default shall be deemed to have occurred so long as the Party receiving the notice of default has taken material affirmative steps to cure such breach within said period and thereafter diligently and continuously prosecutes the same to conclusion.

11.2. <u>Parties Entitled to Enforce Agreement</u>. This Agreement inures to the benefit of RMV, RMV's Successors-in-Interest, the County and the County's Successors-in-Interest, each of which shall have the exclusive right to enforce by proceedings at law or in equity, any and all of the rights, covenants and obligations now or hereafter imposed by the

provisions of this Agreement or any amendment thereto, including, without limitation, (i) the right to prevent the violation of any provisions hereof and (ii) the right to recover damages for such violation. No other person or entity shall have, nor may they be assigned, the right to enforce this Agreement.

11.3. <u>No Waiver; Cumulative Remedies</u>. Failure by a Party to enforce any right, covenant or obligation in any certain instance or on any particular occasion shall not be deemed a waiver of such right, covenant or obligation on any future breach of the same or any other right, covenant or obligation by the other Party. All rights, options and remedies under this Agreement are cumulative; and no one of them shall be exclusive of any other; and each Party shall have the right to pursue any one or all of such rights, options and remedies or any other remedy or relief which may be provided by law, whether or not stated in this Agreement.

12. Miscellaneous Provisions.

12.1. <u>No Prior Assignments</u>. The Parties hereto represent and warrant that they have not heretofore assigned or transferred or purported to assign or transfer to any other person, entity, firm or corporation whatsoever, any claim, debt, liability, demand, obligation, expense, action or causes of action herein released.

12.2. <u>Binding Effect</u>. This Agreement and its terms shall be binding upon (i) RMV, RMV's Successors-in-Interest, and their officers, directors, shareholders, partners, members, agents, representatives, managers, staff and employees, and (ii) the County, the County's Successors-in-Interest, and their elected and appointed officials, officers, directors, shareholders, partners, members, agents, representatives, consultants, managers, staff and employees, as though they were Parties hereto, wherever they may be located. Notwithstanding any other provision to the contrary contained herein, upon any sale or transfer of a Party's entire

interest in the Burdened Property or the RMV Benefitted Property, then the transferring Party shall be entirely freed and relieved of all liability under any and all of its covenants and obligations contained in or derived from this Agreement arising out of any act, occurrence or omission occurring after the consummation of such sale or transfer. The transferee of such interest shall be deemed, without any further agreement or instrument, to have assumed and agreed to carry out any and all covenants and obligations of the transferring Party arising out of any act, occurrence or omission occurring after the consummation of such sale or transfer. Further, in the event that the transferee of such interest agrees in writing to assume any and all covenants and obligations of the transferring Party arising out of any act, occurrence or omission occurring prior to the consummation of such sale or transfer, then the transferring Party shall be entirely freed and relieved of all liabilities under any and all of its covenants and obligations contained or derived from this Agreement whatsoever.

12.3. <u>Settlement of Disputed Claims</u>. The Parties hereto understand and agree that this Agreement is a settlement and compromise of disputed claims, and that no Party's actions under this Agreement shall be construed as an admission of liability.

12.4. Entire Agreement. This writing constitutes the entire agreement among the Parties with respect to the subject matter hereof, and no modification of this Agreement shall be valid unless executed in writing by the Parties hereto. The Parties intend and agree that, except as expressly set forth in this Agreement, the Prior Agreement shall not be affected, amended or altered by this Agreement. Further, none of the Parties to this Agreement shall be bound by any representations, warranties, promises, statements, or information unless expressly set forth herein.

12.5. <u>Factual Investigation</u>. Each Party has conducted its own factual investigation, is not relying on any other Party, and assumes the risk that there are material unknown facts or that facts are other than as is presumed. The Parties further acknowledge that they are aware that they may hereafter discover material facts in addition to or different from those which they now know or believe to be true with respect to the subject matter of this Agreement, and further acknowledge that there may be future events, circumstances, or occurrences materially different from those they know or believe likely to occur, but that it is their intention to enter into and be bound by this Agreement.

12.6. <u>Agreement May Be Pleaded as a Defense</u>. In connection with any demand or cause of action related to a matter released in Sections 8.1 and 8.2, inclusive, this Agreement may be pleaded as a defense by the Parties hereto and shall operate to effect a dismissal of such demand or cause of action.

12.7. <u>Captions</u>. The captions of the various sections in this Agreement are for convenience and organization only, and are not intended to be any part of the body of this Agreement, nor are they intended to be referred to in construing the provisions of this Agreement.

12.8. <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of California.

12.9. <u>Severability</u>. If any terms or provisions of this Agreement or the application of any term(s) or provision(s) of this Agreement to a particular situation, is (are) held by a court of competent jurisdiction to be invalid, void or unenforceable, the remainder of this Agreement or the application of this Agreement to other situations shall remain in full force and effect unless amended or modified by mutual consent of the Parties; provided that, if the

invalidation, voiding or unenforceability would deprive any Party of material benefits derived from this Agreement, or make performance under this Agreement unreasonably difficult, then the Parties shall meet and confer and shall make good faith efforts to amend or modify this Agreement in a manner that is mutually acceptable to the Parties. In the event that the Parties are not able to agree upon such amendments or modification, then this Agreement shall be terminated in its entirety and the parties shall have all rights, obligations and remedies available as of the Effective Date as if this Agreement had not been entered into.

12.10. Notices, Demands and Communications Between the Parties. Formal written notices, demands, correspondence and communications between the Parties shall be sufficiently given if delivered personally (including delivery by private courier), dispatched by certified mail, postage prepaid and return receipt requested, or delivered by nationally recognized overnight courier service, or by electronic facsimile transmission followed by delivery of a "hard" copy to the offices of the Parties indicated below. Such written notices, demands, correspondence and communications may be sent in the same manner to such persons and addresses as any Party may from time-to-time designate in writing at least fifteen (15) days prior to the name and/or address change and as provided in this Section 12.10.

If to RMV:	Rancho Mission Viejo, LLC 28811 Ortega Highway San Juan Capistrano, CA 92675 Attn: V.P. –Planning & Entitlement Facsimile: (949) 488-2676
With copy to:	Brobeck, Phleger & Harrison LLP 38 Technology Drive Irvine, CA 92661 Attn: Drew Jones, Esq. Facsimile: (949) 790-6301

If to County: County of Orange Integrated Waste Management Department 320 North Flower Street, Suite No. 400 Santa Ana, CA 92703 Attn: Director of IWMD Facsimile: (714) 834-4183

With a copy to:

County of Orange Office of the County Counsel 10 Civic Center Plaza, 4th Floor P.O. Box 1379 Santa Ana, CA 92702 Attn: County Counsel Facsimile: (714) 834-2359

Notices personally delivered shall be deemed to have been received upon delivery. Notices delivered by certified mail, as provided above, shall be deemed to have been given and received on the first to occur of (i) actual receipt by any of the addressees designated above as the Party to whom notices are to be sent, or (ii) within five (5) days after a certified letter containing such notice, properly addressed, with postage prepaid, is deposited in the United States mail. Notices delivered by nationally recognized overnight courier service (such as Federal Express) as provided above shall be deemed to have been received twenty-four (24) hours after the date of deposit. Notices delivered by electronic facsimile transmission shall be deemed received upon receipt of sender of electronic confirmation of delivery, provided that a "hard" copy is delivered by overnight courier as provided above.

12.11. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and all the counterparts shall constitute but one and the same agreement, notwithstanding that all Parties hereto are not signatories to the same or original counterpart.

12.12. <u>Nonwaiver</u>. Unless otherwise expressly provided in this Agreement, no waiver by a Party of any provision hereof shall be deemed to have been made unless expressed

in writing and signed by such Party. No delay or omission in the exercise of any right or remedy accruing to any Party upon any breach under this Agreement shall impair such right or remedy or be construed as a waiver of any such breach theretofore or thereafter occurring. The waiver by a Party of any breach of any term, covenant or condition herein stated shall not be deemed to be a waiver of any other term, covenant or condition.

12.13. <u>Authority</u>. The persons signing below represent that they: (i) have the authority to bind their respective Parties to this Agreement; and (ii) have obtained any and all necessary approvals required to executed this Agreement.

12.14. <u>Understanding of Terms</u>. The Parties each hereby affirm and acknowledge that they have read this Agreement, that they know and understand its terms, and have signed it voluntarily and on the advice of counsel. The Parties have had a full and unhindered opportunity to consult with their attorneys, accountants, financial advisors and such other consultants, as they may have desired prior to executing this Agreement.

12.15. <u>Construction</u>. The Parties acknowledge that each Party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendment or exhibits hereto.

12.16. <u>Attachments</u>. All attachments attached hereto shall be incorporated herein by reference as if set forth herein in full.

12.17. <u>No Third Party Beneficiaries</u>. The Parties agree that no third party beneficiary to this Agreement exists and that nothing contained herein shall be construed as giving any other person or entity third party beneficiary status.

12.18. <u>Further Assurances</u>. The Parties shall promptly perform, execute and/or deliver or cause to be performed, executed and/or delivered any and all acts, deeds and assurances, including the delivery of any documents, as either Party may reasonably require in order to carry out the intent and purpose of this Agreement.

12.19. <u>Recordation of Memorandum of Agreement</u>. Neither Party shall record this Agreement nor a short form hereof. However, the County Restrictive Covenant referenced in Section 3 shall be recorded in accordance with the provisions of Section 3.

12.20. <u>Attorneys' Fees</u>. In the event any action shall be instituted in connection with this Covenant, the party prevailing in such action shall be entitled to recover from the other party all of its costs and expenses incurred therein, including, without limitation, reasonable attorneys' fees and costs.

IN WITNESS WHEREOF, the Parties hereto have executed one or more copies of this

Agreement as of the date first set forth above.

Date: October 1/2, 2002 November

COUNTY OF ORANGE

By:

Jánice V. Goss, Director Integrated Waste Management Department

Date: October <u>17</u>, 2002

SAN JUAN PARTNERSHIP NO. I, a California limited partnership

By: O'NEILL PROPERTIES, LLC, a Delaware limited liability company, its General Partner

By: Anthony R loiso Manager

- SAN JUAN PARTNERSHIP NO. IV, a California Limited Partnership
- By: O'NEILL PROPERTIES, LLC, a Delaware limited liability company, its General Partner

By: Anthony R Ioiso Manager

DMB SAN JUAN INVESTMENT NORTH, LLC, a Delaware limited liability company,

By: Rancho Mission Viejo, LLC, a Delaware limited liability company, its authorized agent and manager

By: Donald L. Vodra

Chief Operating Officer

By:

Richard Broming Vice President

APPROVED AS TO FORM:

Dated: October <u>17</u>, 2002

BROBECK, PHLEGER & HARRISON LLP

nto By: C. Jeff Brinkon

Attorneys for DMB SAN JUAN INVESTMENT NORTH and SAN JUAN PARTNERSHIP Nos. I and IV

By: Geoffrey Hunt

Deputy County Counsel

Attorneys for COUNTY OF ORANGE

Dated: October 23, 2002

ATTACHMENT 1

County Restrictive Covenant

COLIB1\MWP\255914.20

RECORDING REQUESTED BY AND WHEN RECORDED MAIL TO:

Rancho Mission Viejo, LLC 28811 Ortega Highway San Juan Capistrano, California 92675 Attn: Vice President - Operations

FOR RECORDER'S OFFICE USE ONLY

COVENANT AND DECLARATION OF RESTRICTIONS

THIS COVENANT AND DECLARATION OF RESTRICTIONS ("Covenant") is made and entered into this <u>22wd</u> day of <u>Cetaber</u> 2002, by and between the COUNTY OF ORANGE, a political subdivision of the State of California ("County"), DMB SAN JUAN INVESTMENT NORTH, LLC, a Delaware limited liability company ("DMBSJIN"), SAN JUAN PARTNERSHIP NO. I, a California limited partnership ("SJP I") and SAN JUAN PARTNERSHIP NO. IV, a California limited partnership ("SJP IV"). DMBSJIN, SJP I and SJP IV are referred to collectively herein as "RMV".

RECITALS

A. The County owns approximately 1,530 acres in southern Orange County, California on which the County operates the Prima Deshecha Landfill (the "PDL"). Prior to the date hereof, the County purchased the real property described on <u>Exhibit "A"</u> attached hereto and depicted on <u>Exhibit "B"</u> attached hereto (the "Burdened Property") from RMV's predecessorsin-interest; the Burdened Property comprises approximately 945 acres of the PDL's 1,530 acres.

B. The current general development plan for the PDL (known as the "2001 GDP") was approved by the County Board of Supervisors on November 6, 2001. The 2001 GDP

divides up the PDL into five (5) zones and designates specific uses for such zones: Zones 1 and 4 are designated for landfilling. The environmental impacts for the 2001 GDP were addressed by Final Program EIR 575 ("FEIR 575").

C. RMV owns fee title to those certain real properties situated in the County of Orange, State of California that are contiguous with the Burdened Property, as more particularly described in <u>Exhibit "C"</u> attached hereto and incorporated herein by this reference (collectively the "Benefitted Property").

D. Pursuant to that certain Settlement Agreement executed by RMV and County dated concurrent herewith, County has agreed to certain limitations and restrictions on: (i) the Landfill Operations (defined in Section 2 below) conducted on those portions of the Burdened Property delineated and described as the "Landfill Operations Area" on Exhibit "B" attached hereto (the "Landfill Operations Area"), and (ii) the development and use of those portions of the Burdened Property delineated and described as the "Restricted Area" on Exhibit "B" attached hereto (the "Restricted Area"), all as set forth in this Covenant for the benefit of RMV and the Benefitted Property. The Landfill Operations Area together with the Restricted Area constitute all of the Burdened Property.

E. The parties desire to record this Covenant to put future owners, lessees and users on notice of the restrictions, conditions, covenants and reservations now or hereafter imposed by the provisions of this Covenant (collectively the "Restrictions").

AGREEMENT

NOW, THEREFORE, County hereby covenants, agrees and declares that the Burdened Property shall be used, operated, held, conveyed and transferred subject to the following Restrictions:

1. Amendment of Prior Agreement. Once this Covenant has been duly executed and recorded in the official records of the County, the provisions of Section I.C.5 of that certain Purchase and Option Agreement executed between the County and RMV's predecessors-ininterest on June 13, 1973 and referenced in that certain deed recorded in Book 10776, Page 926 of the official records of the County (pursuant to which the County purchased the Burdened Property from RMV's predecessors-in-interest) shall no longer be of any force or effect. Notwithstanding the preceding sentence, in the event that (i) the settlement agreement referenced in Recital D above is held by a court of competent jurisdiction to be invalid, void or unenforceable, and (ii) the holding of said court (or a short form memorandum of the same) in which the recording information of this Covenant is specifically referenced is recorded in the official records of County by any party to said agreement or by said court, then this Covenant shall terminate and be of no further force or effect and the provisions of Section I.C.5 of said Purchase and Option Agreement shall be revived and reinstated and again become effective.

2. <u>Restrictions On County Landfill Operations At PDL</u>. County hereby promises, covenants and agrees that it will conduct its Landfill Operations on the Burdened Property in accordance with the requirements set forth in this Section 2 (including Sections 2.1 through 2.5). As used herein, the term "Landfill Operations" shall mean: (i) the storing, collecting, transporting, treating, processing, recycling, reusing, converting, disposing, transferring, placing, burying, relocating or otherwise handling in any manner whatsoever any

solid, semi-solid, liquid or gas waste of any type; (ii) all activities and operations related to closure and/or post-closure of the landfill; (iii) the construction, use, operation, maintenance, repair and/or replacement of buildings, structures, improvements, environmental monitoring and control systems, property and equipment in connection with the activities and operations identified in subsections (i) and (ii) above; (iv) all site grading, cutting, filling and other earth movement work performed in connection with the activities and operations identified in subsections (i), (ii) or (iii) above; and (v) the construction, use, operation, maintenance, repair and/or replacement of roads, paths, trails and accessways (whether temporary or permanent) in connection with the activities and operations (i), (ii), (iii) and (iv) above.

2.1. <u>Limitation on Location of Landfill Operations</u>. Except as provided in Section 3.2 below, all of the Landfill Operations occurring on the Burdened Property shall be conducted exclusively within Landfill Operations Area.

2.2. Further Limitation on Placement of Refuse. Notwithstanding any other provision of this Covenant to the contrary, the County shall not place any waste or refuse on any portion of the Burdened Property that is located outside of the Landfill Operations Area. Furthermore, the height of any waste or refuse placed within the Landfill Operations Area shall not exceed one thousand and ten (1,010) feet above mean sea level unless and until appropriate measures (including any measures required by Applicable Laws (defined below)) are taken to screen from view any portion of such waste or refuse that is visible from the Benefitted Property.

2.3. <u>Design of Landfill</u>. The County shall adjust the design of any landfill within the Burdened Property and/or take such other steps as may be necessary to prevent or

mitigate any landfill-related costs and impacts on the Benefitted Property. Furthermore, to the extent it may legally do so, the County shall take any and all steps necessary to prevent any landfill-related mitigation measures (including, without limitation, any buffer zones) from being imposed on the Benefitted Property. However, if landfill-related mitigation measures are prescribed by the County (or any agency thereof) or by any other governmental agency (whether generally or in connection with land use decisions affecting the Benefitted Property), then the County shall adjust the design of the landfill or otherwise modify its Landfill Operations within the Landfill Operations Area so as to eliminate the need for such mitigation measures. Nothing contained in this Section 2.3 shall restrict the County from imposing any landfill-related fees or taxes so long as such fees or taxes (i) are applied in a non-discriminatory manner to all users of the PDL and (ii) are not applicable solely to new development and/or construction.

2.4. Practices and Procedures To Be Employed on Burdened Property. The County shall conduct its Landfill Operations on the Burdened Property in accordance with: (i) all applicable federal, state and local laws, regulations, codes and standards (collectively "Applicable Laws"), and (ii) applicable industry standards for landfill and waste management facilities located within southern California (provided, however, that County shall not be obligated to retroactively implement any such industry standards unless required to do so by Applicable Laws. Whether or not necessary to comply with the obligations set forth in the preceding sentence, the County shall, at a minimum, employ the following practices and procedures in connection with its Landfill Operations on the Burdened Property:

2.4.1. <u>Litter, Waste and Dust Control</u>. The County shall take all commercially reasonable steps to eliminate the migration of litter, waste, dust and other similar substances or materials from the Burdened Property to the Benefitted Property, including,

without limitation: i) controlling and containing debris, wastes and litter from being blown from the Burdened Property, ii) constructing and maintaining a four foot high or higher litter fence, or its equal, around the working face of the landfill, and iii) utilizing such other measures as required to aid in preventing dust and litter migration beyond the landfill areas within the Burdened Property.

2.4.2. <u>Litter and Waste Removal</u>. The County shall be responsible for the removal of litter, waste and other similar substances originating in the landfill and migrating onto the Benefitted Property, provided that RMV, or the then owner of such land, grants the County reasonable access as necessary.

2.4.3. <u>Ridge Line Buffer</u>. So as to maintain a buffer zone and natural barrier to minimize viewing, noise, dust, litter, and other effects, if any, of Landfill Operations on the Benefitted Property, the height and natural contour of the existing ridge lines most immediately contiguous to the boundary line between the Burdened Property and the Benefitted Property shall not be materially modified by the County; provided, nothing in this provision shall preclude the installation and maintenance of landscaping along said ridge line.

2.4.4. <u>Vector Controls</u>. Vector control for rats, flies, birds, noxious insects, and larva shall be provided so that no significant increase in populations of said species occurs on off-site property due to Landfill Operations. A minimum of 6 inches of soil or other appropriate cover material shall be placed on the top of each day's deposit of refuse.

2.4.5. <u>Gas Controls</u>. An adequate gas control and monitoring program shall be initiated upon start of Landfill Operations to prevent gas migration onto the Benefitted Property.

2.4.6. <u>Drainage</u>. Adequate surface drainage and subdrainage systems shall be installed to prevent entry of rainfall runoff and groundwater into the solid waste fill. Further, all Landfill Operations shall be engineered, constructed, operated and maintained such that surface and ground water from the landfill does not adversely impact the Benefitted Property.

2.4.7. <u>Hours of Operation</u>. The County shall limit Landfill Operations (including, without limitation, the use of mechanized equipment) to the hours between 6:00 A.M. and 6:00 P.M., except in the case of emergency or other unusual situations.

2.4.8. <u>Construction of Perimeter Fence</u>. The County shall construct and maintain a second fence around the entire Burdened Property (but at least six (6) feet below the ridge lines described in Section 2.4.3 above) consisting of five strands of barbed wire, or equal, for a total height of approximately six (6) feet.

2.4.9. <u>Restrictive Access</u>. The County shall implement reasonable access controls to restrict access to designated areas and times for the landfills within the Burdened Property. All gates in landfill fences are to be kept closed and locked during hours when the respective landfill is not open.

2.5. La Pata Avenue Requirements.

2.5.1. <u>Maintenance and Repair</u>. As of the date hereof, the County has constructed La Pata Avenue which provides access to the PDL from Ortega Highway ("La Pata Avenue"). The County shall maintain La Pata Avenue in good condition and repair.

2.5.2. Dust and Litter Controls. The County shall take commercially reasonable steps to control any dust and litter along La Pata Avenue resulting from Landfill Operations. The County shall implement a commercially reasonable litter collection plan, which shall provide for litter collection not less than once each week. The County shall be responsible for the removal of litter, waste and other similar substances originating from users accessing the PDL via La Pata Avenue and migrating onto the Benefitted Property, provided that RMV, or the then owner of such land, grants the County reasonable access as necessary.

3. Restrictions on Restricted Area.

3.1. <u>Grant of Restrictive Covenant</u>. Except as provided in Section 3.2 below, County hereby promises, covenants and agrees to: (i) keep, manage and maintain the Restricted Area in its natural, undeveloped state, (ii) not to conduct any Landfill Operations within the Restricted Area, or (iii) not to construct any buildings, structures or other improvements upon the Restricted Area.

3.2. <u>Permitted Uses</u>. Subject to any Restrictions set forth elsewhere in this Agreement, including, without limitation, those set forth in Section 2 above, the County is hereby permitted to use the Restricted Area for the following purposes ("Permitted Uses"):

3.2.1. <u>Roads</u>. To install, operate, maintain, repair and/or replace public and privately owned and maintained trails and roads in the Restricted Area.

3.2.2. Grazing. To permit livestock grazing.

3.2.3. <u>Utilities: Public Improvements</u>. To install, operate, maintain, repair and/or replace (and to grant easements or licenses to third-parties to do the same) public

utility lines and associated improvements within the Restricted Area, including, without limitation, above-ground and below-ground electrical transmission facilities, fuel and gas pipeline facilities, water, water reservoirs and pipelines, sewer, and hard-wired telephone, video and data transmission facilities; provided, however, that in no event shall any such facilities be located above ground if the same is or would be visible from any portion of the Benefitted Property. To the extent permitted by law, the rights permitted in this Section 3.2.3 expressly exclude the right to install, operate or maintain any wireless telecommunication facilities within the Restricted Area which serve the general public or are operated as a commercial, for-profit use (provided, nothing contained in this Section 3.2.3 shall restrict the County from installing, operating or using wireless telecommunication facilities within the Restricted Area for the County's own, internal use).

3.2.4. <u>Existing Easements</u>. To continue the use of the Restricted Area for any purposes provided for in any easements of record existing as of the date hereof and to exercise any rights to consolidate and/or relocate any such easements.

3.2.5. <u>Flood Control</u>. To install, operate, maintain, repair and/or replace flood control, drainage facilities and ancillary and appurtenant facilities.

3.2.6. <u>Hiking and Riding Trails; Other Recreational Uses</u>. Provided that any necessary permits or other regulatory approvals are first obtained, to install, operate, maintain, repair and/or replace one or more regional and/or community hiking and/or equestrian trails, parks and related improvements and/or to engage in any other recreational uses; provided, however, that any such trails, improvements and/or uses shall be located at an elevation that is at

least ten (10) feet below and to the south and west of the existing ridgeline located on or near the boundary line between the Burdened Property and the Benefitted Property.

3.2.7. <u>Prescribed Fires: Access</u>. To conduct prescribed fires in accordance with the requirements, direction, authorization or recommendation of the local fire authority for purposes of the health and/or safety of surrounding properties and populations and otherwise to have any and all access to the Restricted Area in order to fight or otherwise mitigate fires.

3.2.8. Scientific Research. To conduct scientific research.

3.2.9. <u>Grading</u>. To perform remedial grading with revegetated slopes utilizing native species.

3.2.10. <u>Mitigation</u>. To comply with the requirements of any public agency relating to environmental, geological, hydrological, archaeological, paleontological, aesthetic and/or any other environmental or similar land use related mitigation and/or restoration measures, commitments and conditions (collectively, "Mitigation Actions") with respect to the implementation of the 2001 GDP (and any and all other such mitigation measures and activities that may so burden the Restricted Area).

3.2.11. <u>Regulatory Compliance</u>. To conduct any and all activities and operations as may be necessary to comply with Applicable Laws in connection with the landfill's closure and post-closure activities, including, but not limited to, the installation, operation, maintenance, repair and/or replacement of probes, monitoring wells and fences; provided, however, that to the extent such activities and operations can be conducted within the Landfill

Operations Area, then such activities and operations shall be conducted within Landfill Operations Area. Further, the County may install, operate, maintain, repair and/or replace such probes, monitoring wells and fences within the Restricted Area as may be necessary to comply with Applicable Laws in connection with the Landfill's operations; provided, however, that to the extent such activities and operations can be conducted within the Landfill Operations Area, then such activities and operations shall be conducted within Landfill Operations Area.

3.2.12. <u>Biological Mitigation</u>. To conduct habitat and species mitigation, conservation or other similar operations pursuant to any approved Natural Community Conservation Plan, Habitat Conservation Plan or other similar plan of conservation.

3.2.13. <u>Landfill Site Grading and/or Soil Filling</u>. To perform site grading and/or soil filling (to maximize capacity) in support of Landfill Operations so long as the same are not visible from the Benefitted Property. Such activities shall not, nor shall they be construed to, include or permit the placement of any waste or refuse within the Restricted Area.

3.3. Access. All activities and uses within the Restricted Area shall be accessed through the Burdened Property; nothing contained herein shall permit the County to access the Restricted Area by means of the RMV Benefitted Area without the express written approval of the then owner or owners of the portions of the RMV Benefitted Area over which access is desired.

4. <u>Costs of Compliance</u>. The County shall have all responsibility and shall bear all costs and liabilities of any kind related to the County's compliance with this Covenant and the Restrictions contained herein.

5. <u>Amendment</u>. This Covenant may only be modified or amended by a document recorded in the official records of Orange County, California, and signed by County (or its successor) and by those having at least 67% of the voting rights of the fee owners of the Benefitted Property. For purposes of the preceding sentence, each owner of the Benefitted Property shall have one (1) vote for each full five (5) acres of land within the Benefitted Property owned by such person (and no voting rights shall be attributed to any remainder portions of land); voting rights shall be determined on the overall acreage within the Benefitted Property owned by a single person or entity, whether such acreage consists of one or more, contiguous or non-contiguous, parcels of land. For example purposes only, if an owner owns six (6) acres within the Benefitted Property, such owner shall only be entitled to any votes.

6. Enforcement.

6.1. <u>Notice and Cure</u>. In the case of a breach of any obligation hereunder, the County shall not be deemed to be in default of this Covenant if County cures such breach within ninety (90) days of receiving notice from RMV (or any other person is entitled to enforce this Covenant) of such breach; provided, however, that if such breach is not capable of being cured within said period, no default shall be deemed to have occurred so long as County has taken material affirmative steps to cure such breach within said period and thereafter diligently and continuously prosecutes the same to conclusion.

6.2. <u>Parties Entitled to Enforce Covenant</u>. Only persons or entities who own at least five (5) acres of land within the Benefitted Property (with said acreage threshold being satisfied individually by a single owner or collectively by a group of owners) shall have the right

to enforce by proceedings at law or in equity, any and all of the Restrictions now or hereafter imposed by the provisions of this Covenant or any amendment hereto, including, without limitation, (i) the right to prevent the violation of any Restrictions, and (ii) the right to recover damages for such violation. Notwithstanding the preceding sentence, in no event shall an "association," as such term is defined in California Civil Code Section 1351 (as the same may be amended from time to time), have the right to enforce any or all of the Restrictions or this Covenant.

6.3. Non-Binding Mediation. Any dispute or claim between or among the parties with respect to this Covenant (a "Dispute") shall be subject first to a non-binding mediation prior to any party instituting litigation with regard to such Dispute. The party instituting the Dispute shall submit the Dispute to mediation pursuant to the mediation procedures adopted by the JAMS-Endispute (Orange County division) or any successor thereto or to any other independent entity providing similar services upon which the parties to the Dispute may mutually agree. No person shall serve as a mediator in any Dispute in which such person has a financial or personal interest in the result of the mediation, except by the written consent of all parties to the Dispute. Prior to accepting any appointment, the prospective mediator shall disclose any circumstances likely to create a presumption of bias or to prevent a prompt commencement of the mediation process. Within ten (10) days of the selection of the mediator, each party to the Dispute shall submit a brief memorandum setting forth its position with regard to the issues to be resolved. The mediation shall commence within ten (10) days following submittal of the memoranda to the mediator and shall conclude within fifteen (15) days from the commencement of the mediation unless the parties to the Dispute mutually agree to extend the mediation period. The mediation shall be held in Orange County or such other

place as is mutually acceptable by the parties to the Dispute. The mediation shall be non-binding upon the parties, and the mediator does not have the authority to impose a settlement on any party to the Dispute. Any admissions, offers of compromise or settlement negotiations or communications at the mediation shall be excluded in any subsequent dispute resolution forum. All of the mediation costs and expenses shall be borne equally by the parties to the Dispute unless agreed to otherwise.

6.4. <u>No Waiver; Cumulative Remedies</u>. Failure by RMV (or any other party entitled to enforce any Restrictions) to enforce any Restriction in any certain instance or on any particular occasion shall not be deemed a waiver of such right on any future breach of the same or any other Restriction by County. All rights, options and remedies of RMV under this Covenant are cumulative; and no one of them shall be exclusive of any other; and RMV shall have the right to pursue any one or all of such rights, options and remedies or any other remedy or relief which may be provided by law, whether or not stated in this Covenant.

7. <u>Captions</u>. The captions used herein are for convenience only and are not a part of this Covenant and do not in any way limit or amplify the terms and provisions hereof.

8. <u>Governing Law and Venue</u>. This Covenant shall be governed by and construed under the laws of the State of California. In the event of any legal action to enforce or interpret this Covenant, the sole and exclusive venue shall be a court of competent jurisdiction located in Orange County; California, and County shall submit to the jurisdiction of such court.

9. <u>Severability</u>. In the event that any portion of this Covenant shall become invalid, void, unenforceable or against public policy, for any reason, or shall be held by any court of competent jurisdiction to be invalid, void, unenforceable or against public policy, the remaining

portions of this Covenant shall not be affected thereby and shall remain in force and effect to the full extent permissible by law; provided, however, that if the invalidity, voiding or unenforceability would deprive any party of material benefits derived from this Covenant, then the parties shall meet and confer and shall make good faith efforts to amend or modify this Covenant in a manner that is mutually acceptable to the parties.

10. <u>Gender and Number</u>. In this Covenant (unless the context requires otherwise), the masculine, feminine and neuter genders and the singular and the plural include one another.

11. <u>Covenants to Run with the Land; Term</u>. The Burdened Property shall be held, developed, conveyed, hypothecated, encumbered, leased, rented, used and occupied subject to the Restrictions. The Restrictions are intended and shall be construed as covenants and conditions running with and binding upon the Burdened Property and equitable servitudes thereupon and every part thereof; and all and each of the Restrictions shall be binding upon and burden all persons having or acquiring any right, title or interest in the Burdened Property (during their ownership of such interest), or any part thereof, and their successors and assigns. Subject to the limitations on enforcement set forth in Section 6.2 above, the Restrictions shall inure to the benefit of the Benefitted Property, the owners thereof and their successors and assigns; and the Benefitted Property shall be deemed the dominant tenement for purposes of the Restrictions.

12. <u>Notices, Demands and Communications Between the Parties</u>. Formal written notices, demands, correspondence and communications between the parties shall be sufficiently given if delivered personally (including delivery by private courier), dispatched by certified mail, postage prepaid and return receipt requested, or delivered by nationally recognized overnight

courier service, or by electronic facsimile transmission followed by delivery of a "hard" copy to the offices of the parties indicated below. Such written notices, demands, correspondence and communications may be sent in the same manner to such persons and addresses as any party may from time-to-time designate in writing at least fifteen (15) days prior to the name and/or address change and as provided in this Section 11.

If to RMV:	Rancho Mission Viejo, LLC 28811 Ortega Highway San Juan Capistrano, CA 92675 Attn: Vice President - Operations Facsimile: (949) 248-1763
If to County:	County of Orange

Attn: Director of IWMD Facsimile: (714) 834-4183 Notices personally delivered shall be deemed to have been received upon delivery. Notices delivered by certified mail, as provided above, shall be deemed to have been given and received on the first to occur of (a) actual receipt by any of the addressees designated above as the party to whom notices are to be sent, or (b) within five (5) days after a certified letter containing such notice, properly addressed, with postage prepaid, is deposited in the United States mail. Notices delivered by nationally recognized overnight courier service (such as Federal Express) as provided above shall be deemed to have been received twenty-four (24) hours after the date of deposit. Notices delivered by electronic facsimile transmission shall be deemed received upon receipt of sender of electronic confirmation of delivery, provided that a "hard" copy is delivered by overnight courier as provided above.

320 N. Flower, Suite 400 Santa Ana, CA 92703

Integrated Waste Management Department

13. <u>Effect of Covenant</u>. This Covenant is made for the purposes set forth in the Recitals to this Covenant and RMV makes no warranties or representations, express or implied,

as to the binding effect or enforceability of all or any portion of this Covenant, or as to the compliance of any of these provisions with public laws, ordinances and regulations applicable thereto.

14. Rights of Mortgagees.

14.1. <u>No Impairment</u>. The breach of any covenants, conditions or restrictions herein contained shall not defeat, invalidate nor impair the obligation or priority of any mortgage or deed of trust now or hereafter executed and constituting a lien upon the Burdened Property or any portion thereof which is made in good faith and for value; provided, however, that any party, including the holder of the mortgage or deed of trust, who acquires title through private or judicial foreclosure, trustee's sale or deed in lieu of foreclosure (a "Foreclosure-Purchaser") and all successors and assigns of such Foreclosure-Purchaser shall take title subject to all of the covenants, conditions and restrictions contained in this Covenant with respect to such Burdened Property.

14.2. Foreclosure-Purchaser. A Foreclosure-Purchaser shall not be liable for damages arising from the breach of any Restrictions performed or which were to have been performed prior to the time such Foreclosure-Purchaser acquires title to all or any portion of the Burdened Property; provided, such Foreclosure-Purchaser shall have six (6) months after acquisition of title to all or a portion of the Burdened Property to repair or replace any improvements which were constructed in violation of any of the requirements under the Covenant.

15. <u>Authority</u>. Each party represents and warrants to the other party that it has full right, power and authority to make, enter into and perform its obligations under this Covenant; that the making and entering into this Covenant and the performance thereof will not violate any laws, ordinances, restrictive covenants or other agreements under which such party is bound; and that the undersigned officers/agents are duly authorized to execute this Covenant and to bind the party for which he/she signs.

16. <u>Understanding of Terms</u>. The parties each hereby affirm and acknowledge that they have read this Covenant, that they know and understand its terms, and have signed it voluntarily and on the advice of counsel. The parties have had a full and unhindered opportunity to consult with their attorneys, accountants, financial advisors and such other consultants, as they may have desired prior to executing this Covenant.

17. <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed this Covenant and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Covenant or any amendment or exhibits hereto.

18. **Exhibits.** All exhibits attached hereto shall be incorporated herein by reference as if set forth herein in full.

19. <u>No Third Party Beneficiaries</u>. The parties agree that no third party beneficiary to this Covenant exists and that nothing contained herein shall be construed as giving any other person or entity third party beneficiary status.

20. <u>Further Assurances</u>. The parties shall promptly perform, execute and/or deliver or cause to be performed, executed and/or delivered any and all acts, deeds and assurances, including the delivery of any documents, as either party may reasonably require in order to carry out the intent and purpose of this Covenant.

IN WITNESS WHEREOF, the parties have caused this Covenant to be executed as of the day and year first written above.

Date: October ____, 2002

COUNTY OF ORANGE

By:

Janiée V. Goss, Director Integrated Waste Management Department APPROVED AS TO FORM:

COUNT By: Geoffi Hunt

Deputy County Counsel

Date: October <u>17</u>, 2002

SAN JUAN PARTNERSHIP NO. I, a California limited partnership

By: O'NEILL PROPERTIES, LLC, a Delaware limited liability company, its General Partner

By: 🖌 Anthony R Moiso Manager

DMB SAN JUAN INVESTMENT NORTH, LLC, a Delaware limited liability company,

By: Rancho Mission Viejo, LLC, a Delaware limited liability company, its authorized agent and manager

By: Donald L. Vodra · Chief Operating Officer By: **Richard Broming** Vice President

SAN JUAN PARTNERSHIP NO. IV, a California Limited Partnership

Manage

By: O'NEILL PROPERTIES, LLC, a Delaware limited liability company, its General Partner

By: 4 Anthony Moiso

OCOLIB1\MWP\255918.12

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

State of California)
b	ss.
County of Dramie	
On 10-24-02 hefore me	Dale C. madi Notare Public
Date , Detoro inc, _	Name and Title of Onicer (e.g., "Jane Dee, Notary Public")
personally appeared Junice V.	
	Name(s) of Signer(s)
	D'personally known to me
	proved to me on the basis of satisfactory
	evidence
	to be the person(s) whose name(s) is/are
	subscribed to the within instrument and
COMM #1223106	acknowledged to me that he/she/they executed
NOTARY PUBLIC-CALIFORNIA S	the same in his/her/their authorized
My Comm. Exp. June 28, 2003	capacity(ies), and that by his/her/their
	signature(s) on the instrument the person(s), or
	the entity upon behalf of which the person(s)
	acted, executed the instrument.
	WITNESS my hand and official seal.
	Dele Con A
Place Notary Seal Above	Signature of Notary Public/
C	OPTIONAL
Though the information below is not required by and could prevent fraudulent removal	law, it may prove valuable to persons relying on the document and reattachment of this form to another document.
Description of Attached Document	
Title or Type of Document:	
	Number of Descen
Document Date:	Number of Pages:
Signer(s) Other Than Named Above:	
· · · · · · · · · · · · · · · · · · ·	
Capacity(ies) Claimed by Signer	
Signer's Name:	RIGHT THUMBPRINT OF SIGNER
🗀 Individual	Top of thumb here
Corporate Officer — Thie(s): Partner — C Limited Concerel	
□ Attorney in Fact	
Guardian or Conservator	
Guardian or Conservator Other:	
Guardian or Conservator Other:	

;

© 1997 National Notary Association • 9350 De Soto Ave., P.O. Box 2402 • Chatsworth, CA 91313-2402 Prod. No. 5907 Reorder: Call Toll-Free 1-800-876-6827

.

STATE OF CALIFORNIA COUNTY OF ORANGE

On OCTOBER 17, 2002, before me, TINA T. BAKER, a Notary Public in and for said state, personally appeared ANTHONY R. MOISO as Manager for SAN JUAN PARTNERSHIP NO. I, personally known to me to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument, the person, or the entity upon behalf of which the person acted, executed the instrument.

WITNESS my hand and official seal.

STATE OF CALIFORNIA

COUNTY OF ORANGE



Signature of Notary

State of California

My Commission Expires: August 29, 2004

On OCTOBER 17, 2002, before me, TINA T. BAKER, a Notary Public in and for said state, personally appeared ANTHONY R. MOISO as Manager for SAN JUAN PARTNERSHIP NO. IV, personally known to me to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument, the person, or the entity upon behalf of which the person acted, executed the instrument.

WITNESS my hand and official seal.

lina 2 Balcer

Signature of Notary

State of California

My Commission Expires: August 29, 2004


STATE OF CALIFORNIA COUNTY OF ORANGE

On October 17, 2002, before me, TINA T. BAKER, a Notary Public in and for said state, personally appeared DONALD L. VODRA as Chief Operating Officer and RICHARD BROMING as Vice President for DMB SAN JUAN INVESTMENT NORTH, LLC, personally known to me to be the persons whose names are subscribed to the within instrument and acknowledged to me that they executed the same in their authorized capacities, and that by their signatures on the instrument, the persons, or the entity upon behalf of which the persons acted, executed the instrument.

WITNESS my hand and official seal.



J. Baken

Signature of Notary

State of California

My Commission Expires: August 29, 2004

EXHIBIT "A"

Burdened Property Legal Description

(see following page)

EXHIBIT "A"

Burdened Property Legal Description

All that real property situated in the unincorporated territory of Orange County, California more particularly described as follows:

Parcel A

All that real property described in that certain Grant Deed in favor of said County recorded on June 29, 1973 in Book 10776, Pages 926 through 941, inclusive, of the Official Records of said County;

EXCEPTING THEREFROM those portions of the above-described property that were granted as temporary construction easements in said Grant Deed.

Parcel B

All that real property described in that certain Grant Deed in favor of said County recorded on July 1, 1974 in Book 11185, Pages 15 through 19, inclusive, of the Official Records of said County.

Parcel C

All that real property described in that certain Grant Deed in favor of said County recorded on August 6, 1975 in Book 11476, Pages 200 through 203, inclusive, of the Official Records of said County.

EXHIBIT "B"

:

County Burdened Property

(see following 8 pages)

٠,

















EXHIBIT "C"

z

RMV Benefitted Property

(see following 2 pages)



G: \10075501 \PH801 \EX080124A.DWG

	·		
CC INST. NO. (N19717'53"W PCL. 108 265.25') (N55'35'09"W 162.46') SW'LY LINE OF PER 10776 920 PCL. 108 NO. 5557 EXP. 9/30/01 CURVE DATA [] (N1520'26"W 431.74') () (Δ=6'22'30" R=1650.00	2007-07 20070508835, 0.F. LA PATA AVENUE 6, O.R. -(Δ=108'31", R=3050.00 L=60.79') PCL. 109 PCL. 109 SWLY LINE DO'L=183.58PGL. 108	P.O.B. (N68*17'04"E RAD) (N53'D6'02"E 427.79') SE'LY LINE PCL 109	
 (2) (\$\Delta = 7'35'10" R=2950.00' L=390.59') (2) (\$\Delta = 7'35'10" R=2950.00' L=390.59') (3) INDICATES RECORD INFORMATION PER INST. NO. 20010508635 O.R. THOSE PORTIONS OF PARCELS 108 & 109 OF CERTIFICATE OF COMPLIANCE NO. 2001-01 IN THE UNINCORPORATED TERRITORY OF THE COUNTY OF ORANGE, STATE OF CALIFORNIA, AS SHOWN ON THAT CERTAIN DOCUMENT RECORDED JULY 26, 2001 AS INSTRUMENT NO. 20010508635. OF OFFICIAL RECORDS, IN THE OFFICE OF THE COUNTY DESCRIPTED AS EQUI OWS: 			
BEGINNING AT THE GENERAL NORTHEASTERLY TERMINUS OF THAT CERTAIN COURSE SHOWN AS "N53'06'02"E 427.79" FOR THAT PORTION OF THE SOUTHEASTERLY LINE OF SAID PARCEL 109 AS SHOWN ON SAID CERTIFICATE OF COMPLIANCE NO. 200101; THENCE ALONG SAID SOUTHEASTERLY LINE AND THE SOUTHWESTERLY LINE OF SAID PARCEL 109 THE FOLLOWING COURSES: S53'06'02" W 427.79 FEET, AND N19'17'47" W 2585.59 FEET TO THE NORTHWESTERLY CORNER OF SAID PARCEL 109, THENCE CONTINUING ALONG THE SOUTHWESTERLY LINE OF SAID PARCEL 108, N19'17'47"W 1613.17 FEET; THENCE LEAVING SAID SOUTHWESTERLY LINE N68'12'55"E 200.02' TO THE SOUTHWESTERLY LINE OF LA PATA AVENUE AS DESCRIBED IN THE DEED RECORDED IN BOOK 10776, PAGE 926 OF SAID OFFICIAL RECORDS, THENCE ALONG SAID SOUTHWESTERLY LINE OF LA PATA AVENUE, THE FOLLOWING COURSES: S19'17'53"E 265.25 FEET, S55'35'09"E 162.46 FEET, S21'47'05"E 1208.79 FEET TO THE BEGINNING OF A CURVE CONCAVE NORTHEASTERLY HAVING A RADIUS OF 3050.00 FEET, SOUTHEASTERLY 60.79 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 1'08'31", S22'55'36"E 1411.86 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 2950.00 FEET, SOUTHEASTERLY 390.59 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 1'08'31", S22'55'36"E 1411.86 FEET TO THE BEGINNING OF A CURVE THROUGH A CENTRAL ANGLE OF 1'08'31", S22'55'36"E 1411.86 FEET TO THE BEGINNING OF A CURVE THROUGH A CENTRAL ANGLE OF 7'35'10", S15'20'26"E 431.74 FEET TO THE BEGINNING OF A CURVE THROUGH A CENTRAL ANGLE OF 7'35'10", S15'20'26"E 431.74 FEET TO THE BEGINNING OF A CURVE THROUGH A CENTRAL ANGLE OF 1'55'10", S15'20'26"E 431.74 FEET TO THE BEGINNING OF A CURVE CONCAVE NORTHEASTERLY HAVING A RADIUS OF 1650.00 FEET, SOUTHEASTERLY 183.59' ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 6'22'30" TO THE POINT OF BEGINNING. SHT 2 OF 2			
HUITT ~ ZOLLARS Huitt-Zollara, Inc. 430 Exchange, Butle 200, Irvine, CA 92602-1309 Phone (714) 734-5100 Fax (714) 734-5155 APPROVED BY	MAP AND LEGAL DESCRIPTION OF EXCEPTION AREA	SCALE 1"=800' DRAINN BY RHH CHECKED BY JRD DATE 9/16/02 JOB MO.	
	L	10-0/35-01	

....

.

G: \10075501 \MAP75501DWG/LL075506.DWG

ATTACHMENT 2

No-Build Area

•



1.3

G: \10075501 \PH801 \EX080125A.DWG



G: \10075501 \PH801 \EX080125A.DWG



١

G: \10075501\PH801\EX080125A.DWG



G: \10075501\PH801\EX080125A.DWG



G: \10075501 \PH801 \EX080125A.DWG

ATTACHMENT 3

Prior Agreement

OFFICE COPY

7 GA 126-1.01, 1.1 and 1.2 Refuse Disposal Station 26 8 Prima Deshecha Canada

9 X2-1, 1.1-1.6 La Pata Avenue 10

1 2

3

5 6

11

19

. .

.# A

.01 23

PURCHASE AND OPTION AGREEMENT

12 THIS AGREEMENT is made and entered into this 13^{-4} day of -12.251973, by and between the CCUNTY OF ORANGE, a political subdivision of th 13 State of California, hereinafter referred to as "COUNTY", and CROCKER NATIONAL BANK, a national banking association, formerly Crocker-Citisens National Bank, successor to Citizens National Bank, formerly known as 14 Citizens National Trust and Savings Bank of Los Angeles, as Trustee, under Decree of Final Distribution, had in the matter of the Estate of Jerome O'Neill, deceased; Alice O'Neill Avery, and Richard Jerome 15 16 O'Neill, as Trustees under two separate Declarations of Trust dated April 18, 1968, for the respective benefits of Alice O'Neill Avery and Richard Jerome O'Neill, hereinafter referred to as "OWNER". 17 18

RECITALS

1. The Forster Canyon Disposal Station, which serves the southeasterly section of the County, is presently near its maximum capacity. 20

2. A study was prepared by the Refuse Disposal Division of the County 21 || Road Department to locate a replacement site for the existing refuse 22 disposal station.

3. As a result of the above study, COUNTY has determined to acquire 23 the 1,524-acre Prima Deshecha Canada site, shown on the map marked "EXHIBIT A", attached hereto and made a part hereof, for a refuse 24 disposal station. 25

4. While COUNTY desires to acquire OWNER'S portion of said site through acquisition agreement, because of the pressing need and lack of alternati 26 sites, owner has been advised that COUNTY would proceed to acquire the subject property through eminent domain, if a voluntary sale thereof 27 by OWNER could not be effected. 28

5. COUNTY'S Board of Supervisors made a formal determination of the necessity for public use of the subject property as a refuse disposal 29 station by adopting a condemnation resolution, No. 73-556, on May 22, 1973, and has filed an action in eminent domain covering the subject 30 property. 31

6. OWNER owns approximately 945 acres of the proposed refuse disposal station site, shown as Parcels GA 126-1.01, 1.1, and 1.2 on "EXHIBIT 32 B', attached hereto and made a part hereof.

7. OWNER also owns approximately 46 acres which will be required for 1 the construction of the La Pata Avenue access road to the proposed sit The road right of way to be acquired, designated as Parcels X2-1, 1.1-is shown on the map marked "EXHIBIT C", attached hereto and made a par 2 3 8. COUNTY desires at this time to acquire only Parcels GA 126-1.01 and 4 the road right of way designated as Parcels X2-1, 1.1-1.6. 5 OWNER agrees to provide COUNTY with the option to purchase Parcels GA 126-1.1 and 126-1.2, described in "EIHIBIT D", attached hereto and 6 made a part hereof. 7 10. Concurrently with the execution of this Agreement, OWNER has executed and delivered to Stephen L. Salyer, Real Property Agent for COUNTY, a 8 Grant Deed conveying OWNER'S fee interest in Parcels GA 126-1.01 and X2-1, 1.1-1.4/ and a Temporary Easement Deed for construction purposes 9 over Parcels X2-1.5, 1.6 to COUNTY. The second 10 NOW, THEREFORE, it is agreed by and between the parties hereto as follo 11 I. 12 PURCHASE OF PARCELS GA 126-1.01 AND 13 X2-1, 1.1-1.6 14 COUNTY will: A. 15 1. Pay to the OWNER the sum of One Million Five Hundred Thousand Dollars (\$1,500,000) in cash, within thirty days after: 16 a. Fee title to Parcels GA 126-1.01 and X2-1.1, 1.2, 1.3, 17 and 1.4 has vested in the COUNTY in accordance with the terms and conditions of Clause IB below; 18 The temporary construction easements designated as Parcels Ъ. 19 X2-1.5, 1.6 have vested in the COUNTY in accordance with the terms and conditions of Clause IB below. 20 2. Forthwith order a title policy of insurance covering the subjec 21 acquisition from The TI Corporation (of California) for the full amount of the consideration. 22 в. OWNER will: 23 1. Do all things necessary to vest in COUNTY: 24 a. Adequate title for construction purposes to the temporary 25 construction easements designated as Parcel X2-1.5, 1.6; 26 b. Fee title to Parcels GA 126-1.01 and X2-1, 1.1-1.4 (and upon exercise of option, fee title to Parcels GA 126-1.1, 1.2); 27 Title to be vested in COUNTY free and clear of all liens, 28 encumbrances, assessments, easements, agreements, licenses, leases (recorded or unrecorded), and taxes except: 29 (1) Any installment of General and Special County and 30 City taxes, if any, that is not entirely earned at the time title is vested in the COUNTY, and all taxes subsequent 31 thereto. 32 (However, OWNER shall remain liable for payment of any 04.02 .123 -2-

earned taxes, pursuant to the provisions of Section 4986. 1 California Revenue and Taxation Code, as amended, and the COUNTY is authorized to pay, from the amount shown 2 in Clause IA-1 above (and upon exercise of option, the amount shown in Clause IIA-4 below) any earned delinquent 3 taxes, together with penalties and interest thereon, delinquent or non-delinquent assessments or bonds). 4 5 (2) Easement for telephone and telegraph purposes as gran to the Southern California Telephone Company by Deed reco. 6 in Book 1121, Page 423, of Official Records. 14 7 (3) An easement 150 feet in width as granted to San Diego Gas and Electric Company, as recorded in Book 7308, Page 8 1 374, of Official Records. 6 Q. J. 9 ((4) An exclusive easement 250 feet in width as granted to Southern California Edison Company as recorded in Book 748 10 Page 714, of Official Records. ot 11 (5) An easement 10 feet in width as granted to San Diego Pipeline Company as recorded in Book 7894, Page 222, of 12 13 (6) Terms, covenants, conditions and restrictions set forth in agreement elective to the Rancho Mission Viejo Agricul-14 tural Preserve as recorded in Book 8884, Page 206, of Official Records. 15 (7) Mineral rights reserved in the Deed from OWNER to COUNT 16 covering Parcels GA 126-1.01 and X2-1, 1.1-1.4 (and upon exercise of option, Parcels GA 126-1.1, 1.2) below the 17 depth of 300 feet below sea level, excluding all rights 18 Pay for real property transfer taxes as required by law, cost 2. 19 of the policy of title insurance from The TI Corporation (of California), and all costs not specified to be borne by COUNTY 20 necessary to convey title to COUNTY as herein specified. 21 Warrant that there are no oral or written leases on any portion 3. of Parcels GA 126-1.01, or X2-1, 1.1-1.6 exceeding a period of 22 thirty (30) days and OWNER agrees to hold COUNTY harmless and reimburse COUNTY for any and all of its losses and expenses 23 occasioned by reason of any lease of said property held by any tenant of OWNER for a period exceeding thirty (30) days. 24 •,... C. It is mutually understood and agreed: 25 1.15 1. COUNTY shall forward a copy of the final construction plans for 26 the La Pata Avenue access road to OWNER for review prior to initiation of construction of said access road. 27 2. COUNTY shall prepare deeds conveying the necessary slope case-28 ments for the La Pata Avenue access road from OWNER to COUNTY, upon 29 completion of the construction of said road. Said slope easements shall cover the actual slope areas, as constructed for said 30 \hat{r} : road. . . Said deeds shall be forwarded to OWNER within thirty (30) days 31 following filing of a Notice of Completion for the access road 32 1.03 :23 -3-

OWNER shall execute and return said deeds to COUNTY within thirty (30) days of their receipt. OWNER'S execution of said deeds shall not be unreasonably withheld.

3. The twelve (12) foot structural steel plate pipe previously planned as a cattle undercrossing under the La Pata Avenue access road, is hereby deleted from the construction plans. It is furthe mutually understood between the parties hereto that the compensati herein provided for includes, but is not limited to, any and all damages that may accrue to the OWNER'S remaining property as a result of the deletion of the subject cattle undercrossing.

4. The undersigned GRANTOR hereby agrees and consents to the dismissal of any eminent domain action which may be filed by the COUNTY against said Parcels GA 126-1.01, 1.1, 1.2 and X2-1, 1.1-1. and also waives any and all claims to any money that may be deposi in the Superior Court in said action.

5. OWNER owns extensive lands which abut upon the proposed dispo site and access road. Such abutting lands, commonly known and designated as the "Rancho Mission Viejo", may suffer a substantial decline in value in the event COUNTY'S landfill operating practice: on the subject lands are not conducted in accordance with certain minimum standards. It is, therefore, mutually understood and agreed that the conveyance by OWNER to COUNTY of the proposed disposal site, together with the conveyance by OWNER to COUNTY of the temporary construction easements, permanent slope easements, and the La Pata Avenue access road are, and each of them is, subject to the following equitable servitudes:

a. Access Road Requirements

(1) A minimum thirty (30) foot paved access road constructed of asphalt over aggregate base shall be provided by COUNTY. The connection to Ortega Highway shall conform with Orange County Road Department and State Division of Highway requirements.

(2) Control of dust and litter along the access road shall the responsibility of COUNTY and shall be accomplished by litter collection not less than once each week.

(3) A four (4) foot high hog wire fence, or equal, topped by two strands of barbed wire for a total height of approximately six feet, shall be constructed along the access road property line to constrain litter.

b. Site Requirements

(1) Control of litter blowing from the site and unauthorized landfill site access shall be accomplished by the construction and maintenance, around the working landfill area, of a four foot high hog wire fence, or equal, topped by two strands of barbed wire for a total height of approximately six feet.

A second fence shall be constructed around the entire property consisting of five strands of barbed wire, or equal, for a total height of approximately six feet.

Gates in landfill fence are to be kept closed and locked during hours when the landfill is not open.

-4-

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

04.04 .123

 $= \{ i_{i}, j_{i}\}$

(2) Adequate water shall be applied to solid waste and cov 1 soil as required to aid in preventing dust and litter migr 2 tion beyond the landfill property boundary. 3 (3) During the life of the landfill operation, COUNTY shal. be responsible for the removal of litter originating in the landfill and blowing onto OWNER'S adjacent property provide 4 that OWNER, or the appropriate possessory interest, provide 5 6 (4) The existing ridge line is to be maintained as a buffer zone and barrier to inhibit viewing, noise, dust, litter, 7 and other effects of landfill-associated operations from adjacent property. All landfill site grading, outting, 8 filling, and construction of improvements shall be done at least fifty (50) vertical feet below the existing ridge 9 line, except when grading above this level as necessary to stabilize the ridge. 10 (5) All on-site roads shall be at least fifty (50) vertical 11 feet below the aforementioned ridge line. A 20-foot paved roadway shall be installed and maintained within the 12 actual fenced disposal site area for a minimum of 300 feet or until the road reaches a point fifty (50) vertical 13 feet below the ridge line. 14 (6) Vector control for rats, flies, birds, noxious insects, and larva shall be provided so that no increase in popula-15 tions of said species occurs on off-site property due to landfill operations. A minimum of 6 inches of soil 16 covering shall be placed on the top of each day's deposit of refuse. Records of vector inspections and resulting 17 actions are to be available to OWNER. 18 (7) An adequate gas control and monitoring program shall be initiated upon start of landfilling to prevent gas 19 migration onto OWNER'S adjacent property. Gas sampling and testing shall be done at least semi-annually and 20 <u>.</u>. the results transmitted to OWNER. 21 (8) Adequate surface drainage and subdrainage systems shall be installed to prevent entry of rainfall runoff 22 and groundwater into the solid waste fill. 23 ·.. • (9) COUNTY agrees to limit operation of the refuse disposal station to the hours between 6:00 m.m. and 6:00 p.m. except 24 - - - - in the case of emergency situations. 25 (10) OWNER shall at all reasonable times have the right to inspect the subject property and refuse disposal operations. 26 c. COUNTY does hereby covenant that it will conduct its land-27 fill operating practices in accordance with the foregoing requirements. These requirements, being equitable servitudes appurtemant to and for the express benefit of OWNER'S said 28 abutting lands, shall at all times be binding upon, and shall 29 be enforceable against COUNTY, its successors and assigns, or 30 against any and all persons, corporations, or other entities $\frac{1}{2}$ who should at any time hereafter become the owner of the lands 31 and interests so conveyed, by OWNER, its successors and assigns, either at law or in equity. 32)4.05 It is mutually understood and agreed that each of such conveyances, including the conveyances of the option parcels hereinafter provided for, in the event said options be exer--5-

Nr. Nr.

cised, shall, by appropriate reference incorporate therein these equitable servitudes.

6. All notices pursuant to this Agreement should be addressed as set forth below or as either party may hereinafter designate by written notice and shall be sent through the United States Mail:

TO: COUNTY

 $\{ y_{i}, \dots, y_{i} \}$

1

2

3

4

5

6

7

8

- 9

10

11

13

14

15

16 17

18

19

20 21

22

23

24

25

26

27

28

29

30

31

32

150

.

80,

. . .

÷.

· · · ;

5. . . .

1 ...i

 \sim

TO: OWNER

County of Orange Department of Real Property Services 400 Civic Center Drive West Santa Ana, California 92701

Rancho Mission Viejo Post Office Box 2035 Mission Viejo, California 92675

II

OPTION TO PURCHASE

A. OWNER hereby grants to COUNTY the right and option to purchase 12 Parcels GA 126-1.1 and 1.2, shown on the map marked "Exhibit B" and described in "Exhibit D". This option is granted on the following terms:

1. COUNTY'S right to exercise the option for Parcel GA 126-1.1 shall expire one year from the date of this Agreement.

2. COUNTY'S right to exercise the option for Parcel GA 126-1.2 shall be contingent upon its having exercised the option for Parcel GA 126-1.1 and shall expire two years from the date of this Agreement

3. An option shall be considered exercised when COUNTY, through its Director of Real Property Services, has mailed a signed, written notice to OWNER of the COUNTY'S Board of Supervisors' intention to exercise the option.

4. The purchase price upon exercise of these options shall be One Million Dollars (\$1,000,000) in cash for Parcel GA 126-1.1 and Three Hundred Thirty-Six Thousand Dollars (\$336,000) in cash for Parcel GA 126-1.2.

- B. Upon Exercise of Option:
 - 1. COUNTY will:

a. Forthwith prepare and forward to OWNER a grant deed conveying to COUNTY the fee interest in the subject option parcel, as described in "EXHIBIT D".

b. Pay to OWNER a sum as set forth in Clause IIA-4 above within thirty (30) days after fee title to the subject option parcel has vested in COUNTY in accordance with the terms and conditions of Clause IB above.

c. Forthwith order a title policy of insurance covering the subject acquisition from The TI Corporation (of California), for the full amount of the consideration,

2. OWNER will:

a. Execute and deliver to COUNTY the grant deed, prepared by

04.06 .123

-6-

COUNTY, conveying the fee interest in the subject option parcel to COUNTY, within thirty (30) days following receipt of said grant deed from COUNTY.

04.07

b. Do all things necessary to vest fee title in COUNTY to the subject option parcel, as specified in Clause IB.

c. Pay for real property transfer taxes as required by law, cost of the policy of title insurance from the TI Corporation (of California), and all costs not specified to be borne by COUNTY necessary to convey title to COUNTY as herein specified.

d. Warrant that there are no oral or written leases on any portion of Parcels GA 126-1.1, 1.2 exceeding a period of thirty (30) days, and OWNER agrees to hold COUNTY harmless and reimburs COUNTY for any and all of its losses and expenses occasioned by reason of any lease of said property held by any tenant of OWNER for a period exceeding thirty (30) days.

-7-

1 CROCKER NATIONAL BANK, as Trustee u Decree of Final Distribution in the 2 matter of the Estate of Jerome O'Ne deceased 3 4 By lister/ in Vice President 5 6 Į. By Ú. Trust Officer 7 8 ALICE O'NEILL AVERY and RICHARD JERC O'NEILL, as Trustees under a Declara tion of Trust dated April 8, 1968, f the benefit of Alice O'Neill Avery 9 10 11 Cheil aven By ALICE O'NEILL AVERY 12 13 1hr. RICHARD JEROME O'NEILL 14 15 RICHARD JEROME O'NEILL and ALICE O'NI APPROVED AS TO FORM: AVERY, as Trustees under a Declaratic 16 Adrian Kuyper of Trust dated April 18, 1968, for th benefit of Richard Jerome O'Neill Bylichian Schule County Counsel 17 Į (ĊĈ. Вy 18 RICHARD JEROME O'NEILL 19 RECOMMENDED FOR APPROVAL: 20 By <u>Glip</u> <u>Chell</u> L. McConville ý. þ Road Commissioner and Coupt 21 Surveyor 22 ZHA CAUA 23 Stanley E. Krause, Director CONFORMED Dept. of Real Property Services 24 Stenken Вy 25 Real Property 26 Director Βv 27 Assistant 28 ATTEST: 29 W. E. ST JOHN, County Clerk of the County of Orange and ex 30 officio Clerk of the Board of COUNTY OF ORANGE 31 Supervisors of Orange County Seel By<u>c</u> Deputy Ronald W. Caspers 32 S/d June Alexander Вy Chairman, Board of Supervisors 4.08 123





	PRIMA DESHECHA	
NUNE	14.01 (4) 6-2126	HEVELED
4-17-7	3 160 00 2450	
	IIY 31 4002	

THIS IS NOT A SURVEY OF D LAND BUT IS COMPLED FROM DAT SHOWN BY OFFICIAL RECORDS AN SURVEY DATA ON FILE IN THE OFFIC










Proj	REFUSE DISPOS
·• -•	STATION #26
Project No.	GA 125-1.1
Date:	4-12-73

GRANT DEED

That portion of the Rancho Mission Viejo or La Paz as shown on a map recorded December 19, 1867, in Book 1, Pages 63 and 64 of Patents, Records of Los Angeles County, California, described as follows:

Commencing at a 2" Iron Pipe in the Southwesterly line of sai Rancho Mission Viejo or La Paz as shown on a map filed in Book 92, Page 15, Records of Surveys, Records of Orange County, California, distant thereon S. 19° 17' 42" E., 7376.12 feet from the Rancho Mission Viejo Corner No. 3 as shown on said Record of Survey map; thence along said Rancho line S. 19" 17' 42" E., 2650.00 feet; thence leaving said Rancho line following along a line which passe: through a point in the boundary line of the land shown on said Record of Survey Map distant thereon S. 15° 17' 51" E., 2517.46 feet from the Northerly terminus of that certain course in said boundary line shown as "N. 15° 17' 51" W., 3639.95 feet", N. 70° 52' 17" E., 4303.81 feet to the TRUE POINT OF BEGINNING; thence S. 19° 07' 36" E., 3735.21 feet to the boundary line of the land shown on said Record of Survey map; thence along said boundary line through the following described courses: S. 39° 31' 40" W., 2087.60 feet; thence N. 39° 02' 02" W., 2591.37 feet; thence S. 35° 32' 42" W., 1577.06 feet; thence N. 66° 09' 36" W., 468.95 feet to said Southwesterly line of the Rancho Mission Viejo or La Paz; thence along said Rancho line N. 19° 17' 42" W., 2976.85 feet to the Westerly terminus of the aforedescribed course having a bearing of "N. 70° 52' 17" E."; thence along said course N. 70° 52' 17" E., 4303.81 feet to the TRUE POINT OF BEGINNING.

The aforedescribed parcel is included within the boundary of the land shown on a map filed in Book 92, Page 15, Records of Surveys, Records of said Orange County.

APPROVED AS TO DESCRIPTION L. McCONVILLE BOAD COMMISSIONER & COUNTY SURVEYOR 9-1/10

Deputy County Surveyor

Proje REFUSE DISPOSAI STATION #26 Project No. GA 126-1.2 Date: 4-12-73

GRANT DEED

That portion of the Rancho Mission Viejo or La Paz as shown on a map recorded December 19, 1867, in Book 1, Pages 63 and 64 of Patents, Records of Los Angeles County, California, described as follows:

Commencing at a 2" Iron Pipe in the Southwesterly line of said Rancho Mission Viejo or La Paz as shown on a map filed in Book 92, Page 15, Records of Surveys, Records of Orange County, California, distant thereon S. 19º 17' 42" E., 7376.12 feet from the Rancho Mission Viejo Corner No. 3 as shown on said Record of Survey map; thence along said Rancho line S. 19° 17' 42" E., 2650.00 feet; thence leaving said Rancho line following along a line which passes through a point in the boundary line of the land shown on said Record of Survey map distant thereon S. 15° 17' 51" E., 2517.46 feet from the Northerly terminus of that certain course in said boundary line shown as "N. 15° 17' 51" W., 3639.95 feet", N. 70° 52' 17" E., 4303.81 feet to the TRUE POINT OF BEGINNING; thence S. 19° 07' 36" E., 3735.21 feet to the boundary line of the land shown on said Record of Survey map; thence along said boundary line through the following described courses: N. 39° 31' 40" E., 1138.79 feet; thence N. 18° 16' 46" E., 1113.08 feet; thence N. 31° 26' 35" W., 1165.59 feet; thence N. 15° 17' 51" W., 1122.49 feet to the Easterly prolongation of the aforedescribed course described as "N. 70° 52' 17" E., 4303.81 feet"; thence along said prolongation S. 70° 52' 17".W., 1475.36 feet to the TRUE POINT OF BEGINNING.

The aforedescribed parcel is included within the boundary of the land shown on a map filed in Book 92, Page 15 of Records of Surveys, Records of said Orange County.

APPROVED AS TO DESCRIPTION L. McCONVILLE BOAD COMMISSIONER & COUNTY, SURVEYOR 1.00 County Surveyo

ATTACHMENT 4

RMV Benefitted Property

.



G: \10075501 \PH801 \EX080124A.DWG



G: \10075501 \MAP 75501DWG/LL075506.DWG

ATTACHMENT 5

Resolution No. 01-380

.

	1
	2
	3 RESOLUTION OF THE BOARD OF SUPERVISORS OF
	4 ORANGE COUNTY, CALIFORNIA
	5 November 6, 2001
	6 WHEREAS, Draft Environmental Impact Report No. 575 (DEIR 575) was prepared for the 2001 Prin Deshecha Landfill General Development Plan to address the environmental effects, mitigation measures, an project alternatives associated with the proposed project; and
. •	8 WHEREAS, on March 29, 1999, the County issued a Notice of Preparation/Initial Study for the 200 Prima Deshecha General Development Plan EIR, and caused that Notice of Preparation to be distributed to a responsible agencies, trustee agencies and interested parties; and
1 (1)	WHEREAS, pursuant to the initial study prepared for the 2001 Prima Deshecha General Developmen Plan EIR, and comments received in response to the Notice of Preparation, the County prepared Dra Environmental Impact Report 575 (State Clearinghouse Number 99041035); and
12	WHEREAS, a Notice of Completion for this DEIR was filed on February 5, 2001, giving public notic of the availability of the DEIR for review and comment; and
13	WHEREAS, public comments have been received on the Draft EIR, and responses to those comment
14	have been prepared and provided to the Board of Supervisors in a separately bound document entitled "Fina Program Environmental Impact Report - Responses to Public Comments" (the "Responses to Comments"); and
16	WHEREAS, consistent with CEQA Guidelines Section 15132, the DEIR and appendices, the Response to Comments and IWMD Staff Reports to the Planning Commission and Board of Supervisors, including al minutes, transcripts, attachments, incorporation, and references, comprise the proposed Final Environmenta Impact Report for the Project, and contain all information specified by that CEQA Guideline; and
. 18 19	WHEREAS, Memoranda of Understanding regarding the Prima Deshecha Landfill between the County of Orange and the Cities of San Juan Capistrano and San Clemente were signed by the County on September 12. 1995 and July 1, 1997, respectively, to ensure the implementation of mitigation measures; and
20 21	WHEREAS, the IWMD prepared a proposed Final Responses to Comments document for the Planning Commission for review as an advisory body to the Board of Supervisors on such matters; and
22	WHEREAS, a noticed public hearing for October 9, 2001 was continued to October 24, 2001, to receive and consider public testimony with respect to the Final EIR; and
23	WHEREAS, the Planning Commission acted in conformity with the IWMD recommendation that it
24	WHEREAS the Blancing Comparison in the proposed Final EIR be certified as adequate; and
25	and
26	
27	// RESOLUTION NO 01-380
28	Approve 2001 General Development Plan for Prima Deshecha Property
	1

1

	1 WHEREAS	, a mitigation monitoring and reporting program has been ciraited to meet the requirements
	2 to ensure compliance	e with project changes imposed and mitigation measures imposed to avoid or substantia
	3 included in the Final	EIR and incorporated herein by reference, defines the following for each mitigation measu
	4 1. <u>Method and</u> review of ev	Timing of Verification - In each case, a method and time for verification of the mitigation,
	5 selected are a addressed an	designed to ensure that impact related components of project implementation are adequate d do not proceed without establishing that the mitigation is assured
	6 2. Responsible	Person - In each case, a public official is assored in the mining the
	 7 ensuring that inadvertently 8 who grants the 	t the mitigation is carried out. To guarantee that the mitigation measure as responsible f overlooked in connection with the issuance of a later permit, the supervising public offici e permit called for in the performance is named.
	9 3. <u>Definition of</u>	Mitigation - In each case (except where a mitigation, such as a gentechnical report is
1	10 known proce 10 mitigation, eit be taken in mi	dure or commonly understood term), the mitigation measure contains the criteria report is a well there in the form of adherence to certain adopted regulations or identification of the steps t tigation; and
I	WHEREAS,	Section 21081 of CEOA and Section 15091 of the Cuidadian
l	12 Supervisors make one completed, identifying	or more of the following findings prior to approval of a project for which an EIR has been one or more significant effects of the project along with statements of f
1	13 each finding:	sector of the project along with statements of facts supportin
14	14 <u>Finding 1</u> - Ch substantially le	hanges or alterations have been required in, or incorporated into, the project that avoid o seen the significant environmental effect as identified in the final EIR.
. 16	16 Finding 2 - Su agency and not can and should	ch changes or alterations are within the responsibility and jurisdiction of another public the agency making the finding. Such changes have been adopted by such other agency of be adopted by such other agency.
	Finding 3 - Spe	cific economic, legal, social, technological, or other considerations, including provision of
18	8 employment op alternatives ider	portunities for highly trained workers, make infeasible the mitigation measures or project atified in the final EIR.
19	NOW THEREFORE,	BE IT RESOLVED THAT:
20	1. The Board of Su	pervisors hereby certifies the Final EIR prepared for the Prima Deshecha Leadfill and inter-
21	as complete and approval of a C	adequate in that it addresses all environmental effects of the proposed project, including ieneral Development Plan, and fully complies with the requirements of the California
22	following element	Quality Act and State CEQA Guidelines. The Final EIR will be composed of the
23	a. Draft EL	R;
24 26	b. Technica	I Appendices to the Draft EIR;
23	c. Commen	ts received on the draft EIR and responses to these comments:
20	d. Integrated	HWaste Management Department staff report dated October 9, 2001
27	e. Planning	Commission minutes;
28	//	
		7
	••	- *

-

	I	f.	Board of Supervisors staff report, Agenda Item Transmittal, resolution, and minutes;
	2	g	All attachments, incorporations, and references delineated in a. through f, above,
	3	A O	of the above information referred to in this resolution has been or will be on file with the Countyinge Integrated Waste Management Department, 320 North Flower Street, Santa Ana, California.
	5	2. T w fin St he	Board of Supervisors makes the findings contained in the attached Statement of Findings and Fa h respect to significant impacts identified in the Final EIR and find that each fact in support of lings is true and is based upon substantial evidence in the record, including the Final EIR. T ement of Findings and Facts is attached hereto and incorporated into Exhibit A and incorpora in by this reference.
	8	5. Th th:	Board finds that the Final EIR has identified all significant environmental effects of the project a there are no known potential environmental impacts not addressed in the Final EIR.
9		Th Fa	Board finds that all significant effects of the project are set forth in the Statement of Findings a s and the Final EIR.
11	5	. Th fea the	Board finds that the Final EIR has described all reasonable alternatives to the project that cou- bly obtain the basic objectives of the project (including the "No Project" Alternative), even whe alternatives might impede the attainment of project objectives and rnight be more costly.
13	6	. Thi and	Board finds that no substantial evidence has been presented which would call into question the fac onclusions in the EIR.
14	7.	The Coc	Board finds that no significant new information has been added to this EIR pursuant to Californ of Regulations section 15088.5 such that recirculation for additional public review is necessary.
16 17	8.	Thi: occi avoi "Mi	Board finds that, although Final EIR 575 identifies certain significant environmental effects that ma if a General Development Plan is approved, all significant effects that can feasibly be mitigated of ed have been reduced to an acceptable level by the imposition of mitigation measures set forth in th gation Monitoring and Reporting Program," attached hereto and marked as Exhibit B.
18 19 20	9.	This Gen proc 2108	Board finds that the "Mitigation Monitoring and Reporting Program" for the 2001 Prima Deshech al Development Plan, attached hereto and marked as Exhibit B, establishes a mechanism and lures for implementing and verifying the mitigations pursuant to Public Resources Code section .6.
21 22	10.	This Gene mitig to or	Board adopts the "Mitigation Monitoring and Reporting Program" for the 2001 Prima Deshecha al Development Plan, attached hereto and marked as Exhibit B and made a part hereof. These tion measures shall be incorporated into the 2001 Prima Deshecha General Development Plan prior oncurrent with project implementation.
23 24 25 26	11.	This Deve have by the unave 2001 Consi	oard finds that the unavoidable significant adverse effects of the 2001 Prima Deshecha General pment Plan project identified in Exhibit A attached hereto (Statement of Findings and Facts), that of been reduced to a level of less than significant have been substantially lessened in their severity imposition of the mitigation measures identified in Exhibit B. This Board finds that the remaining lable significant impacts are clearly outweighed by the economic, social, and other benefits of the rima Deshecha General Development Plan project, as set forth in the "Statement of Overriding erations," attached hereto as Exhibit C and made a part hereof.
27 28	12.	This signifi docum part he	bard adopts the recitation of overriding considerations, notwithstanding certain unavoidable ant environmental effects which cannot feasibly be substantially mitigated as set forth in the at entitled "Statement of Overriding Considerations," attached hereto as Exhibit C and made a cof.
			-3-

•••

	1 13. This Board finds that Final EIR 575 reflects the independent review and judgment of the County Orange.
2	14. This Board finds that Final EIR 575 serves as adequate and appropriate environmental documentation f the Proposed 2001 Prima Deshecha General Development Plan Project.
٤	BE IT FURTHER RESOLVED THAT this Board of Supervisors hereby certifies Final EIR 575
5	Deshecha General Development Plan Project and fully complies with the requirements of the CEQA Statute, the State CEQA Guidelines, and the County's environmental analysis procedures. This Board of Supervisors beset
G	approves the 2007 Prima Deshecha General Development Plan with the following conditions:
7	I. "No later than ninety (90) days following approval of the 2001 General Development Plan (GDP) by the Orange County Board of Supervisors, the Integrated Waste Management Department (IWMD). Discussional of the second s
8	in cooperation with the City Manager of the City of San Juan Capistrano, will convene a Task Force t determine the feasibility of minimizing traffic noise associated with haul trucks on that segment of Or
9	Highway between Rancho Viejo Road, and the easterly City limit. The purpose and responsibility of th Task Force will be to develop within a 12-month period from date of formation, a landfill traffic nois management program shall include, but not limited to the following:
11	• retrofitting heavy commercial, trucks which haul waste to the landfill with noise damping accompanies
12	to eliminate the metal-on-metal impact noise and rattling associated with such trucks.
13	 periodic road inspection, and repair at the cost of IWMD for La Pata Avenue (between Orteg: Highway and the landfill) and Ortega Highway (between Rancho Vieio Road and the easterly air
14	traffic and maintaining an even, consistent road surface.
15	 work with CalTrans on implementing prohibition on the use of "Jake" brakes along La Pata Avenue (between Ortega Highway and the landfill) and Ortega Highway (between Contega Highway)
16	the easterly city limit)."
17	2. This Board approves in concept the alignment for La Pata as proposed, with the understanding that future modifications to the alignment may be necessary. Any future modifications to the alignment may be necessary.
18	defined through coordination with the City of San Clemente, and approval subject to subsequent CEQA analysis and documentation.
19	3. Following approval by the Orange County Board of Supervisors of the 2001 prime price in the
- 20	Development Plan (GDP), and prior to any grading of landfill Zone 4, the Integrated Waste Management Department (IWMD) Director, in cooperation with the City Management the City of State
21	prepare a Zone 4 Viewshed Protection Plan (VPP) for viewpoints, identified by the City Manager, City of San Clemente. The preliminary public viewpoints are: A – the approximate contraction of the city of the ci
22	School site at the corner of La Pata and Hermosa; $B - K-8$ school site; and $C - Salida$ intersection with unnamed street. The VPP will involve a landscaping plan landform or contour and in a school site.
23	implementation schedule, mutually agreed to by the City and County.
24	IWMD will review and consult with the City of San Clemente regarding traffic, water quality, viewsheds, and landscape mitigation whenever IWMD reviews and undates the Prime Dechashe Salid W
25	Facilities Permit.
26	
27	//
28	
11	

For the purpose of protecting San Diego Gas & Electric (SDG&E) facilities and access to those facilities I prior to any grading, development of projects, or establishment of traits on Prima Deshecha proper which require encroachment into the SDG&E right-of-way, or which affect those facilities or acce 2 thereto, IWMD will coordinate with Sempra Energy. Implementation of those elements of the plans th would affect SDG&E facilities will comply with SDG&E requirements pertaining to the design 3 operation of the project to protect SDG&E facilities, including recovery of any cost from the County Orange from any temporary or permanent relocation of facilities and/or temporary outages. 4 5. Following approval by the Orange County Board of Supervisors of the 2001 Prima Deshecha GDP, prior 5 to any grading of landfill Zone 4, or at an appropriate time in the planning entitlement process for the RMV property to be developed to the north and east of the Prima Deshecha property - such time to be 6 determined by the Director of Planning for the County of Orange - the Integrated Waste Management Department (IWMD) Director, will conduct a Zone 4 viewshed analysis for public viewpoints from the 7 RMV property identified by the Director of Planning. Depending on the exposure of the landfill Zone 4 to the RMV development, a Viewshed Protection Plan (VPP) may be required. If required, the VPP will 8 be prepared by the IWMD Director and may involve a landscaping plan. a landform or contour grading plan, and implementation schedule, mutually agreed to by the Director of Planning and the IWMD 9 10 6. Representatives of the County Planning Department will, as soon as may be practical, meet with Rancho Mission Viejo (RMV) to discuss any long range proposals RMV has for development of the property it 11 owns adjacent to the north-eastern edge of the Landfill. The purpose of such discussions shall be to work towards development of a mutually acceptable means to plan for compatibility between landfill activities 12 conducted by the County in Zone 4 and potential future development of the property adjacent to the 13 Landfill owned by RMV. 14 15 16 17 ${}^{\prime\prime}$ 18 11 19 11 20 // 21 $^{\prime\prime}$ 22 11 23 II24 \parallel 25 \parallel 26 27 // 28

The foregoing was passed and adopted by the following vote of the Orange County Board of Supervisors, on November 06, 2001, to wit:

AYES:

Supervisors:

THOMAS W. WILSON, TODD SPITZER, CHARLES V. SMITH JAMES W. SILVA, CYNTHIA P. COAD

NOES: EXCUSED: ABSTAINED:

Supervisor(s): Supervisor(s): Supervisor(s):

cynthia P. Cond

STATE OF CALIFORNIA

COUNTY OF ORANGE

I, DARLENE J. BLOOM, Clerk of the Board of Orange County, California, hereby certify that a copy of this document has been delivered to the Chair of the Board and that the above and foregoing Resolution was duly and regularly adopted by the Orange County Board of Supervisors.

IN WITNESS WHEREOF, I have hereto set my hand and seal.



DARLE Clerk of the Board

County of Orange, State of California

Resolution No:	01-380
Agenda Date:	11/06/2001
Item No:	56



I certify that the foregoing is a true and correct copy of the Resolution adopted by the Board of Supervisors, Orange County, State of California

DARLENE J. BLOOM, Clerk of the Board of Supervisors

Deputy

By:

APPENDIX E

GLA Report January 2004



GeoLogic Associates

Geologists, Hydrogeologists and Engineers

January 6, 2004 GLA JN 9637

Bryan A. Stirrat & Associates 1360 Valley Vista Drive Diamond Bar, CA 91765

Attention: Christine Arbogast, Vice President

ANALYSIS OF GROUNDWATER RESOURCES IN ZONE 4 PRIMA DESHECHA LANDFILL ORANGE COUNTY, CALIFORNIA

GeoLogic Associates (GLA) is pleased to provide this letter report describing our analysis of groundwater resources available in the Zone 4 development area of the Prima Deshecha Landfill. The analysis presented herein is based on the available geologic, hydrogeologic, and climatic data for the site. It is noted that geotechnical and groundwater studies for the Zone 4 Master Plan were conducted for the specific purpose of obtaining geologic and hydrogeologic design criteria for the future landfill. As a result, there is little site-specific information for evaluating the actual groundwater resources in Zone 4, as much of this resource lies below the elevations that were explored to obtain landfill design information. Site specific geologic information is utilized wherever possible; and where specific data were not available, the references or justification for the assumptions used are provided.

BACKGROUND

As part of the ongoing development of the Prima Deshecha Landfill (PDLF), the United States Army Corps of Engineers, United States Fish and Wildlife Service, the State of California Department of Fish and Game, and the Regional Water Quality Control Board (resource agencies) have required the County of Orange Integrated Waste Management Department (COIWMD) to put forth efforts to replace sensitive biological resources disturbed or removed for landfill operations and /or construction of ancillary facilities. Landslides that occurred as a result of heavy rainfall associated with the 1997/1998 "El Nino" required stabilization in order to further the development of the landfill Master Plan. While numerous stabilization options were considered, the stabilization option selected as the most technically sound and cost-effective resulted in re-alignment of Prima Deshecha Canada Creek (creek), the principal surface water course that runs through the landfill property. Realignment of the creek would result in removal or disturbance of large quantities of riparian habitat. Approval of the preferred landslide mitigation alternative by the resource agencies was contingent upon restoration of the habitat in the realigned creek channel. Currently, the primary source of water for the creek is supplied by precipitation runoff and from a groundwater spring located near the center of the proposed Zone 4 landfill in the eastern portion of the site. On average, the PDLF receives approximately 10.2 inches of rainfall per year (GLA, 2002; Converse

Consultants, 1992). Approximately 90 percent of the precipitation falls on the site between December and March. After that time, the only source of surface water to the creek is from spring flow. Based on field observations and geological investigations (GLA, 2002), the spring is apparently fed from groundwater stored in the Monterey Formation and uppermost San Onofre Breccia. A splay of the Forster Fault juxtaposes low-permeability claystone of the Capistrano Formation against the higher permeability breccia and conglomerate of the San Onofre Breccia. As a result of this permeability contrast and the local topography, groundwater surfaces and continues to flow in the creek. At some distance from the spring, the surface water submerges into the alluvium. That distance varies throughout the year and is apparently dependent upon the flow rate of the spring, the time since last rainfall, the amount of vegetation utilizing the surface water and alluvial groundwater, the amount of alluvial soils in the channel, and other factors. The spring has provided some water to the ground surface throughout the operating life of the landfill, and the resource agencies have indicated that the preferred source of water to the habitat restoration areas is the spring water.

Recognizing that the source of the spring water is tied into recharge of water bearing strata in Zone 4, and that future landfill development in this area would reduce the area of recharge to the spring, BAS requested that GLA estimate the amount of water available in these strata and determine the length of time that water could be provided in to the spring, during (or after) the development of Zone 4. Utilizing limited, available geologic, hydrologic, hydrogeologic, and climatic data for the site, GLA analyzed the available water resources in Zone 4. Geologic and hydrogeologic studies for the Zone 4 area have been conducted to evaluate the geotechnical character of the soils and bedrock in this area, and have not evaluated groundwater resources to any detail. As a result, the analysis presented herein includes assumptions pertaining to the hydrogeologic character of similar bedrock units. Furthermore, this analysis only provides an assessment of water quality and its effect on the proposed habitat mitigation areas is provided.

ANALYSIS ASSUMPTIONS

The following table summarizes the model scenarios analyzed and the ranges of input parameters.

INPUT PARAMETER	SCENARIOS				
	Conservative	Moderately Conservative	Moderate	Moderately Optimistic	Optimistic
Volume of Saturated San Onofre Breccia	47,115,000 cubic yards				
Volume of Saturated Monterey Sandstone		5,495,000 cubic yards			
Volume of Saturated Monterey Siltstone		20,192	2,500 cubic yards		
Porosity of the San Onofre Breccia	5%	7%	7%	7%	8%
Specific Yield of the San Onofre Breccia	1%	3%	3%	3%	5%
Porosity of the Monterey Sandstone	5%	10%	10%	10%	15%
Specific Yield of the Monterey Sandstone	5%	10%	10%	10%	15%
Porosity of the Monterey Siltstone	15%	20%	20%	20%	25%
Specific Yield of the Monterey Siltstone	1%	2%	2%	2%	2.5%
Flow Rate of the Spring	4.5 gpm	4.5 gpm	4 gpm	3.5 gpm	3.5 gpm
Annual Rainfall	9.5 inches	9.5 inches	10.2 inches	12.5 inches	14.0 inches
Rainfall Runoff	75%	75%	70%	65%	65%
Evapotranspiration	25%	25%	20%	15%	15%

The derivation of these assumed values is described below.

The Bedrock is Uniformly Saturated: The results of field investigations conducted for the Zone 4 Master Plan suggest that a portion of Monterey Formation located in the northern half of Zone 4 is unsaturated; however, for this analysis, it is assumed that the bedrock beneath the groundwater equipotential contours is uniformly saturated. The groundwater equipotential contours were developed from groundwater elevation data collected from eleven groundwater-monitoring wells screened in bedrock at the PDLF. The data used in this analysis were collected in 1997 and represent "average" rainfall conditions (Figure 1). Groundwater elevation data collected after this time would reflect "El Nino" conditions, and may be unusually high.

<u>Water Source</u>: Portions of the Monterey Formation and San Onofre Breccia within the topographic catch area of the spring provide water for the Zone 4 spring. This assumption follows generally accepted hydrogeologic principles and provides a finite area from which to determine the potential volume of available groundwater. This assumption is graphically presented on Figures 2, 3, and 4.

Quantity of Water Available in the Bedrock: The quantity of water in the bedrock is equal to the volume of the formation under the groundwater potentiometric surface within the catch area of the spring multiplied by the formation porosity: If total saturation is assumed (as stated in Assumption No. 2), then the volume of water is equivalent to the pore volume of the of the formation. Porosity values were not obtained as part of the Zone 4 Master Plan geotechnical investigations. The saturated interval of the Monterey Formation includes an upper siltstone member and a lower sandstone member. The San Onofre Breccia is a cemented sandstone and breccia/conglomerate with little observed primary porosity. Secondary porosity in the San Onofre Breccia includes jointing and faulting of the rock, and is likely to be the prime conveyance mechanism for groundwater in this unit. For the analysis, porosity values of these formations were varied using representative values for similar rock types (Driscoll, 1986).

Quantity of Free Groundwater: Free groundwater is that quantity of groundwater that can flow to a spring under gravity or to a well by pumping. The specific yield of an aquifer is that percentage of the water volume that can flow to a spring or well. Water left in the aquifer is held there by molecular attraction to soil particles and by capillarity (surface tension). Sufficient field data were not obtained during the Zone 4 Master Plan investigations to determine the specific yield of the bedrock units in Zone 4. As a result, specific yield values were assigned to the Monterey Formation members and San Onofre Breccia based on published values for similar bedrock materials (Driscoll, 1986).

For this analysis, all available free groundwater in the Monterey Formation and San Onofre Breccia is assumed to be discharged through the spring. The analysis assumes that there is no vertical movement of groundwater below the spring elevation, and that there is no lateral movement of groundwater beyond the catch area of the spring and through other minor seeps and springs. As a result, this assumption provides more groundwater to the spring than is likely to be available under natural conditions.

Spring Flow Rate: For this analysis, the rate of spring flow varies between 3.15 and 4.5 gallons per minute (gpm). Although the 1992 Converse Report indicates flow rates of 2 to 5 gpm, no historical stream gauging data are available, though field observations suggest that these values are appropriate. A minimum value of 3.15 gallons per minute was used herein to be consistent with a hydrologic evaluation for the Zone 1 landslide remediation project habitat mitigation program (GLA, 2002). The analysis assumes that the spring flows all year long at a constant rate.

Rainfall, Runoff, and Evapotranspiration Data: Historical rainfall data were obtained from the California Irrigation Management Information System (CIMIS). Ten years of historical data from a weather station in Irvine, California indicate that the average annual rainfall in this area is 10.2 inches. Conservative analyses utilize lower values that might occur during drought periods. Optimistic analyses utilized higher values that might occur during a series of "El Nino" years. Rainfall rates ranging from 9.5 to 14.0 inches per year were considered to fall within a range of reasonable, sustainable rainfall values for this analysis.

The percentage of rainfall lost to runoff was estimated based on soil type, hydraulic gradient in this area of Zone 4, and general intensity of storm events for the landfill site. Balcom and Bosanko soils typically overlie the siltstone units of the Monterey Formation (which, in turn, overlie the Monterey Sandstone and San Onofre Breccia). These soils have very slow infiltration rates ranging from 0.06 to 0.2 inches per hour (Converse Consultants, 1992). The amount of precipitation that is absorbed into the soil to become available for groundwater recharge was estimated by varying the runoff and evapotranspiration values.

ANALYTICAL RESULTS

Modeling was performed using basic mathematical formulae to determine the length of time water might be available for habitat recharge using the constants and variables described above. Based on these calculations, the "worst case" or most conservative scenario would indicate that the groundwater resources in Zone 4 could only supply water for a period of about 6 years after the recharge area is impacted. Based on the Zone 4 phasing proposed in the Amended 2001 General Development Plan for the PDLF, the greater portion of the recharge area is anticipated to be impacted by 2045. The "best case" or most optimistic scenarios suggests that the Zone 4 groundwater resources could supply water to the habitat mitigation areas for a period of up to 52 years after the recharge area is impacted. These results are summarized in the table below:

INPUT PARAMETER	SCENARIOS				
	Conservative	Moderately Conservative	Moderate	Moderately Optimistic	Optimistic
Volume of Saturated San Onofre Breccia	47,115,000 cubic yards				
Volume of Saturated Monterey Sandstone	5,495,000 cubic yards				
Volume of Saturated Monterey Siltstone	20,192,500 cubic yards				
Porosity of the San Onofre Breccia	5%	7%	7%	7%	8%
Specific Yield of the San Onofre Breccia	1%	3%	3%	3%	5%
Porosity of the Monterey Sandstone	5%	10%	10%	10%	15%
Specific Yield of the Monterey Sandstone	5%	10%	10%	10%	15%
Porosity of the Monterey Siltstone	15%	20%	20%	20%	25%
Specific Yield of the Monterey Siltstone	1%	2%	2%	2%	2.5%
Flow Rate of the Spring	4.5 gpm	4.5 gpm	4 gpm	3.5 gpm	3.5 gpm
Annual Rainfall	9.5 inches	9.5 inches	10.2 inches	12.5 inches	14.0 inches
Rainfall Runoff	75%	75%	70%	65%	65%
Evapotranspiration	25%	25%	20%	15%	15%
ZARAM TRANSPORT	Stephen	September 2	Sec. 12 57 - 18 8	计专用字 与化于同学的	STAR IN

CONCLUSION

The Zone 4 groundwater availability analyses presented herein were performed to obtain a gross quantity of water that might be available for habitat restoration uses at the Prima Deshecha Landfill. It is not a groundwater flow model nor were groundwater-flow modeling codes used in this analysis. More detailed analysis of the available groundwater resources in Zone 4 should be preceded by a hydrogeologic investigation designed to obtain specific water storage and transmission characteristics of the local water-bearing units. It should be noted, however, that a more detailed analysis may provide a tighter range of results, but it is unlikely that the results would indicate greater resource sustainability.

The available quantities of water determined in this analysis are based on gravity flow of water stored at or above the level of the Zone 4 spring. Additional water may be available at depths below the spring level and could be developed using deep groundwater extraction wells.

This evaluation makes no assessment of the compatibility of the quality of groundwater in Zone 4 with the proposed plant species in the habitat restoration areas. A riparian ecologist should review groundwater chemistry data obtained from Zone 4 monitoring wells to determine the compatibility of the groundwater chemistry with the proposed habitat, particularly under a groundwater extraction scenario.

Groundwater conditions change over time due to local or regional climatic changes, withdrawal or addition of groundwater by man, and by changes in the geometry and recharge conditions of the groundwater basin by grading or natural causes. Dependence on a single source of water for the habitat restoration area should consider that such changes may affect the quantity of available groundwater.

This letter report is based on the available geologic and hydrogeologic data for the Zone 4 area at Prima Deshecha Landfill. Our firm should be notified if the geologic or

hydrogeologic conditions are found to differ from those described in this report, since this may require a re-evaluation of the conclusions presented herein. This letter report has not been prepared for use by parties and projects other than those named or described herein. It may not contain sufficient information for other parties or other purposes. This letter report has been prepared in accordance with generally accepted hydrogeologic practices, and makes no warranties, either expressed or implied, as to the professional advice or data included in it.

GeoLogic Associates

methor

Khn M. Hower, CEG Senior Geologist

REFERENCES

Converse Consultants, 1992, "Geology and Soils Environmental Impact Report – Prima Deshecha Landfill, prepared for P&D Technologies, April 30, 1992.

Driscoll, Fletcher G., 1986, Groundwater and Wells, Second Edition.

- GeoLogic Associates, 2002, "Geotechnical Investigation Zone 4, Prima Deshecha Landfill Master Plan," prepared for the County of Orange Integrated Waste Management Department, January 2002.
- GeoLogic Associates, 2002, "Revised Hydrologic Evaluation of the Landslide Remediation Project Habitat Mitigation Program – Prima Deshecha Landfill, Orange County, California," prepared for the County of Orange Integrated Waste Management Department, January 30, 2002.









Ĩ

Volume of Saturated San Onofre Breccia (Tso - to spring elevation) Volume of Saturated Monterey Formation Sandstone (Tmss) Volume of Saturated Monterey Siltstone (Tmsi)

,

47,115,000 cubic yards 5,495,000 cubic yards 20,192,500 cubic yards

DETAILED ANALYSIS - MOST CONSERVATIVE SCENAL All saturated units provide water to the spring. Low-end published values of porosity and specific yield. With and constant flow, rate from spring.	210	
Righ-end constant now rate now spring.		
Relow Average Rainfall		
High runoff and high evapotranspiration		
Volume of Tmss	: 5,495,000	cubic vards
convert to cubic feet (x27)	148,365,000	cubic feet
convert to gallons (x7.48)	: 1,109,770,200	gallons
determine approx. volume of water = porosity (x0.05)	55,488,510	gallons of water in Tm sandstone today
determine available quantity of water = specific yield (x0.05)	2,774,426	gallons of water in Tm sandstone today that can flow to a well or spring
Volume of saturated Tmsi	20,192,500	cubic yards
convert to cubic feet (x27):	545,197,500	cubic feet
convert to gallons (x7.48):	4,078,077,300	gallons
determine approx. volume of water = porosity (x0.15):	611,711,595	gallons of water in Tm siltstone today.
determine available quantity of water = specific yield (x0.01):	6,117,116	gallons of water in Tm siltstone today that can flow to a well or spring.
Volume of saturated Tso:	47,115,000	cubic yards
convert to cubic feet (x27):	1,272,105,000	cubic feet
convert to gailons (x7.48):	9,515,345,400	gallons
determine approx. Volume of water = porosity (x0.05):	475,767,270	gallons of water in Tso today
determine available quantity of water = specific yield $(x0.01)$:	4,757,673	gallons of water in the San Onotre Breccia that can flow to a well or spring
Area of externed bedrock in the Spring catch area of Zone 4:	13,049,214	gallons of water in bedrock that can flow to a well or spring.
Area of saturated Tm in Zone A unaffected by landfill development:	107	
Area of saturated Tm in Zone 4 affected by landfill development:	135	autos acras
Area of saturated 1 m in 2010 (areated by landin development Average annual precipitation	95	inches
Loss to runoff:	75	nercent
Precipitation available for infiltration:	2.4	inches
Loss to evapotranspiration and plant growth:	25	percent
Precipitation available for groundwater recharge:	1.8	inches
Total volume of water available for recharge in natural condition:	9,053,968	gallons of water for groundwater recharge per year.
Total volume of free water available (specific yield):	90,540	gallons of free groundwater
Loss of water each year due to spring discharge:	2,365,200	gallons of water in spring discharge
Net loss of water due to spring discharge:	2,274,660	gallons per year
Landfill Development Effects on Recharge Area:	Amount of Free	Water in the Bedrock (gallons)
Year I (Phase A - No effect)	13,649,214	
Year 2 (Phase A - No effect)	11,374,554	
Year 3 (Phase A - No effect)	9,099,894	
Year 4 (Phase A - No effect)	6,825,233	
Year 5 (Phase B - 11% Initiation Reduction)	4,540,613	
I Car O (First D - 11% Influention Reduction)	2,233,994	
I Call / (Finance $D = 1176$ similation (Collection) Verse 9 (Dasse $R = 1196$ Infiltration Deduction)	Dry	
Ver Q (Phase R - 11% Infiltration Deduction)	Dry	
Vers 10 (Phase B - 11% Infiltration Reduction)	Day	
1 Cal IV (1 mass D 11 fe initiation ACtivation)	Diy	

ed San Osofre Breccia (Tso - to spring elevation) 47,115,000 cabic yards 5,495,000 cubic yards Volume of Saturated Monterey Formation Sandstone (Tissa) Volume of Saturated Monterey Siltstone (Tissa) 20,192,500 cubic yards DETAILED ANALYSIS - MODERATELY CONSERVATIVE SCENARIO All saturated units provide water to the spring Moderate published values of porosity and specific yield. High-end constant flow rate from spring. Recharge varies by year based on Zone 4 de Below Average Rainfall High runoff and high evapotranspiration Volume of Tunss: 5,495,000 cubic yards convert to cubic feet (x27): 148.365.000 cubic feet convert to galloum (27.44): 1,109,770,200 gallou determine approx. volume of water = porosity (20.1): 110,977,020 gallou ine available quantity of water = specific yield (20.1): 110,977,702 gallou 110.977.020 gallons of water in Tay sandstone today 11,097,702 gallons of water is Tm sandstone today that can flow to a well or spring Volume of saturated Test: convert to cubic fact (x27): 20 192 500 cmbic vants 545,197,500 cubic feet convert to gallons (x7.48): 4,078,077,300 gallons determine approx. volume of water - porosity (x0.2): \$15,615,460 gallons of water in Tm siltstone today. 16,312,309 gailous of water in Tar substance today that can flow to a well or spring, 47,115,000 cabic yards e available qu maity of water = specific yield (x0.02): Val ue of saturated Tao: convert to cabic feet (x27): 1,272,105,000 cabic feet convert to gallous (x7.48): 9,515,345,400 gallo determine approx. volume of water = porticity (vo.07); determine available quantity of water = apocific yield (vo.03); Total available water (TsoH2O+TmasH2O+TmaiH2O); Area of submitted behavior in the Spring carchi area of Zone 4; a of submitted Tm in Zone 4 matElicated by landfill development: 666,074,178 gallons of water in Tso today 19,982,225 gallons of water in the San Onofre Breccia that can flow to a well or spring 47,392,237 gallons of water in bedrock that can flow to a well or spring. 187 астек 52 acres Area of saturated Tm in Zone 4 affected by landfill devel 135 acres 9.5 inches 75 percent Average sumal precipitation Logs to resolf Precipitation available for infiltration: 2.4 inches 25 perce Loss to evapotranspiration and plant growth Pres tion available for groundwater recharge. 1.8 inches of water available for recharge in natural condition: lotal volume of free water available (specific yield): 9,053,968 gallons of water for groundwater rec 90,540 gallons of free groundwater 2,365,200 gallons of water in spring discharge ndwater recharge per year. Total vol and of proc wants available (approxime) areas, of water each year due to spring discharge: Net loss of water due to spring discharge: --Loss of w 2,274,660 gailons per year Landfill Development Effects on Recharge Area: Amount of Free Water in the Bedrock (gallons) Year 1 (Phane A - No effect) Year 2 (Phane A - No effect) 47,392,237 45,117,576 Year 3 (Phase A - No effect) Year 4 (Phase A - No effect) 42,842,916 40,568,256 Year 5 (Phase B - 11% Infiltration Reduction) 38,283,636 Year 3 (Phase B - 11% Infiltration Keduciton) Year 6 (Phase B - 11% Infiltration Reduction) Year 7 (Phase B - 11% Infiltration Reduction) Year 9 (Phase B - 11% Infiltration Reduction) Year 9 (Phase B - 11% Infiltration Reduction) 35,999,016 33,714,397 31,429,777 29.145.157 Year 2 (man: o ~ 11% antitoration Reduction) Year 10 (Plane B - 11% infiltration Reduction) Year 11 (Plane B - 11% infiltration Reduction) 26,860,537 74 575 918 Year 12 (Phase B - 11% Infiltration Reduction) 22 291 299 Year 13 (Phase C - 11% Infiltration Reduction 20.006.671 Year 14 (Phase C - 11% infiltration Reduction) Year 14 (Phase C - 11% infiltration Reduction) Year 15 (Phase C - 11% infiltration Reduction) 17,722,059 15,437,439 Year 16 (Phase C - 11% Infiltration Reduction) 13,152,819 Year 17 (Phase C - 11% Infiltration Reduction) 10.168 200 Year 18 (Phase C - 11% Infiltration Reduction) Year 19 (Phase C - 11% Infiltration Reduction) 8,583,580 6.298,960 Year 20 (Phase C - 11% Infiltration Reduction) 4,014,341 Year 21 (Phase C - 11% Infiltration Reduction) 1.729.721 Year 22 (Phase C - 11% Infiltration Reduction) Year 23 (Phase C - 11% Infiltration Reduction) Year 23 (Phase C - 11% Infiltration Reduction) Dry Dry Year 24 (Phase D - 51% infiltration reduction) Year 25 (Phase D - 51% infiltration reduction) Dry Dry Year 26 (Phase D - 51% infiltration reduction) Dry Year 27 (Phase D - 51% infiltration reduction) Dry Year 28 (Phase D - 51% infiltration reduction) Year 29 (Phase E - 51% infiltration reduction) Dry Dry Year 30 (Phase E - 66% infiltration reduction) Dry Year 31 (Phase F - 69% Infiltration Reduction) Dry Dry Year 32 (Phase F - 69% Infiltration Reduction) Year 33 (Phase F - 69% Infiltration Reduction) Drv Year 34 (Phase F - 69% Infiltration Reduction) Dry Year 35 (Phase F - 69% Infiltration Reduction) Dry Dry Year 36 (Plane F - 69% Infiltration Roduction) Dry Dry Year 37 (Phase F - 69% Infiltration Reduction) Year 38 (Phase F - 69% Infiltration Reduction) Year 39 (Phase G - 72% infiltration Reduction) Dry Dry Year 40 (Phase G - 72% Infiltration Reduction) Dry Dry Year 41 (Phase G - 72% Infiltration Reduction) Year 42 (Phase H - 72% Infiltration Reduction) Dry Dry Year 43 (Phase H - 72% Infiltration Roduction) Year 44 (Phase H - 72% Infiltration Reduction) Dry Dry Year 45 (Phase H - 72% Infiltration Reduction) Year 46 (Phane I - 72% Infiltration Reduction) Year 47 (Phase I - 72 % Infiltration Reduction) Dry Year 48 (Phase I - 72% Infiltration Reduction) Dry Year 49 (Closure - 72 % Infiltration Reduction) Dry Year 50 (Closure - 72% Infiltration Reduction) Drv

. •• •

. . .

Volume of Saturated San Onofie Breecia (Tao - to spring elevation) Volume of Saturated Monterey Formation Sandstone (Taus) Volume of Saturated Montorey Silestone (Tausi) 47,115,000 cubic yards 5,495,000 cubic yards 20.192.500 cubic varies **DETAILED ANALYSIS - MODERATE SCENARIO** All saturated units provide water to the spring, Modernic published values of poresity and specific yield. Moderate constant flow rate from spring. Recharge varies by year based on Zone 4 develop Average Raisfall Moderate runoff and moderate evapetranspiration Volume of Tuss 5,495,000 cubic yards 148,365,000 cubic feet convert to cubic feet (x27): convert to gallons (x7.48): 1,109,770,200 gallons e of water = peroxity (x0.1): 110,977,020 gallons of water in Tin sandstone today ю арриох, чо 11,097,702 gallens of water in Ter sandstone today that can flow to a well or spring 20,192,500 cabic yards ne available antity of water - specific yield (s0.1): Volume of saturated Tami: convert to cubic feet (x27); 545,197,500 cubic feet convert to gallons (x7.48): 4,078,077,300 gallons determine approx. volume of water = poronity (x0.2): ac available quantity of water = specific yield (x0.02): Volume of saturated Tao: 815(61),460 gallons of water in Tu siltstone today. 16,312,309 gallons of water in Tu siltstone today that can flow to a well or spri 47,115,000 cubic yards convert to cabic fact (x27): 1,272,105,000 cabic fact convert to gallons (x7.48): 9,515,345,400 gallons z approx. volu ne of water = parasity (x0.07); 666,074,178 gallons of water in Tso today determine available quantity of water - specific yield (x0.03): Total available water (TaoH2O+TmaiH2O+TmaiH2O): 19,982,225 gallons of water in the San Onofre Breccia that can flow to a well or spring 47,392,237 gallons of water in bedrock that can flow to a well or spring. Area of saturated bodrock in the Spring catch area of Zane 4 Area of saturated Tan in Zone 4 multiceted by landfill development: Area of saturated Tan in Zone 4 affected by landfill development; 187 acres 52 acres 135 acres Average annual pre 10.2 inches 70 percent 3.1 inches Lors to mooff Precipitation available for infiltration: Loss to evapotranspiration and plant growth: 20 percent Precipitation available for groundwater rech ianjo: 2.4 inches recommended and the second sec 12,443,011 gallons of water for groundwater recharge per year. 124,430 gallons of free groundwater Total vol 2.102.400 gallons of water in spring discharge Loss of water each year day to suring discharge: Net loss of water due to spring discharge: 1,977,970 gallons per year Landfill Development Effects on Recharge Area: Ann ount of Free Water in the Bedrock (gallons) Year 1 (Plane A - No effect) Year 2 (Plane A - No effect) Year 3 (Plane A - No effect) 47.392.237 45,414,267 43,436,297 Year 5 (Phase B - 11% Infiltration Reduction) 41,458,327 39,466,670 Year 6 (Phase B - 11% Infiltration Reduction) 37,475,012 Year 7 (Phase B - 11% infiltration Reduction) Year 8 (Phase B - 11% infiltration Reduction) 35,483,355 33,491,698 31,500,041 Year 9 (Phase B - 11% Infiltration Reduction) Year 10 (Phase B - 11% Infiltration Reduction) 29,504,384 Year 11 (Plane B - 11% Infiltration Reduction) Year 12 (Plane B - 11% Infiltration Reduction) Year 12 (Plane B - 11% Infiltration Reduction) 27,516,726 21 121 069 Year 13 (Phase C - 11% Infiltration Reduction) 23,533,412 Year 14 (Phase C - 11% Infiltration Reduction) 21,541,755 Year 15 (Phase C - 11% Infiltration Reduction) Year 16 (Phase C - 11% Infiltration Reduction) 17.558.440 15,566,783 Year 17 (Plane C - 11% Infiltration Reduction) Year 18 (Phase C - 11% infiltration Reduction) Year 19 (Phase C - 11% Infiltration Reduction) Year 20 (Phase C - 11% Infiltration Reduction) 11,583,469 9,591,812 Year 21 (Phase C - 11% Infiltration Reduction) Year 22 (Phase C - 11% Infiltration Reduction) 7,600,154 5 608 497 Year 23 (Plane C - 11% infiltration Reduction) Year 24 (Plane D - 51% infiltration reduction) 3,616,840 1.575.411 Year 25 (Phase D - 51% infiltration reduction) Dry Year 26 (Phase D - 51% infiltration reduction) Dry Year 27 (Phase D - 51% infiltration reduction) Year 28 (Phase D - 51% infiltration reduction) Dry Dry Dry Year 29 (Phase E - 66% infiltration reduction) Year 30 (Phase E - 66% infiltration reduction) Dry Dry Year 31 (Phase F - 69% Infiltration Reduction) Year 32 (Phase F - 69% Infiltration Reduction) Dry Dry Year 33 (Phase F - 69% Infiltration Reduction) Year 34 (Phase F - 69% Infiltration Reduction) Dry Dry Year 35 (Phase F - 69% Infiltration Reduction) Year 35 (Phase F - 69% Infiltration Reduction) Year 36 (Phase F - 69% Infiltration Reduction) Dry Dry Your 37 (Phase F - 69% Infiltration Reduction) Dry Year 38 (Phase F - 69% Infiltration Reduction) (ear 39 (Plum: G - 72% Infiltration Reduction) Year 40 (Phase G - 72% Infiltration Reduction) Dry Year 41 (Phase G - 72% Infiltration Reduction) Dry Year 42 (Phase H - 72% Infiltration Reduction) Dтy Year 43 (Phase H - 72% Infiltration Reduction) Dry Year 44 (Phase H - 72% isfiltration Reduction) Year 45 (Phase H - 72% isfiltration Reduction) Dry Dry Dry Year 46 (Phase I - 72% Infiltration Reduction) Year 47 (Phase I - 72 % Infiltration Reduction) Dry Dry Year 48 (Phase 1 - 72% Infiltration Reduction) Year 49 (Closure - 72 % Infiltration Reduction) Year 50 (Closure - 72% Infiltration Reduction) Dry

Dry

• _ • •

47.115.000 cubic vante Volume of Saturated San Osofie Breccia (Tso - to spring elevation) Volume of Saturated Montercy Formation Sandstone (Taus) Volume of Saturated Montercy Formation Sandstone (Taus) 5,495,000 cubic yards 20.192.500 cabic varia DETAILED ANALYSIS - MODERATELY OPTIMISTIC SCENARIO All saturated units provide water to the spring. Mederate published values of porosity and specific yield. Low constant flow rate from spring. Recharge varies by year based on Zone 4 development. Above Average Rainfall Low runoff and low evapotranspiration Volume of Tauss: 5,495,000 cubic yards 148,365,000 cubic feet convert to cubic feet (x27); contrar to callar net (x2.7), 173,000 cubic text convert to gallams (x7.48); 1,109,770,200 gallans of water in Tin sandstate today determine approx. volume of water m porosity (x0.1); 110,977,202 gallans of water in Tin sandstate today mine available quantity of water - specific yield (x0.1); 11,097,702 gallans of water in Tin sandstate today that can flow to a well or spring Volume of saturated Tawi: 2,0172,500 cubic yards crunine available quantity of water - specific yield (x0, 1): Volume of saturated Teuri: convert to cubic feet (x27): 545,197,500 cubic feet convert to gallons (x7.48): 4,078,077,300 gallons determine approx. volume of water - poreity (x0.2): 815,615,660 gallons of water in Tm sittstone today. ne available quantity of water - specific yield (x0.02): 16,312,309 gallons of water in Tm sittstone today that can flow to a well or spring. ine available quantity of water - specific yield (x0.02): Volume of saturated Tac: 47,115,000 cabic yard convert to cabic fact (x27): 1,272,105,000 cabic fact 47,115,000 cabic yanta convert to gallons (x7.48): 9,515,345,400 gallons determine approx. volume of water = provinity (x0.07): 666,074,178 galloas of water in Tso today mine available quantity of water = specific yield (x0.03): 19,982,225 galloas of water in the Sm Ot Total available water (TsoH2O+TmsaH2O+TmaiH2O): 47,392,237 galloas of water in bedrock th 19,982,225 gallons of water in the San Onofic Breccis that can flow to a well or spring 47,392,237 gallons of water in bedrock that can flow to a well or spring. Area of saturated bedrock in the Spring catch area of Zone 4 Area of saturated Tan in Zone 4 unaffected by landfill development: Area of saturated Tan in Zone 4 affected by landfill development: 187 acres 52 acres 135 acres 12.5 inches Average stand precipi tion: Loss to mooff 65 percent 4.4 inches Precipitation available for infiltration: Loss to evapotranspiration and plant growth: Precipitation available for grownlwater recharge: 15 percent 3.7 inches e of water available for recharge in antural condition: 18,902,143 gallous of water for groundwater recharge per year. where a vanishes for recording at manual community. al volume of free water available (specific yield): Loss of water each year due to spring discharge: Net loss of water due to spring discharge: 189,021 gallons of free groundwater 1,839,600 gallons of water in spring discharge Total v 1,650,579 gallons per year Landfill Development Effects on Recharge Area: An ount of Free Water in the Bedrock (gallons) Year I (Phase A - No effect) Year 2 (Phase A - No effect) 47.392.237 45,741,658 Year 3 (Plane A - No effect) 44.091.079 Year 5 (Phase B - 11% Infiltration Reduction) 42,440,501 40,769,130 Year 6 (Phase B - 11% Infiltration Reduction) 39,097,759 Year 7 (Phase B - 11% Infiltration Reduction) 37,426,388 Year 8 (Phase B - 11% Infiltration Reduction) Year 9 (Phase B - 11% Infiltration Reduction) 35,755,017 34.083.646 Yew 10 (Phase B - 11% Infiltration Reduction) Year 11 (Phase B - 11% Infiltration Reduction) 32,412,275 30,740,904 Year 12 (Phase B - 11% Infiltration Reduction) Year 13 (Phase C - 11% Infiltration Reduction) 29,069,533 27,398,163 Year 14 (Phase C - 11% Infiltration Reduction) Year 15 (Phase C - 11% Infiltration Reduction) 25,726,792 24,055,421 Year 16 (Plane C - 11% infiltration Reduction) Year 17 (Plane C - 11% infiltration Reduction) Year 13 (Plane C - 11% infiltration Reduction) Year 19 (Plane C - 11% infiltration Reduction) 22,384,050 20,712,679 19,041,308 17,369,937 15,698,566 Year 20 (Phase C - 11% Infiltration Reduction) Year 21 (Phase C - 11% Infiltration Reduction) Year 22 (Phase C - 11% Infiltration Reduction) 17 155 824 Year 23 (Phase C - 11% Infiltration Reduction) 10,684,453 Year 24 (Phase D - 51% infiltration reduction) Year 25 (Phase D - 51% infiltration reduction) Year 26 (Phase D - 51% infiltration reduction) 8,937,474 7,190,494 5.443.515 Year 27 (Phase D - 51% infiltration reduction) 3,696,535 Year 28 (Phase D - 51% infiltration reduction) 1 949 556 Year 29 (Phase E - 66% Infiltration reduction) Year 30 (Phase E - 66% Infiltration reduction) 174,223 Dn Year 31 (Phase F - 69% Infiltration Roduction) Dry Year 32 (Phase F - 69% Infiltration Reduction) Dry Year 33 (Phase F - 69% Infiltration Reduction) Dry Year 34 (Phase F - 69% Infiltration Reduction) Dry Year 35 (Phase F - 69% Infiltration Reduction) Dry Dry Year 36 (Phase F - 69% Infiltration Reduction) Year 37 (Phase F - 69% Infiltration Reduction) Dry Year 38 (Phase F - 69% Infiltration Reduction) Year 39 (Phase G - 72% Infiltration Reduction) Dry Dry Dry Year 40 (Phase G - 72% Infiltration Reduction) Dry Dry Year 41 (Phase G - 72% Infiltration Reduction) Year 42 (Phase H - 72% Infiltration Reduction) Year 43 (Phase H - 72% Infiltration Reduction) Dry Dry Year 44 (Phase H - 72% Infiltration Reduction) Year 45 (Phase H - 72% Infiltration Reduction) Dry Dry Year 46 (Phase I - 72% Infiltration Reduction) Year 47 (Phase I - 72 % Infiltration Reduction) Dry Year 48 (Phase I - 72% infiltration Reduction) Dry Dry Year 49 (Closure - 72 % Infiltration Reduction) Year 50 (Closure - 72% Infiltration Reduction)

Dry

47,115,000 cubic yarda 5,495,000 cubic yarda 20,192,500 cubic yarda e of Saturated San Onofre Breccis (Tao - to spring elevation) turated Monterey Fou Volume of Saturated Monterey Founation Sanda Volume of Saturated Monterey Siltstone (Tmsi) uc (Taus)

1. 1

DETAILED ANALYSIS - OPTIMISTIC SCENARIO

Alastariado ArALISIS- OF LIMISTIC & LAVARDO Alastariado unita provide water in the spring. High published values of porenity and specific yield. Low constains flow rute from upring. Racharge varies by year based on Zone 4 developme Above Average Balafiel Low receil and low exploremapication Volume of Trass 5,495,000 cubic yarda convert to cubic feet (x27); 148,365,000 cubic feet convert to control per (2,27). 199,500,000 control per (2,27). convert to guillous (27,49): 1,109,770,200 guillous determine approx. volume of water = ponentiy (20.15): 166,465,530 guillous of water in Tm sandstone today inc available quanchiy of water = specific yield (20.15): 24,969,830 guillous of water in Tm sandstone today or today that can flow to a well or spi rmine available quantity of weater = specific yield (c0. 15): 24,969,850 gallons of water in Tin standatose: today that can flow to a well or spring Volume of seturated Tami: 20,192,500 cable yeards convert to quick for Q27: 345,197,500 qallons determine approx. volume of water = provide (v0.25): 10,197,500 qallons determine approx. volume of water = provide (v0.25): 10,197,300 qallons volume of saturated Tami: 27,193,19,325 gallons of water in Tin silutone today. Volume of saturated Tami: 47,115,000 qallons of water in Tin silutone today that can flow to a well or spring. Volume of saturated Tami: 71,15,000 qallons for water in Tin silutone today that can flow to a well or spring. Volume of saturated Tami: 71,15,000 qallons for water in Tin silutone today that can flow to a well or spring. Volume of saturated Tami: 71,15,000 qallons for converse to gallons (r2,7437) 9,515,345,400 gallons of water in Tin today mains available water = provide yield (sto.05): 11,061,342 gallons of water in The today Total arealistic water = provide yield (sto.05): 11,061,342 gallons of water in the Sam Quork that can flow to a well or spring. Total arealistic water (r0.470,771,302) 83,519,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. 18,719,194 gallons of water in bohook that eas flow to a well or spring. ine available quantity of water = specific yield (20.15): Volume of asturated Tensi: emine available quantity of water = specific yield (x0.025): Volume of saturated Tao: Area of saturated badrock in the Spring estable area of Zone 4: Area of saturated Tm in Zone 4 usaffected by landfill development: Area of saturated Tm in Zone 4 affected by landfill development: 187 acres 52 norm 135 norm 14.0 inches Average manual procipitation: 65 percent 4.9 inches 15 percent 4.2 inches Long to mmolf Law to remote Precipitation available for indistration: Loss to evaportnapiration and plant growth Precipitation available for groundwater recharge: rrecipitation available for recharge in natural condition: e of water available for recharge in natural condition: Total volume of free water available (spoodife yield): Loss of water each year due to spring discharge: Not loss of water due to spring discharge: 21,170,400 gallons of water for groundwater recharge per year. 211,704 gallons of free groundwater 1,839,600 gallons of water in spring discharge 1,627,896 guilous per your antfill Development Effects on Racharge Ava.: Year 1 (Phase A - No effect) Year 2 (Phase A - No effect) Year 2 (Phase A - No effect) Year 3 (Phase A - No effect) Year 5 (Phase B - 11% Infiltration Robustion) Year 6 (Phase B - 11% Infiltration Robustion) Year 7 (Phase B - 11% Infiltration Robustion) Year 9 (Phase B - 11% Infiltration Robustion) Year 9 (Phase B - 11% Infiltration Robustion) Year 9 (Phase B - 11% Infiltration Robustion) annum of Free Weter in the Bedrock (galler 83,519,194 86,891,298 Land6H De 85,263,402 83,635,506 81,984,323 80,333,139 78 681 956 77,030,772 75,379,589

Dn

Year 10 (Phase B - 11% Infiltration Reduction) 73,728,406 Year 11 (Phase B - 11% infiltration Reduction) Year 12 (Phase B - 11% infiltration Reduction) Year 13 (Phase B - 11% infiltration Reduction) Year 14 (Phase C - 11% infiltration Reduction) 72,077,222 70,426,039 68,774,855 67.123.672 Year 15 (Plane C - 11% influence Robuston) Year 15 (Plane C - 11% influence Robuston) Year 16 (Plane C - 11% influence Robuston) Year 17 (Plane C - 11% influence Robuston) 65,472,488 63,821,305 62,170,122 Yeu: 17 (Phase C - 11% Infiltration Reduction) Yeu: 18 (Phase C - 11% Infiltration Reduction) Yeu: 20 (Phase C - 11% Infiltration Reduction) Yeu: 20 (Phase C - 11% Infiltration Reduction) Yeu: 22 (Phase C - 11% Infiltration Reduction) Yeu: 23 (Phase C - 11% Infiltration Reduction) Yeu: 23 (Phase D - 11% Infiltration Reduction) Yeu: 23 (Phase D - 51% infiltration multicion) Yeu: 25 (Phase D - 51% infiltration multicion) Yeu: 25 (Phase D - 51% infiltration multicion) 60.518.938 58 867 755 57,216,571 55,565,388 53,914,204 52,263,021 50,527,156 48,791,291 Year 26 (Phase D - 51% infiltration reduction) 47.055.426 Yar 26 (Phase D - 51% infiltration relation) Year 27 (Phase D - 19% infiltration relation) Year 29 (Phase D - 51% infiltration relation) Year 29 (Phase E - 66% infiltration relation) Year 30 (Phase E - 66% infiltration relation) Year 31 (Phase E - 66% infiltration Relation) Year 33 (Phase F - 66% infiltration Relation) Year 33 (Phase F - 66% infiltration Relation) Year 33 (Phase F - 66% infiltration Relation) 45,319,561 43,583,696 41,816,075 40.048.454 38,274,483 36,500,511 34,726,539 Year 34 (Phase F - 69% Infiltration Reduction) Year 35 (Phase F - 69% Infiltration Reduction) Year 36 (Phase F - 69% Infiltration Reduction) Year 37 (Phase F - 69% Infiltration Reduction) 12,952,567 31,178,596 29,404,624 27,630,652 Year 38 (Phane F - 69% Infiltration Reduction) Year 39 (Phane G - 72% Infiltration Reduction) 25,856,680 24,076,358 Year 40 (Phase G - 72% Infiltration Reduction) 22,296,035 Year 41 (Phase G - 72% Infiltration Reduction) Year 42 (Phase H - 72% Infiltration Reduction) Year 43 (Phase H - 72% Infiltration Reduction) Year 44 (Phase H - 72% Infiltration Reduction) 20 515 712 18,735,389 16,955,066 15,174,743 Year 45 (Phaser H - 72% Indistantion Raduction) Year 46 (Phaser H - 72% Indistriction Raduction) Year 47 (Phase I - 72% Indistriction Reduction) 13,394,420 11,614,097 9,833,775 Year 48 (Phase 1 - 72% Infiltration Reduction) Year 49 (Closure - 72 % Infiltration Reduction) 8.053.452 6,273,125 Year 50 (Closure - 72% Infiltration Reduction) 4,492,800 Year 51 (Year 1 of Post-Closure) Year 52 (Year 2 of Post-Closure) Year 53 (Year 3 of Post-Closure) 2 712 483 932,160 Dh Year 54 (Year 4 of Post-Clonare) Dn Year 54 (Year 4 or Post-Closure) Year 55 (Year 5 of Post-Closure) Year 56 (Year 6 of Post-Closure) Year 57 (Year 7 of Post-Closure) Dry Dry Dry Dry Year 58 (Year 8 of Post-Closure) Year 59 (Year 9 of Post-Closure)

Year 60 (Year 10 of Post-Closure)

APPENDIX F

Delineation of Jurisdictional Waters Prima Deshecha Landfill June 7, 2004

DELINEATION OF JURISDICTIONAL WATERS

Prima Deshecha Landfill County of Orange, California



Prepared for:

BonTerra Consulting

151 Kalmus Drive, Suite E-200 Costa Mesa, California 92625 Contact: **Mr. Gary Medeiros** 714.444.9199

County of Orange

Integrated Waste Management Department 320 N. Flower Street Santa Ana, California 92703-5000 Contact: Mr. John Tzeng 949.728.3045

Prepared by:

RBF Consulting

14725 Alton Parkway Irvine, California 92618 Contact: **Mr. Bruce R. Grove, Jr.** 949.855.3686 **Mr. Richard Beck** 949.855.3687

> June 7, 2004 JN 10-102970-10914

DELINEATION OF JURISDICTIONAL WATERS

Prima Deshecha Landfill County of Orange, California

Prepared For:

BonTerra Consulting 151 Kalmus Drive, Suite E-200 Costa Mesa, California 92625 Contact: Mr. Gary Medeiros 714/444-9199

County of Orange Integrated Waste Management Department 320 N. Flower Street Santa Ana, California 92703-5000 *Contact: John Tzeng* 949/728-3045

Prepared By:

RBF Consulting

14725 Alton Parkway Irvine, California 92618 Contacts: Mr. Bruce R. Grove, Jr. 949/855-3686 Mr. Richard Beck 949/855-3687

June 7, 2004 Revisal of February 20, 2004 JN 10-102970



June 7, 2004

Mr. Gary Medeiros BonTerra Consulting 151 Kalmus Drive, E-200 Costa Mesa, California 92626

SUBJECT: UPDATED DELINEATION OF JURISDICTIONAL WATERS Prima Deshecha Landfill, County of Orange

Dear Mr. Medeiros:

On behalf of RBF Consulting (RBF), we are pleased to submit this Delineation of Jurisdictional Waters for the above referenced project. The enclosed delineation was conducted in May 2003 to document the regulatory authority of the U.S. Army Corps of Engineers (ACOE), California Department of Fish and Game's (CDFG), and Regional Water Quality Control Board's (RWQCB) jurisdiction pursuant to the Federal Clean Water Act (CWA) and the State Fish and Game Code. The project area was surveyed pursuant to the ACOE's *1987 Wetland Delineation Manuel*, to identify evidence of flows, riparian vegetation, and hydric soils.

This report presents RBF's best effort at determining the jurisdictional boundaries using the most up-to-date regulations, written policy, and guidance from the regulatory agencies. However, as with any jurisdictional delineation, only the regulatory agencies can make a final determination of jurisdiction. Generally, this would be a written concurrence in the form of a Jurisdictional Determination (JD) letter.

Please note that based on a detailed review of current site conditions, our research has indicated that it will be necessary for the project applicant to successfully obtain the following permits prior to commencement of any construction activities within the delineated jurisdictional areas: Army Corps of Engineers 404 Permit, 1602 Streambed Alternation Agreement, and 401 Water Quality Certification.

Please do not hesitate to contact me at 949/855-3687, or Bruce Grove at 949/855-3686, if you or your staff has any questions or require further information.

Sincerely,

and Beek

Richard Beck Regulatory Coordinator Environmental Services-Special Projects

wh France

Bruce R. Grove Jr., REA Project Manager Environmental Services-Special Projects



14725 Alton Parkway, Irvine, CA 92618-2027 = P.O. Box 57057, Irvine, CA 92619-7057 = 949,472,3505 = Fax 949,472,8373

Offices located throughout California, Arizona & Nevada 🔳 www.RBF.com

DELINEATION OF JURISDICTIONAL WATERS

Prima Deshecha Landfill County of Orange, California

The undersigned certify that this report is a complete and accurate account of the findings and conclusions of a jurisdictional "waters of the U.S." (including wetlands) and "waters of the State" determination for the above-referenced project.



Richard Beck

Richard Beck Environmental Analyst/ Regulatory Coordinator

Burg 9

Bruce R. Grove, Jr. Project Manager, REA Environmental Services-Special Projects

June 2004

TABLE OF CONTENTS

1.0 1.1	INTRODUCTION AND PURPOSE	1 1
2.0 2.1 2.2 2.3	SUMMARY OF REGULATIONS Army Corps of Engineers Regional Water Quality Control Board California Department of Fish and Game	5
2.4	Activities Requiring Permits	7
3.0 3.1	METHODOLOGY Vegetation	8 8
3.3 3.4	Soils Literature Review	9 9
4.0	SITE CONDITIONS	16
4 1	Vegetation	
42	Hydrology	
4.3	Soils	10 16
4.3	Functions and Values	
5.0	FINDINGS	20
5.1	U.S. Army Corps of Engineers Determination	
5.2	California Regional Water Quality Control Board Determination	
5.3	California Department of Fish and Game (1602) Determination	
6.0	CONCLUSION OF REGULATORY APPROVAL PROCESS	25
6.1	Army Corps of Engineers	
6.2	Regional Water Quality Control Board	
6.3	California Department of Fish and Game	
6.4	Global Recommendations	
7.0	REFERENCES	28
LIST	OF EXHIBITS	

1.	Regional vicinity	2
2.	Project Site	3
3.	Site Vicinity	4
4.	Soil Survey Map	.15
5A.	On-Site Photographs	.17
5B.	On-Site Photographs	.18
6.	ACOE Jurisdictional Map	.21
7.	CDFG Jurisdictional Map	.24

<u>Page</u>

J,
LIST OF TABLES

1.	Jurisdictional Summary (Wetlands)	.20
2.	Jurisdictional Summary (ACOE Non-wetlands)	.22
3.	Jurisdictional Summary (CDFG)	.23

APPENDIX

A. Wetland Data I	Forms
-------------------	-------

- Β.
- Additional Site Photographs Vegetation Map (provided by BonTerra Consulting) C.



1.0 INTRODUCTION AND PURPOSE

This delineation was prepared for the County of Orange, Integrated Waste Management Department (IWMD) in order to delineate the U.S. Army Corps of Engineers' (ACOE), California Department of Fish and Game's (CDFG), and Regional Water Quality Control Board's (RWQCB) jurisdictional authority for drainages located within the Prima Deshecha Landfill, herein referred to as the project site.

The project site is located within the County of Orange and City of San Juan Capistrano, California (T.8S, R.7W, Sections 3, 4, 8, 9, 10, 15, 16, 17, SBBM) (refer to Exhibits 1-3). More specifically, the project site is located within the 1,530-acre landfill site currently under the management/ownership of the IWMD. The project site is located approximately 2.5 miles east of the Interstate 5 Freeway and 1.0-mile south of Ortega Highway; on-site access is provided via La Pata Road. Once on-site, access is generally provided via existing unimproved and improved roadways.

This delineation has been designed to document the authority of the regulatory agencies, the methodology undertaken by RBF Consulting (RBF) to document jurisdictional authority, and the findings made by RBF within the boundaries of the project site. This report presents our best effort at determining the jurisdictional boundaries using the most up-to-date regulations, written policy, and guidance from the regulatory agencies; however, only the regulatory agencies can make a final determination of jurisdictional boundaries.

1.1 Project Site Background

The IWMD currently owns and operates a portion of the project site that is utilized for landfill operations. The proposed project consists of the Second Amendment to the 2001 Prima Deshecha General Development Plan. A complete project description shall be submitted along with a formal application submittal to the resource agencies.



PRIMA DESHECHA LANDFILL • JURISDICTIONAL DELINEATION



Regional Vicinity

Exhibit 1



Exhibit 2



Source: National Geographics USGS.



not to scale

N G 2/19/04 JN 10-102970-10914



PRIMA DESHECHA LANDFILL • JURISDICTIONAL DELINEATION
Site Vicinity

DELINEATION OF JURISDICTIONAL WATERS

Prima Deshecha Landfill County of Orange, California

Prepared For:

BonTerra Consulting 151 Kalmus Drive, Suite E-200 Costa Mesa, California 92625 Contact: Mr. Gary Medeiros 714/444-9199

County of Orange Integrated Waste Management Department 320 N. Flower Street Santa Ana, California 92703-5000 *Contact: John Tzeng* 949/728-3045

Prepared By:

RBF Consulting

14725 Alton Parkway Irvine, California 92618 Contacts: Mr. Bruce R. Grove, Jr. 949/855-3686 Mr. Richard Beck 949/855-3687

June 7, 2004 Revisal of February 20, 2004 JN 10-102970



June 7, 2004

Mr. Gary Medeiros BONTERRA CONSULTING 151 Kalmus Drive, E-200 Costa Mesa, California 92626

SUBJECT: UPDATED DELINEATION OF JURISDICTIONAL WATERS Prima Deshecha Landfill, County of Orange

Dear Mr. Medeiros:

On behalf of RBF Consulting (RBF), we are pleased to submit this Delineation of Jurisdictional Waters for the above referenced project. The enclosed delineation was conducted in May 2003 to document the regulatory authority of the U.S. Army Corps of Engineers (ACOE), California Department of Fish and Game's (CDFG), and Regional Water Quality Control Board's (RWQCB) jurisdiction pursuant to the Federal Clean Water Act (CWA) and the State Fish and Game Code. The project area was surveyed pursuant to the ACOE's *1987 Wetland Delineation Manuel*, to identify evidence of flows, riparian vegetation, and hydric soils.

This report presents RBF's best effort at determining the jurisdictional boundaries using the most up-to-date regulations, written policy, and guidance from the regulatory agencies. However, as with any jurisdictional delineation, only the regulatory agencies can make a final determination of jurisdiction. Generally, this would be a written concurrence in the form of a Jurisdictional Determination (JD) letter.

Please note that based on a detailed review of current site conditions, our research has indicated that it will be necessary for the project applicant to successfully obtain the following permits prior to commencement of any construction activities within the delineated jurisdictional areas: Army Corps of Engineers 404 Permit, 1602 Streambed Alternation Agreement, and 401 Water Quality Certification.

Please do not hesitate to contact me at 949/855-3687, or Bruce Grove at 949/855-3686, if you or your staff has any questions or require further information.

Sincerely,

chand Beck

Richard Beck Regulatory Coordinator Environmental Services-Special Projects

M. F. From

Bruce R. Grove Jr., REA Project Manager Environmental Services-Special Projects



14725 Alton Parkway, Itvine, CA 92618-2027 # P.O. Box 57057, Itvine, CA 92619 7057 # 949 472.3505 # Fax 949.472.8373

Offices located throughout California, Arizona & Nevada 🔳 www.RB5.com

DELINEATION OF JURISDICTIONAL WATERS

Prima Deshecha Landfill County of Orange, California

The undersigned certify that this report is a complete and accurate account of the findings and conclusions of a jurisdictional "waters of the U.S." (including wetlands) and "waters of the State" determination for the above-referenced project.



Richard Beck

Richard Beck Environmental Analyst/ Regulatory Coordinator

Burg M.

Bruce R. Grove, Jr. Project Manager, REA Environmental Services-Special Projects

June 2004

TABLE OF CONTENTS

1.1 Project Site Background. 1 2.0 SUMMARY OF REGULATIONS 5 2.1 Army Corps of Engineers 5 2.2 Regional Water Quality Control Board. 6 2.3 California Department of Fish and Game 6 2.4 Activities Requiring Permits. 7 3.0 METHODOLOGY. 8 3.1 Vegetation 8 3.2 Hydrology. 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology. 16 4.3 Fonctions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 22 5.2 California Regional Water Quality Control Board Determination 22 5.2 California Department of Fish and Game 26 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 7.0 REFER	1.0	INTRODUCTION AND PURPOSE	1
2.0 SUMMARY OF REGULATIONS 5 2.1 Army Corps of Engineers 5 2.2 Regional Water Quality Control Board 6 2.3 California Department of Fish and Game 6 2.4 Activities Requiring Permits 7 3.0 METHODOLOGY 8 3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 9 5.0 FINDINGS 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 JU.S. Army Corps of Engineers Determination 22 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 6.5 California Department of Fish and Game 26 6.4 Global Recommendations </td <td>1.1</td> <td>Project Site Background</td> <td>1</td>	1.1	Project Site Background	1
2.1 Army Corps of Engineers 5 2.2 Regional Water Quality Control Board 6 2.3 California Department of Fish and Game 6 2.4 Activities Requiring Permits 7 3.0 METHODOLOGY 8 3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 9 3.4 Literature Review 9 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 5.3 California Department of Fish and Game (1602) Determi	2.0	SUMMARY OF REGULATIONS	5
2.2 Regional Water Quality Control Board	2.1	Army Corps of Engineers	
2.3 California Department of Fish and Game 6 2.4 Activities Requiring Permits 7 3.0 METHODOLOGY 8 3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.4 Hydrology 16 4.5 Soils 16 4.6 Soils 16 4.7 FUNDINGS 16 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25	2.2	Regional Water Quality Control Board	6
2.4 Activities Requiring Permits 7 3.0 METHODOLOGY 8 3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Soils 16 4.3 Soils 16 4.3 Soils 16 4.4 Hydrology 16 4.3 Soils 16 4.4 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Department of Fish and Game (1602) Determination 22 5.3 California Department of Fish and Game 25 6.4 Global Recommendations 25 6.5 Regional Water Quality Control Board 25 <t< td=""><td>2.3</td><td>California Department of Fish and Game</td><td>6</td></t<>	2.3	California Department of Fish and Game	6
3.0 METHODOLOGY 8 3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 20 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 3 3 3 1 Regional V	2.4	Activities Requiring Permits	7
3.1 Vegetation 8 3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 3 3 3 1 Regional Vicini	3.0	METHODOLOGY	8
3.2 Hydrology 9 3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 3 3 3 1 Regional Vicinity 4 4 Soil Survey Map 15 5.4	3.1	Vegetation	
3.3 Soils 9 3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Soils 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 3 3 3 1 Regional Vicinity 4 2 Project Site<	3.2	Hydrology	
3.4 Literature Review 9 4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Soils 16 4.3 Functions and Values 16 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 3 3 3 1 Regional Vicinity 2 2 Project Site 3 3 3 Site Vicinity 4 4 </td <td>3.3</td> <td>Soils</td> <td></td>	3.3	Soils	
4.0 SITE CONDITIONS 16 4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 20 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 1 Regional Vicinity 2 1 Regional Vicinity 4 3 Site Vicinity 4 4 Soil Survey Map 15 5A. On-Site Photographs 15 <td>3.4</td> <td>Literature Review</td> <td>9</td>	3.4	Literature Review	9
4.1 Vegetation 16 4.2 Hydrology 16 4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 2 1 Regional Vicinity 4 4 Soil Survey Map 15 5A. On-Site Photographs 17	4.0	SITE CONDITIONS	
4.2 Hydrology	4.1	Vegetation	
4.3 Soils 16 4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 2 1 Regional Vicinity 2 2 Project Site 3 3 Site Vicinity 4 4 Soil Survey Map 15 5A. On-Site Photographs 17	4.2	Hydrology	
4.3 Functions and Values 19 5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 1 Regional Vicinity 2 1. Regional Vicinity 2 2. Project Site 3 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 15	4.3	Soils	
5.0 FINDINGS 20 5.1 U.S. Army Corps of Engineers Determination 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 2 1 Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	4.3	Functions and Values	
5.1 U.S. Army Corps of Engineers Determination. 20 5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	5.0	FINDINGS	20
5.2 California Regional Water Quality Control Board Determination 22 5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 7.0 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	5.1	U.S. Army Corps of Engineers Determination	20
5.3 California Department of Fish and Game (1602) Determination 22 6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 28 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	5.2	California Regional Water Quality Control Board Determination	
6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS 25 6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 2 28 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	5.3	California Department of Fish and Game (1602) Determination	22
6.1 Army Corps of Engineers 25 6.2 Regional Water Quality Control Board 25 6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 28 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	6.0	CONCLUSION OF REGULATORY APPROVAL PROCESS	25
6.2 Regional Water Quality Control Board	6.1	Army Corps of Engineers	
6.3 California Department of Fish and Game 26 6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 28 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	6.2	Regional Water Quality Control Board	25
6.4 Global Recommendations 26 7.0 REFERENCES 28 LIST OF EXHIBITS 28 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	6.3	California Department of Fish and Game	
7.0 REFERENCES 28 LIST OF EXHIBITS 2 1. Regional Vicinity 2 2. Project Site 3 3. Site Vicinity 4 4. Soil Survey Map 15 5A. On-Site Photographs 17	6.4	Global Recommendations	26
LIST OF EXHIBITS 2 1. Regional Vicinity	7.0	REFERENCES	
1. Regional Vicinity	LIST	OF EXHIBITS	
 Project Site	1.	Regional Vicinity	2
 3. Site Vicinity	2.	Project Site	3
4. Soil Survey Map	3.	Site Vicinity	4
5A. On-Site Photographs	4.	Soil Survey Map	15
	5A.	On-Site Photographs	

CDFG Jurisdictional Map......24

5B.

6.

7.

Page

LIST OF TABLES

1.	Jurisdictional Summary (Wetlands)	20
2.	Jurisdictional Summary (ACOE Non-wetlands)	22
3.	Jurisdictional Summary (CDFG)	23
.		

APPENDIX

A. Wetla	nd Data For	ms
----------	-------------	----

- Β.
- Additional Site Photographs Vegetation Map (provided by BonTerra Consulting) C.



1.0 INTRODUCTION AND PURPOSE

This delineation was prepared for the County of Orange, Integrated Waste Management Department (IWMD) in order to delineate the U.S. Army Corps of Engineers' (ACOE), California Department of Fish and Game's (CDFG), and Regional Water Quality Control Board's (RWQCB) jurisdictional authority for drainages located within the Prima Deshecha Landfill, herein referred to as the project site.

The project site is located within the County of Orange and City of San Juan Capistrano, California (T.8S, R.7W, Sections 3, 4, 8, 9, 10, 15, 16, 17, SBBM) (refer to Exhibits 1-3). More specifically, the project site is located within the 1,530-acre landfill site currently under the management/ownership of the IWMD. The project site is located approximately 2.5 miles east of the Interstate 5 Freeway and 1.0-mile south of Ortega Highway; on-site access is provided via La Pata Road. Once on-site, access is generally provided via existing unimproved and improved roadways.

This delineation has been designed to document the authority of the regulatory agencies, the methodology undertaken by RBF Consulting (RBF) to document jurisdictional authority, and the findings made by RBF within the boundaries of the project site. This report presents our best effort at determining the jurisdictional boundaries using the most up-to-date regulations, written policy, and guidance from the regulatory agencies; however, only the regulatory agencies can make a final determination of jurisdictional boundaries.

1.1 Project Site Background

The IWMD currently owns and operates a portion of the project site that is utilized for landfill operations. The proposed project consists of the Second Amendment to the 2001 Prima Deshecha General Development Plan. A complete project description shall be submitted along with a formal application submittal to the resource agencies.



PRIMA DESHECHA LANDFILL • JURISDICTIONAL DELINEATION





CONSULTING

2/19/04 JN 10-102970-10914

Exhibit 1



Exhibit 2

2.0 SUMMARY OF REGULATIONS

There are three (3) key agencies that regulate activities within inland streams, wetlands, and riparian areas in California. The ACOE Regulatory Branch regulates activities pursuant to Section 404 of the Federal Clean Water Act (CWA), and Section 10 of the Rivers and Harbors Act. Of the State agencies, CDFG regulates activities under the Fish and Game Code Section 1600-1607, and the RWQCB pursuant to Section 401 of the CWA and the California Porter-Cologne Act.

2.1 Army Corps of Engineers

The ACOE has regulatory authority over the discharge of dredged or fill material into the waters of the United States under Section 404 of the CWA. The ACOE and Environmental Protection Agency (EPA) recently clarified and simplified the definition of "fill material" to include any "material placed in waters of the United States where the material has the effect of: (i) Replacing any portion of a water of the United States with dry land; or (ii) Changing the bottom elevation of any portion of the waters of the United States." Examples include, but are not limited to sand, rock, clay, construction debris, wood chips, and "materials used to create any structure or infrastructure in the waters of the United States." The term "waters of the United States" includes the following:

(1) all waters that have, are, or may be used in interstate or foreign commerce (including sightseeing or hunting), including all waters subject to the ebb and flow of the tide;

(2) wetlands;

(3) all waters such as interstate lakes, rivers, streams (including intermittent streams), mudflats, sandflats, wetlands, sloughs, prairie potholes, wet meadows, playa lakes, or natural ponds; the use, degradation or destruction of which could affect interstate or foreign commerce;

(4) all impoundments of water mentioned above;

(5) all tributaries of waters mentioned above;

(6) the territorial seas; and

(7) all wetlands adjacent to the waters mentioned above.

Under this definition, and in the absence of wetlands, the limits of the ACOE's jurisdiction in non-tidal waters extend to the ordinary high water mark (OHWM), which is defined as "...that line on the shore established by the fluctuations of water and indicated by physical characteristics such as a clear, natural line impressed on the bank, shelving, changes in the character of soil, destruction of terrestrial vegetation, the presence of litter and debris, or other appropriate means that consider the characteristics of the surrounding areas (33 CFR §328.3(e))."

Wetlands, a subset of jurisdictional waters, are jointly defined by the ACOE and EPA as "those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions (33 CFR §328.3(b))". Wetlands generally include swamps, marshes, bogs, and similar areas. The process in which jurisdictional areas (if any) are identified is further discussed in Section 3.0, Methodology.

It should be noted that a major change in wetland regulation occurred on January 9, 2001, when the U.S. Supreme Court issued the decision, *Solid Waste Agency of Northern Cook County v. U.S. Army Corps of Engineers et al* (SWANCC). The SWANCC decision limited the scope of the ACOE's Section 404 CWA regulatory permitting program as applied to isolated waters. The Supreme Court struck down the ACOE's jurisdictional authority over isolated, non-navigable, intrastate waters that are not tributary or adjacent to navigable waters or tributaries (i.e., wetland conditions). Overall, the Court held that Congress did not intend for isolated, non-navigable water conditions to be covered within Section 404 of the CWA, since they are not considered to be true "waters of the U.S."

2.2 Regional Water Quality Control Board

The RWQCB is the primary agency responsible for protecting water quality in California. The RWQCB regulates discharges to surface waters under the Federal CWA and the California Porter-Cologne Water Quality Control Act. The RWQCB's jurisdiction extends to all waters of the State and to all waters of the United States, including wetlands (isolated and non-isolated conditions).

Section 401 of the CWA gives the RWQCB the authority to regulate through 401 Certification any proposed federally permitted activity, which may affect water quality. Among such activities are discharges of dredged or fill material permitted by the ACOE pursuant to Section 404 of the CWA. Section 401 requires the RWQCB to provide "certification that there is reasonable assurance that an activity which may result in the discharge to waters of the United States will not violate water quality standards." Water Quality Certification must be based on a finding that the proposed discharge will comply with water quality standards, of which are found as numeric and narrative objectives in each of the nine (9) Regional Board's Basin Plan.

The Porter-Cologne Water Quality Control Act gives the State very broad authority to regulate waters of the State, which are defined as any surface water or groundwater, including saline waters. The Porter-Cologne has become an important tool in the post SWANCC era, with respect to the State's authority over isolated waters. Generally, any person proposing to discharge waste into a water body that could affect its water quality must file a Report of Waste Discharge (should there be no Section 404 nexus). Although "waste" is partially defined as any waste substance associated with human habitation, the RWQCB also interprets this to *include fill* discharged into water bodies.

2.3 California Department of Fish and Game

Historically, the State of California regulated activities in rivers, streams, and lakes pursuant to Sections 1600-1607 of the California Fish and Game Code. Legislation that took effect on January 1, 2004 repealed Fish and Game Code sections 1600-1607 and added Fish and Game Code sections 1600-1616. The most important issue to note with this change is that now there is no separation between private/public notifications (previously 1601/1603). Fish and Game Code section 1602 requires any person, state or local governmental agency, or public utility to notify the CDFG before beginning any activity that will do one or more of the following:

- 1) substantially obstruct or divert the natural flow of a river, stream, or lake;
- 2) substantially change or use any material from the bed, channel, or bank of a river, stream, or lake; or

3) deposit or dispose of debris, waste, or other material containing crumbled, flaked, or ground pavement where it can pass into a river, stream, or lake.

This notification process is referred to as a 1602 Streambed Alteration Agreement (SAA). Fish and Game Code section 1602 applies to all perennial, intermittent, and ephemeral rivers, streams, and lakes in the state.

Jurisdictional limits of the CDFG are not as clearly defined by regulation as those of the ACOE. While they closely resemble the limits described by ACOE regulations, they include riparian habitat supported by a river, stream, or lake regardless of the presence or absence of hydric soils and saturated soil conditions. Generally, the CDFG takes jurisdiction to the top of bank of the stream or to the outer limit of the adjacent riparian vegetation (outer drip line), whichever is greater. Notification is generally required for any project that will take place in or in the vicinity of a river, stream, lake, or their tributaries. This includes rivers or streams that flow at least periodically or permanently through a bed or channel with banks that support fish or other aquatic life and watercourses having a surface or subsurface flow that support or have supported riparian vegetation.

2.4 Activities Requiring Permits

Any development proposal that involves impacting drainages, streams, or wetlands on the site through filling, stockpiling, conversion to a storm drain, channelization, bank stabilization, road or utility line crossings, or any other modification would require permits from the ACOE, the RWQCB, and the CDFG before any development could commence on the project site. Both *permanent* and *temporary* impacts are regulated and would therefore trigger the need for permits.

There are two (2) different permit categories utilized by the ACOE, which include either a Nationwide Permit (NWP) or Individual Permit (IP). The specific permit required is primarily based on project description and jurisdictional impacts. The ACOE will not issue its authorization until the RWQCB completes the Section 401 Water Quality Certification. Processing of the 401 Certification with the RWQCB and SAA with the CDFG can occur concurrently with the ACOE permit process, since the agencies can utilize the same information and analysis. Applications to both the RWQCB and the CDFG require submittal of a valid California Environmental Quality Act (CEQA) document along with the application.



3.0 METHODOLOGY

Prior to visiting the project site, RBF conducted a review of United States Geological Survey (USGS) topographic maps, *aerial photographs*, the Soil Survey of Orange County and Western Part of Riverside County, California (dated 1978); and the State of California Hydric Soils List, (dated 1995), to identify areas that **may** fall under an agency's jurisdiction (refer to Section 3.4, *Literature Review*, for a complete discussion).

ACOE jurisdictional wetlands are delineated using the methods outlined in the ACOE Wetland Delineation Manual (1987). The methodology set forth in the 1987 Manual is based on the following three (3) indicators that are normally present in wetlands: (1) hydrology providing permanent or periodic inundation by groundwater or surface water, (2) hydric soils, and (3) hydrophytic vegetation. In order to be considered a wetland, an area must exhibit at least minimal hydric characteristics within these three parameters. As described in Section 2.0, ACOE non-wetland waters of the U.S. are delineated based on the limits of the OHWM as determined by erosion, the deposition of vegetation or debris, and changes in the vegetation. The RWQCB shares ACOE jurisdiction, unless isolated conditions are present. In the presence of isolated conditions, the RWQCB takes jurisdiction via the OHWM and/or the 3-parameter wetland methodology utilized by the ACOE. CDFG's jurisdiction is defined to the top of bank of the stream/channel or to the limit of the adjacent riparian vegetation.

Analysis presented in this document consists of field surveys and verification of current conditions conducted on May 5, 6, and 30, 2003. While in the field, jurisdictional areas were recorded onto a base map at an approximate scale of 1"= 300' using the topographic contours and visible landmarks as guidelines. It should be noted that areas within the northwestern portion of the landfill were not accessible during the course of this delineation due to current landfill activities. Once in the field, vegetation, soils, and evidence of hydrology were examined via the methodology listed below:

3.1 Vegetation

Nearly 5,000 plant types in the United States may occur in wetlands. These plants, known as hydrophytic vegetation are listed in regional publications of the U.S. Fish and Wildlife Service (USFWS). Cover of vegetation is estimated and is ranked according to their dominance. Species that contribute to a cumulative total of 50% of the total dominant coverage, plus any species that comprise at least 20% (also known as the "50/20 rule") of the total dominant coverage are recorded on a wetland data sheet (included in Appendix A, *Wetland Data Forms*). Wetland indicator status is assigned to each species using *The List of Plant Species that Occur in Wetlands* (USFWS, 1988). If greater than 50% of the dominant species from all strata were Obligate, Facultative-wetland, or Facultative species, the criteria for wetland vegetation was considered to be met. Plant indicator status categories are described below:

- Obligate Wetland (OBL): Plants that occur almost always (estimated >99 percent) in wetlands under natural conditions, but which may also occur rarely (estimated <1 percent) in non-wetlands (i.e., cattail or pickelweed).
- Facultative Wetland (FACW): Plants that occur usually (estimated >67 to 99 percent) in wetlands, but also occur (estimated 1 to 33 percent) in non-wetlands (i.e., mulefat or willow).

- Facultative (FAC): Plants with similar likelihood (estimated 33 to 67 percent) of occurring in both wetlands and non-wetlands.
- Facultative Upland (FACU): Plants that occur sometimes (estimated 1 to <33 percent) in wetlands, but occur more often (estimated >67 to 99 percent) in non-wetlands.
- Obligate Upland (UPL): Plants that occur rarely (estimated 1 percent) in wetlands, but occur almost always (estimated >99 percent) in non-wetlands under natural conditions.

3.2 Hydrology

If wetland vegetation criteria is met, the presence of wetland hydrology is evaluated at each transect by recording the extent of observed surface flows, depth of inundation, depth to saturated soils, and depth to free water in the soil test pits. In addition, indicators of wetland or riverine hydrology are recorded including the OHWM, drift lines, rack, debris, and sediment deposits. The lateral extent of the hydrology indicators are used as a guide for locating soil pits for evaluation of hydric soils and jurisdictional areas. In portions of the stream where the flow is divided by multiple channels with intermediate sand bars, the entire area between the channels is considered within the OHWM and the wetland hydrology indicator is considered met for the entire area.

3.3 Soils

There are approximately 2,000 named soils in the United States that occur in wetlands. Such soils, called hydric soils, have characteristics that indicate they were developed in conditions where soil oxygen is limited by the presence of saturated soil for long periods during the growing season.

Once in the field, soil characteristics are verified by digging soil pits along each transect to a depth of at least 16 inches. Soil pit locations are usually placed within the drainage invert or within adjoining vegetation. At each soil pit, the soil texture and color are recorded by comparison with standard plates within a Munsell Soil Chart (1994). Munsell Soil Charts aid in designating color labels to soils, based by degrees of three simple variables- hue, value, and chroma. Any indicators of hydric soils, such as redoximorphic features, buried organic matter, organic streaking, reduced soil conditions, gleyed or low-chroma soils, or sulfuric odor are also recorded. A hydric soil is a soil that formed under conditions of saturation, flooding, or ponding long enough during the growing season to develop anaerobic conditions (as previously listed) in the upper 16 inches. The concept of hydric soils includes soils developed under sufficiently wet conditions to support the growth and regeneration of hydrophytic vegetation. Soils that are sufficiently wet because of artificial measures are included in the concept of hydric soils. It should also be noted that the limits of wetland hydrology indicators are used as a guide for locating soil pits. If any hydric soil features are located, progressive pits are dug moving laterally away from the active channel until hydric features are no longer present within the top 16 inches of the soil profile.

3.4 Literature Review

As previously mentioned, RBF conducted a review of USGS topographic maps, San Juan Capistrano (Photorevised 1981), Cañada Governadora (Photorevised 1997), San Clemente (Photorevised 1975), and Dana Point (Photorevised 1975), California Quadrangles; aerial

photographs, provided by Eagle Aerial (dated 2002); the Soil Survey of Orange County and Western Part Riverside County, California (1978); and the State of California Hydric Soils List, (1995) prior to visiting the site. Review of relevant literature and materials often help preliminarily identify areas that **may** fall under an agency's jurisdiction. Examples of relevant information include, USGS blueline streams, vegetation map or aerial photographs, and hydric soils as listed within the U.S. Department of Agriculture (USDA) Soil Surveys. A summary of RBF's literature review is provided below (refer to Section 7.0, for a complete list of references used during the course of this delineation):

USGS Topographic Quadrangles, San Juan Capistrano (Photorevised 1981), Canada Governadora (Photorevised 1997), San Clemente (Photorevised 1975), and Dana Point (Photorevised 1975): The USGS maps show geological formations and their characteristics, describing the physical setting of an area through contour lines and major surface features including lakes, rivers, streams, buildings, landmarks, and other factors that may fall under an agency's jurisdiction. Additionally, the maps depict topography through color and contour lines, which are helpful in determining elevations and tatitude and longitude within a project site.

Based on the USGS California Quadrangles, on-site topography ranges from approximately 300 to 1000 feet above mean sea level (msl). Seven (7) USGS blue line streams are located within the boundaries of the project site. Blue line streams are generally referred to as streams that flow at least during part of the year. A stream does not need to be a specific size to be considered a blue line; however, these streams are generally considered jurisdictional by State and Federal agencies, as they are often tributaries to larger "navigable" jurisdictional waters. No on-site lakes, marshes, or swamps were noted during the review of the USGS topographic map.

Aerial Photograph: Prior to the May 2003 field visit, RBF reviewed an existing aerial photograph, provided by Eagle Aerial (flown in 2002) for the project site. Aerial photographs can be useful during the delineation process, as the photographs often indicate drainages and vegetation (i.e. riparian vegetation) present within the boundaries of the subject site (if any).

According to the aerial photograph, the eastern portion of the subject site consists of hilly topography. The western portion of the site consists of major grading, which is a result of current and historical landfill activities (Zones 1 and 2). Both riparian and upland habitat appear to be present on the aerial, with the riparian zones distinctly following areas of anticipated drainages.

Soil Survey: Orange County and Western Part of Riverside, California (1978): On-site and adjoining soils were researched prior to the May 2003 field visit. The presence of hydric soils is initially investigated by comparing the mapped soil series for the site to the County list of hydric soils. Soil surveys furnish soil maps and interpretations originally needed in giving technical assistance to farmers and ranchers; in guiding other decisions about soil selection, use, and management; and in planning research and disseminating the results of the research. In addition, soil surveys are now heavily utilized in order to obtain soil information within respect to potential wetland environments and jursidctional areas (i.e., soil characteristics, drainage, and color). According to the Orange County and Western Part of Riverside County, California Soil Survey, dated 1978, the proposed project site is situated on (1) general soil association, the Alo-Basanko association. The Alo-Basanko association consists of strongly sloping to steep, well-drained clays on coastal foothills. In addition to general soil associations, the project site is underlain by nine (9) soil series with multiple phases. The following is a brief description of the series, which underlie the proposed project site:

Alo clay, 9 to 15 percent slopes (100): This strongly sloping soil generally occurs on ridges and toe slopes in the foothills. About 5 percent of this mapping soil is included in areas of Bosanko clay, 9 to 15 percent slopes; 5 percent Anaheim clay loam; and 3 percent Balcolm clay loam. If the soil is bare, runoff is medium and the erosion hazard is moderate. Available water capacity is 3.5 to 7.0 inches. Present land use is citrus, dryland barley, pasture, range, and urban development. Subgroup Typic Chromoxererts.

Alo clay, 15 to 30 percent slopes (101): This moderately steep soil generally occurs on broad ridge tops in the foothills. About 5 percent of this mapping unit is included areas of Bosanko clay, 15 to 30 percent slopes; 5 percent Anaheim clay loam, 15 to 30 percent slopes; 3 percent slopes; 3 percent Balcom clay loam, 15 to 30 percent slopes; and 2 percent gravelly soils on some ridge tops. If the soil is bare, runoff is rapid and erosion hazard is high. Available water capacity is 3.5 to 6.0 inches. Present land use is dryland barley, pasture, range, and urban development. Subgroup Typic Chromoxererts.

Alo clay, 30 to 50 percent slopes (102): This steep soil generally occurs on side slopes in the foothills. About 5 percent of this mapping unit is included areas of Anaheim clay loam, 30 to 50 percent slopes; 5 percent Calleguas clay loam, 50 to 75 percent slopes, eroded; 2 percent Balcom clay loam, 30 to 50 percent slopes; and 2 percent Basanko clay, 30 to 50 percent slopes. If the soil is bare, runoff is rapid and the erosion hazard is high. Available water capacity is 3.5 to 5.5 inches. Present land use is range and watershed. Subgroup Typic Chromoxererts.

Anaheim clay loam, 15 to 30 percent slopes (108): This moderately steep soil generally occurs on broad ridgetops and north-facing side slopes. About 5 percent of this mapping until is included areas of Alo clay; 5 percent Anaheiim loam; 5 percent Nacimiento clay loam; 2 percent Cieneba sandy loam; and 3 percent Balcom clay loam. If the soil is bare, runoff is rapid and the erosion hazard is high. Permeability is moderately slow. Available water capacity is 4.0 to 7.0 inches. Present land use is pasture and range. Subgroup Pachic Haploxerolls.

Balcom clay loam, 15 to 30 percent slopes (112): This moderately steep soil generally occurs on hill ridgetops. About 5 percent of this mapping unit is included areas of Bosanko clay, 15 to 30 percent slopes; 4 percent Calleguas clay loam; 4 percent Cieneba sandy loam, 15 to 30 percent slopes; and some soils, similar to the Balcom soil, that have a dark gray surface layer. If the soil is bare, runoff is rapid and the erosion hazard is high. Available water capacity is 4 to 6 inches. Present land use is urban development, dryland barely, and dry pasture. Subgroup Calcixerollic Xerochrepts.

Bosanko clay, 9 to 15 percent slopes (126): This strongly sloping soil generally occurs on broad hilltop ridges and on toe slopes. About 5 percent of this mapping unit is included areas of Balcom clay loam, 9 to 15 percent slopes; 4 percent Alo clay, 9 to 15 percent slopes; and 10 percent soils that are similar to this Bosanko soil and are more than 40 inches deep. If the soil is bare, runoff is medium and the erosion hazard is moderate. Available water capacity is 3.5 to 6.5 inches. Present land use is citrus, dryland barley, pasture range, and urban development. Subgroup Chromic Pelloxererts.

Bosanko clay, 15 to 30 percent slopes (127): This moderately steep soil occurs on broad hilltop ridges. About 7 percent of this mapping unit is included areas of Balcom clay loam, 15 to 30 percent slopes; and 5 percent Alo clay, 15 to 30 percent slopes. If the soil is bare, runoff is rapid and the erosion hazard is moderate. Available water capacity is 3.5 to 6.0 inches. Present land use is dryland barley, pasture, range, and urban development. Subgroup Chromic Pelloxererts.

Bosanko clay, 30 to 50 percent slopes (128): This steep soil generally occurs on north-facing hillsides. About 7 percent of this mapping unit is included areas of Balcom clay loam, 30 to 50 percent slopes; and 5 percent Alo clay, 30 to 50 percent slopes. If the soil is bare, runoff is rapid and the erosion hazard is high. Available water capacity is 3.0 to 5.5 inches. Present land use is range and watershed. Subgroup Chromic Pelioxererts.

Botella loam, 2 to 9 percent slopes (131): This gently sloping to moderately sloping soil generally occurs on alluvial fans in long, narrow foothill valleys. About 5 percent of this mapping unit is included areas where slopes at the lower ends of fans or valleys are 0 to 2 percent; 5 percent Capistrano sandy loam, 2 to 9 percent slopes; 5 percent Botella clay loam, 2 to 9 percent slopes; and 3 percent Sorrento sandy loam, 2 to 9 percent slopes. If the soil is bare, runoff is medium and the erosion hazard is moderate. Present land use is irrigated citrus, dryland small grain, pasture, and range. Subgroup Pachic Argixerolls.

Botella clay loam, 2 to 9 percent slopes (132): This gently sloping to moderately sloping soil generally occurs on alluvial fans in narrow foothill valleys. About 5 percent of this mapping unit is included areas where slopes at the lower ends of fans or valleys are 0 to 2 percent; 5 percent Botella Loam, 2 to 9 percent slopes; 5 percent Sorrento clay loam; and 3 percent Mocho loam. If the soil is bare, runoff is medium and the erosion hazard is moderate. Present land use is irrigated citrus, dryland small grain, pasture, and range. Subgroup Pachic Argixerolls.

Calleguas clay loam, 50 to 75 percent slopes, eroded (134): This very steep soil generally has south-facing slopes. As much as 75 percent of the original surface layer has been lost in areas that have been cultivated, overgrazed, or burned because of sheet, rill, and gully erosion. Geologic erosion is active, and soil slipping is common. Many areas are a succession of short, vertical, exposures, or "cat steps". About 5 percent of this mapping unit is included areas of Cieneba sandy loam; 5 percent Balcom clay loam; 3 percent Anaheim clay loam; and 10 percent less sloping or steeper Calleguas clay loam. If the soil is bare, runoff is rapid and the erosion hazard is high. Present land use is range, watershed, and urban development. Subgroup Typic Xerorthents.

Cropley clay, 2 to 9 percent slopes (149): This gently sloping to moderately sloping soil generally occurs as irregular, oblong areas. About 10 percent of this mapping unit is included areas of Cropley clay, 0 to 2 percent slopes; 10 percent a soil that is very dark grayish brown to dark grayish brown clay but is otherwise similar to this Cropley soil; 7 percent Bosanko clay; and 3 percent Botella clay laom, 2 to 9 percent slopes. If the soil is bare, runoff is medium and the erosion hazard is slight. Present land use is citrus, barely, pasture, range, and urban development. Subgroup Chromic Pelloxererts.

Soper gravelly loam, 30 to 50 percent slopes (202): This steep soil generally occurs on hillsides. About 5 percent of this mapping unit is included areas of Yorba gravelly sandy loam, 2 percent Gabino gravelly clay loam, 5 percent Soper cobbly loam, 5 percent less sloping or steeper Soper gravelly loams, 2 percent Cieneba-Rock outcrop complex, and 2 percent Cieneba sandy loam. If the soil is bare, runoff is rapid and the erosion hazard is high. Available water capacity is 2.5 to 5.0 inches. Present land use is range, watershed, and wildlife habitat. Subgroup Typic Argixerolls.

Soper-Rock outcrop complex, 30 to 75 percent slopes (204): This mapping unit commonly occurs on hillsides and ridges. It is 10 to 15 percent Rock outcrop. The Soper soil is severely eroded. About 3 percent of this mapping unit is included areas of Anaheim loam, 5 percent Cieneba sandy loam, and 20 percent Soper cobbiy loam. If the soil is bare, runoff is rapid and the erosion hazard is high. Available water capacity is 2.5 to 3.5 inches. Present land use is range, watershed, and wildlife habitat. Subgroup Typic Argixerolls.

Sorrento clay loam, 2 to 9 percent slopes (209): This gently sloping to moderately sloping soil generally occurs on upper valley fans and along stream channels in 10- to 100-acre areas. About 10 percent of this mapping unit is included areas of Sorrento loam, 2 to 9 percent slopes; 5 percent soils that are noncalcareous throughout but are otherwise similar to his Sorrento soil; 5 percent Mocho loam, 2 to 9 percent slopes; 3 percent Botella clay loam, 2 to 9 percent slopes; and 3 percent areas of this Sorrento soil where slopes are somewhat steeper than 9 percent. If the soil is bare, runoff is slow to medium and the erosion hazard is slight to moderate. Available water capacity is 11.0 to 13.0 inches. Present land use is irrigated crops, citrus, and urban development. Subgroup Calcic Haploxerolls.

Based on the Orange County and Western Part of Riverside County, California Soil Survey the above soil series are typical of hydric soils due to the low permeability and porosity characteristic of high clay content soils. Soil pits were dug throughout the subject site to verify the list findings (refer to Section 4.2, *Site Conditions-Soils*, for a discussion of on-site soils). Also, refer to Exhibit 4, *Soil Survey Map*, for an illustration of approximate on-site soil locations.

Hydric Soils List of California (1995): RBF reviewed the Hydric Soils List of California, provided by the Natural Resources Conservation Service (NRCS), dated December 15, 1995 in an effort to verify whether or not on-site soils are considered to be hydric. Lists of hydric soils along with soil survey maps are good off-site ancillary tools to assist in wetland determinations, but as expected, they are not a substitute for on-site investigations. According to list, none of the above-mentioned soil series are listed as hydric.



- <u>Local Climate</u>: The local climate is typical of the Southern California Coastal Region. Winters are cool and moist; nearly all of the precipitation falls in winter. Summers are mild, warm, and dry. Average annual rainfall at most of the lower elevations of the region is approximately 14 inches. For this purposes of this delineation, the growing season is considered to be 365 days a year.
- <u>Groundwater Depth</u>: According to the soil survey mentioned above, soils within the project site generally reveal a groundwater depth of greater than six (6)-feet below ground surface (bgs).
- <u>Coastal Zone</u>: The project site is not located within the California Coastal Zone.



Exhibit 4

APP 40X:Martin

GUIJIOGNOD

4.0 SITE CONDITIONS

As described in Section 1.0, the proposed project is located in southern Orange County. Onsite access is generally provided via existing paved roadways, specifically La Pata Avenue. Refer to Sections 4.1 through 4.3, below, for discussion with respect to the three (3) wetland parameters defined in Section 3.0.

4.1 Vegetation

The vegetation present on the project site primarily consists of Grasslands; Coastal Sage Scrub, which includes, Mulefat Scrub, Sagebrush Scrub, Coyote Bush Scrub, Southern Willow Scrub; Mixed Sage Scrub/Grassland Ecotone; Woodland; Freshwater and Alkali Marsh; and Riparian. The Riparian vegetation consists of freshwater marsh, alkali marsh, southern willow scrub, and mulefat scrub. The riparian vegetation is primarily located along Prima Deshecha Cañada (Drainage G), where vegetation is most abundant within the project site.

A large majority of the project site appeared to consist of non-native species (e.g., thistle and mustard) during the May 2003 field visits. This type of vegetation occurred within and outside of jurisdictional drainages and was often dominant. In cases where hydrophytic vegetation was present (i.e., on-site wetlands), evidence of hydrology and soil samples were then warranted. Dominant hydrophytic vegetation present within the boundaries of the project site during the May 2003 field visits included willow and mulefat; however, other hydrophytes were noted as well.

4.2 Hydrology

As discussed during the topographic map review, several blue line streams and unnamed drainages are located within the boundaries of the project site. Prima Deshecha Cañada is the dominant drainage and flows generally in a southwestern direction, originating in the eastern portion of the project site. The majority of jurisdictional waters on-site consist of ephemeral drainage tributaries of Prima Deshecha Cañada, which generally contain water flow only during storm events.

Water flow was observed within Prima Deshecha Cañada and within the associated on-site spring during the May 2003 field visits. Depth to surface water ranged from approximately 2 inches to 2.5 feet (within the spring/open water area). Many of the ephemeral on-site drainages displayed evidence of hydrology (via drift/debris lines, scouring, and cut) sufficient to document the OHWM, and therefore met the hydrology standard for both wetland and non-wetland waters.

4.3 Soils

Soils within the boundaries of the project site were found to be consistent with those previously mentioned during the literature review in Section 3.4. Generally, on-site soils consisted of clays and loams (limited to heavy flow areas). Evidence of organic matter from surrounding vegetation was noted near Wetland 1 and Drainage G. Overall, on-site soil colors were typically light brow/grey in nature (Munsell 10 YR 3/3, 3/1 and 5YR 4/2); on-site soils appeared to be dry to moist, depending on the drainage and location. Soils within riparian and hydrophytic areas were consistently hydric (gleyed soils), due to the clayey soil moisture content. Refer to Appendix A, *Wetland Data Forms*, for specific detail on each of the soil pits dug during the May 2003 field visits.

Exhibit 5A

2/19/04 JN 10-102970-10914

PRIMA DESHECHA LANDFILL • JURISDICTIONAL DELINEATION Site Vicinity

View looking at open water / wetland habitat within the northeastern portion of the project site.



















Typical on-site drainage. Many drainages were not incised due to the dominant on-site soils (clays).



View looking upstream at Drainage V.



View of native riparian habitat (wetland) near the disturbed landslide remediation area.

2/19/04 10-102970-10914

PRIMA DESHECHA LANDFILL • JURISDICTIONAL DELINEATION

Exhibit 5B



4.3 Functions and Values

The project site is located within two (2) major watersheds, the Prima Deshecha Canada and Segunda Deshecha Canada watersheds. The project site primarily occupies the northernmost portion of the Prima Deshecha Cañada. In fact, the Prima Deshecha Canada watershed covers approximately 1,298 acres, which is roughly 84 percent of the project site. Prima Deshecha Canada is oriented such that flow occurs predominately in a southwesterly direction. A perennial spring is located in the northernmost reaches of Prima Deshecha Canada, at outcrops of the San Onofre Breccia near the head of the canyon. Surface flow continues over an area of approximately 1,200 linear feet, beyond which point all spring water has infiltrated into the alluvium of the streambed.

Wetlands and waters offer many functions and values beneficial to the general public and to the environment. Since each site is dynamic and unique, the functions and values vary greatly depending on hydrology, soil, vegetation, size, and location of the wetland in the region. Although jurisdictional areas may not serve all functions, each works together as part of a complex integrated system. A functional assessment of the jurisdictional areas on-site shall be provided during the formal resource agency application process.



5.0 FINDINGS

This delineation was prepared for the County of Orange IWMD in order to delineate the ACOE's, CDFG's, and RWQCB's jurisdictional authority for drainages located within the Prima Deshecha Landfill. This report presents RBF's best effort at determining the jurisdictional boundaries using the most up-to-date regulations, written policy, and guidance from the regulatory agencies. However, as with any jurisdictional boundaries within a project site/property. Jurisdictional boundaries are broken down specifically by agency and are described below.

5.1 U.S. Army Corps of Engineers (ACOE)

Wetland Determination:

As previously noted in Section 2.1, an area must exhibit **all three (3)** of the wetland parameters described in the *ACOE Wetland Delineation Manual* to be considered a jurisdictional wetland. Based on the results of the field investigations, it was determined that portions of the project site contained all three parameters. Based on the site conditions, **3.39-acres** of ACOE jurisdictional wetlands are present within the boundaries of the project site (refer to Table 1 and Exhibit 6).

Wetland	ACOE Area (Acres)	
W-1	0.40	de in free
W-2	0.76	
٧٧-3	0.30	
W-4	1.77	
W-5	0.02	
W-6	0.14	
Totals	3.39	. • •

Table 1 ACOE Jurisdictional Summary (Wetlands)

Waters of the U.S. (non-wetland) Determination:

Areas within the project site exhibited water flow and evidence of hydrology (Drift and debris lines, scouring, and cut) sufficient to document the OHWM, thus meeting the criteria for ACOE jurisdictional waters (non-wetland). The OHWM ranged from 2 feet to 20 feet. Based on the results of the field observations and data collection, **6.56-acres** of ACOE jurisdictional "waters of the U.S." are located within the boundaries of the project site (refer to Table 2).







Drainage	Linear Feet	Corps Width (feet)	Square Footage	Corps Area (Acres)
A	900	6	5,400	0.12
В	4,200	Varies	18,700	0.43
С	500	2	1,000	0.02
D	800	8	6,400	0.15
Е	400	2	800	0.02
F	2,800	2	5,600	0.13
G	5,800	Varies	35,600	0.82
H	400	2	800	0.02
1	500	2	1,000	0.02
U.	400	2	800	0.02
К	1,090	Varies	3,800	0.09
L	500	3	1,500	0.03
М	1,100	Varies	2,700	0.06
N	1,180	Varies	3,160	0.07
0	500	2	1,000	0.02
P	700	3	2,100	0.05
Q	3,550	Varies	10,500	0.24
R	2,000	3	6,000	0.14
S	500	2	1,000	0.02
Т	2,800	Varies	13,600	0.31
<u> </u>	1,300	2	2,600	0.06
V	2,200	Varies	25,800	0.59
W	300	2	600	0.01
X	2,000	2	4,000	0.09
Υ	2,400	4	9,600	0.22
Z	2,000	Varies	14,600	0.34
AA	3,500	Varies	17,600	0.40
BB	600	4	2,400	0.05
"Spring"	NA	NA	NA	0.24
Basin	NA	NA	NA	1.78
Totals	44.920	NA	198,660	6.56

Table 2 Jurisdictional Summary (ACOE Non-Wetland)

5.2 California Regional Water Quality Control Board Determination (RWQCB)

No isolated conditions were observed within the boundaries of the project site; therefore, the RWQCB follows that of ACOE jurisdictional "waters of the U.S."

5.3 California Department of Fish and Game Determination (CDFG)

The CDFG jurisdiction within the drainages extends from the top of bank to top of bank, and in areas of riparian vegetation, to the outer drip line. For the purposes of this delineation, the CDFG width for most drainages varied from 2 feet to 100 feet. Many times the jurisdiction

.

ł.,

expanded due to the presence of stands of riparian vegetation such as willow and/or mulefat. Based on the results of the field observations and data collection, **16.08-acres** of CDFG jurisdiction is located within the boundaries of the project site (refer to Table 3 and Exhibit 7).

Streambed	Linear Feet	CDFG Width (In feet)	Square Footage	CDFG Area (Acres)
A	900	6	5,400	0.12
8	4,200	Varies	26,700	0.61
С	500	2	1,000	0.02
D	800	8	6,400	0.15
Ē	400	4	1,600	0.04
F	2,800	Varies	9,600	0.22
G	10,600	Varies	203,200	7.21
н	400	10	4,000	0.09
1	500	2	1,000	0.02
J	400	16	6,400	0,15
к	1,090	Varies	11,800 .	0.27
L	700	Varies	11,000	0.25
М	1,170	Varies	3,170	0.07
N	1,180	Varies	6,960	0.16
0	500	10	5,000	0.11
Р	700	6	4,200	0.10
Q	3,550	Varies	23,060	0.53
R	2,000	20	40,000	0.92
S	500	2	1,000	0.02
Т	2,800	Varies	13,600	0.31
U.	1,300	Varies	2,600	0.06
V	2,500	Varies	28,800	0.66
W	300	6	1,800	0.04
x	2,000	2	4,000	0.09
Y	2,400	Varies	19,200	0.44
Z	2,500	Varies	38,500	0.88
AA	3,500	Varies	29,100	0.67
BB	600	6	3,600	0.08
Basin	NA	NA	NA	1.78
Totals	50,790	NA	512,690	16.08

Table 3 Jurisdictional Summary (CDFG)

6.0 CONCLUSION OF REGULATORY APPROVAL PROCESS

The following is a summary of the various permits, agreements, and certifications required before construction activities take place within the above-mentioned jurisdictional areas.

6.1 U.S. Army Corps of Engineers (ACOE)

The ACOE regulates discharges of dredged fill material into "waters of the United States" pursuant to Section 404 of the CWA. A federal permit will be required from the ACOE Regulatory Branch-Los Angeles District Office since improvements associated with the proposed improvements will result in the discharge of material within the ACOE's jurisdiction.

The project is currently under preliminary design and conceptual planning; therefore, specific impacts (permanent and temporary) to jurisdictional areas remain undefined. Based on the amount of jurisdiction on-site, the type of proposed project (i.e., land filling), and based on preliminary correspondence with the ACOE, it is anticipated that the proposed activities can be authorized via an IP/Letter of Permission (LOP).

An IP usually has a 30-day comment period under public noticing, though the time limit can be as short as 15 days. Processing time generally takes 9-12 months, but it is not uncommon for the processing time to last 1-3 years depending on the complexity and size of the subject project. The IP process generally involves a Pre-Application Field Meeting; environmental documentation (e.g., jurisdictional delineation, site plans, project purpose, location, duration, etc.); submittal of a Pre-Construction Notification (PCN); consultations with other agencies (as appropriate); a Section 404(B)(1) Alternatives Analysis and compliance determination; and a final ACOE permit decision. Prior to issuance of the ACOE permit, a CWA Section 401 Water Quality Certification from the RWQCB must be obtained.

6.2 Regional Water Quality Control Board (RWQCB)

For the ACOE 404 permit to be approved, a 401 Water Quality Certification from the San Diego RWQCB will be required. The RWQCB requires that CEQA compliance be obtained prior to obtaining the 401 Certification.

Once an application has been deemed complete, the RWQCB has between 60 days and 1 year in which to make a decision. According to regulations of the ACOE, the State has 60 days from the date of receipt of a valid request for water quality standards certification (33 CFR Section 325.2 (b) (1) (ii)). The ACOE district engineer may specify a longer (up to one year) or shorter time, if he or she determines that a longer or shorter time is reasonable (33 CFR Section 325.2 (b) (1) (ii)). If processing and review of the 401 application will take more than 60 days, the RWQCB will request additional time from the ACOE. Please note that even when an application has been deemed complete, the RWQCB has the option of denial without prejudice. This is not a reflection on the project, but a means to stop the clock until the required information has been required.

As required by 23 California Code of Regulations (CCR) § 3858 (a), the RWQCB is required to have a *minimum 21 day public comment period* before any action is taken on a 401 application. The period closes when the RWQCB acts on the 401 application. The public comment period does not close after a certain number of days because proposed projects tend to change

through the 401 process and the public is allowed to review and comment on the changed project. The public comment period starts as soon as an application has been received.

Additionally, the RWQCB requires that water quality concerns related to urban storm water runoff be addressed. Any 401 Certification application submitted to the RWQCB should incorporate the use of BMPs for the treatment of pollutants carried by storm water runoff in order to be considered a complete application. The RWQCB also requires a 401 Certification Application Fee, which is dependent on the amount of impacts.

6.3 California Department of Fish and Game (CDFG)

As noted within this delineation, on-site drainages within the project site meet the CDFG's definition as streambed. Since improvements associated with the proposed project will impact CDFG Jurisdiction, a 1602 Streambed Alteration Agreement (SAA) must be obtained prior to impact.

After the CDFG is notified, the CDFG will determine whether the notification package (application) is complete. The CDFG will make this determination within 30 calendar days of receiving the notification package if the application is for a regular agreement (i.e., an agreement for a term of five years or less). Once the notification package is complete, the CDFG will process it as described below. The 30-day time period does *not* apply to notifications for long-term agreements (i.e., agreements for a term greater than five years).

If a SAA is required, the CDFG conducts an onsite inspection, if necessary, and prepares a draft agreement. The draft agreement will include measures to protect fish and wildlife resources while conducting the project. For regular agreements, the CDFG will submit a draft agreement to the applicant within 60 calendar days after the notification is deemed complete. Again, the 60-day time period does not apply to notifications for long-term agreements, since these are often large or complex projects.

The applicant then has 30 calendar days to notify the CDFG whether the measures in the draft agreement are acceptable. After the CDFG receives the signed draft agreement, it will make it final by signing it. The CDFG Agreement will require a \$1390.50 fee and CEQA compliance is necessary in order for the SAA to be issued.

6.4 Global Recommendations

Agency Concurrence and Pre-Application Field Meeting:

Prior to the application process, it is highly recommended that the delineation be forwarded to each of the regulatory agencies for their concurrence. Once the delineation is approved, RBF has found it extremely beneficial and pro-active to have an on-site meeting with the ACOE, CDFG, and RWQCB to discuss potential permitting strategies and mitigation opportunities (if any). In short, these Pre-Application Field Meetings often help streamline the permitting process.

Potential Mitigation:

Mitigation can take several forms. It can consist of (1) avoidance or minimization of impacts, (2) compensation in the form of habitat creation, or (3) compensation through participation in a mitigation bank. The first type of mitigation (avoidance or minimization of impacts) is preferred by the agencies and should be investigated to the maximum extent possible. For any project that impacts riparian vegetation, it is preferred by the agencies that compensation through the creation of habitat be performed on-site and in kind (i.e., riparian woodland for riparian woodland; sandy bottom for sandy bottom). The exact requirements of any special conditions established in the permit would be dictated by the regulatory agencies following review of the formally submitted project application.
7.0 REFERENCES

Aerial Photograph, provided by Eagle Aerial, flown in 2002.

California Department of Fish and Game, Lake and Streambed Alteration Program, http://www.dfg.ca.gov/1600/index.html

Common Riparian Plants of California, Pickleweed Press 1996.

Common Wetland Plants of Coastal California, Pickleweed Press 1996.

Final Environmental Impact Report No. 575, General Development Plan, Prima Deshecha, 2001.

Munsell Soil Color Charts, 1994.

National List of Vascular Plant Species that Occur in Wetlands, U.S. Fish and Wildlife Service, 1988.

Natural Resources Conservation Services, Hydric Soils List of California, 1995. http://soils.usda.gov/soil_use/hydric/main.htm

Site Visits conducted on May 5, 6, and 28, 2003.

Soil Survey, Orange County and Western Part of Riverside County, California, 1978.

Thomas Brothers Map, Orange and Los Angeles Counties, 2003.

U.S. Army Corps of Engineers, Los Angeles District Regulatory Program, http://www.spl.usace.army.mil/

<u>U.S. Army Corps of Engineers</u>, Final Summary Report: Guidelines for Jurisdictional Determinations for Waters of the United States in the Arid Southwest, June 2001.

U.S. Army Corps of Engineers, Wetland Delineation Manual, 1987.

U.S. Fish and Wildlife Service, http://endangered.fws.gov/consultations/index.html

<u>USGS Topographic Maps</u>, Canada Gobernadora (photorevised 1997), Dana Point (photorevised 1975), San Clemente (photorevised 1975), San Juan Capistrano (photorevised 1981), California, Quadrangles.

Vegetation Map, provided by BonTerra Consulting 2003.

A) Wetland Data Forms

ž

100 C 100 - 10

1-120

ġ

1

.

は空かけない

Distances and

the state of the second

W WALLAND

and we say

ş

national sector

P.0050000

-

. .

Appendix

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u><u>-</u></u>	Date May 5, 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R Beck / C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Marsh
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 3-4

VEGETATION

.

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Yerba Mansa	herb	# OBL	9		
2 Muletat	Shrub	FACW	10		
3			11		
4			12		
5	·		13		
6	· · ·		14		
7			15		
8			16		
Percent of Dominant Species that a	are OBL, FAC	W, or FAC (e	xcluding FAC-) 790%.		-
Remarks Limited we 3-4 Mulefa	dy s t shr	pecies ub-s	noted.		

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Feviewed) Other No Recorded Data Available FIELD OBSERVATIONS			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands
			Secondary Indicators (2 or more Required):
Depth of Surface Water	None	(ពា)	Oxidized Root Channels in Upper 12 Inches
Depth to Free Water in Pit	716.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	716.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

Map Unit Name (S	Series and Phase)	Calleguas cla	y loam	Drainage Class:	WD			
axonomy (Subgr	oup) Typic Xer	orthents	Field Observations Confirm Mapped Type? YES NO					
		PROF	LE DESCRIPTION					
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Con Structure	cretions, , etc.		
1-2	0	1042 3-1	None	NA	Organic	Mat		
2-16	A	10YR 3.1	Nore	NA	clay			
16-	B_	2.5YR/2.51	None	NA	chan			
					J			
<u>_</u>	ļ							
<u></u>	 	······						
****					· · · · · · · · · · · · · · · · · · ·			
	l		<u> </u>	.				
Histic Epip Sulfidic Od Aquic Mois Reducing C Gleyed or L	edon lor ture Regime Conditions _ow-Chroma Colo	75	High Or Organic Listed o Listed o Other (E	ganic Content in Surfac Streaking in Sandy Soi In Local Hydric Soils Lis In National Hydric Soils I Explain in Remarks)	e Layer in Sand Is t List	ly Soils		
Wois top	st soils O lau	, heavy jer.	organic	material c	a long	·		
TLAND DETE	ERMINATION tion Present?	YES NO	ls this Sampling P	oint Within a Watland?				

Remarks 15-17 wide, upstream of spring. Flat wetland area.

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	H'	Date May 5, 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. BECK/C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Usland
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 2

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator	
1 Thistle	shrub	UPL	9			
2 Mulefat E	Shrub	FACW	10			
3 41	1.		11			
4			12			
5			13			
6		· · · ·	14			
7			15			
8			16			
Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	xcluding FAC-) 51.			
Similar to plot ID1. One () muletat shrub noted.						

IT DROLOGI					
 Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Peviewed) Other No Recorded Data Available 			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits		
FIELD OBSERVATIONS		Drainage Patterns in Wetlands			
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):		
Depth to Free Water in Pit	716.0	(in ⁾	Water-Stained Leaves		
Depth to Suturated Soil	716.0	(in)	FAC-Neutral Test		

Taxonomy (Subgroup) Typic Chromostever+5 Field Observations Confirm Mapped Type? YES NO PROFILE DESCRIPTION PROFILE DESCRIPTION Texture, Concreding Mottle Texture, Concreding Mottle Texture, Concreding Mottle Intervention Matrix Color Mottle Colors Mottle Texture, Concreding Mottle Texture, Concreding Mottle Intervention None NA Silling - C Intervention None NA Silling - C Intervention Intervention None NA Silling - C Intervention Intervention Intervention Intervention Silling - C Intervention Intervention Intervention Intervention Silling - C Intervention Intervention Intervention Intervention Intervention Intervention Intervention Intervention Intervention	Map Unit Name (S	Series and Phase	e): Alo clay	~ _	Drainage Class:	
PROFILE DESCRIPTION Depth (inches) Horizon Matrix Color (Munsell Moist) Mottle Colors Abundance/Contrast Mottle Structure, e I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 A IOTR 3/2 None NA Sill hard C I-16 I INDICATORS: Indicator Indicator Inditor Indicator <tht< td=""><td colspan="3">Taxonomy (Subgroup) Typic Chromoseverts</td><td>Field Observation:</td><td>s Confirm Mapped Type</td><td>? YES NO</td></tht<>	Taxonomy (Subgroup) Typic Chromoseverts			Field Observation:	s Confirm Mapped Type	? YES NO
Depth (inches) Horizon Matrix Color (Munsell Moist) Mottle Colors Mottle Abundance/Contrast Texture, Concr Structure, e I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL A IotR 3/2 None NA Silling - C I-IL IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 I-IL IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 I-IL IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 IotR 3/2 <td< td=""><td></td><td></td><td>PROFI</td><td>LE DESCRIPTION</td><td>·····</td><td></td></td<>			PROFI	LE DESCRIPTION	·····	
Indext strain None NA Silling - C Image: Strain Strain Image: Strain Strain Silling - C Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Strain Image: Strain Image: Strain	Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretion Structure, etc
Histosol Concretions Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	1-16	<u>A</u>	10YR 32	None	NA	Silty - Clar
Histosol Concretions Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	······································					
Histosol Concretions Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)		 				
HYDRJC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)					 	
HYDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)						
Histosol Concretions Histosol Histo Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy S Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)		· · · · · · · · · · · · · · · · · · ·				
	Histosol Histic Epipe Sulfidic Odd Aquic Moist Reducing C Gleyed or L	edon or ture Regime tonditions ow-Chroma Colo	rs	Concretii Concretii High Org Organic : Listed on Listed on Other (E)	ons Janic Content in Surface Streaking in Sandy Soils Local Hydric Soils List National Hydric Soils Li Aplain in Remarks)	e Layer in Sandy Soils S
(lid(KS)	marks:		······································			

Hydrophytic Vegetation Present?	YES (NO)	
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wotland?
Hydric Soils Present?	YES NO	The and camping tome ventility ventility ventility ventility ventility
Remarks Clayen soils	, no a	lominat FACt veg.
	· · · · · · · · · · · · · · · · · · ·	

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Dailaga Decherha Lands	rill	Date May 5, 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator P Back (C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Non-native
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Third SD	shrub	U.P.L	9		
2			10		
3			11		
4		1	12		
5	h	<u> </u>	13		
6	<u></u>		14		
7			15		
8		1	16		
Percent of Dominant Species tha	it are OBL, FAC	W, or FAC (excluding FAC-) O /.		
Remarks Aren i inde	rt domi	nated	by thistle.		

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Reviewed) Other No Recorded Data Available			WETLAND HYDROLOGY INDICATORS Primary Indicators: (Vd. of Stour / Cut Inundated OHWM. Saturated in Upper 12 Inches 4/FT. Water Marks Drift Lines Sediment Deposits
FIELD OBSERVATIONS			L. Drainage Patterns in Wetlands
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0	(in) :	Water-Stained Leaves
Depth to Saturated Soil	76.0	(in)	Cher (Explain in Remarks)

Map Unit Name (S	Map Unit Name (Series and Phase): Alo Clay			Drainage Class: Ѡ	D
Taxonomy (Subgr	oup) Typic Chi	rom oxever +5	Field Observations	Confirm Mapped Type	YES NO
		PROFI	E DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.
1-16	A	2.54/3	None	NA	Silty-Clay
			· · · · · · · · · · · · · · · · · · ·		1
<u> </u>					
<u> </u>		<u>+</u>			·
			·		·
			· · · · · · · · · · · · · · · · · · ·		. <u></u>
	[,,				
Histosol Histic Epipe Sulfidic Ode Aquic Moist Reducing C	edon or ture Regime conditions ow-Chroma Colors	s	Concretic Concretic High Org Organic Listed on Listed on Other (E)	ons anic Content in Surface Streaking in Sandy Soils Local Hydric Soils List National Hydric Soils Li kplain in Remarks)	Layer in Sandy Soils s
emarks: Roc.IC o	utcroppine	مح أم الحودة	ur noted	•	

Hydrophytic Vegetation Present?	YES (NO)		
Wetiand Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	YES NO
Hydric Soils Present?	YES NO		
Remarks Jurisdictional	tributo	ry (F)	
		,	

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands		Date May 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator PROCK/C.Johnson		state California
Do Normal Circumstances exist on the site?	(YES) NO	Community ID Marsh
Is the site significantly disturbed (Atypical Situation)?	YES (NO)	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 5-6

٦

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
	1 Yerba Marsa	herb	OBL	9		
1	2			10		
	3			11		
	4			12		
4	5			13		
	6			14		
-	7	Í		15		
-	8			16		
	Percent of Dominant Species that a	are OBL, FAC	W, or FAC (excluding FAC-) 790 /.		
-	Remarks Similar Content.	to. Thick	plots er vec	3-4 more layer.	water	

HYDROLOGY

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availa	in Remarks) Gauge Levitwed) ble	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Vater Marks Crift Lines Sediment Deposits
FIELD OBSERV	ATIONS	Drainage Patterns in Wetlands
Depth of Surface Water	None Mulot king	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0 (in	Water-Stained Leaves
Depth to Saturated Soil	716.6 (in	General Test General Test Other (Explain in Remarks)

Map Unit Name (S	Series and Phase);	Calleguas clas	loam	Drainage Class: 🕻	N _
Taxonomy (Subgr	oup) Typic Xev	orthents	Field Observations	Confirm Mapped Type	YES NO
		PROFIL	E DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-2	0	10 4231	None	NA	O. Mater
2-16	A	107R3-1	None	1	Clan
16-	<u> </u>	2.5YR/2.5	None		Clan
					<u> </u>
				, , , , , , , , , , , , , , , , , , , ,	
					[
		·	·		· · · · · · · · · · · · · · · · · · ·
······]			<u> </u>	
		HYDRIC S			
	edoa			ons Contant in Surface	the second s
	lor			anic Coment in Surface Streaking in Sandy Soil	e Layer in Sandy Solls
🗖 Aquic Mois	sture Regime		Listed on	Local Hydric Soils List	
Reducing (Conditions		🔲 Listed on	National Hydric Soils L	list
Sieyed or I	Low-Chroma Color	5	🗋 Other (E	xplain in Remarks)	
Remarks:	· · · · · · · · · · · · · · · · · · ·		·········		

ydrophytic Vegetation Present?	YES NO	
etland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?
ydric Soils Present?	YES NO	
emarks		

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u></u>	Date May 2003
Applicant/Owner County of Orange-	TWMA	County Orange
Investigator R. Beck /C. Johnson		state California
Do Normal Circumstances exist on the site?	(YES) NO	Community ID Marsh /Spring
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Piot ID 7

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1	Thistle	Shrub	WPL	9		
2	Cattarl	Shrub	OBL	10		
3	Mulerat	Shrub	FACUS	11		
4	Salt Grass	herb	8L	12		
5	Munwort	Shrub	FACW	13		
6				14		
7				15		
8				16		
Per	cent of Dominant Species that a	re OBL, FAC\	N, or FAC (e	xcluding FAC-) 780 /.	·-	
Ren	Thick vege Water prese	tation	n. A.	top of spring.		

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availa	in Remarks) Gauge eviewed) ble		WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSERV	ATIONS		Drainage Patterns in Wetlands
Depth of Surface Water	2.0	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	0	(in)	Water-Stained Leaves
Depth to Saturated Soil	1.0	(in)	General Test General Test Other (Explain in Remarks)

ap Unit Name (Se	eries and Phase):	Calleguas clay	loan	Drainage Class:	WA			
axonomy (Subgro	up)-Typic Xeron	thents	Field Observations Confirm Mapped Type? YES NO					
		PROFIL	E DESCRIPTION					
Depth (inches)	Horizon	Matrix Color (Munsell Mojst)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.			
1-16	A	2.54/3.1	None	NA	Silty Clay			
					· · · · · · · · · · · · · · · · · · ·			
	₩. 							
					· · · · · · · · · · · · · · · · · · ·			
		- *						
				·	<u> </u>			
				ions				
Histic Epipe	don			nanic Content in Surface	Laver in Sandy Soils			
Sulfidic Odo	ĸ		Organic	Streaking in Sandy Soil	s			
Aquic Moist	u r e Regime		Listed or	n Local Hydric Soils List				
Reducing Co	onditions		Listed or	n National Hydric Soils L	ist			
Gleyed or Lo	ow-Chroma Colors	š	🖾 Other (E	xplain in Remarks)				
			·····					

Hydrophytic Vegetation Present?	YES NO	
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	TES NO	
Remarks		······································

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u></u>	Date May 5 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck / C. Johnson	-	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Narsh Spring
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 8

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator			
1	Muletont	Shrub	Fren	9					
2	Saltarass	herb	OBL	10					
3				11					
4				12					
5				13					
6	·			14					
7				15					
8				16					
Perce	ent of Dominant Species that a	re OBL, FAC	W, or FAC (e	xcluding FAC-)					
Rema	Remarks Mulefat à Sultanss along water's edge.								

HYDROLOGY

i

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availa	in Remarks) Gauge Leviewed)	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSERV	ATIONS	Drainage Patterns in Wetlands
Depth of Surface Water	Nore (in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	4.0 (in)	Water-Stained Leaves
Depth to Saturated Soil	2.0 (in)	General Test General Test General Test General Test General Test General Test General Science

Map Unit Name (S	eries and Phase)	<i>Calleguas</i>	clar	loam	Drainage Class: 🚺	<u>d</u> t			
Taxonomy (Subgro	oup) Typic Xe	rothents		Field Observations	Confirm Mapped Type	? YES NO			
		F	ROFIL	E DESCRIPTION					
Depth (inches)	Horizon	Matrix Co (Munsell M	olor loist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.			
1-16	<u> </u>	31	YR	None	NA	Silly. Clay			
				···		/ /			
				<u> </u>					
] 							
				<u></u>					
	<u>.</u>	<u> </u>				<u> </u>			
		HYI	DRIC S						
	oda-				ons				
	or				janic Content in Surface Streaking in Sandy Sail	e Layer in Sandy Soils			
	lure Renime			Listed or	Streaking in Sandy Soil				
	Conditions				National Hydric Soils I	ist			
Gleyed or L	ow-Chroma Colo	rs		Other (E	xplain in Remarks)				
Remarks:	Remarks:								

1

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	(YES) NO		
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	•
Hydric Soils Present?	CYES NO		

Remarks

On-site spring. Open water noted.

4

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u>411</u>	Date May 5,2003
Applicant / Owner County of Orange-	TWMD	County Orange
Investigator R. Beck /C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Marsh / spring
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 9

VEGETATION

Γ	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1	Mulefat	Shrub	FACW	9		
2	Saltquase	herb	OBL	10		
3				11		
4				12		
5	······································			13		
6	······································			14		
7	v		T	15		
8				16		
P	ercent of Dominant Species that	it are OBL, FAC	W, or FAC (excluding FAC-) 770%	· · · · · · · · · · · · · · · · · · ·	
R	emarks					
	Mundet 2 SAL	longest al	ong was	ter's edge.		
	MUMERTALI & SOM		ر	N		
1						

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Peviewed) Other No Recorded Data Available FIELD OBSERVATIONS			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands		
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):		
Depth to Free Water in Pit	4.0	(in)	Water-Stained Leaves Local Soil Survey Data		
Depth to Saturated Soil	2.0	(in)	FAC-Neutral Test Other (Explain in Remarks)		

Map Unit Name (S	Series and Phase)	colleguas cla	loam	Drainage Class: WD		
axonomy (Subgr	OUP) TYPE Xe	<i>votuents</i>	Field Observations	Confirm Mapped Type?	YES NO	
·		PROFIL	E DESCRIPTION	······································		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.	
1-16	A	3/148	None	NA	Silty-clay	
<u> </u>		· · · · · · · · · · · · · · · · · · ·				
······						
F 3		HYDRIC S	SOIL INDICATORS:			
Histosol				ions		
	edon		High Or	ganic Content in Surface	Layer in Sandy Soils	
Sulfidic Od	lor		Organic Streaking in Sandy Soils			
LI Aquic Mois	ture Regime		Listed o	n Local Hydric Soils List		
Reducing (Conditions		Listed of	n National Hydric Soils L	ist	
Gleyed or l	Low-Chroma Colo	rs	D Other (E	xplain in Remarks)		
ernarks:	·	·····		· · · · · · · · · · · · · · · · · · ·		
	·					

ł

Hydrophytic Vegetation Present?	YES	NO	
Wetland Hydrology Present?	VES	NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	TES	NO	
Remarks			
On-site spring)		
open water v	whed.		
• 1			
· · · ·			

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	H.	Date May 5, 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck (C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Marsh / Spring
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID /O

VEGETATION

ĺ	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator				
	1 Mulefat	Shrub	PACW	<u>9</u>						
	2 Salterrass	hens	OBr	10						
	3			11						
	4			12						
	5			13						
ľ	6	······································		14						
Ê	7			15						
99- 1	8			16						
	Percent of Dominant Species that are OBL, FACW, or FAC (excluding FAC-) > 70%									
Ī	Remarks	Remarks								
Mulefat & saltgrass along water's edge.										

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Peviewed) Other No Recorded Data Available FIELD OBSERVATIONS			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands
Dopth of Surface Water		(in)	Secondary Indicators (2 or more Required):
	None	(m)	Oxidized Root Channels in Upper 12 Inches
Depth to Free Water in Pit	4.0	(in)	 Water-Stained Leaves Local Soil Survey Data
Depth to Saturated Soil	2.0	(in)	General Test Other (Explain in Remarks)

Map Unit Name (S	Series and Phase):	Callegues clay	loam	Drainage Class;	
Taxonomy (Subgr	oup) Typic Xer	othents	Field Observations	Confirm Mapped Type	YES NO
		PROFIL	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-14	A	3/148	None	NA	Silty - clay
		· · · · · · · · · · · · · · · · · · ·			
				· · · · · · · · · · · · · · · · · · ·	
	· · · · ·				
		HIDRIC :			
	edon			ions ia Caataat ia Suufaar	
Sulfidic Oc	lor			Streaking in Sandy Soil	e Layer in Sandy Solis
Aquic Moisture Regime			Listed or	n Local Hydric Soils List	3
Reducing (Conditions		Listed of	n National Hydric Soils L	ist
Gleyed or I	Low-Chroma Color	s	Other (E	xplain in Remarks)	
Remarks:			·····		

Hydrophytic Vegetation Present?		NO				
Wetland Hydrology Present?	প্ৰিন্থ	NO	Is this Sampling Point Within a Wetland?	ES	NO	ŗ
Hydric Soils Present?			7			
Remarks						
Open wate	w not	ed				

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u>-</u>	Date May 5,2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Marsh / spring
is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Muletat	Shrub	FACW	9		
2 Saltgrass	here	OBL	10		
3			11		
4			12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that	t are OBL, FAC	W, or FAC (excluding FAC-) 770%		
Remarks Multat & So	utopass	ومعله	water's edge		

 Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Peviewed) Other No Recorded Data Available 			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSER	RVATIONS		L Drainage Patterns in Wetlands
Depth of Surface Water	Nove	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	4.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	2.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

Map Unit Name (S	eries and Phase)	Callequar clay	loan	Drainage Class: 🕠			
Taxonomy (Subgro	oup) Typic Xe	evothents	Field Observations Confirm Mapped Type? (YES) NO				
		PROFIL	E DESCRIPTION				
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.		
1-16	<u>A</u>	3/1 YR	None	NA	Silty-aay		
· · · · · · · · · · · · · · · · · · ·							
			*				
					· · · · · · · · · · · · · · · · · · ·		
		<u> </u>					
F1		HYDRIC S					
				ons			
	edon			ganic Content in Surface	Layer in Sandy Soils		
	ture Regime			Streaking in Sandy Sola	5		
Reducina (Conditions			National Hydric Soils List	ist		
Gleyed or Low-Chroma Colors Other (Explain in Remarks)							
Remarks:							

ł

Hydrophytic Vegetation Present?	VES NO	
Wetland Hydrology Present?	TES NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	NO NO	7
Remarks		_1
open water h	040	

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u>Gill</u>	Date May 5,2003
Applicant/Owner County of Orange-	IWMA	County Orange
Investigator R. Beck/C. Johnson	· · · · · · · · · · · · · · · · · · ·	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Uplance.
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 12

VEGETATION

Domina	nt Plant Species	Stratum	Indicator	Ì	Dominant Plant Species	Stratum	Indicator
1 Thi	stle	Shab	UPL	9			
2 Mu	lefert	shrub	FACW	10			
3	<u>, , , , , , , , , , , , , , , , , , , </u>			11			
4				12			
5				13			
6				14			
7	, ,, -= un, y			15			
8				16			
Percent of Do	ominant Species that a	ire OBL, FAC	W, or FAC (e	exclud	ing FAC-) 450 /		
Remarks This	He, limit	ed m	ulesat	•			

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Feuiewed) Other No Recorded Data Available			WETLAND HYDROLOGY INDICATORS Primary Indicators: OHWM AUtod Inundated OFT Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSERV	ATIONS		L. Drainage Patterns in Wetlands
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0	(in)	Water-Stained Leaves Local Soil Survey Data
Depth to Saturated Soil	716.0	(in)	General Test General Test Other (Explain in Remarks)

ap Unit Name (S	Series and Phase):	Callegues cla	y loan	Drainage Class:	-
axonomy (Subgr	DUP) TYPIC XC	vor thents	Field Observation	s Confirm Mapped Type	? (YES) NO
		PROFIL	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretion: Structure, etc.
1-16	4	DYR32	Nore	AG	Silty Clay
		<u> </u>	<u> </u>		•
					· · · · · · · · · · · · · · · · · · ·
	<u> </u>				
				<u></u>	·
		HYDRIC S	SOIL INDICATORS		
Histosol				tions	
Histic Epip	edon		🗍 High Or	rganic Content in Surfac	e Layer in Sandy Soil
🔲 Sulfidic Od	lor		🔲 Organir	- c Streaking in Sandy Soi	ls
Aquic Mois	sture Regime			on Local Hydric Soils Lis	t
	Conditions		Listed c	n National Hydric Soils	r Fist
	Low-Chroma Colo	re de la companya de		Evolois is Pamarks)	LI3.
······································		·······			
marks;					

I

:

Hydrophytic Vegetation Present?	YES NO		_
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	YES NO
Hydric Soils Present?	(ES) NO	1	
Remarks			
Turiediction	al Ta	Lutara	
Junearcho			
		÷	

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u>All</u>	Date May 5,2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck/C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Ripanian .
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 13-14

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Mulpfut	Shrub	FACW	9		
2 Sn/thru55	herb	032	10		
3			11		
4			12		
5			13		
6 ·			14		
7			15		
8			16		
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	excluding FAC-) 750 7.		
Remarks hydrophytic	vegete	tion	in drainage	- 1	

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Feuiewex) Other No Recorded Data Available EIELD OBSERVATIONS			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands
Depth of Surface Water	2 0.5	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	4.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	4.0	(in)	☐ FAC-Neutral Test ☐ Other (Explain in Remarks)

Taxonomy (Subgroup) Typic Xerror trunts Field Observations Confirm Mapped Type? YES NO PROFILE DESCRIPTION PROFILE DESCRIPTION Mattix Colors Mattle Colors Mattle Colors Mattle Colors Structure, concretions 1-16 A OTK 3 Nome NA Structure, etc. 1-16 A OTK 3 Nome Nome Structure, etc. 1-16 A OTK 3 Nome Nome Structure, etc. Hybric Solo Concretions Concretions High Organic Content in Surface Layer in Sandy	Map Unit Name (S	eries and Phase): Calleguas cla	y loam	Drainage Class:	~
PROFILE DESCRIPTION Depth (inches) Horizon Matrix Color (Munsell Moist) Mottle Colors Mottle Abundance/Contrast Texture, Concretions I-ILe A D/K 3 Nor NA S; I+g-Cla I-ILe A D/K 3 I I I I I-ILe A D/K 3 I Nor NA S; I+g-Cla I-ILe A D/K 3 I </td <td>Taxonomy (Subgro</td> <td>oup) Typic Xe</td> <td>vor thents</td> <td>Field Observation</td> <td>is Confirm Mapped Type</td> <td>P? YES NO</td>	Taxonomy (Subgro	oup) Typic Xe	vor thents	Field Observation	is Confirm Mapped Type	P? YES NO
Depth (inches) Horizon Matrix Celor (Munsell Moist) Mottle Colors (Munsell Moist) Mottle Abundance/Contrast Texture, Concretions Structure, etcj. I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Norc NA Si Ity - Cla I-IL A DIR 3 Interview Interview Interview I-IL A DIR 3 Interview Interview Interview I-IL A DIR 3 Interview Interview Interview Interview I-IL A Interview Interview Interview Interview Interview Interview I-IL Interview Interview Interview Interview Interview Interview I-IL Interv			PROF	LE DESCRIPTION		
I-ILG A D/K 3 None NA Si I+y-Cla Image: Stress of the str	Depth (inches)	Horizon	Matrix C o lor (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contras	Texture, Concretions t Structure, etcy
Image: Substant Street Stre	1-16	A	107K 31	None	NA	silty-Cla
Histosol Image: Concretions Histosol Concretions Histic Epipedon Image: Concretions Sulfidic Odor Image: Concretions Aquic Moisture Regime Image: Listed on Local Hydric Soils List Reducing Conditions Image: Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	<u></u>					
Histosol Image: Concretions Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)						
Histosol Histosol Histic Epipedon Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	· · · · · · · · · · · · · · · · · · ·					
Histosol Concretions Histosol High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	<u> </u>					
HYDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	<u> </u>					
HYDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)			······································			
Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)		for a second	HYDRIC	SOIL INDICATORS	· · · · · · · · · · · · · · · · · · ·	
Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks)	🔲 Histosof				tions	
Sulfidic Odor Organic Streaking in Sandy Soils Organic Streaking in Sandy Soils Organic Streaking in Sandy Soils Listed on Local Hydric Soils List I Reducing Conditions Gleyed or Low-Chroma Colors Other (Explain in Remarks)	Histic Epipe	edon		🛛 High O	rganic Content in Surfac	ce Layer in Sandy Soils
Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Emarks	Sulfidic Od	or minor	ON SK-)	🗌 Organi	c Streaking in Sandy So	oils
Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Emarks	Aquic Mois	ture Regime		Listed (on Local Hydric Soils Lis	st
Gleyed or Low-Chroma Colors Other (Explain in Remarks)	Reducing Conditions					
Remarks:	Gleyed or Low-Chroma Colors Other (Explain in Remarks)					
	emarks:					

C. Andrewson and Providence of the local division of the local div

Hydrophytic Vegetation Present?	YES NO		A	
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	VES NO	
Hydric Soils Present?	VES NO	-		
Remarks			······	

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u></u>	Date May 5, 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R Reck /C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Near Rocal
Is the site significantly disturbed (Atypical Situation)?	(YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Piot ID 15-18

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Muletat	Shrub		9		
2 thistle	Shrub		10		
3			11		
4			12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	excluding FAC-) 750		
Remarks Some vegeta Over near u	ition a unimpa	ppeare oved r	d to be grade. bad, multiplin	l)Fill crea	red

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs Other No Recorded Data Available			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Designed Batierpo in Wetlands
FIELD OBSEF	RVATIONS		Drainage Patterns in Wettanos
Depth of Surface Water	20	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	2.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	4.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

Map Unit Name (Se	eries and Phase)	Drainage Class:				
Taxonomy (Subgroup) Typic Chromoxerers Fiel			Field Observations	S Confirm Mapped Type?	YES NO	
······································		PROFIL	E DESCRIPTION			
Depth (inches)	Horizon	Matrix Color (Munsell Moist),	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.	
1-16	A	104R3/1	Nor	NA	Silty Cla.	
					·····	
				· · · · · · · · · · · · · · · · · · ·		
<u>-</u>			<u> </u>			
			SOIL INDICATORS:			
Histosol				ions		
🔲 Histic Epipe	edon		🔲 High Or	ganic Content in Surface	e Layer in Sandy Soils	
🔹 🕽 🖉 uifidic Odd				- Streaking in Sandy Soil	s	
Aquic Moisture Regime						
Reducing Conditions						
Gleyed or L	ow-Chroma Cold	ors	🔲 Other (E	Explain in Remarks)		
lemarks:	. 1	· 1				
Very	vet i	moist cl	ays.			
- /	·		/			

Hydrophytic Vegetation Present?	CYES NO	
Wetland Hydrology Present?	VER NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	VES NO	
Remarks Area disturb	ed. Ro	ughly 487 wide.

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	All	Date May 5, 2003
Applicant / Owner County of Orange-	IWMD	County Orange
Investigator R. Beck C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Ripanin
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID /9

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator	
1 Mukfat	Shrub	FACW	9			
2			10			
3			11		<u>_</u>	
4			12			
5			13			
6			14			
7			15			
8			16			
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	xcluding FAC-) 780/.			
Remarks Trimarily all mulefat. Some non-natives within center						

Recorded Data (Describe i Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availal	in Remarks) Gauge eviewed) ble	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits	
FIELD OBSERV	ATIONS		Drainage Patterns in Wetlands
Depth of Surface Water	None	Secondary Indicators (2 or more Required):	
Depth to Free Water in Pit 716.0 (in)			Water-Stained Leaves
Depth to Saturated Soil	716.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

Map Unit Name (S	Series and Phase): Calleguas clai	loam	Drainage Class:	
Taxonomy (Subgr	oup) Typic Xe	vorthents	Field Observations	Confirm Mapped Type	YES NO
		PROFI	LE DESCRIPTION	*	
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	A	104R 3/1	None	NA	Silty-Clay
	ļ 				/ 1
					· · · · · · · · · · · · · · · · · · ·
	}				
			· · · · · · · · · · · · · · · · · · ·		
<u> </u>		-			· · · · · · · · · · · · · · · · · · ·
	<u> </u>	HYDRIC	L SOIL INDICATORS:		
Histosol				ions	
🔲 Histic Epip	edon		🗌 High Org	ganic Content in Surface	e Layer in Sandy Soils
🔲 Sulfidic Od	lor		🔲 Organic	Streaking in Sandy Soil	S
Aquic Mois	sture Regime		Listed or	n Local Hydric Soils List	
Reducing (Conditions			n National Hydric Soils I	_ist
Gleyed or I	Low-Chroma Cold	ors	Ll Other (E	xplain in Remarks)	
Remarks:		20 201	denter	1-	
Kocky	on sid		e a nord		
					1

			······································			
Hydrophytic Vegetation Present?	YES	NO				
Wetland Hydrology Present?	YES	NO	Is this Sampling Point Within a Wetland?	YES	NO	
Hydric Soils Present?	YES	NO				
Remarks						

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u> </u>	Date May 5 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Reparison
Is the site significantly disturbed (Atypical Situation)?	YES 0	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 20

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1			9		
2			10		
3			11		
4			12		
5			13		
6			14		
7			15		
8			16		· · ·
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	excluding FAC-) 780/		
Remarks Same a	5 P(a	st 19	w/ mulebat	•	

Recorded Data (Describe in Rer Stream, Lake, or Tide Gaug Aerial Photographs Gevie Other No Recorded Data Available	marks) e wed)	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Dreipage Patterns in Wetlands	
FIELD OBSERVATIO	641		
Depth of Surface Water (in)		Secondary Indicators (2 or more Required):	
Depth to Free Water in Pit (in)		Water-Stained Leaves Local Soil Survey Data	
Depth to Saturated Soil	(in)	General Test General Test Other (Explain in Remarks)	

Map Unit Name (S	eries and Phase)	Callegnas clay	bam	Drainage Class:	-			
Taxonomy (Subgro	DUP) TYPIC Xer	ortheats	Field Observations Confirm Mapped Type? (YES) NO					
PROFI			LE DESCRIPTION					
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Murisell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.			
		HYDRIC	SOIL INDICATORS:					
Histosol Histic Epipe	edon or		Concreti High Org Organic	ions ganic Content in Surface Streaking in Sandy Soil	e Layer in Sandy Soils s			
🔲 Aquic Mois	ture Regime		Listed on Local Hydric Soils List					
Reducing Conditions Gleyed or Low-Chroma Colors			 Listed on National Hydric Soils List Other (Explain in Remarks) 					
Remarks:								

1

J.

1

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	(YES) NO	\overline{D}
Wetland Hydrology Present?	(YES) NO	Is this Sampling Point Within a Wetland? CYES NO
Hydric Soils Present?	(TES) NO	
Continuetio	n of	P(o+1)
		·

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	HII	Date	May 5.2	2003
Applicant/Owner County of Orange-	TWMD	County	Orangé	
Investigator R Beck / C. Johnson		State	Califor	nia
Do Normal Circumstances exist on the site?	YES NO	Commu	nity ID	R.p. Vez
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transec	et 1D	,
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID	21	

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator		
1 Mulesat	Shab	FACIN	9				
2 Willow	Tren	FACW	10				
3			11				
4			12				
5			13				
6			14				
7			15				
8	•		16				
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	excluding FAC-)				
Remarks Wear Coubird Station.							

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other	in Remarks) Gauge Ceviewed) ble	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Device a Duty of Marketer	
FIELD OBSER\	ATIONS		L Drainage Patterns in Wetlands
Depth of Surface Water	Nore	Secondary Indicators (2 or more Required):	
Depth to Free Water in Pit 76.0 (in)		Water-Stained Leaves	
Depth to Saturated Soil	716.0	(in)	☐ FAC-Neutral Test ☐ Other (Explain in Remarks)

Taxonomy (Subgroup) Typic Xerortructs Field Observations Confirm Mapped Type? YES NO PROFILE DESCRIPTION Depth Horizon Matrix Color (Munsell Moist) Mottle Colors Abundance/Contrast Texture, Concreti Structure, etc Image: Im	Map Unit Name (S	Series and Phase	Calleguas cl	ay loam	Drainage Class:		
PROFILE DESCRIPTION Depth (inches) Horizon Matrix Color (Munsell Moist) Mottle Colors Mottle Abundance/Contrast Texture, Concreti Structure, etc I-IC I NTR3/I Nmc NA CIGY I-IC I Indication Indication Indication Indication I-IS Histosol Indication Indication Indication	Taxonomy (Subgr	oup) Typic X	evortunts	Field Observations	Confirm Mapped Type'	YES NO	
Depth (inches) Horizon Matrix Color (Munsell Moist) Mottle Colors Abundance/Contrast Texture, Concreti Structure, etc I-IC I NTR3/I Now NA CIGY I-IC I I I I I I I-IC I I I I I I I-IC I I I I I I I I-IC I I I I I I I I			PROFIL	E DESCRIPTION			
Initial Initia	Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.	
HYDRIC SOIL INDICATORS: Histosol Histosol Histo Epipedon Sulfidic Odor Aquic Moisture Regime Reducing Conditions Gleyed or Low-Chroma Colors Histo Soils List Gleyed or Low-Chroma Colors Histo Aquic Advice Advi	1-16	A	NTR3/1	Nore	NA	Clay	
Histosol HyDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area. Very Clayey	<u>.</u>						
Histosol Histosol Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy So Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area. Very Clayey							
Histosol HyDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Hypical of area, Very Clayey							
Histosol Concretions Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Sol Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area, Very Clayey							
HYDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area. Very Clayey							
Hydric Soil INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area. Very Clayey	.		-			· · · · · · · · · · · · · · · · · · ·	
HYDRIC SOIL INDICATORS: Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Soils Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soil's Mpical of area, Men Clayey		•					
Histosol Concretions Histic Epipedon High Organic Content in Surface Layer in Sandy Sol Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area, Very Clayey			HYDRIC S	SOIL INDICATORS:			
 Histic Epipedon Sulfidic Odor Aquic Moisture Regime Reducing Conditions Gleyed or Low-Chroma Colors High Organic Content in Surface Layer in Sandy Sol Organic Streaking in Sandy Soils Listed on Local Hydric Soils List Other (Explain in Remarks) 	Histosol			Concret	ions		
Sulfidic Odor Organic Streaking in Sandy Soils Aquic Moisture Regime Listed on Local Hydric Soils List Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils Mpical of area, Very Clayey	🔲 Histic Epip	edon		🔲 High Org	ganic Content in Surface	e Layer in Sandy Soils	
 Aquic Moisture Regime Reducing Conditions Gleyed or Low-Chroma Colors Emarks: Soils Mpical of area. Very Clayey 	🔲 Sulfidic Od	for		Organic	Streaking in Sandy Soil	s	
Reducing Conditions Listed on National Hydric Soils List Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Other (Explain in Remarks) Soils Hypical of area, Very Clayey	🗍 Aquic Mois	sture Regime		Listed o	n Local Hvdric Soils List		
Gleyed or Low-Chroma Colors Other (Explain in Remarks) emarks: Soils typical of area. Very Clayey				Listed on National Hydric Soils List			
Soils typical of area. Very clayey	Gleyed or L	Low-Chroma Cold	Irs	Other (E	xplain in Remarks)		
Soils typical of area. Very clayey	emarks:					· · · · · · · · · · · · · · · · · · ·	
Sours apprend on areas, at partycy	e de	Lace		cea Vi	m claup	,	
	Jona	gpica			1	7	
		,				1	

WETLAND DETERMINATION

Wetland Hydrology Present? YES NO Is this Sampling Point Within a Wetland? YES NO Hydric Soils Present? YES NO	Hydrophytic Vegetation Present?	(YES)NO	
Hydric Soils Present? YES NO	Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?
	Hydric Soils Present?	YES NO	

Remarks

pear road.

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u><u><u></u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	Date May 5,2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck /C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Rip Veg.
Is the site significantly disturbed (Atypical Situation)?	YES D	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 23

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Muletat	Shrus	FHCW	9		
2 Willow	Tree	PACW	10		
3			11		
4			12		
5			13		
6			14		
7		·····	15		
8			16		
Percent of Dominant Species that	t are OBL, FAC	W, or FAC (excluding FAC-) (00%		
Remarks Near coubird	d Statio	~			

Recorded Data (Descrit Stream, Lake, or Tic Aerial Photographs Other No Recorded Data Ava	ilable	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands	
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit کول. ۵ (in)		Water-Stained Leaves	
Depth to Saturated Soil	716.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

S	О	۱L	S
---	---	----	---

Map Unit Name (S	Series and Phase)	: Calleguas cu	ay loam	Drainage Class:	
Taxonomy (Subgr	OUP) TYPIC X	evor thents	Field Observations	Confirm Mapped Type?	YES NO
		PROFIL	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	M	104R3/1	None	ACI	Clary
		······			
	<u>_</u>				
· ·				· · · · · · · · · · · · · · · · · · ·	
			¦		
<u>, , , , , , , , , , , , , , , , , , , </u>	<u>.</u>	HYDRIC	SOIL INDICATORS:	<u> </u>	ļ
🗔 Histosol			Concreti	ons	
Histic Epip	bedon		🔲 High Org	janic Content in Surface	e Layer in Sandy Soils
Sulfidic Od	dor		Organic	Streaking in Sandy Soil	s
Aquic Moisture Regime Listed on Local Hydric Soils List					
	Conditions		Listed or	ι National Hydric Soils I	list
Gleyed or	Low-Chroma Cold	ors	🔤 🗌 Other (E	xplain in Remarks)	
Remarks:		<u> </u>			
Soils	typical of	f avra. Ven	y clayey.		
,	_		- • •		

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	E	NO				
Wetland Hydrology Present?	YES	NO	Is this Sampling Point Within a Wetland?	YED	NO	
Hydric Soils Present?	(TES)	NO				
Demente						

Remarks

Near road.

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands		Date May 6, 2003
Applicant / Owner County of Orange-	TWMD	County Orange
Investigator R. Beck C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Upland
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 24

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Thistle.	Shrub	UPL	9		
2 Muletat <10%.	shrub	FACW	10		
3			11		
4			12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	excluding FAC-) 450/.		
Remarks Near road, H	istle a	abunde	ant.		

Recorded Data (Describe Stream, Lake, or Tid Aeriai Photographs Other No Recorded Data Avail	e in Remarks) e Gauge Reviewed) able	WETLAND HYDROLOGY INDICATORS Primary Indicators: OHW M noted Inundated 3 FF Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Draipage Patterns in Wetlands
Depth of Surface Water	Nore (in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0 (in)	Water-Stained Leaves
Depth to Saturated Soil	76.0 (in)	FAC-Neutral Test Other (Explain in Remarks)
SOILS

avanamy (Ruber			WEYCENT STOPLE			
axonomy (Subgro	oup) Typic Ch	romoxeverts	Field Observations	Confirm Mapped Type'	? (YES)NO	
		PROFI	LE DESCRIPTION		and the second sec	
Depth (inches)	Horizon	Matrix Color (Munsell Meist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.	
1-/6	A	104R33			Clay-ban	
			 		ļ	
					· · · · · · · · · · · · · · · · · · ·	
					,	
·	·					
	w	<u>_</u>		<u> </u>		
		HYDRIC	SOIL INDICATORS:			
Histosol			Concreti	ons		
Histic Epipe	ədon		🛄 High Org	ganic Content in Surface	e Layer in Sandy Soils	
	OF		U Organic	Streaking in Sandy Soil	S	
	ture Regime		Listed or	n Local Hydric Soils List		
	onditions		Listed on National Hydric Soils List			
LI Gleyed or L	ow-Chroma Colo	rs	🖵 Other (E	xplain in Remarks)		
marks:				11 7		
Sails	dry c	lays inf	ermixed	with #	Sands.	
	-					

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	YES (NO)			Í
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	YES	(NO)
Hydric Soils Present?	YES	5		
Jurisdictional	draina	ge.		

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands		Date	May	62	2003	
Applicant/Owner County of Orange-	TWMD	County	Orb	inge		
Investigator R. Beck C. Johnson		State	Cal	ifori	nia	
Do Normal Circumstances exist on the site?	YES NO	Commu	nity ID	S.EAS	f Area	
Is the site significantly disturbed (Atypical Situation)?	YES C	Transec	t ID 🕳			
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID	25	-		

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1	see notes.			9		
2		1		10		
3				11		
4				12		
5				13		
6				14		
7				15		
8				16		
F	Percent of Dominant Species that a	are OBL, FAC	W, or FAC (e	excluding FAC-)		
F	Remarks Bank area Part Slide Loce Plu	Som Mul ot Z	e 70% etat 6).	son Oak. Pos just down st	sible ream.	

Recorded Data (Describe i Stream, Lake, or Tide Aerial Photographs Other	in Remarks) Gauge eviewed) Die	WETLAND HYDROLOGY INDICATORS Primary Indicators: DHWM OF Inundated /2 FF. Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSERV	ATIONS	Drainage Patterns in Wetlands
Depth of Surface Water	None (in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	御716.0 (in)	Water-Stained Leaves Local Soil Survey Data
Depth to Saturated Soil	7 6,6 (in)	FAC-Neutral Test Other (Explain in Remarks)

SOILS

avanaany / O. L	······································	WEANKO CLAY, "	T TO IS PENCENT Slope		
	UP) Chromic P	WIOXEVEN 15	Field Observations	Confirm Mapped Type	YES NO
-		PROFIL	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist) 	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	A	NYRZI	3 Nore	1A	Clause
		3	/	· · · · ·	
		+	<u>+</u>		
		<u></u>	<u> </u>		
			 		· · · · · · · · · · · · · · · · · · ·
					· · · · · · · · · · · · · · · · · · ·
Aquic Moisti Reducing Co Gleyed or Lo	ure Regime onditions ow-Chroma Colo	rs	Listed on Listed on Other (E)	Local Hydric Soils List National Hydric Soils L xplain in Remarks)	.ist
TLAND DETEI	RMINATION		1	······	· · · · · · · · · · · · · · · · · · ·
TLAND DETEI	RMINATION	YES NO		· · · · · · · · · · · · · · · · · · ·	
TLAND DETEI drophytic Vegetati etland Hydrology F	RMINATION ion Present? Present?	YES NO	Is this Sampling Po	int Within a Wetland?	YES NO
TLAND DETEI drophytic Vegetati etland Hydrology F dric Soils Present	RMINATION ion Present? Present? ?	YES NO YES NO YES NO	Is this Sampling Po	int Within a Wetland?	YES NO

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands		Date May 6,2003
Applicant/Owner County of Orange.	TWMD	County Orange
Investigator R. Beck C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID S. EAST Aren .
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 24

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator		
1 Saltarass	herb	OBL	9				
2 Mulcint	Shrub.	FAEW	10				
3 Picklewred <10%	herb	OBL	11				
4			12				
5			13				
6			14				
7			15				
8			16 //				
Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	excluding FAC-)7 80/1		-		
Remarks Pickleweed on alope. Very small & Minor. Multint w/in invert.							

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availal FIELD OBSERV	in Remarks) Gauge eviewed) ble	WETLAND HYDROLOGY INDICATORS Primary Indicators: Strong Witten Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands	
Depth of Surface Water	None	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	716.0	(in)	General Test General Test General Test Other (Explain in Remarks)

SOILS

Map Unit Name (Se	eries and Phase)	Bosanko clay e	1-15 % SLopes	Drainage Class: 🖒	JD
ʻaxonomy (Subgro	oup) Chromic	Pelioxeverts	Field Observations	s Confirm Mapped Type'	YES NO
		PROFI	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Coldr (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	<u> </u>	104231	None	NA	Clayer,
	· · · · · · · · · · · · · · · · · · ·				
	₩. ₩	<u> </u>			
	·				·
		HYDRIC	SOIL INDICATORS:		
🔲 Histosol				ions	
Histic Epipe	don		🔲 High Or	ganic Content in Surface	e Layer in Sandy Soils
Sulfidic Odc	рг		🗋 Organic	Streaking in Sandy Soil	s
🔲 Aquic Moist	ure Regime		🗆 Listed o	n Local Hydric Soils List	
🔲 Reducing C	onditions		Listed o	n National Hydric Soils L	ist
Gleyed or Lo	ow-Chroma Colo	rs	🗉 🗌 Other (B	Explain in Remarks)	
rmarks:					·
		·····			
TLAND DETE	RMINATION	~ ¬			

Hydrophytic Vegetation Present?	(YES) NO	
Wetland Hydrology Present?	VED NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	YES NO	
Wetland habit	tat nea	r site boundary.

1

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u>	Date May 6 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck /C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID S. East Avea
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 27

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Saltomass	here	081	9		
2 Mule fet	Shines	PHCW	10		
3 Pickleweg <10%	Nerro	OBL	11		
4			12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that	t are OBL, FAC	W, or FAC (e	excluding FAC-) 🥱 802		
Remarks Pickleweed on Muletat W/in	Slope. V invert	eng sma	el & minor.		

 Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs Ceviewed Other No Recorded Data Available 			WETLAND HYDROLOGY INDICATORS Primary Indicators: Strong cut a Inundated 12.41. Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits		
FIELD OBSER	VATIONS		Drainage Patterns in Wetlands		
Depth of Surface Water None (in)			Secondary Indicators (2 or more Required):		
Depth to Free Water in Pit > \\u. 0 (in)		Water-Stained Leaves			
Depth to Saturated Soil	٥.جا ١ <	General FAC-Neutral Test			

SOILS

Map Unit Name (Series and Phase): Bosanco (Loy, 9013			13 % slopes	Drainage Class: ເມເ	> ~		
Taxonomy (Subgre	oup) Chromit	Pelioxeverts	Field Observations	Field Observations Confirm Mapped Type? (YES) NO			
		PROFI	LE DESCRIPTION				
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.		
1-16	A	104R 3/1	None	NA	Clayey		
,	· · · · · · · · · · · · · · · · · · ·				· · · · · · · · · · · · · · · · · · ·		
· · · · · · · · · · · · · · · · · · ·							
		<u></u>					
[L		
-		HYDRIC	SOIL INDICATORS:				
Histosol				ions			
Histic Epipe	edon			ganic Content in Surface	+ Layer in Sandy Solis		
	ture Regime			Streaking in Sandy Sola n Local Hydric Soils List	5		
Reducing (Conditions			n National Hydric Soils L	_ist		
🖾 Gleyed or t	Low-Chroma Color	rs	🔲 Other (E	Other (Explain in Remarks)			
Remarks:				Marth Anna ann an			

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	VES NO	
Netland Hydrology Present?	RES NO	Is this Sampling Point Within a Wetland?
lydric Soils Present?	NO NO	
Wetland habito	t near	site boundary.

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

1987	COE	Wetlands	Delineation	Manual	

Project'Site Prima Deshecha Lands		Date May 30, 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck (C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Reparison
is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 28

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1	plustat a	301	FACW	9		
2	•			10		
3				11		
4				12		
5				13		
6		:		14		
7				15		
8				16 /		
Per	rcent of Dominant Species that a	are OBL, FAC	W, or FAC (e	excluding FAC-) <20		
Rei	Thistle on	slope	3 10	m-natives.		

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Peviewed) Other No Recorded Data Available			WETLAND HYDROLOGY INDICATORS Primary Indicators:
FIELD OBSE	RVATIONS		Drainage Patterns in Wetlands
Depth of Surface Water	Nore	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	216.0	(in)	Water-Stained Leaves
Depth to Saturated Soil	716.0	(in)	☐ FAC-Neutral Test ☐ Other (Explain in Remarks)

iap Unit Name (9	Series and Phase)	Calleguas clay	loam	Drainage Class: 🚺				
axonomy (Subgr	oup) Typic Ke	vorthents	Field Observations	Confirm Mapped Type?	YES NO			
		PROF	LE DESCRIPTION					
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretion Structure, etc.			
1.16	A	104 53			Clay			
		HYDRIC	SOIL INDICATORS:	_				
Histosol				ions				
Histic Epi	oedon		🗌 High Org	ganic Content in Surfac	e Layer in Sandy Sol			
Sulfidic O	dor		🗌 Organic	Streaking in Sandy Soil	s			
🗋 Aquic Moi	sture Regime		Listed on Local Hydric Soils List					
Reducing	Conditions		Listed o	n National Hydric Soils I	List			
Gleyed or	Low-Chroma Col	ors	🔲 Other (E	Explain in Remarks)				
emarks:	/	· · · · · · · · · · · · · · · · · · ·	/					
Ar h	ndrie	indica	fors a	insat 1	5			
100 1			<i>p</i>		•			
BANNI	34 Ch	975.	,					

. -'

.

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	YES (NO)			~
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?	YES	(NO)
Hydric Soils Present?	YES NO		_	<u> </u>
Remarks		· · · · · · · · · · · · · · · · · · ·		

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u><u> </u></u>	Date May 30, 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck /C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Reparison
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 29

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator		
1 W. How	Et.	FACW	9				
2	Shrub	Tree	10				
3 Scurpis 4/0/			11				
4			12				
5			13				
6			14				
7			15		•		
8			16				
Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	xcluding FAC-) 100				
Solid willow stand. Pit@ end near distuised grading.							

Recorded Data (Descr Siream, Lake, or T Aerial Photographs Other No Recorded Data Av	ide in Remarks) ide Gauge (Feviewed) ailable	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Forift Lines Sediment Deposits	
FIELD OBSERVATIONS			Drainage Patterns in Wetlands
Depth of Surface Water	Nor	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit >16.0 (in)		Water-Stained Leaves Local Soil Survey Data	
Depth to Saturated Soil 716.0 (in)		☐ FAC-Neutral Test ☐ Other (Explain in Remarks)	

p Unit Name (S	eries and Phase)	Botella clay i	oam	Drainage Class: 🕻	
conomy (Subgro	oup) Pachic 1	traixerolls	Field Observations	Confirm Mapped Type	YES NO
		PROFIL	E DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	A	10YR 3/2	SY 2.5/	20/.	Silly CH
			· · · · · · · · · · · · · · · · · · ·		
		HYDRIC	SOIL INDICATORS:		<u> </u>
Histosol	edon		Concret High Or	ions ganic Content in Surfac	e Layer in Sandy Soils
🔲 Sulfidic Oc	lor		🗌 Organic	Streaking in Sandy Soi	ls
🔲 Aquic Mois	sture Regime		Listed o	n Local Hydric Soils Lis	t
	Conditions		Listed o	n National Hydric Soils	List
Gleyed or I	Low-Chroma Cold	ors	🔟 Other (f	Explain in Remarks)	
Dark 1 Dark 1 Dxidi	notizor mottle	streak(in oot chan	ig.		

MINATION

lydrophytic Vegetation Present?	(YES NO	
Vetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland? (YES) NO
lydric Soils Present?	YES NO	
Remarks		

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands		Date May 30, 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck / C. Johnson	_	State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Rearian
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Piot ID 24 30-3/

VEGETATION

	Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
	1 Willow	Tree	FACE	9		
:	2			10 .		
	3			11		
:	4	·····		12		
	5			13		
	6			14		• · • · • • • • • • • • • • • • • • • •
L,	7			15		
	8			16		
1	Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	xcluding FAC-) 100 / .		
	Remarks		······································			······································
·						

HYDROLOGY

i

 Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availation 	e in Remarks) e Gauge Feulewed) Souths outst. able	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits
FIELD OBSER	VATIONS	Drainage Patterns in Wetlands
Depth of Surface Water	0.5 " (in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0 (in)	Water-Stained Leaves
Depth to Saturated Soil	716.0.(in)	FAC-Neutral Test Other (Explain in Remarks)

Map Unit Name (S	eries and Phase)	Botella clay	\oam	Drainage Class: 🕨	
Taxonomy (Subgroup) Pachic Avaixer oils		Field Observations Confirm Mapped Type? (YES NO			
		PROFI	LE DESCRIPTION		
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions Structure, etc.
1-16	<u>A</u>	10YK 3/2		- -	Chay.
	· · · · · · · · · · · · · · · · · · · ·	.			
		· · · · · · · · · · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·
			<u> </u>		
		HYDRIC	SOIL INDICATORS:		1
Histosol Histic Epipe	edon		Concret	iions ganic Content in Surfac Streaking in Sandy Soi	e Layer in Sandy Soils Is
	ture Regime		Listed o	n Local Hydric Soils Lis	 t
	Conditions		🔲 Listed o	n National Hydric Soils	List
Gleyed or L	ow-Chroma Cold	ors	🗌 Other (B	Explain in Remarks)	
Remarks: Moi's	soils.				

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	YES NO		~	
Wetland Hydrology Present?	NO NO	Is this Sampling Point Within a Wetland?	Ē	NO
Hydric Soils Present?	NO NO	·		
Remarks			····-	

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Land!	<u> </u>	Date May 30, 2003
Applicant / Owner County of Orange-	IWMD	County Orange
Investigator R. Beck/C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Reparison
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
ts the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 201 32-33

VEGETATION

O^{dd}

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Willow	Tree	FACU	9		
2 Muletat	shrib	FACW	10		
3 Thistle < 20%			11		
4			12		
5			13		
6			14		
7			15		
8			16 1		
Percent of Dominant Species that a	re OBL, FAC	W, or FAC (e	excluding FAC-) 90 .	·	
Remarks		<u> </u>			·····

 Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs (Reviewed) Other No Recorded Data Available FIELD OBSERVATIONS 			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands
Depth of Surface Water	Nore	(in)	Secondary Indicators (2 or more Required):
Depth to Free Water in Pit	716.0	(in)	Water-Stained Leaves Local Soil Survey Data
Depth to Saturated Soil 7/(C.O (in)		FAC-Neutral Test Other (Explain in Remarks)	

so	۱L	S
----	----	---

xonomy (Subgrou	p) PACHIC .	Argixerolls	Field Observation	s Confirm Mapped Type	? (YES) NO	
		PROFIL	E DESCRIPTION			
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Contrast	Texture, Concretions, Structure, etc.	
1-16	A	104R3/2	<u> </u>		Clayey,	
					-Some loa.	
					347	
					67	
	-					
					· · · · · · · · · · · · · · · · · · ·	
		HYDRIC S	SOIL INDICATORS	:		
🗋 Histosol				tions		
Histic Epiped	don		High O	rganic Content in Surfac	e Layer in Sandy Soils	
🔲 Sulfidic Odo	r		U Organi	c Streaking in Sandy So	ils	
🔲 Aquic Moistu	ure Regime		Listed on Local Hydric Soils List			
, 🔲 Reducing Co	onditions		Listed	on National Hydric Soils	List	
	w-Chroma Cold	ors	🗌 Other (Explain in Remarks)		
Gleyed or Lo					•	

DMINATION

Hydrophytic Vegetation Present?	NO VES NO	
Wetland Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland? (FES) NO
Hydric Soils Present?	NO NO	
Remarks		

ROUTINE WETLAND DETERMINATION

(1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	HI	Date May 30, 2003
Applicant/Owner County of Orange-	IWMD	County Orange
Investigator R. Beck C. Johnson		state California
Do Normal Circumstances exist on the site?	YES NO	Community ID Drainage
ts the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES	Plot ID 34

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Mulerat	Shrillo	FACID	9		
2 Willow	Shrub	FACW	10		
3			11		
4	· · · · · · · · · · · · · · · · · · ·		12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that	are OBL, FAC	W, or FAC (e	excluding FAC-) 780 /.		
Remarks Globern / tor	jon n	ofed	on slopes.		

Recorded Data (Describe Stream, Lake, or Tide Aerial Photographs Other No Recorded Data Availal	in Remarks) Gauge Evilwed) ble	WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Drift Lines Sediment Deposits Drainage Patterns in Wetlands	
Depth of Surface Water	None	Secondary Indicators (2 or more Required):	
Depth to Free Water in Pit 76.0 (in)			Water-Stained Leaves
Depth to Saturated Soil 76.0 (in)			General Test General Test Other (Explain in Remarks)

SOILS

Map Unit Name (S	eries and Phase):	Calleguas ale	<u>ur loan</u>	Drainage Class:	~
laxonomy (Subgro	up) Typic Xer	ertuents	Field Observat	ions Confirm Mapped Type	YES NO
		PRO	FILE DESCRIPTIO	N	
Depth (inches)	Horizon	Matrix Color (Munsell Moist	Mottle Color (Munsell Mois	s Mottle t) Abundance/Contrast	Texture, Concretions, Structure, etc.
1-16	<u> </u>	544/2	• •		Clayey
	· · · · · · · · · · · · · · · · · · ·				
l		LHYDRI		<u> </u>	
Histosol Histic Epipe Sulfidic Odd Aquic Moist Reducing C Gleyed or L	edon or ture Regime conditions ow-Chroma Color	s	Con High Orga Liste Liste	cretions Organic Content in Surfac anic Streaking in Sandy Soi ed on Local Hydric Soils Lis ed on National Hydric Soils I er (Explain in Remarks)	e Layer in Sandy Soils Is t List
Joils from	stonel slope.	that have	pred. F	oossille 5/10	le

Hydrophytic Vegetation Present?	YES NO			
Wetland Hydrology Present?	ES NO	Is this Sampling Point Within a Wetland?	NO NO	
Hydric Soils Present?	KES NO		—	
Remarks		. 4		

ROUTINE WETLAND DETERMINATION (1987 COE Wetlands Delineation Manual)

Project/Site Prima Deshecha Lands	<u>fill</u>	Date May 30, 2003
Applicant/Owner County of Orange-	TWMD	County Orange
Investigator R. Beck/C. Johnson		State California
Do Normal Circumstances exist on the site?	YES NO	Community ID Reparison
Is the site significantly disturbed (Atypical Situation)?	YES NO	Transect ID
Is the area a potential Problem Area? (If needed, explain on reverse)	YES NO	Plot ID 229 35

VEGETATION

Dominant Plant Species	Stratum	Indicator	Dominant Plant Species	Stratum	Indicator
1 Muletat	shrub	FACW	9		
2 Saltarass KID'l.	herb	OBL	10 ·		
3			11		
4			12		
5			13		
6			14		
7			15		
8			16		
Percent of Dominant Species that a	ire OBL, FAC	W, or FAC (e	xcluding FAC-)		
Remarks Small mule	fat si	hrubs	in Arninag	€.	

Recorded Data (Describe in Remarks) Stream, Lake, or Tide Gauge Aerial Photographs <i>Feviewed</i> Other No Recorded Data Available			WETLAND HYDROLOGY INDICATORS Primary Indicators: Inundated Saturated in Upper 12 Inches Water Marks Sediment Deposits
FIELD OBSER	VATIONS		Drainage Patterns in Wetlands
Depth of Surface Water	Depth of Surface Water Nore (in)		Secondary Indicators (2 or more Required):
Depth to Free Water in Pit $7/6.0$ (in)		Water-Stained Leaves Local Soil Survey Data	
Depth to Saturated Soll	716.0	(in)	FAC-Neutral Test Other (Explain in Remarks)

SOILS

Map Unit Name (S	eries and Phase):	Calleguas clay	lean	Drainage Class:	WD		
Taxonomy (Subgro	OUP) Typic Xer	orthents	Field Observations	s Confirm Mapped T	ype? (YES)NO		
PROFIL			LE DESCRIPTION				
Depth (inches)	Horizon	Matrix Color (Munsell Moist)	Mottle Colors (Munsell Moist)	Mottle Abundance/Cont	Texture, Concretions, rast Structure, etc.		
1-16	A	54 4/2	7.54R6	3 10-20	1. Clayey		
					/ /		
		·					
,	· · ·· · · · · · · · · · · · · · · · ·						
	· · · · ·						
		HYDRIC	SOIL INDICATORS				
🔲 Histosol				tions			
Histic Epip	edon		High Or	ganic Content in Su	rface Layer in Sandy Solls		
🗋 Sulfidic Od	for		🗌 Organic	Streaking in Sandy	' Soils		
Aquic Mois	sture Regime		Listed on Local Hydric Soils List				
Reducing (Conditions		Listed on National Hydric Soils List				
Gleyed or l	Gleyed or Low-Chroma Colors			Explain in Remarks)			
Remarks: Mottles present. Clayey, moist soils within invert.							

WETLAND DETERMINATION

Hydrophytic Vegetation Present?	KES NO	
Wetiand Hydrology Present?	YES NO	Is this Sampling Point Within a Wetland?
Hydric Soils Present?	MED NO	
Remarks		

B) Additional Site Photographs

BALL NAV

虞

Statistical State

ų





Street,









(Access)





BURNESS.











Summe

;



Summer



in sum




Station of



Monor

C) Vegetation Map

۰.,

And a state of

the strength

₩00 J.N

1. N. 1. W. W.

and the first

i



APPENDIX G

Pre-Mitigation Plan



Project Boundary Project Boundary Area of Disturbance - 1069.36 Acres Area of Disturbance - 1069.36 Acres Grading Contours Grading Pre-Mitigation (Phases 2, 92 Acres Existing Native Grass - 15.3 Acres Proposed Pre-Mitigation (Phases 2 & 3) Coastal Sage Scrub - 108.14 Acres Native Grass - 15.3 Acres Native Grass - 15.3 Acres Riparian mfs - 6.0 Acres Native Grass - 15.3 Acres Proposed Pre-Mitigation (Phases 2 & 3) Other Areas Mathematic Existing Landside Remediation Mitigation La Pala Ave. Feasibility Study Area Mathematic Mathematic

D.Projects/OCIW/MD/J002/premma-Jimery/Ex_onsile_premit_065005.mxd





APPENDIX H

Regional Environmental Enhancement Opportunities Plan (REEOP)

Regional Environmental Enhancement Opportunities Program (REEOP):

The Regional Environmental Enhancement Opportunities Plan (REEOP) identifies restoration, enhancement, and creation opportunities on the PDL property that can be made available to satisfy potential mitigation requirements for (non-PDL) public and private projects in the region. The REEOP document will serve to identify these opportunities in conjunction with state and federal resource agency permit authorizations. SEIR 597 provides the CEQA documentation that would allow other public agencies or private landowners to use areas within Prima identified by the REEOP at the time of landfill closure for restoration mitigation purposes, if IWMD authorizes this activity and the entity has received the required regulatory permits needed to authorize impacts to resources under jurisdiction of the regulatory agencies.

The area available for regional enhancement opportunities can be found both outside the ultimate limits of active landfilling operations, as well as within completed areas of phased landfill development (Exhibit H-1, Environmental Opportunities). Consistent with the planning approach used for the Pre-Mitigation Plan, the conceptual REEOP design has been developed to accommodate the potential future expansion of Camino de los Mares through the southwestern portion of the project site. This plan primarily provides for native grassland and coastal sage scrub opportunities based on existing site soils and hydrology. Please note that once a final alignment for the extension of La Pata Avenue is approved, the REEOP will be revised accordingly to incorporate the refined rights-of-way and to identify additional mitigation area opportunities.

The conceptual design of the Regional Enhancement Opportunities Plan is based on the following goals:

- Select areas that are currently disturbed or contain non-native plant species. Enhancement of these areas will involve the overall net habitat values of the area.
- Select potential restoration opportunity areas within the landfill property *limits.* Selection will be based on site constraints such as landfill final cover protection, recreation uses identified by the 2001 GDP, and long-term operation and maintenance including landslide remediation/stabilization.
- Select appropriate habitat restoration types. Select appropriate habitat restoration types based on soils, slope, aspect, hydrology, and other site conditions.
- Incorporate viewshed protection elements. Incorporate viewshed protection requirements from City/County MOUs and agreements with adjacent landowners.
- Maintain consistency with the commitments, reservations and restrictions

contained in the agreement between the County and the Rancho Mission Viejo Company. Ensure that use of the restoration areas and opportunities identified in the REEOP upholds the Right of First Refusal (after County requirements are met) for the Rancho Mission Viejo Company within the Burdened Property.

